

Litigation and Regulatory 5055 North Point Pkwy Alpharetta, GA 30022

November 13, 2008

### Transmittal Letter No. 08-01

#### VIA E-FILING

Ms. Beth Salak, Director Division of Competitive Markets and Enforcement Florida Public Service Commission Attn: Tariff Section 2540 Shumard Oak Boulevard Tallahassee, Florida 32399-0850

RE: <u>Verizon Business Services: Tariff No. 6</u>
Introduce Tariff Dedicated to Large Business and Enterprise Customers

Dear Ms. Salak:

MCI Communication Services, Inc. d/b/a Verizon Business Services ("Verizon Business") hereby files with your office an original of its new interexchange FL PSC Tariff No. 6 for services dedicated to large business and enterprise customers. These services were previously in Verizon Business FL PSC Tariff No. 3 and FL PSC Tariff No. 5. Tariff No. 3 and Tariff No. 5 have been revised in Transmittal Letter No. 08-11 and 08-07, respectively, to include only residential and small business ("Mass Market") services.

Verizon Business proposes these changes and requests an effective date of November 14, 2008, so as to have a single tariff dedicated to interexchange services provided to large business and enterprise customers with terms, conditions, and product descriptions similar to those in other states. These customers are typically multi-state or multinational customers who desire as much uniformity as possible for the services they obtain from Verizon Business regardless of where service is provisioned.

Letter to Ms. Beth Salak November 13, 2008 Page 2

If you have any questions regarding this filing, please contact me either at (888) 215-5680 or sandy.chandler@verizonbusiness.com.

Respectively submitted,

Tariff Manager Verizon Business

Enclosure

## INTEREXCHANGE TELECOMMUNICATIONS SERVICES TARIFF FOR MCI COMMUNICATIONS SERVICES, INC. d/b/a VERIZON BUSINESS SERVICES

# **APPLICABILITY**

This tariff contains the rules, regulations, descriptions and rates applicable to the furnishing of interexchange telecommunications services offered by Verizon Business Services (hereinafter referred to as the "Company") within the State of Florida to large business and enterprise (i.e., non-mass market) customers. Interexchange telecommunications services offered to residential and small business customers are not contained in this tariff, but instead can be found in Company tariff Nos. 3 and 5.

The title page and pages 1-567 inclusive of this tariff are effective as of the date shown.

<u>Page</u> 1 2	Revision Original
3	Original
4	Original Original
5	Original
6	Original
7	Original
8	Original
9	Original
10	Original
11	Original
12	Original
13	Original
14	Original
15	Original
16	Original
17	Original
18	Original
19	Original
20	Original
21	Original
22	Original
23	Original
24	Original
25 26	Original
27	Original Original
28	Original
29	Original
30	Original
31	Original
32	Original
33	Original
34	Original
35	Original
36	Original
37	Original
38	Original
39	Original
40	Original

<sup>\*</sup> New or Revised Page

78 Original 79 Original 80 Original	Page 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75 76 77	Revision Original
	75 76 77 78 79	Original Original Original Original Original

<sup>\*</sup> New or Revised Page

<u>Page</u>	Revision
286	Original
287	Original
288	Original
289	Original
290	Original
291	Original
292	Original
293	Original
294	Original
295	Original
296	Original
297	Original
298	Original
299	Original
300	Original
301	Original
302	Original
303	Original
304	Original
305	Original
306	Original
307	Original
308	_
309	Original
	Original
310	Original
311	Original
312	Original
313	Original
314	Original
315	Original
316	Original
317	Original
318	Original
319	Original
320	Original
321	Original
322	Original
323	Original
324	Original
325	Original
326	Original
327	Original
328	Original
329	Original
330	Original
	_

<sup>\*</sup> New or Revised Page

Page 331 332 333 334 335 336 337 338 339 340 341 342 343 344 345 346 347 348 349 350 351 352 353 354 355 356 357 358 359 360 361 362 363 364 365 367 368 369 370 371 372 373 374 375	Revision Original

<sup>\*</sup> New or Revised Page

<sup>\*</sup> New or Revised Page

Page 421 422 423 424 425 426 427 428 429 430 431 432 433 434 435 436 437 438 439 440 441 442 443 444 445 446 447 448 449 450 451 452 453	Revision Original
	-
	-
454	Original
455 456	Original
457	Original Original
458	Original
459	Original
460 461	Original
462	Original Original
463	Original
464	Original
465	Original

<sup>\*</sup> New or Revised Page

Page 466 467 468 469 470 471 472 473 474 475 476 477 478 479 480 481 482 483 484 485 486 487 488 489 490 491 491 492 493 494 495 496 497 498 499 500 501 502 503 504 505 506 507 508	Revision Original
506	Original
509	Original
510	Original

<sup>\*</sup> New or Revised Page

Page 511 512 513 514 515 516 517 518 519 520 521 522 523 524 525 526 527 528 529 530 531 532 533 534 535 536 537 538 539 540 541 542 543 544 545 546 547	Revision Original
545	Original Original
	3

<sup>\*</sup> New or Revised Page

<u>Page</u> 551 552	Revision Original Original
553	Original
554	Original
555	Original
556	Original
557	Original
558	Original
559	Original
560	Original
561	Original
562	Original
563	Original
564	Original
565	Original
566	Original
567	Original

<sup>\*</sup> New or Revised Page

# GENERAL TERMS AND CONDITIONS OF SERVICE FOR TELECOMMUNICATIONS SERVICES

# **TABLE OF CONTENTS**

		<u> </u>	5405
l.	Scop	pe and Application	<u>PAGE</u> 19
	1. 2. 3. 4.	Scope Order of Precedence Service Provider General Definitions	19 19 19 20
II.	Term	Terms and Conditions	
	1.	Services A. Use of Service B. Service Limitations C. General Provisions of Service	24 24 29 31
	2.	Term A. Term B. Service Cancellation or Change by Customer C. Service Cancellation by Company D. Notice E. Service Transitioning	32 32 32 33 36 36
	3.	Minimum Annual Volume Commitment, Underutilization and Early Termination Charges	37
	4.	Rates, Charges and Payment Terms  A. Payment Terms  B. Service Credits  C. Fraud	37 37 42 44
	5.	Termination	47
	6.	Disclaimer of Warranties and Limitation of Liability A. Warranty Disclaimer B. Liability Disclaimer C. Liability Limitation D. Limitation of Action E. Indemnification F. Third Party Acts G. Dangerous Conditions H. Premises Liability I. Risk-Based Pricing	47 47 47 48 48 48 48 49
7.	Additional Terms and Conditions for Toll-Free Service 49		49

# GENERAL TERMS AND CONDITIONS OF SERVICE FOR TELECOMMUNICATIONS SERVICES

# TABLE OF CONTENTS (CONT'D.)

	<ol> <li>8.</li> <li>9.</li> </ol>	Miscellaneous Provisions  A. Intentionally Omitted.  B. Confidential Information  C. CPNI Consent  D. Protection of Customer CPNI  E. Customer Data Privacy  F. Assignment  G. Supplemental Customer Obligations Pertaining to Customer-obtained Facilities; Security; Site Access; and Customer Information  H. Software and Documentation  I. Company Marks  J. Other Provisions  .01 Intentionally Omitted .02 No Waiver .03 Severability .04 Dispute Resolution .05 Compliance with Laws .06 Service Order .07 Purchase Order .08 Independent Contractor Relationship; No Agency .09 No Third-Party Beneficiaries .10 Interpretation .11 Headings .12 Signatures  Telecommunications Service Priority (TSP)	PAGE 51 51 51 52 54 55 55 56 57 57 57 57 58 58 58 58 59 59 59 59 60
	9. 10.	TSP Special Construction	63
III.	Other ( 1. 2. 3. 4.	Charges and Surcharges Carrier Access Charge Payphone Use Surcharge Special Access Surcharge Customer Support Service Charges A. No Fault Found Dispatch Charge B. Reconnection Charge	65 65 66 66 66 66
IV.	Fund		68
V.	Individ	lual Case Basis Agreements	68
VI.	Intrasta A. B. C.	ate Long Distance Services Product Description Definitions Features and Options	69 69 70 70

# GENERAL TERMS AND CONDITIONS OF SERVICE FOR TELECOMMUNICATIONS SERVICES

# TABLE OF CONTENTS (CONT'D.)

VII.	<ol> <li>Rates and Charges</li> <li>Intrastate Access</li> <li>Usage</li> <li>Intrastate Operator Assisted</li> <li>Intrastate Features and Feature Packages</li> <li>Optional Calling Plans         <ul> <li>Verizon Business Services III – Optional Calling Plan</li> <li>Verizon Business Services I – Optional Calling Plan</li> <li>Verizon Business Services I – Optional Calling Plan</li> <li>Intrastate Long Distance Directory Assistance</li> </ul> </li> </ol>	PAGE 72 72 75 76 78 78 78 89 102 112
VIII.	Intrastate Network Services Local Access Service  1. Service Description 1.1 Intrastate Analog Local Access 1.2 Intrastate DSO (Hubless) Access 1.3 Intrastate T-1 Digital Access 1.4 Intrastate DS-3 Local Access 1.5 Intrastate SONET Access 2. Special Pricing Plans and Credits 3. Other Non-Recurring Charges	113 113 113 126 150 176 201 202 209
IX.	Intrastate U.S. Private Line Services  1. Service Description  2. Definitions  3. Features and Options  4. Rates & Charges  5. Terms and Conditions  6. Service Level Agreements	212 212 213 214 214 236 237
X.	State Specific Plans and Programs	
XI.	Currently Offered Promotions	
XII.	Expired Promotions	
XIII.	Grandfathered Services	

72

#### SCOPE AND APPLICATION

- 1. <u>Scope</u>: This tariff applies to intrastate interexchange telecommunications service. The various services, plans, programs, rates and charges (individually or collectively, "Provisions") set forth below may include some non-regulated Provisions. To the extent non-regulated Provisions are indicated, these are only for a matter of convenience and do not mean that such non-regulated Provisions are subject to the jurisdiction of this Commission.
- 2. <u>Order of Precedence</u>: In the event of an inconsistency between a General Term and a service-specific term or condition in this tariff, the service-specific tariff term or condition shall prevail.
- 3. <u>Service Provider</u>: Services are provided by MCI Communications Services, Inc., d/b/a Verizon Business Services.

## 4. General Definitions.

These General Definitions also apply to capitalized terms contained in written contracts between Company and Customer ("Signed Contract") which incorporate this tariff by reference and do not otherwise define those capitalized terms in the written contract.

<u>Access</u>: A means by which Company service is provided to a Customer. Access may be "Dedicated," in which case it is available to the Customer on a full-time, unshared, basis, or it may be "Switched," in which case it is available to the Customer and others on a usage, shared, basis.

<u>Access Coordination</u>: The design, ordering, installation coordination, pre-service testing, service turn-up, maintenance and ongoing coordination of testing and trouble resolution for all Company-provided Local Access Channels. In the case of customer-provided Local Access Channels, the maintenance, ongoing coordination of testing and trouble resolution.

Address of Record: For purposes of the CPNI rules, a Company address of record is any address associated with the account for at least 30 days in a Company system such that Company can immediately notify the customer if the address of record changes, such as a billing address and a contract notice address. So long as Company does not have the ability to immediately notify customers if service addresses changes, a service address is not an address of record.

<u>Affinity Group</u>: An association or group (not organized for the purpose of qualifying for service discounts) with common business or other interests that agrees to market Company service(s).

<u>Administrative Change</u>: The modification of an existing circuit or a dedicated access line or port at the request of the Customer that involves changes to authorization codes, speed numbers, route guide, consolidation of billing, verification of testing performed by parties other than the Company, or any other administrative change not covered by a billing record change.

# 4. <u>General Definitions (Cont'd)</u>

<u>Alternate Access</u>: A form of Local Access in which the provider is an entity, other than the Local Exchange Carrier, authorized or permitted to provide such service.

ASR or Access Service Request: An order placed with a Local Access provider for Local Access.

Answer Supervision: The return of a signal to the calling party disclosing that a call was answered.

<u>Application for Service</u>: A standard order or enrollment form that includes all pertinent billing, technical and other descriptive information needed by the Company to provide service.

<u>Authorized User</u>: A person, firm, corporation or other entity that either is authorized by the Customer to receive or send communications or is placed in a position by the Customer, either through acts or omissions, to send or receive communications.

<u>Bandwidth</u>: The total frequency band, in hertz, allocated for a channel.

<u>Called Station</u>: The station called, or the terminating point of a call.

Calling Station: The station at which a call is originates.

Channel or Circuit: A communications path between two or more points.

<u>Channel Termination:</u> The point at which the Company's channel originates, terminates, or drops for the insertion or removal of a Customer's signal.

<u>Collect Call</u>: A call for which applicable charges are billed to the telephone number of the called station of the Customer.

<u>Commercial Service</u>: A switched network service involving dial station originations for which the Customer pays a rate that is described as a business or commercial rate in the applicable local exchange service tariff for switched service.

<u>Company</u>: Verizon Business Services, or any affiliated entity designated as the provider of any service offered in the tariff.

<u>Customer</u>: The person, firm, corporation or other entity which orders service and is responsible for the payment of all charges for service and for compliance with Company contract and tariff requirements. The term "customer" also includes a person, firm, corporation or other entity that either knowingly or unknowingly accesses service and completes a communication over the Company's network. For Resp Org Service, the Customer is the person, firm, corporation or other entity that selects or is directed to select the Company as the Responsible Organization (Resp Org) for a toll-free telephone number. For purposes of SMS Resp Org Changes, the customer is the person, firm, corporation, or other entity that submits the change request.

<u>Customer Premises</u>: A space owned, leased or otherwise under the control of Customer or Authorized User at which Service is provided (including without limitation locations of interconnection, termination and origination).

## 4. General Definitions (Cont'd)

<u>Dedicated Access/Termination</u>: An access line service consisting of a continuously connected circuit between a Customer Premises or serving telephone company central office and a Company terminal, available to the Customer on a full-time, unshared, basis, which is used for the origination or termination of services.

<u>Due Date</u>: The date on which payment for service by the Customer is due.

End-to-End: Customer Premise to Customer Premise.

<u>Exemption Certificate</u>: A written notification provided by a Customer certifying that its dedicated facility should be exempted from the monthly Special Access Surcharge because: (a) the facility terminates in a device not capable of interconnecting service with the local exchange network; or (b) the facility is associated with a Switched Access Service that is subject to Carrier Common Line Charges.

<u>Expedite</u>: A Service Order that is processed at the request of the Customer in a time period shorter than the Company standard Service interval.

Gbps: Billions of bits per second.

Guide: The Service Publication and Price Guide, found on-line at www.verizonbusiness.com/guide.

<u>ICO (Independent Company)</u>: An ILEC (Incumbent Local Exchange Carrier) not affiliated with an RBOC (Regional Bell Operating Company) which provides exchange service in areas not served by an RBOC.

<u>ILEC (Incumbent Local Exchange Carrier)</u>: The dominant phone carrier providing exchange service within a geographic area as determined by the FCC.

<u>Installation</u>: The provision of connections for new or additional service.

<u>Institutional Phones</u>: Telephones, other than payphones, located in public institutions such as universities, prisons, or public offices, or in hotels or motels, or in other premises where the Customer may not be able to control access to the phones.

<u>Interexchange (IXC) Service</u>: The portion of a Channel or Circuit between a Company designated Point-of-Presence in one exchange and a Company designated Point-of-Presence in another exchange.

<u>Interruption</u>: A condition that arises when service or a portion thereof is inoperative.

<u>Invoice</u>: A document or other communique that discloses the Company's charges for service furnished to the Customer.

kbps: Kilobits per second.

<u>LATA (Local Access Transport Area)</u>: A geographic area established by law and regulation for the provision and administration of telecommunications services.

<u>Local Access Channel</u>: The connection between a Customer Premises and a Company Point-of-Presence.

<u>Local Access</u>: The portion of service between a Customer Premises and a Company designated Point-of-Presence.

## 4. General Definitions (Cont'd)

Local Exchange Carrier (LEC): A company which furnishes exchange telephone service.

Locations: Places from or at which service is furnished.

Mbps: Megabits per second

MCI Legacy Company: An affiliate of Verizon that was an affiliate of MCI, Inc. prior to the acquisition of MCI, Inc. by Verizon Communications Inc., including one or more of the following entities (without limitation): MCI Communications Services, Inc. d/b/a Verizon Business Services; MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services of Virginia, Inc. d/b/a Verizon Access Transmission Services of Virginia; and MCImetro Access Transmission Services of Massachusetts, Inc. d/b/a Verizon Access Transmission Services of Massachusetts, and applicable affiliated operating companies outside the United States.

Mileage or Mileage Band: The airline mileage between two cities can be calculated using the Vertical (V) and Horizontal (H) Coordinates of the serving wire centers associated with the Company's Terminal Locations using the following formula

$$\sqrt{((V_1-V_2)^2+(H_1-H_2)^2)/10}$$

where  $V_1$  and  $H_1$  correspond to the V & H coordinates of City 1 and  $V_2$  and  $H_2$  correspond to the V & H coordinates of City 2.

Example:		<u>H</u>
City 1 - New York City 2 - Chicago	4997 5986	1406 3426
	$V_1$ $V_2$ $H_1$ $H_2$ $\sqrt{((4997-5986)^2+(1406-3426)^2)/10}$	
	√(505852.1)=711.2	

Airline Mileage = 712 miles (Result will always be rounded to the next highest mile).

<u>Multiline Terminating Device</u>: Switching equipment, key telephone type systems or other similar customer premises terminating equipment which is capable of terminating more than one access line.

Operator Station: Service that requires the assistance of an operator to complete a call.

<u>Person-to-Person</u>: Service for which the person originating a call specifies to an operator a particular person, mobile station, department, extension, or office to be reached.

Physical Change: A service modification that requires some physical change of service.

<u>Point-of-Presence (POP)</u>: A location at which the Company maintains a Terminal Location for purposes of providing service.

<u>Promotion</u>: Periodic financial inducement offered by the Company to new and/or existing Customers of service to subscribe to and use new or additional service(s).

# 4. <u>General Definitions (Cont'd)</u>

Rate: The price per unit of service.

Rate Center: A specified geographical location used for determining mileage measurements.

RBOC (Regional Bell Operating Company): One of the seven LECs established in the U.S. Department of Justice 1984 Consent Decree with AT&T. The RBOC carriers are Ameritech, Verizon (NYNEX) or Verizon North, Verizon (Bell Atlantic) or Verizon South, Bell South, Pacific Bell (PacBell), Southwestern Bell and US West (Qwest).

<u>Redundancy</u>: An offering of alternate service through the use of one or more different routings, circuits, and/or additional equipment

Requested Service Date: The date requested by the Customer for the commencement of service and agreed to by the Company.

Restoration: The re-establishment of service.

Route Diversity: Two channels furnished partially or entirely over two physically separate routes.

<u>Service</u>: Any Company product or service provided to, or obtained by, the Customer, any Authorized User, or a third party.

<u>Serving Wire Center</u>: The physical location within a Local Exchange Carrier's central office or other service provider's facility used to determine mileage sensitive rates. A serving wire center is associated with each Customer location and each Company location.

#### II. TERMS AND CONDITIONS

## 1. <u>SERVICES</u>

# 1. A <u>Use of Service</u>

- .01 Customer and Authorized Users: Service may be used for Customer or Authorized User communications. Customer is responsible for Authorized User's use of Service and for Authorized User's compliance with the requirements of this tariff. A resale Customer must, without limitation: (a) transact/interact with its customers without reference to the Company or any of its affiliates, unless the Company agrees otherwise in writing; (b) either acquire its own Carrier Identification Code(s) (CIC(s)) or indemnify the Company for all costs incurred by the Company as a result of the Customer's use of a Company CIC; and (c) comply with the registration requirements contained in Section 64.1195 of the FCC's Rules and Regulations. Resale Customers may purchase and use Service only as explicitly authorized by Company and subject to any limitations imposed by the underlying suppliers involved in providing Service.
- .02 To the extent Customer and Company allow another entity to purchase Service, Customer remains responsible for such entity's use of the Service, both in compliance with the applicable terms and for payment of applicable charges. In such arrangements, Company has no obligation to deal with such entities and may choose to interact solely with Customer for all purposes.
- .03 Intentionally left blank.
- .04 Automatic Number Identification: Any Customer of Automatic Number Identification (ANI) or charge number services on calls may:
  - .041 use the telephone number and billing information for billing and collection, routing, screening, and completion of the originating telephone subscriber's call or transaction, or for services directly related to the originating telephone subscriber's call or transaction:

## 1. <u>SERVICES (CONT'D.)</u>

# 1. A <u>Use of Service</u> (Cont'd.)

- .042 not reuse or sell the telephone number or billing information without first notifying the originating telephone subscriber and obtaining the affirmative consent of such subscriber for such reuse or sale; and
- .043 not disclose, except as expressly permitted, any information derived from the Automatic Number Identification, or charge number service for any purpose other than performing the services or transactions that are the subject of the originating telephone subscriber's call; for ensuring network performance security and the effectiveness of call delivery; for compiling, using, and disclosing aggregate information; and for complying with applicable law or legal process.

These requirements are not intended to prevent a person to whom Automatic Number Identification or charge number services are provided from using the telephone number and billing information provided pursuant to such service, and any information derived from the Automatic Number Identification or charge number services, or from the analysis of the characteristics of a telecommunications transmission, to offer a product or service that is directly related to the products or services previously acquired by that Customer from such person. Use of such information is subject to the requirements of 47 CFR Sec. 64.1200 and Sec. 64.1504(c).

- .05 Calling Party Number: A Customer's calling party number (CPN) may be identified and disclosed to the called party. This may be prevented if the Customer first dials \*67 (or 1167 for rotary or pulse-dialing phones) to place a call. Customers with per-line blocking first must dial \*82 (or 1182 for rotary or pulse-dialing phones) to allow for the presentation of CPN to called parties.
- .06 Interconnection with Another Carrier: Services may be interconnected with service(s) or facilities provided by another common carrier pursuant to the following conditions:
  - .061 Interconnection may take place at a Company premise, at the premise of another common carrier, or at the premise of the Customer or its Authorized User. Unless otherwise indicated, Service is not part of a joint undertaking between the Company and any other common carrier or carriers.
  - Any special facilities needed to achieve compatibility between the Company's Service and the service or facilities of another common carrier will be provided at the Customer's expense. At the Customers request and upon appropriate authorization by the Customer, the Company will undertake to make the necessary interconnection arrangements.
  - .063 When Services of Company are interconnected with and/or terminated in any service and/or equipment of another common carrier, Customer shall comply with any applicable tariff regulations of and/or contractual obligations it has to the other common carrier.

## 1. <u>SERVICES (CONT'D.)</u>

# 1. A <u>Use of Service</u> (Cont'd.)

- .07 Company is appointed Customer's agent to arrange interconnection from the Company's point of presence (POP) to the Customer's facilities, where necessary for Company to provide Service, unless otherwise specified. Customer is responsible for payment of local access line charges for such interconnections secured on its behalf. The rates charged for local access service are subject to change by the providers of such local access service. Customer acknowledges that Company may rely on other carriers for installation and testing of local access lines. Company is not liable for services and equipment not provided by the Company.
- .08 Interconnection with Customer–Provided Systems or Facilities: Services may be interconnected with Customer-provided systems or facilities (including equipment) provided by the Customer, an Authorized User, or their representative(s) pursuant to the following conditions:
  - .081 Customer is responsible for the installation, operation and maintenance of Customer Equipment (including without limitation Terminal Equipment such as teleprinters, handsets or data sets), situated at a Customer Premises, at Customer's expense, except as otherwise provided for in the Customer's Signed Contract; and
  - the characteristics and performance of Customer Equipment to be interconnected with Service provided by the Company must be and remain compliant with requirements imposed by the Federal Communications Commission, state commissions, and/or industry standards, as these may be modified from time to time.
  - .083 Company is not responsible for the performance of Customer Equipment, including without limitation the through transmission of signals generated by Customer Equipment or for the quality of, or defects in, such transmission; the reception of signals by Customer Equipment; or network control signaling where such signaling is performed by network control signaling Customer Equipment.

- II. TERMS AND CONDITIONS (CONT'D.)
  - 1. <u>SERVICES (CONT'D.)</u>
  - 1. A <u>Use of Service</u> (Cont'd.)
    - .09 Customer Interconnection Responsibilities:
      - .091 Customer is responsible for taking all steps necessary to interconnect Service with Terminal Equipment, communications systems, or other facilities of Customer or any third party acting as its agent (collectively "Customer Equipment" at Customer Premises, including the payment of associated interconnection costs and those associated with Customer personnel and other service providers, the securing of rights-of-way and conduit, and the furnishing of electrical power (including the agreed-upon AC or DC current type), heating, ventilating and cooling. Additional required Customer undertakings include, but are not limited to, obtaining, installing and maintaining all equipment, materials and supplies necessary to interconnect the Customer Equipment, to Service, as well as fulfillment of the following: (a) securing all licenses, permits, and other arrangements necessary for interconnection; (b) ensuring that Customer Equipment are properly interfaced with Service and that emitted signals: (i) are of the proper mode, bandwidth, power, data speed and signal level for the intended use of the Customer; (ii) are fully compliant with the minimum protective criteria generally accepted in the telephone industry, any criteria set forth in this tariff relating to Terminal Equipment, and other appropriate criteria as may be prescribed by the Company; and (iii) do not damage Company facilities, injure its personnel or degrade Service to other customers. If the Federal Communications Commission or some other appropriate certifying body certifies particular Customer Equipment as being technically acceptable for direct electrical connection with Service, the Company will permit such equipment to be connected with Service without the use of protective interface devices. If the Customer (or its agent, contractor, or Authorized User) fails to maintain and operate its Customer Equipment properly, with resulting imminent harm to the Company's network, personnel or other customer Services, the Company may, upon written notice, require repair, maintenance, or the use of protective interface equipment at the Customer's expense. If this action fails to produce satisfactory quality and safety results, the Company, immediately upon written notice, may terminate the Customer's Service without liability. During any such period of suspension, any credit allowance for service interruptions does not apply. Customer shall reimburse Company for damage to its facilities caused by the negligence or willful act of Customer in using Customer Equipment or a Service. Any costs associated with meeting these obligations, including costs of altering the structure to permit installation of the Company-provided facilities. shall be borne entirely by, or may be charged by the Company to, the Customer. Company may require Customer to demonstrate its compliance with this section prior to accepting an Application for Service.
      - .092 Customer must provide access to and release of Company equipment on Customer Premises as reasonably requested for maintenance as well as testing and repair related to failures of equipment or Service. Additional charges may apply when maintenance, testing and/or repair is performed outside of regular business hours at Customer's request.

- 1. <u>SERVICES (CONT'D.)</u>
- 1. A <u>Use of Service</u> (Cont'd.)
  - .093 Customer must identify, monitor, remove and dispose of any hazardous material (e.g. friable asbestos) prior to any Company work at that location. If, in the Company's opinion, a Service location poses risks such that injury or damage to Company's employees or property might result from work being performed by Company, Customer is responsible for that work, including without limitation, installing and maintaining Company facilities and equipment (at Company direction).
  - .094 Customer is responsible for ensuring that no one other than qualified and authorized Customer personnel attempt to adjust, modify, move or otherwise interfere in any way with the continuous operation of Company equipment located at Customer Premises.
  - .095 Customer is responsible for all loss regardless of cause (other than directly resulting from an act or omission of the Company) to Company equipment on Customer Premises.
  - .10 Access to Company Systems: When a Customer is permitted access to Company computer systems and data ("Systems") for purposes of managing and maintaining service, the Customer must:
    - .101 access Systems only to the extent necessary to manage and maintain service, as access into Systems beyond that authorized may result in civil and/or criminal penalties:
    - .102 not disclose or use information, which may be learned as a consequence of access to Systems, except as may be necessary to manage and maintain service. Customers must take all reasonable measures to prevent those not having a need to know from acquiring such information;
    - .103 not in any manner disclose, provide, or otherwise make available, in whole or in part, Systems documentation or any related or other confidential material except to those having a need to know as an incident of the operation of the Customer's Service. Systems remains the property of the Company and may not be copied, reproduced or otherwise disseminated without the prior written consent of the Company;
    - .104 take all other reasonable measures to maintain the confidentiality of Systems. Such measures shall include the use of Personal Identification Numbers (PINs) and passwords selected by and known only to the Customer's individual authorized users. Systems telephone numbers and dial-up access numbers assigned to the Customer by the Company and PINs or any aspect of access and sign-on methodology revealed to the Customer shall not be posted or shared with others under any circumstances. Customers shall follow normal logoff procedures prior to leaving a terminal unattended and are required to report to the Company any known or suspected attempt by others to obtain unauthorized access to Systems; and

# 1. <u>SERVICES (CONT'D.)</u>

# 1. A <u>Use of Service</u> (Cont'd.)

.105 immediately notify the Company if a security device assigned to a Customer for dial-up access is lost, stolen, or misplaced.

#### 1. B Service Limitations

- .01 No Unlawful Use: Service may not be used for: (a) an unlawful purpose; or (b) making calls that employ automatic dialing devices and terminate into electronic Information Services, pay-per-call services, or other domestic or international audiotext services.
- .02 Facilities Used for Service: The facilities used to provide Service will be exclusively of the Company's choosing. The Company may at any time substitute facilities used to provide Service, or it may substitute comparable Service for the Service being provided to Customer. In no event will title to any of the facilities used by the Company to furnish Service vest in the Customer or in an Authorized User. At Customer's request and cost, Company will provide alternative or redundant Service during facility or service substitutions, where commercially reasonable. Customer will continue to pay for any such additional Service provided by Company until cancelled by Customer with at least thirty days written notice.
- .03 Availability of Facilities and Authority: Service is offered and furnished subject to the availability of facilities and authority required to provide it. In particular and without limitation, Company may stop providing a service or product without liability if (a) a third party that had been providing an underlying component of that service or product stops providing it to Company, (b) a material portion of the facilities used to provide Service is rendered unavailable by condemnation or casualty, or (c) an applicable law or regulation would require the alteration of the Service or be violated by its provision. Notwithstanding any other provision in this tariff, the Company may: (a) deny Customer Service requests and thereby limit or allocate Service among Customers; or (b) limit or allocate the facilities available for Service, if necessary, in order to: (i) manage its network in an efficient manner; (ii) meet reasonable Customer Service expectations; or (iii) furnish Service to existing and future customers based on current and projected available capacity, taking into account forecasted Customer requirements. The Company may terminate Service or available billing options associated with Service upon written notice to affected Customers whenever necessitated by circumstances beyond the Company's control, including situations involving Customer call volumes or patterns that result, or could result, in Company network blockages or the degradation of Service furnished to the Customer or to other customers. Company also reserves the right to abandon Service altogether, with appropriate notice, without liability.
- .04 Customer has no property right to any telephone number or any other designation associated with Services furnished by the Company. Company reserves the right to change such numbers or other designations, or the meaning associated with them, at its discretion.
- .05 Call Blocking: Without notice, the Company may block calls to or from certain countries, country codes, cities, city codes, NXX exchanges, individual telephone stations, groups or ranges of individual telephone stations or calls that use certain authorization codes,

- 1. <u>SERVICES (CONT'D.)</u>
- 1. B <u>Service Limitations</u> (Cont'd.)

whenever the Company deems it necessary to take such action to prevent: (a) the unlawful use of Service; (b) fraud; (c) nonpayment for Service; (d) the use of Service in violation of the requirements of this tariff; or (e) Company network blockage or the degradation of service furnished to the Customer or to other customers; or when actions taken by foreign governments or foreign telecommunications agencies, or responsive actions taken by the U. S. Government or state government or any instrumentality thereof, render it impossible or impracticable to provide service. The Company will unblock as soon as it determines it can do so without undue risk, and it will, upon request by an affected Customer, assign new authorization codes to replace any that were deactivated. Whenever call blocking occurs on lines presubscribed to the Company, Customers or former Customers will be unable to make calls via any of the Company's CICs or those of its affiliates. The Company may refuse to accept calling or credit card, collect calling and/or third number calls which it determines are or may be fraudulent, or it may limit the use of these billing options to or from certain countries or areas, including all or part of the United States, Puerto Rico, and the U.S. Virgin Islands.

- .06 976 Access: Access will not be provided to local 976 numbers.
- .07 Intentionally left blank.
- .08 Service Non-Use: If service is made available to a Customer and the Customer has failed to place that service into actual and substantial use during the six-month period immediately following its availability, or, if during any service term, the Customer has not actually and substantially used the available service for any consecutive six-month period, the Company may discontinue the service after furnishing written notice to the Customer regarding such non-use and, thereafter, the Customer fails to place the service into actual and substantial use during the three-month period immediately following therendition of notice. As used in this section, "actual and substantial use" shall mean a pattern of use that discloses an intention on the part of the Customer to use the service.
- .09 Emergency Restoration of Service: The use and restoration of service during emergency conditions will be performed in accordance with Part 64, Subpart D, Appendix A, of the Rules and Regulations of the Federal Communications Commission.
- .10 Customer Modifications: Neither the Customer nor any Authorized User may rearrange, disconnect, remove, modify, or attempt to repair service, or permit others to rearrange, disconnect, remove, modify, or attempt to repair service, without the prior written consent of the Company.

## 1. <u>SERVICES (CONT'D.)</u>

# 1. C General Provisions of Service

- .01 Intentionally Omitted.
- .02 Minimum Service Period: Except as otherwise expressly provided for in this tariff or in a Signed Contract, service is furnished and billed on the basis of a minimum period of thirty days, beginning on the date service is first made available to the Customer for use or on the date service otherwise commences.
- .03 24 X 7 Service: Except as otherwise expressly provided elsewhere in this tariff or in a contract, service is offered and provided on a monthly basis, twenty-four hours per day, seven days per week.
- .04 Monthly Billing Period: For the purpose of computing charges, a month consists of thirty days.
- .05 Service Application Information: The name of the Customer and any Authorized User(s) must be contained on any Service Application or enrollment form furnished by the Company and, in addition, the Customer must furnish all other information necessary for the Company to provide service.
- Collocation: The Company will entertain requests from Customers or prospective Customers, on a first-come, first-served basis, to collocate Customer Equipment at Company Terminals. Grant of a collocation request is subject to: (a) the Company's current and forecasted operational or physical space requirements, taking into account available space at a Terminal; (b) any applicable lease or occupancy restrictions imposed on the Company; (c) the technical and operational compatibility of the Customer's system or facilities with services; (d) the Company's security requirements; and (e) terms and conditions to which the Customer contractually will commit.
- .07 Service Inspection: The Company, upon reasonable notice, may:
  - .071 require that service be made available for inspection, testing or adjustment so that it can be maintained in a satisfactory condition. During any period of such inspection, testing or adjustment of service, no service credit will be given for the time during which service was unavailable to the Customer; or;
  - .072 undertake inspections or testing as may be necessary to determine whether the requirements of this tariff are being satisfied in connection with the installation, operation or maintenance of service. The Company may suspend, and subsequently discontinue, the provision of service at any time, without penalty, if the Customer or Authorized User fails or refuses to comply with these inspection, testing or adjustment requirements.

# 1. <u>SERVICES (CONT'D.)</u>

# 1. C General Provisions of Service (Cont'd.)

.08 Governmental Charges: The Company may adjust its rates and charges, or impose additional rates and charges, in order to recover amounts it is required or permitted by governmental or quasi-governmental authorities ("Governmental Charges") to collect from or pay to others in support of statutory or regulatory programs. Examples of such Governmental Charges include, but are not limited to, Universal Service funding and compensation payable to payphone service providers for use of their payphones to access Company service.

## 2. TERM

#### 2. A Term

The term for which service is provided is thirty (30) days, unless otherwise specified in this tariff with regard to a specific product or in a Signed Contract.

- .01 <u>Service Cancellation Request Received From Local Exchange Carrier</u>: If the Company cancels a Customer account or service at the request of the Local Exchange Carrier (LEC) serving the Customer, but the Customer nevertheless continues to complete calls over the Company's network by dialing 1+ or a Company CIC Code (e.g., 1010222), then, for a period not to exceed sixty (60) days from the date the Company first received the service cancellation request, the following provisions shall apply:
- .02 Customers who had subscribed to Long Distance Options A, B, C-1, or C-2, will receive service under the terms and conditions, including rates and charges, set forth under the offering to which the customer subscribed at the time of cancellation under those services.

# 2. B Service Cancellation or Change by Customer

- .01 Service Cancellation: If a Customer cancels an order for service before service commences, or before completion of a minimum service period, or before completion of some other period mutually agreed to between the Customer and the Company, the Customer will be billed and required to pay the Company for its unrecovered and nonrecoverable expenditures or liabilities incurred to establish, provide and terminate service for Customer, including without limitation termination charges paid to third parties, and all recurring charges identified in the Signed Contract.
- .02 Intentionally left blank.
- .03 Change of Service: When a change of service involves the continued use of service, installation charges will not apply to the service being continued in use. The minimum service period for the service will be deemed to have begun on the date the original service was first available to the Customer.

## 2. TERM (CONT'D.)

# 2. C Service Cancellation by Company

- .01 The Company may discontinue the furnishing of any and/or all service(s), or cancel a Customer's service or Customer accounts, without incurring any liability:
  - .011 Immediately, upon seven (7) days written notice to the Customer, if:
    - .0111 providing service would violate any applicable law, regulation, court order, or other legal authority;
    - .0112 any sum owed by the Customer has not been paid within thirty (30) days of the invoice date and remains unpaid during the seven (7) day period following the delivery of written notice of non-payment to the Customer;
    - .0113 the Customer fails to comply with a request by the Company for security for the payment of service;
    - .0114 the Customer subscribes to a calling card service and has not used the service (with the exception of calls to Directory Assistance) for 180 days. In such case, the Company may deactivate the card. If the Customer wishes to renew service (e.g., upon reopening its business), the Company will promptly supply a new card;
    - .0115 the Customer has failed to fulfill a contractual commitment to pay for service previously furnished to the Customer; or
    - .0116 the Customer is non-compliant with any other provision or requirement set forth in the tariff or in any contract between the Company and the Customer.
  - .012 Immediately, and without notice, if:
    - .0121 the Customer refuses to furnish information that: (1) is essential to the Company or its billing agent to invoice service; or (2) pertains to the Customer's credit-worthiness, its past or current use of common carrier communications service, or its planned use of such service;
    - .0122 the Customer has provided false information regarding the Customer's identity, address, credit-worthiness, past or current use of common carrier communications service, the planned use of service, or the Customer's status under federal and/or state low income programs;
    - .0123 the Customer either refuses to pay when billed for service or indicates to the Company or an entity billing on the Company's behalf that it does not intend to pay for service;

- 2. TERM (CONT'D.)
- 2. C Service Cancellation by Company (Cont'd.)
  - .0124 a Customer subscribed to a service accessed with an authorization code has not used the service (with the exception of calls to Directory Assistance) for 90 days. In such case the Company may deactivate the authorization code. If the Customer wishes to renew service (e.g., upon reopening its business), the Company will promptly supply a new authorization code. This provision does not apply to Customers whose service is accessed by dialing "1+" into central offices where equal exchange access is available;
  - .0125 a Customer has not used a Service (of any type) for 12 months;
  - .0126 the Customer or prospective Customer uses service to transmit or receive a message, locate a person, or otherwise give or obtain information without payment for service;
  - .0127 the Customer uses, or attempts to use, service with the intent to avoid the payment, either in whole or in part, of the charges for the service by:
    - .01271 using or attempting to use service by rearranging, tampering with, or making connections to service in an unauthorized manner;
    - .01272 using tricks, schemes, false or invalid numbers, false credit devices, electronic devices; or
    - .01273 any other fraudulent means or devices.
  - .0128 the Customer uses service to entice callers to dial certain numbers and thereby incur charges without informing the callers that such charges will be incurred:
  - .0129 the Customer previously was provided with notice of noncompliance with any provision in this tariff, took corrective action within the requisite seven (7) day period to comply with the provision, but thereafter engages in the same non-compliant activity;
  - .01210 the Customer acts, or fails to act, in a manner that hinders or frustrates any investigation by the Company or others having legal authority to investigate Customer compliance with this tariff or with the Customer's other legal obligations;
  - .01211 the Customer fails to pass back to the Company appropriate answer supervision so that a call can be billed;

- II. TERMS AND CONDITIONS (CONT'D.)
  - 2. TERM (CONT'D.)
  - 2. C Service Cancellation by Company (Cont'd.)
    - .01212 the Customer is reselling service, having elected to use a Company Carrier Identification Code (CIC), but fails or refuses to indemnify the Company for costs incurred by the Company as the result of its use of a Company CIC;
    - 0.1213 the Customer uses service for an unlawful purpose, or in a manner that interferes with service to other users; or
    - .01214 the Company reasonably determines, in its sole judgment, that such action is necessary to prevent or protect against fraud or to otherwise protect its personnel, agents, facilities or services.
    - Discontinuance or cancellation of service by the Company will not relieve the Customer of any obligation to pay for service previously furnished to the Customer or for any termination or other charges. Upon the Company's discontinuance or cancellation of service to the Customer, all applicable charges, including termination charges, become due. The remedies set forth above are in addition to all other remedies that may be available to the Company at law or in equity or under any other provision of this tariff or a Signed Contract.

## 2. TERM (CONT'D.)

## 2. D Notice

Except as otherwise expressly provided for in this tariff or in a Signed Contract, service will be provided until canceled by the Customer with the required prior written notice or until canceled by the Company pursuant to the notice requirements set forth in Section 2.C. For service provided exclusively solely within this state, Customer must provide 30 days written notice. Disconnection notices must be labeled conspicuously "Disconnect Request." Notice is deemed effective only if Customer receives a written confirmation of receipt from Company; Customer should contact its account representative or Customer Service if it does not receive such confirmation within 5 business days. Notice must be in writing and either transmitted via overnight courier, electronic mail, hand delivery or certified or registered mail, postage prepaid and return receipt requested to the following addresses. Except as provided otherwise, notices will be deemed to have been given when received. This provision does not waive or otherwise affect any contractual charges or damages that may arise from the termination of a service or a contract. Customer remains responsible for any use of service (including related recurring charges), even if service is disconnected after the requested disconnection date.

To Company:

Verizon Business Services 6415-6455 Business Center Drive Highlands Ranch, CO 80130

Attn: Customer Service

Email: notice@verizonbusiness.com

With Copy to:

Verizon Business Services 22001 Loudoun County Parkway

Ashburn, VA 20147

Attn: Vice President, Legal

To Customer, notice may be provided to any of the following:

- the notice address in a Signed Contract that has not been superceded even if the contract term has expired
- an address to which invoices are sent
- any other address reasonably calculated to achieve actual notice under the circumstances

## 2. E Service Transitioning

If a Customer voluntarily discontinues service or has its service terminated for cause, including for non-payment of amounts due and owing the Company, the Company will have no obligation whatsoever to assist the Customer in any respect in transitioning to another provider's service.

# MINIMUM ANNUAL VOLUME COMMITMENT, UNDERUTILIZATION AND EARLY TERMINATION CHARGES

Minimum Annual Volume Commitments and Underutilization and/or Early Termination Charges, if applicable, are established in the tariff for each service or in the Signed Contract.

#### 4. RATES, CHARGES AND PAYMENT TERMS

## 4. A Payment Terms

.01 Service Charges: The Customer is responsible for payment to the Company, or to any entity designated by the Company as its collection agent, for all charges associated with service furnished to the Customer or any Authorized User, including any applicable underutilization or early termination charges. This responsibility for payment is not changed by virtue of any use, misuse, or abuse of the Customer's service or Customerprovided systems, facilities or services interconnected to the Customer's service, which use, misuse, or abuse may be occasioned by third parties including, without limitation, the Customer's employees or other members of the public. Payments must be made at the address designated in the invoice or other such place as Company may designate. Amounts not paid or disputed under the terms of the Company Customer Agreement on or before 30 days from invoice date or such other due date set forth as provided above are considered past due. For the purpose of billing, service is deemed to start on the day the service and its associated equipment, if any, are installed or otherwise made available for Customer's use. Where billing is based upon Customer usage, Customers will be billed for all usage starting on the date usage begins.

## Charges for service may be:

- .011 non-recurring or one-time charges that are payable when the service with which they are associated has been performed. If an entity other than the Company (e.g., another carrier or a supplier) imposes charges on the Company in connection with service provided to a Customer, those charges, along with any applicable Company charges, will be charged to the Customer;
- .012 recurring charges which are fixed in amount and not dependent on usage are billed in advance; or
- .013 usage charges billed after each usage cycle associated with service.

The charges for which the Customer is responsible are the rates and charges in effect at the end of the monthly billing period applicable to the Customer for the service provided.

# .02 Billing Increment:

.021 The duration of a call or other billing unit is rated in the billing increments applicable to the service. If the final interval of a call or other billing unit is less than the applicable billing increment, it will be rounded up to a full increment for purposes of billing.

- 4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)
- 4. A Payment Terms (Cont'd.)
  - .022 Computed usage charges or credits for each call are rounded to the nearest cent (except as noted below for the minimum charge).
  - .023 Billable time for switched telecommunications service is the duration of time between the called station answering and the called or calling station disconnecting, provided duration may be rounded in accordance with specific descriptions in this tariff.
  - .03 Minimum Charge: The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$ .01. For example, if a Customer's ECR platform or transport charge calculates to anything less than \$.01 (e.g., \$.004, each such charge will be billed at \$.01).
  - .04 Payment Period: Invoices are due and payable in U.S. dollars within thirty (30) days of the invoice date, unless the Company expressly indicates otherwise on the invoice or in a contract between the Company and the Customer. If the Company becomes concerned at any time about the ability of a Customer to satisfy its payment obligation, the Company, in its sole discretion, may require that the Customer pay its invoices within a specified number of lesser days and to make such payments in cash or the equivalent of cash. A late payment charge equal to the lesser of: (i) one and one-half percent (1.5%) per month, compounded, or (ii) the maximum amount allowed by applicable law will be applied against past due amounts, including taxes. The late payment charge will be applied to any disputed portion of the unpaid balance that is resolved against the customer. A charge is considered disputed on the date the Company receives sufficient documentation to raise a bona fide question about the charge which Company is able to effectively investigate. A dispute is deemed resolved on the date the Company completes its investigation and notifies the Customer of its disposition. A check return charge will be assessed for checks with insufficient funds or non-existing accounts.
  - .05 Outstanding Balance Payments: The Company may require a Customer that owes an outstanding balance to any Company affiliate to pay the outstanding amount in full before the commencement of service.
  - Security Deposits: Customers or prospective Customers whose financial condition either is not known or not acceptable to the Company may be requested and required at any time to provide the Company with a security deposit. Such deposits must be paid in cash or the equivalent of cash. Security deposits must be in an amount equal to the applicable installation charges, if any, and/or up to one-half of the annual volume commitment (or other minimum purchase amount). Any Customer or prospective Customer may also be required at any time, whether before or after the commencement of service, to provide such other assurances of, or security for, the payment of charges for its services as the Company may deem necessary including, without limitation, advance payments for service, payments in cash or cash equivalents, third party guarantees of payment, pledges or other grants of security interests in the customer's assets, and other similar arrangements. The Company also may establish toll usage limits for Customers or prospective Customers, or it may require from the Customer a commercial credit card

#### 4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)

#### 4. A Payment Terms (Cont'd.)

account number against which future usage can be charged. Any required deposit or toll usage limits may be increased or reduced by the Company as a result of its experiences with the Customer. In the case of a cash deposit, simple interest at the rate of six percent (6%) annually will be paid for the period during which the deposit is held by the Company. unless a different rate has been established by the appropriate legal authority in the jurisdiction in which service is being provided. At the Company's election, a deposit may be refunded by crediting it against the Customer's account at any time.

- .07 Security Compliance: The Company may refuse to accept or process service orders between the time of its request for a security deposit or commercial credit card account number against which service charges can be applied and the time of a Customer's compliance with the request.
- .08 Past Due Accounts: The Company may refuse to furnish service if any Customer account with the Company is past due.
- .09 Installation Charges: All installation charges assume installation at normal locations under normal working conditions. Any installation made under other circumstances is subject to an additional charge.
- .10 Intentionally Omitted.
- .11 Application of Service Credits: Service credits due Customer that are related to, or based on, service usage will be applied before the application of taxes and any Governmental Charges; and service-related credits due Customer that are not related to, or based on, service usage will be applied after the application of taxes and any Governmental Charges.

#### .12 Taxes

- .121 The Company's rates and charges for service are exclusive of the following charges that will be assessed, and which shall not be counted toward the attainment of any volume or revenue commitment and will not be discounted:
  - applicable federal, state, local, and foreign sales, use, excise, utility, gross receipts and value added taxes;
  - any tax imposed by an authority on the benefits of a promotion offered by the Company involving services or goods of a third party;
  - .1213 other taxes:
  - .1214 tax-like charges to recover amounts the Company is required or permitted by a governmental or quasi-governmental authority to collect from others or pay to others in support of statutory or regulatory funds or programs;

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

- II. TERMS AND CONDITIONS (CONT'D.)
  - 4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)
  - 4. A Payment Terms (Cont'd.)
    - .1215 other tax-like charges;
    - .1216 a tax-related surcharge imposed on all charges (net of bad debts) for outbound service originating in, or inbound service terminating in, a jurisdiction which levies, or asserts a claim of right to levy:
      - .12161 a gross receipts tax, a license tax, or other tax-like charge on the Company's operations in that jurisdiction based on the Company's gross receipts, revenues or operations in that jurisdiction.
    - All taxes, tax-like charges and the tax-related surcharges are referred to collectively as "Tax(es)." The Company may elect to impose and collect such Taxes, unless otherwise constrained by court order or direction. The Customer agrees to pay all Taxes imposed. If the Company has collected Taxes and a challenged Tax is found to have been invalid and unenforceable, and if the amounts collected were retained by the Company or delivered over to the jurisdiction and returned to the Company, the Company, in its sole discretion, may reduce service rates for a fixed period of time in the future in order to flow-through to customers an amount equivalent to the amounts collected, credit or refund such amounts to affected customers (less its reasonable administrative costs), or negotiate an arrangement with the jurisdiction to provide a future benefit for customers in that jurisdiction. Customer will pay any Taxes that become applicable retroactively.
    - .123 If the Customer provides the Company with a duly authorized exemption certificate, the Company will exempt the Customer in accordance with law, effective on the date the Company receives the certificate.
    - .124 Taxes based on the Company's net income will be the Company's sole responsibility.
    - .13 Notice of Dispute: If the Customer does not give written notice to the Company of a dispute with respect to invoiced charges or the application of Taxes within six (6) months of the date of the invoice, the invoice, or application of taxes, shall be deemed to be correct and binding on the Customer for all purposes.
    - .14 Collection Expenses: If the Company incurs any fees or expenses, including attorneys' fees, in collecting, or attempting to collect, any charges owed the Company, the Customer will be liable to the Company for the payment of all such fees and expenses reasonably incurred.
    - .15 Treatment of Credit Balances: If a Customer whose account has been closed has a credit balance showing, the Company will transfer the credit to another account, if the Customer has one, or will mail a check for the balance to the Customer at its last known address according to the records of the Company.

## 4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)

# 4. A Payment Terms (Cont'd.)

- .16 Early Termination Charges: No provision for payment of a sum upon termination of service prior to the end of a committed term (whether the sum is called a termination charge, termination liability, or is otherwise designated), where termination is (a) by the Customer in the absence of a material breach by the Company of its obligations, or (b) by the Company when permitted or required, shall be construed as an alternative performance or in any other manner a grant of permission or right to the Customer to terminate service prior to the end of the committed term. Any such early termination will be deemed to be a material breach by the Customer of its contractual obligation to the Company.
- .17 Check Payments: In the event payment is made by check and the Customer's check is not honored by the institution on which it was drawn, the Company will impose, and the Customer will be required to pay, a \$15 fee, in addition to all other remedies available to the Company.
- Delay in Billing Capability: If billing systems or other required support is not available to bill for service, any service feature, any service-related surcharge, or any other charge element at the time of service provision, the Company will bill for that service, feature, surcharge or other charge element as soon as it is capable of doing so.
- Unbilled Charges: Except in cases involving fraud, the Company will invoice previously unbilled charges for service if the invoice date is no later than one hundred and eighty (180) days from the end of the monthly billing period in which the charges occurred. In cases involving fraud, the Company will invoice previously unbilled charges if the invoice date is no later than eighteen (18) months from the end of the monthly billing period in which the charges occurred.
- .20 Intentionally left blank.
- .21 Unbillable Charges: Calls may not be placed using a local exchange company's calling card if that local exchange company is not obligated to invoice such calls on Company's behalf, and calls may not be placed or received using 10-10XXX dialing or collect or third party calling conventions, e.g., 1-800-COLLECT, if the serving local exchange carrier is not obligated to provide billing name and address (BNA) information to Company in connection with such calls.
- .22 Applicable Local Exchange Carrier Terms: The terms and conditions for billing, payment and collection, including without limitation, any late payment charge, specified in the applicable local exchange carrier's tariff apply to charges of Company for which the local exchange carrier serves as the billing agent for Company or buys Company's accounts receivables.
- .23 If an entity (other than the Company) imposes charges on the Company, in addition to its own internal costs, in connection with a Service, those charges may be passed on to the Customer.

- II. TERMS AND CONDITIONS (CONT'D.)
  - 4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)

### 4. B Service Credits

The following credits for service interruptions apply, except for Directory Assistance, and except as otherwise expressly provided for in this tariff or in a Signed Contract.

- .01 For services for which charges are specified on the basis of per minute of use, or on usage of a fraction of a minute, the following credits will be made for an interrupted call that can be remedied by redialing the called number.
  - .011 Usage Charges: A credit will be made for that portion of a call which is interrupted due to poor transmission (e.g. noisy circuit), one-way transmission (one party is unable to hear the other), or involuntary disconnection caused by deficiencies in service. A Customer also may be granted a credit for reaching a wrong number. To receive a credit, the Customer must notify his or her Customer Service Representative and furnish information, including the called number, the service subscribed to, the difficulty experienced, and the approximate time the call was placed.
  - .012 When a call is involuntarily disconnected, the Customer will be given a credit equivalent to the charge for the initial minute of the call made to reestablish communications with the other party. When a domestic call has been interrupted by poor transmission or one-way transmission, the Customer will be given a credit of the Customer's choosing, either (a) \$2.65 per reported call, or (b) an amount equal to the basic or standard charges for the reported call, up to a maximum of thirty (30) minutes. A Customer who has reached a wrong number will be given a credit equal to the charge for the initial minute of the call to the wrong number, if he or she reports the situation promptly to a Customer Service Representative. This credit is limited to \$100 over a twelve (12) month period.
- .02 Recurring Charges: For services for which charges are specified on the basis of a monthly recurring charge, the following credits will be made for service interruptions.
  - .021 If service is interrupted for a period of less than two (2) hours, no credit will be given.
  - .022 When service is interrupted for a period between two (2) and twenty-four (24) hours, a credit will be given in an amount equal to one thirtieth (1/30) of the applicable monthly recurring service charge.
  - .023 When service is interrupted for a period of more than twenty-four (24) hours, a credit will be given in an amount equal to one thirtieth (1/30) of the applicable monthly recurring service charge for each twenty-four (24) hour period or fraction thereof that service is interrupted.

- 4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)
- 4. B Service Credits (Cont'd.)
  - .03 Alternative Credit: In lieu of the credits provided for in the previous three subsections, the Company may apply a credit against usage or monthly recurring charges in an amount not to exceed \$1,000 per Customer or account, per monthly billing period, whenever the Company determines, in its sole discretion, that such a credit is warranted due to considerations involving the delivery of past service to the Customer or an account.
  - .04 No Credit: No credits will be given for interruptions:
    - .041 caused by the act or omission of the Customer or an Authorized User;
    - due to the failure of power, equipment, systems, facilities or services not provided by the Company;
    - during any period during which the Company or its representatives are not afforded access to the premises where access lines or Company facilities associated with the Customer's or Authorized User's service are located;
    - during any period when the Customer or an Authorized User has released service to the Company or its representative for maintenance, service rearrangement, or the implementation of a Customer service order;
    - .045 during any period when the Customer has chosen not to release service for testing or repair and the Customer continues to use service on an impaired basis:
    - .046 due to network busy conditions;

or

- .047 not reported to the Company.
- .05 Interruption Period: An interruption period begins when the Company receives notification that service has been interrupted and service has been released for testing and repair. An interruption period ends when service is restored. If the Customer reports that service has been interrupted, but refuses to release it for testing and repair, the service is deemed to be impaired, but not interrupted for crediting purposes.
- .06 Affected Service: Only those segments of service actually affected by an interruption are considered in determining the credit amount.
- .07 Use of Alternative Service: If the Customer elects to use another means of communicating after a service interruption has occurred, or during a period when he or she is unable to use service, the Customer is responsible for paying the charges for the alternative service used and will not be reimbursed by the Company other than via the appropriate credit as set forth in this tariff.

## 4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)

# 4. B Service Credits (Cont'd.)

- .08 Extent of Service: For purposes of crediting, "service" includes facilities provided by the Company and/or a participating international carrier or an overseas administration connecting with such international carrier.
- .09 Redundancy: At the Customer's request and cost, the Company will undertake to make arrangements for service redundancy, whenever possible.

## 4. C Fraud

- .01 Theft of Service: Except with respect to usage involving the use, misuse or abuse of Customer Premises Equipment (CPE) interconnected with service, or as otherwise provided in this section or in any Signed Contract, the Company will issue full credit for invoiced charges for Customer calls determined to the reasonable satisfaction of Company to have resulted from a "theft of service." A "theft of service" is the unauthorized use of the Customer's service following its theft by a third person over whom neither the Customer nor an Authorized User possesses an ability to control. Under no circumstance will credit be issued for service use resulting from the acts or omissions of the Customer or any Authorized User, or from the acts of any of the Customer's or Authorized User's employees, former employees, agents, vendors or independent contractors. To qualify for a credit, the Customer must: (1) notify the Company in writing within ninety (90) days of the Customer's receipt of the first invoice containing alleged unauthorized service use; and (2) co-operate fully with the Company in connection with any investigation, prosecution or litigation arising from such theft of service. The Customer's written notification must identify with specificity the service use for which the Customer is seeking credit.
- .02 Remote Toll Fraud Program: The Remote Toll Fraud Program provides the procedure required for Customer fraud credit requests and limits a qualifying Customer's liability for verified Remote Toll Fraud usage charges. For purposes of this Program, Remote Toll Fraud is defined as: (i) the placement of unauthorized outbound calls to international locations by using the Customer's outbound international service; or (ii) unauthorized use of eligible toll-free service originating from a domestic location(s) and compromising certain CPE associated with the service(s) located in the United States. The following terms and conditions apply to the Remote Toll Fraud Program:
  - .021 To be eligible to receive benefits under this Program, the Customer must:
    - .0211 provide the Company with requested information concerning any password(s) used to access CPE involved in any suspected Remote Toll Fraud:
    - .0212 co-operate with the Company in all efforts to identify, prevent or eliminate suspected or confirmed Remote Toll Fraud. For purposes of notification, the Customer must furnish the designated Company Representative with: (i) the names of Customer personnel involved in the Program, including individuals capable of being contacted 24 hours per day, seven

# 4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)

# 4. C Fraud (Cont'd.)

days per week; and (ii) Customer pager, cellular or off-hour telephone numbers. This information must be furnished initially by the Customer and updated, as necessary, in order for the Customer to become, and remain, eligible to receive benefits under the Program;

- .0213 provide follow up information regarding the nature of any potentially fraudulent usage within forty-eight (48) hours of a Company notification to the Customer. Provide Company with access to its CPE within twenty-four (24) hours of a Company request, and the Customer must allow the Company Representative to investigate the current and/or former CPE configuration. Under no circumstance will the CPE configuration be, or be deemed to be, the responsibility of the Company;
- .0214 immediately notify the Company whenever Remote Toll Fraud is suspected or detected, irrespective of whether other common carrier(s) are involved;
- .0215 notify the Company in writing within sixty (60) days of receipt of the first Company invoice containing any suspected Remote Toll Fraud usage charges. If the Customer fails to notify the Company, all usage charges for which written notice has not been provided will not qualify as Remote Toll Fraud usage charges. The Customer's notice must establish with specificity (e.g., by way of call detail records) to the reasonable satisfaction of the Company that the suspected usage charges are covered under the Program; and
- .0216 notify the Company in writing within sixty (60) days of the end of the Remote Toll Fraud incident of the means by which the fraud occurred, if known, and the changes made to the Customer's CPE to prevent future Remote Toll Fraud.
- .022 Program Benefits and Other Program Conditions:
  - .0221 CPE eligible for participation in the Program is limited to a single Private Branch Exchange (PBX) or a single electronic key system located on Customer Premises within the United States.
  - .0222 Under no circumstance will the Program cover, nor will Remote Toll Fraud be defined to include, calls placed by wireless devices, dial-around (10-10-XXX) calls, Operator Service calls, toll-free or 900 pay-per-call traffic, or calls made through any non-Company conferencing service or Centrex system.

# 4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)

# 4. C Fraud (Cont'd.)

- .0223 For each incident of Remote Toll Fraud, an eligible Customer will be liable to the Company for the first \$10,000 of Remote Toll Fraud usage charges for calls occurring prior to notification. For purposes of this section notification is defined as (i) notice to the Customer from an authorized representative of the Company's Fraud Prevention organization or Account Team representative; or (ii) notice from the Customer to the Company's Fraud Prevention organization or Account Team representative of suspected remote access fraud. An eligible Customer's pre-notification liability will not exceed \$10,000 per Remote Toll Fraud incident, but the Customer will be liable to the Company for all Remote Toll Fraud usage charges for calls occurring after notification is given by the Company to the Customer or by the Customer to the Company.
- .0224 The \$10,000 limitation of Customer liability will not apply to CPE for which a credit previously was given, or to any other CPE connected to CPE for which a credit previously was given, until a "30 day fraud-free period" has elapsed. This period will be measured beginning with the date of the last fraudulent call of the last incident affecting the same CPE.
- .0225 CPE not owned or leased by the Customer and not subject to the Customer's direct control, whether on Customer Premises or elsewhere, is not eligible under the Program.
- .0226 The Program will not cover any Remote Toll Fraud usage charges resulting from the negligent or intentional acts of the Customer, its employees, former employees, agents, vendors or independent contractors.
- .0227 Credited Remote Toll Fraud usage charge amounts will be excluded from the ascertainment of volume/term discount levels and satisfaction of any applicable volume/revenue requirements.
- .0228 With respect to any credit amount awarded to Customer under this Program, the Company is subrogated to any and all rights of the Customer with respect to any associated claims against third parties (including, without limitation, any person who made the unauthorized calls resulting in the credit amount given).
- .023 Failure of the Customer to comply with any of its obligations under the Remote Toll Fraud Program will disqualify the Customer from current and future participation in the Program at all Customer locations.

4. RATES, CHARGES AND PAYMENT TERMS (CONT'D.)

# 4. C Fraud (Cont'd.)

.03 Company may, but is not required to, advise Customer of abnormal calling patterns or other possible unauthorized use of facilities or Service assigned to the Customer.

Additionally, the Company may, but is not required to, block calls on authorization codes which Company believes to be unauthorized or fraudulent.

### 5. TERMINATION

Termination rights of the Company and the Customer are established in this tariff or in any Signed Contract.

- 6. DISCLAIMER OF WARRANTIES AND LIMITATION OF LIABILITY
- 6. A WARRANTY DISCLAIMER. EXCEPT AS SPECIFICALLY SET FORTH IN THIS TARIFF OR IN A SIGNED AGREEMENT, COMPANY MAKES NO WARRANTIES, EXPRESS OR IMPLIED, AS TO ANY COMPANY SERVICES, SOFTWARE OR DOCUMENTATION. COMPANY SPECIFICALLY DISCLAIMS ANY AND ALL IMPLIED WARRANTIES, INCLUDING WITHOUT LIMITATION ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE, NONINFRINGEMENT OF THIRD-PARTY RIGHTS, OR ANY WARRANTIES ARISING FROM A COURSE OF DEALING, USAGE OR TRADE PRACTICE.
- 6. B LIABILITY DISCLAIMER. Except for credits for service interruptions, as provided for in Section 4.B, the Company will not be liable for any failure of performance due to causes beyond its reasonable control, including, but not limited to, acts of God, fires, floods or other catastrophes; terrorist acts, embargo, national emergencies, insurrections, riots or wars, strikes, lockouts, work stoppages or other labor difficulties; unavailability of equipment, software or parts from vendors; preemption of service to restore service in compliance with Part 64, Subpart D, Appendix A, of the FCC's Rules and Regulations; any law, order, regulation or other action of any governing authority or agency thereof; or changes requested by Customer.
- 6. C LIABILITY LIMITATION. The liability of the Company in tort for willful misconduct, if established as a result of judicial or administrative proceedings, is not limited by this tariff. Except as noted in the preceding sentence, the Company's liability, if any, is limited to an amount equal to the credit, if any, provided by this tariff. The limitations in this sub-section apply, without limitation, to any claim or suit, by a Customer or by any others, for damages associated with the ordering (including the reservation of any specific number for use with a Service), installation, provision. termination, maintenance, repair, interruption, restoration or billing of any Service (including component features, options or facilities) offered under this tariff, including without limitation from mistakes, omissions, interruptions, delays, errors or defects of any kind. NOTWITHSTANDING ANY PROVISION OF THIS TARIFF TO THE CONTRARY, IN NO EVENT WILL THE COMPANY BE LIABLE TO THE CUSTOMER OR OTHERS FOR ANY INDIRECT, SPECIAL, INCIDENTAL, CONSEQUENTIAL, EXEMPLARY OR PUNITIVE LOSS OR DAMAGE OF ANY KIND. INCLUDING WITHOUT LIMITATION LOSS OF USE OR LOST BUSINESS REVENUE, PROFITS OR GOODWILL, UNDER ANY THEORY OF TORT, CONTRACT, INDEMNITY, WARRANTY, STRICT LIABILITY, NEGLIGENCE OR OTHERWISE (WHETHER OR NOT THE COMPANY KNEW OR SHOULD HAVE KNOWN OF THE POSSIBILITY OF SUCH LOSS OR DAMAGE).

- II. TERMS AND CONDITIONS (CONT'D.)
  - 6. DISCLAIMER OF WARRANTIES AND LIMITATION OF LIABILITY (CONT'D.)
  - 6. D LIMITATION OF ACTION. No claim, action or other proceeding against the Company may be brought more than two years after the service to which it relates is rendered.
  - 6. E INDEMNIFICATION. The Company will be indemnified, defended and held harmless by the Customer, and by others authorized to use service, against all claims of loss, expense or damage arising from the use of service, including without limitation:
    - .01 claims of libel, slander, invasion of privacy, or infringement of copyright arising out of the material, data, information or other content transmitted via service;
    - .02 patent infringement claims arising from the combining or connecting of service with systems or facilities furnished by the Customer or others;
    - claims by the Customer or others for any personal injury to, or death of, any person or persons, and for any loss, damage or destruction of any property, whether owned by the Customer or others, caused or claimed to have been caused, directly or indirectly, by the installation, operation, failure of operation, maintenance, removal, presence, condition, location or use of service furnished by the Company at any location at which there is present an atmosphere that is explosive, prone to fire, dangerous, or otherwise unsuited for service installation;
    - .04 claims of criminal enterprise;
    - .05 claims arising out of the failure of Customer-provided equipment, facilities, systems or services;
    - .06 claims for personal injury or death of any person related to the installation, maintenance, location, condition, operation, failure, presence, use or removal of equipment or wiring provided by Company, if not solely caused by negligence of Company;
    - .07 claims arising out of the use by Customer of Services which use has been restricted or limited by action of a government agency having jurisdiction over the Customer, the Company or its Services;

or

- all other claims arising out of any act or omission of the Customer or an Authorized User in connection with service provided by the Company.
- 6. F THIRD PARTY ACTS. The Company will not be liable for any act or omission of any other person furnishing a portion of service, or for any act or omission of a third party, including those vendors participating in offerings made to Customers that involve service, or for damages associated with service or facilities which the Company does not provide, or for damages which result from the operation of Customer-provided systems, facilities or services interconnected with the Company's service.
- 6. G DANGEROUS CONDITIONS. The Company makes no guarantee or warranty with respect to service installations at locations at which there is present an atmosphere that is explosive, prone to fire, dangerous or otherwise unsuitable for such installations.

## 6. DISCLAIMER OF WARRANTIES AND LIMITATION OF LIABILITY (CONT'D.)

- 6. H PREMISES LIABILITY. The Company is not liable for any defacement of, or damage to, the premises of a Customer or Authorized User resulting from the furnishing of service, or the attachment to service by the Company of instruments, apparatus and associated wiring on such Customer Premises, or from the installation or removal thereof, when such defacement or damage is not the result of Company negligence, including that of its agents or contractors. No agents or employees of other carriers shall be deemed to be agents or employees of Company.
- 6. I RISK-BASED PRICING. The rates and charges in this tariff (as well as its other terms and conditions) are based and dependent on the tariff's disclaimers and limitations of warranties and liability.
- 7. ADDITIONAL TERMS AND CONDITIONS FOR TOLL-FREE SERVICE.

In addition to the other General Terms, for toll-free service (including features), the following apply:

- .01 At the Customer's request, the Company will perform the function of a Resp Org, specifically: (1) searching for, and reserving, toll-free telephone numbers with the toll-free Service Management System (SMS/toll-free); (2) creating and maintaining toll-free telephone number Customer records with the SMS/toll-free; and (3) providing a single-point-of-contact for trouble reporting involving toll-free service.
- .02. In its capacity as Resp Org, the Company will reserve, assign, activate or change, upon request, toll-free numbers for a Customer or prospective Customer, and will administer toll-free numbers in accordance with the rules of the Federal Communications Commission, customary industry standards and practices, the terms of this tariff, and the effective procedures of the SMS/toll-free. Customers may request the reservation, assignment or activation of toll-free numbers on their own behalf, or a Customer which resells toll-free services may make such requests on behalf of its customers. A Customer who resells toll-free services must provide to any of its customers or prospective customers, upon reasonable request, information concerning the status of a particular toll-free number(s) in which the customer or prospective customer has an interest and, if applicable, the identity of the Resp Org(s) for the toll-free number(s). When a resale Customer decides (or learns of its customer's decision) not to utilize the reserved, assigned or activated toll-free number, the Customer must notify the Company within forty-eight hours so that the Company can release the toll-free number into the pool of numbers available for assignment in accordance with industry standards and practices.
- .03 With respect to any claim that arises out of the Company's acting as a Resp Org or relates to the Company's provision of toll-free service, including without limitation toll-free Directory Assistance and toll-free service provided with a number or numbers other than the one(s) provided by the Company to the Customer, the liability of the Company is limited to the lesser of (a) the actual monetary damages incurred and proved by the Customer as the direct result of the Company's actions, or (b) \$1,000.
- A Customer of toll-free service is responsible for payment for all calls placed to or via the Customer's service number(s). This responsibility is not changed by virtue of any use, misuse, or abuse of the Customer's service or interconnected Customer-provided system facilities or service, which use, misuse or abuse may be occasioned by third parties including, without limitation, the Customer's employees and members of the public who dial the Customer's toll-free service number(s) either by mistake or with the intent to abuse service.

# 7. ADDITIONAL TERMS AND CONDITIONS FOR TOLL-FREE SERVICE (CONT'D)

- .05 If a Customer accumulates more than \$1,000 of undisputed Company toll-free service charges, the Company may refuse to honor any Customer request for a Resp Org change until such undisputed charges are paid in full.
- .06 The following pertain to the use of toll-free numbers:
  - .06.1 Notwithstanding any other provision in this tariff, if the Company determines that a toll-free number associated with service provided by the Company is being used in violation of the requirements of this tariff, the Company may terminate service immediately. Contemporaneously with service termination, the Company will notify the Customer by certified mail of the action it has taken and the reasons therefor.
  - .06.2 If the Company terminates toll-free service pursuant to Section 6.1 above, the Company will retain control for four (4) months of all toll-free numbers associated with terminated service. During the four-month period, the Company will not transfer the number to any other Customer, will refuse to reconnect the number for the previous Customer, except upon direction from the Federal Communications Commission or a court or commission of competent jurisdiction; will refuse to honor transfer of service arrangements between the disconnected Customer and any third party; and will refuse to honor any change of Resp Org forms issued by the terminated Customer.
  - .06.3 If the Federal Communications Commission, state commission, or a court of competent jurisdiction orders the Company to return the number to the control of the terminated Customer, or if the Company determines that its tariff requirements have not been violated, the Company will reestablish service without charge to the Customer. The Company also will not impose any underutilization or other charge as a result of a service disconnection that is overruled by the Federal Communications Commission of a court of competent jurisdiction.
  - .06.4 At the end of the four-month period, assuming there is no outstanding challenge to the actions of the Company, the Company will return control of the toll free number to the numbering authority for availability on a first-come, first-served basis pursuant to existing industry practices.
  - .06.5 The termination of service by the Company pursuant to this subsection does not relieve the Customer of any obligation to pay the Company for charges due and owing for service furnished up to the time of service termination. In the event service is terminated for cause and the Customer is committed to a term or other plan for which charges apply in the event of Customer termination for convenience, the Company will charge, and the Customer will be obliged to pay, as though it had terminated service for its own convenience.
  - .06.6 Answer Supervision: A call begins when a call is terminated on or passes through Customer's Premises equipment. The Customer must provide answer supervision back to the Company point of connection when the toll free service is connected to switching equipment or a Customer-provided communications system. In such cases, the equipment or system must provide appropriate supervision so that the measure of chargeable time begins upon the delivery of the call to the customer's switching equipment or communications system and ends upon termination of the call.

## 7. ADDITIONAL TERMS AND CONDITIONS FOR TOLL-FREE SERVICE (CONT'D)

- .07 The Company, upon written Customer request, may institute (and subsequently remove) call blocking to the Customer's toll-free telephone number(s) from certain countries, cities, NXX exchanges, or individual telephone stations in order to prevent the receipt of telephone calls made for the purpose of annoying, abusing, threatening or harassing any person at the called number. The Company reserves the right to limit the number of requests for toll-free call blocking per Customer.
- .08 Toll-free telephone numbers may not be hoarded, marketed or sold by Customers, except as permitted by the rules and policies of the Federal Communications Commission.

# 8. MISCELLANEOUS PROVISIONS

- 8. A Intentionally Omitted
- 8. B Confidential Information

Except as required by law or regulation, each party promises that during the Term and for three years after, it will use the other party's Confidential Information only for proper purposes, not disclose it to third parties except as provided below, and protect it from disclosure using the same degree of care it uses for its own Confidential Information (but no less than a reasonable degree of care). "Confidential Information" means information (in whatever form) designated as confidential by the disclosing party by conspicuous markings (if tangible Confidential Information) or by announcement at the time of initial disclosure (if oral Confidential Information) or if not so marked or announced should reasonably have been understood as confidential to the disclosing party (or one of its affiliates or subcontractors), either because of legends or other markings, the circumstances of disclosure or the nature of the information itself and that (i) relates to services provided under this tariff; (ii) relates to the disclosing party's customers, products, services, developments, trade secrets, know-how or personnel; and (iii) is received by the receiving party from the disclosing party during the Term. Confidential Information does not include information that: (a) is in the possession of the receiving party free of any obligation of confidentiality at the time of its disclosure; (b) is or becomes publicly known other than by a breach of this provision; (c) is received without restriction from a non-party free to disclose it; or (d) is developed independently by the receiving party without reference to the Confidential Information. In addition, information, whether or not Confidential Information, may be disclosed by a receiving party as may be required by applicable law, rule, regulation, or lawful process provided that the receiving party, to the extent permitted by applicable law, rule, regulation, or lawful process, first notifies the disclosing party in order to permit the disclosing party to seek reasonable protective arrangements.

## 8. MISCELLANEOUS PROVISIONS (CONT'D.)

### 8. C CPNI Consent

Company acknowledges that it has a duty, and Customer has a right, under federal and/or state law to protect the confidentiality of Customer's CPNI. CPNI includes information relating to the quantity, technical configuration, type, destination, location, and amount of use of the telecommunications services Customer purchases from Company, as well as related local and toll billing information, made available to Company solely by virtue of Customer's relationship with Company. With Customer consent, Company may share Customer CPNI and other Confidential Information among its affiliates (collectively, "Verizon"), including Verizon Wireless, and with agents and partners, so that all may use this information to offer Customer the full range of products and services offered by Verizon and its affiliates, including local, long distance, wireless, and Internet services (see www.verizon.com for a description of Verizon companies and services). Customer consents to Verizon using and disclosing Customer CPNI and Confidential Information as described above. Customer may refuse CPNI consent by notifying Verizon in writing at cpni-notices@verizonwireless.com and cpni-notices@verizonbusiness.com of Customer's decision to withhold Customer's consent. Customer's consent or refusal to consent will remain valid until Customer otherwise advises Verizon, and in either case, will not affect Verizon's provision of service to Customer.

#### 8. D Protection of Customer CPNI

- .01 Verizon will protect the confidentiality of Customer CPNI in accordance applicable laws, rules and regulations. Verizon may access, use, and disclose Customer CPNI as permitted or required by applicable laws, rules, and regulations or this tariff.
- .02 Provided that Customer is served by at least one dedicated Verizon representative under the Service Agreements (that can be reached by Customer by means other than calling through a call center) and as permitted or required by applicable law, Verizon may provide Customer CPNI (including, without restriction, call detail) to representatives authorized by Customer ("Authorized Customer Representatives" as defined below) in accordance with the following:
- Verizon may provide Customer CPNI to Authorized Customer Representatives via any means authorized by Verizon that is not prohibited by applicable laws, rules, or regulations, including, without restriction: to the Customer's email address(es) of record (if any) or other email addresses furnished by Authorized Customer Representatives, to the Customer's telephone number(s) of record or other telephone numbers provided by Authorized Customer Representatives, to the Customer's postal (US Mail) address(es) of record or to other postal addresses furnished by Authorized Customer Representatives, or via Verizon's on-line customer portal or other on-line communication mechanism.
- .04 Authorized Customer Representatives include Customer employees, Customer agents, or Customer contractors, other than Verizon, who have existing relationships on behalf of Customer with Verizon customer service, account, or other Verizon representatives and all other persons authorized in written notice(s) (including email) from Customer to Verizon. Authorized Customer Representatives shall remain such until Customer notifies Verizon in writing that they are no longer Authorized Customer Representatives as described below. Customer agrees, and will cause Authorized Customer

- 8. <u>MISCELLANEOUS PROVISIONS (CONT'D.)</u>
- 8. D Protection of Customer CPNI (Cont'd.)

Representatives, to abide by reasonable authentication and password procedures developed by Verizon in connection with disclosure of Customer CPNI to Authorized Customer Representatives.

- .05 Customer's notices of authorization or deauthorization must be sent to your service or account manager, and must contain the following information:
  - the name, title, postal address, email address, and telephone number of the person authorized or deauthorized
  - that the person is being authorized, or is no longer authorized, (as applicable) to access CPNI
  - the full corporate name of the Customer whose CPNI (and whose affiliates' CPNI) the person can access (or can no longer access as applicable)
- During the Service Agreements, Customer will at all times have designated representatives ("CPNI Authorizers") with the power to authorize Customer representatives to access CPNI under this tariff. Additions or removals of CPNI Authorizers will be effective within a reasonable period after Verizon has received a signed writing of the change, including the affected person(s)' name, title, postal address, email address, and telephone number.
- .07 Customer agrees that it will, consistent with all legal requirements, authenticate as authorized to access CPNI any person it provides with the means to obtain such access. Customer also will upon reasonable request provide to Verizon the identity and contact information for personnel authorized by Customer to access Service and CPNI, as well as any personal with the authority to authorize other personnel to access Service or CPNI. Customer agrees to cooperate with Verizon's reasonable authentication and security procedures for access to CPNI, including, without limitation, password resets and re-authentication of personnel.

## 8. MISCELLANEOUS PROVISIONS (CONT'D.)

# 8. E Customer Data Privacy

Customer acknowledges that the Company, its affiliates and agents will, by virtue of the provision of service, come into possession of information and data regarding Customer, its employees and Authorized Users. This information and data ("Customer Data") shall include, but not be limited to, data transmissions (including the originating and destination numbers and IP addresses, date, time and duration of voice or data transmissions, and other data necessary for the establishment, billing or maintenance of the transmission), data containing personal and/or private information of Customer, its employees or Authorized Users, and other data provided to or obtained by the Company, its affiliates and agents in connection with the provision of service. Customer acknowledges and agrees that the Company, and its affiliates and agents, may use, process and/or transfer Customer Data (including intra-group transfers and transfers to entities in countries that do not provide statutory protections for personal information): (1) in connection with provisioning of service; (2) to incorporate the Customer Data into databases controlled by the Company and its affiliates for the purpose of providing service, administration, provisioning, billing and reconciliation, verification of Customer identity and solvency, maintenance, support and product development, fraud detection and prevention, sales, revenue and customer analysis and reporting, market and customer use analysis, and (3) to communicate to Customer regarding products and services of the Company and its affiliates by voice, letter, fax or email. Customer may withdraw consent for such communications (or any use, transfer or processing of Customer Data except for that required to provision, administer, bill or account for the service) by sending written notice to the Company. Customer's consent is effective until withdrawn. Customer warrants that it has obtained or will obtain all legally required consents and permissions from relevant parties (including data subjects) for the use, processing and transfer of Customer Data as described in this Section.

- II. TERMS AND CONDITIONS (CONT'D.)
  - 8. MISCELLANEOUS PROVISIONS (CONT'D.)
  - 8. F Assignment

Customer and Company may assign a Signed Contract or any of its rights under it to an affiliate or successor upon notice to the other party. A Customer affiliate or successor must meet Company's creditworthiness standards for the assignment to become effective. All other assignments (or other transfers of rights or obligations) without prior written consent are void.

8. G Supplemental Customer Obligations Pertaining to Customer-obtained Facilities; Security; Site Access; and Customer Information

In addition to the obligations of Customer contained elsewhere in this tariff, Customer also will be responsible for satisfying the following requirements in connection with the receipt of service:

- .01 <u>Customer-Obtained Facilities.</u> Customer is responsible for obtaining, installing, and maintaining all equipment, software, wiring, power sources, telephone connections and/or communications services necessary for interconnecting with the Company's service ("Facilities"). Customer is responsible for ensuring that such Facilities are and remain compatible with service. The Company is not responsible for the availability, capacity and/or condition of any Facilities provided by third parties. Customer hereby grants to the Company all licenses, waivers, consents, or registrations necessary to deliver, install, and maintain Company-provided equipment on Customer or Authorized User premises.
- .02 <u>Company-Provided Equipment</u>. Customer agrees to operate Company-provided equipment in accordance with instructions of Company or Company's agent. Failure to do so will void any Company responsibility for interruption of service and may make Customer responsible for damage to equipment. Customer agrees to return to Company all Company-provided equipment delivered to Customer within five (5) days of termination of the service in connection with which the equipment was used. That equipment must be in the same condition as when delivered to Customer, normal wear and tear only excepted. Customer shall reimburse Company, upon demand, for any costs incurred by Company due to Customer's failure to comply with this provision.
- .03 <u>Security.</u> Customer, at its expense, will take all reasonable steps necessary to preserve and protect Company-provided equipment, software, data and systems located on Customer's premises or, otherwise, in Customer's control and used in connection with Company service, whether owned by Customer, the Company, or a Company affiliate or subcontractor.

Customer acknowledges and agrees that the Company will not be liable, either in contract or in tort, for any loss resulting from any unauthorized access to, alteration of, or use of Facilities used in connection with service. Customer agrees to safeguard account passwords and other information used to provide access to Company products and services. Customer shall require its end users to rotate passwords periodically, but no less often than every ninety (90) days. Customer shall cooperate fully with Company to promptly mitigate any unauthorized use or disclosure of Customer passwords or other authentication information.

## 8. MISCELLANEOUS PROVISIONS (CONT'D.)

- 8. G Supplemental Customer Obligations Pertaining to Customer-obtained Facilities; Security; Site Access; and Customer Information (Cont'd).
  - .04 Customer Site Access. Customer agrees to allow the Company and its affiliates or subcontractors and their respective employees and agents access to Customer or Authorized User premises at which service is being or will be provided (including access to associated equipment), without limitation for inspecting, installing, testing, repairing or removing service, on an unrestricted basis, 24 hours a day and 7 days a week.
  - .05 Customer Information. The Customer must furnish the Company with all information needed by the Company to install, maintain, change or remove service including, without limitation, circuit installation and disconnection authorization information.

### 8. H Software and Documentation

Where software is provided by Company under license from a third party, Customer's rights to use the software are subject to that license and Customer may be required to execute a separate software license agreement in a form satisfactory to the underlying third-party licensor. Software and related documentation provided by the Company to Customer in connection with service and not otherwise subject to either a separate Signed Contract or to an accompanying shrink wrap license (collectively the "Software") is subject to the following:

- In consideration for payment of any applicable fees, Customer is granted a personal, non-exclusive, non-transferable license to use the Software, in object code form only, solely in connection with service for Customer's internal business purposes on Customerowned or Customer-leased equipment (the "License"). Customer may not use the Software (i) in connection with the products and/or services of any third party, or (ii) to provide services for the benefit of any third party, including without limitation as a service bureau.
- .02 Customer may make one copy of the Software, other than the documentation, for archival or back-up purposes only, provided that any copyright and other proprietary rights notices are reproduced on such copy. Customer may not make any copies of documentation provided as part of the Software.
- .03 Customer may not: (i) attempt to reverse engineer, decompile, disassemble or otherwise translate or modify the Software in any manner; or (ii) sell, assign, license, sublicense or otherwise transfer, transmit or convey Software, or any copies or modifications thereof, or any interest therein, to any third party.
- .04 All rights in the Software, including without limitation any patents, copyrights and any other intellectual property rights therein, shall remain the exclusive property of the Company and/or its licensors. Customer agrees that the Software is the proprietary and confidential information of the Company and/or its licensors subject to the provisions of any contract between the parties pertaining to "Confidential Information."

## 8. <u>MISCELLANEOUS PROVISIONS (CONT'D.)</u>

- 8. H Software and Documentation (Cont'd).
  - .05 Except to the extent otherwise expressly agreed by the parties in writing, the Company has no obligation to provide maintenance or other support of any kind for the Software, including without limitation any error corrections, updates, enhancements or other modifications.
  - The License will immediately terminate upon the earlier of: (i) termination or expiration of any contract between the Company and the Customer pertaining to the Software; (ii) termination of the service with which the Software is intended for use; or (iii) failure of Customer to comply with any provisions of this Subsection. Upon termination of any License, at the Company's option, Customer will promptly either: (i) destroy all copies of the Software in its possession; or (ii) return all such copies to the Company, and in either event provide a written officer's certification confirming the same.

# 8. I Company Marks

A Customer may not: (a) use any service mark or trade mark of the Company or any of its affiliates, or of which the Company or any of its affiliates is a licensee, or (b) refer to the Company or any of its affiliates in connection with any product, equipment, offering, promotion (including without limitation in any press release, advertising or other publication) of the Customer or of a third party on behalf, or with the authorization, of the Customer, without the written approval of the Company affiliate. Customer agrees that: (a) any permitted use of a Company mark by it is for the exclusive benefit of the Company or the affiliate; (b) all good will resulting from use of the mark vests solely in the Company or the affiliate; and, (c) it will neither have nor make any claim in or to such mark.

#### 8. J Other Provisions

- .01 Intentionally Omitted.
- No Waiver: Neither the Company's nor the Customer's failure, at any time, to enforce any right or remedy available in this tariff will be interpreted as a waiver of such party's right to enforce each and every provision of the tariff in the future.
- .03 Severability: All provisions of this tariff are severable, and the invalidity or unenforceability of any provision will not affect the validity or enforceability of the remaining provisions. The remaining provisions will be interpreted in such a manner as to carry out the full intention of the parties.

- II. TERMS AND CONDITIONS (CONT'D.)
  - 8. MISCELLANEOUS PROVISIONS (CONT'D.)
  - 8. J Other Provisions (Cont'd).
    - .04 Dispute Resolution: Any controversy, claim, or dispute ("Disputed Claim") arising out of or relating to a service agreement between Company and Customer, except for claims relating to indemnity, infringement, or confidentiality obligations or matters relating to injunctions or other equitable relief (together "Equitable Claims"), are first subject to a thirty (30) day negotiation period between the Company and Customer in which each shall disclose to the other all such documents, facts, statements and any other information which are reasonably requested and are relevant to the dispute in question. If such negotiations fail to resolve the dispute within thirty (30) calendar days, Disputed Clams must be resolved by binding arbitration of a single arbitrator in accordance with the rules of the American Arbitration Association. The decision of the arbitrator must be based upon the service agreement and applicable law. The decision of the arbitrator must be reduced to writing, is final and binding except for fraud, misconduct, or errors of law, and judgment upon the decision rendered may be entered in any court having jurisdiction. In all arbitrations, the arbitrator must give effect to applicable statutes of limitation subject to limitation of actions terms in the service agreement, and has no authority to award relief in excess of what the service agreement provides or to order consolidation or class arbitrations. The arbitrator has no authority to award punitive damages in any Disputed Claim. Any such claims arising under such a service agreement must be pursued on an individual basis in accordance with the procedure noted above. Even if applicable law permits class actions or class arbitrations, the dispute resolution procedure specified here applies and Company and customer waive any rights to pursue any claim arising under the service agreement on a class basis. The arbitration will be held in a mutually agreed to location, and is final and binding. No Disputed Claim (including an Equitable Claim) may be brought more than 2 years after the time the claim first could have been brought.
    - .05 Compliance with Laws: Customer and Company shall comply with all applicable federal, state, and local laws, ordinances, regulations and codes in its performance under a service agreement, including without limitation the export, import, customs, and foreign corrupt practices laws of the United States or any country in which Customer receives equipment, software or services.
    - .06 Service Order. Customers may be required to enter into written Service Orders which contain or reference the name of Customer, a specific description of the service ordered, the rates to be charged, the duration of the services, and the applicable terms and conditions.
    - .07 Purchase Order: A Customer purchase order or similar document is evidence only of Customer's intention to purchase equipment, software and/or services. Except for a provision evidencing an intent to be bound by the terms and conditions of an agreement between Customer and Company, the terms and conditions of a Customer purchase order or similar document will be disregarded and have no force or effect; instead, the terms and conditions of the relevant agreement between Customer an Company will govern.

- II. TERMS AND CONDITIONS (CONT'D.)
  - 8. <u>MISCELLANEOUS PROVISIONS (CONT'D.)</u>
  - 8. J Other Provisions (Cont'd).
    - Independent Contractor Relationship; No Agency: The personnel of Customer and of Company are not agents or employees of the other and Customer and Company are each an independent contractor for all purposes and at all times in connection with a service agreement. Neither Customer nor Company has the right or authority to, and shall not, assume or create any obligation of any nature whatsoever on behalf of the other or bind the other in any respect whatsoever. Customer and Company each shall indemnify, hold harmless and defend the other against any liabilities, claims, losses and damages (including costs, expenses and reasonable attorneys' fees) arising out of its failure to comply with this provision and any laws, rules or regulations applicable to this provision.
    - .09 No Third-Party Beneficiaries. Except to the extent explicitly provided, any Company Signed Contract is solely for the benefit of Company and Customer and creates no rights in any third-parties not a party to that contract.
    - .10 Interpretation: No service agreement may be construed or interpreted for or against either Customer or Company because that party drafted or caused that party's legal representative to draft any of its provisions.
    - .11 Headings: The Section headings used in this tariff or a service agreement are for reference and convenience only and may not be considered in their interpretation.
    - .12 Signatures: Any requirement for a signature in a Company Customer Agreement (or an amendment to it) may be satisfied by a facsimile transmission of an original signature.

- II. TERMS AND CONDITIONS (CONT'D.)
  - 9. Telecommunication Service Priority (TSP)
  - I. DESCRIPTION:

The Telecommunications Service Priority (TSP) program is a federally-established program under which the Office of Priority Telecommunications in the Executive Office of the President prioritizes the restoration and provisioning of telecommunications services – including services to private companies and institutions — that support national security or emergency preparedness (NS/EP). The FCC defines telecommunications services under the TSP program to include the sending and receiving of signals or most any kind, by virtually any means. NS/EP services are those used to maintain a state of readiness or to respond to and manage any event or crisis (local, national, or international) that causes or could cause injury or harm to the population, damage to or loss of property, or that degrades or threatens the NS/EP posture of the United States. For telecommunications services enrolled in the program, the Company will provision and restore TSP-coded circuits, and provide TSP Special Construction services, under the terms set forth in this TSP service product description, and as required by the FCC's TSP regulations (currently at 47 CFR Part 64, Subpart D, Appendix A), and other applicable law. TSP services are in two categories: Priority Provisioning (including Emergency Provisioning and Essential Provisioning) and Priority Restoration.

- II. FEATURES: The following features are available on a per-circuit basis. A Customer may subscribe to either Emergency Provisioning or Essential Provisioning for a circuit, but may not subscribe to both.
- 1. Emergency Provisioning is provided by the Company in response to an emergency, when the Customer's need for a service is critical and must be provisioned at the earliest possible time, without regard to the cost to the Customer. In Emergency Provisioning the Company will take immediate action to allocate the resources necessary to provision circuit(s) and any related special construction assigned an Emergency Provisioning priority level as soon as possible, including dispatching personnel outside normal Company business hours.
- Essential Provisioning is provided for new essential NS/EP service that must be installed by a specific date that cannot be met using normal Company business procedures. In Essential Provisioning, the Company will adjust its resources to make its best effort to provision the circuit(s) and any related special construction assigned an Essential Provisioning priority level, by the requested service due date, based on the priority level assigned.
- 3. <u>Priority Restoration</u> designation establishes priorities for restoring NS/EP service in the event of an outage or failure of multiple services. The Company will dispatch personnel outside normal business hours if necessary to restore circuit(s) (and provide any related special construction) assigned a Priority Restoration level of 1, 2, or 3. The Company will dispatch personnel outside normal business hours to restore circuits (and provide any related special construction) assigned a Priority Restoration level of 4 or 5 only when the next business day is more than 24 hours away.

- II. TERMS AND CONDITIONS (CONT'D.)
  - 9. Telecommunication Service Priority (TSP) (Cont'd)
  - III. RATES AND CHARGES:
  - 1. Feature Charges: The following feature charges apply. Circuit-based charges set out below apply to the provisioning or restoration of circuits assigned a TSP priority level. To the extent other work is needed to provision or restore the telecommunications service, beyond the circuit itself, TSP Special Construction Charges will apply. Pricing for any services beyond circuit provisioning and restoration, and TSP Special Construction, will be negotiated by the Customer and Company on a case-by-case basis. Pricing for the TSP provisioning or restoration of services which are not provided through one or more circuits dedicated to a particular Customer also will be negotiated on a case-by-case basis.
    - 1. <u>Non-Recurring Charges:</u>
      - Emergency Provisioning and Essential Provisioning: The following one-time, per circuit charges apply for Emergency Provisioning and Essential Provisioning, depending on whether such provisioning includes local access channel coordination by the Company:
        - \$460 per circuit for circuits without local access channel coordination \$715 per circuit for circuits with one local access channel coordination \$715 per circuit for each additional local access channel coordination
      - 2. <u>Priority Restoration:</u> The following one-time per circuit charges apply for Priority Restoration, depending on whether installation of the TSP priority code includes local access channel coordination by the Company:
        - \$305 per circuit for circuits without local access channel coordination \$710 per circuit for circuits with one local access channel coordination \$710 per circuit for each additional local access channel coordination
      - 3. <u>Change Charges:</u> The following one-time per-circuit charges apply for each change to a TSP Priority level or to the service to be provisioned or restored other than Local Access Channels, depending on whether the provisioning or restoration includes Company-provided local access channel coordination:
        - \$45 per circuit for circuits without local access channel coordination \$195 per circuit for circuits with one local access channel coordination \$195 per circuit for each additional local access channel coordination

- II. TERMS AND CONDITIONS (CONT'D.)
  - 9. Telecommunication Service Priority (TSP) (Cont'd)
  - III. RATES AND CHARGES (Cont'd)
  - 1. Feature Charges (Cont'd)
    - 1. <u>Non-Recurring Charges</u> (Cont'd)
      - 4. <u>Local Access Channel Charges:</u> The following non-recurring per-circuit charges apply to each local access channel, based on the feature associated with the circuit and the state in which the channel is located:

	<u>Feature/Charge</u>		
Emergency Provisioning and Essential Provisioning	Priority Restoration	Feature Change	
\$89.17	\$85.31	\$86.82	

- 2. Monthly Recurring Charges:
  - 1. <u>Priority Restoration:</u> The following per-circuit monthly recurring charges apply for Priority Restoration:
    - \$16 for circuits without local access channel coordination
    - 16 for circuits with one local access channel coordination
    - 16 for each additional local access channel coordination
  - 2. <u>Local Access Channel Charges</u>: The following monthly recurring per-circuit charges apply to each local access channel associated with Priority Restoration, based on the state in which the channel is located:

\$0.94

- IV. TERMS AND CONDITIONS: In addition to the Guide General Terms and Conditions that apply to telecommunications service and to Internet, Enhanced and other\_non-Telecommunications products and services, the following apply:
- After being assigned a TSP Authorization Code for a service, the Customer must transmit the code to the Company via a service order.
- The Company will not accept TSP assignments or orders without an assigned TSP Authorization Code.

- II. TERMS AND CONDITIONS (CONT'D.)
  - 10. TSP Special Construction

Rates and charges for TSP Special Construction will be based on the costs incurred by the Company and may include (A) non-recurring type charges, (B) recurring type charges, (C) termination liabilities, (D) underutilization liabilities, or (E) combinations thereof.

- 1. <u>Basis for Cost Computation:</u> TSP Special Construction costs may include one or more of the following items to the extent that they are applicable:
- (A) Cost installed of the facilities to be provided, including estimated cost for rearrangements of existing facilities. Cost installed includes the cost of:
  - (1) equipment and materials provided or used,
  - (2) engineering, labor and supervision (including necessary overtime),
  - (3) transportation,
  - (4) rights-of-way, and,
  - (5) express shipping of equipment and other expedite charges deemed necessary by the Company,
- (B) Cost of maintenance,
- (C) Depreciation on the estimated cost installed of any facilities provided, based on the anticipated useful service life of the facilities with an appropriate allowance for the estimated net salvage,
- (D) Administration, taxes and uncollectible revenue on the basis of reasonable average costs for these items.
- (E) License preparation, processing and related fees,
- (F) Tariff preparation, processing and related fees,
- (G) Any other identifiable costs related to the facilities provided, and,
- (H) An amount for return and contingencies.

- II. TERMS AND CONDITIONS (CONT'D.)
  - 10. TSP Special Construction (Cont'd)
  - 2. Termination Liability: To the extent that there is no other requirement for use by the Company, a termination liability may apply for facilities specially constructed at the request of the customer to meet its requirements.
    - The termination liability period is the estimated service life of the facilities provided.
    - 2. The amounts of the maximum termination liability is equal to the estimated amounts for:
    - (A) Cost installed of the facilities provided including estimated costs for rearrangements of existing facilities and/or construction of new facilities as appropriate, less net salvage. Cost installed includes the cost of:
      - (1) equipment and materials provided or used,
      - (2) engineering, labor and supervision,
      - (3) transportation, and,
      - (4) rights-of-way
    - (B) License preparation, processing, and related fees,
    - (C) Tariff preparation, processing, and related fees, and,
    - (D) Any other identifiable costs related to the specially constructed or rearranged facilities.
    - 3. The applicable termination liability charge is calculated by multiplying the sum of the amounts determined as set forth in 2.2 preceding by a factor related to the unexpired period of liability and the discount rates for return and contingencies. The amount determined in 2.2 (A) preceding shall be adjusted to reflect the predetermined estimated net salvage, including any reuse of the facilities provided. This product is adjusted to reflect applicable taxes.

### III. OTHER CHARGES AND SURCHARGES

## 1. <u>CARRIER ACCESS CHARGE (CAC)</u>

A monthly charge of \$2.05 applies to each Multi-line Business Line presubscribed to Company service and which accesses service via switched access. A monthly charge of \$0.20 applies to each presubscribed line used by a Customer to access Company service via Local Exchange Carrier-provided Centrex, and a monthly charge of \$0.41 applies to each presubscribed line used by the Customer to access Company service via Local Exchange Carrier-provided Primary Rate Interface (PRI). A "line" is a connection, furnished by a Local Exchange Carrier, that is presubscribed to the Company for the purpose of carrying the Customer's telecommunications service traffic. The line-type determination is based upon Local Exchange Carrier-provided information.

The CAC will not be eligible to receive promotional or any other discounts.

For any Customer executing a Special Customer Arrangement (SCA) before January 1, 1998, which subsequently became effective and either: (a) expressly forecloses any increase in CAC charges, or (b) contains a provision that limits by capping any increase in CAC charges, CAC charges will not apply during: (i) the remainder of the original term of service, so long as the rates and charges, including regulations that affect rates and charges, for service are not amended; and, (ii) any optional term of service following the original term of service which the SCA permits the customer to invoke by unilateral action, so long as the rates and charges, including regulations that affect rates and charges for service are not amended. Otherwise, CAC charges will apply at the beginning of any new SCA term of service, including any month-to-month term of service, following the expiration of any original or optional service term.

In addition, if an SCA is amended to add a service (with service features) not being furnished under the SCA, CAC charges will not apply to the amended SCA so long as the requirements set forth in (i) and (ii), immediately above, continue to be met. Otherwise, CAC charges will apply at the beginning of any new SCA term of service, including any month-to-month term of service, following the expiration of any original or optional service term, or the first monthly billing period following the amendment of the SCA to add new service. If the service added is Long Distance Voice Services, including Local Network Connection, and that service is being substituted for MCI 800 Service, Option B, Option C and/or networkMCI One; and if the Customer subscribes to new, non-resold, exchange service provided by an affiliate of the Company, CAC charges will not apply if the Customer pays charges for the substituted service(s) that equal or exceed the charges for the service substituted for, after application of all discounts or credits, and independent of any new Long Distance Voice Services or Long Distance Voice Services features to which the Customer subscribes.

# III. OTHER CHARGES AND SURCHARGES (Cont'd)

## PAYPHONE USE SURCHARGE

An undiscountable charge of \$0.55 per call will apply to each call originating from a domestic payphone or associated facility used to access Company services. This charge, which is in addition to applicable call usage and other charges associated with Company service, applies to the use of the instrument used to access Company service and is unrelated to the Company service accessed from the payphone or associated facility.

The payphone use surcharge will not apply to: calls using Telecommunications Relay Service; calls originated by Customers who are certified as having qualified hearing or speech impairments; and calls placed from payphones at which the Customer pays for service by inserting coins during the progress of the call. Federal Universal Service Fund (FUSF) applies.

## 3. SPECIAL ACCESS SURCHARGE

The following Special Access Surcharge will apply as indicated in other sections of this tariff.

\$27.71

# 4. <u>CUSTOMER SUPPORT SERVICE CHARGES</u>

### A. NO FAULT FOUND DISPATCH CHARGE:

A non-recurring charge applies when a Company representative is dispatched to the Customer's premises at the request of the Customer to investigate a suspected issue with any Company service, and the Company representative responds to the dispatch and confirms the proper functioning of such Company service. This can include, but is not limited to, dispatches requested to: assist with identifying a problem which turns out to be within the scope of the Customer/vendor-maintained equipment or network; provide technical assistance with Customer-or vendor-maintained network and equipment issues that are outside the scope of the Company's responsibility; or tag Company's demarcation point and, in doing so, the Company representative finds the circuit is clearly marked.

The following charges apply per visit to the Customer's Premises, based on time of the visit:

Time of Day/Charge	Charge
Normal Working Hours	\$265
Outside of Normal Working Hours	400

For purposes of this provision, "Normal Working Hours" are defined as Monday to Friday, excluding New Year's Day, Martin Luther King Jr. Day, Presidents' Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day, Day after Thanksgiving and Christmas Day, 7 AM to 7 PM in the time zone of the Customer's premises; a visit to Customer premises which begins or ends outside of Normal Working Hours is "Outside of Normal Working Hours."

## III. OTHER CHARGES AND SURCHARGES

- 4. <u>CUSTOMER SUPPORT SERVICE CHARGES</u> (Cont'd)
  - B. <u>RECONNECTION CHARGE:</u>

For Intrastate Telecommunications services, a \$20 non-recurring charge applies per line presubscribed to the Company, excluding payphones, and per Company-provided authorization code which the Company unblocks following Call Blocking.

#### IV. FUND

With the Fund, eligible Customers will receive specific dollar amounts which may be deposited into accounts in the names of those Customers. These amounts may be utilized toward the purchase by Customers of Company and participating vendors' products and services that have been pre-approved by Company for the use of Fund dollars.

Eligible dollar amounts are defined as promotional dollars available with the subscription to (a) certain promotions of Verizon Business Services, (b) certain Option B, Option C, networkMCI One and Long Distance Voice Services promotions, and (c) certain Special Customer Arrangements. Any employee or agent of the Customer may place orders for vendors' products using the Customer's Fund account. Unless otherwise specified in this tariff, the Customer may choose to convert the balance in its Fund account to invoice credits. The invoice credit amount, upon Customer's request, will be applied on a prorata basis to the Customer's first invoice following the end of the annual period in which the Customer makes such request and in each subsequent 12-month period of the Customer's term of service.

If the Customer subscribes to and terminates Company service, any balances remaining in the Fund at the time of termination will be forfeited. The Company may choose to apply the remaining Fund as invoice credits to offset amounts owed to the Company. If the Customer's service agreement expires, the Company has the right to forfeit any remaining Fund balance 30 days from the date of expiration. Any promotional amount(s) previously used towards participating vendor products or Company invoice credits may be charged back to the Customer upon termination.

The Company may, at any time, choose to discontinue this offering, in which case it will provide notice to each participating Customer. Any Fund balance remaining upon the completion of one full billing period following notice of discontinuance will be forfeited. Any dollar amounts in the Customer's Fund account which are forfeited by the Customer will become the property of the Company.

The Fund balance will be unavailable to a Customer during any period in which the Customer has failed to comply with the Payment Arrangements specified in this tariff or satisfy a monthly annual volume commitment set forth in the Customer's Company service agreement.

### V. INDIVIDUAL CASE BASIS (ICB) ARRANGEMENTS.

Rates, terms or conditions for ICB arrangements will be developed on a case-by-case basis in response to a bona fide request from a Customer or prospective Customer to develop a competitive bid for a service offered under this tariff. Rates, terms or conditions quoted in response to such competitive requests may be different than those specified for such services in this tariff. ICB rates will be offered to the Customer in writing and on a non-discriminatory basis. The Company may also enter into ICB arrangements in response to a Customer request for special arrangements which may include engineering, installation, construction, facilities, assembly, purchase or lease of facilities, and/or other special services not otherwise offered under this tariff. Company, at its option, may provide the requested services. Appropriate recurring and/or non-recurring charges will be developed accordingly.

### VI. INTRASTATE LONG DISTANCE VOICE SERVICES

## A. PRODUCT DESCRIPTION:

An outbound, customized telecommunications service that may include an inbound service option using Business Line Access or Dedicated Access Line terminations and/or a virtual private network. Service is available for single or multi-location companies using switched, dedicated and calling card origination and switched and dedicated termination.

- 1. Intentionally Omitted
- 2. <u>Discontinuance of Service:</u> Customers who discontinue Long Distance Voice Services service will be billed at their contract rates for a period of up to thirty (30) days, after which they will be billed for any service usage at the standard tariff rates applicable to service.
- 3. Switched Digital Service: Intrastate Switched Data Service is available.

Switched Digital Services are not available to Feature Option A Customers. These are Dial-up services for transmitting data at speeds of 56 kbps and 64 kbps and at speeds in multiples of 64 kbps.

Domestic Switched Data 56 and 64 kbps services may be accessed using either: T-1 Digital Access Primary Rate Interface (PRI); PRI provided by a Local Exchange Carrier (LEC); LEC-provided Basic Rate Interface (BRI) access; Digital Data Service (DDS) local loops; or, LEC-provided Switched Digital Access (SDA). SDA access to Switched Data is available in selected locations.

Switched Digital Services at multiples of 64 kbps may be accessed through Company Primary Rate Interface (PRI) access lines, or LEC-provided PRI. The Customer is responsible for obtaining the LEC-provided PRI access line which connects the Customer's premises to the LEC's central office. The Customer is billed directly by the LEC for this access type. Switched services at multiples of 64 kbps require that T-1 local access lines be equipped with B8ZS line coding from the LEC.

- A. <u>Outbound:</u> Switched Digital offers dial-up service for transmitting data at speeds of 56 kbps and 64 kbps and at speeds in multiples of 64 kbps up to 1536 kbps between locations. Switched Digital offers H0 and H11 services, which are dial-up services for transmitting data at speeds of 384 kbps and 1536 kbps (equivalent to 6 x 64 kbps and 24 x 64 kbps, respectively).
- B. <u>Inbound:</u> For Customers subscribing to Feature Option B and Feature Option C-2, Toll Free Digital Service (inbound) offers dial-up service for transmitting data at speeds of 56 kbps and 64 kbps and at speeds in multiples of 64 kbps between locations.

- VI. INTRASTATE LONG DISTANCE VOICE SERVICES (CONT'D.)
  - A. <u>PRODUCT DESCRIPTION (CONT'D.)</u>
    - 4. Calculation of Charges
      - 4.1 <u>Call Rounding</u>: Unless otherwise specified, calls are subject to the following rounding rules, on a per-call basis, for billing purposes:

Call Type	Initial Period	Additional Increments
Outbound Service (except Operator Assisted)	18 seconds	6 seconds
Outbound Service		
Operator Assisted	60 seconds	60 seconds
Toll Free Inbound Service	18 seconds	6 seconds

- 4.2 <u>Charge Rounding</u>: If the computed charge includes a fraction of a cent, the fraction is rounded to the nearest whole cent.
- B. <u>DEFINITIONS</u>: In addition to the General Definitions, the following definitions apply:

Dedicated: A non-switched connection between a Customer Premises and an MCI Legacy Company POP.

Local Network Connection: A switched connection between a Customer's Premises and an MCI Legacy Company Point of Presence (POP).

Switched: A switched connection between a Customer Premises and a Company POP which is not provided by an MCI Legacy Company.

- C. <u>FEATURES AND OPTIONS</u>: Intrastate Long Distance Voice Services offers Features which are also offered with and inherently part of interstate Long Distance Voice Services. Rates, terms and conditions for these Features can be found in the Guide where interstate Long Distance Voice Services are described. Those features which inherently cannot be used with Intrastate Long Distance Voice Services, such as international service, are not included. Additionally, Card Access (Calling Card) is available, as described below.
  - Card Access (Calling Card): A telephone calling card issued by Company at the Customer's request, which enables the Customer or user(s) authorized by the Customer (Authorized Users) to place calls over Company's network and to have the charges for such calls billed to the Customer's account. Company offers the following types of intrastate Card Access:
- 1.1 Standard Card: Standard Card enables Customer to use intrastate outbound Long Distance Voice Services and is available via a toll-free number from any touch-tone dialing phone using DTMF (dual tone multi-frequency). Authorized Users can dial 7-digit and 10-digit private dialing plan numbers (which are numbers that the Company creates for Customer using the Company's network database and intelligent platform and that can be translated to a 10-digit United States number), and 10-digit public numbers (which are phone numbers issued by a local exchange carrier).

- VI. INTRASTATE LONG DISTANCE VOICE SERVICES (CONT'D.)
  - C. <u>FEATURES AND OPTIONS (CONT'D.)</u>
    - Card Access (Cont'd)
      - 1.1.1 <u>Standard Card Usage Charges:</u> The usage charges in the Long Distance Voice Service Usage Rate Schedule in this tariff apply for outbound intrastate Long Distance Voice Services based on the originating location.
      - 1.1.2 <u>Standard Card Surcharge:</u> The following per-call charges apply to each Standard Card call other than calls to Intrastate Long Distance Directory Assistance. This charge will not apply to Customized Message Announcement.

The following per call calling card access surcharges apply based on the Customer calling plan:

<b>Current Services</b>	Non-Current Services			
Verizon Business Services (VBS) I, II, and III	On-net	networkMCI One	Vnet	Vision
\$0.90	\$0.90	\$0.85	\$0.40	\$0.95

- 1.2 Chargeable Calling Card: Chargeable Calling Card enables the Customer to establish a fixed dollar amount available per Calling Card that is charged against Customer's account as calls are made. If an Authorized User tries to exceed the fixed amount, he/she will be unable to make calls. As with Standard Card, Authorized Users can dial 7-digit and 10-digit private dialing plan numbers, 10-digit public numbers, and numbers that terminate in the international locations listed under Standard Card—Usage Charges. Customers can establish Chargeable Calling Cards that are renewable at a previously-budgeted amount or expire after the initial amount is expended. Unless otherwise specified, standard tariff provisions and rates and charges for Calling Card apply.
  - 1.2.1 No Resale: Chargeable Calling Cards cannot be resold by Customer. Violation of this resale prohibition is deemed a material breach establishing cause for the Company's immediate termination of Customer's Chargeable Calling Cards. Customer will indemnify, defend, and hold the Company harmless from and against all claims of loss, damage, cost or expense arising from Customer's attempted or actual resale of Chargeable Calling Cards.
  - 1.2.2 Co-Branding: Customer may request that its logo or other identifying image be placed on the front of the Chargeable Calling Card. The inclusion of such logo or image is entirely within the Company's discretion, and the Company may refuse to include such logo or image. Customer will indemnify, defend, and hold the Company harmless from and against any loss, damage, cost or expense resulting from any third-party claim or allegation that any image or branding placed on the Calling Card at Customer's request violates the service mark, trademark or other intellectual property rights of any third party.

#### VII. RATES AND CHARGES

#### 1. <u>INTRASTATE ACCESS</u>:

- 1.1 Outbound Service
  - 1.1.1 Dedicated Access/Termination
  - A. <u>Intrastate T-1 Digital Access /Termination</u>: For Intrastate T-1 Digital Access described in the tariff section on Intrastate Network Services Local Access Services, Termination monthly and non-recurring charges in effect on the date of the invoice will apply.

In addition, a Customer may allow another Long Distance Voice Services Customer to terminate calls on the first Customer's network via dedicated access. This feature is available at the request of the Customer on whose network the calls are to terminate. Such requests must be submitted to the Company in writing. Other Customers authorized to make such calls must dial a private dialing plan number.

- B. <u>Intrastate Analog Access /Termination</u>: For Intrastate Analog Access, described in described in the tariff section on Intrastate Network Services Local Access Services, monthly and non-recurring charges in effect on the date of the invoice will apply.
- 1.1.2 <u>Switched Access</u>: A \$5 monthly minimum charge will apply to each Customer location using switched access. The Customer will be charged the difference between the Customer's actual usage charges (including surcharges) and \$5 per location during each monthly billing period. This charge does not apply to Customers who subscribe to Feature Option C-1.
  - A. Calling Card charges apply.
  - B. Intrastate Toll Free Remote Access: This feature is not available for Feature Option A. Allows access to the Customer's network via a Customer-specific toll free number from anywhere within the same state. This feature is available for calling within the Customer's private network, with both 7- and 10-digit dialing plans, via switched access only. This feature is not available via Card Access. Per-call surcharges apply in addition to applicable intrastate usage charges.

For Customers subscribing to Feature Option B, a \$0.25 per call surcharge will apply to each intrastate Toll Free Remote Access call.

For Customers subscribing to Feature Option C-1, a \$0.25 per call surcharge will apply to each intrastate Toll Free Remote Access call which originates over "Switched" facilities and terminates over "Switched" facilities.

For Customers subscribing to Feature Option C-1, a \$0.15 per call surcharge will apply to each intrastate Toll Free Remote Access call which originates over "Switched" facilities and terminates over "Dedicated" facilities.

# 1. <u>INTRASTATE ACCESS (CONT'D.)</u>:

#### 1.2 <u>Inbound Service</u>

- 1.2.1 Dedicated Access Termination
- A. Intrastate Analog Access charges apply, as described in described in the tariff section on Intrastate Network Services Local Access Services.
- B. Intrastate T-1 Digital Access charges apply, as defined in described in the tariff section on Intrastate Network Services Local Access Services.
- C. Other Forms of Dedicated Access: At the Customer's request and where feasible, the Company may permit the Customer to obtain other forms of access to the Customer's inbound service terminating location, or the Company may procure other forms of access to such location. All applicable recurring and non-recurring charges for such service will be calculated on an individual case basis, in accordance with the charges assessed by the Local Exchange Carrier or other access provider. The installation and monthly charges for any interface equipment associated with such access that is provided by the Company shall be calculated on an individual case basis. If the Company procures access for the Customer from the Local Exchange Carrier, the Customer's use of such access shall be in conformity with the regulations and other terms and conditions under which the Local Exchange Carrier provides such access.
- D. A \$100 per month service fee per service group for Dedicated Access Termination will apply.
- 1.2.2 <u>Business Line Termination</u>: The following non-recurring and monthly recurring charges will apply for Business Line termination:

	Charges	
	Non-Recurring Monthly	
per service group	\$0.00	\$30

- 1.2.3 Additional Charges for Inbound Service Access:
- A. <u>Special Access Surcharge</u>: The Special Access Surcharge will apply for Dedicated Access Termination for inbound service. It will apply only to Customers subscribing to Feature Option B and Feature Option C-2. The Special Access Surcharge, imposed by the Local Exchange Carrier, will not apply to those Customers who furnish the Company with an Exemption Certificate. It will apply for each voice grade equivalent channel will apply.

#### 1. <u>INTRASTATE ACCESS (CONT'D.)</u>:

Access Term Discount Plan: The following are available to Customers of Feature Option A only. Customers who order intrastate Dedicated Access Line Service from the Company or a Company affiliate may receive an Access Term Discount. Any Customer who signs a new contract for intrastate Long Distance Voice Services Option A and has intrastate Dedicated Access Service provided by the Company will automatically receive the Access Term Discount Plan on all intrastate Dedicated Access circuits. Any Customer who has not signed a Long Distance Voice Services Option A contract but has a contractual commitment to keeping its intrastate dedicated access circuits provided by the Company for at least 12 months will also receive the Access Term Discount Plan discounts on these circuits.

Other charges (e.g., Access Coordination, Central Office Connection, Network Connection, and non-recurring charges) described herein are not discountable under these plans. For purposes of the Access Term Discount Plan, intrastate Dedicated Access Line Service only includes Intrastate Analog Access, Intrastate T-1 Digital Access, and Intrastate DSO (Hubless) Access.

1.3.1 <u>Discounts:</u> The following discounts will apply to those Customers who have an Access Term Discount Plan on their dedicated access circuits. The discount applies to the Local Access Channel Monthly Recurring Channel Charges only.

For Customers who have signed a Long Distance Voice Services Option A contract, the Term Commitment of the contract will determine the discount percentage to use. For Customers who have not signed a Long Distance Voice Services Option A contract but have a contractual commitment to keeping its dedicated access circuits provided by the Company or an affiliate of the Company for at least 12 months, the length of time the Customer is committed to its dedicated access continually being provided by the Company determines the discount percentage to use.

Term Commitment (Months)	Analog Local Access Discount	DS0 (Hubless) Local Access or DDS Local Access Discount	T-1 Digital Access Discount
12 to 23	0%	0%	5%
24 to 35	4	5	10
36 to 47	5	8	20
48 to 59	6	10	22
60 or more	7	15	24

#### 2. USAGE

2.1 <u>Inbound and Outbound Service:</u> The following per-minute rates apply based on origination type:

Those on Verizon Business Services (VBS) I and VBS II plans prior to February 1, 2007:

1 Y	ear	2 Y	ear	3 Y	'ear
Switched	Dedicated or	Switched	Dedicated or	Switched	Dedicated or
(including	Local	(including	Local	(including	Local
Card Access)		Card Access)		Card Access)	
\$0.0760	\$0.0370	\$0.0750	\$0.0365	\$0.0740	\$0.0355

Those on VBS II plans effective on or after 2/01/07, and those on VBS III plans:

1 Year		2 Year		3 Year	
Switched	Dedicated or	Switched	Dedicated or	Switched	Dedicated or
(including	Local	(including	Local	(including	Local
Card Access)		Card Access)		Card Access)	
\$0.0750	\$0.0360	\$0.0740	\$0.0350	\$0.0730	\$0.0345

# 2.2 Intrastate Switched Digital Service.

2.2.1. Outbound: The following per-minute rates apply to outbound Switched Data service at speeds of 56 kbps and 64 kbps, at speeds in multiples of 64 kbps up to 1536 kbps and H0 and H11 services between intrastate locations, based on origination and termination types and speed:

Origination	Termination	Speed/Per-Mir	
<u>Type</u>	<u>Type</u>	56/64 kbps	<u>128+</u> kbps
Local Number Connection	Local Number Connection	\$0.1676	\$0.1444
Local Number Connection	Dedicated	0.1726	0.1494
Local Number Connection	Switched	0.1826	0.1594
Dedicated	Local Number Connection	0.1826	0.1594
Dedicated	Dedicated	0.1826	0.1594
Dedicated	Switched	0.1926	0.1694
Switched	Local Number Connection	0.2284	0.1999
Switched	Dedicated	0.2284	0.1999
Switched	Switched	0.2384	0.2099

- 2.2 Intrastate Switched Digital Service (Cont'd)
- 2.2.2 Inbound: The following per-minute rates apply to inbound Switched Data service at speeds of 56 kbps and 64 kbps, at speeds in multiples of 64 kbps up to 1536 kbps and H0 and H11 services between intrastate locations, based on origination and termination types and speed:

Origination	Termination	Speed/Per-Min	
<u>Type</u>	<u>Type</u>	<u>56/64 kbps</u>	<u>128+</u> <u>kbps</u>
Local Number Connection Local Number Connection	Local Number Connection Dedicated	\$0.1676 0.1826	\$0.1444 0.1594
Local Number Connection	Switched	0.2284	0.1999
Switched	Local Number Connection	0.1826	0.1594
Switched	Dedicated	0.1926	0.1694
Switched	Switched	0.2384	0.2099

#### INTRASTATE OPERATOR ASSISTED

The following charges apply to outbound Operator Assisted calls which originate and terminate via Switched (excluding Card) and Dedicated access only within the same state. For calls terminating via Local Network Connection facilities, a termination credit of \$0.01 per minute will apply. For calls originating over Local Network Connection facilities, an origination credit of \$0.045 per minute will apply.

3.1 Per Minute Charges: For intrastate calls within the classifications detailed below, (excluding collect calls and calls which are billed by a third party), the following call classifications and usage rates will apply:

#### 3.1.1 Call Classifications:

- Calls made from Company presubscribed payphones or institutional phones and which are completed with the assistance of a Company operator:
- Calls made from private residential or business phones presubscribed to the Company and which are completed with the assistance of a Company operator; and
- Calls made from private residential or business phones by Casual Callers, which are completed with the assistance of a Company operator.
- 3.1.2 Operator Assisted Per Minute Usage Charges for intrastate calls.

Mileage	Day 1 <sup>st</sup>	Day Add'l	Evening 1 <sup>st</sup> Min	Evening Add'l Min	Night/Weekend 1 <sup>st</sup> Min	Night/Weekend Add'l Minute
	Minute	Min	I IVIIII	Additiviiii	1 141111	Add I Williate
ALL	\$0.30	\$0.30	\$0.30	\$0.30	\$0.30	\$0.30

"Day" calls are from 8 am to 4:59 pm Mon - Fri; "Evening" calls are 5:00 pm to 10:59 pm, Mon-Fri, and Sun 5:00 PM - 10:59 PM; and, "Night/Weekend" calls are 11:00 PM Friday - 5:00 PM Sunday, 11:00 PM Sunday - 7:59 AM Monday, and 11:00 PM to 7:59 AM, Mon - Sun.

#### 3. <u>INTRASTATE OPERATOR ASSISTED (CONT'D.)</u>:

3.2 <u>Operator Services Surcharges</u>: The following undiscountable surcharges apply at the rates as set forth in the chart below:

Collect Person to Person - Operator assisted call wherein the caller asks if a specific person is available at the terminating number. If so, that individual decides whether to accept the call. If they do accept, then the call is completed and they will be billed for the call. If the individual requested is not home or does not accept the call, then the call is not completed.

Collect Station to Station – Customer places a call through use of Company Operator Services to reach another number and, if accepted by the terminating (called) party, the call is completed and billed to the terminating number.

1st Party Person to Person – Customer asks the Company Operator Services to place a call to reach a specific person. Charges may be billed, and the call may be completed, only if the person is available.

1st Party Station to Station – Customer places call through use of Company Operator Services to reach one number. Charges may be billed only if the call is connected.

Third Party - Customer places call through use of Company Operator Services to reach one number, but bill the call to a third number which is neither the originating nor the terminating (called) number. Charges may be billed, and the call may be completed, only if the operator gains verbal acceptance from the third party.

Operator Dialed - The Company dials the number that the Customer requests.

Call Type	Surcharge
Collect Person-to-Person	\$3.90
Collect Station-to-Station	\$2.10
1 <sup>st</sup> Party Person-to-Person	\$3.90
1 <sup>st</sup> Party Station-to-Station	\$2.10
3 <sup>rd</sup> Party	\$1.00
Operator Dialed	\$1.15

4. <u>INTRASTATE FEATURES AND FEATURE PACKAGES</u>: See Section III feature descriptions and charges.

#### 5. OPTIONAL CALLING PLANS:

5.1 Verizon Business Services III – Optional Calling Plan

The following provisions apply to Customers who subscribe to Verizon Business Services III only. The plan is available to Feature Option 1 Customers only.

- 1. LD Voice Package
- 1.1 <u>Monthly Recurring Charges:</u> Customer will pay the following flat rate monthly recurring charge, fixed for the service term, for all intrastate and interstate outbound Long Distance Voice Services calls from any single Customer site up to 20,000 minutes per month originating from a T1 or PRI purchased from Company.

Service Term (Years)	Monthly Recurring Charge
1	\$725
2	685
3	650

- 1.2 Overage Charge: An overage charge of \$0.05 applies for each minute of intrastate and interstate outbound Long Distance Voice Services calls that exceeds 20,000 in a month.
- 1.3 Terms and Conditions: Customer understands that the LD Voice Package is restricted in the following manner:
- International long distance, Inbound (toll free), and Calling Card minutes are NOT included.
- (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the LD Voice Package.
- (iii) Customer may not utilize the LD Voice Package in any call center environment or in connection with any such similar environment.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION OF THE LD VOICE PACKAGE BY VERIZON AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.1 Verizon Business Services III – Optional Calling Plan (Cont'd.)
    - 2. Verizon Business Services III Local and Long Distance
      - A. Eligibility: To be eligible for this plan, Customers:
      - must designate a Company affiliate as its exchange service carrier and intrastate calling and as its carrier for intraLATA toll calling;
      - must subscribe both to certain intrastate service as described in the appropriate intrastate Company or Company-affiliate tariff or other appropriate governing document ("Companion Intrastate Service") and to certain exchange service as described in the appropriate Companyaffiliate exchange service tariff or other appropriate governing document ("Companion Local Service") concurrent with enrollment in this plan;
      - may not receive any discounts or the benefits of any promotional offering;
      - must subscribe to service under Product Packages Guide Types 22 or 23.

The Company may cancel a Customer's service and Customer accounts, without incurring any liability, immediately, and without notice, if the Customer fails to satisfy the following conditions. The Customer may not:

- subscribe to service under this plan for more than 30 lines or 24 trunks per Customer location;
- use auto-dialers or any similar type of device in connection with service under this plan; and,
- use service under this plan in a call center environment or in connection with any similar such application.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

#### 5. OPTIONAL CALLING PLANS (CONT'D.):

- 5.1 <u>Verizon Business Services III Optional Calling Plan (Cont'd.)</u>
  - 2. Verizon Business Services III Local and Long Distance (Cont'd.)
    - B. <u>Definitions:</u> For the purposes of this plan, the following definitions apply:

Eligible Intrastate Service: Verizon Business Services III Long Distance Voice Services Feature Option A Outbound Service (excluding Calling Card) usage that originates and terminates in one state as set forth in the Companion Intrastate tariffs or other appropriate governing documents.

Eligible Interstate Usage: Verizon Business Services III Long Distance Voice Services Feature Option A Outbound Service (excluding Calling Card) usage that originates in the U.S. Mainland and Hawaii and terminates in the U.S. Mainland, Alaska, Hawaii, Puerto Rico, the U.S Virgin Islands, Guam, and CNMI.

Existing Customers: Customers who, at the time of subscription to the plan, are receiving service under this tariff.

New Customers: Customers who, at the time of subscription to the plan, are not receiving service under this tariff.

C. Monthly Recurring Charges:

A monthly recurring charge will apply for each Offering under this plan. The monthly recurring charge is in lieu of the rates and charges as set forth in Product Packages Guide Types 22 and 23 for services eligible for this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Intrastate Service and Companion Local Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per-Local T-1 or per-ISDN-PRI basis.

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.1 <u>Verizon Business Services III Optional Calling Plan (Cont'd.)</u>
      - 2. <u>Verizon Business Services III Local and Long Distance (Cont'd.)</u>
        - C. <u>Monthly Recurring Charges (Cont'd.):</u>

The following monthly recurring charges apply:

C.1 <u>Monthly Recurring Charges:</u> The chart below shows the applicable monthly recurring charges for Customers who subscribe to this plan and who currently subscribe to Companion Intrastate Service and Companion Local Service as described in the applicable Company affiliate state tariff, if any, or if not in the state tariff, in the intrastate portion of the Guide.

Offering	Charge	
Α	\$	56
В		61
C (grandfathered – see Offering C Flat with Cap above)	1,4	100
C Flat with Cap	1,3	325

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.1 <u>Verizon Business Services III Optional Calling Plan (Cont'd.)</u>
      - 2. Verizon Business Services III Local and Long Distance (Cont'd.)
        - D. <u>Benefits:</u> Customers will receive the following benefits on in lieu of the benefits set forth in Product Packages Guide Types 22 and 23 for service provided under this plan.
        - Offering A Flat with Cap: Upon installation of Companion Interstate D.1 Service and Companion Local Service, Customer will receive an allotment of 800 minutes per monthly period per Local Line that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment, Customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.
        - Offering B Flat with Cap: Upon installation of Companion Interstate D.2 Service and Companion Local Service, Customer will receive an allotment of 800 minutes per monthly period per Local Trunk that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment, Customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.1 <u>Verizon Business Services III Optional Calling Plan (Cont'd.)</u>
      - 2. Verizon Business Services III Local and Long Distance (Cont'd.)
        - D. Benefits (Cont'd.):
        - D.3 Offering C Flat with Cap: Upon installation of Companion Interstate Service and Companion Local Service, Customer will receive an allotment of 20,000 minutes per monthly period per Local T1 or ISDN-PRI that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment, Customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.
        - E. Other Benefits and Discounts: The following discounts are identical to, and shall not be in addition to, any discounts applicable to Companion Intrastate Service and Companion Local Service.
        - E.1 A Customer who subscribes to this and demonstrates to the Company's reasonable satisfaction that the Customer will accept another carrier's offer in absence of any further inducement will receive discounts, not to exceed the following amounts, on monthly recurring plan, optional feature and feature package charges in lieu of all other discounts, based on the Customer's term of service under this plan:

Term of Service (Years)	<u>Discount</u>
1	5%
2 +	15

E.2 Customers who subscribe to domestic Verizon Business Services I Long Distance Voice Services under Product Packages Guide Types 22 or 23 will receive the discounts set forth in the Product Package for domestic Verizon Business Services I Long Distance Voice Services on Overage Usage charges.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.1 <u>Verizon Business Services III Optional Calling Plan (Cont'd.)</u>
    - 2. Verizon Business Services III Local and Long Distance (Cont'd.)
      - F. Termination of Service:
      - F.1 The following provisions will apply to Customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing Customers who disconnect Companion Local Service only, plan service and Companion Intrastate Service will terminate and the Customer will be automatically re-subscribed to the service offerings to which the Customer subscribed at the time of subscription to this plan under this tariff and the Companion Intrastate Service tariff.

For existing Customers who disconnect Companion Local Service and Companion Intrastate Service, plan service and Companion Intrastate Service will terminate and the Customer will then be automatically resubscribed to the service offering under this tariff to which the Customer subscribed at the time of subscription to this plan.

For new Customers who disconnect Companion Local Service only, plan service and Companion Intrastate Service will terminate and the Customer will be automatically subscribed to Verizon Business Services I Long Distance Voice Services Feature Option A under this tariff and Verizon Business Services I Long Distance Voice Services under the Companion Intrastate Service tariff.

For new Customers who disconnect Companion Local Service and Companion Intrastate Service, plan service and Companion Intrastate Service will terminate and the Customer will be automatically subscribed to Verizon Business Services I Long Distance Voice Services Feature Option A under this tariff and Verizon Business Services I Long Distance Voice Services under the Companion Intrastate Service tariff for intraLATA intrastate service only.

- VII. RATES AND CHARGES (CONT'D.)
  - 5. <u>OPTIONAL CALLING PLANS (CONT'D.):</u>
    - 5.1 <u>Verizon Business Services III Optional Calling Plan (Cont'd.)</u>
      - 2. Verizon Business Services III Local and Long Distance (Cont'd.)
        - G. Other Conditions:
        - G.1 <u>Billing:</u> Customers subscribing to service under this plan via a Company-designated Internet site will receive Electronic Billing invoicing only.
        - G2 Customers subscribing to this plan may not receive the benefits of any other discounts, including term plan discounts, or promotions.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.1 <u>Verizon Business Services III Optional Calling Plan (Cont'd.)</u>
    - 3. <u>LD Voice 800 Minute Package for Verizon Business Lines or Trunks</u>
    - 3.1 Customer will pay the following flat rate monthly recurring charge, fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 800 minutes per month originating from a Business Line or Trunk purchased from Company.

Agreement Term (Years)	Monthly Recurring Charge
One	\$28
Two	26
Three	24

- 3.2 Overage Charge: An overage charge of \$0.05 applies for each minute of intrastate and interstate outbound Long Distance Voice Services calls that exceeds 800 in a month.
- 3.3 <u>Terms and Conditions:</u> Customer understands that the LD Voice 800 Minute Package is restricted in the following manner:
  - International LD, Inbound (toll free), and Calling Card minutes are NOT included.
  - (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 800 Minute Package.
  - (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
  - (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services III Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 800 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.1 <u>Verizon Business Services III Optional Calling Plan (Cont'd.)</u>
    - 4. LD Voice 500 Minute Package for Verizon Business Lines or Trunks
    - 4.1 Customer will pay the following flat rate monthly recurring charge, fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 500 minutes per month originating from a Business Line or Trunk purchased from Company.

Agreement Term (Years) Monthly Recurring C	
One	\$18
Two	17
Three	16

- 4.2 Overage Charge: An overage charge of \$0.05 applies for each minute of intrastate and interstate outbound Long Distance Voice Services calls that exceeds 500 in a month.
- 4.3 <u>Terms and Conditions:</u> Customer understands that the LD Voice 500 Minute Package is restricted in the following manner:
  - International LD, Inbound (toll free), and Calling Card minutes are NOT included.
  - (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 500 Minute Package.
  - (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
  - (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services III Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 500 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.1 <u>Verizon Business Services III Optional Calling Plan (Cont'd.)</u>
    - 5. <u>LD Voice 300 Minute Package for Verizon Business Lines or Trunks</u>
    - 5.1 Customer will pay the following flat rate monthly recurring charge, fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 300 minutes per month originating from a Business Line or Trunk purchased from Company.

Agreement Term (Years)	m (Years) Monthly Recurring Charge	
One	\$12	
Two	11	
Three	10	

- 5.2 Overage Charge: An overage charge of \$0.05 applies for each minute of intrastate and interstate outbound Long Distance Voice Services calls that exceeds 300 in a month.
- 5.3 Terms and Conditions: Customer understands that the LD Voice 300 Minute Package is restricted in the following manner:
  - (i) International LD, Inbound (toll free), and Calling Card minutes are NOT included.
  - (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 300 Minute Package.
  - (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
  - (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services III Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 300 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.2 <u>Verizon Business Services II Optional Calling Plan</u>

The following provisions apply to Customers who subscribe to Verizon Business Services II only. The plan is available to Feature Option 1 Customers only.

- 1. LD Voice Package
- 1.1 <u>Monthly Recurring Charges:</u> Customer will pay the following flat rate monthly recurring charge, fixed for the service term, for all intrastate and interstate outbound Long Distance Voice Services calls from any single Customer site up to 20,000 minutes per month originating from a T1 or PRI purchased from Company.

Service Term (Years)	Monthly Recurring Charge
1	\$725
2	685
3	650

- 1.2 Overage Charge: An overage charge of \$0.05 applies for each minute of intrastate and interstate outbound Long Distance Voice Services calls that exceeds 20,000 in a month.
- 1.3 Terms and Conditions: Customer understands that the LD Voice Package is restricted in the following manner:
  - (i) International long distance, Inbound (toll free), and Calling Card minutes are NOT included.
  - (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the LD Voice Package.
  - (iii) Customer may not utilize the LD Voice Package in any call center environment or in connection with any such similar environment.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION OF THE LD VOICE PACKAGE BY VERIZON AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.2 Verizon Business Services II – Optional Calling Plan (Cont'd.)
    - 2. Verizon Business Services II Local and Long Distance
      - A. Eligibility: To be eligible for this plan, Customers:
      - must designate a Company affiliate as its exchange service carrier and intrastate calling and as its carrier for intraLATA toll calling;
      - must subscribe both to certain intrastate service as described in the appropriate intrastate Company or Company-affiliate tariff or other appropriate governing document ("Companion Intrastate Service") and to certain exchange service as described in the appropriate Companyaffiliate exchange service tariff or other appropriate governing document ("Companion Local Service") concurrent with enrollment in this plan;
      - may not receive any discounts or the benefits of any promotional offering;
      - must subscribe to service under Product Packages Guide Types 18, 19, 20 or 21.

The Company may cancel a Customer's service and Customer accounts, without incurring any liability, immediately, and without notice, if the Customer fails to satisfy the following conditions. The Customer may not:

- subscribe to service under this plan for more than 30 lines or 24 trunks per Customer location;
- use auto-dialers or any similar type of device in connection with service under this plan; and,
- use service under this plan in a call center environment or in connection with any similar such application.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.2 <u>Verizon Business Services II Optional Calling Plan (Cont'd.)</u>
    - 2. Verizon Business Services II Local and Long Distance (Cont'd.)
      - B. <u>Definitions:</u> For the purposes of this plan, the following definitions apply:

Eligible Intrastate Service: Verizon Business Services II Long Distance Voice Services Feature Option A Outbound Service (excluding Calling Card) usage that originates and terminates in one state as set forth in the Companion Intrastate tariffs or other appropriate governing documents.

Eligible Interstate Usage: Verizon Business Services II Long Distance Voice Services Feature Option A Outbound Service (excluding Calling Card) usage that originates in the U.S. Mainland and Hawaii and terminates in the U.S. Mainland, Alaska, Hawaii, Puerto Rico, the U.S Virgin Islands, Guam, and CNMI.

Existing Customers: Customers who, at the time of subscription to the plan, are receiving service under this tariff or other appropriate governing document containing the Companion Intrastate Service.

New Customers: Customers who, at the time of subscription to the plan, are not receiving service under this tariff or other appropriate governing document containing the Companion Intrastate Service.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.2 <u>Verizon Business Services II Optional Calling Plan (Cont'd.)</u>
    - 2. Verizon Business Services II Local and Long Distance (Cont'd.)
      - C. Monthly Recurring Charges:

A monthly recurring charge will apply for each Offering under this plan. The monthly recurring charge is in lieu of the rates and charges as set forth in Product Packages Guide Types 18, 19, 20 and 21 for services eligible for this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Intrastate Service and Companion Local Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per-Local T-1 or per-ISDN-PRI basis

As of July 1, 2007 the existing "Offering A" is being replaced by a new "Offering A". The new offering A monthly recurring charge will apply to new and renewed Local Line service. Existing Offering A monthly recurring charges for Local Line service will continue to apply until the end of the service agreement in effect as of July 1, 2007.

As of July 1, 2007 the existing "Offering B" is being replaced by a new "Offering B". The new Offering B monthly recurring charge will apply to new and renewed Local Trunk service. Existing Offering B monthly recurring charges for Local Trunk service will continue to apply until the end of the service agreement in effect as of July 1, 2007.

As of April 1, 2007 "Offering C" is being replaced by "Offering C Flat with Cap" and the Offering C Flat with Cap monthly recurring charge will apply to new and renewed T1 or PRI service. Existing Offering C monthly recurring charges for T1 and PRI service will continue to apply until the end of the service agreement in effect as of April 1, 2007.

C.1.1 The following month recurring charges apply to Offering A and Offering B monthly recurring charges for Local Line service existing prior to July 1, 2007 and Offering C monthly recurring charges for Local Line service existing prior to April 1, 2007.

The chart below shows the applicable monthly recurring charges for Customers who subscribe to this plan and who currently subscribe to Companion Intrastate Service and Companion Local Service as described in the applicable Company affiliate state tariff, if any, or if not in the state tariff, in the intrastate portion of the Guide.

Offering	Charge
A	\$ 67
В	65
С	1,400

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.2 <u>Verizon Business Services II Optional Calling Plan (Cont'd.)</u>
      - 2. <u>Verizon Business Services II Local and Long Distance (Cont'd.)</u>
        - C. <u>Monthly Recurring Charges (Cont'd.):</u>
        - C.1.2 <u>Benefits:</u> Customers will receive the following benefits on in lieu of the benefits set forth in Product Packages Guide Types 18, 19, 20 and 21 for service provided under this plan prior to July 1, 2007 for Offering A and Offering B and prior to April 1, 2007 for Offering C.
        - i. Offering A: Prior to July 1, 2007, upon installation of interstate service and Companion Local Service, the Customer will receive unlimited Eligible Interstate Usage, Eligible Intrastate Service usage and unlimited exchange service usage as described in the Companion Intrastate Service tariffs and Companion Local Service tariffs or other appropriate governing documents.
        - ii. Offering B: Prior to July 1, 2007, upon installation of interstate service and Companion Local Service, a Customer who subscribes to Offering B will receive an allotment of 1,250 minutes per monthly period per local trunk subscribed to under this offering that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service as described in the Companion Intrastate Service tariff in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Interstate Usage that exceeds this allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment, the Customer will be charged \$0.05 per minute for Eligible Interstate Usage for the portion of the call after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.

The Customer will receive exchange service usage as described in the Companion Local Service tariffs or other appropriate governing document.

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.2 <u>Verizon Business Services II Optional Calling Plan (Cont'd.)</u>
      - 2. <u>Verizon Business Services II Local and Long Distance (Cont'd.)</u>
        - C. Monthly Recurring Charges (Cont'd.):
        - C.1.2 Benefits (Con't.):
        - iii. Offering C: Prior to April 1, 2007, upon installation of Companion Interstate Service and Companion Local Service, Customer will receive an allotment of 30,000 minutes per monthly period per Local T1 or ISDN-PRI that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment, Customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.
        - C.2 Monthly Recurring Charges as of July 1, 2007: The charts below show the applicable monthly recurring charges for Customers who subscribe to this plan and who currently subscribe to Companion Intrastate Service and Companion Local Service as described in the applicable Company affiliate state tariff, if any, or if not in the state tariff, in the intrastate portion of the Guide.

Offering	Charge	)
A	\$	56
В		61
C (grandfathered – see Offering C Flat		100
with Cap above)	1,	,400
C Flat with Cap	1,	325

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.2 <u>Verizon Business Services II Optional Calling Plan (Cont'd.)</u>
      - 2. Verizon Business Services II Local and Long Distance (Cont'd.)
        - D. <u>Benefits:</u> Customers will receive the following benefits on in lieu of the benefits set forth in Product Packages Guide Types 18, 19, 20 and 21 for service provided under this plan.
        - Offering A Flat with Cap: As of July 1, 2007, existing "Offering A" is D.1 being replaced by "Offering A Flat with Cap". Existing Offering A monthly recurring charges for Local Trunk service will continue to apply until the end of the service agreement in effect as of July 1, 2007. Upon installation of Companion Interstate Service and Companion Local Service, Customer will receive an allotment of 800 minutes per monthly period per Local Line that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment, Customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.2 <u>Verizon Business Services II Optional Calling Plan (Cont'd.)</u>
      - 2. Verizon Business Services II Local and Long Distance (Cont'd.)
        - D. Benefits (Cont'd.):
        - **D.2** Offering B Flat with Cap: As of July 1, 2007, the existing "Offering B" is being replaced by a new "Offering B Flat with Cap". Existing Offering B monthly recurring charges for Local Trunk service will continue to apply until the end of the service agreement in effect as of July 1, 2007. The Upon installation of Companion Interstate Service and Companion Local Service, Customer will receive an allotment of 800 minutes per monthly period per Local Trunk that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment, Customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.
        - D.3 Offering C Flat with Cap: Upon installation of Companion Interstate Service and Companion Local Service, Customer will receive an allotment of 20,000 minutes per monthly period per Local T1 or ISDN-PRI that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment. Customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.

- VII. RATES AND CHARGES (CONT'D.)
- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.2 <u>Verizon Business Services II Optional Calling Plan (Cont'd.)</u>
    - 2. Verizon Business Services II Local and Long Distance (Cont'd.)
      - E. <u>Other Benefits and Discounts</u>: The following discounts are identical to, and shall not be in addition to, any discounts applicable to Companion Intrastate Service and Companion Local Service.
      - E.1 A Customer who subscribes to this and demonstrates to the Company's reasonable satisfaction that the Customer will accept another carrier's offer in absence of any further inducement will receive discounts, not to exceed the following amounts, on monthly recurring plan, optional feature and feature package charges in lieu of all other discounts, based on the Customer's term of service under this plan:

Term of Service (Years)	<u>Discount</u>	
1	5%	
2 +	15	

E.2 Customers who subscribe to domestic Verizon Business Services II Long Distance Voice Services under Product Packages Guide Types 18, 19, 20 or 21 will receive the discounts set forth in the Product Package for domestic Verizon Business Services II Long Distance Voice Services on Overage Usage charges.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.2 <u>Verizon Business Services II Optional Calling Plan (Cont'd.)</u>
    - 2. <u>Verizon Business Services II Local and Long Distance (Cont'd.)</u>
      - F. Termination of Service:
      - F.1 The following provisions will apply to Customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing Customers who disconnect Companion Local Service only, plan service and Companion Intrastate Service will terminate and the Customer will be automatically re-subscribed to the service offerings to which the Customer subscribed at the time of subscription to this plan under this tariff and the Companion Intrastate Service tariff.

For existing Customers who disconnect Companion Local Service and Companion Intrastate Service, plan service and Companion Intrastate Service will terminate and the Customer will then be automatically resubscribed to the service offering under this tariff to which the Customer subscribed at the time of subscription to this plan.

For new Customers who disconnect Companion Local Service only, plan service and Companion Intrastate Service will terminate and the Customer will be automatically subscribed to Verizon Business Services I Long Distance Voice Services Feature Option A under this tariff and Verizon Business Services I Long Distance Voice Services under the Companion Intrastate Service tariff.

For new Customers who disconnect Companion Local Service and Companion Intrastate Service, plan service and Companion Intrastate Service will terminate and the Customer will be automatically subscribed to Verizon Business Services I Long Distance Voice Services Feature Option A under this tariff and Verizon Business Services I Long Distance Voice Services under the Companion Intrastate Service tariff for intraLATA intrastate service only.

- G. Other Conditions:
- G.1 <u>Billing:</u> Customers subscribing to service under this plan via a Company-designated Internet site will receive Electronic Billing invoicing only.
- G2 Customers subscribing to this plan may not receive the benefits of any other discounts, including term plan discounts, or promotions.

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.2 <u>Verizon Business Services II Optional Calling Plan (Cont'd.)</u>
      - 3. <u>LD Voice 800 Minute Package for Verizon Business Lines or Trunks</u>
      - 3.1 Customer will pay the following flat rate monthly recurring charge, fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 800 minutes per month originating from a Business Line or Trunk purchased from Company.

Agreement Term (Years)	Monthly Recurring Charge
One	\$28
Two	26
Three	24

- 3.2 Overage Charge: An overage charge of \$0.05 applies for each minute of intrastate and interstate outbound Long Distance Voice Services calls that exceeds 800 in a month.
- 3.3 <u>Terms and Conditions:</u> Customer understands that the LD Voice 800 Minute Package is restricted in the following manner:
  - International LD, Inbound (toll free), and Calling Card minutes are NOT included.
  - (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 800 Minute Package.
  - (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
  - (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services II Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 800 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.2 Verizon Business Services II Optional Calling Plan (Cont'd.)
    - 4. LD Voice 500 Minute Package for Verizon Business Lines or Trunks
    - 4.1 Customer will pay the following flat rate monthly recurring charge, fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 500 minutes per month originating from a Business Line or Trunk purchased from Company.

Agreement Term (Years)	Monthly Recurring Charge
One	\$18
Two	17
Three	16

- 4.2 Overage Charge: An overage charge of \$0.05 applies for each minute of intrastate and interstate outbound Long Distance Voice Services calls that exceeds 500 in a month.
- 4.3 <u>Terms and Conditions:</u> Customer understands that the LD Voice 500 Minute Package is restricted in the following manner:
  - International LD, Inbound (toll free), and Calling Card minutes are NOT included.
  - (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 500 Minute Package.
  - (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
  - (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services II Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 500 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

- LD Voice 300 Minute Package for Verizon Business Lines or Trunks
- 5.1 Customer will pay the following flat rate monthly recurring charge, fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 300 minutes per month originating from a Business Line or Trunk purchased from Company.

Agreement Term (Years)	erm (Years) Monthly Recurring Charge	
One	\$12	
Two	11	
Three	10	

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.2 Verizon Business Services II Optional Calling Plan (Cont'd.)
    - 5. LD Voice 300 Minute Package for Verizon Business Lines or Trunks (Cont'd.)
    - 5.2 Overage Charge: An overage charge of \$0.05 applies for each minute of intrastate and interstate outbound Long Distance Voice Services calls that exceeds 300 in a month.
    - 5.3 Terms and Conditions: Customer understands that the LD Voice 300 Minute Package is restricted in the following manner:
      - (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
      - (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 300 Minute Package.
      - (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
      - (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services II Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 300 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.3 Verizon Business Services I Optional Calling Plans

The following provisions apply to Customers who subscribe to Verizon Business Services I only. These plans are available to Feature Option 1 Customers only.

- 1. LD Voice Package
- 1.1 <u>Monthly Recurring Charges:</u> Customer will pay the following flat rate monthly recurring charge, fixed for the service term, for all intrastate and interstate outbound Long Distance Voice Services calls from any single Customer site up to 20,000 minutes per month originating from a T1 or PRI purchased from Company.

Monthly Recurring Charge
\$725
685
650

- 1.2 <u>Overage Charge:</u> An overage charge of \$0.05 applies for each minute of intrastate and interstate outbound Long Distance Voice Services calls that exceeds 20,000 in a month.
- 1.3 Terms and Conditions: Customer understands that the LD Voice Package is restricted in the following manner:
  - (i) International long distance, Inbound (toll free), and Calling Card minutes are NOT included.
  - (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the LD Voice Package.
  - (iii) Customer may not utilize the LD Voice Package in any call center environment or in connection with any such similar environment.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION OF THE LD VOICE PACKAGE BY VERIZON AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.3 Verizon Business Services I – Optional Calling Plans (Cont'd.)
    - 2. Verizon Business Services I Local and Long Distance: The Verizon Business Services I Local and Long Distance optional calling plan is not available to Customers after March 1, 2005.
      - A. Eligibility: To be eligible for this plan, Customers:
      - must designate a Company affiliate as its exchange service carrier and intrastate calling and as its carrier for intraLATA toll calling;
      - must subscribe both to certain intrastate service as described in the appropriate intrastate Company or Company-affiliate tariff or other appropriate governing document ("Companion Intrastate Service") and to certain exchange service as described in the appropriate Companyaffiliate exchange service tariff or other appropriate governing document ("Companion Local Service") concurrent with enrollment in this plan;
      - may not receive any discounts or the benefits of any promotional offering;
      - must subscribe to service under Product Packages Guide Types 13, 14, 15 or 16.

The Company may cancel a Customer's service and Customer accounts, without incurring any liability, immediately, and without notice, if the Customer fails to satisfy the following conditions. The Customer may not:

- subscribe to service under this plan for more than 30 lines or 24 trunks per Customer location;
- use auto-dialers or any similar type of device in connection with service under this plan; and,
- use service under this plan in a call center environment or in connection with any similar such application.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

#### 5. OPTIONAL CALLING PLANS (CONT'D.):

#### 5.3 <u>Verizon Business Services I – Optional Calling Plans (Cont'd.)</u>

#### Verizon Business Services I Local and Long Distance (Cont'd.):

#### B. <u>Definitions:</u> For the purposes of this plan, the following definitions apply:

Eligible Intrastate Service: Verizon Business Services I Long Distance Voice Services Feature Option A Outbound Service (excluding Calling Card) usage that originates and terminates in one state as set forth in the Companion Intrastate tariffs or other appropriate governing documents.

Eligible Interstate Usage: Verizon Business Services I Long Distance Voice Services Feature Option A Outbound Service (excluding Calling Card) usage that originates in the U.S. Mainland and Hawaii and terminates in the U.S. Mainland, Alaska, Hawaii, Puerto Rico, the U.S Virgin Islands, Guam, and CNMI.

Existing Customers: Customers who, at the time of subscription to the plan, are receiving service under this tariff or other appropriate governing document containing the Companion Intrastate Service.

New Customers: Customers who, at the time of subscription to the plan, are not receiving service under this tariff or other appropriate governing document containing the Companion Intrastate Service.

#### C. Monthly Recurring Charges:

A monthly recurring charge will apply for each Offering under this plan. The monthly recurring charge is in lieu of the rates and charges as set forth in Product Packages Guide Types 13, 14, 15 and 16 for services eligible for this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Intrastate Service and Companion Local Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per-Local T-1 or per-ISDN-PRI basis.

The chart below shows the applicable monthly recurring charges for Customers who subscribe to this plan and who currently subscribe to Companion Intrastate Service and Companion Local Service as described in the applicable Company affiliate state tariff, if any, or if not in the state tariff, in the intrastate portion of the Guide.

Offering	Charge	
Α	\$45	(Winter Park Serving
		Areas)
Α	55	(Tampa Serving Areas)
Α	50	(all other Serving Areas
		in Florida)
В	65	
С	1.400	

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.3 <u>Verizon Business Services I Optional Calling Plans (Cont'd.)</u>
      - 2. Verizon Business Services I Local and Long Distance (Cont'd.):
        - D. <u>Benefits:</u> Customers will receive the following benefits on in lieu of the benefits set forth in Product Packages Guide Types 13, 14, 15 and 16 for service provided under this plan.
        - D.1 Offering A: Upon installation of interstate service and Companion Local Service, the Customer will receive unlimited Eligible Interstate Usage, Eligible Intrastate Service usage and unlimited exchange service usage as described in the Companion Intrastate Service tariffs and Companion Local Service tariffs or other appropriate governing documents.
        - D.2 Offering B: Upon installation of interstate service and Companion Local Service, a Customer who subscribes to Offering B will receive an allotment of 1,250 minutes per monthly period per local trunk subscribed to under this offering that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service as described in the Companion Intrastate Service tariff in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Interstate Usage that exceeds this allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment, the Customer will be charged \$0.05 per minute for Eligible Interstate Usage for the portion of the call after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.

The Customer will receive exchange service usage as described in the Companion Local Service tariffs or other appropriate governing document.

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.3 <u>Verizon Business Services I Optional Calling Plans (Cont'd.)</u>
      - 2. Verizon Business Services I Local and Long Distance (Cont'd.):
        - D. Benefits (Cont'd.):
        - D.3 Offering C: Upon installation of Companion Interstate Service and Companion Local Service, Customer will receive an allotment of 30,000 minutes per monthly period per Local T1 or ISDN-PRI that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of Customer's monthly allotment and ends after completion of the allotment. Customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment. For Customers enrolling in this plan after the first day of a billing cycle. whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.
        - E. Other Benefits and Discounts: The following discounts are identical to, and shall not be in addition to, any discounts applicable to Companion Intrastate Service and Companion Local Service.
        - E.1 A Customer who subscribes to this and demonstrates to the Company's reasonable satisfaction that the Customer will accept another carrier's offer in absence of any further inducement will receive discounts, not to exceed the following amounts, on monthly recurring plan, optional feature and feature package charges in lieu of all other discounts, based on the Customer's term of service under this plan:

Term of Service (Years)	<u>Discount</u>
1	5%
2 +	15%

E.2 Customers who subscribe to domestic Verizon Business Services I Long Distance Voice Services under Product Packages Guide Types 13, 14, 15 or 16 will receive the discounts set forth in the Product Package for domestic Verizon Business Services I Long Distance Voice Services on Overage Usage charges.

## VII. RATES AND CHARGES (CONT'D.)

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.3 <u>Verizon Business Services I Optional Calling Plans (Cont'd.)</u>
    - Verizon Business Services I Local and Long Distance (Cont'd.):
      - F. Termination of Service:
      - F.1 The following provisions will apply to Customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing Customers who disconnect Companion Local Service only, plan service and Companion Intrastate Service will terminate and the Customer will be automatically re-subscribed to the service offerings to which the Customer subscribed at the time of subscription to this plan under this tariff and the Companion Intrastate Service tariff.

For existing Customers who disconnect Companion Local Service and Companion Intrastate Service, plan service and Companion Intrastate Service will terminate and the Customer will then be automatically resubscribed to the service offering under this tariff to which the Customer subscribed at the time of subscription to this plan.

For new Customers who disconnect Companion Local Service only, plan service and Companion Intrastate Service will terminate and the Customer will be automatically subscribed to Verizon Business Services I Long Distance Voice Services Feature Option A under this tariff and Verizon Business Services I Long Distance Voice Services under the Companion Intrastate Service tariff.

For new Customers who disconnect Companion Local Service and Companion Intrastate Service, plan service and Companion Intrastate Service will terminate and the Customer will be automatically subscribed to Verizon Business Services I Long Distance Voice Services Feature Option A under this tariff and Verizon Business Services I Long Distance Voice Services under the Companion Intrastate Service tariff for intraLATA intrastate service only.

- G. Other Conditions:
- G.1 <u>Billing:</u> Customers subscribing to service under this plan via a Company-designated Internet site will receive Electronic Billing invoicing only.
- G.2 Customers subscribing to this plan may not receive the benefits of any other discounts, including term plan discounts, or promotions.

#### VII. RATES AND CHARGES (CONT'D.)

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.3 <u>Verizon Business Services I – Optional Calling Plans (Cont'd.)</u>
    - 3. Verizon Business Services I Local and Long Distance Line Solution
      - A. Eligibility: To be eligible for this plan, the Customer:
      - must designate the Company as its exchange service and intrastate telecommunications service carrier;
      - must subscribe both to certain intrastate service as described in the appropriate intrastate Company tariff or other appropriate governing document ("Companion Intrastate Service") and to certain exchange service as described in the appropriate Company exchange service tariff or other appropriate governing document ("Companion Local Service") concurrent with enrollment in this plan;
      - must subscribe to service under Product Packages Guide Types 13, 14, 15 or 16, and,
      - may not receive any discounts or the benefits of any promotional offerina.

The Company may cancel a Customer's service and Customer accounts, without incurring any liability, immediately, and without notice, if the Customer fails to satisfy the following conditions. The Customer may not:

- subscribe to service under this plan for more than 30 lines per Customer location;
- use auto-dialers or any similar type of device in connection with service under this plan; and,
- use service under this plan in a call center environment or in connection with any similar such application.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

### VII. RATES AND CHARGES (CONT'D.)

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.3 <u>Verizon Business Services I Optional Calling Plans (Cont'd.)</u>
    - 3. Verizon Business Services I Local and Long Distance Line Solution (Cont'd.)
      - B. <u>Definitions:</u> For the purposes of this plan, the following definitions apply:

Eligible Intrastate Service: Verizon Business Services I Long Distance Voice Services Feature Option A Outbound Service (excluding Calling Card) usage that originates and terminates in one state as set forth in the Companion Intrastate tariffs or other appropriate governing documents.

Eligible Interstate Usage: Verizon Business Services I Long Distance Voice Services Feature Option A Outbound Service (excluding Calling Card) usage that originates in the U.S. Mainland and Hawaii and terminates in the U.S. Mainland, Alaska, Hawaii, Puerto Rico, the U.S Virgin Islands, Guam, and CNMI.

Existing Customers: Customers who, at the time of subscription to the plan, are receiving service under this tariff and the tariff or other appropriate governing document containing the Companion Intrastate Service.

Local Line: A single, voice-grade communications channel, including an Automatic Number Identification (ANI.)

New Customers: Customers who, at the time of subscription to the plan, are not receiving service under this tariff and the tariff or other appropriate governing document containing the Companion Intrastate Service.

- C. <u>Availability:</u> Companion Intrastate Service and Companion Local Service is available where an MCI Legacy company provides local exchange service in this state.
- D. Monthly Recurring Charges: A \$60 monthly recurring charge will apply. The monthly recurring charge is in lieu of the rates and charges as set forth in Product Packages Guide Types 13, 14, 15 and 16 for services eligible for this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Intrastate Service and Companion Local Service.
- E. <u>Benefits:</u> Customers will receive the following benefits in lieu of the benefits set forth in Product Packages Guide Types 13, 14, 15 and 16 for service provided under this plan:

Upon installation of interstate service and Companion Local Service, the Customer will receive unlimited Eligible Interstate Usage, Eligible Intrastate Service usage and exchange service usage as described in the Companion Intrastate Service tariffs and Companion Local Service tariffs or other appropriate governing documents.

- VII. RATES AND CHARGES (CONT'D.)
  - 5. OPTIONAL CALLING PLANS (CONT'D.):
    - 5.3 Verizon Business Services I Optional Calling Plans (Cont'd.)
      - 3. Verizon Business Services I Local and Long Distance Line Solution (Cont'd.)
        - F. Other Benefits and Discounts: The following discounts are identical to, and shall not be in addition to, any discounts applicable to Companion Intrastate Service and Companion Local Service.
        - F.1 A Customer who subscribes to this plan and demonstrates to the Company's reasonable satisfaction that the Customer will accept another carrier's offer in absence of any further inducement will receive discounts, not to exceed the following amounts, on monthly recurring plan, optional feature and feature package charges in lieu of all other discounts, based on the Customer's term of service under this plan:

Term of Service (Years)	Discount
1	5%
2 +	15

F.2 Customers who subscribe to domestic Verizon Business Services I Long Distance Voice Services under Product Packages Guide Types 13, 14, 15 or 16 will receive the discounts set forth in the Product Package for domestic Verizon Business Services I Long Distance Voice Services on Overage Usage charges.

## VII. RATES AND CHARGES (CONT'D.)

- 5. OPTIONAL CALLING PLANS (CONT'D.):
  - 5.3 <u>Verizon Business Services I Optional Calling Plans (Cont'd.)</u>
    - 3. Verizon Business Services I Local and Long Distance Line Solution (Cont'd.)
      - G. <u>Termination of Service:</u> The following provisions will apply to Customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing Customers who disconnect Companion Local Service only, plan service and Companion Intrastate Service will terminate and the Customer will be automatically re-subscribed to the service offerings to which the Customer subscribed at the time of subscription to this plan under this tariff and the Companion Intrastate Service tariff.

For existing Customers who disconnect Companion Local Service and Companion Intrastate Service, plan service and Companion Intrastate Service will terminate and the Customer will then be automatically resubscribed to the service offering under this tariff to which the Customer subscribed at the time of subscription to this plan.

For new Customers who disconnect Companion Local Service only, plan service and Companion Intrastate Service will terminate and the Customer will be automatically subscribed to Verizon Business Services I Long Distance Voice Services Feature Option A under this tariff and Verizon Business Services I Long Distance Voice Services under the Companion Intrastate Service tariff.

For new Customers who disconnect Companion Local Service and Companion Intrastate Service, plan service and Companion Intrastate Service will terminate and the Customer will be automatically subscribed to Verizon Business Services I Long Distance Voice Services Feature Option A under this tariff and Verizon Business Services I Long Distance Voice Services under the Companion Intrastate Service tariff for intraLATA intrastate service only.

- H. Other Conditions:
- H.1 <u>Billing:</u> Customers subscribing to service under this plan via a Company-designated Internet site will receive Electronic Billing invoicing only.

### VII. RATES AND CHARGES (CONT'D.)

#### 6. INTRASTATE LONG DISTANCE DIRECTORY ASSISTANCE

Except as otherwise specified in this tariff, an undiscountable per call charge as set forth in the following table will apply to each call requesting Directory Assistance for a intrastate telephone number A charge will apply to each Directory Assistance call irrespective of whether the requested telephone number can be furnished. Directory Assistance calls will not count toward, nor be calculated as part of, applicable volume discounts. Directory Assistance calls may be placed with the assistance of an operator.

Intrastate Directory Assistance Per-Call Charge
\$1.40

A credit will apply to each Directory Assistance call if the Customer notifies a Company Service representative that he or she: (1) experienced poor transmission or was cut-off during the call; (2) was given an incorrect telephone number; or (3) inadvertently misdialed and reached Directory Assistance using the wrong area code.

Presubscribed Customers who have been certified in writing as being unable to access or use a manual directory because of a visual or other physical impairment will be eligible to receive a Directory Assistance credit against the per-call charge (and any applicable operator-assistance surcharges) by dialing Area Code + 555-1212. A licensed physician, optometrist, appropriate federal or state agency, or appropriate approved private agency must make certification on a form available upon request from the Company. When completed, the written certification must be returned to a designated company service center or, at the Company's election, delivered to the Customer's Local Exchange Carrier.

Credits may be used by: 1) the visually or otherwise physically impaired customer; 2) an organization established specifically for the purpose of assisting the visually or otherwise physically impaired; or 3) a business where all owner(s) and/or employees of the business on the premises at which a call originates have been certified as visually or otherwise physically impaired. Credits will apply only to Directory Assistance calls made by dialing Area Code + 555-1212 from the telephone number of the certified visually or otherwise physically impaired person or organization. Only one telephone number per location is entitled to this credit. A maximum of fifty Directory Assistance calls per monthly billing period will be eligible for crediting, and no single credit for a Directory Assistance call may exceed \$5.00.

A Customer may request that the Directory Assistance operator complete an intrastate telephone call to a number obtained in a Directory Assistance call. For any call completed, the Customer will be charged the additional undiscountable Operator Dialed surcharge set forth in the Operator Services Surcharges tariff section above, plus the applicable per-minute rate for the Company service to which the customer is presubscribed. This feature is not available to those accessing Directory Assistance by dialing 1010, plus a Company CIC, or for calls to toll-free, 500, 700, 900 and 976 numbers.

- 1. <u>SERVICE DESCRIPTION</u>: Intrastate Network Services Local Access Service is provided by Company. Intrastate Network Services Local Access Service is a point-to-point service which provides the Customer access to Company network services through a circuit from the Customer's designated premises to the Company's Point-of-Presence (POP). The connections of the circuit may, or may not, be provided by Company facilities but the circuit itself is provided by the Company. Intrastate Network Services Local Access Service may be provided in the following arrangements:
  - 1.1. Intrastate Analog Local Access
  - 1.2. Intrastate DS0 (Hubless) Access
  - 1.3. Intrastate T1 Digital Access
  - 1.4. Intrastate DS-3 Local Access
  - 1.5. Intrastate Sonet Access
- 1.1 <u>Intrastate Analog Local Access</u>
  - 1.1.1 Product Description: Intrastate Analog Local Access is provided by the Company.
    Intrastate Analog Local Access is a service that provides channels with voice frequency transmission capability in the nominal frequency range of 300 to 3000 Hz and may be terminated over two-wire or four-wire.
  - 1.1.2. Features and Options:
    - 1.1.2.1 Signaling: The following monthly recurring and installation charges will apply, per local access channel.

<u>Monthly</u>	Installation
\$20.00	\$0.00

1.1.2.2 Data Conditioning: The following monthly recurring and installation charges will apply, per local access channel.

<u>Monthly</u>	<u>Installation</u>
\$5.00	2.00

1.1.2.3 Access Integration Option: A form of access which allows Customers to utilize their dedicated access lines to carry traffic for both an inbound and an outbound service over the same channels. This feature is available in conjunction with inbound-Dedicated Termination only, MCI Vision, Prism I, Prism II, and Option 3.

<u>Monthly</u>	<u>Installation</u>
\$0.00	\$0.00

- 1.1.2.4 Access Pricing Plans (APP). APP, as described below in Section 2.1, is available to Pre-Verizon Business Services (VBS) I Customers only.
- 1.1.2.5 Intrastate Access Term Discounts, as described below in Section 2.2, are available to Pre-Verizon Business Services I Customers only.

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.1 <u>Intrastate Analog Local Access (Cont'd.)</u>
  - 1.1.3. Rates and Charges: Where rates or charges are specific to a particular pricing program (i.e., Verizon Business Services (VBS) I, II, or III) those rates and charges only apply where a Customer has enrolled in that pricing program. Rates and charges which are not specific to a particular pricing program apply to all Customers.
    - 1.1.3.1 Monthly Recurring Channel Charges:
      - 1.1.3.1.1 The following monthly recurring charges apply to circuits installed under VBS III, based on the Serving Wire Center of the circuit:

Wire Center	Rate						
ABDLFLXA	\$200.00	BRTNFLXX	\$170.00	DBRYFLMA	\$200.00	FTLDFLMR	\$140.00
ALCHFLXA	\$160.00	BSHNFLXA	\$250.00	DDCYFLXA	\$280.00	FTLDFLOA	\$150.00
ALFAFLXA	\$200.00	BVHLFLXA	\$260.00	DELDFLMA	\$220.00	FTLDFLPL	\$170.00
ALFRFLXA	\$200.00	BWLGFLXA	\$240.00	DESTFLXA	\$280.00	FTLDFLSF	\$140.00
ALSPFLXA	\$190.00	BYBHFLMA	\$160.00	DFSPFLXA	\$330.00	FTLDFLSG	\$180.00
ALTRFLXA	\$260.00	BYSHFLXA	\$190.00	DLBHFLKP	\$190.00	FTLDFLSU	\$180.00
ALVAFLXA	\$240.00	CCBHFLAF	\$200.00	DLBHFLMA	\$160.00	FTLDFLTA	\$140.00
ANMRFLXA	\$210.00	CCBHFLMA	\$170.00	DLSPFLMA	\$230.00	FTLDFLWN	\$190.00
APPKFLXA	\$240.00	CDKYFLMA	\$310.00	DNDNFLXA	\$190.00	FTMBFLXA	\$230.00
ARCDFLXA	\$270.00	CFLDFLMA	\$260.00	DNLNFLWM	\$250.00	FTMDFLXA	\$240.00
ARCHFLMA	\$210.00	CFVLFLXA	\$240.00	DRBHFLDF	\$170.00	FTMYFLMA	\$270.00
ASTRFLXA	\$270.00	CHLKFLXA	\$300.00	DRBHFLMA	\$170.00	FTMYFLXA	\$150.00
AVPKFLXA	\$150.00	CHPLFLJA	\$150.00	DUNDFLXA	\$240.00	FTMYFLXB	\$210.00
BAKRFLXA	\$250.00	CHSWFLXA	\$280.00	DWPKFLXA	\$170.00	FTMYFLXC	\$210.00
BARTFLXA	\$270.00	CHTHFLXA	\$350.00	DYBHFLFN	\$160.00	FTPRFLMA	\$150.00
BAYUFLXA	\$180.00	CITRFLXA	\$140.00	DYBHFLMA	\$150.00	FTWBFLXA	\$270.00
BBPKFLXA	\$310.00	CLHNFLXA	\$140.00	DYBHFLMN	\$160.00	FTWBFLXB	\$270.00
BCGRFLXA	\$250.00	CLMTFLXA	\$300.00	DYBHFLOB	\$190.00	FTWBFLXC	\$250.00
BCRTFLBT	\$150.00	CLTNFLXA	\$310.00	DYBHFLOS	\$200.00	FTWHFLXA	\$140.00
BCRTFLMA	\$140.00	CLWRFLXA	\$150.00	DYBHFLPO	\$190.00	GCSPFLCN	\$230.00
BCRTFLSA	\$170.00	CNSDFLXA	\$180.00	DYBHFLWS	\$200.00	GCVLFLMA	\$210.00
BGPIFLMA	\$330.00	CNTMFLLE	\$210.00	EGLLFLBG	\$180.00	GDRGFLXA	\$240.00
BHPKFLXA	\$160.00	COCOFLMA	\$150.00	EGLLFLIH	\$190.00	GENVFLMA	\$210.00
BKVLFLJF	\$310.00	COCOFLME	\$180.00	ENWDFLXA	\$370.00	GLBRFLMC	\$200.00
BLDWFLMA	\$220.00	CPCRFLXA	\$220.00	EORNFLMA	\$210.00	GLDLFLXA	\$330.00
BLGLFLMA	\$210.00	CPCRFLXB	\$220.00	ESTSFLXA	\$270.00	GLGCFLXA	\$270.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.1 Intrastate Analog Local Access (Cont'd.)

1.1.3.1 Monthly Recurring Channel Charges: (Cont'd)
1.1.3.1.1 Circuits installed under VBS III (Cont'd)

Wire Center	Rate						
BLTWFLXA	\$340.00	CPHZFLXA	\$260.00	ETVLFLAA	\$280.00	GLRDFLXA	\$220.00
BLVWFLXA	\$230.00	CRCYFLXA	\$140.00	EVRGFLXA	\$320.00	GNBOFLXA	\$340.00
BNFYFLXA	\$200.00	CRRVFLXA	\$270.00	FHSDFLXA	\$160.00	GNDYFLXA	\$160.00
BNNLFLMA	\$230.00	CRVWFLXA	\$280.00	FLBHFLMA	\$220.00	GNVLFLXA	\$280.00
BNSPFLXA	\$250.00	CRWDFLXA	\$170.00	FLRHFLXA	\$140.00	GNWDFLXA	\$230.00
BORAFLXA	\$160.00	CSCYFLBA	\$290.00	FRBHFLFP	\$240.00	GRETFLXA	\$340.00
BRBAFLXA	\$180.00	CSLBFLXA	\$240.00	FRPTFLXA	\$320.00	GSVLFLAN	\$150.00
BRFRFLXA	\$150.00	CTDLFLXA	\$200.00	FRSTFLXA	\$340.00	GSVLFLMA	\$150.00
BRJTFLXA	\$280.00	CYGRFLXA	\$220.00	FTGRFLMA	\$200.00	GSVLFLNW	\$150.00
BRKRFLXA	\$130.00	CYLKFLXA	\$220.00	FTLDFLCR	\$170.00	GVLDFLXA	\$290.00
BRNDFLXA	\$170.00	CYLKFLXB	\$220.00	FTLDFLCY	\$160.00	HAVNFLMA	\$340.00
BRSNFLMA	\$230.00	DBRYFLDL	\$200.00	FTLDFLJA	\$180.00	HBSDFLMA	\$220.00
HDSNFLXA	\$240.00	JCVLFLWF	\$140.00	MDBGFLPM	\$220.00	MTLDFLAP	\$190.00
HGLDFLXA	\$230.00	JCVMFLLI	\$190.00	MDSNFLXA	\$300.00	MTLDFLXA	\$150.00
HGSPFLXA	\$150.00	JCVMFLUF	\$190.00	MIAMFLAE	\$140.00	MTVRFLXA	\$270.00
HLNVFLMA	\$240.00	JNGSFLXA	\$180.00	MIAMFLAL	\$160.00	MXVLFLMA	\$220.00
HLRDFLXA	\$150.00	JPTRFLGN	\$230.00	MIAMFLAP	\$150.00	MYCYFLXA	\$280.00
HLWDFLED	\$210.00	JPTRFLMA	\$190.00	MIAMFLBA	\$150.00	NDADFLAA	\$160.00
HLWDFLHA	\$150.00	JSPRFLXA	\$160.00	MIAMFLBC	\$180.00	NDADFLAC	\$190.00
HLWDFLMA	\$170.00	KGLKFLXA	\$240.00	MIAMFLBR	\$170.00	NDADFLBR	\$150.00
HLWDFLPE	\$150.00	KNVLFLXA	\$310.00	MIAMFLCA	\$150.00	NDADFLGG	\$160.00
HLWDFLWH	\$190.00	KSSMFLXA	\$250.00	MIAMFLDB	\$190.00	NDADFLOL	\$150.00
HMSPFLXA	\$280.00	KSSMFLXB	\$250.00	MIAMFLDI	\$140.00	NFMYFLXA	\$210.00
HMSTFLEA	\$200.00	KSSMFLXD	\$240.00	MIAMFLFL	\$170.00	NFMYFLXB	\$220.00
HMSTFLHM	\$170.00	KYHGFLMA	\$230.00	MIAMFLGR	\$140.00	NGBHFLXA	\$180.00
HMSTFLNA	\$210.00	KYLRFLLS	\$260.00	MIAMFLHL	\$150.00	NKLRFLMA	\$240.00
HMSTFLTE	\$310.00	KYLRFLMA	\$270.00	MIAMFLIC	\$200.00	NNPLFLXA	\$260.00
HNCYFLXA	\$250.00	KYSTFLXA	\$200.00	MIAMFLKE	\$190.00	NPLSFLXC	\$270.00
HNCYFLXN	\$280.00	KYWSFLMA	\$240.00	MIAMFLME	\$150.00	NPLSFLXD	\$270.00
HOWYFLXA	\$280.00	LBLLFLXA	\$260.00	MIAMFLNM	\$170.00	NPRCFLXA	\$220.00
HSNGFLXA	\$130.00	LDLKFLXA	\$250.00	MIAMFLNS	\$160.00	NRPTFLXA	\$330.00
HTISFLMA	\$190.00	LEE FLXA	\$310.00	MIAMFLOL	\$150.00	NRSDFLXA	\$170.00
HWTHFLMA	\$220.00	LGBKFLXA	\$190.00	MIAMFLPB	\$150.00	NSBHFLMA	\$160.00
HYPKFLXA	\$170.00	LHACFLXA	\$230.00	MIAMFLPL	\$140.00	NWBYFLMA	\$210.00
IMKLFLXA	\$260.00	LKALFLXA	\$230.00	MIAMFLRR	\$150.00	OCALFLXA	\$150.00
INLKFLXA	\$380.00	LKBNFLXB	\$170.00	MIAMFLSH	\$190.00	OCALFLXB	\$210.00
INRKFLXX	\$190.00	LKBRFLXA	\$230.00	MIAMFLSO	\$150.00	OCALFLXC	\$210.00
INTRFLXA	\$130.00	LKBTFLXA	\$140.00	MIAMFLUJ	\$170.00	OCALFLXJ	\$220.00
INTWFLXA	\$270.00	LKCYFLMA	\$160.00	MIAMFLWD	\$200.00	OCNFFLXA	\$240.00
INVRFLXA	\$260.00	LKHLFLXA	\$220.00	MIAMFLWM	\$150.00	OJUSFLTL	\$240.00
IONAFLXA	\$230.00	LKLDFLXA	\$190.00	MIANFLWK	\$160.00	OKCBFLXA	\$290.00
ISLMFLMA	\$290.00	LKLDFLXE	\$150.00	MIANFLYI	\$140.00	OKHLFLMA	\$250.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.1 Intrastate Analog Local Access (Cont'd.)

1.1.3.1 Monthly Recurring Channel Charges: (Cont'd)
1.1.3.1.1 Circuits installed under VBS III (Cont'd)

Wire Center	Rate						
JAY FLMA	\$260.00	LKLDFLXN	\$260.00	MIAQFL06	\$160.00	OKLWFLXA	\$240.00
JCBHFLAB	\$190.00	LKMRFLHE	\$200.00	MIATFLAD	\$140.00	OLDSFLXA	\$190.00
JCBHFLMA	\$150.00	LKMRFLMA	\$220.00	MICCFLBB	\$220.00	OLTWFLLN	\$270.00
JCBHFLSP	\$190.00	LKPCFLXA	\$240.00	MLBRFLMA	\$150.00	ORCYFLXA	\$240.00
JCVLFLAI	\$210.00	LKWLFLXA	\$280.00	MLBYFLXA	\$270.00	ORCYFLXC	\$210.00
JCVLFLAR	\$150.00	LKWLFLXE	\$340.00	MLRSFLXA	\$140.00	ORLDFL60	\$140.00
JCVLFLBW	\$160.00	LLMNFLXA	\$170.00	MLTNFLRA	\$220.00	ORLDFLAP	\$150.00
JCVLFLCL	\$140.00	LNLKFLXA	\$230.00	MNDRFLAV	\$150.00	ORLDFLCL	\$140.00
JCVLFLFC	\$190.00	LRGOFLXA	\$170.00	MNDRFLLO	\$160.00	ORLDFLLU	\$140.00
JCVLFLGH	\$210.00	LRVLFLXA	\$160.00	MNDRFLLW	\$200.00	ORLDFLMA	\$140.00
JCVLFLIA	\$180.00	LSBGFLXA	\$160.00	MNLKFLXA	\$250.00	ORLDFLOE	\$140.00
JCVLFLJT	\$160.00	LUTZFLXA	\$210.00	MNSNFLMA	\$260.00	ORLDFLPC	\$140.00
JCVLFLLF	\$160.00	LVOKFLXA	\$130.00	MNTIFLXA	\$260.00	ORLDFLPH	\$150.00
JCVLFLNO	\$150.00	LWTYFLXA	\$240.00	MOISFLXA	\$300.00	ORLDFLSA	\$170.00
JCVLFLOG	\$160.00	LYHNFLOH	\$160.00	MOLNFLXA	\$190.00	ORLDFLSO	\$160.00
JCVLFLOW	\$160.00	MALNFLXA	\$230.00	MRDCFLXA	\$260.00	ORLDFLWR	\$160.00
JCVLFLRV	\$150.00	MAYOFLXA	\$170.00	MRHNFLXA	\$300.00	ORLDFLXH	\$140.00
JCVLFLSJ	\$150.00	MCINFLXA	\$140.00	MRNNFLXA	\$190.00	ORLEFLGV	\$190.00
JCVLFLSM	\$150.00	MCLNFLXA	\$270.00	MRTHFLVE	\$350.00	ORLEFLKT	\$280.00
JCVLFLWC	\$150.00	MCNPFLMA	\$210.00	MTDRFLXA	\$280.00	ORLFFLEJ	\$280.00

- SERVICE DESCRIPTION (CONT'D.)
- 1.1 Intrastate Analog Local Access (Cont'd.)

1.1.3.1 Monthly Recurring Channel Charges: (Cont'd)

1.1.3.1.1 Circuits installed under VBS III (Cont'd)

Wire Center	Rate						
ORLFFLHX	\$280.00	RAFRFLXA	\$150.00	TAMPFLEM	\$180.00	WNPKFLTL	\$160.00
ORLFFLQR	\$280.00	RSKNFLXA	\$220.00	TAMPFLKO	\$180.00	WNPKFLXA	\$150.00
ORPKFLMA	\$200.00	RYHLFLXA	\$230.00	TAMPFLXA	\$180.00	WNPKFLXE	\$160.00
ORPKFLRW	\$200.00	SARKFLXA	\$210.00	TAMPFLXE	\$160.00	WPBHFL58	\$150.00
ORSPFLXA	\$150.00	SBNGFLXA	\$220.00	TAMPFLXX	\$160.00	WPBHFLAN	\$150.00
OSPRFLXA	\$240.00	SBSTFLFE	\$210.00	TAMQFLHU	\$280.00	WPBHFLGA	\$150.00
OVIDFLCA	\$200.00	SBSTFLMA	\$210.00	TAMQFLPM	\$280.00	WPBHFLGR	\$190.00
PACEFLPV	\$200.00	SCPKFLXA	\$230.00	TAMQFLRY	\$280.00	WPBHFLHH	\$150.00
PAHKFLMA	\$270.00	SEKYFLXA	\$220.00	TAMQFLTA	\$280.00	WPBHFLLE	\$190.00
PANCFLXA	\$260.00	SGBEFLXA	\$200.00	TAMSFLCZ	\$280.00	WPBHFLRB	\$150.00
PCBHFLNT	\$150.00	SGBHFLXA	\$330.00	THNTFLXA	\$210.00	WPBHFLRP	\$200.00
PKCYFLXA	\$240.00	SGKYFLMA	\$310.00	TLCHFLXA	\$270.00	WSSDFLXA	\$130.00
PLCSFLMA	\$210.00	SHLMFLXA	\$270.00	TLHSFLXA	\$150.00	WSTVFLXA	\$220.00
PLMTFLXA	\$210.00	SKWYFLXA	\$180.00	TLHSFLXB	\$250.00	WWSPFLHI	\$290.00
PLSLFLXA	\$170.00	SLHLFLXA	\$230.00	TLHSFLXC	\$250.00	WWSPFLSH	\$300.00
PLTKFLMA	\$160.00	SLSPFLXA	\$210.00	TLHSFLXD	\$250.00	YBCTFLXA	\$130.00
PMBHFLCS	\$160.00	SMNLFLXA	\$170.00	TLHSFLXE	\$240.00	YNFNFLMA	\$230.00
PMBHFLED	\$160.00	SNANFLXA	\$290.00	TLHSFLXF	\$270.00	YNTWFLMA	\$280.00
PMBHFLFE	\$150.00	SNDSFLXA	\$250.00	TLHSFLXG	\$290.00	YULEFLMA	\$230.00
PMBHFLMA	\$150.00	SNFRFLMA	\$160.00	TLHSFLXH	\$270.00	ZLSPFLXA	\$240.00
PMBHFLNP	\$150.00	SNISFLXA	\$240.00	TMTRFLXA	\$220.00	ZPHYFLXA	\$190.00
PMBHFLTA	\$150.00	SNRSFLXA	\$310.00	TRENFLMA	\$250.00		
PMPKFLMA	\$210.00	SNSNFLXA	\$280.00	TRSPFLXA	\$210.00		
PNCRFLXA	\$240.00	SNSPFLXA	\$230.00	TTVLFLMA	\$180.00		
PNCYFLCA	\$190.00	SPBGFLXA	\$160.00	TVRSFLXA	\$270.00		
PNCYFLMA	\$150.00	SPBGFLXS	\$190.00	UMTLFLXA	\$270.00		
PNGRFLXA	\$250.00	SPCPFLXA	\$260.00	UNVRFLXA	\$190.00		
PNISFLXA	\$240.00	SPRGFLXA	\$210.00	VENCFLXA	\$300.00		
PNLNFLXA	\$350.00	SRSTFLXA	\$150.00	VENCFLXS	\$310.00		
PNLSFLXA	\$170.00	SSDSFLXA	\$220.00	VERNFLMA	\$220.00		
PNSCFLBL	\$160.00	SSPRFLXA	\$260.00	VLPRFLXA	\$290.00		
PNSCFLFP	\$150.00	STAGFLBS	\$240.00	VLPRFLXB	\$290.00		
PNSCFLHC	\$200.00	STAGFLMA	\$210.00	VRBHFLBE	\$190.00		
PNSCFLPB	\$220.00	STAGFLSH	\$240.00	VRBHFLMA	\$150.00		
PNSCFLWA	\$160.00	STAGFLWG	\$230.00	WALDFLXA	\$130.00		
PNVDFLMA	\$200.00	STCDFLXA	\$270.00	WCHLFLXA	\$240.00		
POINFLXA	\$310.00	STGRFLXA	\$190.00	WELKFLMA	\$210.00		

- SERVICE DESCRIPTION (CONT'D.)
- 1.1 Intrastate Analog Local Access (Cont'd.)

1.1.3.1 Monthly Recurring Channel Charges: (Cont'd)
1.1.3.1.1 Circuits installed under VBS III (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
PRRNFLCC	\$200.00	STMKFLXA	\$250.00	WHSPFLXA	\$130.00
PRRNFLMA	\$200.00	STRKFLXA	\$250.00	WIMMFLXA	\$230.00
PRRYFLXA	\$490.00	STRTFLMA	\$190.00	WLBRFLXA	\$130.00
PRSHFLXA	\$260.00	SVSPFLXA	\$220.00	WLCHFLXA	\$230.00
PRSNFLFD	\$250.00	SVSSFLXA	\$230.00	WLCRFLXA	\$170.00
PSDNFLXA	\$180.00	SWTHFLXA	\$160.00	WLHLFLXA	\$350.00
PTCTFLXA	\$270.00	SYHSFLCC	\$210.00	WLSTFLXA	\$230.00
PTCYFLXA	\$240.00	TAFBFLXA	\$230.00	WLWDFLXA	\$240.00
PTSJFLXA	\$360.00	TAMPFL52	\$170.00	WNDRFLXA	\$230.00
PTSLFLMA	\$190.00	TAMPFLCM	\$180.00	WNGRFLXA	\$240.00
PTSLFLSO	\$150.00	TAMPFLCO	\$180.00	WNHNFLBU	\$220.00
QNCYFLXA	\$340.00	TAMPFLDM	\$180.00	WNHNFLXC	\$190.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.1 <u>Intrastate Analog Local Access (Cont'd.)</u>
  - 1.1.3.1 Monthly Recurring Channel Charges: (Cont'd)
    - 1.1.3.1.2 The following monthly recurring charges apply for circuits installed on or after May 1, 2002 that are not assessed VBS III Monthly Recurring Local Loop Charges, based on the Serving Wire Center of the circuit:

Wire Center	Rate						
ABDLFLXA	\$63.80	BWLGFLXA	\$124.30	DNLNFLWM	\$129.10	FTWBFLXA	\$63.80
ALCHFLXA	\$189.95	BYBHFLAK	\$113.10	DRBHFLDF	\$98.90	FTWBFLXB	\$97.25
ALFAFLXA	\$106.70	BYBHFLMA	\$113.10	DRBHFLMA	\$97.25	FTWBFLXC	\$106.70
ALFRFLXA	\$134.15	BYSHFLXA	\$105.50	DUNDFLXA	\$109.90	FTWHFLXA	\$199.25
ALSPFLXA	\$103.85	CCBHFLAF	\$105.50	DWPKFLXA	\$239.55	GCSPFLCN	\$119.50
ALTHFLXA	\$109.90	CCBHFLMA	\$105.50	DYBHFLFN	\$98.90	GCVLFLMA	\$163.60
ALTRFLXA	\$114.70	CDKYFLMA	\$177.55	DYBHFLMA	\$63.80	GDRGFLXA	\$124.30
ALVAFLXA	\$119.50	CFLDFLMA	\$146.55	DYBHFLOB	\$100.55	GENVFLMA	\$117.90
ANMRFLXA	\$117.90	CFVLFLXA	\$124.30	DYBHFLOS	\$106.70	GLBRFLMC	\$109.90
APLCFLXA	\$169.80	CHLKFLXA	\$172.90	DYBHFLPO	\$102.20	GLDLFLXA	\$158.95
APPKFLXA	\$111.50	CHPLFLJA	\$149.65	DYBHFLWS	\$63.80	GLGCFLXA	\$143.45
ARCDFLXA	\$148.10	CHSWFLXA	\$155.85	EGLLFLBG	\$98.90	GLRDFLXA	\$105.50
ARCHFLMA	\$116.30	CHTHFLXA	\$127.50	EGLLFLIH	\$100.55	GNBOFLXA	\$124.30
ARPNFLXA	\$177.55	CITRFLXA	\$117.90	ENWDFLXA	\$137.25	GNDYFLXA	\$102.20
ASTRFLXA	\$148.10	CLHNFLXA	\$122.70	EORNFLMA	\$122.70	GNVLFLXA	\$154.30
AVPKFLXA	\$63.80	CLMTFLXA	\$168.25	ESPNFLXA	\$169.80	GNWDFLXA	\$140.35
BAKRFLXA	\$135.70	CLTNFLXA	\$183.75	ESTSFLXA	\$148.10	GRETFLXA	\$134.15
BARTFLXA	\$109.90	CLWRFLXA	\$63.80	ETVLFLAA	\$360.00	GSVLFLAN	\$63.80
BAYUFLXA	\$103.85	CNSDFLXA	\$100.55	EVRGFLXA	\$188.40	GSVLFLMA	\$63.80
BBPKFLXA	\$129.10	CNTMFLLE	\$108.30	FHSDFLXA	\$106.70	GSVLFLNW	\$102.20
BCGRFLXA	\$132.60	COCOFLMA	\$63.80	FLBHFLMA	\$122.70	GSVLFLSS	\$93.95
BCRTFLBB	\$100.55	COCOFLME	\$97.25	FLRHFLXA	\$149.65	GVLDFLXA	\$165.15
BCRTFLBT	\$100.55	CPCRFLXA	\$105.50	FRBHFLFP	\$132.30	HAVNFLMA	\$154.30
BCRTFLMA	\$63.80	CPCRFLXB	\$102.20	FRPTFLXA	\$137.25	HBSDFLMA	\$113.10
BCRTFLSA	\$103.85	CPHZFLXA	\$140.35	FRSTFLXA	\$134.15	HDSNFLXA	\$135.70
BGPIFLMA	\$259.70	CRBLFLXA	\$165.15	FTGRFLMA	\$116.30	HGLDFLXA	\$102.20
BHPKFLXA	\$100.55	CRCYFLXA	\$146.55	FTLDFLCR	\$98.90	HGSPFLXA	\$196.15
BKVLFLJF	\$165.15	CRRVFLXA	\$146.55	FTLDFLCY	\$100.55	HLNVFLMA	\$119.50
BLDWFLMA	\$122.70	CRVWFLXA	\$132.30	FTLDFLJA	\$102.20	HLRDFLXA	\$137.25
BLGLFLMA	\$152.75	CRWDFLXA	\$108.30	FTLDFLMR	\$63.80	HLWDFLED	\$102.20
BLTWFLXA	\$63.80	CSCYFLBA	\$166.70	FTLDFLOA	\$63.80	HLWDFLHA	\$100.55

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.1 <u>Intrastate Analog Local Access (Cont'd.)</u>
  - 1.1.3.1 Monthly Recurring Channel Charges: (Cont'd)

1.1.3.1.2 Circuits installed on or after May 1, 2002 (Cont'd)

Wire Center	Rate						
BLVWFLXA	\$108.30	CSLBFLXA	\$105.50	FTLDFLOV	\$95.60	HLWDFLMA	\$102.20
BNFYFLXA	\$151.20	CTDLFLXA	\$141.90	FTLDFLPL	\$97.25	HLWDFLPE	\$103.85
BNNLFLMA	\$127.50	CYGRFLXA	\$106.70	FTLDFLSF	\$117.90	HLWDFLWH	\$100.55
BNSPFLXA	\$127.50	CYLKFLXA	\$106.70	FTLDFLSG	\$106.70	HMSPFLXA	\$151.20
BORAFLXA	\$219.40	CYLKFLXB	\$106.70	FTLDFLSU	\$98.90	HMSTFLEA	\$135.70
BRBAFLXA	\$105.50	DBRYFLDL	\$109.90	FTLDFLTA	\$63.80	HMSTFLHM	\$135.70
BRFRFLXA	\$216.30	DBRYFLMA	\$106.70	FTLDFLTN	\$93.95	HMSTFLNA	\$130.70
BRJTFLXA	\$122.70	DDCYFLXA	\$180.65	FTLDFLWN	\$111.50	HMSTFLTE	\$137.25
BRKRFLXA	\$119.50	DELDFLMA	\$127.50	FTMBFLXA	\$116.30	HNCYFLXA	\$109.90
BRNDFLXA	\$106.70	DESTFLXA	\$102.20	FTMDFLXA	\$125.90	HNCYFLXN	\$114.70
BRSNFLMA	\$130.70	DFSPFLXA	\$148.10	FTMYFLMA	\$102.20	HOWYFLXA	\$154.30
BRSTFLXA	\$98.90	DLBHFLKP	\$106.70	FTMYFLXA	\$63.80	HSFRFLXA	\$117.90
BRTNFLXX	\$109.90	DLBHFLMA	\$106.70	FTMYFLXB	\$98.90	HSNGFLXA	\$119.50
BSHNFLXA	\$148.10	DLSPFLMA	\$125.90	FTMYFLXC	\$97.25	HTISFLMA	\$102.20
BVHLFLXA	\$63.80	DNDNFLXA	\$100.55	FTPRFLMA	\$63.80	HWTHFLMA	\$117.90
HYPKFLXA	\$93.95	LBLLFLXA	\$137.25	MIAMFLGR	\$95.60	NDADFLGG	\$63.80
IMKLFLXA	\$141.90	LDLKFLXA	\$129.10	MIAMFLHL	\$106.70	NDADFLOL	\$97.25
INLKFLXA	\$143.45	LGBKFLXA	\$105.50	MIAMFLIC	\$105.50	NFMYFLXA	\$97.25
INRKFLXX	\$103.85	LHACFLXA	\$116.30	MIAMFLKE	\$106.70	NFMYFLXB	\$100.55
INTRFLXA	\$154.30	LKALFLXA	\$98.90	MIAMFLME	\$63.80	NGBHFLXA	\$106.70
INTWFLXA	\$122.70	LKBNFLXB	\$116.30	MIAMFLNM	\$98.90	NKLRFLMA	\$145.00
INVRFLXA	\$134.15	LKBRFLXA	\$103.85	MIAMFLNS	\$98.90	NNPLFLXA	\$134.15
IONAFLXA	\$111.50	LKBTFLXA	\$163.60	MIAMFLOL	\$100.55	NPLSFLXC	\$151.20
ISLMFLMA	\$194.60	LKCYFLMA	\$185.30	MIAMFLPB	\$100.55	NPLSFLXD	\$146.55
JAY FLMA	\$140.35	LKHLFLXA	\$119.50	MIAMFLPL	\$106.70	NPRCFLXA	\$129.10
JCBHFLAB	\$109.90	LKLDFLXA	\$100.55	MIAMFLRR	\$105.50	NRPTFLXA	\$134.15
JCBHFLMA	\$113.10	LKLDFLXE	\$177.55	MIAMFLSH	\$100.55	NRSDFLXA	\$97.25
JCBHFLSP	\$106.70	LKLDFLXN	\$105.50	MIAMFLSO	\$113.10	NSBHFLMA	\$116.30
JCVLFLAI	\$97.25	LKMRFL01	\$63.80	MIAMFLWD	\$117.90	NWBYFLMA	\$121.10
JCVLFLAR	\$102.20	LKMRFLHE	\$63.80	MIAMFLWM	\$102.20	OCALFLDB	\$63.80
JCVLFLBW	\$103.85	LKMRFLMA	\$63.80	MIAMFLYJ	\$119.50	OCALFLXA	\$63.80
JCVLFLCL	\$63.80	LKMYFLMA	\$103.85	MIANFLPV	\$105.50	OCALFLXB	\$97.25
JCVLFLFC	\$102.20	LKPCFLXA	\$130.70	MIANFLWK	\$105.50	OCALFLXC	\$97.25
JCVLFLGH	\$102.20	LKWLFLXA	\$119.50	MIANFLYI	\$97.25	OCALFLXJ	\$102.20
JCVLFLIA	\$109.90	LKWLFLXE	\$132.60	MIAQFL06	\$97.25	OCNFFLXA	\$116.30
JCVLFLJB	\$63.80	LKWOFLIY	\$220.67	MIATFLAD	\$98.90	OJUSFLTL	\$98.90
JCVLFLJT	\$98.90	LLMNFLXA	\$98.90	MICCFLBB	\$122.70	OKCBFLXA	\$168.25
JCVLFLLF	\$100.55	LNLKFLXA	\$129.10	MLBRFLMA	\$179.10	OKHLFLMA	\$134.15
JCVLFLNO	\$102.20	LRGOFLXA	\$98.90	MLBYFLXA	\$111.50	OKLWFLXA	\$117.90
JCVLFLOG	\$63.80	LRHLFLXA	\$152.75	MLRSFLXA	\$121.10	OLDSFLXA	\$109.90
JCVLFLOW	\$106.70	LRVLFLXA	\$228.70	MLTNFLRA	\$114.70	OLTWFLLN	\$154.30

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.1 <u>Intrastate Analog Local Access (Cont'd.)</u>
  - 1.1.3.1 Monthly Recurring Channel Charges: (Cont'd)

## 1.1.3.1.2 Circuits installed on or after May 1, 2002 (Cont'd)

Wire Center	Rate						
JCVLFLRV	\$98.90	LSBGFLXA	\$138.80	MNDRFLAV	\$63.80	ORCYFLXA	\$114.70
JCVLFLSJ	\$102.20	LUTZFLXA	\$114.70	MNDRFLLO	\$98.90	ORCYFLXC	\$117.90
JCVLFLSM	\$95.60	LVOKFLXA	\$216.30	MNDRFLLW	\$105.50	ORLDFL60	\$63.80
JCVLFLWC	\$105.50	LWTYFLXA	\$141.90	MNLKFLXA	\$135.70	ORLDFLAP	\$102.20
JCVLFLWF	\$93.95	LYHNFLOH	\$63.80	MNSNFLMA	\$141.90	ORLDFLCL	\$95.60
JCVMFLED	\$105.50	MALNFLXA	\$149.65	MNTIFLXA	\$132.60	ORLDFLLU	\$108.30
JCVMFLLI	\$100.55	MAYOFLXA	\$236.45	MOISFLXA	\$168.25	ORLDFLMA	\$63.80
JCVMFLLR	\$93.95	MCINFLXA	\$119.50	MOLNFLXA	\$240.00	ORLDFLOE	\$63.80
JCVMFLUF	\$103.85	MCLNFLXA	\$135.70	MRDCFLXA	\$140.35	ORLDFLPC	\$102.20
JNGSFLXA	\$228.70	MCNPFLMA	\$109.90	MRHNFLXA	\$171.35	ORLDFLPH	\$102.20
JPTRFLGN	\$200.00	MDBGFLPM	\$125.90	MRNNFLXA	\$132.60	ORLDFLSA	\$103.85
JPTRFLMA	\$119.50	MDSNFLXA	\$174.45	MRTHFLVE	\$231.80	ORLDFLXH	\$93.95
JSPRFLXA	\$213.20	MIAMFLAC	\$95.60	MTDRFLXA	\$154.30	ORLEFLGV	\$102.20
KGLKFLXA	\$140.35	MIAMFLAE	\$98.90	MTLDFLAP	\$102.20	ORPKFLMA	\$106.70
KNVLFLXA	\$137.25	MIAMFLAL	\$95.60	MTLDFLBR	\$102.20	ORPKFLRW	\$113.10
KSSMFLXA	\$121.10	MIAMFLAP	\$98.90	MTLDFLTC	\$103.85	ORSPFLXA	\$132.30
KSSMFLXB	\$117.90	MIAMFLBA	\$97.25	MTLDFLXA	\$103.85	OSPRFLXA	\$109.90
KSSMFLXD	\$117.90	MIAMFLBC	\$95.60	MTVRFLXA	\$122.70	OVIDFLCA	\$108.30
KYHGFLMA	\$125.90	MIAMFLBR	\$100.55	MXVLFLMA	\$129.10	PACEFLPV	\$106.70
KYLRFLLS	\$168.25	MIAMFLCA	\$108.30	MYCYFLXA	\$130.70	PAHKFLMA	\$149.65
KYLRFLMA	\$177.55	MIAMFLDA	\$97.25	NDADFLAA	\$97.25	PANCFLXA	\$138.80
KYSTFLXA	\$114.70	MIAMFLDB	\$105.50	NDADFLAC	\$97.25	PBGRFLEZ	\$108.30
KYWSFLMA	\$298.45	MIAMFLFL	\$95.60	NDADFLBR	\$98.90	PCBHFLNT	\$111.50

**RYHLFLXA** 

\$168.25 | TAMPFLDM

## VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)

- 1. SERVICE DESCRIPTION (CONT'D.)
- 1.1 Intrastate Analog Local Access (Cont'd.)
  - 1.1.3.1 Monthly Recurring Channel Charges: (Cont'd)

1.1.3.1.2 Circuits installed on or after May 1, 2002 (Cont'd)

		T	_	T			
Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
PKCYFLXA	\$106.70	SARKFLXA	\$97.25	TAMPFLEM	\$230.00	WLWDFLXA	\$129.10
PLBYFLAO	\$97.25	SBNGFLXA	\$108.30	TAMPFLKO	\$124.30	WNDRFLXA	\$109.90
PLCSFLMA	\$134.15	SBSTFLFE	\$117.90	TAMPFLUT	\$103.85	WNGRFLXA	\$113.10
PLMTFLXA	\$117.90	SBSTFLMA	\$113.10	TAMPFLXA	\$157.40	WNHNFLXC	\$100.55
PLSLFLXA	\$113.10	SCPKFLXA	\$113.10	TAMPFLXE	\$102.20	WNPKFLTL	\$100.55
PLTKFLMA	\$132.60	SEKYFLXA	\$100.55	TAMPFLXX	\$63.80	WNPKFLXA	\$98.90
PMBHFLCS	\$105.50	SGBEFLXA	\$103.85	TAMQFLHU	\$119.50	WNPKFLXE	\$100.55
PMBHFLDR	\$97.25	SGBHFLXA	\$138.80	TAMQFLPM	\$109.90	WPBHFL58	\$63.80
PMBHFLED	\$95.60	SGKYFLMA	\$270.55	TAMQFLRY	\$165.15	WPBHFLAN	\$63.80
PMBHFLFE	\$63.80	SHLMFLXA	\$97.25	TAMQFLSZ	\$103.85	WPBHFLGA	\$105.50
PMBHFLJK	\$95.60	SKWYFLXA	\$97.25	TAMQFLTA	\$63.80	WPBHFLGR	\$108.30
PMBHFLMA	\$102.20	SLHLFLXA	\$117.90	TAMSFLZK	\$106.70	WPBHFLHH	\$100.55
PMBHFLNP	\$102.20	SLSPFLXA	\$102.20	THBHFLXA	\$125.90	WPBHFLLE	\$103.85
PMBHFLTA	\$100.55	SMNLFLXA	\$97.25	THNTFLXA	\$113.10	WPBHFLRB	\$102.20
PMPKFLMA	\$143.45	SNANFLXA	\$183.75	TLCHFLXA	\$169.80	WPBHFLRP	\$109.90
PNCRFLXA	\$122.70	SNDSFLXA	\$124.30	TLHSFLXA	\$63.80	WSSDFLXA	\$97.25
PNCYFLCA	\$100.55	SNFRFLMA	\$102.20	TLHSFLXB	\$97.25	WSTVFLXA	\$152.75
PNCYFLMA	\$63.80	SNISFLXA	\$125.90	TLHSFLXC	\$97.25	WWHTFLXA	\$130.70
PNGRFLXA	\$130.70	SNRSFLXA	\$127.50	TLHSFLXD	\$97.25	WWSPFLHI	\$176.00
PNISFLXA	\$117.90	SNSNFLXA	\$149.65	TLHSFLXE	\$93.95	WWSPFLSH	\$182.20
PNLNFLXA	\$162.05	SNSPFLXA	\$122.70	TLHSFLXF	\$103.85	YBCTFLXA	\$95.60
PNLSFLXA	\$100.55	SPBGFLXA	\$63.80	TLHSFLXG	\$108.30	YNFNFLMA	\$121.10
PNSCFLBL	\$103.85	SPBGFLXS	\$100.55	TLHSFLXH	\$102.20	YNTWFLMA	\$148.10
PNSCFLFP	\$63.80	SPRGFLXA	\$102.20	TMTRFLXA	\$135.70	YULEFLMA	\$125.90
PNSCFLHC	\$102.20	SRSTFLXA	\$63.80	TRENFLMA	\$138.80	ZLSPFLXA	\$124.30
PNSCFLPB	\$121.10	SSDSFLXA	\$100.55	TRSPFLXA	\$113.10	ZPHYFLXA	\$127.50
PNSCFLUP	\$103.85	SSPRFLXA	\$135.70	TTVLFLMA	\$121.10		
PNSCFLWA	\$105.50	STAGFLBS	\$103.85	TVRSFLXA	\$148.10		
PNVDFLMA	\$109.90	STAGFLMA	\$63.80	UMTLFLXA	\$143.45		
POINFLXA	\$125.90	STAGFLSH	\$102.20	UNVRFLXA	\$106.70		
PRRNFLCC	\$117.90	STAGFLWG	\$109.90	VENCFLXA	\$121.10		
PRRNFLMA	\$116.30	STCDFLXA	\$127.50	VENCFLXS	\$127.50		
PRRYFLXA	\$166.70	STGRFLXA	\$105.50	VERNFLMA	\$134.15		
PRSHFLXA	\$121.10	STMKFLXA	\$125.90	VLPRFLXA	\$108.30		
PRSNFLFD	\$135.70	STRKFLXA	\$151.20	VLPRFLXB	\$113.10		
PSDNFLXA	\$102.20	STRTFLMA	\$63.80	VRBHFLBE	\$97.25		
PTCTFLXA	\$135.70	SVSPFLXA	\$100.55	VRBHFLMA	\$63.80		
PTCYFLXA	\$137.25	SVSSFLXA	\$109.90	WALDFLXA	\$114.70		
PTSJFLXA	\$143.45	SWTHFLXA	\$105.50	WCHLFLXA	\$124.30		
PTSLFLMA	\$108.30	SYHSFLCC	\$125.90	WELKFLMA	\$148.10		
PTSLFLSO	\$102.20	TAFBFLXA	\$105.50	WHSPFLXA	\$194.60	1	
PXTNFLXA	\$157.40	TAMPFL41	\$98.90	WIMMFLXA	\$121.10	1	
QNCYFLXA	\$138.80	TAMPFL52	\$93.95	WLBRFLXA	\$200.80	1	
RAFRFLXA	\$151.20	TAMPFLCM	\$98.90	WLCHFLXA	\$124.30		
RSKNFLXA	\$114.70	TAMPFLCO	\$102.20	WLCRFLXA	\$100.55	1	
DVIII ELVA	Φ400.0E		Φ02.0F	VALL OTEL XA	040400	1	

ISSUED: November 13, 2008 EFFECTIVE: November 14, 2008

**WLSTFLXA** 

\$124.30

\$93.95

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.1 Intrastate Analog Local Access (Cont'd.)
  - 1.1.3.1 Monthly Recurring Channel Charges: (Cont'd)
    - 1.1.3.1.3 Pre-VBS III Monthly Recurring Local Loop Charges: The following monthly recurring charges apply, for circuits installed before May 1, 2002 that are not assessed VBS III Monthly Recurring Local Loop Charges, based on the Serving Wire Center of the circuit:

**SEE RATE TABLE FOR SECTION 1.1.3.1.2** 

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.1 <u>Intrastate Analog Local Access (Cont'd.)</u>
    - 1.1.3.2 Non-Recurring Charges: A \$200 non-recurring channel charge applies.
    - 1.1.3.3 <u>Network Connection Charge:</u> The following monthly recurring and non-recurring charges will apply, per connection, only for circuits installed on or after October 1, 1998, if access is Customer-provided.

Monthly Installation \$60 \$0.00

1.1.3.4 The Special Access Surcharge is applied.

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. SERVICE DESCRIPTION (CONT'D.)
  - 1.1 <u>Intrastate Analog Local Access (Cont'd.)</u>
    - 1.1.3.5. Other Non-Recurring Charges:
      - 1.1.3.5.1 Customer Support Services charges apply.
      - 1.1.3.5.2 Intrastate Administrative Non-Recurring Charges apply.
    - 1.1.4. TERMS AND CONDITIONS: The tariff general terms and conditions apply, except that the Credit Allowance for Interruption of Local Access Service as set forth in Section 2.3 below also applies.

### 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>

#### 1.2 Intrastate DS0 (Hubless) Access

1.2.1 <u>Product Description</u>: Intrastate DS0 (Hubless) Access is provided by Company. Intrastate DS0 (Hubless) Access is a digital local access arrangement offering bandwidth up to 64 kbps that can be used to Customer's premises to the Company's Point of Presence (POP) to provide Customer access to Company services. It is known as "Hubless DDS" because it does not physically pass through the centralized Local Exchange Carrier wire center or hub.

### 1.2.2 Rates and Charges:

Where rates or charges are specific to a particular pricing program (i.e., Verizon Business Services (VBS) I, II, or III) those rates and charges only apply where a Customer has enrolled in that pricing program. Rates and charges which are not specific to a particular pricing program apply to all Customers.

#### 1.2.2.1. Local Access Channel Monthly Recurring Channel Charges:

1.2.2.1.1 <u>Verizon Business Services III Charges:</u> The following monthly recurring per-circuit charges, based on the Serving Wire Center of the circuit:

Wire Center	Rate						
ABDLFLXA	\$200.00	BNNLFLMA	\$230.00	CNSDFLXA	\$180.00	DRBHFLMA	\$170.00
ALCHFLXA	\$160.00	BNSPFLXA	\$250.00	CNTMFLLE	\$210.00	DUNDFLXA	\$240.00
ALFAFLXA	\$200.00	BORAFLXA	\$160.00	COCOFLMA	\$150.00	DWPKFLXA	\$170.00
ALFRFLXA	\$200.00	BRBAFLXA	\$180.00	COCOFLME	\$180.00	DYBHFLFN	\$160.00
ALSPFLXA	\$190.00	BRFRFLXA	\$150.00	CPCRFLXA	\$220.00	DYBHFLMA	\$150.00
ALTRFLXA	\$260.00	BRJTFLXA	\$280.00	CPCRFLXB	\$220.00	DYBHFLMN	\$160.00
ALVAFLXA	\$240.00	BRKRFLXA	\$130.00	CPHZFLXA	\$260.00	DYBHFLOB	\$190.00
ANMRFLXA	\$210.00	BRNDFLXA	\$170.00	CRCYFLXA	\$140.00	DYBHFLOS	\$200.00
APPKFLXA	\$240.00	BRSNFLMA	\$230.00	CRRVFLXA	\$270.00	DYBHFLPO	\$190.00
ARCDFLXA	\$270.00	BRTNFLXX	\$170.00	CRVWFLXA	\$280.00	DYBHFLWS	\$200.00
ARCHFLMA	\$210.00	BSHNFLXA	\$250.00	CRWDFLXA	\$170.00	EGLLFLBG	\$180.00
ASTRFLXA	\$270.00	BVHLFLXA	\$260.00	CSCYFLBA	\$290.00	EGLLFLIH	\$190.00
AVPKFLXA	\$150.00	BWLGFLXA	\$240.00	CSLBFLXA	\$240.00	ENWDFLXA	\$370.00
BAKRFLXA	\$250.00	BYBHFLMA	\$160.00	CTDLFLXA	\$200.00	EORNFLMA	\$210.00
BARTFLXA	\$270.00	BYSHFLXA	\$190.00	CYGRFLXA	\$220.00	ESTSFLXA	\$270.00
BAYUFLXA	\$180.00	CCBHFLAF	\$200.00	CYLKFLXA	\$220.00	ETVLFLAA	\$280.00
BBPKFLXA	\$310.00	CCBHFLMA	\$170.00	CYLKFLXB	\$220.00	EVRGFLXA	\$320.00
BCGRFLXA	\$250.00	CDKYFLMA	\$310.00	DBRYFLDL	\$200.00	FHSDFLXA	\$160.00
BCRTFLBT	\$150.00	CFLDFLMA	\$260.00	DBRYFLMA	\$200.00	FLBHFLMA	\$220.00
BCRTFLMA	\$140.00	CFVLFLXA	\$240.00	DDCYFLXA	\$280.00	FLRHFLXA	\$140.00
BCRTFLSA	\$170.00	CHLKFLXA	\$300.00	DELDFLMA	\$220.00	FRBHFLFP	\$240.00
BGPIFLMA	\$330.00	CHPLFLJA	\$150.00	DESTFLXA	\$280.00	FRPTFLXA	\$320.00
BHPKFLXA	\$160.00	CHSWFLXA	\$280.00	DFSPFLXA	\$330.00	FRSTFLXA	\$340.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

# 1.2.2.1.1 <u>Verizon Business Services III Charges</u>: (Cont'd)

Wire Center	Rate						
BKVLFLJF	\$310.00	CHTHFLXA	\$350.00	DLBHFLKP	\$190.00	FTGRFLMA	\$200.00
BLDWFLMA	\$220.00	CITRFLXA	\$140.00	DLBHFLMA	\$160.00	FTLDFLCR	\$170.00
BLGLFLMA	\$210.00	CLHNFLXA	\$140.00	DLSPFLMA	\$230.00	FTLDFLCY	\$160.00
BLTWFLXA	\$340.00	CLMTFLXA	\$300.00	DNDNFLXA	\$190.00	FTLDFLJA	\$180.00
BLVWFLXA	\$230.00	CLTNFLXA	\$310.00	DNLNFLWM	\$250.00	FTLDFLMR	\$140.00
BNFYFLXA	\$200.00	CLWRFLXA	\$150.00	DRBHFLDF	\$170.00	FTLDFLOA	\$150.00

Wire Center	Rate						
FTLDFLPL	\$170.00	HMSTFLNA	\$210.00	KYHGFLMA	\$230.00	MIAMFLFL	\$170.00
FTLDFLSF	\$140.00	HMSTFLTE	\$310.00	KYLRFLLS	\$260.00	MIAMFLGR	\$140.00
FTLDFLSG	\$180.00	HNCYFLXA	\$250.00	KYLRFLMA	\$270.00	MIAMFLHL	\$150.00
FTLDFLSU	\$180.00	HNCYFLXN	\$280.00	KYSTFLXA	\$200.00	MIAMFLIC	\$200.00
FTLDFLTA	\$140.00	HOWYFLXA	\$280.00	KYWSFLMA	\$240.00	MIAMFLKE	\$190.00
FTLDFLWN	\$190.00	HSNGFLXA	\$130.00	LBLLFLXA	\$260.00	MIAMFLME	\$150.00
FTMBFLXA	\$230.00	HTISFLMA	\$190.00	LDLKFLXA	\$250.00	MIAMFLNM	\$170.00
FTMDFLXA	\$240.00	HWTHFLMA	\$220.00	LEE FLXA	\$310.00	MIAMFLNS	\$160.00
FTMYFLMA	\$270.00	HYPKFLXA	\$170.00	LGBKFLXA	\$190.00	MIAMFLOL	\$150.00
FTMYFLXA	\$150.00	IMKLFLXA	\$260.00	LHACFLXA	\$230.00	MIAMFLPB	\$150.00
FTMYFLXB	\$210.00	INLKFLXA	\$380.00	LKALFLXA	\$230.00	MIAMFLPL	\$140.00
FTMYFLXC	\$210.00	INRKFLXX	\$190.00	LKBNFLXB	\$170.00	MIAMFLRR	\$150.00
FTPRFLMA	\$150.00	INTRFLXA	\$130.00	LKBRFLXA	\$230.00	MIAMFLSH	\$190.00
FTWBFLXA	\$270.00	INTWFLXA	\$270.00	LKBTFLXA	\$140.00	MIAMFLSO	\$150.00
FTWBFLXB	\$270.00	INVRFLXA	\$260.00	LKCYFLMA	\$160.00	MIAMFLUJ	\$170.00
FTWBFLXC	\$250.00	IONAFLXA	\$230.00	LKHLFLXA	\$220.00	MIAMFLWD	\$200.00
FTWHFLXA	\$140.00	ISLMFLMA	\$290.00	LKLDFLXA	\$190.00	MIAMFLWM	\$150.00
GCSPFLCN	\$230.00	JAY FLMA	\$260.00	LKLDFLXE	\$150.00	MIANFLWK	\$160.00
GCVLFLMA	\$210.00	JCBHFLAB	\$190.00	LKLDFLXN	\$260.00	MIANFLYI	\$140.00
GDRGFLXA	\$240.00	JCBHFLMA	\$150.00	LKMRFLHE	\$200.00	MIAQFL06	\$160.00
GENVFLMA	\$210.00	JCBHFLSP	\$190.00	LKMRFLMA	\$220.00	MIATFLAD	\$140.00
GLBRFLMC	\$200.00	JCVLFLAI	\$210.00	LKPCFLXA	\$240.00	MICCFLBB	\$220.00
GLDLFLXA	\$330.00	JCVLFLAR	\$150.00	LKWLFLXA	\$280.00	MLBRFLMA	\$150.00
GLGCFLXA	\$270.00	JCVLFLBW	\$160.00	LKWLFLXE	\$340.00	MLBYFLXA	\$270.00
GLRDFLXA	\$220.00	JCVLFLCL	\$140.00	LLMNFLXA	\$170.00	MLRSFLXA	\$140.00
GNBOFLXA	\$340.00	JCVLFLFC	\$190.00	LNLKFLXA	\$230.00	MLTNFLRA	\$220.00
GNDYFLXA	\$160.00	JCVLFLGH	\$210.00	LRGOFLXA	\$170.00	MNDRFLAV	\$150.00
GNVLFLXA	\$280.00	JCVLFLIA	\$180.00	LRVLFLXA	\$160.00	MNDRFLLO	\$160.00
GNWDFLXA	\$230.00	JCVLFLJT	\$160.00	LSBGFLXA	\$160.00	MNDRFLLW	\$200.00
GRETFLXA	\$340.00	JCVLFLLF	\$160.00	LUTZFLXA	\$210.00	MNLKFLXA	\$250.00
GSVLFLAN	\$150.00	JCVLFLNO	\$150.00	LVOKFLXA	\$130.00	MNSNFLMA	\$260.00
GSVLFLMA	\$150.00	JCVLFLOG	\$160.00	LWTYFLXA	\$240.00	MNTIFLXA	\$260.00
GSVLFLNW	\$150.00	JCVLFLOW	\$160.00	LYHNFLOH	\$160.00	MOISFLXA	\$300.00
GVLDFLXA	\$290.00	JCVLFLRV	\$150.00	MALNFLXA	\$230.00	MOLNFLXA	\$190.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

## 1.2.2.1.1 <u>Verizon Business Services III Charges</u>: (Cont'd)

Wire Center	Rate						
HAVNFLMA	\$340.00	JCVLFLSJ	\$150.00	MAYOFLXA	\$170.00	MRDCFLXA	\$260.00
HBSDFLMA	\$220.00	JCVLFLSM	\$150.00	MCINFLXA	\$140.00	MRHNFLXA	\$300.00
HDSNFLXA	\$240.00	JCVLFLWC	\$150.00	MCLNFLXA	\$270.00	MRNNFLXA	\$190.00
HGLDFLXA	\$230.00	JCVLFLWF	\$140.00	MCNPFLMA	\$210.00	MRTHFLVE	\$350.00
HGSPFLXA	\$150.00	JCVMFLLI	\$190.00	MDBGFLPM	\$220.00	MTDRFLXA	\$280.00
HLNVFLMA	\$240.00	JCVMFLUF	\$190.00	MDSNFLXA	\$300.00	MTLDFLAP	\$190.00
HLRDFLXA	\$150.00	JNGSFLXA	\$180.00	MIAMFLAE	\$140.00	MTLDFLXA	\$150.00
HLWDFLED	\$210.00	JPTRFLGN	\$230.00	MIAMFLAL	\$160.00	MTVRFLXA	\$270.00
HLWDFLHA	\$150.00	JPTRFLMA	\$190.00	MIAMFLAP	\$150.00	MXVLFLMA	\$220.00
HLWDFLMA	\$170.00	JSPRFLXA	\$160.00	MIAMFLBA	\$150.00	MYCYFLXA	\$280.00
HLWDFLPE	\$150.00	KGLKFLXA	\$240.00	MIAMFLBC	\$180.00	NDADFLAA	\$160.00
HLWDFLWH	\$190.00	KNVLFLXA	\$310.00	MIAMFLBR	\$170.00	NDADFLAC	\$190.00
HMSPFLXA	\$280.00	KSSMFLXA	\$250.00	MIAMFLCA	\$150.00	NDADFLBR	\$150.00
HMSTFLEA	\$200.00	KSSMFLXB	\$250.00	MIAMFLDB	\$190.00	NDADFLGG	\$160.00
HMSTFLHM	\$170.00	KSSMFLXD	\$240.00	MIAMFLDI	\$140.00	NDADFLOL	\$150.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

## 1.2.2.1.1 <u>Verizon Business Services III Charges</u>: (Cont'd)

Wire Center	Rate						
NFMYFLXA	\$210.00	PAHKFLMA	\$270.00	SCPKFLXA	\$230.00	TAMQFLPM	\$280.00
NFMYFLXB	\$220.00	PANCFLXA	\$260.00	SEKYFLXA	\$220.00	TAMQFLRY	\$280.00
NGBHFLXA	\$180.00	PCBHFLNT	\$150.00	SGBEFLXA	\$200.00	TAMQFLTA	\$280.00
NKLRFLMA	\$240.00	PKCYFLXA	\$240.00	SGBHFLXA	\$330.00	TAMSFLCZ	\$280.00
NNPLFLXA	\$260.00	PLCSFLMA	\$210.00	SGKYFLMA	\$310.00	THNTFLXA	\$210.00
NPLSFLXC	\$270.00	PLMTFLXA	\$210.00	SHLMFLXA	\$270.00	TLCHFLXA	\$270.00
NPLSFLXD	\$270.00	PLSLFLXA	\$170.00	SKWYFLXA	\$180.00	TLHSFLXA	\$150.00
NPRCFLXA	\$220.00	PLTKFLMA	\$160.00	SLHLFLXA	\$230.00	TLHSFLXB	\$250.00
NRPTFLXA	\$330.00	PMBHFLCS	\$160.00	SLSPFLXA	\$210.00	TLHSFLXC	\$250.00
NRSDFLXA	\$170.00	PMBHFLED	\$160.00	SMNLFLXA	\$170.00	TLHSFLXD	\$250.00
NSBHFLMA	\$160.00	PMBHFLFE	\$150.00	SNANFLXA	\$290.00	TLHSFLXE	\$240.00
NWBYFLMA	\$210.00	PMBHFLMA	\$150.00	SNDSFLXA	\$250.00	TLHSFLXF	\$270.00
OCALFLXA	\$150.00	PMBHFLNP	\$150.00	SNFRFLMA	\$160.00	TLHSFLXG	\$290.00
OCALFLXB	\$210.00	PMBHFLTA	\$150.00	SNISFLXA	\$240.00	TLHSFLXH	\$270.00
OCALFLXC	\$210.00	PMPKFLMA	\$210.00	SNRSFLXA	\$310.00	TMTRFLXA	\$220.00
OCALFLXJ	\$220.00	PNCRFLXA	\$240.00	SNSNFLXA	\$280.00	TRENFLMA	\$250.00
OCNFFLXA	\$240.00	PNCYFLCA	\$190.00	SNSPFLXA	\$230.00	TRSPFLXA	\$210.00
OJUSFLTL	\$240.00	PNCYFLMA	\$150.00	SPBGFLXA	\$160.00	TTVLFLMA	\$180.00
OKCBFLXA	\$290.00	PNGRFLXA	\$250.00	SPBGFLXS	\$190.00	TVRSFLXA	\$270.00
OKHLFLMA	\$250.00	PNISFLXA	\$240.00	SPCPFLXA	\$260.00	UMTLFLXA	\$270.00
OKLWFLXA	\$240.00	PNLNFLXA	\$350.00	SPRGFLXA	\$210.00	UNVRFLXA	\$190.00
OLDSFLXA	\$190.00	PNLSFLXA	\$170.00	SRSTFLXA	\$150.00	VENCFLXA	\$300.00
OLTWFLLN	\$270.00	PNSCFLBL	\$160.00	SSDSFLXA	\$220.00	VENCFLXS	\$310.00
ORCYFLXA	\$240.00	PNSCFLFP	\$150.00	SSPRFLXA	\$260.00	VERNFLMA	\$220.00
ORCYFLXC	\$210.00	PNSCFLHC	\$200.00	STAGFLBS	\$240.00	VLPRFLXA	\$290.00
ORLDFL60	\$140.00	PNSCFLPB	\$220.00	STAGFLMA	\$210.00	VLPRFLXB	\$290.00
ORLDFLAP	\$150.00	PNSCFLWA	\$160.00	STAGFLSH	\$240.00	VRBHFLBE	\$190.00
ORLDFLCL	\$140.00	PNVDFLMA	\$200.00	STAGFLWG	\$230.00	VRBHFLMA	\$150.00
ORLDFLLU	\$140.00	POINFLXA	\$310.00	STCDFLXA	\$270.00	WALDFLXA	\$130.00
ORLDFLMA	\$140.00	PRRNFLCC	\$200.00	STGRFLXA	\$190.00	WCHLFLXA	\$240.00
ORLDFLOE	\$140.00	PRRNFLMA	\$200.00	STMKFLXA	\$250.00	WELKFLMA	\$210.00
ORLDFLPC	\$140.00	PRRYFLXA	\$490.00	STRKFLXA	\$250.00	WHSPFLXA	\$130.00
ORLDFLPH	\$150.00	PRSHFLXA	\$260.00	STRTFLMA	\$190.00	WIMMFLXA	\$230.00
ORLDFLSA	\$170.00	PRSNFLFD	\$250.00	SVSPFLXA	\$220.00	WLBRFLXA	\$130.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 <u>Intrastate DS0 (Hubless) Access</u> (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

## 1.2.2.1.1 <u>Verizon Business Services III Charges</u>: (Cont'd)

Wire Center	Rate						
ORLDFLSO	\$160.00	PSDNFLXA	\$180.00	SVSSFLXA	\$230.00	WLCHFLXA	\$230.00
ORLDFLWR	\$160.00	PTCTFLXA	\$270.00	SWTHFLXA	\$160.00	WLCRFLXA	\$170.00
ORLDFLXH	\$140.00	PTCYFLXA	\$240.00	SYHSFLCC	\$210.00	WLHLFLXA	\$350.00
ORLEFLGV	\$190.00	PTSJFLXA	\$360.00	TAFBFLXA	\$230.00	WLSTFLXA	\$230.00
ORLEFLKT	\$280.00	PTSLFLMA	\$190.00	TAMPFL52	\$170.00	WLWDFLXA	\$240.00
ORLFFLEJ	\$280.00	PTSLFLSO	\$150.00	TAMPFLCM	\$180.00	WNDRFLXA	\$230.00
ORLFFLHX	\$280.00	QNCYFLXA	\$340.00	TAMPFLCO	\$180.00	WNGRFLXA	\$240.00
ORLFFLQR	\$280.00	RAFRFLXA	\$150.00	TAMPFLDM	\$180.00	WNHNFLBU	\$220.00
ORPKFLMA	\$200.00	RSKNFLXA	\$220.00	TAMPFLEM	\$180.00	WNHNFLXC	\$190.00
ORPKFLRW	\$200.00	RYHLFLXA	\$230.00	TAMPFLKO	\$180.00	WNPKFLTL	\$160.00
ORSPFLXA	\$150.00	SARKFLXA	\$210.00	TAMPFLXA	\$180.00	WNPKFLXA	\$150.00
OSPRFLXA	\$240.00	SBNGFLXA	\$220.00	TAMPFLXE	\$160.00	WNPKFLXE	\$160.00
OVIDFLCA	\$200.00	SBSTFLFE	\$210.00	TAMPFLXX	\$160.00	WPBHFL58	\$150.00
PACEFLPV	\$200.00	SBSTFLMA	\$210.00	TAMQFLHU	\$280.00	WPBHFLAN	\$150.00
WPBHFLGA	\$150.00	WSSDFLXA	\$130.00	YNTWFLMA	\$280.00		
WPBHFLGR	\$190.00	WSTVFLXA	\$220.00	YULEFLMA	\$230.00		
WPBHFLHH	\$150.00	WWSPFLHI	\$290.00	ZLSPFLXA	\$240.00		
WPBHFLLE	\$190.00	WWSPFLSH	\$300.00	ZPHYFLXA	\$190.00	]	
WPBHFLRB	\$150.00	YBCTFLXA	\$130.00			_	
WPBHFLRP	\$200.00	YNFNFLMA	\$230.00	]			

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access
  - 1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)
    - 1.2.2.1.2 <u>Verizon Business Services II Charges:</u> The following monthly recurring per-circuit charges, based on the Serving Wire Center of the circuit:

Wire	D-1-	Wine Oracles	D-1-	M/2 - 0 1	Dete	W O	D-1-
Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
ABDLFLXA	\$240.00	CFLDFLMA	\$250.00	DYBHFLPO	\$170.00	GSVLFLNW	\$150.00
ALCHFLXA	\$180.00	CFVLFLXA	\$300.00	EGLLFLBG	\$150.00	GVLDFLXA	\$300.00
ALFAFLXA	\$240.00	CHLKFLXA	\$300.00	EGLLFLIH	\$170.00	HAVNFLMA	\$340.00
ALFRFLXA	\$250.00	CHPLFLJA	\$150.00	ENWDFLXA	\$370.00	HBSDFLMA	\$210.00
ALSPFLXA	\$150.00	CHSWFLXA	\$230.00	EORNFLMA	\$200.00	HDSNFLXA	\$270.00
ALTRFLXA	\$150.00	CHTHFLXA	\$340.00	ESTSFLXA	\$300.00	HGLDFLXA	\$250.00
ALVAFLXA	\$300.00	CITRFLXA	\$150.00	EVRGFLXA	\$300.00	HGSPFLXA	\$170.00
ANMRFLXA	\$200.00	CLHNFLXA	\$150.00	FHSDFLXA	\$210.00	HLNVFLMA	\$230.00
APPKFLXA	\$250.00	CLMTFLXA	\$300.00	FLBHFLMA	\$210.00	HLRDFLXA	\$150.00
ARCDFLXA	\$300.00	CLTNFLXA	\$300.00	FLRHFLXA	\$150.00	HLWDFLHA	\$180.00
ARCHFLMA	\$200.00	CLWRFLXA	\$210.00	FRBHFLFP	\$200.00	HLWDFLMA	\$180.00
ASTRFLXA	\$300.00	CNSDFLXA	\$250.00	FRPTFLXA	\$300.00	HLWDFLPE	\$150.00
AVPKFLXA	\$150.00	CNTMFLLE	\$190.00	FRSTFLXA	\$380.00	HLWDFLWH	\$170.00
BAKRFLXA	\$300.00	COCOFLMA	\$150.00	FTGRFLMA	\$180.00	HMSPFLXA	\$300.00
BARTFLXA	\$290.00	COCOFLME	\$170.00	FTLDFLCR	\$170.00	HMSTFLEA	\$150.00
BAYUFLXA	\$250.00	COCYFL13	\$150.00	FTLDFLCY	\$170.00	HMSTFLHM	\$150.00
BBPKFLXA	\$350.00	CPCRFLXA	\$250.00	FTLDFLJA	\$150.00	HMSTFLNA	\$180.00
BCGRFLXA	\$300.00	CPCRFLXB	\$250.00	FTLDFLMR	\$150.00	HNCYFLXA	\$280.00
BCRTFLBT	\$150.00	CPHZFLXA	\$300.00	FTLDFLOA	\$150.00	HNCYFLXN	\$330.00
BCRTFLMA	\$150.00	CRCYFLXA	\$150.00	FTLDFLPL	\$170.00	HOWYFLXA	\$300.00
BCRTFLSA	\$180.00	CRRVFLXA	\$300.00	FTLDFLSG	\$170.00	HSNGFLXA	\$150.00
BGPIFLMA	\$310.00	CRVWFLXA	\$250.00	FTLDFLSU	\$170.00	HTISFLMA	\$180.00
BHPKFLXA	\$210.00	CRWDFLXA	\$240.00	FTLDFLWN	\$180.00	HWTHFLMA	\$200.00
BKVLFLJF	\$250.00	CSCYFLBA	\$280.00	FTMBFLXA	\$300.00	HYPKFLXA	\$220.00
BLDWFLMA	\$200.00	CSLBFLXA	\$250.00	FTMDFLXA	\$300.00	IMKLFLXA	\$300.00
BLGLFLMA	\$180.00	CTDLFLXA	\$280.00	FTMYFLXA	\$150.00	INLKFLXA	\$420.00
BLTWFLXA	\$340.00	CYGRFLXA	\$250.00	FTMYFLXB	\$200.00	INRKFLXX	\$260.00
BLVWFLXA	\$300.00	CYLKFLXA	\$250.00	FTMYFLXC	\$250.00	INTRFLXA	\$150.00
BNFYFLXA	\$250.00	CYLKFLXB	\$300.00	FTPRFLMA	\$150.00	INTWFLXA	\$250.00
BNNLFLMA	\$200.00	DBRYFLDL	\$180.00	FTWBFLXA	\$200.00	INVRFLXA	\$300.00
BNSPFLXA	\$300.00	DBRYFLMA	\$180.00	FTWBFLXB	\$250.00	IONAFLXA	\$300.00
BORAFLXA	\$180.00	DDCYFLXA	\$300.00	FTWBFLXC	\$300.00	ISLMFLMA	\$270.00
BRBAFLXA	\$200.00	DELDFLMA	\$150.00	FTWHFLXA	\$150.00	JAY FLMA	\$250.00
BRFRFLXA	\$150.00	DESTFLXA	\$250.00	GCSPFLCN	\$210.00	JCBHFLAB	\$170.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.2.2.1.2 <u>Verizon Business Services II Charges</u>: (Cont'd)

Wire Center	Rate						
BRJTFLXA	\$310.00	DFSPFLXA	\$300.00	GCVLFLMA	\$190.00	JCBHFLMA	\$150.00
BRKRFLXA	\$150.00	DLBHFLKP	\$170.00	GDRGFLXA	\$300.00	JCBHFLSP	\$170.00
BRNDFLXA	\$240.00	DLBHFLMA	\$150.00	GENVFLMA	\$190.00	JCVLFLAR	\$150.00
BRSNFLMA	\$200.00	DLSPFLMA	\$210.00	GLBRFLMC	\$180.00	JCVLFLBW	\$170.00
BRTNFLXX	\$200.00	DNDNFLXA	\$250.00	GLDLFLXA	\$300.00	JCVLFLCL	\$150.00
BSHNFLXA	\$250.00	DNLNFLWM	\$200.00	GLGCFLXA	\$300.00	JCVLFLFC	\$170.00
BVHLFLXA	\$250.00	DRBHFLMA	\$170.00	GLRDFLXA	\$150.00	JCVLFLIA	\$150.00
BWLGFLXA	\$300.00	DUNDFLXA	\$270.00	GNBOFLXA	\$250.00	JCVLFLJT	\$170.00
BYBHFLMA	\$150.00	DWPKFLXA	\$200.00	GNDYFLXA	\$230.00	JCVLFLLF	\$150.00
BYSHFLXA	\$240.00	DYBHFLFN	\$150.00	GNVLFLXA	\$300.00	JCVLFLNO	\$150.00
CCBHFLAF	\$180.00	DYBHFLMA	\$150.00	GNWDFLXA	\$300.00	JCVLFLOW	\$150.00
CCBHFLMA	\$150.00	DYBHFLOB	\$170.00	GRETFLXA	\$360.00	JCVLFLRV	\$150.00
CDKYFLMA	\$300.00	DYBHFLOS	\$180.00	GSVLFLMA	\$150.00	JCVLFLSJ	\$150.00
JCVLFLSM	\$150.00	MIAMFLAE	\$150.00	NDADFLOL	\$170.00	PMBHFLTA	\$170.00
JCVLFLWC	\$150.00	MIAMFLAL	\$170.00	NFMYFLXA	\$250.00	PMPKFLMA	\$200.00
JNGSFLXA	\$200.00	MIAMFLAP	\$150.00	NFMYFLXB	\$250.00	PNCRFLXA	\$270.00
JPTRFLMA	\$190.00	MIAMFLBA	\$150.00	NGBHFLXA	\$270.00	PNCYFLCA	\$170.00
JSPRFLXA	\$180.00	MIAMFLBC	\$150.00	NKLRFLMA	\$200.00	PNCYFLMA	\$150.00
KGLKFLXA	\$300.00	MIAMFLBR	\$150.00	NNPLFLXA	\$300.00	PNGRFLXA	\$300.00
KNVLFLXA	\$250.00	MIAMFLCA	\$150.00	NPLSFLXC	\$300.00	PNISFLXA	\$300.00
KSSMFLXA	\$150.00	MIAMFLDB	\$170.00	NPLSFLXD	\$300.00	PNLNFLXA	\$300.00
KSSMFLXB	\$300.00	MIAMFLFL	\$150.00	NPRCFLXA	\$220.00	PNLSFLXA	\$240.00
KSSMFLXD	\$300.00	MIAMFLGR	\$150.00	NRPTFLXA	\$400.00	PNSCFLBL	\$150.00
KYHGFLMA	\$200.00	MIAMFLHL	\$150.00	NRSDFLXA	\$150.00	PNSCFLFP	\$150.00
KYLRFLLS	\$230.00	MIAMFLIC	\$150.00	NSBHFLMA	\$150.00	PNSCFLHC	\$180.00
KYLRFLMA	\$240.00	MIAMFLKE	\$180.00	NWBYFLMA	\$210.00	PNSCFLPB	\$210.00
KYSTFLXA	\$250.00	MIAMFLME	\$150.00	OCALFLXA	\$150.00	PNSCFLWA	\$180.00
KYWSFLMA	\$210.00	MIAMFLNM	\$170.00	OCALFLXB	\$150.00	PNVDFLMA	\$180.00
LBLLFLXA	\$300.00	MIAMFLNS	\$170.00	OCALFLXC	\$250.00	POINFLXA	\$350.00
LDLKFLXA	\$300.00	MIAMFLOL	\$170.00	OCALFLXJ	\$250.00	PRRNFLMA	\$150.00
LEE FLXA	\$300.00	MIAMFLPB	\$150.00	OCNFFLXA	\$300.00	PRRYFLXA	\$500.00
LGBKFLXA	\$200.00	MIAMFLPL	\$150.00	OKCBFLXA	\$300.00	PRSHFLXA	\$200.00
LHACFLXA	\$300.00	MIAMFLRR	\$170.00	OKHLFLMA	\$200.00	PRSNFLFD	\$240.00
LKALFLXA	\$260.00	MIAMFLSH	\$170.00	OKLWFLXA	\$300.00	PSDNFLXA	\$260.00
LKBNFLXB	\$250.00	MIAMFLSO	\$150.00	OLDSFLXA	\$240.00	PTCTFLXA	\$250.00
LKBRFLXA	\$150.00	MIAMFLWD	\$150.00	OLTWFLLN	\$260.00	PTCYFLXA	\$270.00
LKBTFLXA	\$150.00	MIAMFLWM	\$150.00	ORCYFLXA	\$300.00	PTSJFLXA	\$350.00
LKCYFLMA	\$150.00	MICCFLBB	\$210.00	ORCYFLXC	\$300.00	PTSLFLMA	\$150.00
LKHLFLXA	\$300.00	MLBRFLMA	\$150.00	ORLDFLAP	\$170.00	PTSLFLSO	\$150.00
LKLDFLXA	\$210.00	MLBYFLXA	\$290.00	ORLDFLCL	\$150.00	QNCYFLXA	\$300.00
LKLDFLXE	\$210.00	MLRSFLXA	\$150.00	ORLDFLMA	\$150.00	RAFRFLXA	\$170.00

SWTHFLXA

SYHSFLCC

\$210.00

\$200.00

WNDRFLXA

WNGRFLXA

## VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 <u>Intrastate DS0 (Hubless) Access</u> (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.2.2.1.2 <u>Verizon Business Services II Charges</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
LKLDFLXN	\$260.00	MLTNFLRA	\$200.00	ORLDFLPC	\$150.00	RSKNFLXA	\$260.00
LKMRFLMA	\$200.00	MNDRFLAV	\$150.00	ORLDFLPH	\$150.00	RYHLFLXA	\$300.00
LKPCFLXA	\$300.00	MNDRFLLO	\$150.00	ORLDFLSA	\$150.00	SARKFLXA	\$230.00
LKWLFLXA	\$310.00	MNDRFLLW	\$190.00	ORPKFLMA	\$180.00	SBNGFLXA	\$300.00
LKWLFLXE	\$380.00	MNLKFLXA	\$290.00	ORPKFLRW	\$180.00	SBSTFLFE	\$250.00
LLMNFLXA	\$240.00	MNSNFLMA	\$200.00	ORSPFLXA	\$170.00	SBSTFLMA	\$250.00
LNLKFLXA	\$280.00	MNTIFLXA	\$300.00	OSPRFLXA	\$290.00	SCPKFLXA	\$300.00
LRGOFLXA	\$240.00	MOISFLXA	\$300.00	OVIDFLCA	\$150.00	SEKYFLXA	\$250.00
LRVLFLXA	\$180.00	MOLNFLXA	\$200.00	PACEFLPV	\$190.00	SGBEFLXA	\$260.00
LSBGFLXA	\$200.00	MRDCFLXA	\$300.00	PAHKFLMA	\$200.00	SGBHFLXA	\$250.00
LUTZFLXA	\$250.00	MRHNFLXA	\$300.00	PANCFLXA	\$300.00	SGKYFLMA	\$290.00
LVOKFLXA	\$150.00	MRNNFLXA	\$200.00	PCBHFLNT	\$150.00	SHLMFLXA	\$250.00
LWTYFLXA	\$300.00	MRTHFLVE	\$330.00	PKCYFLXA	\$290.00	SKWYFLXA	\$240.00
LYHNFLOH	\$150.00	MTDRFLXA	\$300.00	PLCSFLMA	\$230.00	SLHLFLXA	\$300.00
MALNFLXA	\$300.00	MTLDFLXA	\$150.00	PLMTFLXA	\$200.00	SLSPFLXA	\$150.00
MAYOFLXA	\$190.00	MTVRFLXA	\$300.00	PLSLFLXA	\$200.00	SMNLFLXA	\$220.00
MCINFLXA	\$150.00	MXVLFLMA	\$210.00	PLTKFLMA	\$150.00	SNANFLXA	\$300.00
MCLNFLXA	\$300.00	MYCYFLXA	\$250.00	PMBHFLCS	\$170.00	SNDSFLXA	\$300.00
MCNPFLMA	\$190.00	NDADFLAC	\$170.00	PMBHFLFE	\$150.00	SNFRFLMA	\$150.00
MDBGFLPM	\$200.00	NDADFLBR	\$150.00	PMBHFLMA	\$150.00	SNISFLXA	\$300.00
MDSNFLXA	\$300.00	NDADFLGG	\$150.00	PMBHFLNP	\$150.00	SNRSFLXA	\$300.00
SNSNFLXA	\$300.00	UMTLFLXA	\$300.00	ZPHYFLXA	\$200.00		
SNSPFLXA	\$240.00	UNVRFLXA	\$240.00			_	
SPBGFLXA	\$210.00	VENCFLXA	\$340.00				
SPBGFLXS	\$250.00	VENCFLXS	\$370.00				
SPCPFLXA	\$300.00	VERNFLMA	\$200.00				
SPRGFLXA	\$260.00	VLPRFLXA	\$300.00				
SRSTFLXA	\$150.00	VLPRFLXB	\$300.00				
SSDSFLXA	\$250.00	VRBHFLBE	\$150.00				
SSPRFLXA	\$300.00	VRBHFLMA	\$150.00				
STAGFLBS	\$180.00	WALDFLXA	\$150.00				
STAGFLMA	\$150.00	WCHLFLXA	\$240.00				
STAGFLSH	\$180.00	WELKFLMA	\$200.00				
STAGFLWG	\$190.00	WHSPFLXA	\$150.00				
STCDFLXA	\$300.00	WIMMFLXA	\$270.00				
STGRFLXA	\$260.00	WLBRFLXA	\$150.00				
STMKFLXA	\$300.00	WLCHFLXA	\$270.00				
STRKFLXA	\$300.00	WLCRFLXA	\$150.00				
STRTFLMA	\$150.00	WLHLFLXA	\$250.00				
SVSPFLXA	\$250.00	WLSTFLXA	\$300.00				
SVSSFLXA	\$300.00	WLWDFLXA	\$300.00	]			
CVA/TUELVA	<u> </u>		<b>#050 00</b>				

ISSUED: November 13, 2008 EFFECTIVE: November 14, 2008

\$250.00

\$300.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 <u>Intrastate DS0 (Hubless) Access</u> (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.2.2.1.2 <u>Verizon Business Services II Charges</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate
TAFBFLXA	\$250.00	WNHNFLXC	\$220.00
TAMPFLXA	\$150.00	WNPKFLXA	\$150.00
TAMPFLXE	\$210.00	WNPKFLXE	\$150.00
TAMPFLXX	\$210.00	WPBHFLAN	\$150.00
THNTFLXA	\$250.00	WPBHFLGA	\$150.00
TLCHFLXA	\$300.00	WPBHFLGR	\$170.00
TLHSFLXA	\$150.00	WPBHFLHH	\$150.00
TLHSFLXB	\$200.00	WPBHFLLE	\$150.00
TLHSFLXC	\$200.00	WPBHFLRB	\$150.00
TLHSFLXD	\$200.00	WPBHFLRP	\$150.00
TLHSFLXE	\$200.00	WSSDFLXA	\$150.00
TLHSFLXF	\$250.00	WSTVFLXA	\$300.00
TLHSFLXG	\$300.00	WWSPFLHI	\$200.00
TLHSFLXH	\$250.00	WWSPFLSH	\$200.00
TMTRFLXA	\$240.00	YBCTFLXA	\$150.00
TRENFLMA	\$240.00	YNFNFLMA	\$210.00
TRSPFLXA	\$270.00	YNTWFLMA	\$250.00
TTVLFLMA	\$150.00	YULEFLMA	\$190.00
TVRSFLXA	\$300.00	ZLSPFLXA	\$300.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 <u>Intrastate DS0 (Hubless) Access</u> (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.3 <u>Verizon Business Services I Charges:</u> The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate						
ABDLFLXA	\$150.00	CFVLFLXA	\$250.00	DYBHFLWS	\$200.00	GRETFLXA	\$250.00
ALCHFLXA	\$250.00	CHLKFLXA	\$250.00	EGLLFLBG	\$250.00	GSVLFLAN	\$150.00
ALFAFLXA	\$150.00	CHPLFLJA	\$250.00	EGLLFLIH	\$250.00	GSVLFLMA	\$200.00
ALFRFLXA	\$250.00	CHSWFLXA	\$250.00	ENWDFLXA	\$150.00	GSVLFLNW	\$250.00
ALSPFLXA	\$250.00	CHTHFLXA	\$250.00	EORNFLMA	\$250.00	GVLDFLXA	\$250.00
ALTRFLXA	\$150.00	CITRFLXA	\$250.00	ESTSFLXA	\$250.00	HAVNFLMA	\$250.00
ALVAFLXA	\$250.00	CLHNFLXA	\$250.00	ETVLFLAA	\$200.00	HBSDFLMA	\$250.00
ANMRFLXA	\$150.00	CLMTFLXA	\$250.00	EVRGFLXA	\$250.00	HDSNFLXA	\$150.00
APPKFLXA	\$250.00	CLTNFLXA	\$250.00	FHSDFLXA	\$150.00	HGLDFLXA	\$150.00
ARCDFLXA	\$250.00	CLWRFLXA	\$150.00	FLBHFLMA	\$250.00	HGSPFLXA	\$250.00
ARCHFLMA	\$250.00	CNSDFLXA	\$150.00	FLRHFLXA	\$250.00	HLNVFLMA	\$250.00
ASTRFLXA	\$250.00	CNTMFLLE	\$250.00	FRBHFLFP	\$250.00	HLRDFLXA	\$250.00
AVPKFLXA	\$150.00	COCOFLMA	\$200.00	FRPTFLXA	\$250.00	HLWDFLED	\$200.00
BAKRFLXA	\$250.00	COCOFLME	\$250.00	FRSTFLXA	\$150.00	HLWDFLHA	\$250.00
BARTFLXA	\$150.00	COCYFL13	\$150.00	FTGRFLMA	\$250.00	HLWDFLMA	\$250.00
BAYUFLXA	\$150.00	CPCRFLXA	\$200.00	FTLDFLCR	\$250.00	HLWDFLPE	\$250.00
BBPKFLXA	\$150.00	CPCRFLXB	\$200.00	FTLDFLCY	\$250.00	HLWDFLWH	\$250.00
BCGRFLXA	\$250.00	CPHZFLXA	\$250.00	FTLDFLJA	\$250.00	HMSPFLXA	\$250.00
BCRTFLBT	\$250.00	CRCYFLXA	\$250.00	FTLDFLMR	\$150.00	HMSTFLEA	\$250.00
BCRTFLMA	\$200.00	CRRVFLXA	\$250.00	FTLDFLOA	\$200.00	HMSTFLHM	\$250.00
BCRTFLSA	\$250.00	CRVWFLXA	\$250.00	FTLDFLPL	\$250.00	HMSTFLNA	\$250.00
BGPIFLMA	\$250.00	CRWDFLXA	\$150.00	FTLDFLSG	\$250.00	HNCYFLXA	\$150.00
BHPKFLXA	\$150.00	CSCYFLBA	\$250.00	FTLDFLSU	\$250.00	HNCYFLXN	\$150.00
BKVLFLJF	\$250.00	CSLBFLXA	\$250.00	FTLDFLTA	\$200.00	HOWYFLXA	\$250.00
BLDWFLMA	\$250.00	CTDLFLXA	\$250.00	FTLDFLWN	\$250.00	HSNGFLXA	\$250.00
BLGLFLMA	\$250.00	CYGRFLXA	\$150.00	FTMBFLXA	\$250.00	HTISFLMA	\$250.00
BLTWFLXA	\$200.00	CYLKFLXA	\$200.00	FTMDFLXA	\$250.00	HWTHFLMA	\$250.00
BLVWFLXA	\$250.00	CYLKFLXB	\$250.00	FTMYFLMA	\$250.00	HYPKFLXA	\$150.00
BNFYFLXA	\$250.00	DBRYFLDL	\$250.00	FTMYFLXA	\$150.00	IMKLFLXA	\$250.00
BNNLFLMA	\$250.00	DBRYFLMA	\$250.00	FTMYFLXB	\$150.00	INLKFLXA	\$150.00
BNSPFLXA	\$250.00	DDCYFLXA	\$250.00	FTMYFLXC	\$200.00	INRKFLXX	\$150.00
BORAFLXA	\$250.00	DELDFLMA	\$250.00	FTPRFLMA	\$200.00	INTRFLXA	\$250.00
BRBAFLXA	\$150.00	DESTFLXA	\$200.00	FTWBFLXA	\$150.00	INTWFLXA	\$250.00
BRFRFLXA	\$250.00	DFSPFLXA	\$250.00	FTWBFLXB	\$200.00	INVRFLXA	\$250.00
BRJTFLXA	\$150.00	DLBHFLKP	\$250.00	FTWBFLXC	\$250.00	IONAFLXA	\$250.00
BRKRFLXA	\$250.00	DLBHFLMA	\$250.00	FTWHFLXA	\$250.00	ISLMFLMA	\$250.00
BRNDFLXA	\$150.00	DLSPFLMA	\$250.00	GCSPFLCN	\$250.00	JAY FLMA	\$250.00
BRSNFLMA	\$250.00	DNDNFLXA	\$150.00	GCVLFLMA	\$250.00	JCBHFLAB	\$250.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 <u>Intrastate DS0 (Hubless) Access</u> (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.3 <u>Verizon Business Services I Charges:</u> (Cont'd)

Wire Center	Rate						
BRTNFLXX	\$150.00	DNLNFLWM	\$250.00	GDRGFLXA	\$250.00	JCBHFLMA	\$250.00
BSHNFLXA	\$250.00	DRBHFLDF	\$200.00	GENVFLMA	\$250.00	JCBHFLSP	\$250.00
BVHLFLXA	\$200.00	DRBHFLMA	\$250.00	GLBRFLMC	\$250.00	JCVLFLAI	\$200.00
BWLGFLXA	\$250.00	DUNDFLXA	\$150.00	GLDLFLXA	\$250.00	JCVLFLAR	\$250.00
BYBHFLMA	\$250.00	DWPKFLXA	\$250.00	GLGCFLXA	\$250.00	JCVLFLBW	\$150.00
BYSHFLXA	\$150.00	DYBHFLFN	\$250.00	GLRDFLXA	\$250.00	JCVLFLCL	\$150.00
CCBHFLAF	\$250.00	DYBHFLMA	\$150.00	GNBOFLXA	\$200.00	JCVLFLFC	\$250.00
CCBHFLMA	\$250.00	DYBHFLOB	\$250.00	GNDYFLXA	\$150.00	JCVLFLGH	\$200.00
CDKYFLMA	\$250.00	DYBHFLOS	\$250.00	GNVLFLXA	\$250.00	JCVLFLIA	\$250.00
CFLDFLMA	\$250.00	DYBHFLPO	\$250.00	GNWDFLXA	\$250.00	JCVLFLJT	\$250.00
JCVLFLLF	\$250.00	LWTYFLXA	\$250.00	MOLNFLXA	\$200.00	ORLDFLSA	\$250.00
JCVLFLNO	\$250.00	LYHNFLOH	\$200.00	MRDCFLXA	\$250.00	ORLDFLSO	\$200.00
JCVLFLOG	\$200.00	MALNFLXA	\$250.00	MRHNFLXA	\$250.00	ORLDFLXH	\$200.00
JCVLFLOW	\$250.00	MAYOFLXA	\$250.00	MRNNFLXA	\$250.00	ORLEFLGV	\$200.00
JCVLFLRV	\$250.00	MCINFLXA	\$250.00	MRTHFLVE	\$250.00	ORLFFLEJ	\$250.00
JCVLFLSJ	\$150.00	MCLNFLXA	\$250.00	MTDRFLXA	\$250.00	ORLFFLQR	\$250.00
JCVLFLSM	\$150.00	MCNPFLMA	\$250.00	MTLDFLAP	\$200.00	ORPKFLMA	\$250.00
JCVLFLWC	\$250.00	MDBGFLPM	\$250.00	MTLDFLXA	\$250.00	ORPKFLRW	\$250.00
JCVMFLLI	\$200.00	MDSNFLXA	\$250.00	MTVRFLXA	\$250.00	ORSPFLXA	\$250.00
JCVMFLUF	\$200.00	MIAMFLAE	\$150.00	MXVLFLMA	\$250.00	OSPRFLXA	\$150.00
JNGSFLXA	\$250.00	MIAMFLAL	\$150.00	MYCYFLXA	\$200.00	OVIDFLCA	\$250.00
JPTRFLMA	\$250.00	MIAMFLAP	\$250.00	NDADFLAA	\$200.00	PACEFLPV	\$250.00
JSPRFLXA	\$250.00	MIAMFLBA	\$150.00	NDADFLAC	\$150.00	PAHKFLMA	\$250.00
KGLKFLXA	\$250.00	MIAMFLBC	\$150.00	NDADFLBR	\$150.00	PANCFLXA	\$250.00
KNVLFLXA	\$250.00	MIAMFLBR	\$250.00	NDADFLGG	\$200.00	PCBHFLNT	\$250.00
KSSMFLXA	\$250.00	MIAMFLCA	\$150.00	NDADFLOL	\$150.00	PKCYFLXA	\$150.00
KSSMFLXB	\$250.00	MIAMFLDB	\$150.00	NFMYFLXA	\$200.00	PLCSFLMA	\$250.00
KSSMFLXD	\$250.00	MIAMFLFL	\$250.00	NFMYFLXB	\$200.00	PLMTFLXA	\$150.00
KYHGFLMA	\$250.00	MIAMFLGR	\$150.00	NGBHFLXA	\$150.00	PLSLFLXA	\$150.00
KYLRFLLS	\$250.00	MIAMFLHL	\$250.00	NKLRFLMA	\$250.00	PLTKFLMA	\$250.00
KYLRFLMA	\$250.00	MIAMFLIC	\$250.00	NNPLFLXA	\$250.00	PMBHFLCS	\$250.00
KYSTFLXA	\$150.00	MIAMFLKE	\$250.00	NPLSFLXC	\$250.00	PMBHFLED	\$250.00
KYWSFLMA	\$250.00	MIAMFLME	\$150.00	NPLSFLXD	\$250.00	PMBHFLFE	\$150.00
LBLLFLXA	\$250.00	MIAMFLNM	\$250.00	NPRCFLXA	\$150.00	PMBHFLMA	\$150.00
LDLKFLXA	\$250.00	MIAMFLNS	\$150.00	NRPTFLXA	\$150.00	PMBHFLNP	\$150.00
LEE FLXA	\$250.00	MIAMFLOL	\$150.00	NRSDFLXA	\$150.00	PMBHFLTA	\$150.00
LGBKFLXA	\$150.00	MIAMFLPB	\$150.00	NSBHFLMA	\$250.00	PMPKFLMA	\$250.00
LHACFLXA	\$250.00	MIAMFLPL	\$150.00	NWBYFLMA	\$250.00	PNCRFLXA	\$150.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 <u>Intrastate DS0 (Hubless) Access</u> (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.3 <u>Verizon Business Services I Charges:</u> (Cont'd)

Wire Center	Rate						
LKALFLXA	\$150.00	MIAMFLRR	\$150.00	OCALFLXA	\$150.00	PNCYFLCA	\$250.00
LKBNFLXB	\$250.00	MIAMFLSH	\$250.00	OCALFLXB	\$200.00	PNCYFLMA	\$200.00
LKBRFLXA	\$250.00	MIAMFLSO	\$150.00	OCALFLXC	\$200.00	PNGRFLXA	\$250.00
LKBTFLXA	\$250.00	MIAMFLWD	\$250.00	OCALFLXJ	\$200.00	PNISFLXA	\$250.00
LKCYFLMA	\$250.00	MIAMFLWM	\$150.00	OCNFFLXA	\$250.00	PNLNFLXA	\$250.00
LKHLFLXA	\$250.00	MIANFLYI	\$200.00	OJUSFLTL	\$200.00	PNLSFLXA	\$150.00
LKLDFLXA	\$150.00	MIAQFL06	\$200.00	OKCBFLXA	\$250.00	PNSCFLBL	\$250.00
LKLDFLXE	\$150.00	MIATFLAD	\$200.00	OKHLFLMA	\$250.00	PNSCFLFP	\$200.00
LKLDFLXN	\$150.00	MICCFLBB	\$250.00	OKLWFLXA	\$250.00	PNSCFLHC	\$250.00
LKMRFLMA	\$200.00	MLBRFLMA	\$200.00	OLDSFLXA	\$150.00	PNSCFLPB	\$250.00
LKPCFLXA	\$250.00	MLBYFLXA	\$150.00	OLTWFLLN	\$250.00	PNSCFLWA	\$150.00
LKWLFLXA	\$150.00	MLRSFLXA	\$250.00	ORCYFLXA	\$250.00	PNVDFLMA	\$250.00
LKWLFLXE	\$150.00	MLTNFLRA	\$250.00	ORCYFLXC	\$250.00	POINFLXA	\$150.00
LLMNFLXA	\$150.00	MNDRFLAV	\$200.00	ORLDFL60	\$200.00	PRRNFLMA	\$150.00
LNLKFLXA	\$150.00	MNDRFLLO	\$250.00	ORLDFLAP	\$250.00	PRRYFLXA	\$250.00
LRGOFLXA	\$150.00	MNDRFLLW	\$250.00	ORLDFLCL	\$150.00	PRSHFLXA	\$150.00
LRVLFLXA	\$250.00	MNLKFLXA	\$150.00	ORLDFLLU	\$200.00	PRSNFLFD	\$250.00
LSBGFLXA	\$250.00	MNSNFLMA	\$250.00	ORLDFLMA	\$150.00	PSDNFLXA	\$150.00
LUTZFLXA	\$150.00	MNTIFLXA	\$250.00	ORLDFLPC	\$150.00	PTCTFLXA	\$250.00
LVOKFLXA	\$250.00	MOISFLXA	\$250.00	ORLDFLPH	\$250.00	PTCYFLXA	\$150.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 <u>Intrastate DS0 (Hubless) Access</u> (Cont'd)

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.3 <u>Verizon Business Services I Charges:</u> (Cont'd)

	T	1	T	T	T
Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
PTSJFLXA	\$250.00	STRTFLMA	\$200.00	WCHLFLXA	\$250.00
PTSLFLMA	\$250.00	SVSPFLXA	\$200.00	WELKFLMA	\$250.00
PTSLFLSO	\$250.00	SVSSFLXA	\$250.00	WHSPFLXA	\$250.00
QNCYFLXA	\$250.00	SWTHFLXA	\$150.00	WIMMFLXA	\$150.00
RAFRFLXA	\$250.00	SYHSFLCC	\$250.00	WLBRFLXA	\$250.00
RSKNFLXA	\$150.00	TAFBFLXA	\$250.00	WLCHFLXA	\$150.00
RYHLFLXA	\$250.00	TAMPFL52	\$200.00	WLCRFLXA	\$150.00
SARKFLXA	\$150.00	TAMPFLCM	\$200.00	WLHLFLXA	\$250.00
SBNGFLXA	\$250.00	TAMPFLCO	\$200.00	WLSTFLXA	\$250.00
SBSTFLFE	\$250.00	TAMPFLDM	\$200.00	WLWDFLXA	\$250.00
SBSTFLMA	\$250.00	TAMPFLKO	\$200.00	WNDRFLXA	\$250.00
SCPKFLXA	\$250.00	TAMPFLXA	\$150.00	WNGRFLXA	\$250.00
SEKYFLXA	\$150.00	TAMPFLXE	\$150.00	WNHNFLBU	\$200.00
SGBEFLXA	\$150.00	TAMPFLXX	\$150.00	WNHNFLXC	\$150.00
SGBHFLXA	\$250.00	TAMQFLHU	\$200.00	WNPKFLXA	\$250.00
SGKYFLMA	\$250.00	TAMQFLPM	\$200.00	WNPKFLXE	\$150.00
SHLMFLXA	\$200.00	TAMQFLTA	\$200.00	WPBHFL58	\$150.00
SKWYFLXA	\$150.00	TAMSFLCZ	\$200.00	WPBHFLAN	\$200.00
SLHLFLXA	\$250.00	THNTFLXA	\$150.00	WPBHFLGA	\$250.00
SLSPFLXA	\$150.00	TLCHFLXA	\$250.00	WPBHFLGR	\$250.00
SMNLFLXA	\$150.00	TLHSFLXA	\$150.00	WPBHFLHH	\$250.00
SNANFLXA	\$250.00	TLHSFLXB	\$150.00	WPBHFLLE	\$250.00
SNDSFLXA	\$250.00	TLHSFLXC	\$150.00	WPBHFLRB	\$250.00
SNFRFLMA	\$250.00	TLHSFLXD	\$150.00	WPBHFLRP	\$250.00
SNISFLXA	\$250.00	TLHSFLXE	\$150.00	WSSDFLXA	\$150.00
SNRSFLXA	\$250.00	TLHSFLXF	\$200.00	WSTVFLXA	\$250.00
SNSNFLXA	\$250.00	TLHSFLXG	\$250.00	WWSPFLHI	\$250.00
SNSPFLXA	\$150.00	TLHSFLXH	\$200.00	WWSPFLSH	\$250.00
SPBGFLXA	\$150.00	TMTRFLXA	\$150.00	YBCTFLXA	\$150.00
SPBGFLXS	\$150.00	TRENFLMA	\$250.00	YNFNFLMA	\$250.00
SPCPFLXA	\$250.00	TRSPFLXA	\$150.00	YNTWFLMA	\$250.00
SPRGFLXA	\$150.00	TTVLFLMA	\$250.00	YULEFLMA	\$250.00
SRSTFLXA	\$150.00	TVRSFLXA	\$250.00	ZLSPFLXA	\$250.00
SSDSFLXA	\$150.00	UMTLFLXA	\$250.00	ZPHYFLXA	\$150.00
SSPRFLXA	\$250.00	UNVRFLXA	\$150.00		
STAGFLBS	\$250.00	VENCFLXA	\$150.00		
STAGFLMA	\$200.00	VENCFLXS	\$200.00		
STAGFLSH	\$250.00	VERNFLMA	\$250.00		
STAGFLWG	\$250.00	VLPRFLXA	\$250.00		
STCDFLXA	\$250.00	VLPRFLXB	\$250.00		
STGRFLXA	\$150.00	VRBHFLBE	\$250.00		
STMKFLXA	\$250.00	VRBHFLMA	\$200.00		
STRKFLXA	\$250.00	WALDFLXA	\$250.00		

- SERVICE DESCRIPTION (CONT'D.)
- 1.2 Intrastate DS0 (Hubless) Access

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: The following monthly recurring per-circuit charges, based on Serving Wire Center of the circuit:

Pre-VBS Charges for low-speed (2.4kps, 4.8kps, 9.6kps)

Wire Center	Rate						
ABDLFLXA	\$98.70	BWLGFLXA	\$154.65	DNLNFLWM	\$160.20	FTWBFLXA	\$98.70
ALCHFLXA	\$231.05	BYBHFLAK	\$141.70	DRBHFLDF	\$125.25	FTWBFLXB	\$123.35
ALFAFLXA	\$134.30	BYBHFLMA	\$141.70	DRBHFLMA	\$123.35	FTWBFLXC	\$134.30
ALFRFLXA	\$166.25	BYSHFLXA	\$132.85	DUNDFLXA	\$138.00	FTWHFLXA	\$241.85
ALSPFLXA	\$130.95	CCBHFLAF	\$132.85	DWPKFLXA	\$288.65	GCSPFLCN	\$149.10
ALTHFLXA	\$138.00	CCBHFLMA	\$132.85	DYBHFLFN	\$125.25	GCVLFLMA	\$200.45
ALTRFLXA	\$143.55	CDKYFLMA	\$216.65	DYBHFLMA	\$98.70	GDRGFLXA	\$154.65
ALVAFLXA	\$149.10	CFLDFLMA	\$180.65	DYBHFLOB	\$127.15	GENVFLMA	\$147.25
ANMRFLXA	\$147.25	CFVLFLXA	\$154.65	DYBHFLOS	\$134.30	GLBRFLMC	\$138.00
APLCFLXA	\$207.65	CHLKFLXA	\$211.25	DYBHFLPO	\$129.05	GLDLFLXA	\$195.05
APPKFLXA	\$139.85	CHPLFLJA	\$184.25	DYBHFLWS	\$98.70	GLGCFLXA	\$177.05
ARCDFLXA	\$182.45	CHSWFLXA	\$191.45	EGLLFLBG	\$125.25	GLRDFLXA	\$132.85
ARCHFLMA	\$145.40	CHTHFLXA	\$158.35	EGLLFLIH	\$127.15	GNBOFLXA	\$154.65
ARPNFLXA	\$216.65	CITRFLXA	\$147.25	ENWDFLXA	\$169.85	GNDYFLXA	\$129.05
ASTRFLXA	\$182.45	CLHNFLXA	\$152.80	EORNFLMA	\$152.80	GNVLFLXA	\$189.65
AVPKFLXA	\$98.70	CLMTFLXA	\$205.85	ESPNFLXA	\$207.65	GNWDFLXA	\$173.45
BAKRFLXA	\$168.05	CLTNFLXA	\$223.85	ESTSFLXA	\$182.45	GRETFLXA	\$166.25
BARTFLXA	\$138.00	CLWRFLXA	\$98.70	ETVLFLAA	\$360.00	GSVLFLAN	\$98.70
BAYUFLXA	\$130.95	CNSDFLXA	\$127.15	EVRGFLXA	\$229.25	GSVLFLMA	\$98.70
BBPKFLXA	\$160.20	CNTMFLLE	\$145.55	FHSDFLXA	\$134.30	GSVLFLNW	\$129.05
BCGRFLXA	\$164.45	COCOFLMA	\$98.70	FLBHFLMA	\$152.80	GSVLFLSS	\$119.55
BCRTFLBB	\$127.15	COCOFLME	\$123.35	FLRHFLXA	\$184.25	GVLDFLXA	\$202.25
BCRTFLBT	\$127.15	CPCRFLXA	\$132.85	FRBHFLFP	\$163.90	HAVNFLMA	\$189.65
BCRTFLMA	\$98.70	CPCRFLXB	\$129.05	FRPTFLXA	\$169.85	HBSDFLMA	\$141.70
BCRTFLSA	\$130.95	CPHZFLXA	\$173.45	FRSTFLXA	\$166.25	HDSNFLXA	\$168.05
BGPIFLMA	\$312.05	CRBLFLXA	\$202.25	FTGRFLMA	\$145.40	HGLDFLXA	\$129.05
BHPKFLXA	\$127.15	CRCYFLXA	\$180.65	FTLDFLCR	\$125.25	HGSPFLXA	\$238.25
BKVLFLJF	\$202.25	CRRVFLXA	\$180.65	FTLDFLCY	\$127.15	HLNVFLMA	\$149.10
BLDWFLMA	\$152.80	CRVWFLXA	\$163.90	FTLDFLJA	\$129.05	HLRDFLXA	\$169.85

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 <u>Intrastate DS0 (Hubless) Access</u>

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: (Cont'd)

Pre-VBS Charges for low-speed (2.4kps, 4.8kps, 9.6kps)

Wire Center	Rate						
BLGLFLMA	\$187.85	CRWDFLXA	\$136.15	FTLDFLMR	\$98.70	HLWDFLED	\$129.05
BLTWFLXA	\$98.70	CSCYFLBA	\$204.05	FTLDFLOA	\$98.70	HLWDFLHA	\$127.15
BLVWFLXA	\$136.15	CSLBFLXA	\$132.85	FTLDFLOV	\$121.45	HLWDFLMA	\$129.05
BNFYFLXA	\$186.05	CTDLFLXA	\$175.25	FTLDFLPL	\$123.35	HLWDFLPE	\$130.95
BNNLFLMA	\$158.35	CYGRFLXA	\$134.30	FTLDFLSF	\$147.25	HLWDFLWH	\$127.15
BNSPFLXA	\$158.35	CYLKFLXA	\$134.30	FTLDFLSG	\$134.30	HMSPFLXA	\$186.05
BORAFLXA	\$265.25	CYLKFLXB	\$134.30	FTLDFLSU	\$125.25	HMSTFLEA	\$168.05
BRBAFLXA	\$132.85	DBRYFLDL	\$138.00	FTLDFLTA	\$98.70	HMSTFLHM	\$168.05
BRFRFLXA	\$261.65	DBRYFLMA	\$134.30	FTLDFLTN	\$119.55	HMSTFLNA	\$162.05
BRJTFLXA	\$152.80	DDCYFLXA	\$220.25	FTLDFLWN	\$139.85	HMSTFLTE	\$169.85
BRKRFLXA	\$149.10	DELDFLMA	\$158.35	FTMBFLXA	\$145.40	HNCYFLXA	\$138.00
BRNDFLXA	\$134.30	DESTFLXA	\$129.05	FTMDFLXA	\$156.50	HNCYFLXN	\$143.55
BRSNFLMA	\$162.05	DFSPFLXA	\$182.45	FTMYFLMA	\$129.05	HOWYFLXA	\$189.65
BRSTFLXA	\$125.25	DLBHFLKP	\$134.30	FTMYFLXA	\$98.70	HSFRFLXA	\$147.25
BRTNFLXX	\$138.00	DLBHFLMA	\$134.30	FTMYFLXB	\$125.25	HSNGFLXA	\$149.10
BSHNFLXA	\$182.45	DLSPFLMA	\$156.50	FTMYFLXC	\$123.35	HTISFLMA	\$129.05
BVHLFLXA	\$98.70	DNDNFLXA	\$127.15	FTPRFLMA	\$256.25	HWTHFLMA	\$147.25
HYPKFLXA	\$119.55	KYLRFLMA	\$216.65	MIAMFLBC	\$121.45	MTLDFLBR	\$129.05
IMKLFLXA	\$175.25	KYSTFLXA	\$143.55	MIAMFLBR	\$127.15	MTLDFLTC	\$130.95
INLKFLXA	\$177.05	KYWSFLMA	\$357.05	MIAMFLCA	\$136.15	MTLDFLXA	\$130.95
INRKFLXX	\$130.95	LBLLFLXA	\$169.85	MIAMFLDA	\$123.35	MTVRFLXA	\$152.80
INTRFLXA	\$189.65	LDLKFLXA	\$160.20	MIAMFLDB	\$132.85	MXVLFLMA	\$160.20
INTWFLXA	\$152.80	LGBKFLXA	\$132.85	MIAMFLFL	\$121.45	MYCYFLXA	\$162.05
INVRFLXA	\$166.25	LHACFLXA	\$145.40	MIAMFLGR	\$121.45	NDADFLAA	\$123.35
IONAFLXA	\$139.85	LKALFLXA	\$125.25	MIAMFLHL	\$134.30	NDADFLAC	\$123.35
ISLMFLMA	\$236.45	LKBNFLXB	\$145.40	MIAMFLIC	\$132.85	NDADFLBR	\$125.25
JAY FLMA	\$173.45	LKBRFLXA	\$130.95	MIAMFLKE	\$134.30	NDADFLGG	\$98.70
JCBHFLAB	\$138.00	LKBTFLXA	\$200.45	MIAMFLME	\$98.70	NDADFLOL	\$123.35
JCBHFLMA	\$141.70	LKCYFLMA	\$225.65	MIAMFLNM	\$125.25	NFMYFLXA	\$123.35
JCBHFLSP	\$134.30	LKHLFLXA	\$149.10	MIAMFLNS	\$125.25	NFMYFLXB	\$127.15
JCVLFLAI	\$123.35	LKLDFLXA	\$127.15	MIAMFLOL	\$127.15	NGBHFLXA	\$134.30

- SERVICE DESCRIPTION (CONT'D.)
- 1.2 Intrastate DS0 (Hubless) Access

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: (Cont'd)

Pre-VBS Charges for low-speed (2.4kps, 4.8kps, 9.6kps)

Wire Center	Rate						
JCVLFLAR	\$129.05	LKLDFLXE	\$216.65	MIAMFLPB	\$127.15	NKLRFLMA	\$178.85
JCVLFLBW	\$130.95	LKLDFLXN	\$132.85	MIAMFLPL	\$134.30	NNPLFLXA	\$166.25
JCVLFLCL	\$98.70	LKMRFL01	\$98.70	MIAMFLRR	\$132.85	NPLSFLXC	\$186.05
JCVLFLFC	\$129.05	LKMRFLHE	\$98.70	MIAMFLSH	\$127.15	NPLSFLXD	\$180.65
JCVLFLGH	\$129.05	LKMRFLMA	\$98.70	MIAMFLSO	\$141.70	NPRCFLXA	\$160.20
JCVLFLIA	\$138.00	LKMYFLMA	\$130.95	MIAMFLWD	\$147.25	NRPTFLXA	\$166.25
JCVLFLJB	\$98.70	LKPCFLXA	\$162.05	MIAMFLWM	\$129.05	NRSDFLXA	\$123.35
JCVLFLJT	\$125.25	LKWLFLXA	\$149.10	MIAMFLYJ	\$149.10	NSBHFLMA	\$145.40
JCVLFLLF	\$127.15	LKWLFLXE	\$164.45	MIANFLPV	\$132.85	NWBYFLMA	\$150.95
JCVLFLNO	\$129.05	LKWOFLIY	\$220.67	MIANFLWK	\$132.85	OCALFLDB	\$98.70
JCVLFLOG	\$98.70	LLMNFLXA	\$125.25	MIANFLYI	\$180.00	OCALFLXA	\$98.70
JCVLFLOW	\$134.30	LNLKFLXA	\$160.20	MIAQFL06	\$123.35	OCALFLXB	\$123.35
JCVLFLRV	\$125.25	LRGOFLXA	\$125.25	MIATFLAD	\$125.25	OCALFLXC	\$123.35
JCVLFLSJ	\$129.05	LRHLFLXA	\$187.85	MICCFLBB	\$152.80	OCALFLXJ	\$129.05
JCVLFLSM	\$121.45	LRVLFLXA	\$276.05	MLBRFLMA	\$218.45	OCNFFLXA	\$145.40
JCVLFLWC	\$132.85	LSBGFLXA	\$171.65	MLBYFLXA	\$139.85	OJUSFLTL	\$125.25
JCVLFLWF	\$119.55	LUTZFLXA	\$143.55	MLRSFLXA	\$150.95	OKCBFLXA	\$205.85
JCVMFLED	\$132.85	LVOKFLXA	\$261.65	MLTNFLRA	\$143.55	OKHLFLMA	\$166.25
JCVMFLLI	\$127.15	LWTYFLXA	\$175.25	MNDRFLAV	\$98.70	OKLWFLXA	\$147.25
JCVMFLLR	\$119.55	LYHNFLOH	\$98.70	MNDRFLLO	\$125.25	OLDSFLXA	\$138.00
JCVMFLUF	\$130.95	MALNFLXA	\$184.25	MNDRFLLW	\$132.85	OLTWFLLN	\$189.65
JNGSFLXA	\$276.05	MAYOFLXA	\$285.05	MNLKFLXA	\$168.05	ORCYFLXA	\$143.55
JPTRFLGN	\$300.00	MCINFLXA	\$149.10	MNSNFLMA	\$175.25	ORCYFLXC	\$147.25
JPTRFLMA	\$149.10	MCLNFLXA	\$168.05	MNTIFLXA	\$164.45	ORLDFL60	\$98.70
JSPRFLXA	\$258.05	MCNPFLMA	\$138.00	MOISFLXA	\$205.85	ORLDFLAP	\$129.05
KGLKFLXA	\$173.45	MDBGFLPM	\$156.50	MOLNFLXA	\$240.00	ORLDFLCL	\$121.45
KNVLFLXA	\$169.85	MDSNFLXA	\$213.05	MRDCFLXA	\$173.45	ORLDFLLU	\$136.15
KSSMFLXA	\$150.95	MIAMFLAC	\$121.45	MRHNFLXA	\$209.45	ORLDFLMA	\$98.70
KSSMFLXB	\$147.25	MIAMFLAE	\$125.25	MRNNFLXA	\$164.45	ORLDFLOE	\$98.70
KSSMFLXD	\$147.25	MIAMFLAL	\$121.45	MRTHFLVE	\$279.65	ORLDFLPC	\$129.05
KYHGFLMA	\$156.50	MIAMFLAP	\$125.25	MTDRFLXA	\$189.65	ORLDFLPH	\$129.05
KYLRFLLS	\$205.85	MIAMFLBA	\$123.35	MTLDFLAP	\$129.05	ORLDFLSA	\$130.95

- SERVICE DESCRIPTION (CONT'D.)
- 1.2 Intrastate DS0 (Hubless) Access

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: (Cont'd)

Pre-VBS Charges for low-speed (2.4kps, 4.8kps, 9.6kps)

Wire Center	Rate						
ORLDFLXH	\$119.55	PSDNFLXA	\$129.05	SVSPFLXA	\$127.15	WALDFLXA	\$143.55
ORLEFLGV	\$129.05	PTCTFLXA	\$168.05	SVSSFLXA	\$138.00	WCHLFLXA	\$154.65
ORPKFLMA	\$134.30	PTCYFLXA	\$169.85	SWTHFLXA	\$132.85	WELKFLMA	\$182.45
ORPKFLRW	\$141.70	PTSJFLXA	\$177.05	SYHSFLCC	\$156.50	WHSPFLXA	\$236.45
ORSPFLXA	\$163.90	PTSLFLMA	\$136.15	TAFBFLXA	\$132.85	WIMMFLXA	\$150.95
OSPRFLXA	\$138.00	PTSLFLSO	\$129.05	TAMPFL41	\$125.25	WLBRFLXA	\$243.65
OVIDFLCA	\$136.15	PXTNFLXA	\$193.25	TAMPFL52	\$119.55	WLCHFLXA	\$154.65
PACEFLPV	\$134.30	QNCYFLXA	\$171.65	TAMPFLCM	\$125.25	WLCRFLXA	\$127.15
PAHKFLMA	\$184.25	RAFRFLXA	\$186.05	TAMPFLCO	\$129.05	WLSTFLXA	\$154.65
PANCFLXA	\$171.65	RSKNFLXA	\$143.55	TAMPFLDM	\$119.55	WLWDFLXA	\$160.20
PBGRFLEZ	\$136.15	RYHLFLXA	\$205.85	TAMPFLEM	\$230.00	WNDRFLXA	\$138.00
PCBHFLNT	\$139.85	SARKFLXA	\$123.35	TAMPFLKO	\$154.65	WNGRFLXA	\$141.70
PKCYFLXA	\$134.30	SBNGFLXA	\$136.15	TAMPFLUT	\$130.95	WNHNFLXC	\$127.15
PLBYFLAO	\$123.35	SBSTFLFE	\$147.25	TAMPFLXA	\$193.25	WNPKFLTL	\$127.15
PLCSFLMA	\$166.25	SBSTFLMA	\$141.70	TAMPFLXE	\$129.05	WNPKFLXA	\$125.25
PLMTFLXA	\$147.25	SCPKFLXA	\$141.70	TAMPFLXX	\$98.70	WNPKFLXE	\$127.15
PLSLFLXA	\$141.70	SEKYFLXA	\$127.15	TAMQFLHU	\$149.10	WPBHFL58	\$98.70
PLTKFLMA	\$164.45	SGBEFLXA	\$130.95	TAMQFLPM	\$138.00	WPBHFLAN	\$98.70
PMBHFLCS	\$132.85	SGBHFLXA	\$171.65	TAMQFLRY	\$202.25	WPBHFLGA	\$132.85
PMBHFLDR	\$123.35	SGKYFLMA	\$324.65	TAMQFLSZ	\$130.95	WPBHFLGR	\$136.15
PMBHFLED	\$121.45	SHLMFLXA	\$123.35	TAMQFLTA	\$98.70	WPBHFLHH	\$127.15
PMBHFLFE	\$98.70	SKWYFLXA	\$123.35	TAMSFLZK	\$134.30	WPBHFLLE	\$130.95
PMBHFLJK	\$121.45	SLHLFLXA	\$147.25	THBHFLXA	\$156.50	WPBHFLRB	\$129.05
PMBHFLMA	\$129.05	SLSPFLXA	\$129.05	THNTFLXA	\$141.70	WPBHFLRP	\$138.00
PMBHFLNP	\$129.05	SMNLFLXA	\$123.35	TLCHFLXA	\$207.65	WSSDFLXA	\$123.35
PMBHFLTA	\$127.15	SNANFLXA	\$223.85	TLHSFLXA	\$98.70	WSTVFLXA	\$187.85
PMPKFLMA	\$177.05	SNDSFLXA	\$154.65	TLHSFLXB	\$123.35	WWHTFLXA	\$162.05
PNCRFLXA	\$152.80	SNFRFLMA	\$129.05	TLHSFLXC	\$123.35	WWSPFLHI	\$214.85
PNCYFLCA	\$127.15	SNISFLXA	\$156.50	TLHSFLXD	\$123.35	WWSPFLSH	\$222.05
PNCYFLMA	\$98.70	SNRSFLXA	\$158.35	TLHSFLXE	\$119.55	YBCTFLXA	\$121.45
PNGRFLXA	\$162.05	SNSNFLXA	\$184.25	TLHSFLXF	\$130.95	YNFNFLMA	\$150.95
PNISFLXA	\$147.25	SNSPFLXA	\$152.80	TLHSFLXG	\$136.15	YNTWFLMA	\$182.45

- SERVICE DESCRIPTION (CONT'D.)
- 1.2 Intrastate DS0 (Hubless) Access

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: (Cont'd)

Pre-VBS Charges for low-speed (2.4kps, 4.8kps, 9.6kps)

Wire Center	Rate						
PNLNFLXA	\$198.65	SPBGFLXA	\$98.70	TLHSFLXH	\$129.05	YULEFLMA	\$156.50
PNLSFLXA	\$127.15	SPBGFLXS	\$127.15	TMTRFLXA	\$168.05	ZLSPFLXA	\$154.65
PNSCFLBL	\$130.95	SPRGFLXA	\$129.05	TRENFLMA	\$171.65	ZPHYFLXA	\$158.35
PNSCFLFP	\$98.70	SRSTFLXA	\$98.70	TRSPFLXA	\$141.70		
PNSCFLHC	\$129.05	SSDSFLXA	\$127.15	TTVLFLMA	\$150.95		
PNSCFLPB	\$150.95	SSPRFLXA	\$168.05	TVRSFLXA	\$182.45		
PNSCFLUP	\$130.95	STAGFLBS	\$130.95	UMTLFLXA	\$177.05		
PNSCFLWA	\$132.85	STAGFLMA	\$98.70	UNVRFLXA	\$134.30		
PNVDFLMA	\$138.00	STAGFLSH	\$129.05	VENCFLXA	\$150.95		
POINFLXA	\$156.50	STAGFLWG	\$138.00	VENCFLXS	\$158.35		
PRRNFLCC	\$147.25	STCDFLXA	\$158.35	VERNFLMA	\$166.25		
PRRNFLMA	\$145.40	STGRFLXA	\$132.85	VLPRFLXA	\$136.15		
PRRYFLXA	\$204.05	STMKFLXA	\$156.50	VLPRFLXB	\$141.70		
PRSHFLXA	\$150.95	STRKFLXA	\$186.05	VRBHFLBE	\$123.35		
PRSNFLFD	\$168.05	STRTFLMA	\$98.70	VRBHFLMA	\$98.70		

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: The following monthly recurring per-circuit charges, based on Serving Wire Center of the circuit:

Wire Center	Rate						
ABDLFLXA	\$138.20	BWLGFLXA	\$249.75	DNLNFLWM	\$260.85	FTWBFLXA	\$138.20
ALCHFLXA	\$402.55	BYBHFLAK	\$223.85	DRBHFLDF	\$190.95	FTWBFLXB	\$187.15
ALFAFLXA	\$209.05	BYBHFLMA	\$223.85	DRBHFLMA	\$187.15	FTWBFLXC	\$209.05
ALFRFLXA	\$272.95	BYSHFLXA	\$206.15	DUNDFLXA	\$216.45	FTWHFLXA	\$424.15
ALSPFLXA	\$202.35	CCBHFLAF	\$206.15	DWPKFLXA	\$517.75	GCSPFLCN	\$238.65
ALTHFLXA	\$216.45	CCBHFLMA	\$206.15	DYBHFLFN	\$190.95	GCVLFLMA	\$341.35
ALTRFLXA	\$227.55	CDKYFLMA	\$373.75	DYBHFLMA	\$138.20	GDRGFLXA	\$249.75
ALVAFLXA	\$238.65	CFLDFLMA	\$301.75	DYBHFLOB	\$194.75	GENVFLMA	\$234.95
ANMRFLXA	\$234.95	CFVLFLXA	\$249.75	DYBHFLOS	\$209.05	GLBRFLMC	\$216.45
APLCFLXA	\$355.75	CHLKFLXA	\$362.95	DYBHFLPO	\$198.55	GLDLFLXA	\$330.55
APPKFLXA	\$220.15	CHPLFLJA	\$308.95	DYBHFLWS	\$138.20	GLGCFLXA	\$294.55
ARCDFLXA	\$305.35	CHSWFLXA	\$323.35	EGLLFLBG	\$190.95	GLRDFLXA	\$206.15
ARCHFLMA	\$231.25	CHTHFLXA	\$257.15	EGLLFLIH	\$194.75	GNBOFLXA	\$249.75
ARPNFLXA	\$373.75	CITRFLXA	\$234.95	ENWDFLXA	\$280.15	GNDYFLXA	\$198.55
ASTRFLXA	\$305.35	CLHNFLXA	\$246.05	EORNFLMA	\$246.05	GNVLFLXA	\$319.75
AVPKFLXA	\$138.20	CLMTFLXA	\$352.15	ESPNFLXA	\$355.75	GNWDFLXA	\$287.35
BAKRFLXA	\$276.55	CLTNFLXA	\$388.15	ESTSFLXA	\$305.35	GRETFLXA	\$272.95
BARTFLXA	\$216.45	CLWRFLXA	\$138.20	ETVLFLAA	\$360.00	GSVLFLAN	\$138.20
BAYUFLXA	\$202.35	CNSDFLXA	\$194.75	EVRGFLXA	\$398.95	GSVLFLMA	\$138.20
BBPKFLXA	\$260.85	CNTMFLLE	\$240.55	FHSDFLXA	\$209.05	GSVLFLNW	\$198.55
BCGRFLXA	\$269.35	COCOFLMA	\$138.20	FLBHFLMA	\$246.05	GSVLFLSS	\$179.55
BCRTFLBB	\$194.75	COCOFLME	\$187.15	FLRHFLXA	\$308.95	GVLDFLXA	\$344.95
BCRTFLBT	\$194.75	CPCRFLXA	\$206.15	FRBHFLFP	\$268.25	HAVNFLMA	\$319.75
BCRTFLMA	\$138.20	CPCRFLXB	\$198.55	FRPTFLXA	\$280.15	HBSDFLMA	\$223.85
BCRTFLSA	\$202.35	CPHZFLXA	\$287.35	FRSTFLXA	\$272.95	HDSNFLXA	\$276.55
BGPIFLMA	\$564.55	CRBLFLXA	\$344.95	FTGRFLMA	\$231.25	HGLDFLXA	\$198.55
BHPKFLXA	\$194.75	CRCYFLXA	\$301.75	FTLDFLCR	\$190.95	HGSPFLXA	\$416.95
BKVLFLJF	\$344.95	CRRVFLXA	\$301.75	FTLDFLCY	\$194.75	HLNVFLMA	\$238.65
BLDWFLMA	\$246.05	CRVWFLXA	\$268.25	FTLDFLJA	\$198.55	HLRDFLXA	\$280.15
BLGLFLMA	\$316.15	CRWDFLXA	\$212.75	FTLDFLMR	\$138.20	HLWDFLED	\$198.55
BLTWFLXA	\$138.20	CSCYFLBA	\$348.55	FTLDFLOA	\$138.20	HLWDFLHA	\$194.75
BLVWFLXA	\$212.75	CSLBFLXA	\$206.15	FTLDFLOV	\$183.35	HLWDFLMA	\$198.55
BNFYFLXA	\$312.55	CTDLFLXA	\$290.95	FTLDFLPL	\$187.15	HLWDFLPE	\$202.35
BNNLFLMA	\$257.15	CYGRFLXA	\$209.05	FTLDFLSF	\$234.95	HLWDFLWH	\$194.75
BNSPFLXA	\$257.15	CYLKFLXA	\$209.05	FTLDFLSG	\$209.05	HMSPFLXA	\$312.55

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: The following monthly recurring per-circuit charges, based on Serving Wire Center of the circuit:

Wire Center	Rate						
BORAFLXA	\$470.95	CYLKFLXB	\$209.05	FTLDFLSU	\$190.95	HMSTFLEA	\$276.55
BRBAFLXA	\$206.15	DBRYFLDL	\$216.45	FTLDFLTA	\$138.20	HMSTFLHM	\$276.55
BRFRFLXA	\$463.75	DBRYFLMA	\$209.05	FTLDFLTN	\$179.55	HMSTFLNA	\$264.55
BRJTFLXA	\$246.05	DDCYFLXA	\$380.95	FTLDFLWN	\$220.15	HMSTFLTE	\$280.15
BRKRFLXA	\$238.65	DELDFLMA	\$257.15	FTMBFLXA	\$231.25	HNCYFLXA	\$216.45
BRNDFLXA	\$209.05	DESTFLXA	\$198.55	FTMDFLXA	\$253.45	HNCYFLXN	\$227.55
BRSNFLMA	\$264.55	DFSPFLXA	\$305.35	FTMYFLMA	\$198.55	HOWYFLXA	\$319.75
BRSTFLXA	\$190.95	DLBHFLKP	\$209.05	FTMYFLXA	\$138.20	HSFRFLXA	\$234.95
BRTNFLXX	\$216.45	DLBHFLMA	\$209.05	FTMYFLXB	\$190.95	HSNGFLXA	\$238.65
BSHNFLXA	\$305.35	DLSPFLMA	\$253.45	FTMYFLXC	\$187.15	HTISFLMA	\$198.55
BVHLFLXA	\$138.20	DNDNFLXA	\$194.75	FTPRFLMA	\$452.95	HWTHFLMA	\$234.95
HYPKFLXA	\$179.55	KYSTFLXA	\$227.55	MIAMFLCA	\$212.75	MTVRFLXA	\$246.05
IMKLFLXA	\$290.95	KYWSFLMA	\$654.55	MIAMFLDA	\$187.15	MXVLFLMA	\$260.85
INLKFLXA	\$294.55	LBLLFLXA	\$280.15	MIAMFLDB	\$206.15	MYCYFLXA	\$264.55
INRKFLXX	\$202.35	LDLKFLXA	\$260.85	MIAMFLFL	\$183.35	NDADFLAA	\$187.15
INTRFLXA	\$319.75	LGBKFLXA	\$206.15	MIAMFLGR	\$183.35	NDADFLAC	\$187.15
INTWFLXA	\$246.05	LHACFLXA	\$231.25	MIAMFLHL	\$209.05	NDADFLBR	\$190.95
INVRFLXA	\$272.95	LKALFLXA	\$190.95	MIAMFLIC	\$206.15	NDADFLGG	\$138.20
IONAFLXA	\$220.15	LKBNFLXB	\$231.25	MIAMFLKE	\$209.05	NDADFLOL	\$187.15
ISLMFLMA	\$413.35	LKBRFLXA	\$202.35	MIAMFLME	\$138.20	NFMYFLXA	\$187.15
JAY FLMA	\$287.35	LKBTFLXA	\$341.35	MIAMFLNM	\$190.95	NFMYFLXB	\$194.75
JCBHFLAB	\$216.45	LKCYFLMA	\$391.75	MIAMFLNS	\$190.95	NGBHFLXA	\$209.05
JCBHFLMA	\$223.85	LKHLFLXA	\$238.65	MIAMFLOL	\$194.75	NKLRFLMA	\$298.15
JCBHFLSP	\$209.05	LKLDFLXA	\$194.75	MIAMFLPB	\$194.75	NNPLFLXA	\$272.95

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: (Cont'd)

Wire Center	Rate						
JCVLFLAI	\$187.15	LKLDFLXE	\$373.75	MIAMFLPL	\$209.05	NPLSFLXC	\$312.55
JCVLFLAR	\$198.55	LKLDFLXN	\$206.15	MIAMFLRR	\$206.15	NPLSFLXD	\$301.75
JCVLFLBW	\$202.35	LKMRFL01	\$138.20	MIAMFLSH	\$194.75	NPRCFLXA	\$260.85
JCVLFLCL	\$138.20	LKMRFLHE	\$138.20	MIAMFLSO	\$223.85	NRPTFLXA	\$272.95
JCVLFLFC	\$198.55	LKMRFLMA	\$138.20	MIAMFLWD	\$234.95	NRSDFLXA	\$187.15
JCVLFLGH	\$198.55	LKMYFLMA	\$202.35	MIAMFLWM	\$198.55	NSBHFLMA	\$231.25
JCVLFLIA	\$216.45	LKPCFLXA	\$264.55	MIAMFLYJ	\$238.65	NWBYFLMA	\$242.35
JCVLFLJB	\$138.20	LKWLFLXA	\$238.65	MIANFLPV	\$206.15	OCALFLDB	\$138.20
JCVLFLJT	\$190.95	LKWLFLXE	\$269.35	MIANFLWK	\$206.15	OCALFLXA	\$138.20
JCVLFLLF	\$194.75	LKWOFLIY	\$220.67	MIANFLYI	\$180.00	OCALFLXB	\$187.15
JCVLFLNO	\$198.55	LLMNFLXA	\$190.95	MIAQFL06	\$187.15	OCALFLXC	\$187.15
JCVLFLOG	\$138.20	LNLKFLXA	\$260.85	MIATFLAD	\$190.95	OCALFLXJ	\$198.55
JCVLFLOW	\$209.05	LRGOFLXA	\$190.95	MICCFLBB	\$246.05	OCNFFLXA	\$231.25
JCVLFLRV	\$190.95	LRHLFLXA	\$316.15	MLBRFLMA	\$377.35	OJUSFLTL	\$190.95
JCVLFLSJ	\$198.55	LRVLFLXA	\$492.55	MLBYFLXA	\$220.15	OKCBFLXA	\$352.15
JCVLFLSM	\$183.35	LSBGFLXA	\$283.75	MLRSFLXA	\$242.35	OKHLFLMA	\$272.95
JCVLFLWC	\$206.15	LUTZFLXA	\$227.55	MLTNFLRA	\$227.55	OKLWFLXA	\$234.95
JCVLFLWF	\$179.55	LVOKFLXA	\$463.75	MNDRFLAV	\$138.20	OLDSFLXA	\$216.45
JCVMFLED	\$206.15	LWTYFLXA	\$290.95	MNDRFLLO	\$190.95	OLTWFLLN	\$319.75
JCVMFLLI	\$194.75	LYHNFLOH	\$138.20	MNDRFLLW	\$206.15	ORCYFLXA	\$227.55
JCVMFLLR	\$179.55	MALNFLXA	\$308.95	MNLKFLXA	\$276.55	ORCYFLXC	\$234.95
JCVMFLUF	\$202.35	MAYOFLXA	\$510.55	MNSNFLMA	\$290.95	ORLDFL60	\$138.20
JNGSFLXA	\$492.55	MCINFLXA	\$238.65	MNTIFLXA	\$269.35	ORLDFLAP	\$198.55
JPTRFLGN	\$300.00	MCLNFLXA	\$276.55	MOISFLXA	\$352.15	ORLDFLCL	\$183.35
JPTRFLMA	\$238.65	MCNPFLMA	\$216.45	MOLNFLXA	\$240.00	ORLDFLLU	\$212.75
JSPRFLXA	\$456.55	MDBGFLPM	\$253.45	MRDCFLXA	\$287.35	ORLDFLMA	\$138.20
KGLKFLXA	\$287.35	MDSNFLXA	\$366.55	MRHNFLXA	\$359.35	ORLDFLOE	\$138.20
KNVLFLXA	\$280.15	MIAMFLAC	\$183.35	MRNNFLXA	\$269.35	ORLDFLPC	\$198.55
KSSMFLXA	\$242.35	MIAMFLAE	\$190.95	MRTHFLVE	\$499.75	ORLDFLPH	\$198.55
KSSMFLXB	\$234.95	MIAMFLAL	\$183.35	MTDRFLXA	\$319.75	ORLDFLSA	\$202.35
KSSMFLXD	\$234.95	MIAMFLAP	\$190.95	MTLDFLAP	\$198.55	ORLDFLXH	\$179.55
KYHGFLMA	\$253.45	MIAMFLBA	\$187.15	MTLDFLBR	\$198.55	ORLEFLGV	\$198.55
KYLRFLLS	\$352.15	MIAMFLBC	\$183.35	MTLDFLTC	\$202.35	ORPKFLMA	\$209.05
KYLRFLMA	\$373.75	MIAMFLBR	\$194.75	MTLDFLXA	\$202.35	ORPKFLRW	\$223.85

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: (Cont'd)

Wire Center	Rate						
ORSPFLXA	\$268.25	PTSLFLMA	\$212.75	TAFBFLXA	\$206.15	WIMMFLXA	\$242.35
OSPRFLXA	\$216.45	PTSLFLSO	\$198.55	TAMPFL41	\$190.95	WLBRFLXA	\$427.75
OVIDFLCA	\$212.75	PXTNFLXA	\$326.95	TAMPFL52	\$179.55	WLCHFLXA	\$249.75
PACEFLPV	\$209.05	QNCYFLXA	\$283.75	TAMPFLCM	\$190.95	WLCRFLXA	\$194.75
PAHKFLMA	\$308.95	RAFRFLXA	\$312.55	TAMPFLCO	\$198.55	WLSTFLXA	\$249.75
PANCFLXA	\$283.75	RSKNFLXA	\$227.55	TAMPFLDM	\$179.55	WLWDFLXA	\$260.85
PBGRFLEZ	\$212.75	RYHLFLXA	\$352.15	TAMPFLEM	\$230.00	WNDRFLXA	\$216.45
PCBHFLNT	\$220.15	SARKFLXA	\$187.15	TAMPFLKO	\$249.75	WNGRFLXA	\$223.85
PKCYFLXA	\$209.05	SBNGFLXA	\$212.75	TAMPFLUT	\$202.35	WNHNFLXC	\$194.75
PLBYFLAO	\$187.15	SBSTFLFE	\$234.95	TAMPFLXA	\$326.95	WNPKFLTL	\$194.75
PLCSFLMA	\$272.95	SBSTFLMA	\$223.85	TAMPFLXE	\$198.55	WNPKFLXA	\$190.95
PLMTFLXA	\$234.95	SCPKFLXA	\$223.85	TAMPFLXX	\$138.20	WNPKFLXE	\$194.75
PLSLFLXA	\$223.85	SEKYFLXA	\$194.75	TAMQFLHU	\$238.65	WPBHFL58	\$138.20
PLTKFLMA	\$269.35	SGBEFLXA	\$202.35	TAMQFLPM	\$216.45	WPBHFLAN	\$138.20
PMBHFLCS	\$206.15	SGBHFLXA	\$283.75	TAMQFLRY	\$344.95	WPBHFLGA	\$206.15
PMBHFLDR	\$187.15	SGKYFLMA	\$589.75	TAMQFLSZ	\$202.35	WPBHFLGR	\$212.75
PMBHFLED	\$183.35	SHLMFLXA	\$187.15	TAMQFLTA	\$138.20	WPBHFLHH	\$194.75
PMBHFLFE	\$138.20	SKWYFLXA	\$187.15	TAMSFLZK	\$209.05	WPBHFLLE	\$202.35
PMBHFLJK	\$183.35	SLHLFLXA	\$234.95	THBHFLXA	\$253.45	WPBHFLRB	\$198.55
PMBHFLMA	\$198.55	SLSPFLXA	\$198.55	THNTFLXA	\$223.85	WPBHFLRP	\$216.45
PMBHFLNP	\$198.55	SMNLFLXA	\$187.15	TLCHFLXA	\$355.75	WSSDFLXA	\$187.15
PMBHFLTA	\$194.75	SNANFLXA	\$388.15	TLHSFLXA	\$138.20	WSTVFLXA	\$316.15
PMPKFLMA	\$294.55	SNDSFLXA	\$249.75	TLHSFLXB	\$187.15	WWHTFLXA	\$264.55
PNCRFLXA	\$246.05	SNFRFLMA	\$198.55	TLHSFLXC	\$187.15	WWSPFLHI	\$370.15
PNCYFLCA	\$194.75	SNISFLXA	\$253.45	TLHSFLXD	\$187.15	WWSPFLSH	\$384.55
PNCYFLMA	\$138.20	SNRSFLXA	\$257.15	TLHSFLXE	\$179.55	YBCTFLXA	\$183.35
PNGRFLXA	\$264.55	SNSNFLXA	\$308.95	TLHSFLXF	\$202.35	YNFNFLMA	\$242.35
PNISFLXA	\$234.95	SNSPFLXA	\$246.05	TLHSFLXG	\$212.75	YNTWFLMA	\$305.35
PNLNFLXA	\$337.75	SPBGFLXA	\$138.20	TLHSFLXH	\$198.55	YULEFLMA	\$253.45
PNLSFLXA	\$194.75	SPBGFLXS	\$194.75	TMTRFLXA	\$276.55	ZLSPFLXA	\$249.75

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.2 Intrastate DS0 (Hubless) Access

1.2.2.1. Local Access Channel Monthly Recurring Channel Charges (Cont'd)

1.2.2.1.4 Pre-VBS Charges: (Cont'd)

Pre-VBS Charges for higher speed (56kps, 64kps)

Wire Center	Rate						
PNSCFLBL	\$202.35	SPRGFLXA	\$198.55	TRENFLMA	\$283.75	ZPHYFLXA	\$257.15
PNSCFLFP	\$138.20	SRSTFLXA	\$138.20	TRSPFLXA	\$223.85		
PNSCFLHC	\$198.55	SSDSFLXA	\$194.75	TTVLFLMA	\$242.35		
PNSCFLPB	\$242.35	SSPRFLXA	\$276.55	TVRSFLXA	\$305.35		
PNSCFLUP	\$202.35	STAGFLBS	\$202.35	UMTLFLXA	\$294.55		
PNSCFLWA	\$206.15	STAGFLMA	\$138.20	UNVRFLXA	\$209.05		
PNVDFLMA	\$216.45	STAGFLSH	\$198.55	VENCFLXA	\$242.35		
POINFLXA	\$253.45	STAGFLWG	\$216.45	VENCFLXS	\$257.15		
PRRNFLCC	\$234.95	STCDFLXA	\$257.15	VERNFLMA	\$272.95		
PRRNFLMA	\$231.25	STGRFLXA	\$206.15	VLPRFLXA	\$212.75		
PRRYFLXA	\$348.55	STMKFLXA	\$253.45	VLPRFLXB	\$223.85		
PRSHFLXA	\$242.35	STRKFLXA	\$312.55	VRBHFLBE	\$187.15		
PRSNFLFD	\$276.55	STRTFLMA	\$138.20	VRBHFLMA	\$138.20		
PSDNFLXA	\$198.55	SVSPFLXA	\$194.75	WALDFLXA	\$227.55		
PTCTFLXA	\$276.55	SVSSFLXA	\$216.45	WCHLFLXA	\$249.75		
PTCYFLXA	\$280.15	SWTHFLXA	\$206.15	WELKFLMA	\$305.35		
PTSJFLXA	\$294.55	SYHSFLCC	\$253.45	WHSPFLXA	\$413.35		

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.2 <u>Intrastate DS0 (Hubless) Access</u> (Cont'd)
    - 1.2.2 Rates and Charges (Cont'd):
      - 1.2.2.2. <u>Local Access Channel Non-Recurring Charges</u>: A \$200 non-recurring channel charge applies.
      - 1.2.2.3. <u>Network Connection Charge</u>: The following monthly recurring and non-recurring charges will apply, per connection, only for circuits installed on or after October 1, 1998, if access is Customer-provided.

Monthly Installation
\$60 \$0.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.2 <u>Intrastate DS0 (Hubless) Access (Cont'd.)</u>
    - 1.2.2 Rates and Charges (Cont'd.):
      - 1.2.2.4. Other Non-Recurring Charges:
        - 1.2.2.4.1 Customer Support Services charges apply.
        - 1.2.2.4.2 Intrastate Administrative Non-Recurring Charges apply.
    - 1.2.3 Terms and conditions: The tariff general terms and conditions apply, except that the Credit Allowance for Interruption of Local Access Service as set forth in Section 2.3 below also applies.
  - 1.3 Intrastate T1 Digital Access
    - 1.3.1. <u>Product Description</u>: Intrastate T-1 Digital Access service is provided by Company. Intrastate T-1 Digital Access is a high capacity digital local access arrangement that relies on T-1 transmission technology provided by the local exchange carrier and which will be used to connect Customer's Premises to the Company's Point-of-Presence (POP) to provide Customer access to Company services.
    - 1.3.2 Features and Options:
      - 1.3.2.1. <u>Channelized Access Line with Telco M24 Multiplexer</u>: Charges will apply on an individual case basis.
      - 1.3.2.2. <u>Access Integration Option</u>: A form of T-1 Digital Access which allows Customers to utilize their dedicated access lines to carry traffic for both an inbound and an outbound service over the same circuits.

<u>Monthly</u>	<u>Installation</u>
\$0.00	\$0.00

- 1. SERVICE DESCRIPTION (CONT'D.)
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.2 <u>Features and Options (Cont'd.)</u>:
    - 1.3.2.3. <u>Integrated Services Digital Network (ISDN) Service</u>: Intrastate ISDN integrates voice, data, and video communications services via standard interfaces.
      - 1.3.2.3.1 Primary Rate Interface (PRI): Through Intrastate PRI, a
        Customer can share access facilities among the following Intrastate Long
        Distance Voice Services of MCI Communications Services: outbound
        service and inbound toll-free services. An attribute of PRI, Call-by-Call
        Service Configuration, allows for these services to share dynamically
        allocated individual circuits within the PRI. Intrastate Private Line
        Service is also available with the above combinations.

The PRI consists of a 64 kbps D channel and 23 B channels of 64 kbps each. The bearer, or B, channels are used to access Company services supported over the PRI. The D channels are used to carry signaling and control information for the associated B channels.

In addition to providing basic access to the services specified above, the PRI interface can be configured to support the following advanced ISDN features: Call-by-Call Service Configuration, Non-Facility Associated Signaling, D-Channel Backup, Automatic Number Identification (ANI), Dialed Number identification Service (DNIS) and, for intrastate outbound service with Feature Option C-1, Calling Station Identification (CSI). Call-by-Call Service Configuration allows the B channels of a PRI to be shared between multiple services carried on the PRI. With Non-Facility Associated Signaling, one PRI D channel can control up to a maximum of 20 T-1 Digital Access channels. D Channel Backup provides a redundant hot stand-by of the PRIs primary D Channel. Automatic Number Identification (ANI) allows a Customer to receive the originating caller's working telephone number provided this information is forwarded to the Company by the caller's Local Exchange Carrier (LEC). DNIS provides the called party with the number dialed by the Customer or a translated version of the dialed number. The delivery of ANI and DNIS is ordered along with intrastate outbound service with Feature Option C-1 CSI is provided to intrastate outbound service with Feature Option C-1 where Customers order PRI. Intrastate T-1 Digital Access is required to obtain intrastate ISDN PRI. Each time a change in the number of Private Line DS0s is required, a PRI reconfiguration charge will be assessed.

After initial installation of the PRI trunk group, a Customer may choose to add or delete groups of B channels in DS1 blocks. Each time the trunk group size is changed, a trunk group rearrangement charge will be assessed.

Customers who elect PRI will be assessed the following monthly recurring charges, in addition to the rates in Section 1.3.3 below:

Feature Monthly
Per D Channel: \$110.00

Rearrange trunk group B channels per occurrence: \$200.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
    - 1.3.2 <u>Features and Options (Cont'd.)</u>:
      - 1.3.2.3. Integrated Services Digital Network (ISDN) Service (Cont'd.):
        - 1.3.2.3.2 <u>Call-by-Call Service Configuration</u>: Allows the B channels within a PRI to be shared between services subscribed to on the PRI. Call-by-Call Service Configuration can be used in the following Intrastate Long Distance Voice Services of MCI Communications Services: outbound services and inbound toll-free service. The Customer must specify at the time of PRI order if Call-by-Call Service Configuration is desired and the service combination. The Customer also becomes a Customer of each service. If the PRI is initially dedicated to a single service, it can be reconfigured to support Call-by-Call Service Configuration, and the rearrangement charge specified below will apply. Each time a service is added or deleted from an existing PRI trunk group, a service rearrangement charge will be assessed.

Customers who elect Call-by-Call Service Configuration will be assessed the following monthly recurring charges, in addition to the rates in Section 1.3.3 below:

Feature Monthly

Call-by-Call: \$ 0.00

Trunk group rearrangement per-occurrence: \$200.00

- 1.3.2.4. Access Pricing Plans (APP) as described below in Section 2.1 are available to Pre-Verizon Business Services I Customers only.
- 1.3.2.5. Intrastate Access Term Discounts as described below in Section 2.2 are available to Pre-Verizon Business Services I Customers only.

## 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>

## 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:

#### 1.3.3 Rates and Charges:

Where rates or charges are specific to a particular pricing program (i.e., Verizon Business Services (VBS) I, II, or III) those rates and charges only apply where a Customer has enrolled in that pricing program. Rates and charges which are not specific to a particular pricing program apply to all Customers.

## 1.3.3.1. Local Access Channel Monthly Recurring Channel Charges:

1.3.3.1.1 <u>Verizon Business Services III Charges:</u> The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
ABDLFLXA	\$370.00	BRNDFLXA	\$220.00	CYGRFLXA	\$380.00	FRPTFLXA	\$860.00
ALCHFLXA	\$570.00	BRSNFLMA	\$540.00	CYLKFLXA	\$320.00	FRSTFLXA	\$490.00
ALFAFLXA	\$270.00	BRTNFLXX	\$220.00	CYLKFLXB	\$360.00	FTGRFLMA	\$390.00
ALFRFLXA	\$490.00	BSHNFLXA	\$850.00	DBRYFLDL	\$310.00	FTLDFLCR	\$220.00
ALSPFLXA	\$240.00	BVHLFLXA	\$640.00	DBRYFLMA	\$400.00	FTLDFLCY	\$210.00
ALTRFLXA	\$420.00	BWLGFLXA	\$570.00	DDCYFLXA	\$990.00	FTLDFLJA	\$230.00
ALVAFLXA	\$540.00	BYBHFLMA	\$280.00	DELDFLMA	\$510.00	FTLDFLMR	\$190.00
ANMRFLXA	\$270.00	BYSHFLXA	\$230.00	DESTFLXA	\$620.00	FTLDFLOA	\$230.00
APPKFLXA	\$330.00	CCBHFLAF	\$360.00	DFSPFLXA	\$900.00	FTLDFLPL	\$210.00
ARCDFLXA	\$700.00	CCBHFLMA	\$270.00	DLBHFLKP	\$240.00	FTLDFLSF	\$370.00
ARCHFLMA	\$440.00	CDKYFLMA	\$890.00	DLBHFLMA	\$270.00	FTLDFLSG	\$230.00
ASTRFLXA	\$900.00	CFLDFLMA	\$660.00	DLSPFLMA	\$500.00	FTLDFLSU	\$230.00
AVPKFLXA	\$240.00	CFVLFLXA	\$570.00	DNDNFLXA	\$220.00	FTLDFLTA	\$370.00
BAKRFLXA	\$620.00	CHLKFLXA	\$860.00	DNLNFLWM	\$670.00	FTLDFLWN	\$240.00
BARTFLXA	\$360.00	CHPLFLJA	\$230.00	DRBHFLDF	\$500.00	FTMBFLXA	\$520.00
BAYUFLXA	\$210.00	CHSWFLXA	\$770.00	DRBHFLMA	\$210.00	FTMDFLXA	\$580.00
BBPKFLXA	\$480.00	CHTHFLXA	\$1,030.00	DUNDFLXA	\$380.00	FTMYFLMA	\$490.00
BCGRFLXA	\$620.00	CITRFLXA	\$340.00	DWPKFLXA	\$520.00	FTMYFLXA	\$270.00
BCRTFLBT	\$230.00	CLHNFLXA	\$310.00	DYBHFLFN	\$220.00	FTMYFLXB	\$420.00
BCRTFLMA	\$190.00	CLMTFLXA	\$1,020.00	DYBHFLMA	\$190.00	FTMYFLXC	\$340.00
BCRTFLSA	\$220.00	CLTNFLXA	\$920.00	DYBHFLMN	\$500.00	FTPRFLMA	\$230.00
BGPIFLMA	\$820.00	CLWRFLXA	\$170.00	DYBHFLOB	\$220.00	FTWBFLXA	\$590.00
BHPKFLXA	\$170.00	CNSDFLXA	\$210.00	DYBHFLOS	\$320.00	FTWBFLXB	\$610.00
BKVLFLJF	\$920.00	CNTMFLLE	\$430.00	DYBHFLPO	\$220.00	FTWBFLXC	\$550.00
BLDWFLMA	\$460.00	COCOFLMA	\$190.00	DYBHFLWS	\$610.00	FTWHFLXA	\$510.00
BLGLFLMA	\$690.00	COCOFLME	\$300.00	EGLLFLBG	\$260.00	GCSPFLCN	\$430.00
BLTWFLXA	\$970.00	CPCRFLXA	\$410.00	EGLLFLIH	\$330.00	GCVLFLMA	\$460.00
BLVWFLXA	\$660.00	CPCRFLXB	\$440.00	ENWDFLXA	\$470.00	GDRGFLXA	\$670.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.1 <u>Verizon Business Services III Charges:</u> (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
BNFYFLXA	\$480.00	CPHZFLXA	\$610.00	EORNFLMA	\$440.00	GENVFLMA	\$460.00
BNNLFLMA	\$450.00	CRCYFLXA	\$440.00	ESTSFLXA	\$900.00	GLBRFLMC	\$370.00
BNSPFLXA	\$600.00	CRRVFLXA	\$690.00	ETVLFLAA	\$770.00	GLDLFLXA	\$940.00
BORAFLXA	\$450.00	CRVWFLXA	\$710.00	EVRGFLXA	\$840.00	GLGCFLXA	\$570.00
BRBAFLXA	\$210.00	CRWDFLXA	\$200.00	FHSDFLXA	\$170.00	GLRDFLXA	\$330.00
BRFRFLXA	\$550.00	CSCYFLBA	\$800.00	FLBHFLMA	\$440.00	GNBOFLXA	\$830.00
BRJTFLXA	\$380.00	CSLBFLXA	\$360.00	FLRHFLXA	\$380.00	GNDYFLXA	\$170.00
BRKRFLXA	\$320.00	CTDLFLXA	\$470.00	FRBHFLFP	\$510.00	GNVLFLXA	\$750.00
GNWDFLXA	\$590.00	JCVLFLIA	\$230.00	LRGOFLXA	\$220.00	MLTNFLRA	\$290.00
GRETFLXA	\$750.00	JCVLFLJT	\$210.00	LRVLFLXA	\$470.00	MNDRFLAV	\$210.00
GSVLFLAN	\$370.00	JCVLFLLF	\$280.00	LSBGFLXA	\$430.00	MNDRFLLO	\$210.00
GSVLFLMA	\$190.00	JCVLFLNO	\$220.00	LUTZFLXA	\$260.00	MNDRFLLW	\$290.00
GSVLFLNW	\$210.00	JCVLFLOG	\$490.00	LVOKFLXA	\$310.00	MNLKFLXA	\$360.00
GVLDFLXA	\$1,000.00	JCVLFLOW	\$230.00	LWTYFLXA	\$500.00	MNSNFLMA	\$580.00
HAVNFLMA	\$430.00	JCVLFLRV	\$230.00	LYHNFLOH	\$350.00	MNTIFLXA	\$630.00
HBSDFLMA	\$420.00	JCVLFLSJ	\$210.00	MALNFLXA	\$620.00	MOISFLXA	\$650.00
HDSNFLXA	\$290.00	JCVLFLSM	\$210.00	MAYOFLXA	\$570.00	MOLNFLXA	\$410.00
HGLDFLXA	\$350.00	JCVLFLWC	\$240.00	MCINFLXA	\$310.00	MRDCFLXA	\$530.00
HGSPFLXA	\$570.00	JCVLFLWF	\$370.00	MCLNFLXA	\$630.00	MRHNFLXA	\$850.00
HLNVFLMA	\$300.00	JCVMFLLI	\$500.00	MCNPFLMA	\$390.00	MRNNFLXA	\$560.00
HLRDFLXA	\$400.00	JCVMFLUF	\$500.00	MDBGFLPM	\$490.00	MRTHFLVE	\$490.00
HLWDFLED	\$480.00	JNGSFLXA	\$590.00	MDSNFLXA	\$860.00	MTDRFLXA	\$930.00
HLWDFLHA	\$290.00	JPTRFLGN	\$270.00	MIAMFLAE	\$190.00	MTLDFLAP	\$490.00
HLWDFLMA	\$220.00	JPTRFLMA	\$250.00	MIAMFLAL	\$210.00	MTLDFLXA	\$250.00
HLWDFLPE	\$280.00	JSPRFLXA	\$520.00	MIAMFLAP	\$220.00	MTVRFLXA	\$510.00
HLWDFLWH	\$290.00	KGLKFLXA	\$480.00	MIAMFLBA	\$210.00	MXVLFLMA	\$490.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.1 <u>Verizon Business Services III Charges:</u> (Cont'd)

Wire Center	Rate						
HMSPFLXA	\$740.00	KNVLFLXA	\$710.00	MIAMFLBC	\$270.00	MYCYFLXA	\$370.00
HMSTFLEA	\$350.00	KSSMFLXA	\$350.00	MIAMFLBR	\$220.00	NDADFLAA	\$480.00
HMSTFLHM	\$260.00	KSSMFLXB	\$270.00	MIAMFLCA	\$210.00	NDADFLAC	\$200.00
HMSTFLNA	\$390.00	KSSMFLXD	\$450.00	MIAMFLDB	\$280.00	NDADFLBR	\$260.00
HMSTFLTE	\$930.00	KYHGFLMA	\$500.00	MIAMFLDI	\$370.00	NDADFLGG	\$210.00
HNCYFLXA	\$410.00	KYLRFLLS	\$610.00	MIAMFLFL	\$220.00	NDADFLOL	\$210.00
HNCYFLXN	\$490.00	KYLRFLMA	\$670.00	MIAMFLGR	\$190.00	NFMYFLXA	\$410.00
HOWYFLXA	\$930.00	KYSTFLXA	\$250.00	MIAMFLHL	\$220.00	NFMYFLXB	\$430.00
HSNGFLXA	\$350.00	KYWSFLMA	\$420.00	MIAMFLIC	\$290.00	NGBHFLXA	\$210.00
HTISFLMA	\$420.00	LBLLFLXA	\$640.00	MIAMFLKE	\$290.00	NKLRFLMA	\$530.00
HWTHFLMA	\$450.00	LDLKFLXA	\$780.00	MIAMFLME	\$190.00	NNPLFLXA	\$380.00
HYPKFLXA	\$220.00	LEE FLXA	\$850.00	MIAMFLNM	\$220.00	NPLSFLXC	\$550.00
IMKLFLXA	\$660.00	LGBKFLXA	\$250.00	MIAMFLNS	\$210.00	NPLSFLXD	\$400.00
INLKFLXA	\$550.00	LHACFLXA	\$520.00	MIAMFLOL	\$300.00	NPRCFLXA	\$280.00
INRKFLXX	\$240.00	LKALFLXA	\$420.00	MIAMFLPB	\$210.00	NRPTFLXA	\$440.00
INTRFLXA	\$400.00	LKBNFLXB	\$220.00	MIAMFLPL	\$190.00	NRSDFLXA	\$240.00
INTWFLXA	\$630.00	LKBRFLXA	\$320.00	MIAMFLRR	\$220.00	NSBHFLMA	\$260.00
INVRFLXA	\$700.00	LKBTFLXA	\$460.00	MIAMFLSH	\$290.00	NWBYFLMA	\$470.00
IONAFLXA	\$490.00	LKCYFLMA	\$320.00	MIAMFLSO	\$210.00	OCALFLXA	\$440.00
ISLMFLMA	\$770.00	LKHLFLXA	\$500.00	MIAMFLUJ	\$340.00	OCALFLXB	\$550.00
JAY FLMA	\$620.00	LKLDFLXA	\$240.00	MIAMFLWD	\$310.00	OCALFLXC	\$600.00
JCBHFLAB	\$300.00	LKLDFLXE	\$190.00	MIAMFLWM	\$190.00	OCALFLXJ	\$630.00
JCBHFLMA	\$240.00	LKLDFLXN	\$330.00	MIANFLWK	\$370.00	OCNFFLXA	\$640.00
JCBHFLSP	\$300.00	LKMRFLHE	\$280.00	MIANFLYI	\$370.00	OJUSFLTL	\$740.00
JCVLFLAI	\$640.00	LKMRFLMA	\$310.00	MIAQFL06	\$500.00	OKCBFLXA	\$820.00
JCVLFLAR	\$210.00	LKPCFLXA	\$560.00	MIATFLAD	\$370.00	OKHLFLMA	\$470.00
JCVLFLBW	\$210.00	LKWLFLXA	\$380.00	MICCFLBB	\$430.00	OKLWFLXA	\$710.00
JCVLFLCL	\$190.00	LKWLFLXE	\$510.00	MLBRFLMA	\$190.00	OLDSFLXA	\$240.00
JCVLFLFC	\$290.00	LLMNFLXA	\$200.00	MLBYFLXA	\$370.00	OLTWFLLN	\$710.00
JCVLFLGH	\$640.00	LNLKFLXA	\$330.00	MLRSFLXA	\$290.00	ORCYFLXA	\$540.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.1 <u>Verizon Business Services III Charges:</u> (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
ORCYFLXC	\$470.00	PNSCFLHC	\$350.00	STAGFLBS	\$390.00	VLPRFLXA	\$680.00
ORLDFL60	\$370.00	PNSCFLPB	\$400.00	STAGFLMA	\$230.00	VLPRFLXB	\$700.00
ORLDFLAP	\$220.00	PNSCFLWA	\$360.00	STAGFLSH	\$380.00	VRBHFLBE	\$370.00
ORLDFLCL	\$190.00	PNVDFLMA	\$250.00	STAGFLWG	\$450.00	VRBHFLMA	\$290.00
ORLDFLLU	\$370.00	POINFLXA	\$530.00	STCDFLXA	\$520.00	WALDFLXA	\$290.00
ORLDFLMA	\$190.00	PRRNFLCC	\$480.00	STGRFLXA	\$220.00	WCHLFLXA	\$560.00
ORLDFLOE	\$370.00	PRRNFLMA	\$210.00	STMKFLXA	\$580.00	WELKFLMA	\$530.00
ORLDFLPC	\$190.00	PRRYFLXA	\$1,210.00	STRKFLXA	\$540.00	WHSPFLXA	\$460.00
ORLDFLPH	\$280.00	PRSHFLXA	\$300.00	STRTFLMA	\$290.00	WIMMFLXA	\$310.00
ORLDFLSA	\$190.00	PRSNFLFD	\$570.00	SVSPFLXA	\$600.00	WLBRFLXA	\$450.00
ORLDFLSO	\$310.00	PSDNFLXA	\$210.00	SVSSFLXA	\$670.00	WLCHFLXA	\$310.00
ORLDFLWR	\$310.00	PTCTFLXA	\$720.00	SWTHFLXA	\$170.00	WLCRFLXA	\$200.00
ORLDFLXH	\$370.00	PTCYFLXA	\$320.00	SYHSFLCC	\$480.00	WLHLFLXA	\$600.00
ORLEFLGV	\$490.00	PTSJFLXA	\$1,070.00	TAFBFLXA	\$520.00	WLSTFLXA	\$480.00
ORLEFLKT	\$770.00	PTSLFLMA	\$280.00	TAMPFL52	\$500.00	WLWDFLXA	\$760.00
ORLFFLEJ	\$770.00	PTSLFLSO	\$290.00	TAMPFLCM	\$190.00	WNDRFLXA	\$390.00
ORLFFLHX	\$770.00	QNCYFLXA	\$410.00	TAMPFLCO	\$190.00	WNGRFLXA	\$390.00
ORLFFLQR	\$770.00	RAFRFLXA	\$420.00	TAMPFLDM	\$190.00	WNHNFLBU	\$400.00
ORPKFLMA	\$290.00	RSKNFLXA	\$290.00	TAMPFLEM	\$190.00	WNHNFLXC	\$360.00
ORPKFLRW	\$400.00	RYHLFLXA	\$600.00	TAMPFLKO	\$190.00	WNPKFLTL	\$290.00
ORSPFLXA	\$400.00	SARKFLXA	\$310.00	TAMPFLXA	\$190.00	WNPKFLXA	\$250.00
OSPRFLXA	\$370.00	SBNGFLXA	\$460.00	TAMPFLXE	\$170.00	WNPKFLXE	\$290.00
OVIDFLCA	\$370.00	SBSTFLFE	\$470.00	TAMPFLXX	\$170.00	WPBHFL58	\$370.00
PACEFLPV	\$410.00	SBSTFLMA	\$290.00	TAMQFLHU	\$770.00	WPBHFLAN	\$190.00
PAHKFLMA	\$690.00	SCPKFLXA	\$500.00	TAMQFLPM	\$770.00	WPBHFLGA	\$290.00
PANCFLXA	\$660.00	SEKYFLXA	\$330.00	TAMQFLRY	\$770.00	WPBHFLGR	\$240.00
PCBHFLNT	\$290.00	SGBEFLXA	\$230.00	TAMQFLTA	\$770.00	WPBHFLHH	\$220.00
PKCYFLXA	\$410.00	SGBHFLXA	\$870.00	TAMSFLCZ	\$770.00	WPBHFLLE	\$290.00
PLCSFLMA	\$270.00	SGKYFLMA	\$720.00	THNTFLXA	\$270.00	WPBHFLRB	\$220.00
PLMTFLXA	\$270.00	SHLMFLXA	\$620.00	TLCHFLXA	\$930.00	WPBHFLRP	\$320.00
PLSLFLXA	\$220.00	SKWYFLXA	\$240.00	TLHSFLXA	\$280.00	WSSDFLXA	\$170.00
PLTKFLMA	\$300.00	SLHLFLXA	\$520.00	TLHSFLXB	\$320.00	WSTVFLXA	\$560.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 Intrastate T1 Digital Access (Cont'd.):
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.1 <u>Verizon Business Services III Charges:</u> (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
PMBHFLCS	\$210.00	SLSPFLXA	\$230.00	TLHSFLXC	\$400.00	WWSPFLHI	\$740.00
PMBHFLED	\$500.00	SMNLFLXA	\$210.00	TLHSFLXD	\$320.00	WWSPFLSH	\$760.00
PMBHFLFE	\$210.00	SNANFLXA	\$1,000.00	TLHSFLXE	\$420.00	YBCTFLXA	\$170.00
PMBHFLMA	\$210.00	SNDSFLXA	\$720.00	TLHSFLXF	\$440.00	YNFNFLMA	\$540.00
PMBHFLNP	\$210.00	SNFRFLMA	\$250.00	TLHSFLXG	\$540.00	YNTWFLMA	\$550.00
PMBHFLTA	\$210.00	SNISFLXA	\$570.00	TLHSFLXH	\$490.00	YULEFLMA	\$350.00
PMPKFLMA	\$500.00	SNRSFLXA	\$810.00	TMTRFLXA	\$250.00	ZLSPFLXA	\$490.00
PNCRFLXA	\$300.00	SNSNFLXA	\$840.00	TRENFLMA	\$600.00	ZPHYFLXA	\$350.00
PNCYFLCA	\$330.00	SNSPFLXA	\$330.00	TRSPFLXA	\$250.00		
PNCYFLMA	\$190.00	SPBGFLXA	\$170.00	TTVLFLMA	\$300.00		
PNGRFLXA	\$540.00	SPBGFLXS	\$250.00	TVRSFLXA	\$900.00		
PNISFLXA	\$530.00	SPCPFLXA	\$660.00	UMTLFLXA	\$870.00		
PNLNFLXA	\$1,000.00	SPRGFLXA	\$330.00	UNVRFLXA	\$250.00		
PNLSFLXA	\$200.00	SRSTFLXA	\$190.00	VENCFLXA	\$400.00		
PNSCFLBL	\$230.00	SSDSFLXA	\$330.00	VENCFLXS	\$420.00		
PNSCFLFP	\$230.00	SSPRFLXA	\$650.00	VERNFLMA	\$490.00	]	

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
    - 1.3.3 Rates and Charges: (Cont'd)
      - 1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)
        - 1.3.3.1.2 <u>Verizon Business Services II Charges:</u> The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
ABDLFLXA	\$280.00	CDKYFLMA	\$920.00	DYBHFLMA	\$210.00	GNDYFLXA	\$240.00
ALCHFLXA	\$700.00	CFLDFLMA	\$720.00	DYBHFLOB	\$300.00	GNVLFLXA	\$770.00
ALFAFLXA	\$290.00	CFVLFLXA	\$700.00	DYBHFLOS	\$410.00	GNWDFLXA	\$700.00
ALFRFLXA	\$600.00	CHLKFLXA	\$900.00	DYBHFLPO	\$260.00	GRETFLXA	\$1,230.00
ALSPFLXA	\$270.00	CHPLFLJA	\$340.00	EGLLFLBG	\$290.00	GSVLFLMA	\$210.00
ALTRFLXA	\$290.00	CHSWFLXA	\$700.00	EGLLFLIH	\$400.00	GSVLFLNW	\$300.00
ALVAFLXA	\$550.00	CHTHFLXA	\$1,200.00	ENWDFLXA	\$500.00	GVLDFLXA	\$800.00
ANMRFLXA	\$350.00	CITRFLXA	\$400.00	EORNFLMA	\$600.00	HAVNFLMA	\$1,310.00
APPKFLXA	\$360.00	CLHNFLXA	\$380.00	ESTSFLXA	\$720.00	HBSDFLMA	\$500.00
ARCDFLXA	\$700.00	CLMTFLXA	\$820.00	EVRGFLXA	\$900.00	HDSNFLXA	\$310.00
ARCHFLMA	\$540.00	CLTNFLXA	\$940.00	FHSDFLXA	\$240.00	HGLDFLXA	\$350.00
ASTRFLXA	\$700.00	CLWRFLXA	\$270.00	FLBHFLMA	\$500.00	HGSPFLXA	\$700.00
AVPKFLXA	\$260.00	CNSDFLXA	\$250.00	FLRHFLXA	\$530.00	HLNVFLMA	\$750.00
BAKRFLXA	\$700.00	CNTMFLLE	\$550.00	FRBHFLFP	\$410.00	HLRDFLXA	\$470.00
BARTFLXA	\$380.00	COCOFLMA	\$260.00	FRPTFLXA	\$770.00	HLWDFLHA	\$390.00
BAYUFLXA	\$350.00	COCOFLME	\$330.00	FRSTFLXA	\$590.00	HLWDFLMA	\$370.00
BBPKFLXA	\$580.00	COCYFL13	\$300.00	FTGRFLMA	\$510.00	HLWDFLPE	\$280.00
BCGRFLXA	\$690.00	CPCRFLXA	\$410.00	FTLDFLCR	\$260.00	HLWDFLWH	\$390.00
BCRTFLBT	\$270.00	CPCRFLXB	\$440.00	FTLDFLCY	\$260.00	HMSPFLXA	\$740.00
BCRTFLMA	\$210.00	CPHZFLXA	\$700.00	FTLDFLJA	\$270.00	HMSTFLEA	\$410.00
BCRTFLSA	\$270.00	CRCYFLXA	\$680.00	FTLDFLMR	\$210.00	HMSTFLHM	\$300.00
BGPIFLMA	\$900.00	CRRVFLXA	\$700.00	FTLDFLOA	\$340.00	HMSTFLNA	\$490.00
BHPKFLXA	\$240.00	CRVWFLXA	\$640.00	FTLDFLPL	\$260.00	HNCYFLXA	\$440.00
BKVLFLJF	\$380.00	CRWDFLXA	\$290.00	FTLDFLSG	\$330.00	HNCYFLXN	\$510.00
BLDWFLMA	\$600.00	CSCYFLBA	\$900.00	FTLDFLSU	\$270.00	HOWYFLXA	\$790.00
BLGLFLMA	\$530.00	CSLBFLXA	\$370.00	FTLDFLWN	\$350.00	HSNGFLXA	\$480.00
BLTWFLXA	\$1,110.00	CTDLFLXA	\$580.00	FTMBFLXA	\$500.00	HTISFLMA	\$480.00
BLVWFLXA	\$500.00	CYGRFLXA	\$420.00	FTMDFLXA	\$580.00	HWTHFLMA	\$600.00
BNFYFLXA	\$560.00	CYLKFLXA	\$430.00	FTMYFLXA	\$300.00	HYPKFLXA	\$300.00
BNNLFLMA	\$500.00	CYLKFLXB	\$460.00	FTMYFLXB	\$500.00	IMKLFLXA	\$700.00
BNSPFLXA	\$480.00	DBRYFLDL	\$360.00	FTMYFLXC	\$350.00	INLKFLXA	\$660.00
BORAFLXA	\$700.00	DBRYFLMA	\$460.00	FTPRFLMA	\$260.00	INRKFLXX	\$340.00
BRBAFLXA	\$290.00	DDCYFLXA	\$900.00	FTWBFLXA	\$380.00	INTRFLXA	\$590.00

- 1. SERVICE DESCRIPTION (CONT'D.)
- 1.3 Intrastate T1 Digital Access (Cont'd.):
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.2 Verizon Business Services II Charges: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
BRFRFLXA	\$770.00	DELDFLMA	\$400.00	FTWBFLXB	\$530.00	INTWFLXA	\$690.00
BRJTFLXA	\$420.00	DESTFLXA	\$500.00	FTWBFLXC	\$580.00	INVRFLXA	\$700.00
BRKRFLXA	\$400.00	DFSPFLXA	\$830.00	FTWHFLXA	\$700.00	IONAFLXA	\$500.00
BRNDFLXA	\$290.00	DLBHFLKP	\$340.00	GCSPFLCN	\$610.00	ISLMFLMA	\$890.00
BRSNFLMA	\$600.00	DLBHFLMA	\$320.00	GCVLFLMA	\$600.00	JAY FLMA	\$940.00
BRTNFLXX	\$290.00	DLSPFLMA	\$540.00	GDRGFLXA	\$790.00	JCBHFLAB	\$400.00
BSHNFLXA	\$700.00	DNDNFLXA	\$330.00	GENVFLMA	\$590.00	JCBHFLMA	\$350.00
BVHLFLXA	\$640.00	DNLNFLWM	\$540.00	GLBRFLMC	\$450.00	JCBHFLSP	\$400.00
BWLGFLXA	\$570.00	DRBHFLMA	\$260.00	GLDLFLXA	\$840.00	JCVLFLAR	\$240.00
BYBHFLMA	\$360.00	DUNDFLXA	\$440.00	GLGCFLXA	\$700.00	JCVLFLBW	\$270.00
BYSHFLXA	\$280.00	DWPKFLXA	\$800.00	GLRDFLXA	\$350.00	JCVLFLCL	\$210.00
CCBHFLAF	\$350.00	DYBHFLFN	\$290.00	GNBOFLXA	\$550.00	JCVLFLFC	\$350.00
JCVLFLIA	\$350.00	LYHNFLOH	\$350.00	MRTHFLVE	\$1,180.00	PCBHFLNT	\$430.00
JCVLFLJT	\$270.00	MALNFLXA	\$740.00	MTDRFLXA	\$760.00	PKCYFLXA	\$400.00
JCVLFLLF	\$290.00	MAYOFLXA	\$870.00	MTLDFLXA	\$300.00	PLCSFLMA	\$380.00
JCVLFLNO	\$160.00	MCINFLXA	\$410.00	MTVRFLXA	\$580.00	PLMTFLXA	\$280.00
JCVLFLOW	\$330.00	MCLNFLXA	\$680.00	MXVLFLMA	\$660.00	PLSLFLXA	\$290.00
JCVLFLRV	\$280.00	MCNPFLMA	\$510.00	MYCYFLXA	\$410.00	PLTKFLMA	\$420.00
JCVLFLSJ	\$320.00	MDBGFLPM	\$620.00	NDADFLAC	\$380.00	PMBHFLCS	\$410.00
JCVLFLSM	\$240.00	MDSNFLXA	\$900.00	NDADFLBR	\$270.00	PMBHFLFE	\$270.00
JCVLFLWC	\$320.00	MIAMFLAE	\$210.00	NDADFLGG	\$320.00	PMBHFLMA	\$320.00
JNGSFLXA	\$890.00	MIAMFLAL	\$330.00	NDADFLOL	\$410.00	PMBHFLNP	\$320.00
JPTRFLMA	\$390.00	MIAMFLAP	\$160.00	NFMYFLXA	\$480.00	PMBHFLTA	\$390.00
JSPRFLXA	\$800.00	MIAMFLBA	\$320.00	NFMYFLXB	\$500.00	PMPKFLMA	\$700.00
KGLKFLXA	\$680.00	MIAMFLBC	\$310.00	NGBHFLXA	\$350.00	PNCRFLXA	\$350.00
KNVLFLXA	\$850.00	MIAMFLBR	\$280.00	NKLRFLMA	\$610.00	PNCYFLCA	\$400.00
KSSMFLXA	\$360.00	MIAMFLCA	\$160.00	NNPLFLXA	\$490.00	PNCYFLMA	\$260.00
KSSMFLXB	\$400.00	MIAMFLDB	\$350.00	NPLSFLXC	\$700.00	PNGRFLXA	\$700.00
KSSMFLXD	\$470.00	MIAMFLFL	\$280.00	NPLSFLXD	\$500.00	PNISFLXA	\$540.00
KYHGFLMA	\$700.00	MIAMFLGR	\$210.00	NPRCFLXA	\$350.00	PNLNFLXA	\$920.00
KYLRFLLS	\$700.00	MIAMFLHL	\$180.00	NRPTFLXA	\$490.00	PNLSFLXA	\$300.00
KYLRFLMA	\$770.00	MIAMFLIC	\$340.00	NRSDFLXA	\$300.00	PNSCFLBL	\$260.00
KYSTFLXA	\$290.00	MIAMFLKE	\$360.00	NSBHFLMA	\$380.00	PNSCFLFP	\$260.00
KYWSFLMA	\$750.00	MIAMFLME	\$160.00	NWBYFLMA	\$580.00	PNSCFLHC	\$440.00
LBLLFLXA	\$700.00	MIAMFLNM	\$320.00	OCALFLXA	\$340.00	PNSCFLPB	\$580.00
LDLKFLXA	\$800.00	MIAMFLNS	\$320.00	OCALFLXB	\$380.00	PNSCFLWA	\$320.00
LEE FLXA	\$900.00	MIAMFLOL	\$380.00	OCALFLXC	\$500.00	PNVDFLMA	\$350.00
LGBKFLXA	\$350.00	MIAMFLPB	\$260.00	OCALFLXJ	\$500.00	POINFLXA	\$590.00
LHACFLXA	\$530.00	MIAMFLPL	\$160.00	OCNFFLXA	\$530.00	PRRNFLMA	\$340.00
LKALFLXA	\$370.00	MIAMFLRR	\$260.00	OKCBFLXA	\$900.00	PRRYFLXA	\$1,100.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.2 <u>Verizon Business Services II Charges:</u> (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
LKBNFLXB	\$400.00	MIAMFLSH	\$350.00	OKHLFLMA	\$620.00	PRSHFLXA	\$350.00
LKBRFLXA	\$340.00	MIAMFLSO	\$270.00	OKLWFLXA	\$540.00	PRSNFLFD	\$730.00
LKBTFLXA	\$600.00	MIAMFLWD	\$190.00	OLDSFLXA	\$290.00	PSDNFLXA	\$320.00
LKCYFLMA	\$650.00	MIAMFLWM	\$210.00	OLTWFLLN	\$800.00	PTCTFLXA	\$400.00
LKHLFLXA	\$560.00	MICCFLBB	\$500.00	ORCYFLXA	\$550.00	PTCYFLXA	\$290.00
LKLDFLXA	\$330.00	MLBRFLMA	\$210.00	ORCYFLXC	\$550.00	PTSJFLXA	\$990.00
LKLDFLXE	\$280.00	MLBYFLXA	\$410.00	ORLDFLAP	\$290.00	PTSLFLMA	\$350.00
LKLDFLXN	\$310.00	MLRSFLXA	\$470.00	ORLDFLCL	\$210.00	PTSLFLSO	\$350.00
LKMRFLMA	\$300.00	MLTNFLRA	\$620.00	ORLDFLMA	\$210.00	QNCYFLXA	\$770.00
LKPCFLXA	\$610.00	MNDRFLAV	\$290.00	ORLDFLPC	\$210.00	RAFRFLXA	\$560.00
LKWLFLXA	\$450.00	MNDRFLLO	\$160.00	ORLDFLPH	\$350.00	RSKNFLXA	\$350.00
LKWLFLXE	\$620.00	MNDRFLLW	\$400.00	ORLDFLSA	\$280.00	RYHLFLXA	\$780.00
LLMNFLXA	\$290.00	MNLKFLXA	\$490.00	ORPKFLMA	\$480.00	SARKFLXA	\$350.00
LNLKFLXA	\$350.00	MNSNFLMA	\$400.00	ORPKFLRW	\$510.00	SBNGFLXA	\$500.00
LRGOFLXA	\$290.00	MNTIFLXA	\$700.00	ORSPFLXA	\$490.00	SBSTFLFE	\$500.00
LRVLFLXA	\$730.00	MOISFLXA	\$870.00	OSPRFLXA	\$370.00	SBSTFLMA	\$400.00
LSBGFLXA	\$500.00	MOLNFLXA	\$400.00	OVIDFLCA	\$350.00	SCPKFLXA	\$500.00
LUTZFLXA	\$290.00	MRDCFLXA	\$700.00	PACEFLPV	\$530.00	SEKYFLXA	\$350.00
LVOKFLXA	\$520.00	MRHNFLXA	\$900.00	PAHKFLMA	\$700.00	SGBEFLXA	\$350.00
LWTYFLXA	\$700.00	MRNNFLXA	\$700.00	PANCFLXA	\$700.00	SGBHFLXA	\$780.00

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
SGKYFLMA	\$900.00	TLHSFLXD	\$340.00	WSSDFLXA	\$230.00
SHLMFLXA	\$530.00	TLHSFLXE	\$420.00	WSTVFLXA	\$660.00
SKWYFLXA	\$310.00	TLHSFLXF	\$500.00	WWSPFLHI	\$690.00
SLHLFLXA	\$530.00	TLHSFLXG	\$550.00	WWSPFLSH	\$550.00
SLSPFLXA	\$240.00	TLHSFLXH	\$300.00	YBCTFLXA	\$230.00
SMNLFLXA	\$300.00	TMTRFLXA	\$350.00	YNFNFLMA	\$590.00
SNANFLXA	\$900.00	TRENFLMA	\$670.00	YNTWFLMA	\$700.00
SNDSFLXA	\$850.00	TRSPFLXA	\$290.00	YULEFLMA	\$500.00
SNFRFLMA	\$350.00	TTVLFLMA	\$470.00	ZLSPFLXA	\$500.00
SNISFLXA	\$580.00	TVRSFLXA	\$700.00	ZPHYFLXA	\$280.00
SNRSFLXA	\$730.00	UMTLFLXA	\$700.00		
SNSNFLXA	\$820.00	UNVRFLXA	\$340.00		
SNSPFLXA	\$350.00	VENCFLXA	\$430.00		
SPBGFLXA	\$260.00	VENCFLXS	\$500.00		
SPBGFLXS	\$320.00	VERNFLMA	\$630.00		
SPCPFLXA	\$700.00	VLPRFLXA	\$600.00		
SPRGFLXA	\$350.00	VLPRFLXB	\$620.00		

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 Intrastate T1 Digital Access (Cont'd.):
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.2 <u>Verizon Business Services II Charges:</u> (Cont'd)

Wire Center	Rate	Wire Center	Rate
SRSTFLXA	\$170.00	VRBHFLBE	\$350.00
SSDSFLXA	\$350.00	VRBHFLMA	\$250.00
SSPRFLXA	\$700.00	WALDFLXA	\$380.00
STAGFLBS	\$490.00	WCHLFLXA	\$570.00
STAGFLMA	\$260.00	WELKFLMA	\$700.00
STAGFLSH	\$470.00	WHSPFLXA	\$640.00
STAGFLWG	\$570.00	WIMMFLXA	\$310.00
STCDFLXA	\$570.00	WLBRFLXA	\$620.00
STGRFLXA	\$290.00	WLCHFLXA	\$310.00
STMKFLXA	\$700.00	WLCRFLXA	\$300.00
STRKFLXA	\$700.00	WLHLFLXA	\$600.00
STRTFLMA	\$350.00	WLSTFLXA	\$610.00
SVSPFLXA	\$500.00	WLWDFLXA	\$800.00
SVSSFLXA	\$500.00	WNDRFLXA	\$480.00
SWTHFLXA	\$260.00	WNGRFLXA	\$470.00
SYHSFLCC	\$1,220.00	WNHNFLXC	\$350.00
TAFBFLXA	\$660.00	WNPKFLXA	\$260.00
TAMPFLXA	\$250.00	WNPKFLXE	\$300.00
TAMPFLXE	\$260.00	WPBHFLAN	\$250.00
TAMPFLXX	\$260.00	WPBHFLGA	\$320.00
THNTFLXA	\$350.00	WPBHFLGR	\$340.00
TLCHFLXA	\$700.00	WPBHFLHH	\$340.00
TLHSFLXA	\$300.00	WPBHFLLE	\$350.00
TLHSFLXB	\$340.00	WPBHFLRB	\$180.00
TLHSFLXC	\$460.00	WPBHFLRP	\$350.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.3 <u>Verizon Business Services I Charges:</u> The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate						
ADA MNXA	\$800.00	BGLWMNXB	\$400.00	BTLKMNBA	\$400.00	CNTNMNXC	\$600.00
ADBNMNXA	\$600.00	BGLYMNXB	\$600.00	BUHLMNBU	\$300.00	COKTMNXC	\$400.00
ADMSMNXB	\$400.00	BKTNMNXB	\$400.00	BVCKMNXA	\$800.00	COOKMNCO	\$250.00
ADRNMNXA	\$400.00	BLANMNBL	\$200.00	BWBKMNBI	\$800.00	CRGRMNXC	\$800.00
AFTNMNAF	\$300.00	BLATMNXB	\$400.00	BWDLMNXB	\$600.00	CRMRMNXC	\$600.00
AGINMNXA	\$800.00	BLERMNXB	\$600.00	BWVLMNXB	\$600.00	CRRLMNXA	\$800.00
ALBOMNXB	\$400.00	BLGRMNXB	\$400.00	BWVYMNXB	\$800.00	CRSBMNXC	\$300.00
ALBYMNXA	\$400.00	BLHMMNXB	\$800.00	BYLDMNXB	\$300.00	CRTOMNCB	\$300.00
ALDNMNXB	\$600.00	BLKDMNXB	\$800.00	BYRNMNXB	\$300.00	CRWLMNXC	\$400.00
ALLEMNAL	\$400.00	BLLPMNXA	\$400.00	BYRVMNXB	\$600.00	CRYSMNCR	\$200.00
ALMLMNXA	\$400.00	BLPRMNXB	\$400.00	CARLMNXC	\$600.00	CSLKMNXA	\$400.00
ALTRMNXA	\$400.00	BLTNMNCE	\$200.00	CHGVMNXC	\$400.00	CSMSMNXC	\$600.00
ALVDMNXA	\$400.00	BLTNMNNO	\$200.00	CHNDMNXC	\$400.00	CSSLMNCL	\$600.00
ALXNMNXA	\$600.00	BLTNMNSO	\$200.00	CHOKMNXC	\$800.00	CTFDMNCH	\$300.00
ALXNMNXL	\$600.00	BLTRMNXB	\$600.00	CHSHMNCS	\$300.00	CTGVMNCG	\$200.00
AMBYMNXA	\$600.00	BLVWMNXB	\$600.00	CHSKMNXC	\$300.00	CTTNMNXC	\$400.00
ANDLMNXB	\$400.00	BMDJMNAS	\$300.00	CHSTMNCH	\$300.00	CTWDMNXC	\$600.00
ANOKMNAN	\$200.00	BMDJMNBE	\$400.00	CKTNMNCR	\$300.00	CURRMNXC	\$400.00
APPLMNAP	\$600.00	BNSNMNXB	\$600.00	CLBKMNXC	\$600.00	CYLNMNXC	\$400.00
APVYMNXA	\$300.00	BNVLMNXB	\$300.00	CLCYMNXB	\$600.00	CYRSMNXC	\$800.00
ARCOMNXA	\$600.00	BOCKMNXB	\$600.00	CLDNMNCA	\$350.00	DAKTMNXD	\$600.00
ARGYMNXB	\$400.00	BORPMNXB	\$400.00	CLEMMNXC	\$400.00	DANBMNXD	\$600.00
ARTNMNXA	\$400.00	BOVLMNXB	\$400.00	CLFDMNXC	\$800.00	DDCTMNXD	\$400.00
ASHBMNXA	\$600.00	BOYDMNXB	\$800.00	CLGNMNXC	\$300.00	DELNMNXD	\$400.00
ASKVMNXA	\$800.00	BRCLMNXB	\$800.00	CLGVMNXC	\$600.00	DENTMNXD	\$400.00
ASRVMNXA	\$600.00	BRCTMNBC	\$200.00	CLLKMNXA	\$300.00	DLFTMNXD	\$300.00
ATKNMNXA	\$400.00	BRDSMNXA	\$800.00	CLMTMNXA	\$400.00	DLTHMNAF	\$300.00
ATWRMNXA	\$400.00	BRHMMNBR	\$400.00	CLMXMNXC	\$400.00	DLTHMNCB	\$300.00
AURRMNXA	\$400.00	BRKSMNXB	\$600.00	CLQTMNCA	\$300.00	DLTHMNDB	\$300.00
AUSTMNAB	\$350.00	BRNDMNXB	\$800.00	CLRNMNCO	\$300.00	DLTHMNLA	\$300.00
AVOCMNXA	\$400.00	BRNMMNBA	\$400.00	CLRSMNXC	\$400.00	DLTHMNME	\$300.00
AVONMNVO	\$300.00	BRNRMNBR	\$300.00	CLSPMNCB	\$300.00	DLTHMNPL	\$300.00
BBCYMNXB	\$800.00	BROTMNXB	\$600.00	CLTNMNXA	\$800.00	DLTNMNXD	\$600.00
BBTTMNXB	\$600.00	BRPTMNXB	\$400.00	CLWRMNXC	\$300.00	DLVNMNXD	\$600.00
BCKRMNXA	\$400.00	BRRTMNXB	\$800.00	CLWYMNXC	\$600.00	DNHMMNXD	\$600.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
    - 1.3.3 Rates and Charges: (Cont'd)
      - 1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)
        - 1.3.3.1.3 <u>Verizon Business Services I Charges:</u> The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate						
BCKSMNXB	\$400.00	BRRVMNXB	\$400.00	CMBAMNXC	\$600.00	DNLYMNXD	\$800.00
BDGRMNXB	\$800.00	BRSNMNXB	\$400.00	CMBRMNCA	\$400.00	DNNLMNXD	\$400.00
BDTTMNXA	\$800.00	BRTHMNXB	\$400.00	CMFRMNXC	\$400.00	DNSNMNXD	\$600.00
BECDMNXB	\$600.00	BRTNMNXB	\$400.00	CMPBMNXA	\$600.00	DNVSMNXD	\$800.00
BENAMNXB	\$600.00	BRVIMNXB	\$400.00	CMSTMNCO	\$300.00	DRCKMNXD	\$400.00
BFLKMNXB	\$600.00	BRVIMNXR	\$400.00	CNBYMNXC	\$600.00	DRRVMNXD	\$400.00
BFLOMNBU	\$300.00	BRVLMNBU	\$200.00	CNFLMNXC	\$300.00	DRWDMNXD	\$300.00
BGFKMNXB	\$400.00	BRVLMNXA	\$400.00	CNGRMNXC	\$600.00	DSSLMNXD	\$400.00
BGFSMNXB	\$600.00	BRWSMNXA	\$600.00	CNLKMNXB	\$400.00	DTLKMNDL	\$400.00
BGLKMNXA	\$400.00	BTFDMNXB	\$400.00	CNRPMNND	\$200.00	DUNDMNXA	\$400.00

- 1. <u>SERVICE DESCRIPTION (CON</u>T'D.)
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.3 <u>Verizon Business Services I Charges:</u> (Cont'd)

Wire Center	Rate						
DWSNMNXD	\$800.00	FRBRMNXF	\$600.00	GTWYMNXG	\$600.00	HZRNMNXA	\$800.00
DXTRMNXA	\$400.00	FRDHMNXF	\$600.00	GTZKMNXG	\$800.00	IDCRMNXI	\$400.00
EAGNMN05	\$300.00	FRDLMNFR	\$200.00	GVCYMNXG	\$400.00	INFLMNXI	\$800.00
EAGNMNLB	\$200.00	FRFLMNFB	\$400.00	GYLRMNGA	\$400.00	INGRMNXI	\$800.00
ECHNMNXE	\$400.00	FRFXMNXA	\$600.00	HALMMNXH	\$600.00	IONAMNXI	\$400.00
ECHOMNXE	\$600.00	FRLKMNFL	\$300.00	HAMLMNHB	\$300.00	IRNGMNXI	\$400.00
EDPRMNEP	\$200.00	FRPTMNXF	\$400.00	HBNGMNHI	\$300.00	ISBLMNXI	\$600.00
EDPRMNGP	\$200.00	FRSTMNXF	\$600.00	HCKNMNXE	\$600.00	ISLEMNXI	\$600.00
EDVYMNXE	\$400.00	FRTLMNXF	\$600.00	HCKNMNXH	\$600.00	ISLKMNIL	\$300.00
EGBNMNXE	\$400.00	FRTNMNXF	\$400.00	HCTRMNXA	\$600.00	ISNTMNIS	\$400.00
EGLKMNXE	\$600.00	FSHRMNXF	\$400.00	HFMNMNXH	\$800.00	ITSPMNXA	\$600.00
EGTNMNXE	\$600.00	FSTNMNXF	\$600.00	HILSMNXH	\$600.00	IVNHMNXI	\$600.00
EKRVMNBB	\$300.00	FULDMNXA	\$400.00	HLCKMNXH	\$600.00	JCBSMNXJ	\$400.00
EKRVMNER	\$200.00	GARYMNXG	\$600.00	HLCYMNXA	\$400.00	JCSNMNJA	\$300.00
ELDLMNXA	\$400.00	GBBNMNXA	\$800.00	HLFRMNCO	\$300.00	JFRSMNXA	\$600.00
ELGNMNXE	\$300.00	GCKNMNXG	\$600.00	HLLDMNXH	\$600.00	JNVLMNXJ	\$600.00
ELLKMNXE	\$600.00	GCVLMNXA	\$800.00	HLSTMNXH	\$400.00	JRDNMNXJ	\$300.00
ELMRMNXE	\$600.00	GDMDMNXG	\$400.00	HLWYMNXA	\$800.00	JSPRMNXJ	\$600.00
ELRSMNXE	\$400.00	GDMRMNGM	\$800.00	HMBLMNXA	\$600.00	KBTGMNXN	\$600.00
ELWOMNXE	\$600.00	GDRGMNXG	\$600.00	HMCYMNXH	\$600.00	KDYHMNXA	\$400.00
ELY MNXE	\$800.00	GDRPMNGR	\$300.00	HNCCMNXH	\$800.00	KETNMNXK	\$800.00
ELYSMNXE	\$600.00	GDTHMNXG	\$600.00	HNCKMNHI	\$400.00	KEWTMNKE	\$300.00
EMBRMNXE	\$400.00	GHNTMNXG	\$600.00	HNDRMNXH	\$800.00	KLHRMNXK	\$800.00
EMLYMNXA	\$400.00	GLCOMNXG	\$400.00	HNFLMNXH	\$600.00	KLKNMNXK	\$600.00
EMMNMNXE	\$600.00	GLLYMNXG	\$600.00	HNNGMNHE	\$300.00	KLLGMNXA	\$400.00
ENFDMNXE	\$400.00	GLMNMNXG	\$400.00	HNSKMNXH	\$600.00	KMBLMNXK	\$400.00
ERBGMNXE	\$600.00	GLNDMNXA	\$600.00	HNSNMNXH	\$800.00	KMBRMNXK	\$400.00
ERHRMNXE	\$400.00	GLVLMNGL	\$400.00	HNTLMNXH	\$600.00	KNDYMNXA	\$600.00
ERSKMNXA	\$600.00	GLVYMN06	\$300.00	HNVRMNHB	\$300.00	KNYNMNXA	\$400.00
ESTNMNXE	\$600.00	GLVYMNOR	\$200.00	HODLMNXH	\$400.00	KRKHMNXK	\$600.00
ETZNMNXE	\$600.00	GLWDMNGL	\$600.00	HOKHMNXH	\$600.00	KRLSMNXK	\$600.00
EVLTMNEV	\$300.00	GLYNMNXG	\$400.00	HOLTMNXH	\$600.00	KSSNMNXK	\$400.00
EVVLMNXE	\$800.00	GNBSMNXG	\$800.00	HPKNMNHO	\$200.00	KSTRMNXL	\$600.00
EXCLMNEX	\$200.00	GNTRMNXA	\$800.00	HRLDMNXH	\$600.00	KTRVMNXK	\$600.00
EYOTMNXE	\$300.00	GNVCMNXG	\$600.00	HRLKMNXA	\$300.00	LACRMNXL	\$600.00
FAMTMNXF	\$400.00	GNWLMNXG	\$400.00	HRMNMNXA	\$800.00	LAPTMNXL	\$600.00
FDDMMNXF	\$600.00	GODHMNXG	\$600.00	HRMYMNXH	\$400.00	LECTMNXL	\$400.00
FETNMNXF	\$400.00	GRADMNXG	\$400.00	HRWKMNXH	\$600.00	LEDRMNXL	\$400.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 Intrastate T1 Digital Access (Cont'd.):
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.3 <u>Verizon Business Services I Charges:</u> (Cont'd)

Wire Center	Rate						
FKLNMNXF	\$600.00	GRCYMNXG	\$600.00	HSNGMNXH	\$300.00	LEOTMNXL	\$400.00
FLOMMNXF	\$600.00	GRFDMNXG	\$800.00	HSTNMNXH	\$600.00	LERYMNXB	\$400.00
FLWDMNXF	\$400.00	GRFLMNXG	\$600.00	HTRDMNXH	\$400.00	LESRMNLS	\$400.00
FNLDMNFO	\$600.00	GRISMNXG	\$400.00	HTSNMNAA	\$600.00	LFYTMNXA	\$600.00
FNSNMNXF	\$800.00	GRNGMNXG	\$600.00	HTSNMNXH	\$600.00	LGVLMNXL	\$600.00
FOLYMNFO	\$300.00	GRNYMNXG	\$400.00	HVLDMNXA	\$800.00	LKBNMNXL	\$600.00
FONTMNXF	\$400.00	GRSNMNXG	\$600.00	HWLKMNXH	\$400.00	LKBRMNXL	\$600.00
FOTNMNXF	\$400.00	GRVNMNXG	\$400.00	HWLYMNHA	\$300.00	LKCRMNXL	\$600.00
FRAZMNXF	\$400.00	GRYEMNXG	\$600.00	HYFDMNXA	\$400.00	LKCYMNXL	\$600.00
FRBLMNFA	\$400.00	GRYGMNXG	\$800.00	HYLKMNXH	\$400.00	LKFDMNXL	\$300.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.3 Intrastate T1 Digital Access (Cont'd.):
    - 1.3.3 Rates and Charges: (Cont'd)
      - 1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)
        - 1.3.3.1.4 Pre-VBS Charges: The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate						
LKLLMNXL	\$600.00	MNETMNXA	\$600.00	MZPPMNXM	\$400.00	PEASMNXP	\$400.00
LKPKMNXL	\$600.00	MNHGMNXM	\$400.00	NBRNMNNB	\$400.00	PENCMNXP	\$600.00
LKVLMNXL	\$400.00	MNKTMNXM	\$600.00	NCLTMNNC	\$400.00	PIRZMNXA	\$400.00
LKWLMNXL	\$400.00	MNLKMNXM	\$600.00	NCSNMNXN	\$400.00	PKRPMNPR	\$400.00
LMTNMNXA	\$400.00	MNRVMNXM	\$600.00	NEVSMNXN	\$600.00	PLATMNXP	\$400.00
LNBOMNXL	\$400.00	MNTIMNXM	\$400.00	NKLKMNXN	\$300.00	PLGRMNXP	\$400.00
LNCLMNXL	\$400.00	MNTRMNXM	\$600.00	NLVLMNXN	\$400.00	PLMOMNBN	\$200.00
LNCSMNXL	\$600.00	MNVLMNXM	\$400.00	NMRDMNXN	\$400.00	PLMOMNFE	\$200.00
LNDLMNXB	\$600.00	MOLKMNML	\$400.00	NOMEMNXN	\$800.00	PLMRMNXP	\$600.00
LNGBMNXL	\$600.00	MONDMNXA	\$300.00	NRCRMNXN	\$800.00	PLRPMNXP	\$600.00
LNPRMNXL	\$400.00	MORAMNMO	\$600.00	NRFDMNNO	\$400.00	PLSDMNXP	\$400.00
LNRDMNXL	\$800.00	MPBAMNXM	\$600.00	NROPMNXN	\$400.00	PLVWMNXP	\$300.00
LNSTMNXL	\$400.00	MPGVMNXA	\$300.00	NRWDMNXN	\$300.00	PMTNMNXP	\$600.00
LSMRMNXL	\$600.00	MPLKMNXA	\$400.00	NSHWMNNA	\$300.00	PNCYMNPC	\$250.00
LSPRMNXL	\$400.00	MPLSMN07	\$200.00	NSPLMNPR	\$200.00	PNISMNXP	\$400.00
LSTNMNXA	\$400.00	MPLSMNAS	\$200.00	NSSWMNNI	\$300.00	PNMHMNXP	\$800.00
LTFDMNLI	\$400.00	MPLSMNBB	\$200.00	NVRRMNNA	\$300.00	PNRVMNXP	\$400.00
LTFKMNXL	\$600.00	MPLSMNBE	\$200.00	NWBTMNCL	\$200.00	PPSKMNXP	\$800.00
LTFLMNLF	\$350.00	MPLSMNCD	\$400.00	NWFLMNXN	\$600.00	PPSTMNPI	\$600.00
LUCNMNXL	\$600.00	MPLSMNDT	\$200.00	NWGRMNXN	\$400.00	PQLKMNXP	\$400.00
LVRNMNLU	\$600.00	MPLSMNFD	\$300.00	NWLDMNXN	\$400.00	PRBGMNXP	\$600.00
LWRYMN01	\$800.00	MPLSMNFR	\$200.00	NWLNMNXN	\$400.00	PRHMMNXP	\$400.00
LWVLMNXL	\$400.00	MPLSMNFS	\$200.00	NWMNMNXN	\$400.00	PRLKMNXP	\$300.00
LYLEMNXM	\$400.00	MPLSMNGE	\$200.00	NWMRMNXN	\$400.00	PRLYMNXP	\$400.00
LYNDMNXL	\$600.00	MPLSMNMS	\$300.00	NWPRMNXN	\$400.00	PRPRMNXP	\$400.00
MABLMNXM	\$600.00	MPLSMNPE	\$250.00	NWULMNXN	\$600.00	PRTNMN01	\$300.00
MADLMNXM	\$400.00	MPLSMNPI	\$250.00	NYMLMNXN	\$400.00	PRTNMNPR	\$400.00
MAINMNXM	\$400.00	MPLSMNTF	\$200.00	ODINMNXO	\$400.00	PSTNMNXA	\$600.00
MALMMNXM	\$600.00	MPPLMNXM	\$300.00	ODSSMNXO	\$800.00	PTERMNXP	\$600.00
MAYRMNXM	\$400.00	MPTNMNXM	\$600.00	OGLVMNOA	\$400.00	PTSNMNXP	\$400.00
MCGRMNXM	\$600.00	MPWDMNMA	\$200.00	OKBNMNXO	\$300.00	PYVLMNXP	\$600.00
MCINMNXM	\$600.00	MRBLMNMA	\$300.00	OKGVMNOG	\$300.00	RACNMNXR	\$400.00
MDLDMNXA	\$400.00	MRCLMNXM	\$400.00	OKLEMNXO	\$600.00	RAMYMNXR	\$400.00
MDRVMNXM	\$800.00	MRDCMNXM	\$600.00	OLIVMNOL	\$600.00	RANRMNXR	\$800.00
MDSLMNXM	\$800.00	MRGNMNXM	\$600.00	ONAMMNXO	\$400.00	RCFDMN66	\$250.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
    - 1.3.3 Rates and Charges: (Cont'd)
      - 1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)
        - 1.3.3.1.4 Pre-VBS Charges: The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate						
MDSNMNXM	\$800.00	MRRSMNMO	\$600.00	ORNCMNXO	\$400.00	RCFRMNRO	\$300.00
MGRTMNXM	\$800.00	MRSHMNMA	\$400.00	ORR MNXA	\$800.00	RCMDMNXR	\$400.00
MHNMMNMA	\$400.00	MRTTMNXM	\$800.00	ORVLMNOR	\$800.00	RDFLMNQA	\$300.00
MILCMNXM	\$400.00	MRTWMNXM	\$600.00	OSAGMNXO	\$600.00	RDFLMNRA	\$400.00
MILNMNXM	\$800.00	MSSNMNXM	\$400.00	OSKSMNXO	\$600.00	RDFLMNXF	\$600.00
MIVLMNXM	\$800.00	MTGMMNXM	\$400.00	OSLOMNXO	\$400.00	RDFLMNXR	\$600.00
MLNGMNXM	\$800.00	MTIRMNMI	\$300.00	OSSEMNXO	\$250.00	RDLKMNXR	\$800.00
MLRSMNXM	\$400.00	MTLKMNXM	\$300.00	OSTRMNXO	\$400.00	RDWNMNRW	\$200.00
MLRYMNXM	\$600.00	MTLYMNXM	\$400.00	OTNGMNXO	\$400.00	REMRMNXR	\$600.00
MLTNMNXM	\$400.00	MTRSMNXB	\$400.00	OTRTMNXO	\$400.00	RGRSMNXR	\$300.00
MLVLMNXM	\$300.00	MTVDMNMO	\$600.00	OWTNMNOW	\$300.00	RICEMNXR	\$400.00
MNCHMNXM	\$600.00	MYNRMNXM	\$800.00	PALOMNXP	\$400.00	RKDLMNXR	\$400.00

Wire Center	Rate						
RLFLMNXR	\$400.00	SPFDMNXS	\$400.00	TWLKMNXT	\$600.00	WLLSMNXW	\$600.00
RLLGMNXR	\$400.00	SPGVMNXS	\$600.00	TWVYMNXT	\$600.00	WLMRMNWI	\$350.00
RLNGMNXR	\$400.00	SPLSMNST	\$300.00	TYFLMNXT	\$400.00	WLMSMNXW	\$800.00
RNDLMNXR	\$600.00	SPVYMNXA	\$400.00	TYLRMNXT	\$600.00	WLMTMNXA	\$400.00
RNLKMNXA	\$400.00	SQLKMNXS	\$800.00	ULENMNXU	\$400.00	WNDLMNXW	\$600.00
RNVLMNXA	\$600.00	STBNMNXS	\$300.00	UNWDMNXU	\$400.00	WNDMMNWI	\$300.00
ROCHMNRO	\$300.00	STCAMNXS	\$600.00	UPSLMNXU	\$400.00	WNGRMNXW	\$400.00
ROSEMNXA	\$800.00	STCDMNTO	\$300.00	URBNMNXU	\$400.00	WNMNMNXW	\$800.00
RSCYMNRC	\$400.00	STCHMNSC	\$300.00	VCTAMNXV	\$300.00	WNNBMNXW	\$400.00
RSFRMNXR	\$400.00	STCLMNXS	\$600.00	VESTMNXV	\$600.00	WNSKMNXW	\$800.00
RSHMMNXA	\$800.00	STENMNXS	\$600.00	VKNGMNXV	\$600.00	WNSTMNXW	\$400.00
RSMTMNXR	\$300.00	STHRMNXS	\$400.00	VLRDMNXV	\$400.00	WNTHMNXW	\$600.00
RSSLMNXA	\$600.00	STJMMNXS	\$400.00	VNNGMNXV	\$400.00	WOTNMNXW	\$600.00
RSVLMNFY	\$300.00	STJSMNSJ	\$300.00	VRCTMNXV	\$600.00	WRGHMNXW	\$800.00
RSVTMNXR	\$800.00	STLEMNXS	\$600.00	VRDLMNXV	\$300.00	WRRDMNAC	\$300.00
RTHSMNXR	\$400.00	STMCMNXS	\$300.00	VRGNMNVI	\$300.00	WRRDMNXA	\$400.00
RTTNMNXR	\$600.00	STMNMNXS	\$400.00	VRGSMNXV	\$600.00	WRRNMNXW	\$400.00
RYMNMNXA	\$600.00	STPHMNXS	\$600.00	WABNMNXA	\$600.00	WSBKMNXA	\$400.00
RYTNMNRN	\$300.00	STPLMNBE	\$200.00	WABNMNXW	\$600.00	WSPLMNWS	\$200.00
SABNMNSA	\$300.00	STPLMNEM	\$250.00	WACNMNXW	\$300.00	WTKNMNXW	\$800.00
SCHRMNXS	\$800.00	STPLMNHB	\$250.00	WADNMNWA	\$300.00	WTTWMNXW	\$300.00
SCNDMNXS	\$400.00	STPLMNMI	\$200.00	WAFTMNXW	\$300.00	WTVLMNXW	\$600.00
SDVLMNSO	\$300.00	STPLMNMK	\$200.00	WARBMNXA	\$400.00	WVTNMNXW	\$400.00
SEBKMN01	\$800.00	STPLMNSL	\$300.00	WASCMNWA	\$600.00	WYKFMNXW	\$800.00
SEDNMNXS	\$600.00	STPRMNSP	\$350.00	WAWNMNXA	\$400.00	WYNGMNXA	\$600.00
SFTHMNXS	\$400.00	STRBMNXS	\$600.00	WBLKMNWB	\$200.00	WYZTMNWA	\$200.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.3 Intrastate T1 Digital Access (Cont'd.):
    - 1.3.3 Rates and Charges: (Cont'd)
      - 1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)
        - 1.3.3.1.4 Pre-VBS Charges: The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate						
SGLKMNXS	\$400.00	STRNMNXS	\$600.00	WBSHMNWA	\$600.00	ZMBRMNXZ	\$400.00
SHBNMNXS	\$400.00	STVLMNST	\$300.00	WBSSMNXW	\$400.00	ZMFLMNXZ	\$300.00
SHKPMNSH	\$300.00	STWRMNST	\$200.00	WBSTMNXW	\$400.00	ZMMNMNXA	\$400.00
SHLYMNXS	\$400.00	STWTMNXS	\$400.00	WCNCMNXW	\$400.00		
SHVNMNXS	\$600.00	SVEAMNXS	\$600.00	WDLKMNXW	\$600.00		
SHVWMNRI	\$200.00	SVLKMNXS	\$600.00	WDRFMNXW	\$600.00		
SKCTMNSC	\$400.00	SWVLMNSV	\$400.00	WDSTMNXW	\$600.00		
SLBAMNSA	\$400.00	TNTHMNXA	\$800.00	WHERMNXW	\$600.00		
SLEYMNXS	\$600.00	TOFTMNTB	\$600.00	WHKNMNXW	\$600.00		
SLLKMNXS	\$400.00	TOWRMNXA	\$400.00	WHPHMNXW	\$600.00		
SLTNMNXS	\$400.00	TRACMNTR	\$400.00	WHRKMNXW	\$400.00		
SLWYMNXS	\$600.00	TRFLMNTH	\$400.00	WHTNMNXW	\$800.00		
SNBGMNXS	\$600.00	TRMNMNXT	\$400.00	WINOMNWI	\$350.00		
SNBRMNXS	\$400.00	TRMTMNXT	\$400.00	WLCMMNXW	\$400.00		
SNDSMNSA	\$600.00	TRRCMNXT	\$600.00	WLGVMNXW	\$400.00		
SOHNMNXS	\$600.00	TRRVMNXT	\$600.00	WLKRMNXW	\$600.00		
SPCRMNXS	\$400.00	TWHRMNXA	\$400.00	WLLKMNXW	\$400.00		

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.3 Rates and Charges: (Cont'd)
    - 1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.4 Pre-VBS Charges: The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
ABDLFLXA	\$281.80	BYBHFLMA	\$632.10	DWPKFLXA	\$2,289.05	GDRGFLXA	\$788.55
ALCHFLXA	\$1,633.05	BYSHFLXA	\$535.55	DYBHFLFN	\$438.55	GENVFLMA	\$699.15
ALFAFLXA	\$542.70	CCBHFLAF	\$535.55	DYBHFLMA	\$281.80	GLBRFLMC	\$587.40
ALFRFLXA	\$895.05	CCBHFLMA	\$535.55	DYBHFLOB	\$462.80	GLDLFLXA	\$1,223.05
ALSPFLXA	\$511.30	CDKYFLMA	\$1,469.05	DYBHFLOS	\$542.70	GLGCFLXA	\$1,018.05
ALTHFLXA	\$587.40	CFLDFLMA	\$1,059.05	DYBHFLPO	\$487.05	GLRDFLXA	\$535.55
ALTRFLXA	\$654.45	CFVLFLXA	\$788.55	DYBHFLWS	\$281.80	GNBOFLXA	\$788.55
ALVAFLXA	\$721.50	CHLKFLXA	\$1,407.55	EGLLFLBG	\$438.55	GNDYFLXA	\$487.05
ANMRFLXA	\$699.15	CHPLFLJA	\$1,100.05	EGLLFLIH	\$462.80	GNVLFLXA	\$1,161.55
APLCFLXA	\$1,366.55	CHSWFLXA	\$1,182.05	ENWDFLXA	\$936.05	GNWDFLXA	\$977.05
APPKFLXA	\$609.75	CHTHFLXA	\$833.25	EORNFLMA	\$766.20	GRETFLXA	\$895.05
ARCDFLXA	\$1,079.55	CITRFLXA	\$699.15	ESPNFLXA	\$1,366.55	GSVLFLAN	\$281.80
ARCHFLMA	\$676.80	CLHNFLXA	\$766.20	ESTSFLXA	\$1,079.55	GSVLFLMA	\$281.80
ARPNFLXA	\$1,469.05	CLMTFLXA	\$1,346.05	ETVLFLAA	\$462.80	GSVLFLNW	\$487.05
ASTRFLXA	\$1,079.55	CLTNFLXA	\$1,551.05	EVRGFLXA	\$1,612.55	GSVLFLSS	\$365.80
AVPKFLXA	\$281.80	CLWRFLXA	\$281.80	FHSDFLXA	\$542.70	GVLDFLXA	\$1,305.05
BAKRFLXA	\$915.55	CNSDFLXA	\$462.80	FLBHFLMA	\$766.20	HAVNFLMA	\$1,161.55
BARTFLXA	\$587.40	CNTMFLLE	\$625.65	FLRHFLXA	\$1,100.05	HBSDFLMA	\$632.10
BAYUFLXA	\$511.30	COCOFLMA	\$281.80	FRBHFLFP	\$900.30	HDSNFLXA	\$915.55
BBPKFLXA	\$855.60	COCOFLME	\$414.30	FRPTFLXA	\$936.05	HGLDFLXA	\$487.05
BCGRFLXA	\$874.55	CPCRFLXA	\$535.55	FRSTFLXA	\$895.05	HGSPFLXA	\$1,715.05
BCRTFLBB	\$462.80	CPCRFLXB	\$487.05	FTGRFLMA	\$676.80	HLNVFLMA	\$721.50
BCRTFLBT	\$462.80	CPHZFLXA	\$977.05	FTLDFLCR	\$438.55	HLRDFLXA	\$936.05
BCRTFLMA	\$281.80	CRBLFLXA	\$1,305.05	FTLDFLCY	\$462.80	HLWDFLED	\$487.05
BCRTFLSA	\$511.30	CRCYFLXA	\$1,059.05	FTLDFLJA	\$487.05	HLWDFLHA	\$462.80
BGPIFLMA	\$2,555.55	CRRVFLXA	\$1,059.05	FTLDFLMR	\$281.80	HLWDFLMA	\$487.05
BHPKFLXA	\$462.80	CRVWFLXA	\$900.30	FTLDFLOA	\$281.80	HLWDFLPE	\$511.30
BKVLFLJF	\$1,305.05	CRWDFLXA	\$565.05	FTLDFLOV	\$390.05	HLWDFLWH	\$462.80
BLDWFLMA	\$766.20	CSCYFLBA	\$1,325.55	FTLDFLPL	\$414.30	HMSPFLXA	\$1,120.55
BLGLFLMA	\$1,141.05	CSLBFLXA	\$535.55	FTLDFLSF	\$699.15	HMSTFLEA	\$915.55
BLTWFLXA	\$281.80	CTDLFLXA	\$997.55	FTLDFLSG	\$542.70	HMSTFLHM	\$915.55
BLVWFLXA	\$565.05	CYGRFLXA	\$542.70	FTLDFLSU	\$438.55	HMSTFLNA	\$877.95
BNFYFLXA	\$1,120.55	CYLKFLXA	\$542.70	FTLDFLTA	\$281.80	HMSTFLTE	\$936.05
BNNLFLMA	\$833.25	CYLKFLXB	\$542.70	FTLDFLTN	\$365.80	HNCYFLXA	\$587.40

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.4 Pre-VBS Charges: The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate						
BNSPFLXA	\$833.25	DBRYFLDL	\$587.40	FTLDFLWN	\$609.75	HNCYFLXN	\$654.45
BORAFLXA	\$2,022.55	DBRYFLMA	\$542.70	FTMBFLXA	\$676.80	HOWYFLXA	\$1,161.55
BRBAFLXA	\$535.55	DDCYFLXA	\$1,510.05	FTMDFLXA	\$810.90	HSFRFLXA	\$699.15
BRFRFLXA	\$1,981.55	DELDFLMA	\$833.25	FTMYFLMA	\$487.05	HSNGFLXA	\$721.50
BRJTFLXA	\$766.20	DESTFLXA	\$487.05	FTMYFLXA	\$281.80	HTISFLMA	\$487.05
BRKRFLXA	\$721.50	DFSPFLXA	\$1,079.55	FTMYFLXB	\$438.55	HWTHFLMA	\$699.15
BRNDFLXA	\$542.70	DLBHFLKP	\$542.70	FTMYFLXC	\$414.30	HYPKFLXA	\$365.80
BRSNFLMA	\$877.95	DLBHFLMA	\$542.70	FTPRFLMA	\$1,920.05	IMKLFLXA	\$997.55
BRSTFLXA	\$438.55	DLSPFLMA	\$810.90	FTWBFLXA	\$281.80	INLKFLXA	\$1,018.05
BRTNFLXX	\$587.40	DNDNFLXA	\$462.80	FTWBFLXB	\$414.30	INRKFLXX	\$511.30
BSHNFLXA	\$1,079.55	DNLNFLWM	\$855.60	FTWBFLXC	\$542.70	INTRFLXA	\$1,161.55
BVHLFLXA	\$281.80	DRBHFLDF	\$438.55	FTWHFLXA	\$1,756.05	INTWFLXA	\$766.20
BWLGFLXA	\$788.55	DRBHFLMA	\$414.30	GCSPFLCN	\$721.50	INVRFLXA	\$895.05
BYBHFLAK	\$587.00	DUNDFLXA	\$587.40	GCVLFLMA	\$1,284.55	IONAFLXA	\$609.75

Wire Center	Rate						
ISLMFLMA	\$1,694.55	LKBTFLXA	\$1,284.55	MIAMFLNS	\$438.55	NKLRFLMA	\$1,038.55
JAY FLMA	\$977.05	LKCYFLMA	\$1,571.55	MIAMFLOL	\$462.80	NNPLFLXA	\$895.05
JCBHFLAB	\$587.40	LKHLFLXA	\$721.50	MIAMFLPB	\$462.80	NPLSFLXC	\$1,120.55
JCBHFLMA	\$632.10	LKLDFLXA	\$462.80	MIAMFLPL	\$542.70	NPLSFLXD	\$1,059.05
JCBHFLSP	\$542.70	LKLDFLXE	\$1,469.05	MIAMFLRR	\$535.55	NPRCFLXA	\$855.60
JCVLFLAI	\$414.30	LKLDFLXN	\$535.55	MIAMFLSH	\$462.80	NRPTFLXA	\$895.05
JCVLFLAR	\$487.05	LKMRFL01	\$281.80	MIAMFLSO	\$632.10	NRSDFLXA	\$414.30
JCVLFLBW	\$511.30	LKMRFLHE	\$484.20	MIAMFLWD	\$699.15	NSBHFLMA	\$676.80
JCVLFLCL	\$281.80	LKMRFLMA	\$281.80	MIAMFLWM	\$487.05	NWBYFLMA	\$743.85
JCVLFLFC	\$487.05	LKMYFLMA	\$511.30	MIAMFLYJ	\$721.50	OCALFLDB	\$281.80
JCVLFLGH	\$487.05	LKPCFLXA	\$877.95	MIANFLPV	\$535.55	OCALFLXA	\$281.80
JCVLFLIA	\$587.40	LKWLFLXA	\$721.50	MIANFLWK	\$535.55	OCALFLXB	\$414.30
JCVLFLJB	\$281.80	LKWLFLXE	\$874.55	MIANFLYI	\$413.30	OCALFLXC	\$414.30
JCVLFLJT	\$438.55	LKWOFLIY	\$487.05	MIAQFL06	\$414.30	OCALFLXJ	\$487.05
JCVLFLLF	\$462.80	LLMNFLXA	\$438.55	MIATFLAD	\$438.55	OCNFFLXA	\$676.80
JCVLFLNO	\$487.05	LNLKFLXA	\$855.60	MICCFLBB	\$766.20	OJUSFLTL	\$438.55
JCVLFLOG	\$281.80	LRGOFLXA	\$438.55	MLBRFLMA	\$1,489.55	OKCBFLXA	\$1,346.05
JCVLFLOW	\$542.70	LRHLFLXA	\$1,141.05	MLBYFLXA	\$609.75	OKHLFLMA	\$895.05
JCVLFLRV	\$438.55	LRVLFLXA	\$2,145.55	MLRSFLXA	\$743.85	OKLWFLXA	\$699.15
JCVLFLSJ	\$487.05	LSBGFLXA	\$956.55	MLTNFLRA	\$654.45	OLDSFLXA	\$587.40
JCVLFLSM	\$390.05	LUTZFLXA	\$654.45	MNDRFLAV	\$281.80	OLTWFLLN	\$1,161.55
JCVLFLWC	\$535.55	LVOKFLXA	\$1,981.55	MNDRFLLO	\$438.55	ORCYFLXA	\$654.45
JCVLFLWF	\$365.80	LWTYFLXA	\$997.55	MNDRFLLW	\$535.55	ORCYFLXC	\$699.15
JCVMFLED	\$535.55	LYHNFLOH	\$281.80	MNLKFLXA	\$915.55	ORLDFL60	\$281.80
JCVMFLLI	\$462.80	MALNFLXA	\$1,100.05	MNSNFLMA	\$997.55	ORLDFLAP	\$487.05

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 Intrastate T1 Digital Access (Cont'd.):
  - 1.3.3 Rates and Charges: (Cont'd)
    - 1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.4 Pre-VBS Charges: The following monthly recurring per-circuit charges apply, based on Serving Wire Center of the circuit:

Wire Center	Rate						
JCVMFLLR	\$365.80	MAYOFLXA	\$2,248.05	MNTIFLXA	\$874.55	ORLDFLCL	\$390.05
JCVMFLUF	\$511.30	MCINFLXA	\$721.50	MOISFLXA	\$1,346.05	ORLDFLLU	\$565.05
JNGSFLXA	\$2,145.55	MCLNFLXA	\$915.55	MOLNFLXA	\$679.95	ORLDFLMA	\$281.80
JPTRFLGN	\$1,490.00	MCNPFLMA	\$587.40	MRDCFLXA	\$977.05	ORLDFLOE	\$281.80
JPTRFLMA	\$721.50	MDBGFLPM	\$810.90	MRHNFLXA	\$1,387.05	ORLDFLPC	\$487.05
JSPRFLXA	\$1,940.55	MDSNFLXA	\$1,428.05	MRNNFLXA	\$874.55	ORLDFLPH	\$487.05
KGLKFLXA	\$977.05	MIAMFLAC	\$390.05	MRTHFLVE	\$2,186.55	ORLDFLSA	\$511.30
KNVLFLXA	\$936.05	MIAMFLAE	\$438.55	MTDRFLXA	\$1,161.55	ORLDFLXH	\$365.80
KSSMFLXA	\$743.85	MIAMFLAL	\$390.05	MTLDFLAP	\$487.05	ORLEFLGV	\$487.05
KSSMFLXB	\$699.15	MIAMFLAP	\$438.55	MTLDFLBR	\$487.05	ORPKFLMA	\$542.70
KSSMFLXD	\$699.15	MIAMFLBA	\$414.30	MTLDFLTC	\$511.30	ORPKFLRW	\$632.10
KYHGFLMA	\$810.90	MIAMFLBC	\$390.05	MTLDFLXA	\$511.30	ORSPFLXA	\$900.30
KYLRFLLS	\$1,346.05	MIAMFLBR	\$462.80	MTVRFLXA	\$766.20	OSPRFLXA	\$587.40
KYLRFLMA	\$1,469.05	MIAMFLCA	\$565.05	MXVLFLMA	\$855.60	OVIDFLCA	\$565.05
KYSTFLXA	\$654.45	MIAMFLDA	\$414.30	MYCYFLXA	\$877.95	PACEFLPV	\$542.70
KYWSFLMA	\$3,068.05	MIAMFLDB	\$535.55	NDADFLAA	\$414.30	PAHKFLMA	\$1,100.05
LBLLFLXA	\$936.05	MIAMFLFL	\$390.05	NDADFLAC	\$414.30	PANCFLXA	\$956.55
LDLKFLXA	\$855.60	MIAMFLGR	\$390.05	NDADFLBR	\$438.55	PBGRFLEZ	\$565.05
LGBKFLXA	\$535.55	MIAMFLHL	\$542.70	NDADFLGG	\$281.80	PCBHFLNT	\$609.75
LHACFLXA	\$676.80	MIAMFLIC	\$535.55	NDADFLOL	\$414.30	PKCYFLXA	\$542.70
LKALFLXA	\$438.55	MIAMFLKE	\$542.70	NFMYFLXA	\$414.30	PLBYFLAO	\$414.30
LKBNFLXB	\$676.80	MIAMFLME	\$281.80	NFMYFLXB	\$462.80	PLCSFLMA	\$895.05
LKBRFLXA	\$511.30	MIAMFLNM	\$438.55	NGBHFLXA	\$542.70	PLMTFLXA	\$699.15

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.4 Pre VBS Charges: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
PLSLFLXA	\$632.10	SGBEFLXA	\$511.30	TAMQFLRY	\$1,305.05	WPBHFLGR	\$565.05
PLTKFLMA	\$874.55	SGBHFLXA	\$956.55	TAMQFLSZ	\$511.30	WPBHFLHH	\$462.80
PMBHFLCS	\$535.55	SGKYFLMA	\$2,699.05	TAMQFLTA	\$281.80	WPBHFLLE	\$511.30
PMBHFLDR	\$414.30	SHLMFLXA	\$414.30	TAMSFLZK	\$542.70	WPBHFLRB	\$487.05
PMBHFLED	\$390.05	SKWYFLXA	\$414.30	THBHFLXA	\$810.90	WPBHFLRP	\$587.40
PMBHFLFE	\$281.80	SLHLFLXA	\$699.15	THNTFLXA	\$632.10	WSSDFLXA	\$414.30
PMBHFLJK	\$390.05	SLSPFLXA	\$487.05	TLCHFLXA	\$1,366.55	WSTVFLXA	\$1,141.05
PMBHFLMA	\$487.05	SMNLFLXA	\$414.30	TLHSFLXA	\$281.80	WWHTFLXA	\$877.95
PMBHFLNP	\$487.05	SNANFLXA	\$1,551.05	TLHSFLXB	\$414.30	WWSPFLHI	\$1,448.55
PMBHFLTA	\$462.80	SNDSFLXA	\$788.55	TLHSFLXC	\$414.30	WWSPFLSH	\$1,530.55
PMPKFLMA	\$1,018.05	SNFRFLMA	\$487.05	TLHSFLXD	\$414.30	YBCTFLXA	\$390.05
PNCRFLXA	\$766.20	SNISFLXA	\$810.90	TLHSFLXE	\$365.80	YNFNFLMA	\$743.85
PNCYFLCA	\$462.80	SNRSFLXA	\$833.25	TLHSFLXF	\$511.30	YNTWFLMA	\$1,079.55
PNCYFLMA	\$281.80	SNSNFLXA	\$1,100.05	TLHSFLXG	\$565.05	YULEFLMA	\$810.90
PNGRFLXA	\$877.95	SNSPFLXA	\$766.20	TLHSFLXH	\$487.05	ZLSPFLXA	\$788.55
PNISFLXA	\$699.15	SPBGFLXA	\$281.80	TMTRFLXA	\$915.55	ZPHYFLXA	\$833.25
PNLNFLXA	\$1,264.05	SPBGFLXS	\$462.80	TRENFLMA	\$956.55		
PNLSFLXA	\$462.80	SPRGFLXA	\$487.05	TRSPFLXA	\$632.10		
PNSCFLBL	\$511.30	SRSTFLXA	\$281.80	TTVLFLMA	\$743.85		
PNSCFLFP	\$281.80	SSDSFLXA	\$462.80	TVRSFLXA	\$1,079.55		
PNSCFLHC	\$487.05	SSPRFLXA	\$915.55	UMTLFLXA	\$1,018.05		
PNSCFLPB	\$743.85	STAGFLBS	\$511.30	UNVRFLXA	\$542.70		
PNSCFLUP	\$511.30	STAGFLMA	\$281.80	VENCFLXA	\$743.85		
PNSCFLWA	\$535.55	STAGFLSH	\$487.05	VENCFLXS	\$833.25		
PNVDFLMA	\$587.40	STAGFLWG	\$587.40	VERNFLMA	\$895.05		
POINFLXA	\$810.90	STCDFLXA	\$833.25	VLPRFLXA	\$565.05		
PRRNFLCC	\$699.15	STGRFLXA	\$535.55	VLPRFLXB	\$632.10		
PRRNFLMA	\$676.80	STMKFLXA	\$810.90	VRBHFLBE	\$414.30		
PRRYFLXA	\$1,325.55	STRKFLXA	\$1,120.55	VRBHFLMA	\$281.80		
PRSHFLXA	\$743.85	STRTFLMA	\$281.80	WALDFLXA	\$654.45		
PRSNFLFD	\$915.55	SVSPFLXA	\$462.80	WCHLFLXA	\$788.55		
PSDNFLXA	\$487.05	SVSSFLXA	\$587.40	WELKFLMA	\$1,079.55		
PTCTFLXA	\$915.55	SWTHFLXA	\$535.55	WHSPFLXA	\$1,694.55		
PTCYFLXA	\$936.05	SYHSFLCC	\$810.90	WIMMFLXA	\$743.85		
PTSJFLXA	\$1,018.05	TAFBFLXA	\$535.55	WLBRFLXA	\$1,776.55		

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.3 Intrastate T1 Digital Access (Cont'd.):
  - 1.3.3 Rates and Charges: (Cont'd)

1.3.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.3.3.1.4 Pre VBS Charges: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
PTSLFLMA	\$565.05	TAMPFL41	\$438.55	WLCHFLXA	\$788.55
PTSLFLSO	\$487.05	TAMPFL52	\$365.80	WLCRFLXA	\$462.80
PXTNFLXA	\$1,202.55	TAMPFLCM	\$438.55	WLSTFLXA	\$788.55
QNCYFLXA	\$956.55	TAMPFLCO	\$487.05	WLWDFLXA	\$855.60
RAFRFLXA	\$1,120.55	TAMPFLDM	\$365.80	WNDRFLXA	\$587.40
RSKNFLXA	\$654.45	TAMPFLEM	\$365.80	WNGRFLXA	\$632.10
RYHLFLXA	\$1,346.05	TAMPFLKO	\$788.55	WNHNFLXC	\$462.80
SARKFLXA	\$414.30	TAMPFLUT	\$511.30	WNPKFLTL	\$462.80
SBNGFLXA	\$565.05	TAMPFLXA	\$1,202.55	WNPKFLXA	\$438.55
SBSTFLFE	\$699.15	TAMPFLXE	\$487.05	WNPKFLXE	\$462.80
SBSTFLMA	\$632.10	TAMPFLXX	\$281.80	WPBHFL58	\$281.80
SCPKFLXA	\$632.10	TAMQFLHU	\$721.50	WPBHFLAN	\$281.80
SEKYFLXA	\$462.80	TAMQFLPM	\$587.40	WPBHFLGA	\$535.55

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - SERVICE DESCRIPTION (CONT'D.)
  - 1.3 <u>Intrastate T1 Digital Access (Cont'd.)</u>:
    - 1.3.3 Rates and Charges: (Cont'd)
      - 1.3.3.2. <u>Local Access Channel Non-Recurring Charge</u>: A \$200 non-recurring channel charge applies.
      - 1.3.3.3. <u>Network Connection Charge</u>: The following monthly recurring and non-recurring charges will apply, per connection, only for circuits installed on or after October 1, 1998, if access is Customer-provided.

Monthly Installation
\$230 \$0.00

- 1.3.3.4. <u>Cross Connect Charge</u>: A monthly \$50 charge will apply, and a non-recurring charge of \$250 will apply, per connection, for circuits installed on or after October 1, 1998, for the electrical connection of a Company intrastate T-1 Digital Access circuit to an interexchange carrier other than the Company.
- 1.3.3.5. Special Access Surcharge applies.

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.3 Intrastate T1 Digital Access (Cont'd.):
    - 1.3.3 Rates and Charges (Cont'd.):
      - 1.3.3.6. Other Non-Recurring Charges:
        - 1.3.3.6.1 Customer Support Services charges apply.
        - 1.3.3.6.2 Intrastate Administrative Non-Recurring Charges apply.
    - 1.3.4. <u>Terms and Conditions</u>: The tariff general terms and conditions apply, except that the Credit Allowance for Interruption of Local Access Service as set forth in Section 2.3 below also applies.

#### 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>

#### 1.4 Intrastate DS-3 Local Access

- 1.4.1 <u>Product Description</u>: Intrastate DS-3 Local Access is provided by the Company. Intrastate DS-3 Local Access is a high capacity digital local access arrangement that relies on DS-3 transmission technology used by the Local Exchange Carrier, or other access provider, and which will be used to the Customer's Premises to the Company's Point-of-Presence (POP) to provide Customer access to Company services.
- 1.4.2. <u>Features and Options</u>: Features and Options are not available.

# 1.4.3. Rates and Charges:

Where rates or charges are specific to a particular pricing program (i.e., Verizon Business Services (VBS) I, II, or III) those rates and charges only apply where a Customer has enrolled in that pricing program. Rates and charges which are not specific to a particular pricing program apply to all Customers.

#### 1.4.3.1. Local Access Channel Monthly Recurring Channel Charges:

1.4.3.1.1 <u>Verizon Business Services III:</u> The following monthly recurring per-circuit charges apply based on Serving Wire Center of the circuit:

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
ABDLFLXA	\$2,400.00	BNFYFLXA	\$3,920.00	CLTNFLXA	\$7,620.00	DNDNFLXA	\$2,360.00
ALCHFLXA	\$4,000.00	BNNLFLMA	\$2,790.00	CLWRFLXA	\$2,180.00	DNLNFLWM	\$3,540.00
ALFAFLXA	\$2,850.00	BNSPFLXA	\$5,440.00	CNSDFLXA	\$2,120.00	DRBHFLDF	\$2,170.00
ALFRFLXA	\$3,990.00	BORAFLXA	\$4,570.00	CNTMFLLE	\$2,520.00	DRBHFLMA	\$1,880.00
ALSPFLXA	\$2,150.00	BRBAFLXA	\$2,060.00	COCOFLMA	\$1,400.00	DUNDFLXA	\$4,080.00
ALTRFLXA	\$2,550.00	BRFRFLXA	\$4,510.00	COCOFLME	\$1,580.00	DWPKFLXA	\$4,930.00
ALVAFLXA	\$5,110.00	BRJTFLXA	\$2,240.00	CPCRFLXA	\$4,180.00	DYBHFLFN	\$1,940.00
ANMRFLXA	\$1,970.00	BRKRFLXA	\$2,530.00	CPCRFLXB	\$4,370.00	DYBHFLMA	\$1,400.00
APPKFLXA	\$2,400.00	BRNDFLXA	\$1,770.00	CPHZFLXA	\$6,040.00	DYBHFLMN	\$2,170.00
ARCDFLXA	\$5,420.00	BRSNFLMA	\$2,790.00	CRCYFLXA	\$4,060.00	DYBHFLOB	\$1,630.00
ARCHFLMA	\$2,520.00	BRTNFLXX	\$2,000.00	CRRVFLXA	\$4,150.00	DYBHFLOS	\$2,260.00
ASTRFLXA	\$5,000.00	BSHNFLXA	\$5,200.00	CRVWFLXA	\$4,200.00	DYBHFLPO	\$2,000.00
AVPKFLXA	\$2,140.00	BVHLFLXA	\$3,760.00	CRWDFLXA	\$1,940.00	DYBHFLWS	\$2,700.00
BAKRFLXA	\$3,690.00	BWLGFLXA	\$4,490.00	CSCYFLBA	\$3,700.00	EGLLFLBG	\$1,970.00
BARTFLXA	\$2,400.00	BYBHFLMA	\$2,350.00	CSLBFLXA	\$2,610.00	EGLLFLIH	\$2,000.00
BAYUFLXA	\$2,100.00	BYSHFLXA	\$2,630.00	CTDLFLXA	\$3,860.00	ENWDFLXA	\$6,060.00
BBPKFLXA	\$2,820.00	CCBHFLAF	\$2,130.00	CYGRFLXA	\$2,620.00	EORNFLMA	\$2,580.00
BCGRFLXA	\$5,710.00	CCBHFLMA	\$1,670.00	CYLKFLXA	\$4,240.00	ESTSFLXA	\$5,320.00
BCRTFLBT	\$1,600.00	CDKYFLMA	\$3,960.00	CYLKFLXB	\$4,570.00	ETVLFLAA	\$5,750.00
BCRTFLMA	\$1,400.00	CFLDFLMA	\$3,280.00	DBRYFLDL	\$2,840.00	EVRGFLXA	\$8,110.00
BCRTFLSA	\$2,000.00	CFVLFLXA	\$4,470.00	DBRYFLMA	\$2,810.00	FHSDFLXA	\$1,400.00
BGPIFLMA	\$5,460.00	CHLKFLXA	\$6,610.00	DDCYFLXA	\$6,340.00	FLBHFLMA	\$2,660.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.1 <u>Verizon Business Services III</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
BHPKFLXA	\$1,400.00	CHPLFLJA	\$2,350.00	DELDFLMA	\$2,770.00	FLRHFLXA	\$3,490.00
BKVLFLJF	\$4,610.00	CHSWFLXA	\$4,700.00	DESTFLXA	\$3,400.00	FRBHFLFP	\$2,840.00
BLDWFLMA	\$2,650.00	CHTHFLXA	\$8,400.00	DFSPFLXA	\$5,510.00	FRPTFLXA	\$5,200.00
BLGLFLMA	\$3,420.00	CITRFLXA	\$2,740.00	DLBHFLKP	\$2,320.00	FRSTFLXA	\$2,840.00
BLTWFLXA	\$7,770.00	CLHNFLXA	\$2,680.00	DLBHFLMA	\$2,320.00	FTGRFLMA	\$2,520.00
BLVWFLXA	\$4,090.00	CLMTFLXA	\$6,120.00	DLSPFLMA	\$2,770.00	FTLDFLCR	\$1,910.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.1 <u>Verizon Business Services III</u>: (Cont'd)

Wire Center	Rate						
FTLDFLCY	\$1,630.00	HLWDFLPE	\$1,610.00	JSPRFLXA	\$4,460.00	MIAMFLAP	\$1,910.00
FTLDFLJA	\$1,810.00	HLWDFLWH	\$2,230.00	KGLKFLXA	\$3,240.00	MIAMFLBA	\$1,880.00
FTLDFLMR	\$1,400.00	HMSPFLXA	\$4,450.00	KNVLFLXA	\$3,560.00	MIAMFLBC	\$1,850.00
FTLDFLOA	\$1,610.00	HMSTFLEA	\$2,780.00	KSSMFLXA	\$2,240.00	MIAMFLBR	\$1,910.00
FTLDFLPL	\$1,600.00	HMSTFLHM	\$2,780.00	KSSMFLXB	\$2,170.00	MIAMFLCA	\$1,910.00
FTLDFLSF	\$1,620.00	HMSTFLNA	\$2,650.00	KSSMFLXD	\$2,590.00	MIAMFLDB	\$1,970.00
FTLDFLSG	\$2,320.00	HMSTFLTE	\$4,080.00	KYHGFLMA	\$2,710.00	MIAMFLDI	\$1,620.00
FTLDFLSU	\$2,230.00	HNCYFLXA	\$3,420.00	KYLRFLLS	\$3,600.00	MIAMFLFL	\$1,850.00
FTLDFLTA	\$1,620.00	HNCYFLXN	\$4,410.00	KYLRFLMA	\$3,800.00	MIAMFLGR	\$1,400.00
FTLDFLWN	\$2,410.00	HOWYFLXA	\$5,570.00	KYSTFLXA	\$2,740.00	MIAMFLHL	\$1,610.00
FTMBFLXA	\$4,970.00	HSNGFLXA	\$3,470.00	KYWSFLMA	\$6,270.00	MIAMFLIC	\$2,030.00
FTMDFLXA	\$4,550.00	HTISFLMA	\$3,480.00	LBLLFLXA	\$5,840.00	MIAMFLKE	\$2,000.00
FTMYFLMA	\$4,660.00	HWTHFLMA	\$2,550.00	LDLKFLXA	\$4,670.00	MIAMFLME	\$1,400.00
FTMYFLXA	\$3,080.00	HYPKFLXA	\$1,410.00	LEE FLXA	\$6,740.00	MIAMFLNM	\$2,030.00
FTMYFLXB	\$4,240.00	IMKLFLXA	\$6,110.00	LGBKFLXA	\$2,150.00	MIAMFLNS	\$1,910.00
FTMYFLXC	\$3,820.00	INLKFLXA	\$3,090.00	LHACFLXA	\$4,970.00	MIAMFLOL	\$2,030.00
FTPRFLMA	\$2,240.00	INRKFLXX	\$2,520.00	LKALFLXA	\$3,410.00	MIAMFLPB	\$1,940.00
FTWBFLXA	\$3,030.00	INTRFLXA	\$3,700.00	LKBNFLXB	\$2,380.00	MIAMFLPL	\$1,400.00
FTWBFLXB	\$3,370.00	INTWFLXA	\$5,140.00	LKBRFLXA	\$2,310.00	MIAMFLRR	\$1,910.00
FTWBFLXC	\$3,210.00	INVRFLXA	\$4,210.00	LKBTFLXA	\$3,520.00	MIAMFLSH	\$1,940.00
FTWHFLXA	\$4,190.00	IONAFLXA	\$4,770.00	LKCYFLMA	\$4,090.00	MIAMFLSO	\$2,060.00
GCSPFLCN	\$2,780.00	ISLMFLMA	\$4,120.00	LKHLFLXA	\$3,350.00	MIAMFLUJ	\$2,520.00
GCVLFLMA	\$3,620.00	JAY FLMA	\$3,360.00	LKLDFLXA	\$2,440.00	MIAMFLWD	\$2,290.00
GDRGFLXA	\$5,330.00	JCBHFLAB	\$2,380.00	LKLDFLXE	\$1,400.00	MIAMFLWM	\$1,400.00
GENVFLMA	\$2,710.00	JCBHFLMA	\$2,440.00	LKLDFLXN	\$3,080.00	MIANFLWK	\$1,620.00
GLBRFLMC	\$2,000.00	JCBHFLSP	\$2,380.00	LKMRFLHE	\$2,460.00	MIANFLYI	\$1,620.00
GLDLFLXA	\$5,810.00	JCVLFLAI	\$2,820.00	LKMRFLMA	\$2,700.00	MIAQFL06	\$2,170.00
GLGCFLXA	\$6,170.00	JCVLFLAR	\$1,970.00	LKPCFLXA	\$4,420.00	MIATFLAD	\$1,620.00
GLRDFLXA	\$2,190.00	JCVLFLBW	\$2,190.00	LKWLFLXA	\$2,730.00	MICCFLBB	\$4,300.00
GNBOFLXA	\$6,900.00	JCVLFLCL	\$1,400.00	LKWLFLXE	\$2,950.00	MLBRFLMA	\$1,400.00
GNDYFLXA	\$1,400.00	JCVLFLFC	\$1,970.00	LLMNFLXA	\$1,810.00	MLBYFLXA	\$2,200.00
GNVLFLXA	\$5,810.00	JCVLFLGH	\$2,820.00	LNLKFLXA	\$3,300.00	MLRSFLXA	\$2,670.00
GNWDFLXA	\$4,790.00	JCVLFLIA	\$2,290.00	LRGOFLXA	\$2,060.00	MLTNFLRA	\$2,610.00
GRETFLXA	\$8,330.00	JCVLFLJT	\$2,000.00	LRVLFLXA	\$4,740.00	MNDRFLAV	\$2,170.00
GSVLFLAN	\$1,620.00	JCVLFLLF	\$1,940.00	LSBGFLXA	\$4,720.00	MNDRFLLO	\$2,320.00
GSVLFLMA	\$1,400.00	JCVLFLNO	\$1,630.00	LUTZFLXA	\$2,630.00	MNDRFLLW	\$2,410.00
GSVLFLNW	\$1,850.00	JCVLFLOG	\$2,150.00	LVOKFLXA	\$4,530.00	MNLKFLXA	\$4,300.00
GVLDFLXA	\$5,940.00	JCVLFLOW	\$2,230.00	LWTYFLXA	\$3,360.00	MNSNFLMA	\$3,360.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.1 <u>Verizon Business Services III</u>: (Cont'd)

Wire Center	Rate						
HAVNFLMA	\$5,540.00	JCVLFLRV	\$1,600.00	LYHNFLOH	\$2,250.00	MNTIFLXA	\$4,870.00
HBSDFLMA	\$2,790.00	JCVLFLSJ	\$1,970.00	MALNFLXA	\$4,990.00	MOISFLXA	\$7,240.00
HDSNFLXA	\$3,940.00	JCVLFLSM	\$1,560.00	MAYOFLXA	\$4,870.00	MOLNFLXA	\$2,790.00
HGLDFLXA	\$2,520.00	JCVLFLWC	\$2,030.00	MCINFLXA	\$2,660.00	MRDCFLXA	\$6,040.00
HGSPFLXA	\$4,120.00	JCVLFLWF	\$1,620.00	MCLNFLXA	\$4,800.00	MRHNFLXA	\$7,020.00
HLNVFLMA	\$2,710.00	JCVMFLLI	\$2,170.00	MCNPFLMA	\$2,390.00	MRNNFLXA	\$4,520.00
HLRDFLXA	\$2,950.00	JCVMFLUF	\$2,170.00	MDBGFLPM	\$2,740.00	MRTHFLVE	\$4,900.00
HLWDFLED	\$2,120.00	JNGSFLXA	\$4,730.00	MDSNFLXA	\$6,610.00	MTDRFLXA	\$5,600.00
HLWDFLHA	\$2,260.00	JPTRFLGN	\$2,840.00	MIAMFLAE	\$1,400.00	MTLDFLAP	\$2,140.00
HLWDFLMA	\$2,030.00	JPTRFLMA	\$2,580.00	MIAMFLAL	\$1,850.00	MTLDFLXA	\$1,980.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.1 <u>Verizon Business Services III</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
MTVRFLXA	\$2,940.00	ORLFFLEJ	\$5,750.00	PTSLFLMA	\$3,420.00	TAFBFLXA	\$5,060.00
MXVLFLMA	\$2,780.00	ORLFFLHX	\$5,750.00	PTSLFLSO	\$3,500.00	TAMPFL52	\$2,190.00
MYCYFLXA	\$2,490.00	ORLFFLQR	\$5,750.00	QNCYFLXA	\$6,760.00	TAMPFLCM	\$1,400.00
NDADFLAA	\$2,120.00	ORPKFLMA	\$2,420.00	RAFRFLXA	\$3,230.00	TAMPFLCO	\$1,400.00
NDADFLAC	\$2,260.00	ORPKFLRW	\$2,520.00	RSKNFLXA	\$3,410.00	TAMPFLDM	\$1,400.00
NDADFLBR	\$2,290.00	ORSPFLXA	\$2,870.00	RYHLFLXA	\$4,850.00	TAMPFLEM	\$1,400.00
NDADFLGG	\$1,610.00	OSPRFLXA	\$5,030.00	SARKFLXA	\$4,140.00	TAMPFLKO	\$1,400.00
NDADFLOL	\$2,320.00	OVIDFLCA	\$2,350.00	SBNGFLXA	\$3,690.00	TAMPFLXA	\$1,400.00
NFMYFLXA	\$4,170.00	PACEFLPV	\$2,480.00	SBSTFLFE	\$4,130.00	TAMPFLXE	\$1,400.00
NFMYFLXB	\$4,310.00	PAHKFLMA	\$3,420.00	SBSTFLMA	\$4,090.00	TAMPFLXX	\$1,400.00
NGBHFLXA	\$2,120.00	PANCFLXA	\$5,140.00	SCPKFLXA	\$4,840.00	TAMQFLHU	\$5,750.00
NKLRFLMA	\$3,150.00	PCBHFLNT	\$3,540.00	SEKYFLXA	\$4,360.00	TAMQFLPM	\$5,750.00
NNPLFLXA	\$5,400.00	PKCYFLXA	\$3,410.00	SGBEFLXA	\$2,630.00	TAMQFLRY	\$5,750.00
NPLSFLXC	\$6,510.00	PLCSFLMA	\$3,050.00	SGBHFLXA	\$5,250.00	TAMQFLTA	\$5,750.00
NPLSFLXD	\$5,830.00	PLMTFLXA	\$1,910.00	SGKYFLMA	\$5,690.00	TAMSFLCZ	\$5,750.00
NPRCFLXA	\$2,850.00	PLSLFLXA	\$2,020.00	SHLMFLXA	\$3,420.00	THNTFLXA	\$2,740.00
NRPTFLXA	\$6,810.00	PLTKFLMA	\$3,670.00	SKWYFLXA	\$1,940.00	TLCHFLXA	\$5,890.00
NRSDFLXA	\$2,170.00	PMBHFLCS	\$2,320.00	SLHLFLXA	\$4,090.00	TLHSFLXA	\$2,250.00
NSBHFLMA	\$2,550.00	PMBHFLED	\$2,170.00	SLSPFLXA	\$1,660.00	TLHSFLXB	\$3,020.00
NWBYFLMA	\$2,610.00	PMBHFLFE	\$1,650.00	SMNLFLXA	\$1,620.00	TLHSFLXC	\$3,020.00
OCALFLXA	\$3,260.00	PMBHFLMA	\$2,230.00	SNANFLXA	\$6,380.00	TLHSFLXD	\$3,020.00
OCALFLXB	\$3,280.00	PMBHFLNP	\$2,230.00	SNDSFLXA	\$5,730.00	TLHSFLXE	\$3,200.00
OCALFLXC	\$3,700.00	PMBHFLTA	\$2,260.00	SNFRFLMA	\$2,680.00	TLHSFLXF	\$3,270.00
OCALFLXJ	\$3,800.00	PMPKFLMA	\$4,030.00	SNISFLXA	\$5,370.00	TLHSFLXG	\$3,780.00
OCNFFLXA	\$3,880.00	PNCRFLXA	\$2,000.00	SNRSFLXA	\$4,840.00	TLHSFLXH	\$3,520.00
OJUSFLTL	\$3,230.00	PNCYFLCA	\$3,180.00	SNSNFLXA	\$6,090.00	TMTRFLXA	\$1,810.00
OKCBFLXA	\$6,350.00	PNCYFLMA	\$3,150.00	SNSPFLXA	\$3,740.00	TRENFLMA	\$3,110.00
OKHLFLMA	\$3,060.00	PNGRFLXA	\$5,570.00	SPBGFLXA	\$1,400.00	TRSPFLXA	\$3,020.00
OKLWFLXA	\$4,390.00	PNISFLXA	\$5,040.00	SPBGFLXS	\$2,100.00	TTVLFLMA	\$1,980.00
OLDSFLXA	\$2,520.00	PNLNFLXA	\$6,210.00	SPCPFLXA	\$5,140.00	TVRSFLXA	\$5,390.00
OLTWFLLN	\$3,440.00	PNLSFLXA	\$1,870.00	SPRGFLXA	\$4,140.00	UMTLFLXA	\$5,180.00
ORCYFLXA	\$4,150.00	PNSCFLBL	\$1,400.00	SRSTFLXA	\$2,970.00	UNVRFLXA	\$1,400.00
ORCYFLXC	\$3,180.00	PNSCFLFP	\$2,000.00	SSDSFLXA	\$3,770.00	VENCFLXA	\$5,140.00
ORLDFL60	\$1,620.00	PNSCFLHC	\$2,000.00	SSPRFLXA	\$4,190.00	VENCFLXS	\$6,250.00
ORLDFLAP	\$1,940.00	PNSCFLPB	\$2,410.00	STAGFLBS	\$3,370.00	VERNFLMA	\$3,720.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.1 <u>Verizon Business Services III</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
ORLDFLCL	\$1,400.00	PNSCFLWA	\$1,910.00	STAGFLMA	\$2,470.00	VLPRFLXA	\$3,910.00
ORLDFLLU	\$1,620.00	PNVDFLMA	\$2,680.00	STAGFLSH	\$3,340.00	VLPRFLXB	\$4,030.00
ORLDFLMA	\$1,400.00	POINFLXA	\$2,930.00	STAGFLWG	\$2,340.00	VRBHFLBE	\$3,580.00
ORLDFLOE	\$1,620.00	PRRNFLCC	\$2,120.00	STCDFLXA	\$2,930.00	VRBHFLMA	\$3,550.00
ORLDFLPC	\$1,400.00	PRRNFLMA	\$2,390.00	STGRFLXA	\$2,420.00	WALDFLXA	\$2,540.00
ORLDFLPH	\$1,670.00	PRRYFLXA	\$8,840.00	STMKFLXA	\$4,540.00	WCHLFLXA	\$4,420.00
ORLDFLSA	\$1,400.00	PRSHFLXA	\$1,970.00	STRKFLXA	\$3,660.00	WELKFLMA	\$4,060.00
ORLDFLSO	\$2,130.00	PRSNFLFD	\$3,100.00	STRTFLMA	\$3,310.00	WHSPFLXA	\$4,150.00
ORLDFLWR	\$2,130.00	PSDNFLXA	\$2,190.00	SVSPFLXA	\$3,650.00	WIMMFLXA	\$3,850.00
ORLDFLXH	\$1,620.00	PTCTFLXA	\$3,150.00	SVSSFLXA	\$4,140.00	WLBRFLXA	\$4,250.00
ORLEFLGV	\$2,140.00	PTCYFLXA	\$3,020.00	SWTHFLXA	\$1,400.00	WLCHFLXA	\$3,080.00
ORLEFLKT	\$5,750.00	PTSJFLXA	\$8,480.00	SYHSFLCC	\$3,690.00	WLCRFLXA	\$1,870.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.1 <u>Verizon Business Services III</u>: (Cont'd)

Wire Center	Rate						
WLHLFLXA	\$5,390.00	WNPKFLTL	\$2,050.00	WPBHFLHH	\$1,620.00	WWSPFLSH	\$4,960.00
WLSTFLXA	\$3,040.00	WNPKFLXA	\$1,870.00	WPBHFLLE	\$2,000.00	YBCTFLXA	\$1,400.00
WLWDFLXA	\$4,620.00	WNPKFLXE	\$2,050.00	WPBHFLRB	\$1,630.00	YNFNFLMA	\$3,720.00
WNDRFLXA	\$2,360.00	WPBHFL58	\$1,620.00	WPBHFLRP	\$2,390.00	YNTWFLMA	\$3,790.00
WNGRFLXA	\$2,290.00	WPBHFLAN	\$1,400.00	WSSDFLXA	\$1,400.00	YULEFLMA	\$2,590.00
WNHNFLBU	\$3,230.00	WPBHFLGA	\$1,670.00	WSTVFLXA	\$4,520.00	ZLSPFLXA	\$4,180.00
WNHNFLXC	\$2,940.00	WPBHFLGR	\$2,350.00	WWSPFLHI	\$4,830.00	ZPHYFLXA	\$2,370.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
    - 1.4.3. Rates and Charges: (Cont'd)
      - 1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.2 <u>Verizon Business Services II:</u> The following monthly recurring percircuit charges apply based on Serving Wire Center of the circuit:

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
ABDLFLXA	\$2,950.00	CFLDFLMA	\$6,180.00	DYBHFLPO	\$5,340.00	GSVLFLNW	\$4,640.00
ALCHFLXA	\$5,080.00	CFVLFLXA	\$5,920.00	EGLLFLBG	\$4,500.00	GVLDFLXA	\$11,090.00
ALFAFLXA	\$2,100.00	CHLKFLXA	\$8,700.00	EGLLFLIH	\$4,570.00	HAVNFLMA	\$7,330.00
ALFRFLXA	\$5,110.00	CHPLFLJA	\$2,850.00	ENWDFLXA	\$7,920.00	HBSDFLMA	\$3,650.00
ALSPFLXA	\$2,980.00	CHSWFLXA	\$9,750.00	EORNFLMA	\$3,980.00	HDSNFLXA	\$4,720.00
ALTRFLXA	\$3,180.00	CHTHFLXA	\$9,640.00	ESTSFLXA	\$10,000.00	HGLDFLXA	\$3,020.00
ALVAFLXA	\$9,580.00	CITRFLXA	\$5,870.00	EVRGFLXA	\$14,630.00	HGSPFLXA	\$5,200.00
ANMRFLXA	\$2,360.00	CLHNFLXA	\$3,660.00	FHSDFLXA	\$1,600.00	HLNVFLMA	\$6,090.00
APPKFLXA	\$3,360.00	CLMTFLXA	\$11,270.00	FLBHFLMA	\$5,250.00	HLRDFLXA	\$4,070.00
ARCDFLXA	\$7,230.00	CLTNFLXA	\$10,100.00	FLRHFLXA	\$4,450.00	HLWDFLHA	\$3,320.00
ARCHFLMA	\$5,510.00	CLWRFLXA	\$2,610.00	FRBHFLFP	\$3,970.00	HLWDFLMA	\$2,960.00
ASTRFLXA	\$9,330.00	CNSDFLXA	\$2,540.00	FRPTFLXA	\$10,020.00	HLWDFLPE	\$3,470.00
AVPKFLXA	\$2,780.00	CNTMFLLE	\$5,150.00	FRSTFLXA	\$3,400.00	HLWDFLWH	\$3,320.00
BAKRFLXA	\$7,700.00	COCOFLMA	\$3,300.00	FTGRFLMA	\$3,330.00	HMSPFLXA	\$9,410.00
BARTFLXA	\$2,870.00	COCOFLME	\$4,430.00	FTLDFLCR	\$2,670.00	HMSTFLEA	\$4,270.00
BAYUFLXA	\$2,520.00	COCYFL13	\$3,010.00	FTLDFLCY	\$2,380.00	HMSTFLHM	\$4,270.00
BBPKFLXA	\$3,380.00	CPCRFLXA	\$9,950.00	FTLDFLJA	\$2,790.00	HMSTFLNA	\$3,980.00
BCGRFLXA	\$10,800.00	CPCRFLXB	\$10,540.00	FTLDFLMR	\$1,600.00	HNCYFLXA	\$4,100.00
BCRTFLBT	\$4,050.00	CPHZFLXA	\$10,190.00	FTLDFLOA	\$3,560.00	HNCYFLXN	\$4,630.00
BCRTFLMA	\$3,300.00	CRCYFLXA	\$5,110.00	FTLDFLPL	\$2,540.00	HOWYFLXA	\$10,460.00
BCRTFLSA	\$3,900.00	CRRVFLXA	\$8,930.00	FTLDFLSG	\$3,470.00	HSNGFLXA	\$4,420.00
BGPIFLMA	\$7,510.00	CRVWFLXA	\$8,180.00	FTLDFLSU	\$3,250.00	HTISFLMA	\$4,590.00
BHPKFLXA	\$1,600.00	CRWDFLXA	\$2,320.00	FTLDFLWN	\$3,690.00	HWTHFLMA	\$5,590.00
BKVLFLJF	\$8,120.00	CSCYFLBA	\$6,800.00	FTMBFLXA	\$7,560.00	HYPKFLXA	\$2,120.00
BLDWFLMA	\$3,490.00	CSLBFLXA	\$3,540.00	FTMDFLXA	\$5,930.00	IMKLFLXA	\$11,320.00
BLGLFLMA	\$5,640.00	CTDLFLXA	\$4,930.00	FTMYFLXA	\$4,030.00	INLKFLXA	\$3,700.00
BLTWFLXA	\$8,940.00	CYGRFLXA	\$3,830.00	FTMYFLXB	\$6,240.00	INRKFLXX	\$3,020.00
BLVWFLXA	\$8,230.00	CYLKFLXA	\$6,240.00	FTMYFLXC	\$5,480.00	INTRFLXA	\$4,670.00
BNFYFLXA	\$4,760.00	CYLKFLXB	\$10,710.00	FTPRFLMA	\$2,850.00	INTWFLXA	\$6,830.00
BNNLFLMA	\$5,390.00	DBRYFLDL	\$4,410.00	FTWBFLXA	\$7,280.00	INVRFLXA	\$9,010.00
BNSPFLXA	\$12,020.00	DBRYFLMA	\$4,340.00	FTWBFLXB	\$7,540.00	IONAFLXA	\$11,230.00
BORAFLXA	\$5,690.00	DDCYFLXA	\$11,850.00	FTWBFLXC	\$7,140.00	ISLMFLMA	\$5,580.00
BRBAFLXA	\$2,470.00	DELDFLMA	\$6,020.00	FTWHFLXA	\$5,300.00	JAY FLMA	\$6,890.00
BRFRFLXA	\$5,620.00	DESTFLXA	\$7,690.00	GCSPFLCN	\$3,940.00	JCBHFLAB	\$3,180.00
BRJTFLXA	\$2,680.00	DFSPFLXA	\$10,220.00	GCVLFLMA	\$4,510.00	JCBHFLMA	\$2,580.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.2 <u>Verizon Business Services II</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
BRKRFLXA	\$5,420.00	DLBHFLKP	\$4,260.00	GDRGFLXA	\$6,850.00	JCBHFLSP	\$3,130.00
BRNDFLXA	\$2,460.00	DLBHFLMA	\$3,900.00	GENVFLMA	\$4,120.00	JCVLFLAR	\$2,110.00
BRSNFLMA	\$5,480.00	DLSPFLMA	\$5,950.00	GLBRFLMC	\$5,220.00	JCVLFLBW	\$2,190.00
BRTNFLXX	\$2,390.00	DNDNFLXA	\$2,830.00	GLDLFLXA	\$10,480.00	JCVLFLCL	\$1,850.00
BSHNFLXA	\$10,250.00	DNLNFLWM	\$6,560.00	GLGCFLXA	\$12,710.00	JCVLFLFC	\$2,820.00
BVHLFLXA	\$8,530.00	DRBHFLMA	\$3,610.00	GLRDFLXA	\$2,980.00	JCVLFLIA	\$3,030.00
BWLGFLXA	\$5,840.00	DUNDFLXA	\$4,100.00	GNBOFLXA	\$8,450.00	JCVLFLJT	\$2,130.00
BYBHFLMA	\$3,540.00	DWPKFLXA	\$6,100.00	GNDYFLXA	\$1,600.00	JCVLFLLF	\$2,740.00
BYSHFLXA	\$2,070.00	DYBHFLFN	\$4,500.00	GNVLFLXA	\$7,650.00	JCVLFLNO	\$2,110.00
CCBHFLAF	\$4,790.00	DYBHFLMA	\$3,300.00	GNWDFLXA	\$6,150.00	JCVLFLOW	\$2,190.00
CCBHFLMA	\$4,790.00	DYBHFLOB	\$4,570.00	GRETFLXA	\$10,620.00	JCVLFLRV	\$2,050.00
CDKYFLMA	\$7,170.00	DYBHFLOS	\$5,080.00	GSVLFLMA	\$3,300.00	JCVLFLSJ	\$3,690.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.2 <u>Verizon Business Services II</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
JCVLFLSM	\$1,990.00	MDSNFLXA	\$8,700.00	NDADFLBR	\$4,160.00	PMBHFLFE	\$3,250.00
JCVLFLWC	\$2,160.00	MIAMFLAE	\$1,600.00	NDADFLGG	\$3,320.00	PMBHFLMA	\$2,750.00
JNGSFLXA	\$5,910.00	MIAMFLAL	\$1,600.00	NDADFLOL	\$2,960.00	PMBHFLNP	\$2,750.00
JPTRFLMA	\$3,830.00	MIAMFLAP	\$2,590.00	NFMYFLXA	\$6,120.00	PMBHFLTA	\$3,320.00
JSPRFLXA	\$5,620.00	MIAMFLBA	\$2,600.00	NFMYFLXB	\$9,930.00	PMPKFLMA	\$5,080.00
KGLKFLXA	\$4,710.00	MIAMFLBC	\$2,530.00	NGBHFLXA	\$2,540.00	PNCRFLXA	\$2,390.00
KNVLFLXA	\$6,920.00	MIAMFLBR	\$2,670.00	NKLRFLMA	\$3,860.00	PNCYFLCA	\$5,240.00
KSSMFLXA	\$3,650.00	MIAMFLCA	\$2,670.00	NNPLFLXA	\$11,390.00	PNCYFLMA	\$5,170.00
KSSMFLXB	\$3,620.00	MIAMFLDB	\$2,820.00	NPLSFLXC	\$13,230.00	PNGRFLXA	\$9,150.00
KSSMFLXD	\$4,020.00	MIAMFLFL	\$2,600.00	NPLSFLXD	\$11,940.00	PNISFLXA	\$10,970.00
KYHGFLMA	\$5,950.00	MIAMFLGR	\$1,850.00	NPRCFLXA	\$3,420.00	PNLNFLXA	\$11,160.00
KYLRFLLS	\$4,480.00	MIAMFLHL	\$2,820.00	NRPTFLXA	\$3,560.00	PNLSFLXA	\$2,240.00
KYLRFLMA	\$4,970.00	MIAMFLIC	\$2,960.00	NRSDFLXA	\$2,600.00	PNSCFLBL	\$4,720.00
KYSTFLXA	\$2,180.00	MIAMFLKE	\$2,890.00	NSBHFLMA	\$5,510.00	PNSCFLFP	\$3,300.00
KYWSFLMA	\$8,420.00	MIAMFLME	\$2,530.00	NWBYFLMA	\$5,730.00	PNSCFLHC	\$4,640.00
LBLLFLXA	\$9,230.00	MIAMFLNM	\$2,890.00	OCALFLXA	\$6,900.00	PNSCFLPB	\$5,730.00
LDLKFLXA	\$9,160.00	MIAMFLNS	\$2,600.00	OCALFLXB	\$6,970.00	PNSCFLWA	\$4,790.00
LEE FLXA	\$8,870.00	MIAMFLOL	\$3,250.00	OCALFLXC	\$7,510.00	PNVDFLMA	\$3,380.00
LGBKFLXA	\$2,580.00	MIAMFLPB	\$2,560.00	OCALFLXJ	\$7,670.00	POINFLXA	\$5,430.00
LHACFLXA	\$10,190.00	MIAMFLPL	\$1,600.00	OCNFFLXA	\$8,030.00	PRRNFLMA	\$3,400.00
LKALFLXA	\$3,300.00	MIAMFLRR	\$2,670.00	OKCBFLXA	\$8,360.00	PRRYFLXA	\$10,320.00
LKBNFLXB	\$6,520.00	MIAMFLSH	\$2,670.00	OKHLFLMA	\$6,600.00	PRSHFLXA	\$2,360.00
LKBRFLXA	\$3,150.00	MIAMFLSO	\$2,890.00	OKLWFLXA	\$8,660.00	PRSNFLFD	\$6,670.00
LKBTFLXA	\$4,510.00	MIAMFLWD	\$3,320.00	OLDSFLXA	\$3,020.00	PSDNFLXA	\$2,620.00
LKCYFLMA	\$5,130.00	MIAMFLWM	\$1,600.00	OLTWFLLN	\$6,420.00	PTCTFLXA	\$6,070.00
LKHLFLXA	\$5,220.00	MICCFLBB	\$5,590.00	ORCYFLXA	\$5,150.00	PTCYFLXA	\$3,620.00
LKLDFLXA	\$2,920.00	MLBRFLMA	\$3,300.00	ORCYFLXC	\$5,140.00	PTSJFLXA	\$10,240.00
LKLDFLXE	\$1,620.00	MLBYFLXA	\$2,630.00	ORLDFLAP	\$2,820.00	PTSLFLMA	\$4,370.00
LKLDFLXN	\$3,690.00	MLRSFLXA	\$5,970.00	ORLDFLCL	\$1,600.00	PTSLFLSO	\$4,510.00
LKMRFLMA	\$7,210.00	MLTNFLRA	\$5,440.00	ORLDFLMA	\$1,600.00	QNCYFLXA	\$11,130.00
LKPCFLXA	\$6,190.00	MNDRFLAV	\$3,020.00	ORLDFLPC	\$2,820.00	RAFRFLXA	\$4,260.00
LKWLFLXA	\$3,270.00	MNDRFLLO	\$3,120.00	ORLDFLPH	\$2,820.00	RSKNFLXA	\$4,080.00
LKWLFLXE	\$3,540.00	MNDRFLLW	\$3,320.00	ORLDFLSA	\$2,960.00	RYHLFLXA	\$6,150.00
LLMNFLXA	\$2,170.00	MNLKFLXA	\$5,150.00	ORPKFLMA	\$3,190.00	SARKFLXA	\$6,480.00
LNLKFLXA	\$4,220.00	MNSNFLMA	\$6,960.00	ORPKFLRW	\$3,250.00	SBNGFLXA	\$4,970.00
LRGOFLXA	\$2,460.00	MNTIFLXA	\$6,440.00	ORSPFLXA	\$5,990.00	SBSTFLFE	\$5,400.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.2 <u>Verizon Business Services II</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
LRVLFLXA	\$5,910.00	MOISFLXA	\$14,190.00	OSPRFLXA	\$7,680.00	SBSTFLMA	\$5,350.00
LSBGFLXA	\$9,300.00	MOLNFLXA	\$6,080.00	OVIDFLCA	\$3,320.00	SCPKFLXA	\$11,230.00
LUTZFLXA	\$3,150.00	MRDCFLXA	\$9,150.00	PACEFLPV	\$5,080.00	SEKYFLXA	\$6,880.00
LVOKFLXA	\$5,620.00	MRHNFLXA	\$9,320.00	PAHKFLMA	\$5,640.00	SGBEFLXA	\$3,150.00
LWTYFLXA	\$4,760.00	MRNNFLXA	\$5,810.00	PANCFLXA	\$6,790.00	SGBHFLXA	\$10,170.00
LYHNFLOH	\$3,760.00	MRTHFLVE	\$6,710.00	PCBHFLNT	\$5,890.00	SGKYFLMA	\$7,840.00
MALNFLXA	\$6,410.00	MTDRFLXA	\$10,410.00	PKCYFLXA	\$3,830.00	SHLMFLXA	\$7,680.00
MAYOFLXA	\$6,070.00	MTLDFLXA	\$2,710.00	PLCSFLMA	\$5,800.00	SKWYFLXA	\$2,320.00
MCINFLXA	\$5,790.00	MTVRFLXA	\$4,250.00	PLMTFLXA	\$2,280.00	SLHLFLXA	\$5,500.00
MCLNFLXA	\$6,130.00	MXVLFLMA	\$3,650.00	PLSLFLXA	\$2,390.00	SLSPFLXA	\$2,240.00
MCNPFLMA	\$5,220.00	MYCYFLXA	\$2,980.00	PLTKFLMA	\$4,550.00	SMNLFLXA	\$1,940.00
MDBGFLPM	\$3,550.00	NDADFLAC	\$2,910.00	PMBHFLCS	\$3,470.00	SNANFLXA	\$11,930.00

Rate

\$5,200.00

\$8,940.00

\$3,300.00

\$5,470.00

\$2,520.00

## VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.2 <u>Verizon Business Services II</u>: (Cont'd)

	T	T	T	T
Wire Center	Rate	Wire Center	Rate	Wire Center
SNDSFLXA	\$7,370.00	TRENFLMA	\$5,950.00	YNFNFLMA
SNFRFLMA	\$3,980.00	TRSPFLXA	\$3,620.00	YNTWFLMA
SNISFLXA	\$11,930.00	TTVLFLMA	\$5,420.00	YULEFLMA
SNRSFLXA	\$9,740.00	TVRSFLXA	\$10,150.00	ZLSPFLXA
SNSNFLXA	\$6,950.00	UMTLFLXA	\$9,770.00	ZPHYFLXA
SNSPFLXA	\$4,480.00	UNVRFLXA	\$1,600.00	
SPBGFLXA	\$1,600.00	VENCFLXA	\$6,820.00	
SPBGFLXS	\$2,390.00	VENCFLXS	\$9,150.00	
SPCPFLXA	\$6,790.00	VERNFLMA	\$4,600.00	
SPRGFLXA	\$3,030.00	VLPRFLXA	\$8,220.00	
SRSTFLXA	\$4,220.00	VLPRFLXB	\$8,510.00	
SSDSFLXA	\$4,520.00	VRBHFLBE	\$4,600.00	
SSPRFLXA	\$8,070.00	VRBHFLMA	\$4,560.00	
STAGFLBS	\$4,930.00	WALDFLXA	\$5,500.00	
STAGFLMA	\$4,580.00	WCHLFLXA	\$5,840.00	
STAGFLSH	\$4,880.00	WELKFLMA	\$5,150.00	
STAGFLWG	\$4,070.00	WHSPFLXA	\$5,210.00	
STCDFLXA	\$6,350.00	WIMMFLXA	\$2,340.00	
STGRFLXA	\$2,900.00	WLBRFLXA	\$5,340.00	
STMKFLXA	\$6,000.00	WLCHFLXA	\$2,390.00	
STRKFLXA	\$5,110.00	WLCRFLXA	\$3,140.00	
STRTFLMA	\$4,400.00	WLHLFLXA	\$9,310.00	
SVSPFLXA	\$7,370.00	WLSTFLXA	\$6,270.00	
SVSSFLXA	\$8,250.00	WLWDFLXA	\$9,230.00	
SWTHFLXA	\$1,600.00	WNDRFLXA	\$3,580.00	
SYHSFLCC	\$4,700.00	WNGRFLXA	\$3,360.00	
TAFBFLXA	\$6,640.00	WNHNFLXC	\$3,300.00	
TAMPFLXA	\$3,010.00	WNPKFLXA	\$2,550.00	
TAMPFLXE	\$1,600.00	WNPKFLXE	\$3,010.00	
TAMPFLXX	\$1,600.00	WPBHFLAN	\$1,600.00	
THNTFLXA	\$2,180.00	WPBHFLGA	\$2,960.00	
TLCHFLXA	\$11,250.00	WPBHFLGR	\$5,780.00	
TLHSFLXA	\$2,920.00	WPBHFLHH	\$4,910.00	
TLHSFLXB	\$3,930.00	WPBHFLLE	\$2,890.00	
TLHSFLXC	\$3,930.00	WPBHFLRB	\$2,820.00	
TLHSFLXD	\$3,930.00	WPBHFLRP	\$3,400.00	
TLHSFLXE	\$4,170.00	WSSDFLXA	\$1,600.00	
TLHSFLXF	\$4,260.00	WSTVFLXA	\$5,630.00	
TLHSFLXG	\$4,920.00	WWSPFLHI	\$10,840.00	
TLHSFLXH	\$4,590.00	WWSPFLSH	\$11,080.00	
TMTRFLXA	\$2,170.00	YBCTFLXA	\$1,600.00	

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
    - 1.4.3. Rates and Charges: (Cont'd)
      - 1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)
        - 1.4.3.1.3 <u>Verizon Business Services I:</u> The following monthly recurring per-circuit charges apply based on Serving Wire Center of the circuit:

Wire	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
Center	Nate	Wife Ceffice	Nate	Wife Ceffice	Nate	Wife Center	Rate
ABDLFLXA	\$1,600.00	CFVLFLXA	\$9,400.00	DYBHFLWS	\$3,800.00	GRETFLXA	\$7,500.00
ALCHFLXA	\$9,800.00	CHLKFLXA	\$15,600.00	EGLLFLBG	\$8,500.00	GSVLFLAN	\$3,400.00
ALFAFLXA	\$1,700.00	CHPLFLJA	\$4,400.00	EGLLFLIH	\$9,000.00	GSVLFLMA	\$4,400.00
ALFRFLXA	\$10,400.00	CHSWFLXA	\$10,100.00	ENWDFLXA	\$2,900.00	GSVLFLNW	\$9,500.00
ALSPFLXA	\$6,400.00	CHTHFLXA	\$7,600.00	EORNFLMA	\$5,900.00	GVLDFLXA	\$10,900.00
ALTRFLXA	\$1,600.00	CITRFLXA	\$5,500.00	ESTSFLXA	\$9,400.00	HAVNFLMA	\$7,300.00
ALVAFLXA	\$6,900.00	CLHNFLXA	\$5,900.00	ETVLFLAA	\$3,500.00	HBSDFLMA	\$5,500.00
ANMRFLXA	\$1,900.00	CLMTFLXA	\$11,100.00	EVRGFLXA	\$15,600.00	HDSNFLXA	\$2,100.00
APPKFLXA	\$8,400.00	CLTNFLXA	\$16,900.00	FHSDFLXA	\$1,600.00	HGLDFLXA	\$1,600.00
ARCDFLXA	\$6,700.00	CLWRFLXA	\$1,600.00	FLBHFLMA	\$5,900.00	HGSPFLXA	\$10,000.00
ARCHFLMA	\$5,600.00	CNSDFLXA	\$1,600.00	FLRHFLXA	\$7,000.00	HLNVFLMA	\$4,700.00
ASTRFLXA	\$9,400.00	CNTMFLLE	\$5,300.00	FRBHFLFP	\$6,300.00	HLRDFLXA	\$7,200.00
AVPKFLXA	\$3,100.00	COCOFLMA	\$3,400.00	FRPTFLXA	\$8,500.00	HLWDFLED	\$4,900.00
BAKRFLXA	\$8,400.00	COCOFLME	\$4,700.00	FRSTFLXA	\$1,900.00	HLWDFLHA	\$4,800.00
BARTFLXA	\$1,600.00	COCYFL13	\$2,400.00	FTGRFLMA	\$5,600.00	HLWDFLMA	\$4,900.00
BAYUFLXA	\$1,600.00	CPCRFLXA	\$5,700.00	FTLDFLCR	\$8,500.00	HLWDFLPE	\$9,000.00
BBPKFLXA	\$1,800.00	CPCRFLXB	\$5,500.00	FTLDFLCY	\$9,000.00	HLWDFLWH	\$9,500.00
BCGRFLXA	\$10,500.00	CPHZFLXA	\$11,500.00	FTLDFLJA	\$4,900.00	HMSPFLXA	\$9,700.00
BCRTFLBT	\$9,000.00	CRCYFLXA	\$6,800.00	FTLDFLMR	\$4,400.00	HMSTFLEA	\$6,600.00
BCRTFLMA	\$3,400.00	CRRVFLXA	\$9,300.00	FTLDFLOA	\$3,500.00	HMSTFLHM	\$6,600.00
BCRTFLSA	\$5,100.00	CRVWFLXA	\$6,000.00	FTLDFLPL	\$8,500.00	HMSTFLNA	\$6,200.00
BGPIFLMA	\$11,900.00	CRWDFLXA	\$1,700.00	FTLDFLSG	\$5,200.00	HNCYFLXA	\$1,600.00
BHPKFLXA	\$1,600.00	CSCYFLBA	\$8,000.00	FTLDFLSU	\$4,700.00	HNCYFLXN	\$1,600.00
BKVLFLJF	\$7,700.00	CSLBFLXA	\$7,500.00	FTLDFLTA	\$4,500.00	HOWYFLXA	\$9,900.00
BLDWFLMA	\$5,900.00	CTDLFLXA	\$9,700.00	FTLDFLWN	\$5,700.00	HSNGFLXA	\$5,500.00
BLGLFLMA	\$7,400.00	CYGRFLXA	\$1,600.00	FTMBFLXA	\$8,400.00	HTISFLMA	\$4,900.00
BLTWFLXA	\$4,200.00	CYLKFLXA	\$5,900.00	FTMDFLXA	\$4,700.00	HWTHFLMA	\$5,700.00
BLVWFLXA	\$6,000.00	CYLKFLXB	\$5,900.00	FTMYFLMA	\$3,900.00	HYPKFLXA	\$1,600.00
BNFYFLXA	\$8,900.00	DBRYFLDL	\$5,300.00	FTMYFLXA	\$4,500.00	IMKLFLXA	\$11,700.00
BNNLFLMA	\$6,100.00	DBRYFLMA	\$5,200.00	FTMYFLXB	\$5,200.00	INLKFLXA	\$2,100.00
BNSPFLXA	\$9,800.00	DDCYFLXA	\$12,200.00	FTMYFLXC	\$5,400.00	INRKFLXX	\$1,600.00
BORAFLXA	\$10,800.00	DELDFLMA	\$6,100.00	FTPRFLMA	\$3,400.00	INTRFLXA	\$7,100.00
BRBAFLXA	\$2,000.00	DESTFLXA	\$5,500.00	FTWBFLXA	\$3,800.00	INTWFLXA	\$5,600.00
BRFRFLXA	\$10,900.00	DFSPFLXA	\$9,400.00	FTWBFLXB	\$5,100.00	INVRFLXA	\$8,200.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
    - 1.4.3. Rates and Charges: (Cont'd)
      - 1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)
        - 1.4.3.1.3 <u>Verizon Business Services I:</u> The following monthly recurring per-circuit charges apply based on Serving Wire Center of the circuit:

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
BRJTFLXA	\$1,800.00	DLBHFLKP	\$5,200.00	FTWBFLXC	\$5,900.00	IONAFLXA	\$6,300.00
BRKRFLXA	\$5,900.00	DLBHFLMA	\$5,000.00	FTWHFLXA	\$10,000.00	ISLMFLMA	\$9,200.00
BRNDFLXA	\$1,900.00	DLSPFLMA	\$6,000.00	GCSPFLCN	\$5,700.00	JAY FLMA	\$22,600.00
BRSNFLMA	\$6,200.00	DNDNFLXA	\$1,600.00	GCVLFLMA	\$13,200.00	JCBHFLAB	\$5,300.00
BRTNFLXX	\$2,000.00	DNLNFLWM	\$5,800.00	GDRGFLXA	\$7,300.00	JCBHFLMA	\$5,500.00
BSHNFLXA	\$9,400.00	DRBHFLDF	\$4,700.00	GENVFLMA	\$5,700.00	JCBHFLSP	\$5,200.00
BVHLFLXA	\$8,200.00	DRBHFLMA	\$9,000.00	GLBRFLMC	\$5,300.00	JCVLFLAI	\$4,800.00
BWLGFLXA	\$7,300.00	DUNDFLXA	\$1,600.00	GLDLFLXA	\$10,300.00	JCVLFLAR	\$9,500.00
BYBHFLMA	\$5,500.00	DWPKFLXA	\$12,400.00	GLGCFLXA	\$11,900.00	JCVLFLBW	\$4,900.00
BYSHFLXA	\$2,000.00	DYBHFLFN	\$8,500.00	GLRDFLXA	\$6,600.00	JCVLFLCL	\$4,400.00
CCBHFLAF	\$5,000.00	DYBHFLMA	\$4,400.00	GNBOFLXA	\$3,400.00	JCVLFLFC	\$9,500.00
CCBHFLMA	\$5,000.00	DYBHFLOB	\$9,000.00	GNDYFLXA	\$1,600.00	JCVLFLGH	\$4,900.00
CDKYFLMA	\$8,500.00	DYBHFLOS	\$5,200.00	GNVLFLXA	\$13,200.00	JCVLFLIA	\$5,300.00
CFLDFLMA	\$7,100.00	DYBHFLPO	\$9,500.00	GNWDFLXA	\$8,800.00	JCVLFLJT	\$4,700.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.3 <u>Verizon Business Services I</u>: (Cont'd)

Wire Center	Rate						
JCVLFLLF	\$9,000.00	LYHNFLOH	\$4,400.00	MRHNFLXA	\$15,400.00	ORLEFLGV	\$4,900.00
JCVLFLNO	\$9,500.00	MALNFLXA	\$9,600.00	MRNNFLXA	\$5,800.00	ORLFFLEJ	\$3,500.00
JCVLFLOG	\$3,500.00	MAYOFLXA	\$16,900.00	MRTHFLVE	\$10,800.00	ORLFFLQR	\$3,500.00
JCVLFLOW	\$5,200.00	MCINFLXA	\$5,600.00	MTDRFLXA	\$9,900.00	ORPKFLMA	\$5,200.00
JCVLFLRV	\$8,500.00	MCLNFLXA	\$6,500.00	MTLDFLAP	\$4,900.00	ORPKFLRW	\$5,500.00
JCVLFLSJ	\$9,500.00	MCNPFLMA	\$5,300.00	MTLDFLXA	\$5,700.00	ORSPFLXA	\$8,000.00
JCVLFLSM	\$7,400.00	MDBGFLPM	\$6,000.00	MTVRFLXA	\$10,800.00	OSPRFLXA	\$2,500.00
JCVLFLWC	\$5,000.00	MDSNFLXA	\$15,700.00	MXVLFLMA	\$6,100.00	OVIDFLCA	\$9,500.00
JCVMFLLI	\$4,800.00	MIAMFLAE	\$6,900.00	MYCYFLXA	\$2,500.00	PACEFLPV	\$5,200.00
JCVMFLUF	\$6,100.00	MIAMFLAL	\$6,400.00	NDADFLAA	\$4,700.00	PAHKFLMA	\$7,200.00
JNGSFLXA	\$11,800.00	MIAMFLAP	\$6,900.00	NDADFLAC	\$4,700.00	PANCFLXA	\$11,300.00
JPTRFLMA	\$5,700.00	MIAMFLBA	\$5,800.00	NDADFLBR	\$4,400.00	PCBHFLNT	\$5,400.00
JSPRFLXA	\$10,500.00	MIAMFLBC	\$5,300.00	NDADFLGG	\$3,400.00	PKCYFLXA	\$1,600.00
KGLKFLXA	\$6,400.00	MIAMFLBR	\$6,400.00	NDADFLOL	\$4,700.00	PLCSFLMA	\$6,600.00
KNVLFLXA	\$7,000.00	MIAMFLCA	\$6,900.00	NFMYFLXA	\$2,700.00	PLMTFLXA	\$1,900.00
KSSMFLXA	\$4,700.00	MIAMFLDB	\$8,500.00	NFMYFLXB	\$6,500.00	PLSLFLXA	\$2,000.00
KSSMFLXB	\$9,400.00	MIAMFLFL	\$5,800.00	NGBHFLXA	\$1,600.00	PLTKFLMA	\$6,500.00
KSSMFLXD	\$9,700.00	MIAMFLGR	\$2,300.00	NKLRFLMA	\$7,000.00	PMBHFLCS	\$5,000.00
KYHGFLMA	\$6,000.00	MIAMFLHL	\$7,400.00	NNPLFLXA	\$10,600.00	PMBHFLED	\$5,200.00
KYLRFLLS	\$8,000.00	MIAMFLIC	\$8,500.00	NPLSFLXC	\$12,900.00	PMBHFLFE	\$4,400.00
KYLRFLMA	\$8,400.00	MIAMFLKE	\$7,900.00	NPLSFLXD	\$12,000.00	PMBHFLMA	\$9,500.00
KYSTFLXA	\$1,700.00	MIAMFLME	\$5,300.00	NPRCFLXA	\$1,900.00	PMBHFLNP	\$9,500.00
KYWSFLMA	\$13,600.00	MIAMFLNM	\$9,000.00	NRPTFLXA	\$2,800.00	PMBHFLTA	\$4,800.00
LBLLFLXA	\$11,100.00	MIAMFLNS	\$7,400.00	NRSDFLXA	\$2,200.00	PMPKFLMA	\$7,000.00
LDLKFLXA	\$5,800.00	MIAMFLOL	\$8,000.00	NSBHFLMA	\$5,600.00	PNCRFLXA	\$1,900.00
LEE FLXA	\$16,900.00	MIAMFLPB	\$6,900.00	NWBYFLMA	\$5,800.00	PNCYFLCA	\$4,800.00
LGBKFLXA	\$2,100.00	MIAMFLPL	\$2,300.00	OCALFLXA	\$2,700.00	PNCYFLMA	\$3,500.00
LHACFLXA	\$6,700.00	MIAMFLRR	\$8,500.00	OCALFLXB	\$5,100.00	PNGRFLXA	\$10,100.00
LKALFLXA	\$1,600.00	MIAMFLSH	\$7,400.00	OCALFLXC	\$5,100.00	PNISFLXA	\$6,800.00
LKBNFLXB	\$5,700.00	MIAMFLSO	\$9,000.00	OCALFLXJ	\$5,500.00	PNLNFLXA	\$10,600.00
LKBRFLXA	\$7,300.00	MIAMFLWD	\$10,000.00	OCNFFLXA	\$6,700.00	PNLSFLXA	\$1,600.00
LKBTFLXA	\$8,200.00	MIAMFLWM	\$6,400.00	OJUSFLTL	\$4,700.00	PNSCFLBL	\$4,900.00
LKCYFLMA	\$8,800.00	MIANFLYI	\$4,700.00	OKCBFLXA	\$11,100.00	PNSCFLFP	\$4,400.00
LKHLFLXA	\$5,100.00	MIAQFL06	\$4,700.00	OKHLFLMA	\$6,600.00	PNSCFLHC	\$9,500.00
LKLDFLXA	\$1,600.00	MIATFLAD	\$4,900.00	OKLWFLXA	\$6,800.00	PNSCFLPB	\$5,800.00
LKLDFLXE	\$1,600.00	MICCFLBB	\$5,900.00	OLDSFLXA	\$1,600.00	PNSCFLWA	\$5,000.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.3 <u>Verizon Business Services I</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
LKLDFLXN	\$1,600.00	MLBRFLMA	\$4,400.00	OLTWFLLN	\$7,400.00	PNVDFLMA	\$5,300.00
LKMRFLMA	\$3,400.00	MLBYFLXA	\$1,700.00	ORCYFLXA	\$4,700.00	POINFLXA	\$1,800.00
LKPCFLXA	\$5,700.00	MLRSFLXA	\$7,000.00	ORCYFLXC	\$4,800.00	PRRNFLMA	\$5,600.00
LKWLFLXA	\$1,700.00	MLTNFLRA	\$5,500.00	ORLDFL60	\$4,500.00	PRRYFLXA	\$14,100.00
LKWLFLXE	\$1,900.00	MNDRFLAV	\$3,400.00	ORLDFLAP	\$7,400.00	PRSHFLXA	\$1,900.00
LLMNFLXA	\$1,600.00	MNDRFLLO	\$4,700.00	ORLDFLCL	\$5,300.00	PRSNFLFD	\$6,600.00
LNLKFLXA	\$2,000.00	MNDRFLLW	\$5,000.00	ORLDFLLU	\$4,500.00	PSDNFLXA	\$1,600.00
LRGOFLXA	\$1,600.00	MNLKFLXA	\$2,000.00	ORLDFLMA	\$2,300.00	PTCTFLXA	\$5,300.00
LRVLFLXA	\$11,500.00	MNSNFLMA	\$6,900.00	ORLDFLPC	\$7,400.00	PTCYFLXA	\$1,800.00
LSBGFLXA	\$8,600.00	MNTIFLXA	\$10,500.00	ORLDFLPH	\$7,400.00	PTSJFLXA	\$7,100.00
LUTZFLXA	\$1,800.00	MOISFLXA	\$15,000.00	ORLDFLSA	\$7,900.00	PTSLFLMA	\$5,300.00
LVOKFLXA	\$9,700.00	MOLNFLXA	\$3,000.00	ORLDFLSO	\$3,500.00	PTSLFLSO	\$4,900.00
LWTYFLXA	\$6,800.00	MRDCFLXA	\$11,500.00	ORLDFLXH	\$4,500.00	QNCYFLXA	\$7,800.00

- 1. SERVICE DESCRIPTION (CONT'D.)
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.3 <u>Verizon Business Services I</u>: (Cont'd)

Wire Center         Rate         Wire Center         Rate         Wire Center         Rate           RAFRFLXA         \$7,600.00         TAFBFLXA         \$4,900.00         WLCRFLXA         \$1,600.00           RSKNFLXA         \$1,700.00         TAMPFL52         \$2,700.00         WLHLFLXA         \$4,400.00           RYHLFLXA         \$14,700.00         TAMPFLCM         \$2,900.00         WLSTFLXA         \$5,900.00           SARKFLXA         \$2,200.00         TAMPFLCO         \$3,000.00         WLWDFLXA         \$7,700.00           SBNGFLXA         \$5,700.00         TAMPFLDM         \$2,700.00         WNDRFLXA         \$7,900.00           SBSTFLFE         \$5,700.00         TAMPFLKO         \$2,700.00         WNGRFLXA         \$8,400.00           SBSTFLMA         \$5,500.00         TAMPFLXA         \$2,400.00         WNHNFLBU         \$2,700.00           SCPKFLXA         \$6,400.00         TAMPFLXE         \$1,800.00         WNHNFLXC         \$1,600.00           SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WNPKFLXA         \$4,700.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SKWYFLXA         \$1,600.00         THNTFLXA
RSKNFLXA         \$1,700.00         TAMPFL52         \$2,700.00         WLHLFLXA         \$4,400.00           RYHLFLXA         \$14,700.00         TAMPFLCM         \$2,900.00         WLSTFLXA         \$5,900.00           SARKFLXA         \$2,200.00         TAMPFLCO         \$3,000.00         WLWDFLXA         \$7,700.00           SBNGFLXA         \$5,700.00         TAMPFLDM         \$2,700.00         WNDRFLXA         \$7,900.00           SBSTFLFE         \$5,700.00         TAMPFLKO         \$2,700.00         WNGRFLXA         \$8,400.00           SBSTFLMA         \$5,500.00         TAMPFLXA         \$2,400.00         WNHNFLBU         \$2,700.00           SCPKFLXA         \$6,400.00         TAMPFLXE         \$1,800.00         WNHNFLXC         \$1,600.00           SEKYFLXA         \$2,300.00         TAMPFLXX         \$1,600.00         WNPKFLXA         \$4,700.00           SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WPBHFLS         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SKWYFLXA         \$1,600.00         TAMSFLCZ         \$3,100.00         WPBHFLGR         \$5,300.00           SLHLFLXA         \$5,300.00         TLCHFLXA
RYHLFLXA         \$14,700.00         TAMPFLCM         \$2,900.00         WLSTFLXA         \$5,900.00           SARKFLXA         \$2,200.00         TAMPFLCO         \$3,000.00         WLWDFLXA         \$7,700.00           SBNGFLXA         \$5,700.00         TAMPFLDM         \$2,700.00         WNDRFLXA         \$7,900.00           SBSTFLFE         \$5,700.00         TAMPFLKO         \$2,700.00         WNGRFLXA         \$8,400.00           SBSTFLMA         \$5,500.00         TAMPFLXA         \$2,400.00         WNHNFLBU         \$2,700.00           SCPKFLXA         \$6,400.00         TAMPFLXE         \$1,800.00         WNHNFLXC         \$1,600.00           SEKYFLXA         \$2,300.00         TAMPFLXX         \$1,600.00         WNPKFLXA         \$4,700.00           SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WPBHFL58         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLHH         \$9,000.00           SLHLFLXA         \$5,300.00         TLCHFLXA </td
SARKFLXA         \$2,200.00         TAMPFLCO         \$3,000.00         WLWDFLXA         \$7,700.00           SBNGFLXA         \$5,700.00         TAMPFLDM         \$2,700.00         WNDRFLXA         \$7,900.00           SBSTFLFE         \$5,700.00         TAMPFLKO         \$2,700.00         WNGRFLXA         \$8,400.00           SBSTFLMA         \$5,500.00         TAMPFLXA         \$2,400.00         WNHNFLBU         \$2,700.00           SCPKFLXA         \$6,400.00         TAMPFLXE         \$1,800.00         WNHNFLXC         \$1,600.00           SEKYFLXA         \$2,300.00         TAMPFLXX         \$1,600.00         WNPKFLXA         \$4,700.00           SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WPBHFL58         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SBNGFLXA         \$5,700.00         TAMPFLDM         \$2,700.00         WNDRFLXA         \$7,900.00           SBSTFLFE         \$5,700.00         TAMPFLKO         \$2,700.00         WNGRFLXA         \$8,400.00           SBSTFLMA         \$5,500.00         TAMPFLXA         \$2,400.00         WNHNFLBU         \$2,700.00           SCPKFLXA         \$6,400.00         TAMPFLXE         \$1,800.00         WNPKFLXA         \$4,700.00           SEKYFLXA         \$2,300.00         TAMPFLXX         \$1,600.00         WNPKFLXA         \$4,700.00           SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WNPKFLXE         \$1,700.00           SGBHFLXA         \$8,600.00         TAMQFLPM         \$2,800.00         WPBHFL58         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLHH         \$9,000.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$1,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA
SBSTFLFE         \$5,700.00         TAMPFLKO         \$2,700.00         WNGRFLXA         \$8,400.00           SBSTFLMA         \$5,500.00         TAMPFLXA         \$2,400.00         WNHNFLBU         \$2,700.00           SCPKFLXA         \$6,400.00         TAMPFLXE         \$1,800.00         WNHNFLXC         \$1,600.00           SEKYFLXA         \$2,300.00         TAMPFLXX         \$1,600.00         WNPKFLXA         \$4,700.00           SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WNPKFLXE         \$1,700.00           SGBHFLXA         \$8,600.00         TAMQFLPM         \$2,800.00         WPBHFL58         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLHH         \$9,000.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$1,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SBSTFLMA         \$5,500.00         TAMPFLXA         \$2,400.00         WNHNFLBU         \$2,700.00           SCPKFLXA         \$6,400.00         TAMPFLXE         \$1,800.00         WNHNFLXC         \$1,600.00           SEKYFLXA         \$2,300.00         TAMPFLXX         \$1,600.00         WNPKFLXA         \$4,700.00           SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WNPKFLXE         \$1,700.00           SGBHFLXA         \$8,600.00         TAMQFLPM         \$2,800.00         WPBHFL58         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLGR         \$5,300.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SCPKFLXA         \$6,400.00         TAMPFLXE         \$1,800.00         WNHNFLXC         \$1,600.00           SEKYFLXA         \$2,300.00         TAMPFLXX         \$1,600.00         WNPKFLXA         \$4,700.00           SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WNPKFLXE         \$1,700.00           SGBHFLXA         \$8,600.00         TAMQFLPM         \$2,800.00         WPBHFL58         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLGR         \$5,300.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SEKYFLXA         \$2,300.00         TAMPFLXX         \$1,600.00         WNPKFLXA         \$4,700.00           SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WNPKFLXE         \$1,700.00           SGBHFLXA         \$8,600.00         TAMQFLPM         \$2,800.00         WPBHFL58         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLGR         \$5,300.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SGBEFLXA         \$1,600.00         TAMQFLHU         \$2,900.00         WNPKFLXE         \$1,700.00           SGBHFLXA         \$8,600.00         TAMQFLPM         \$2,800.00         WPBHFL58         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLGR         \$5,300.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SGBHFLXA         \$8,600.00         TAMQFLPM         \$2,800.00         WPBHFL58         \$3,400.00           SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLGR         \$5,300.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SGKYFLMA         \$12,400.00         TAMQFLTA         \$2,800.00         WPBHFLAN         \$4,400.00           SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLGR         \$5,300.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLGR         \$5,300.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SHLMFLXA         \$5,100.00         TAMSFLCZ         \$3,100.00         WPBHFLGA         \$5,000.00           SKWYFLXA         \$1,600.00         THNTFLXA         \$1,900.00         WPBHFLGR         \$5,300.00           SLHLFLXA         \$5,300.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SLHLFLXA         \$5,300.00         TLCHFLXA         \$11,300.00         WPBHFLHH         \$9,000.00           SLSPFLXA         \$1,700.00         TLHSFLXA         \$4,500.00         WPBHFLLE         \$4,900.00
SLSPFLXA \$1,700.00 TLHSFLXA \$4,500.00 WPBHFLLE \$4,900.00
SMNI FLYA \$1 700.00 TLHSELYR \$3 700.00 W/PRHELPR \$0 500.00
+ $0$ $0$ $0$ $0$ $0$ $0$ $0$ $0$ $0$ $0$
SNANFLXA \$12,500.00 TLHSFLXC \$5,100.00 WPBHFLRP \$5,700.00
SNDSFLXA \$7,300.00 TLHSFLXD \$3,700.00 WSSDFLXA \$1,600.00
SNFRFLMA \$4,900.00   TLHSFLXE   \$1,600.00   WSTVFLXA   \$12,500.00
SNISFLXA \$9,600.00 TLHSFLXF \$5,600.00 WWSPFLHI \$8,400.00
SNRSFLXA \$8,000.00 TLHSFLXG \$6,000.00 WWSPFLSH \$8,600.00
SNSNFLXA \$7,700.00 TLHSFLXH \$5,500.00 YBCTFLXA \$1,700.00
SNSPFLXA \$1,800.00 TMTRFLXA \$1,800.00 YNFNFLMA \$5,800.00
SPBGFLXA \$1,600.00   TRENFLMA   \$6,800.00   YNTWFLMA   \$6,700.00
SPBGFLXS         \$1,600.00         TRSPFLXA         \$1,700.00         YULEFLMA         \$6,000.00
SPCPFLXA         \$11,300.00         TTVLFLMA         \$5,800.00         ZLSPFLXA         \$7,300.00
SPRGFLXA \$2,300.00 TVRSFLXA \$9,400.00 ZPHYFLXA \$1,900.00
SRSTFLXA \$2,200.00 UMTLFLXA \$9,000.00
SSDSFLXA \$2,300.00 UNVRFLXA \$1,800.00
SSPRFLXA \$8,400.00 VENCFLXA \$2,600.00
STAGFLBS \$4,900.00 VENCFLXS \$2,700.00
STAGFLMA \$3,400.00 VERNFLMA \$6,600.00
STAGFLSH \$4,900.00 VLPRFLXA \$4,100.00
STAGFLWG \$5,300.00 VLPRFLXB \$6,400.00
STCDFLXA \$5,900.00 VRBHFLBE \$4,700.00
STGRFLXA \$1,600.00 VRBHFLMA \$3,400.00
STMKFLXA \$9,600.00 WALDFLXA \$5,300.00
STRKFLXA \$7,300.00 WCHLFLXA \$7,300.00
STRTFLMA \$3,400.00 WELKFLMA \$7,200.00
SVSPFLXA \$5,300.00 WHSPFLXA \$8,800.00
SVSSFLXA \$6,100.00 WIMMFLXA \$1,900.00
0M/THELVA #4 000 00 M/I DDELVA #6 000 00
SWTHFLXA   \$1,600.00   WLBRFLXA   \$9,200.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
    - 1.4.3. Rates and Charges: (Cont'd)
      - 1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.4 Pre-VBS Services: The following monthly recurring per-circuit charges apply based on Serving Wire Center of the circuit:

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
ABDLFLXA	\$3,600.00	BYBHFLAK	\$6,610.00	DRBHFLMA	\$5,260.00	FTWHFLXA	\$13,580.00
ALCHFLXA	\$12,860.00	BYBHFLMA	\$6,610.00	DUNDFLXA	\$6,370.00	GCSPFLCN	\$7,090.00
ALFAFLXA	\$6,130.00	BYSHFLXA	\$5,860.00	DWPKFLXA	\$16,700.00	GCVLFLMA	\$10,820.00
ALFRFLXA	\$8,540.00	CCBHFLAF	\$5,860.00	DYBHFLFN	\$5,380.00	GDRGFLXA	\$7,450.00
ALSPFLXA	\$5,740.00	CCBHFLMA	\$5,860.00	DYBHFLMA	\$3,600.00	GENVFLMA	\$6,970.00
ALTHFLXA	\$6,370.00	CDKYFLMA	\$11,900.00	DYBHFLOB	\$5,500.00	GLBRFLMC	\$6,370.00
ALTRFLXA	\$6,730.00	CFLDFLMA	\$9,500.00	DYBHFLOS	\$6,130.00	GLDLFLXA	\$10,460.00
ALVAFLXA	\$7,090.00	CFVLFLXA	\$7,450.00	DYBHFLPO	\$5,620.00	GLGCFLXA	\$9,260.00
ANMRFLXA	\$6,970.00	CHLKFLXA	\$11,540.00	DYBHFLWS	\$3,600.00	GLRDFLXA	\$5,860.00
APLCFLXA	\$11,300.00	CHPLFLJA	\$9,740.00	EGLLFLBG	\$5,380.00	GNBOFLXA	\$7,450.00
APPKFLXA	\$6,490.00	CHSWFLXA	\$10,220.00	EGLLFLIH	\$5,500.00	GNDYFLXA	\$5,620.00
ARCDFLXA	\$9,620.00	CHTHFLXA	\$7,690.00	ENWDFLXA	\$8,780.00	GNVLFLXA	\$10,100.00
ARCHFLMA	\$6,850.00	CITRFLXA	\$6,970.00	EORNFLMA	\$7,330.00	GNWDFLXA	\$9,020.00
ARPNFLXA	\$11,900.00	CLHNFLXA	\$7,330.00	ESPNFLXA	\$11,300.00	GRETFLXA	\$8,540.00
ASTRFLXA	\$9,620.00	CLMTFLXA	\$11,180.00	ESTSFLXA	\$9,620.00	GSVLFLAN	\$3,600.00
AVPKFLXA	\$3,600.00	CLTNFLXA	\$12,380.00	ETVLFLAA	\$7,390.00	GSVLFLMA	\$3,600.00
BAKRFLXA	\$8,660.00	CLWRFLXA	\$3,600.00	EVRGFLXA	\$12,740.00	GSVLFLNW	\$5,620.00
BARTFLXA	\$6,370.00	CNSDFLXA	\$5,500.00	FHSDFLXA	\$6,130.00	GSVLFLSS	\$5,020.00
BAYUFLXA	\$5,740.00	CNTMFLLE	\$9,256.06	FLBHFLMA	\$7,330.00	GVLDFLXA	\$10,940.00
BBPKFLXA	\$7,810.00	COCOFLMA	\$3,600.00	FLRHFLXA	\$9,740.00	HAVNFLMA	\$10,100.00
BCGRFLXA	\$8,420.00	COCOFLME	\$5,260.00	FRBHFLFP	\$8,050.00	HBSDFLMA	\$6,610.00
BCRTFLBB	\$5,500.00	CPCRFLXA	\$5,860.00	FRPTFLXA	\$8,780.00	HDSNFLXA	\$8,660.00
BCRTFLBT	\$5,500.00	CPCRFLXB	\$5,620.00	FRSTFLXA	\$8,540.00	HGLDFLXA	\$5,620.00
BCRTFLMA	\$3,600.00	CPHZFLXA	\$9,020.00	FTGRFLMA	\$6,850.00	HGSPFLXA	\$13,340.00
BCRTFLSA	\$5,740.00	CRBLFLXA	\$10,940.00	FTLDFLCR	\$5,380.00	HLNVFLMA	\$7,090.00
BGPIFLMA	\$18,260.00	CRCYFLXA	\$9,500.00	FTLDFLCY	\$5,500.00	HLRDFLXA	\$8,780.00
BHPKFLXA	\$5,500.00	CRRVFLXA	\$9,500.00	FTLDFLJA	\$5,620.00	HLWDFLED	\$5,620.00
BKVLFLJF	\$10,940.00	CRVWFLXA	\$8,050.00	FTLDFLMR	\$3,600.00	HLWDFLHA	\$5,500.00
BLDWFLMA	\$7,330.00	CRWDFLXA	\$6,250.00	FTLDFLOA	\$3,600.00	HLWDFLMA	\$5,620.00
BLGLFLMA	\$9,980.00	CSCYFLBA	\$11,060.00	FTLDFLOV	\$5,140.00	HLWDFLPE	\$5,740.00
BLTWFLXA	\$3,600.00	CSLBFLXA	\$5,860.00	FTLDFLPL	\$5,260.00	HLWDFLWH	\$5,500.00
BLVWFLXA	\$6,250.00	CTDLFLXA	\$9,140.00	FTLDFLSF	\$6,970.00	HMSPFLXA	\$9,860.00
BNFYFLXA	\$9,860.00	CYGRFLXA	\$6,130.00	FTLDFLSG	\$6,130.00	HMSTFLEA	\$8,660.00
BNNLFLMA	\$7,690.00	CYLKFLXA	\$6,130.00	FTLDFLSU	\$5,380.00	HMSTFLHM	\$8,660.00
BNSPFLXA	\$7,690.00	CYLKFLXB	\$6,130.00	FTLDFLTA	\$3,600.00	HMSTFLNA	\$7,930.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.4 <u>Pre-VBS Services</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
BORAFLXA	\$15,140.00	DBRYFLDL	\$6,370.00	FTLDFLTN	\$5,020.00	HMSTFLTE	\$8,780.00
BRBAFLXA	\$5,860.00	DBRYFLMA	\$6,130.00	FTLDFLWN	\$6,490.00	HNCYFLXA	\$6,370.00
BRFRFLXA	\$14,900.00	DDCYFLXA	\$12,140.00	FTMBFLXA	\$6,850.00	HNCYFLXN	\$6,730.00
BRJTFLXA	\$7,330.00	DELDFLMA	\$7,690.00	FTMDFLXA	\$7,570.00	HOWYFLXA	\$10,100.00
BRKRFLXA	\$7,090.00	DESTFLXA	\$5,620.00	FTMYFLMA	\$5,620.00	HSFRFLXA	\$6,970.00
BRNDFLXA	\$6,130.00	DFSPFLXA	\$9,620.00	FTMYFLXA	\$3,600.00	HSNGFLXA	\$7,090.00
BRSNFLMA	\$7,930.00	DLBHFLKP	\$6,130.00	FTMYFLXB	\$5,380.00	HTISFLMA	\$5,620.00
BRSTFLXA	\$5,380.00	DLBHFLMA	\$6,130.00	FTMYFLXC	\$5,260.00	HWTHFLMA	\$6,970.00
BRTNFLXX	\$6,370.00	DLSPFLMA	\$7,570.00	FTPRFLMA	\$14,540.00	HYPKFLXA	\$5,020.00
BSHNFLXA	\$9,620.00	DNDNFLXA	\$5,500.00	FTWBFLXA	\$3,600.00	IMKLFLXA	\$9,140.00
BVHLFLXA	\$3,600.00	DNLNFLWM	\$7,810.00	FTWBFLXB	\$5,260.00	INLKFLXA	\$9,260.00
BWLGFLXA	\$7,450.00	DRBHFLDF	\$5,380.00	FTWBFLXC	\$6,130.00	INRKFLXX	\$5,740.00

- 1. SERVICE DESCRIPTION (CONT'D.)
- 1.4 Intrastate DS-3 Local Access (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

Pre-VBS Services: (Cont'd) 1.4.3.1.4

Wire Center	Rate						
INTRFLXA	\$10,100.00	LHACFLXA	\$6,850.00	MIAMFLIC	\$5,860.00	NDADFLOL	\$5,260.00
INTWFLXA	\$7,330.00	LKALFLXA	\$5,380.00	MIAMFLKE	\$6,130.00	NFMYFLXA	\$5,260.00
INVRFLXA	\$8,540.00	LKBNFLXB	\$6,850.00	MIAMFLME	\$3,600.00	NFMYFLXB	\$5,500.00
IONAFLXA	\$6,490.00	LKBRFLXA	\$5,740.00	MIAMFLNM	\$5,380.00	NGBHFLXA	\$6,130.00
ISLMFLMA	\$13,220.00	LKBTFLXA	\$10,820.00	MIAMFLNS	\$5,380.00	NKLRFLMA	\$9,380.00
JAY FLMA	\$9,020.00	LKCYFLMA	\$12,500.00	MIAMFLOL	\$5,500.00	NNPLFLXA	\$8,540.00
JCBHFLAB	\$6,370.00	LKHLFLXA	\$7,090.00	MIAMFLPB	\$5,500.00	NPLSFLXC	\$9,860.00
JCBHFLMA	\$6,610.00	LKLDFLXA	\$5,500.00	MIAMFLPL	\$6,130.00	NPLSFLXD	\$9,500.00
JCBHFLSP	\$6,130.00	LKLDFLXE	\$11,900.00	MIAMFLRR	\$5,860.00	NPRCFLXA	\$7,810.00
JCVLFLAI	\$5,260.00	LKLDFLXN	\$5,860.00	MIAMFLSH	\$5,500.00	NRPTFLXA	\$8,540.00
JCVLFLAR	\$5,620.00	LKMRFL01	\$3,600.00	MIAMFLSO	\$6,610.00	NRSDFLXA	\$5,260.00
JCVLFLBW	\$5,740.00	LKMRFLHE	\$5,140.00	MIAMFLWD	\$6,970.00	NSBHFLMA	\$6,850.00
JCVLFLCL	\$3,600.00	LKMRFLMA	\$3,600.00	MIAMFLWM	\$5,620.00	NWBYFLMA	\$7,210.00
JCVLFLFC	\$5,620.00	LKMYFLMA	\$5,740.00	MIAMFLYJ	\$7,090.00	OCALFLDB	\$3,600.00
JCVLFLGH	\$5,620.00	LKPCFLXA	\$7,930.00	MIANFLPV	\$5,860.00	OCALFLXA	\$3,600.00
JCVLFLIA	\$6,370.00	LKWLFLXA	\$7,090.00	MIANFLWK	\$5,860.00	OCALFLXB	\$5,260.00
JCVLFLJB	\$5,380.00	LKWLFLXE	\$8,420.00	MIANFLYI	\$5,260.00	OCALFLXC	\$5,260.00
JCVLFLJT	\$5,380.00	LKWOFLIY	\$5,620.00	MIAQFL06	\$5,260.00	OCALFLXJ	\$5,620.00
JCVLFLLF	\$5,500.00	LLMNFLXA	\$5,380.00	MIATFLAD	\$5,380.00	OCNFFLXA	\$6,850.00
JCVLFLNO	\$5,620.00	LNLKFLXA	\$7,810.00	MICCFLBB	\$7,330.00	OJUSFLTL	\$5,380.00
JCVLFLOG	\$3,600.00	LRGOFLXA	\$5,380.00	MLBRFLMA	\$12,020.00	OKCBFLXA	\$11,180.00
JCVLFLOW	\$6,130.00	LRHLFLXA	\$9,980.00	MLBYFLXA	\$6,490.00	OKHLFLMA	\$8,540.00
JCVLFLRV	\$5,380.00	LRVLFLXA	\$15,860.00	MLRSFLXA	\$7,210.00	OKLWFLXA	\$6,970.00
JCVLFLSJ	\$5,620.00	LSBGFLXA	\$8,900.00	MLTNFLRA	\$6,730.00	OLDSFLXA	\$6,370.00
JCVLFLSM	\$5,140.00	LUTZFLXA	\$6,730.00	MNDRFLAV	\$3,600.00	OLTWFLLN	\$10,100.00
JCVLFLWC	\$5,860.00	LVOKFLXA	\$14,900.00	MNDRFLLO	\$5,380.00	ORCYFLXA	\$6,730.00
JCVLFLWF	\$5,020.00	LWTYFLXA	\$9,140.00	MNDRFLLW	\$5,860.00	ORCYFLXC	\$6,970.00
JCVMFLED	\$5,860.00	LYHNFLOH	\$3,600.00	MNLKFLXA	\$8,660.00	ORLDFL60	\$3,600.00
JCVMFLLI	\$5,500.00	MALNFLXA	\$9,740.00	MNSNFLMA	\$9,140.00	ORLDFLAP	\$5,620.00
JCVMFLLR	\$5,020.00	MAYOFLXA	\$16,460.00	MNTIFLXA	\$8,420.00	ORLDFLCL	\$5,140.00
JCVMFLUF	\$5,740.00	MCINFLXA	\$7,090.00	MOISFLXA	\$11,180.00	ORLDFLLU	\$6,250.00
JNGSFLXA	\$15,860.00	MCLNFLXA	\$8,660.00	MOLNFLXA	\$3,590.00	ORLDFLMA	\$3,600.00
JPTRFLGN	\$3,650.00	MCNPFLMA	\$6,370.00	MRDCFLXA	\$9,020.00	ORLDFLOE	\$3,600.00
JPTRFLMA	\$7,090.00	MDBGFLPM	\$7,570.00	MRHNFLXA	\$11,420.00	ORLDFLPC	\$5,620.00
JSPRFLXA	\$14,660.00	MDSNFLXA	\$11,660.00	MRNNFLXA	\$8,420.00	ORLDFLPH	\$5,620.00
KGLKFLXA	\$9,020.00	MIAMFLAC	\$5,140.00	MRTHFLVE	\$16,100.00	ORLDFLSA	\$5,740.00
KNVLFLXA	\$8,780.00	MIAMFLAE	\$5,380.00	MTDRFLXA	\$10,100.00	ORLDFLXH	\$5,020.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.4 <u>Pre-VBS Services</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
KSSMFLXA	\$7,210.00	MIAMFLAL	\$5,140.00	MTLDFLAP	\$5,620.00	ORLEFLGV	\$5,620.00
KSSMFLXB	\$6,970.00	MIAMFLAP	\$5,380.00	MTLDFLBR	\$5,620.00	ORPKFLMA	\$6,130.00
KSSMFLXD	\$6,970.00	MIAMFLBA	\$5,260.00	MTLDFLTC	\$5,740.00	ORPKFLRW	\$6,610.00
KYHGFLMA	\$7,570.00	MIAMFLBC	\$5,140.00	MTLDFLXA	\$5,740.00	ORSPFLXA	\$8,050.00
KYLRFLLS	\$11,180.00	MIAMFLBR	\$5,500.00	MTVRFLXA	\$7,330.00	OSPRFLXA	\$6,370.00
KYLRFLMA	\$11,900.00	MIAMFLCA	\$6,250.00	MXVLFLMA	\$7,810.00	OVIDFLCA	\$6,250.00
KYSTFLXA	\$6,730.00	MIAMFLDA	\$5,260.00	MYCYFLXA	\$7,930.00	PACEFLPV	\$6,130.00
KYWSFLMA	\$21,260.00	MIAMFLDB	\$5,860.00	NDADFLAA	\$5,260.00	PAHKFLMA	\$9,740.00
LBLLFLXA	\$8,780.00	MIAMFLFL	\$5,140.00	NDADFLAC	\$5,260.00	PANCFLXA	\$8,900.00
LDLKFLXA	\$7,810.00	MIAMFLGR	\$5,140.00	NDADFLBR	\$5,380.00	PBGRFLEZ	\$6,250.00
LGBKFLXA	\$5,860.00	MIAMFLHL	\$6,130.00	NDADFLGG	\$3,600.00	PCBHFLNT	\$6,490.00

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.4 <u>Pre-VBS Services</u>: (Cont'd)

Wire	Doto	Wine Contain	Dete	Wine Center	Dete	Wine Conten	Dete
Center	Rate	Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
PCBHFLNT	\$6,490.00	SBNGFLXA	\$6,250.00	TAMPFLXA	\$10,340.00	WNPKFLXA	\$5,380.00
PKCYFLXA	\$6,130.00	SBSTFLFE	\$6,970.00	TAMPFLXE	\$5,620.00	WNPKFLXE	\$5,500.00
PLBYFLAO	\$5,260.00	SBSTFLMA	\$6,610.00	TAMPFLXX	\$3,600.00	WPBHFL58	\$3,600.00
PLCSFLMA	\$8,540.00	SCPKFLXA	\$6,610.00	TAMQFLHU	\$7,090.00	WPBHFLAN	\$3,600.00
PLMTFLXA	\$6,970.00	SEKYFLXA	\$5,500.00	TAMQFLPM	\$6,370.00	WPBHFLGA	\$5,860.00
PLSLFLXA	\$6,610.00	SGBEFLXA	\$5,740.00	TAMQFLRY	\$10,940.00	WPBHFLGR	\$6,250.00
PLTKFLMA	\$8,420.00	SGBHFLXA	\$8,900.00	TAMQFLSZ	\$5,740.00	WPBHFLHH	\$5,500.00
PMBHFLCS	\$5,860.00	SGKYFLMA	\$19,100.00	TAMQFLTA	\$3,600.00	WPBHFLLE	\$5,740.00
PMBHFLDR	\$5,260.00	SHLMFLXA	\$5,260.00	TAMSFLZK	\$6,130.00	WPBHFLRB	\$5,620.00
PMBHFLED	\$5,140.00	SKWYFLXA	\$5,260.00	THBHFLXA	\$7,570.00	WPBHFLRP	\$6,370.00
PMBHFLFE	\$3,600.00	SLHLFLXA	\$6,970.00	THNTFLXA	\$6,610.00	WSSDFLXA	\$5,260.00
PMBHFLJK	\$5,140.00	SLSPFLXA	\$5,620.00	TLCHFLXA	\$11,300.00	WSTVFLXA	\$9,980.00
PMBHFLMA	\$5,620.00	SMNLFLXA	\$5,260.00	TLHSFLXA	\$3,600.00	WWHTFLXA	\$7,930.00
PMBHFLNP	\$5,620.00	SNANFLXA	\$12,380.00	TLHSFLXB	\$5,260.00	WWSPFLHI	\$11,780.00
PMBHFLTA	\$5,500.00	SNDSFLXA	\$7,450.00	TLHSFLXC	\$5,260.00	WWSPFLSH	\$12,260.00
PMPKFLMA	\$9,260.00	SNFRFLMA	\$5,620.00	TLHSFLXD	\$5,260.00	YBCTFLXA	\$5,140.00
PNCRFLXA	\$7,330.00	SNISFLXA	\$7,570.00	TLHSFLXE	\$5,020.00	YNFNFLMA	\$7,210.00
PNCYFLCA	\$5,500.00	SNRSFLXA	\$7,690.00	TLHSFLXF	\$5,740.00	YNTWFLMA	\$9,620.00
PNCYFLMA	\$3,600.00	SNSNFLXA	\$9,740.00	TLHSFLXG	\$6,250.00	YULEFLMA	\$7,570.00
PNGRFLXA	\$7,930.00	SNSPFLXA	\$7,330.00	TLHSFLXH	\$5,620.00	ZLSPFLXA	\$7,450.00
PNISFLXA	\$6,970.00	SPBGFLXA	\$3,600.00	TMTRFLXA	\$8,660.00	ZPHYFLXA	\$7,690.00
PNLNFLXA	\$10,700.00	SPBGFLXS	\$5,500.00	TRENFLMA	\$8,900.00		
PNLSFLXA	\$5,500.00	SPRGFLXA	\$5,620.00	TRSPFLXA	\$6,610.00		
PNSCFLBL	\$5,740.00	SRSTFLXA	\$3,600.00	TTVLFLMA	\$7,210.00		
PNSCFLFP	\$3,600.00	SSDSFLXA	\$5,500.00	TVRSFLXA	\$9,620.00		
PNSCFLHC	\$5,620.00	SSPRFLXA	\$8,660.00	UMTLFLXA	\$9,260.00		
PNSCFLPB	\$7,210.00	STAGFLBS	\$5,740.00	UNVRFLXA	\$6,130.00		
PNSCFLUP	\$5,740.00	STAGFLMA	\$3,600.00	VENCFLXA	\$7,210.00		
PNSCFLWA	\$5,860.00	STAGFLSH	\$5,620.00	VENCFLXS	\$7,690.00		
PNVDFLMA	\$6,370.00	STAGFLWG	\$6,370.00	VERNFLMA	\$8,540.00		
POINFLXA	\$7,570.00	STCDFLXA	\$7,690.00	VLPRFLXA	\$6,250.00		
PRRNFLCC	\$6,970.00	STGRFLXA	\$5,860.00	VLPRFLXB	\$6,610.00		
PRRNFLMA	\$6,850.00	STMKFLXA	\$7,570.00	VRBHFLBE	\$5,260.00		
PRRYFLXA	\$11,060.00	STRKFLXA	\$9,860.00	VRBHFLMA	\$3,600.00		
PRSHFLXA	\$7,210.00	STRTFLMA	\$3,600.00	WALDFLXA	\$6,730.00		
PRSNFLFD	\$8,660.00	SVSPFLXA	\$5,500.00	WCHLFLXA	\$7,450.00		
PSDNFLXA	\$5,620.00	SVSSFLXA	\$6,370.00	WELKFLMA	\$9,620.00		

- 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
- 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
  - 1.4.3. Rates and Charges: (Cont'd)

1.4.3.1. Local Access Channel Monthly Recurring Channel Charges: (Cont'd)

1.4.3.1.4 <u>Pre-VBS Services</u>: (Cont'd)

Wire Center	Rate	Wire Center	Rate	Wire Center	Rate
PTCTFLXA	\$8,660.00	SWTHFLXA	\$5,860.00	WHSPFLXA	\$13,220.00
PTCYFLXA	\$8,780.00	SYHSFLCC	\$7,570.00	WIMMFLXA	\$7,210.00
PTSJFLXA	\$9,260.00	TAFBFLXA	\$5,860.00	WLBRFLXA	\$13,700.00
PTSLFLMA	\$6,250.00	TAMPFL41	\$5,380.00	WLCHFLXA	\$7,450.00
PTSLFLSO	\$5,620.00	TAMPFL52	\$5,020.00	WLCRFLXA	\$5,500.00
PXTNFLXA	\$10,340.00	TAMPFLCM	\$5,380.00	WLSTFLXA	\$7,450.00
QNCYFLXA	\$8,900.00	TAMPFLCO	\$5,620.00	WLWDFLXA	\$7,810.00
RAFRFLXA	\$9,860.00	TAMPFLDM	\$5,020.00	WNDRFLXA	\$6,370.00
RSKNFLXA	\$6,730.00	TAMPFLEM	\$1,800.00	WNGRFLXA	\$6,610.00
RYHLFLXA	\$11,180.00	TAMPFLKO	\$7,450.00	WNHNFLXC	\$5,500.00
SARKFLXA	\$5,260.00	TAMPFLUT	\$5,740.00	WNPKFLTL	\$5,500.00

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.4 <u>Intrastate DS-3 Local Access</u> (Cont'd)
    - 1.4.3. Rates and Charges: (Cont'd)
      - 1.4.3.2. <u>Local Access Channel Non-Recurring Charges</u>: A non-recurring channel charge of \$1000 will apply.

### 1.4.3.3. M1/3 Multiplex:

1.4.3.3.1. <u>Verizon Business III:</u> A \$950 monthly recurring charge applies, per Central Office Connection, to demultiplex/multiplex up to 28 T-1 (1.544 Mbps) channels from/into a TDS-45 (44.736 Mbps) digital signal.

#### 1.4.3.4. Network Connection Charge:

1.4.3.4.1. <u>Verizon Business III:</u> A \$1,725 monthly recurring charge applies, per connection, for circuits installed on or after October 1, 1998, if access is Customer-provided.

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 1. <u>SERVICE DESCRIPTION (CONT'D.)</u>
  - 1.4 <u>Intrastate DS-3 Local Access (Cont'd.)</u>
    - 1.4.3. Rates and Charges (Cont'd.):
      - 1.4.3.5. <u>Cross Connect Charge</u>: A monthly \$250 charge will apply, and a non-recurring charge of \$500 will apply, per connection, for circuits installed on or after October 1, 1998, for the electrical connection of a Company affiliate DS-3 Local Access circuit to an interexchange carrier other than the Company.
      - 1.4.3.6. Charges for special arrangements, including engineering, construction, labor, facility costs and other costs associated with making the interconnection compatible between the Customer's and the Company's equipment and/or facilities are additional and at the expense of the Customer and shall be charged on an individual case basis.
      - 1.4.3.7. There will be no allowance for service deferral on the Local Access Channel. All charges for access begin on the original service request date, or the Telco delivery date, which ever is later.
      - 1.4.3.8.Other Non-Recurring Charges:
        - 1.4.3.8.1. Customer Support Services charges apply.
        - 1.4.3.8.2 Intrastate Administrative Non-Recurring Charges apply.
    - 1.4.4. <u>Terms and Conditions</u>: The tariff general terms and conditions apply, except that the Credit Allowance for Interruption of Local Access Service as set forth in Section 2.3 below also applies.

- 1. SERVICE DESCRIPTION (CONT'D.)
- 1.5 Intrastate SONET Access
  - 1.5.1 Product Description: Intrastate SONET Access is provided by Company. Intrastate SONET Access provides OC3 and OC12 local loop access to SONET in conjunction with Intrastate Analog Local Access Intrastate T-1 Digital Access, Intrastate DS0 (Hubless) Access (DS0), and Intrastate DS-3 Local Access.
  - 1.5.2 Features and Options: Features and Options are not available.
  - Rates and Charges: All charges for Intrastate Analog Local Access, Intrastate T-1 Digital 1.5.3 Access, Intrastate DS-0 (Hubless) Access and Intrastate DS-3 Local Access will apply in addition to the following charges for OC3 and OC12 local loops, with the following exceptions:
    - 1.5.3.1. Local Access Channel Monthly Recurring Channel Charges: The monthly recurring channel charges for OC3 and OC12 are calculated on an individual case basis.
    - 1.5.3.2. Local Access Channel Non-Recurring Charges: A non-recurring channel charge of \$3000 will apply per circuit.
    - 1.5.3.3. Network Connection Charge: The following monthly recurring and non-recurring charges will apply, per connection, only for circuits installed on or after October 1, 1998, whether access is Company -provided or Customer-provided.

For SONET DS3 and SONET STS-1, the following charge will apply:

Installation
\$0.00

For SONET OC3 and above, the following charge will apply:

<u>Monthly</u>	Installation
\$3.450	\$0.00

- 1.5.3.4. Charges for special arrangements, including engineering, construction, labor, facility costs and other costs associated with making the interconnection compatible between the Customer's and the Company's equipment and/or facilities are additional and at the expense of the Customer and shall be charged on an individual case basis.
- 1.5.3.5. Other Non-Recurring Charges:
  - 1.5.3.5.1 The Customer Support Services charges apply. The Intrastate Administrative Non-Recurring Charges apply. 1.5.3.5.2
- 1.5.4. Terms and Conditions: The tariff general terms and conditions apply, except that the Credit Allowance for Interruption of Local Access Service as set forth in Section 2.1 below also applies.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

2.0 <u>SPECIAL PRICING PLANS AND CREDITS</u>. Special Pricing Plans and Credits are available to the extent described below.

## 2.1 Access Pricing Plans (APP)

Customers who order Intrastate Network Services Local Access Service from the Company may subscribe to service through an Access Pricing Plan (APP) which qualifies the Customer to receive discounts which apply against the applicable monthly local circuit rates. APPs are available to Pre-Verizon Business Services I customers only. Other charges (e.g., Access Coordination, Central Office Connection, Network Connection, and installation charges) described herein are not discountable under these plans. For purposes of these APPs, Dedicated Access Line Service only includes Intrastate Analog Local Access, Intrastate T-1 Digital Access, and Intrastate DS0 (Hubless) Access. An APP applies only to one access circuit. The plan and its discounts are not transferrable to any other circuit, nor may the APP be transferred from one Customer to another.

## 2.1.1. Commencing an APP

All Customer requests to commence an APP must be made in writing and received by the Company no later than the 15th day of the month preceding the month in which the APP is to commence. A Customer must provide the Company with one month written notice to change an APP. All charges for access facilities will apply until the requested disconnection date. The charges will apply whether or not the Customer uses the circuit. The APP commences the day of installation for a new circuit, or the first day the Initial Circuit can be billed under the plan.

#### 2.1.2. Rate Changes

Customers whose circuits are subscribed through an APP are subject to all rate and discount revisions which are made during the course of the APP's term.

#### 2.1.3. Early Termination

- 2.1.3.1. Customers who terminate service prior to the end of the committed term will be charged an early termination charge equal to the lesser of the following:
  - i. the number of months remaining in the term multiplied by 50 percent of the undiscounted monthly rate at the time of termination; or,
  - ii. one month's undiscounted monthly rate at the time of termination for Customers in a one-year APP; two months' undiscounted monthly rate at the time of termination for Customers in a two-year APP; six months' undiscounted monthly rate at the time of termination for Customers in a three-year APP; eight months' undiscounted monthly rate at the time of termination for Customers in a four-year APP; and ten months' undiscounted monthly rate at the time of termination for Customers in a five-year APP.

### 2.0 SPECIAL PRICING PLANS AND CREDITS (CONT'D.):

## 2.1 Access Pricing Plans (Cont'd.)

- 2.1.3.2. Early termination charges will be waived under the following circumstances:
  - i. The Customer upgrades its service from Intrastate Analog Local Access or Intrastate DDS Local Access or Intrastate DS0 (Hubless) Access to Intrastate T-1 Digital Access, or from Intrastate T-1 Digital Access to Intrastate DS-3 Local Access, the new service is provided from the Company facility to the same Customer location as the disconnected service; and the written request to disconnect the circuit under the APP and to install the upgrade circuit are received by the Company concurrently, with no lapse in service;
  - ii. The Customer replaces an APP for a given circuit with an APP expiring on or after the original APP for the same circuit;
  - iii. The Customer requests that a circuit be moved; and
    - aa. The move is to another location in the U.S. Mainland;
    - bb. The new circuit at the new location is to be installed on or before the date the original circuit is disconnected; and
    - cc.. The new circuit is enrolled in an APP expiring on or after the APP on the original circuit.
    - dd. The Customer is responsible for all applicable installation charges for the new circuit.
  - iv. The Customer replaces an APP with an MCI Vision VIP Plus or an MCI Vision Customized Business Program with a greater volume commitment than the existing MCI Vision commitment, has an MCI Vision term length equal to or greater than their existing MCI Vision term, and expires after the date of the plan.

## 2.1.4. Renewal

Upon expiration of an APP, a circuit will be automatically resubscribed through an APP with the same term commitment as the original plan unless the Customer states, in writing to the Company, that it wishes not to be re-enrolled or to be enrolled in an APP of a different term than the original. Such written notice must be received by the Company no later than 30 days prior to the existing plan's expiration.

### 2.0 SPECIAL PRICING PLANS AND CREDITS (CONT'D.):

## 2.1 Access Pricing Plans (Cont'd.)

#### 2.1.5. Discounts

The following discounts apply to those Customers who have enrolled in an APP term discount arrangement as described herein, in accordance with the terms specified therein.

Service Type/ Discount

Term Commitment (Months)	Analog Local Access	DS0 (Hubless) Local Access/ DDS Local Access	T-1 Digital Access
12	0%	0%	5%
24	4%	5%	10%
36	5%	8%	20%
48	6%	10%	22%
60	7%	15%	24%

2.2 Intrastate Access Term Discounts are available to Pre-Verizon Business Services I Customers only.

Except for Customers which acquire Dedicated Access for Intrastate Long Distance Voice Service in Special Customer Arrangements in which such Access is discounted, any Customer entering into a new On-Net Voice Services Term Plan with Intrastate Network Local Access Service provided by the Company will receive Access Term Discount Plan discounts on all their Pre-Verizon Business Services I access circuits. The discounts will apply only to Local Access Channel Monthly Recurring Channel Charges (and to no other rate elements) associated with Intrastate Analog Local Access, Intrastate DDS Local Access and Intrastate DS0 (Hubless) Access, and Intrastate T-1 Digital Access.

The following discounts, based on term commitment and access type, apply:

Service Type/ Discount

Term Commitment (Months)	Analog Local Access	DS0 (Hubless) Local Access/ DDS Local Access	T-1 Digital Access
12	0%	0%	5%
24	4%	5%	10%
36	5%	8%	20%
48	6%	10%	22%
60	7%	15%	24%

- 2.0 SPECIAL PRICING PLANS AND CREDITS (CONT'D.):
- 2.3 Credit Allowance for Interruption of Local Access Service

For the following services the Credit for Allowance for Interruption of Local Access Services set forth in Sections 2.41 through 2.4.4 applies:

- 1.1. Intrastate Analog Local Access
- 1.2. Intrastate DS0 (Hubless) Access
- 1.3. Intrastate T1 Digital Access
- 1.4. Intrastate DS-3 Local Access
- 2.3.1 In the event of a local access channel outage, a credit will be given in accordance with the following:
  - a. The outage is 30 minutes or greater, measured from the time when the Customer reports the circuit failure or outage to the Company (trouble ticket initiation) and ending when the Company closes the trouble ticket.
  - b. No credits will be given in cases where the Customer either does not initiate a trouble ticket or does not release the circuit to the Company for testing.
  - c. For each circuit meeting the above criteria, the outage credit will equal two times the monthly local access channel charge, prorated for the outage period. The credit may not exceed 100 percent of the monthly local access channel charge.

Example: For a local access channel costing \$1000 per month, a credit for a 5 hour outage would be calculated as follows:

```
5 hours / 24 hours / 30 days = .0069
.0069 X $1000 = $6.90
Credit = $6.90 X 2 = $13.80
```

- d. Two or more interruptions of 30 minutes or more during any period, up to but not including three hours, shall be considered as one interruption.
- 2.3.2 Credits will not be given if the outage or service unavailability resulted in whole or in part from one or more of the following:
  - a. Any act or omission on the part of the Customer, Customer contractors or vendors, or third party (including, but not limited to any local access provider); scheduled maintenance; labor strikes; natural disasters; or force majeure events beyond the reasonable control of the Company (e.g., acts of God, government regulation, natural emergency, etc.);
  - b. Interruptions during periods when the Customer elects not to release the service for testing or repair and continues to use it on an impaired basis;
  - c. Interruptions during any period during which the Company or its agents are not afforded access to the premises where access lines associated with the Customer's services are terminated.

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 2.0 SPECIAL PRICING PLANS AND CREDITS (CONT'D.):
    - 2.4 Guide Discounts for VBS I customers.

Discounts up to the following levels will apply to Verizon Business Services I, in lieu of all discounts, in response to competitive marketplace conditions. To be eligible for these rates, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2.) enroll in a Guide Type 13, 15 or 16 Service Package as set forth on the Guide for a term of service of either one, two, three, four or five years.

2.4.1 <u>DS0 (Hubless) Access:</u> Discounts up to the following levels apply on monthly recurring charges for DS0 (Hubless) Access apply, based on term of service:

Term of Service (Years)/Discount						
<u>1</u>	<u>2</u>	<u>3</u>	<u>4</u>	<u>5</u>		
10%	15%	18%	20%	25%		

2.4.2 <u>T-1 Digital Access and DS-3 Local Access:</u> Discounts up to the following levels apply on monthly recurring charges for T-1 Digital Access and DS-3 Local Access apply, based on term of service:

Term of Service (Years)/Discount						
<u>1</u>	<u>2</u>	<u>3 - 5</u>				
15%	20%	25%				

- VIII. INTRASTATE NETWORK SERVICES LOCAL ACCESS SERVICE (CONT'D.)
  - 2.0 SPECIAL PRICING PLANS AND CREDITS (CONT'D.):
    - 2.5 Guide Discounts for VBS II customers.

Discounts up to the following levels will apply to Verizon Business Services II, in lieu of all discounts, in response to competitive marketplace conditions. To be eligible for these rates, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2.) enroll in a Guide Type 18, 19, 20, or 21 Service Package as set forth on the Guide for a term of service of either one, two, three, four or five years.

2.5.1 <u>DS0 (Hubless) Access:</u> The following discounts on monthly recurring charges for DS0 (Hubless) Access apply, based on term of service:

Term of Service (Years)/Discount						
<u>1</u>	<u>2</u>	<u>3</u>	<u>4</u>	<u>5</u>		
10%	15%	18%	20%	25%		

2.5.2 <u>T-1 Digital Access and DS-3 Local Access</u>: The following discounts on monthly recurring charges for T-1 Digital Access and DS-3 Local Access apply, based on term of service:

Term of S	Service (Years)	<u>'Discount</u>
<u>1</u>	<u>2</u>	<u>3 - 5</u>
15%	20%	25%

## 2.0 SPECIAL PRICING PLANS AND CREDITS (CONT'D.):

### 2.6 Guide Discounts for VBS III customers.

Discounts up to the following levels will apply to Verizon Business Services II, in lieu of all discounts, in response to competitive marketplace conditions. To be eligible for these rates, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2.) enroll in a Guide Type 22 or 23 Service Package as set forth on the Guide for a term of service of either one, two, three, four or five years.

Discounts up to the following levels on monthly recurring charges for Analog, DS0 (Hubless) Access, T-1 Digital Access and DS-3 Local Access apply, based on the master services agreement term to which Customer has committed:

Term of Service (Years)/Discount				
1 2 - 5				
5%	10%			

- 3.0 <u>OTHER NON-RECURRING CHARGES</u>. In addition to the rates set forth above, the following non-recurring charges also apply to the following services:
  - 1.1. Intrastate Analog Local Access
  - 1.2. Intrastate DS0 (Hubless) Access
  - 1.3. Intrastate T1 Digital Access
  - 1.4. Intrastate DS-3 Local Access
  - 1.5. Intrastate Sonet Access

#### 3.1 Intrastate Administrative Non-Recurring Charges

3.1.1. <u>Definitions</u>. As used in this section, the following definitions apply:

Accelerated Expedite: A service order that is processed, at the request of the Customer, with the objective of installing the service in five (5) or fewer business days, whether or not installation is completed in that time period. An accelerated expedite charge will apply to dedicated access service and be assessed on a per circuit basis. For Customer-provided access, the standard Expedite charge will apply as specified in the table below.

Administrative Change: The modification of an existing circuit, at the request of the Customer, that involves changes to Customer name, Customer contact name, Customer phone number, verification of testing performed by parties other than the Company, service rearrangements not involving a physical change and/or any other administrative change not covered by a Billing Record Change.

<u>Billing Record Change:</u> A change in Customer billing address not involving physical retermination of the circuit.

<u>Cancellation of Order:</u> A (i) Customer-initiated request to discontinue processing a service order, either in part or in its entirety, prior to its completion, excluding requests resulting from the Company's determination of the unavailability of facilities to complete the order and/or (ii) discontinuance by the Company of processing a service order resulting from a Customer request to defer installation for a period longer than 30 days. Cancellation of order charges will not apply to circuits for which the Customer reinstates the same service order within 30 days of requesting discontinue processing the order. Cancellation of Order charges will be assessed per cancelled circuit or port per order. Standard Expedite Charges will apply in addition to any cancellation of order charges if Customer cancels an Accelerated or standard Expedite order before installation.

<u>Expedite:</u> A service order that is processed, at the request of the Customer, with the objective of installing the service in a time period shorter than the Company's standard installation time period for that service, whether or not installation is completed in that time period. An expedite charge will be assessed per circuit, per port, per order and/or per device depending on the service as specified in the Charges section below.

<u>Pending Order Change:</u> The modification of a service order, at the request of the Customer, prior to the completion date of the order. Charges may apply per order and per circuit or port for each modification.

3.1 <u>Intrastate Administrative Non-Recurring Charges</u> (Cont'd)

## 3.1.1. <u>Definitions</u>. (Cont'd)

<u>Physical Change:</u> The modification of an existing circuit, at the request of the Customer, requiring physical change or retermination of the circuit.

For access Types 1, 2 and 3 circuits, the Physical Change Charge will apply. In addition, for access Types 2 and 3 circuits, the Company will pass-through non-recurring charges imposed by the ILEC.

For access Type 4 and SONET circuits only, physical change will constitute circuit discontinuance and installation for which standard tariffed discontinuance and installation charges will apply.

<u>Service Date Change:</u> The modification of a service order, at the request of the Customer, to request a new order due date that is within 30 days of the original due date. Charges will apply for each occurrence in addition to other applicable Administrative Non-Recurring Charges.

3.1.2 <u>Charges</u>. The following non-recurring charges apply, as described above. Where more than one charge appears for a particular service in the "Charge" column, each charge listed will apply in the manner described.

		Nonrecurring Charge			
	Per-				Per-
	<u>Circuit</u>	<u>Port</u>	<u>Order</u>	<u>Change</u>	<u>Device</u>
Accelerated Expedite					
Network Services Local Access Services					
Analog, DS0, DS1	\$2,100	n/a	n/a	n/a	n/a
DS3	4,000	n/a	n/a	n/a	n/a
SONET (OCn)	7,000	n/a	n/a	n/a	n/a
Administrative Change					
Network Services Local Access Services	n/a	n/a	n/a	\$60	n/a
U.S. Private Line Services	n/a	n/a	n/a	60	n/a
Billing Record Change					
Network Services Local Access Services	n/a	n/a	n/a	\$60	n/a
U.S. Private Line Services	n/a	n/a	n/a	60	n/a
Cancellation Of Order					
Network Services Local Access Services	\$700	n/a	n/a	n/a	n/a
U.S. Private Line Services	700	n/a	n/a	n/a	n/a

- 3.1 <u>Intrastate Administrative Non-Recurring Charges</u> (Cont'd)
  - 3.1.2 Charges. (Cont'd)

	Nonrecurring Charge				
	Per-	Per- Per- Per- P			Per-
	<u>Circuit</u>	<u>Port</u>	<u>Order</u>	<u>Change</u>	<u>Device</u>
Expedite					
Network Services Local Access Services					
Analog, DS0, DS1	\$1,400	n/a	n/a	n/a	n/a
DS3	3,000	n/a	n/a	n/a	n/a
SONET (OCn)	4,000	n/a	n/a	n/a	n/a
U.S. Private Line Services	500	n/a	n/a	n/a	n/a
Pending Order Change					
Network Services Local Access Services	\$200	n/a	n/a	n/a	n/a
U.S. Private Line Services	200	n/a	n/a	n/a	n/a
Physical Change					
Network Services Local Access Services	**	n/a	n/a	n/a	n/a
U.S. Private Line Services	**	n/a	n/a	n/a	n/a
** Standard Service Installation charges apply					
Service Date Change					
Network Services Local Access Services	\$60	n/a	n/a	n/a	n/a
U.S. Private Line Services	60	n/a	n/a	n/a	n/a

3.2 <u>Special Access Surcharge</u>. The Special Access Surcharge will apply. This is a monthly recurring charge, per local access channel, for circuits installed prior to October 1, 1998. The surcharge, imposed by the Local Exchange Carrier, will not apply to those Customers who furnish the Company with an Exemption Certification.

#### IX. INTRASTATE U.S. PRIVATE LINE SERVICES

1. <u>SERVICE DESCRIPTION</u>: Intrastate U.S. Private Line (USPL) services provides private line transmission service within the state originating in a Company-designated Point-of-Presence (POP) in one Local Access Transport Area (LATA) and terminating in a Company-designated POP in another LATA within the state. USPL offers a suite of analog, digital, and SONET services available on point-to-point, point-to-multi-point, and multipoint configurations. Access to these services is via Dedicated Access or other compatible Company services.

The following USPL services are available:

- 1. <u>Analog and Digital Services</u>: Analog and Digital services provide dedicated analog or digital service capable of supporting voice, data, and video communications via dedicated Intrastate Analog Local Access, Intrastate DS-0 (Hubless) Access, Intrastate T-1 Digital Access, or Intrastate DS-3 Local Access. The following services are available:
  - Analog: Analog service (formerly Voice Grade Private Line or VGPL) provides analog signals at 2.4, 4.8, 9.6, and 19.2 kbps speeds. Multipoint and point-to-multipoint configurations are supported at data speeds of 2.4, 4.8 and 9.6 kbps. Analog circuits support Tie Line (TL), Automatic Ringdown (ARD), Manual Ringdown (MRD), Off-Premises Extension (OPX) and Foreign Exchange (FX) configurations.
  - 1.2 <u>Digital:</u> Digital services transmit simultaneous, full-duplex digital signals at the following speeds. Multipoint and point-to-multipoint configurations are supported at data speeds of 2.4, 4.8, 9.6 and 56 kbps.
    - 1.2.1 DS0 (Digital Signal Level 0): DS0 transmits at 2.4, 4.8, 9.6, 56 and 64kbps.
    - 1.2.2 <u>Fractional DS1 (FDS1):</u> FDS1 transmits at 112/128 kbps through 1344/1536 kbps, in increments of 56/64 kbps.
    - 1.2.3 <u>DS1 (Digital Signal Level 1):</u> DS1 transmits at 1.544 Mbps.
    - 1.2.4 <u>DS3 (Digital Signal Level 3):</u> DS3 transmits at 44.736 Mbps. One DS3 channel provides the equivalent information handling capacity of 28 DS1 channels or 672 voice equivalent circuits. DS3 is available in the following circuit topologies:
      - A. <u>Linear DS3:</u> Linear DS3 is a single DS3 IXC connecting two designated Company terminals on the Company's digital fiber-optic network.
      - B. Restorable DS3: Restorable DS3 is a dedicated circuit that provides redundancy as it is provisioned over a physical ring topology.
- 2. <u>SONET:</u> SONET provides dedicated, point-to-point, simultaneous full-duplex optical private line services with synchronous optical network (SONET) transmission at speeds from 45 Mbps to 622 Mbps. The following bandwidths are available for IXC transport between Company-designated POPs: DS3 (44.736 Mbps), OC3 (155.520 Mbps), and OC12 (622.08 Mbps). Linear, and Restorable circuits are available for all speeds. Concatenated services are available for OC3 and OC12 bandwidths. Other speeds may be available on an ICB basis.

## X. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)

- 2. <u>DEFINITIONS</u>: In addition to the General Definitions, the following apply:
- <u>Tie Line (TL)</u>: A dedicated circuit connecting two locations to establish an internal voice network by interconnecting into each location's PBX or other voice switching device.
- <u>Automatic Ringdown (ARD)</u>: A dedicated circuit connecting two locations to provide immediate voice connection automatically.

#### Mileage Band:

Mileage Band (IXC Length)

The airline mileage between two cities can be calculated using the Vertical (V) and Horizontal (H) Coordinates of the serving wire centers associated with the Company's Terminal Locations.  $\sqrt{((V_1-V_2)^2+(H_1-H_2)^2)/10}$ 

where  $V_1$  and  $H_1$  correspond to the V & H coordinates of City 1 and  $V_2$  and  $H_2$  correspond to the V & H coordinates of City 2.

Example:	<u></u>	<u>H</u>
City 1 - New York City 2 - Chicago	4997 5986	1406 3426
	$V_1 V_2 \sqrt{((4997-5986)^2-1)^2}$	H <sub>1</sub> H <sub>2</sub> +(1406-3426) <sup>2</sup> )/10
	√(50585	52.1)=711.2328

Airline Mileage = 712 miles\*

- \* Result will always be rounded to the next highest mile.
- <u>Manual Ringdown (MRD)</u>: A dedicated circuit connecting two locations to provide immediate voice connection by manual signaling.
- Off-Premises Extension (OPX): A dedicated circuit connecting a distant location to a main PBX to provide the same voice capabilities available at the main Customer location.
- <u>Foreign Exchange (FX)</u>: A dedicated circuit connecting a distant city to provide a "local presence" to callers without the expense of maintaining a physical location in a distant city.
- <u>SONET Concatenated Circuit:</u> A dedicated circuit where several fibers are joined together end-to-end resulting in full bandwidth. Concatenated circuits are noted by a "c", e.g., OC3c.
- SONET Linear Circuit: A dedicated circuit provisioned as a logical SONET ring over a single physical connection.
- <u>SONET Restorable Circuit:</u> A dedicated circuit that provides redundancy as it is provisioned over a physical SONET ring topology.

## IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)

### 3. <u>FEATURES AND OPTIONS</u>:

- 1. <u>Web Digital Reconfiguration Services (Web DRS):</u> Web DRS provides Customers with a web-based platform to manage the configuration of their USPL circuits, connecting or re-routing circuits as needed using the following Web-based tools:
  - 1.1 Switched DS1 Services (Web DRS SWDS1): Web DRS SWDS1 allows Customers to temporarily access full DS1 bandwidth on demand, for example for infrequent high bandwidth applications such as video conferencing. It is available for DS1 only. Web DRS SWDS1 uses a mesh network of Inter Device Trunks (IDTs) that connects strategic Wideband Digital Cross Connects (DXCs). Tail circuits connect Customer to the Switched T1 Wideband DXC. Using the Web DRS tool, Customer selects any two Customer end points for a connection. Usage charges apply only for the time the connections are active. The connections can be enabled at the time of the request or they can be scheduled to go up and down at a later time.
  - 1.2 Fixed Network Reconfiguration Service (Web DRS FNR): Web DRS FNR allows Customer to redirect its DS0, DS1 or DS3 bandwidth when they temporarily need additional capacity at another location for applications such as disaster recovery. Web DRS FNR uses a fixed network of multiple pre-defined Customer circuit routes connected to a single DXC. Using the Web DRS tool, Customer may change the route to which the current Customer capacity is assigned and reassign that capacity to a different circuit route.
- 2. <u>Echo Control</u>: Echo Control provides echo cancellation equipment for DS1 circuits. Echo cancellation devices monitor an incoming signal and generate a negative image which is combined with an echo signal negating the effects of echo on the line. Echo Control is required for each end of a DS1 circuit for IXC lengths of 500 miles or greater when used to transmit voice and analog data.

#### RATES AND CHARGES:

- 1. <u>Monthly Recurring Charges:</u> Monthly recurring IXC charges include fixed (flat) per-circuit and per-mile rate components which are bandwidth sensitive and vary according to Mileage Band.
  - 1.1 <u>Verizon Business Services III (VBS III) Charges:</u> The following IXC per-circuit monthly recurring charges apply.

Analog Monthly Recurring IXC Per-Circuit Charges							
Speed	Mileage Band	Fixed Charge	Per-Mile Charge				
2.4 – 64 kbps	0 - 49	\$450	\$0.00				
	50 - 249	\$400	\$1.00				
	250 - 499	\$300	\$0.80				

- 4. RATES AND CHARGES (CONT'D)
- 1. <u>Monthly Recurring Charges (Cont'd)</u>
  - 1.1 Verizon Business Services III (VBS III) Charges (Cont'd)

	Digital Mon	thly Recurring IXC		
Туре	Speed	Mileage Band	Fixed Charge	Per-Mil Charg
DS0	2.4 – 64 kbps	0 - 49	\$450	\$0.0
		50 - 249	\$400	\$1.0
		250 - 499	\$300	\$0.8
		500 - 999	\$100	\$0.8
Fractional DS1	56/64-504/576 kbps	0 - 9999	\$500	\$1.0
	560/640 kbps	0 - 9999	\$500	\$1.C
	616/704 kbps	0 - 9999	\$500	<b>\$1.0</b>
	672/768 kbps	0 - 9999	\$500	\$1.1
	728/832 kbps	0 - 9999	\$500	\$1.2
	784/896 kbps	0 - 9999	\$500	\$1.3
	840/960 kbps	0 - 9999	\$500	\$1.3
	896/1024 kbps	0 - 9999	\$500	\$1.4
	952/1088 kbps	0 - 9999	\$500	\$1.5
	1008/1152 kbps	0 - 9999	\$500	\$1.5
	1064/1216 kbps	0 - 9999	\$500	\$1.6
	1120/1280 kbps	0 - 9999	\$500	\$1.7
	1176/1344 kbps	0 - 9999	\$500	\$1.7
	1232/1488 kbps	0 - 9999	\$500	\$1.8
	1288/1472 kbps	0 - 9999	\$500	\$1.9
	1344/1536 kbps	0 - 9999	\$500	\$2.0
DS1	1544 kbps	0 - 49	\$500	\$0.0
		50 - 249	\$400	\$2.2
		250 - 499	\$0	\$2.2
		500 - 999	\$0	\$1.4
Linear DS3	44.736 Mbps	0 - 49	\$2,000	0.0
		50 - 249	\$1,500	11.0
		250 - 499	\$0	9.0
		500 - 999	\$0	8.0
Restorable DS3	44.736 Mbps	0 - 49	\$2,500	0.0
		50 - 249	\$2,000	11.0
		250 - 499	\$0	9.0
		500 - 999	\$0	8.0

- 4. RATES AND CHARGES (CONT'D)
- 1. <u>Monthly Recurring Charges</u> (Cont'd)
  - 1.1 <u>Verizon Business Services III (VBS III) Charges</u> (Cont'd)

	SONET Monthly Recurring IXC				
Speed	Туре	Mileage Band	Fixed Charge	Per-Mile Charge	
OC3	Linear	0 - 49	\$3,000	\$0.00	
		50 - 249	\$2,500	\$14.00	
		250 - 499	\$0	\$13.00	
		500 - 999	\$0	\$9.00	
OC3	Restorable	0 - 49	\$3,000	\$0.00	
		50 - 249	\$2,500	\$14.00	
		250 - 499	\$0	\$13.00	
		500 - 999	\$0	\$9.00	
OC12	Linear	0 - 49	\$6,500	\$0.00	
		50 - 249	\$5,500	\$26.00	
		250 - 499	\$0	\$26.00	
		500 - 999	\$0	\$24.00	
OC12	Restorable	0 - 49	\$6,500	\$0.00	
		50 - 249	\$5,500	\$26.00	
		250 - 499	\$0	\$26.00	
		500 - 999	\$0	\$24.00	
SONET STS-1 is not	SONET STS-1 is not available.				

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 4. RATES AND CHARGES (CONT'D)
  - 1. <u>Monthly Recurring Charges</u> (Cont'd)
    - 1.2 <u>Verizon Business Services II (VBS II) Charges:</u> The following IXC per-circuit monthly recurring charges apply. This pricing program is only available to Customers who have a service agreement that provides for it.

Analog Monthly Recurring IXC Per-Circuit Charges				
	Speed		Fixed Charge	Per-Mile Charge
2.4	- 64 kbps	0+	\$475	\$0.30
	Digital Mont	hly Recurring IXC		
Туре	Speed	Mileage Band	Fixed Charge	Per-Mile Charge
DS0:	2.4 – 64 kbps	0+	\$475	\$0.30
Fractional DS1	56/64-504/576 kbps	0+	\$600	\$0.30
Fractional DS1	560/640 kbps	0+	\$600	\$0.37
Fractional DS1	616/704 kbps	0+	\$600	\$0.39
Fractional DS1	672/768 kbps	0+	\$600	\$0.41
Fractional DS1	728/832 kbps	0+	\$600	\$0.44
Fractional DS1	784/896 kbps	0+	\$600	\$0.46
Fractional DS1	840/960 kbps	0+	\$600	\$0.49
Fractional DS1	896/1024 kbps	0+	\$600	\$0.52
Fractional DS1	952/1088 kbps	0+	\$600	\$0.55
Fractional DS1	1008/1152 kbps	0+	\$600	\$0.58
Fractional DS1	1064/1216 kbps	0+	\$600	\$0.61
Fractional DS1	1120/1280 kbps	0+	\$600	\$0.64
Fractional DS1	1176/1344 kbps	0+	\$600	\$0.68
Fractional DS1	1232/1488 kbps	0+	\$600	\$0.72
Fractional DS1	1288/1472 kbps	0+	\$600	\$0.76
Fractional DS1	1344/1536 kbps	0+	\$600	\$0.80
DS1:	1544 kbps	0+	\$600	\$0.85
Linear DS3	44.736 Mbps	0 -100	\$2,400	\$10.00
Linear DS3	44.736 Mbps	101 – 499	\$2,700	\$7.00
Linear DS3	44.736 Mbps	500 +	\$3,200	\$7.00
Restorable DS3	44.736 Mbps	0 – 100	\$2,800	\$11.25
Restorable DS3	44.736 Mbps	101 – 499	\$3,000	\$8.00
Restorable DS3	44.736 Mbps	500 +	\$3,500	\$8.00

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 4. RATES AND CHARGES (CONT'D):
  - 1. <u>Monthly Recurring Charges</u> (Cont'd)
    - 1.2 <u>Verizon Business Services II (VBS II) Charges</u> (Cont'd)

SONET Monthly Recurring IXC				
Speed	Туре	Mileage Band	Fixed Charge	Per-Mile Charge
OC3	Linear	0+	\$7,040	\$16.56
OC3	Restorable	0+	\$8,000	\$18.00
OC12	Linear	0+	\$21,120	\$49.68
OC12	Restorable	0+	\$24,000	\$54.00

As of December 1, 2005, SONET STS-1 is not available for new installations. For SONET STS-1 installations prior to December 1, 2006, the <u>SONET STS-1 monthly recurring charges</u> in Section 4.1.4.2 will apply.

1.3 <u>Verizon Business Services I (VBS I) Charges:</u> The following IXC per-circuit monthly recurring charges apply. This pricing program is only available to Customers who have a service agreement that provides for it.

Analog Monthly Recurring IXC				
Speed	Mileage Band	Fixed Charge	Per-Mile Charge	
2.4 – 64 kbps	0+	\$375	\$0.25	

- 4. RATES AND CHARGES (CONT'D):
- 1. <u>Monthly Recurring Charges</u> (Cont'd)
  - 1.3 <u>Verizon Business Services I (VBS I) Charges</u> (Cont'd)

1.5 <u>venz</u>	Digital M	lonthly Recurring IXC	<u>'</u>	
Туре	Speed	Mileage Band	Per-Circuit Fixed	Per-Mile
			Charge	Charge
DS0	2.4-9.6 kbps	0+	\$375	\$0.25
DS0	56/64 kbps	0+	\$375	\$0.25
Fractional DS1	56/64 kbps	0+	\$600	\$0.30
Fractional DS1	112/128 kbps	0+	\$600	\$0.25
Fractional DS1	168/192 kbps	0+	\$600	\$0.25
Fractional DS1	224/256 kbps	0+	\$600	\$0.25
Fractional DS1	280/320 kbps	0+	\$600	\$0.25
Fractional DS1	336/384 kbps	0+	\$600	\$0.25
Fractional DS1	392/448 kbps	0+	\$600	\$0.25
Fractional DS1	448/512 kbps	0+	\$600	\$0.25
Fractional DS1	504/576 kbps	0+	\$600	\$0.25
Fractional DS1	560/640 kbps	0+	\$600	\$0.37
Fractional DS1	616/704 kbps	0+	\$600	\$0.39
Fractional DS1	672/768 kbps	0+	\$600	\$0.41
Fractional DS1	728/832 kbps	0+	\$600	\$0.44
Fractional DS1	784/896 kbps	0+	\$600	\$0.46
Fractional DS1	840/960 kbps	0+	\$600	\$0.49
Fractional DS1	896/1024 kbps	0+	\$600	\$0.52
Fractional DS1	952/1088 kbps	0+	\$600	\$0.55
Fractional DS1	1008/1152 kbps	0+	\$600	\$0.58
Fractional DS1	1064/1216 kbps	0+	\$600	\$0.61
Fractional DS1	1120/1280 kbps	0+	\$600	\$0.64
Fractional DS1	1176/1344 kbps	0+	\$600	\$0.68
Fractional DS1	1232/1488 kbps	0+	\$600	\$0.72
Fractional DS1	1288/1472 kbps	0+	\$600	\$0.76
Fractional DS1	1344/1536 kbps	0+	\$600	\$0.80
DS1	1544 kbps	0+	\$600	\$0.85
Linear DS3	44.736 Mbps	0 – 100	\$2,400	\$10.00
Linear DS3	44.736 Mbps	101 – 499	\$2,700	\$7.00
Linear DS3	44.736 Mbps	500 +	\$3,200	\$7.00
Restorable DS3	44.736 Mbps	0 – 100	\$3,000	\$12.50
Restorable DS3	44.736 Mbps	101 – 499	\$3,375	\$8.75
Restorable DS3	44.736 Mbps	500 +	\$4,000	\$8.75

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - RATES AND CHARGES (CONT'D):
  - 1. <u>Monthly Recurring Charges</u> (Cont'd)
    - 1.3 <u>Verizon Business Services I (VBS I) Charges</u> (Cont'd)

SONET Monthly Recurring IXC				
Speed	Туре	Mileage Band	Per-Circuit Fixed Charge	Per-Mile Charge
OC3	Linear	0+	\$7,128	\$17.39
OC3	Restorable	0+	\$8,100	\$18.90
OC12	Linear	0+	\$28,512	\$62.56
OC12	Restorable	0+	\$32,400	\$68.00

As of December 1, 2005, SONET STS-1 is not available for new installations. For SONET STS-1 installations prior to December 1, 2006, the <u>SONET STS-1 monthly recurring charges</u> in Section 4.1.4.2 apply.

- 1.4 <u>Pre-VBS I Charges:</u> The following IXC per-circuit monthly recurring charges apply. This pricing program is only available to Customers who have a service agreement that provides for it. As of December 1, 2005, SONET STS-1 is not available for new installations.
  - 1.4.1 <u>Analog and Digital Service:</u> The following IXC per-circuit monthly recurring fixed and per-mile charges for Analog, DSO and Fractional DS1 services apply, based on Mileage Band:

Section 1.4.1 - Service	Beginning Mileage	Ending Mileage	Fixed Rate	Per Mile Rate
VGPL	0	0	\$100.00	\$0.00
VGPL	1	50	\$237.60	\$0.34
VGPL	51	100	\$261.90	\$0.34
VGPL	101	9999	\$281.30	\$0.34
DSO	0	0	\$100.00	\$0.00
DSO	1	50	\$261.90	\$0.34
DSO	51	100	\$261.90	\$0.34
DSO	101	9999	\$281.30	\$0.34
DDS 2.4/4.8/9.6K	0	0	\$100.00	\$0.00
DDS 2.4/4.8/9.6K	1	50	\$149.32	\$1.48
DDS 2.4/4.8/9.6K	51	100	\$149.32	\$1.48
DDS 2.4/4.8/9.6K	101	9999	\$149.32	\$1.48
DDS 56/64K	0	0	\$100.00	\$0.00
DDS 56/64K	1	9999	\$567.43	\$5.40
FracDS1 56/64K	0	0	\$100.00	\$0.00
FracDS1 56/64K	1	50	\$261.90	\$0.34
FracDS1 56/64K	51	100	\$261.90	\$0.34
FracDS1 56/64K	101	9999	\$281.30	\$0.34

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 4. RATES AND CHARGES (CONT'D):
  - 1. Monthly Recurring Charges (Cont'd)
    - 1.4 <u>Pre-VBS I Charges:</u> (Cont'd)
      - 1.4.1 Analog and Digital Service (Cont'd)

Section 1.4.1 - Service	Beginning Mileage	Ending Mileage	Fixed Rate	Per Mile Rate
FracDS1 112/128K	0	0	\$200.00	\$0.00
FracDS1 112/128K	1	50	\$528.65	\$0.61
FracDS1 112/128K	51	100	\$528.65	\$0.61
FracDS1 112/128K	101	9999	\$528.65	\$0.61
FracDS1 168/192K	0	0	\$200.00	\$0.00
FracDS1 168/192K	1	50	\$776.00	\$0.89
FracDS1 168/192K	51	100	\$776.00	\$0.89
FracDS1 168/192K	101	9999	\$776.00	\$0.89
FracDS1 224/256K	0	0	\$200.00	\$0.00
FracDS1 224/256K	1	50	\$1,003.95	\$1.15
FracDS1 224/256K	51	100	\$1,003.95	\$1.15
FracDS1 224/256K	101	9999	\$1,003.95	\$1.15
FracDS1 280/320K	0	0	\$200.00	\$0.00
FracDS1 280/320K	1	50	\$1,246.45	\$1.45
FracDS1 280/320K	51	100	\$1,246.45	\$1.45
FracDS1 280/320K	101	9999	\$1,246.45	\$1.45
FracDS1 336/384K	0	0	\$200.00	\$0.00
FracDS1 336/384K	1	50	\$1,459.85	\$1.70
FracDS1 336/384K	51	100	\$1,459.85	\$1.70
FracDS1 336/384K	101	9999	\$1,459.85	\$1.70
FracDS1 392/448K	0	0	\$200.00	\$0.00
FracDS1 392/448K	1	50	\$1,658.70	\$1.94
FracDS1 392/448K	51	100	\$1,658.70	\$1.94
FracDS1 392/448K	101	9999	\$1,658.70	\$1.94
FracDS1 448/512K	0	0	\$200.00	\$0.00
FracDS1 448/512K	1	50	\$1,838.15	\$2.15
FracDS1 448/512K	51	100	\$1,838.15	\$2.15
FracDS1 448/512K	101	9999	\$1,838.15	\$2.15
FracDS1 504/576K	0	0	\$200.00	\$0.00
FracDS1 504/576K	1	50	\$2,012.75	\$2.36
FracDS1 504/576K	51	100	\$2,012.75	\$2.36
FracDS1 504/576K	101	9999	\$2,012.75	\$2.36
FracDS1 560/640K	0	0	\$200.00	\$0.00
FracDS1 560/640K	1	50	\$2,167.95	\$2.55
FracDS1 560/640K	51	100	\$2,167.95	\$2.55
FracDS1 560/640K	101	9999	\$2,167.95	\$2.55
FracDS1 616/704K	0	0	\$200.00	\$0.00
FracDS1 616/704K	1	50	\$2,308.60	\$2.72
FracDS1 616/704K	51	100	\$2,308.60	\$2.72
FracDS1 616/704K	101	9999	\$2,308.60	\$2.72

- 4. RATES AND CHARGES (CONT'D):
- 1. <u>Monthly Recurring Charges</u> (Cont'd)
  - 1.4 Pre-VBS I Charges: (Cont'd)
    - 1.4.1 Analog and Digital Service (Cont'd)

Section 1.4.1 - Service	Beginning Mileage	Ending Mileage	Fixed Rate	Per Mile Rate
FracDS1 672/768K	0	0	\$200.00	\$0.00
FracDS1 672/768K	1	50	\$2,434.70	\$2.88
FracDS1 672/768K	51	100	\$2,434.70	\$2.88
FracDS1 672/768K	101	9999	\$2,434.70	\$2.88
FracDS1 728/832K	0	0	\$200.00	\$0.00
FracDS1 728/832K	1	9999	\$970.00	\$7.76
FracDS1 784/896K	0	0	\$200.00	\$0.00
FracDS1 784/896K	1	9999	\$970.00	\$7.76
FracDS1 840/960K	0	0	\$200.00	\$0.00
FracDS1 840/960K	1	9999	\$970.00	\$7.76
FracDS1 896/1024K	0	0	\$200.00	\$0.00
FracDS1 896/1024K	1	9999	\$970.00	\$7.76
FracDS1 952/1088K	0	0	\$200.00	\$0.00
FracDS1 952/1088K	1	9999	\$970.00	\$7.76
FracDS1 1008/1152K	0	0	\$200.00	\$0.00
FracDS1 1008/1152K	1	9999	\$970.00	\$7.76
FracDS1 1064/1216K	0	0	\$200.00	\$0.00
FracDS1 1064/1216K	1	9999	\$970.00	\$7.76
FracDS1 1120/1280K	0	0	\$200.00	\$0.00
FracDS1 1120/1280K	1	9999	\$970.00	\$7.76
FracDS1 1176/1344K	0	0	\$200.00	\$0.00
FracDS1 1176/1344K	1	9999	\$970.00	\$7.76
FracDS1 1232/1408K	0	0	\$200.00	\$0.00
FracDS1 1232/1408K	1	9999	\$970.00	\$7.76
FracDS1 1288/1472K	0	0	\$200.00	\$0.00
FracDS1 1288/1472K	1	9999	\$970.00	\$7.76
DS1	0	0	\$200.00	\$0.00
DS1	1	9999	\$970.00	\$7.76
SONET Linear (T45) DS3	0	0	\$0.00	\$0.00
SONET Linear (T45) DS3	1	50	\$12,125.00	\$33.95
SONET Linear (T45) DS3	51	100	\$12,125.00	\$33.95
SONET Linear (T45) DS3	101	9999	\$12,125.00	\$33.95

- 4. RATES AND CHARGES (CONT'D.):
  - 1.4 <u>Pre-VBS I Charges (Cont'd.)</u>:
  - 1.4.2 <u>SONET:</u> The following monthly recurring per-circuit charges apply for restorable circuits based on the city-pair location of the circuit. As of December 1, 2005, SONET STS-1 is not available for new installations:

SONET DS3 rates

Originating POP	Terminating POP	Monthly Recurring Charge
AUB	DYA	\$49,389.17
AUB	JKV	\$52,695.30
AUB	MSB	\$54,240.90
AUB	ORL	\$38,833.20
AUB	PNN	\$73,271.10
AUB	POB	\$52,985.10
AUB	TAU	\$62,317.87
AUB	TMP	\$38,688.30
DYA	JKV	\$45,643.50
DYA	MSB	\$59,843.70
DYA	ORL	\$42,455.70
DYA	PNN	\$74,623.50
DYA	POB	\$58,008.30
DYA	TAU	\$58,008.30
DYA	TMP	\$49,507.50
JKV	MSB	\$67,861.50
JKV	ORL	\$49,507.50
JKV	PNN	\$69,793.50
JKV	POB	\$66,122.70
JKV	TAU	\$55,703.19
JKV	TMP	\$53,951.10
MSB	ORL	\$56,076.30
MSB	PNN	\$93,286.62
MSB	POB	\$35,645.40
MSB	TAU	\$79,656.36
MSB	TMP	\$56,269.50
ORL	PNN	\$73,947.30
ORL	POB	\$54,434.10
ORL	TAU	\$61,916.98
ORL	TMP	\$43,904.70
PNN	POB	\$92,484.84
PNN	TAU	\$57,807.86
PNN	TMP	\$70,276.50
POB	TAU	\$78,453.69
POB	TMP	\$55,303.50
TAU	TMP	\$60,413.64

# 4. RATES AND CHARGES (CONT'D.):

# 1.4 <u>Pre-VBS I Charges (Cont'd.)</u>:

# 1.4.2 <u>SONET:</u> (Cont'd.)

SONET OC3 rates

Originating POP	Terminating POP	Monthly Recurring Charge
AUB	DYA	\$148,167.50
AUB	JKV	\$158,085.90
AUB	MSB	\$162,722.70
AUB	ORL	\$116,499.60
AUB	PNN	\$219,813.30
AUB	POB	\$158,955.30
AUB	TAU	\$186,953.61
AUB	TMP	\$116,064.90
DYA	JKV	\$136,930.50
DYA	MSB	\$179,531.10
DYA	ORL	\$127,367.10
DYA	PNN	\$223,870.50
DYA	POB	\$174,024.90
DYA	TAU	\$174,024.90
DYA	TMP	\$148,522.50
JKV	MSB	\$203,584.50
JKV	ORL	\$148,522.50
JKV	PNN	\$209,380.50
JKV	POB	\$198,368.10
JKV	TAU	\$167,109.55
JKV	TMP	\$161,853.30
MSB	ORL	\$168,228.90
MSB	PNN	\$279,859.86
MSB	POB	\$106,936.20
MSB	TAU	\$238,969.08
MSB	TMP	\$168,808.50
ORL	PNN	\$221,841.90
ORL	POB	\$163,302.30
ORL	TAU	\$185,750.94
ORL	TMP	\$131,714.10
PNN	POB	\$277,454.52
PNN	TAU	\$173,423.57
PNN	TMP	\$210,829.50
POB	TAU	\$235,361.07
POB	TMP	\$165,910.50
TAU	TMP	\$181,240.92

# 4. RATES AND CHARGES (CONT'D.):

# 1.4 <u>Pre-VBS I Charges (Cont'd.)</u>:

# 1.4.2 <u>SONET:</u> (Cont'd.)

**SONET STS-1 rates** 

Originating POP	Terminating POP	Monthly Recurring Charge
AUB	DYA	\$52,846.41
AUB	JKV	\$56,383.98
AUB	MSB	\$58,037.77
AUB	ORL	\$41,551.53
AUB	PNN	\$78,400.08
AUB	POB	\$56,694.06
AUB	TAU	\$66,680.12
AUB	TMP	\$41,396.49
DYA	JKV	\$48,838.55
DYA	MSB	\$64,032.76
DYA	ORL	\$45,427.60
DYA	PNN	\$79,847.15
DYA	POB	\$62,068.89
DYA	TAU	\$62,068.89
DYA	TMP	\$52,973.03
JKV	MSB	\$72,611.81
JKV	ORL	\$52,973.03
JKV	PNN	\$74,679.05
JKV	POB	\$70,751.29
JKV	TAU	\$59,602.41
JKV	TMP	\$57,727.68
MSB	ORL	\$60,001.65
MSB	PNN	\$99,816.69
MSB	POB	\$38,140.58
MSB	TAU	\$85,232.31
MSB	TMP	\$60,208.37
ORL	PNN	\$79,123.62
ORL	POB	\$58,244.49
ORL	TAU	\$66,251.17
ORL	TMP	\$46,978.03
PNN	POB	\$98,958.78
PNN	TAU	\$61,854.41
PNN	TMP	\$75,195.86
POB	TAU	\$83,945.45
POB	TMP	\$59,174.75
TAU	TMP	\$64,642.60

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 4. RATES AND CHARGES (CONT'D.):
    - 1.4 <u>Pre-VBS I Charges (Cont'd.)</u>:
    - 1.4.2 <u>SONET:</u> (Cont'd.)
      - A. The following IXC per-circuit monthly recurring charges apply for linear circuits as well as restorable circuits whose city-pairs are not found in the SONET DS3 and SONET OC3 tables in Section 1.4.2:

Bandwidth	Mileage Band	Fixed Charge	+	Per-Mile Charge
DS3	1 – 100	\$55,777		\$246.11
	101 – 500	\$64,041		\$165.59
	501 +	\$68,377		\$165.59
OC3	0 +	\$98,242		\$308.00

#### 4. RATES AND CHARGES (CONT'D.):

2.0 <u>Discounts:</u> Network Pricing Plan discounts are available to Pre-Verizon Business Service I Customers only and apply to Inter Exchange Channel (IXC) charges.

Network Pricing Plan (NPP) discounts specified below are limited to Pre-Verizon Business Service I Customers only and apply to IXC charges.

Under these plans, a Customer commits to a minimum monthly revenue level of combined usage over a selected term, and receives the corresponding NPP discounts for IXCs under the plan to which the Customer has committed. Network Pricing Plans are available at various minimum monthly IXC revenue levels over terms of 1, 2, 3, 4 or 5 years. For NPPs entered into on or after November 1,1998, the monthly minimum revenue may include recurring charges for Company-provided local loops. Network Pricing Plans with domestic minimum monthly IXC revenue levels of \$75,000 or greater are available only under NPPs executed prior to November 1, 1998.

Customers who subscribe to service as part of the Commercial Affinity Program Plus (CAP Plus) or the Qualified Industry Affinity Program (INAP) are eligible to receive the NPP discounts associated with the 4 year, \$2,000 minimum monthly IXC revenue level. The NPP discounts apply only to Company-provided IXCs, and customers receiving these discounts cannot receive discounts provided under any Network Pricing Plan Data Promotion. Customers who subscribe to service as part of the Commercial Affinity Program are not eligible for these discounts.

The monthly recurring IXC charges under a particular plan are determined by deducting the respective NPP discounts from the monthly charges calculated for each service using the applicable Monthly Plan rates. IXC revenues, calculated using the applicable Monthly Plan rates for each service plus any applicable intrastate and international revenues as explained below, are totaled and compared to the minimum monthly commitment amount. Except as explained below under, "Commencing An NPP," if the aggregate revenue applicable towards the minimum monthly revenue commitment is less than the committed amount, a charge equal to the difference is applied.

Revenues from the Company-provided portion of international dedicated leased line circuits that are enrolled in a selected International Network Pricing Plan, which revenues are calculated at the undiscounted monthly or base international rates, will be counted toward a selected domestic NPP IXC minimum. The customer may choose the Network Pricing Plan to be associated with these international revenues, subject to the terms of Notification Required from Customers.

Intrastate IXC revenue, at the applicable intrastate rates, may also be counted towards a selected NPP IXC minimum.

Except as noted below, relative to commencing a plan, the commitment to a Network Pricing Plan is independent of the circuits in the plan. Circuits may be added or deleted from a Network Pricing Plan subject to the terms for Notification Required From Customers. Other circuit specific regulations, including minimum service interval and notice of discontinuance, are defined elsewhere in this tariff.

Customers enrolled in a Network Pricing Plan are not entitled to any other discount, except for City Pair Discounts, or Product Packages 6-9 ("Grand Slam" program) or for any promotion offered by the company on IXC charges arising from services furnished under the NPP.

## 4. RATES AND CHARGES (CONT'D.):

#### 2.0 <u>Discounts (Cont'd.):</u>

For NPPs entered into prior to November 1, 1998, the monthly minimum commitment must be satisfied by the customer's gross billing before the application of NPP discounts. For NPPs entered into on or after November 1, 1998, the monthly minimum commitment must be satisfied by the customer's net billing after the application of NPP discounts.

<u>Commencing An NPP:</u> Subject to the terms for Notification Required From Customers, the customer must select: the desired NPP term and month in which the plan is to commence; the monthly minimum IXC revenue commitment; and the circuits, which will initially be part of the plan, including ordered circuits which are due for installation within the first three months of the desired plan commencement.

The NPP commences on the first day of the calendar month selected by the customer, or the first day of a subsequent month in which the first circuit can be billed under the plan.

During the first three months of the plan, the Customer is not charged the difference if the aggregate revenue applicable towards the minimum monthly revenue commitment is less than the committed amount. The obligation to satisfy the minimum actually commences in the fourth month of the plan, unless a circuit due within the first three months of the plan has not been made available by the Company to the customer. In that case, the minimum will not apply until the circuit is made available.

#### Terminating a Network Pricing Plan

A customer may terminate a Network Pricing Plan before the end of its committed term without liability, if:

- the Customer replaces the existing Network Pricing Plan or Plans with a Network Pricing Plan having a minimum monthly IXC revenue commitment equal to or greater than the total monthly IXC minimum revenue commitment of the customer's existing Plan or Plans and which expires on or after the expiration date of the Plan or Plans being replaced;
- a rate revision results in lower plan charges for the Plan to which the customer is committed such that the Customer is caused to fall below the committed minimum monthly IXC revenue level, and the customer replaces the existing Plan by committing to a new Plan having the next lower NPP minimum monthly IXC revenue level and which expires on or after the date of the Plan being replaced. For purposes of applying this provision, the Customer will not be penalized for falling below the minimum under the existing Plan and can receive the new Plan charges from the date the rate revision takes effect, if the customer enters into the new Plan within three months of the rate effective date; or.

Except as specified above, the Customer will be billed and required to pay a termination charge equal to 100 percent of the NPP revenue commitment amount for any unexpired portion of the remaining months in the term plan.

- 4. RATES AND CHARGES (CONT'D.):
  - 2.0 <u>Discounts (Cont'd.):</u>
  - 2.1 <u>NPP Discounts:</u> The following discounts are applicable to domestic charges only for monthly recurring IXC charges.

#### VGPL and DS0

Monthly Commitment					
Level	1 Year	2 Years	3 Years	4 Years	5 Years
\$ 500	1%	2%	3%	4%	<del>5</del> %
\$ 1000	1%	2%	3%	4%	5%
\$ 2000	2%	5%	6%	8%	10%
\$ 3000	2%	5%	6%	8%	10%
\$ 4000	2%	5%	6%	8%	10%
\$ 5000	3%	6%	7%	9%	11%
\$ 7,000	3%	6%	7%	9%	11%
\$10,000	4%	7%	8%	10%	12%
\$15,000	4%	7%	8%	10%	12%
\$25,000	5%	8%	9%	11%	13%
\$50,000	8%	11%	12%	14%	16%
\$75,000	7%	10%	11%	13%	15%
\$100,000	8%	11%	12%	14%	16%
\$200,000	9%	12%	13%	15%	17%
\$350,000	10%	13%	14%	16%	18%
\$500,000	10%	13%	14%	16%	18%
\$750,000	10%	13%	14%	16%	18%

# 4. RATES AND CHARGES (CONT'D.):

## 2.0 <u>Discounts (Cont'd.):</u>

# 2.1 NPP Discounts (Cont'd):

# Fractional DS1 (for Speeds 64 Kbps and below)

Monthly Commitment					
Level_	<u>1 Year</u>	2 Years	3 Years	4 Years	5 Years
\$500	1%	2%	3%	4%	5%
\$1,000	1%	2%	3%	4%	5%
\$2,000	2%	5%	6%	8%	10%
\$3,000	2%	5%	6%	8%	10%
\$4,000	2%	5%	6%	8%	10%
\$5,000	3%	6%	7%	9%	11%
\$7,000	3%	6%	7%	9%	11%
\$10,000	4%	7%	8%	10%	12%
\$15,000	4%	7%	8%	10%	12%
\$25,000	5%	8%	9%	11%	13%
\$50,000	8%	11%	12%	14%	16%
\$75,000	7%	10%	11%	13%	15%
\$100,000	8%	11%	12%	14%	16%
\$200,000	9%	12%	13%	15%	17%
\$350,000	10%	13%	14%	16%	18%
\$500,000	10%	13%	14%	16%	18%
\$750,000	10%	13%	14%	16%	18%

# Fractional DS1 (for Speeds 112 Kbps up to 1472 Kbps)

Monthly					
Commitment					
Level_	<u>1 Year</u>	2 Years	3 Years	4 Years	<u>5 Years</u>
\$500	1%	2%	3%	4%	5%
\$1,000	1%	2%	3%	4%	5%
\$2,000	10%	16%	17%	18%	19%
\$3,000	10%	16%	17%	18%	19%
\$4,000	11%	17%	18%	19%	20%
\$5,000	12%	18%	19%	20%	21%
\$7,000	15%	19%	20%	21%	22%
\$10,000	19%	20%	21%	22%	23%
\$15,000	20%	21%	22%	23%	24%
\$25,000	21%	22%	23%	24%	25%
\$50,000	23%	24%	25%	26%	27%
\$75,000	25%	26%	27%	28%	29%
\$100,000	26%	27%	28%	29%	30%
\$200,000	28%	29%	30%	31%	32%
\$350,000	29%	30%	31%	32%	33%
\$500,000	30%	31%	32%	33%	34%
\$750,000	31%	32%	33%	34%	35%

## 4. RATES AND CHARGES (CONT'D.):

# 2.0 <u>Discounts (Cont'd.):</u>

# 2.1 NPP Discounts (Cont'd):

DS1

Monthly Commitment					
Level	1 Year	2 Years	3 Years	4 Years	5 Years
\$ 1,000	4%	5%	6%	8%	10%
\$ 2,000	15%	17%	19%	20%	21%
\$ 3,000	17%	19%	21%	22%	23%
\$ 4,000	19%	21%	23%	25%	26%
\$ 5,000	23%	28%	32%	33%	35%
\$ 7,000	27%	32%	34%	35%	36%
\$10,000	34%	35%	36%	37%	38%
\$15,000	35%	36%	37%	38%	39%
\$25,000	36%	37%	38%	39%	40%
\$50,000	38%	40%	41%	42%	45%
\$75,000	41%	42%	43%	44%	47%
\$100,000	44%	45%	46%	47%	49%
\$200,000	47%	48%	49%	50%	51%
\$350,000	48%	49%	50%	51%	52%
\$500,000	49%	50%	51%	52%	53%
\$750,000	50%	51%	52%	54%	56%

DS3

Monthly

Commitment					
Level	<u>1 Year</u>	2 Years	3 Years	4 Years	5 Years
\$ 2,000	1%	2%	6%	7%	8%
\$ 3,000	1%	2%	6%	7%	8%
\$ 4,000	1%	2%	6%	7%	8%
\$ 5,000	2%	7%	11%	12%	14%
\$ 7,000	2%	7%	11%	12%	14%
\$10,000	13%	14%	15%	16%	17%
\$15,000	13%	14%	15%	16%	17%
\$25,000	15%	16%	17%	18%	19%
\$50,000	17%	18%	19%	20%	21%
\$75,000	18%	19%	20%	21%	22%
\$100,000	19%	21%	22%	23%	24%
\$200,000	21%	23%	24%	25%	29%
\$350,000	25%	27%	28%	29%	33%
\$500,000	32%	35%	38%	42%	45%
\$750,000	33%	36%	39%	43%	47%

#### 4. RATES AND CHARGES (CONT'D.):

#### 2.0 <u>Discounts (Cont'd.):</u>

#### 2.1 NPP Discounts (Cont'd):

#### **SONET**

Monthly Commitment <u>Level</u>	<u>1 Year</u>	2 Years	3 Years	4 Years	<u>5 Years</u>
\$25,000	9%	14%	17%	18%	19%
\$50,000	11%	16%	19%	20%	21%
\$75,000	13%	18%	20%	21%	22%
\$100,000	15%	20%	22%	23%	24%
\$200,000	17%	22%	24%	25%	29%
\$350,000	19%	24%	26%	27%	31%
\$500,000	29%	34%	38%	42%	45%
\$750,000	31%	36%	39%	43%	48%

#### 2.2 Guide Discounts for VBS I Customers

Discounts up to the following levels will apply to Verizon Business Services I, in lieu of all discounts, in response to competitive marketplace conditions. To be eligible for these rates, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2.) enroll in a Guide Type 13, 14, 15 or 16 Service Package as set forth on the Guide for a term of service of either one, two, three, four or five years, applicable to U.S. Private Line monthly recurring Interexchange charges for Analog, DS0, Fractional DS1, DS1, Digital DS3 (Linear and Restoreable), and SONET (Linear and Restoreable), based on term of service and Annual Volume Commitment:

	Term of S	<u>ervice (Years)/</u>	<u>Discount</u>
Annual Volume Commitment	<u>1</u>	2	<u>3 - 5</u>
\$ 6,000 - \$ 11,999	5%	10%	15%
\$ 12,000 - \$ 35,999	10	15	20
\$ 36,000 - \$ 83,999	15	20	25
\$ 84,000 and greater	20	25	30

#### 4. RATES AND CHARGES (CONT'D.):

#### 2.0 Discounts (Cont'd.):

#### 2.3 Guide Discounts for VBS II Customers

Discounts up to the following levels will apply to Verizon Business Services II, in lieu of all discounts, in response to competitive marketplace conditions. To be eligible for these rates, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2.) enroll in a Guide Type 18, 19, 20 or 21 Service Package as set forth on the Guide for a term of service of either one, two, three, four or five years, applicable to U.S. Private Line monthly recurring Interexchange charges for Analog, DS0, Fractional DS1, DS1, Digital DS3 (Linear and Restoreable), and SONET (Linear and Restoreable), based on term of service and Annual Volume Commitment:

			Term of Serv	rice (Years)/Discoι	ınt
 Annual Volume Com	mit	ment	<u>1</u>	<u>2</u>	<u>3 - 5</u>
\$ 3,000 –	\$	5,999	0%	0%	0%
\$ 6,000 —	\$	11,999	5	10	15
\$ 12,000 —	\$	23,999	10	15	20
\$ 24,000 -	\$	35,999	10	15	20
\$ 36,000 –	\$	83,999	15	20	25
\$ 84,000 and greater			20	25	30

#### 4. RATES AND CHARGES (CONT'D.):

#### 2.0 <u>Discounts (Cont'd.):</u>

#### 2.4 Guide Discounts for VBS III Customers

Discounts up to the following levels will apply to Verizon Business Services II, in lieu of all discounts, in response to competitive marketplace conditions. To be eligible for these rates, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2.) enroll in a Guide Type 22 or 23 Service Package as set forth on the Guide for a term of service of either one, two, three, four or five years, applicable to U.S. Private Line monthly recurring Interexchange charges for Analog, DS0, Fractional DS1, DS1, Digital DS3 (Linear and Restoreable), and SONET (Linear and Restoreable), based on term of service and Annual Volume Commitment:

			ent Term Comr /ears)/Discoun	
Annual Volume Commitment		1	2	3 - 5
\$ 1-	\$ 1,199	0%	0%	0%
\$ 1,200 -	\$ 2,999	5	10	15
\$ 3,000 -	\$ 5,999	10	15	20
\$ 6,000 -	\$ 11,999	15	20	25
\$ 12,000 -	\$ 23,999	20	25	25
\$ 24,000 -	\$ 35,999	20	25	30
\$ 36,000 -	\$119,999	25	30	30
\$120,000 and	greater	30	30	30

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 4. RATES AND CHARGES (CONT'D.):
    - 3.0 <u>Feature Charges:</u>
    - 3.1 The following monthly recurring usage charges apply to Web DRS SWDS1, based on Mileage Band. A minimum monthly usage fee of \$500 per circuit applies in any monthly billing period for which the service is ordered.

IXC Mileage	Per Minute Rate
0 – 300	\$0.99
301 – 900	\$1.24

3.2 The following charges apply to Web DRS FNR, based on circuit type:

Circuit Type	Installation	Monthly Recurring Charge
DS0	\$25	\$ 30
DS1	\$258	\$350
Linear DS3	\$400	\$500

3.3 <u>Echo Control Charges:</u> The following charges apply to each end of a DS1 circuit with IXC mileage lengths of 500 miles or greater when used to transmit voice and analog data.

Monthly Recurring Charge Per Circuit End	Installation Charge
\$200	\$1,000

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - RATES AND CHARGES (CONT'D.):
    - 4. Other Non-Recurring Charges:
    - 4.1 Customer Support Services charges apply.
    - 4.2 Intrastate Administrative Non-Recurring Charges apply.
    - 4.3 <u>Installation:</u> The following per-circuit installation charges apply based on circuit type.

Circuit Type	Charge
Analog	\$50
Digital DS0	\$50
Digital FDS1	\$100
Digital DS1	\$200
Digital DS3	\$600
SONET OC3	\$1000
SONET OC12	\$1000

- State taxes and surcharges may apply.
- 5. <u>TERMS AND CONDITIONS:</u> In addition to the General Terms and Conditions, the following applies:
  - Special Access Surcharges for Analog and Digital USPL will not be applied after receipt of an Exemption Certificate from Customer. A credit, not to exceed three months, will be given for a private line surcharge imposed during the period prior to the receipt of the Exemption Certificate.
  - 2. The minimum service term requirement for all SONET circuits is 12 months. If Customer terminates any SONET circuit before its 12-month commitment has expired, except for termination for Cause, such termination shall not be effective until 30 days after Company receives written notice of termination (Termination Date). In addition to paying all accrued but unpaid charges for the service incurred through the Termination Date, for each circuit terminated Customer may be required to pay, within 30 days after such Termination Date: (a) an amount equal 75 percent of the monthly recurring charges for the terminated circuit remaining in the 12-month commitment, if any; plus (b) all fees or early termination fees imposed by the access line provider, if any; plus (c) a pro rata portion of any and all credits received by Customer. However, in no event will Customer's total termination liability exceed the full contract value of the terminated SONET circuit.

- 6. <u>SERVICE LEVEL AGREEMENTS</u>: The following service level agreement (SLA) applies:
  - 1. VBS III U.S. Private Line Service Level Agreement
    - Scope
    - A. General.

This Service Level Agreement ("SLA") applies to Company's U.S. Private Line ("USPL") Service on an end-to-end basis, except where otherwise noted. Customer may receive credits for failure by Company to satisfy the "Service Availability", "Mean Time to Repair" or "Service Installation Period" standards as set forth below. This SLA is only available for customers who pay Verizon Business Service III charges for USPL Service. This SLA includes both service level standards and a service level objective.

This SLA applies to the following USPL services: Linear services, including Analog, Digital Signal Level 0 ("DS0"), Fractional DS1, and Digital Signal Level 1 ("DS1"); and SONET Services, including Restorable DS3, STS-1, OC3, OC3c, OC12 and OC12c, except where noted elsewhere below. This SLA does not apply to Linear DS3 service or to any services made available on an ICB basis.

The Mean Time to Repair and Service Installation Standards do not apply to SONET service.

This SLA applies only to "Hard Outages", i.e., an inability to exchange data between Customer end points. Slow service or other service degradation is not considered a Hard Outage.

Except as set forth in this SLA, Company makes no claims regarding the availability or performance of USPL service.

B. Access Required.

Local access service is required for USPL and is provided by MCI Legacy Company. The level of Service Availability and MTTR is dependent upon the access design and method used to deliver USPL Service. The term "Type" is used to label or categorize the various designs and methods used to provide local access. Local access types are as follows:

Type 1 circuits are those for which the local loop is furnished wholly via MCI Legacy Company facilities;

Type 2 circuits are those for which the local loop is furnished in part via MCI Legacy Company facilities; and,

Type 3 circuits are those for which the local loop is not furnished via MCI Legacy Company facilities, but are ordered and billed on Customer's behalf by Company.

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.):</u>
    - 1. VBS III U.S. Private Line Service Level Agreement (Cont'd.)
      - II. Service Level Standards
      - A. Service Availability
        - Service Availability Standard. The Service Availability Standard applies only to "Hard Outages", i.e., an inability to exchange data end to end. Service is not considered unavailable until at least 60 seconds of unavailability have elapsed during any 15 minute interval.

Service Type	Service Availability
Linear (excludes Linear DS3)	
Type 1, 2 or 3	99.99%
SONET	
Type 1	99.99%
Type 2 or 3	99.99% (IXC portion only)

2. Calculation: To calculate the percentage of Service Availability, add the total number of minutes that Customer's circuit was available to Customer to exchange data end-to-end during a billing period (i.e., not subject to a Hard Outage), plus any minutes during which the circuit was not available to exchange data between Customer end points, due to events described in Section V, Exclusions. Divide the sum by the total number of minutes in the billing month. Multiply the quotient by 100.

The total number of minutes in a billing month is as follows:

Month in Days	Total Minutes
31 Day Month	44,640
30 Day Month	43,200
29 Day Month	41,760
28 Day Month	40,320

- B. Mean Time to Repair ("MTTR")
  - MTTR Standard. The MTTR Standard applies only to "Hard Outages". Slow service or other service degradation is not considered a Hard Outage.

Service Type	MTTR (hours)
Linear (excluding Linear DS3)	
Type 1, 2 or 3	4
SONET	4

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.)</u>:
    - 1. VBS III U.S. Private Line Service Level Agreement (Cont'd.)
      - II. Service Level Standards (Cont'd.)
      - B. Mean Time to Repair ("MTTR") (Cont'd.)

Calculation: MTTR is the average time to repair all Hard Outages for a particular circuit that Customer has reported as described below over a calendar month period. The time to repair a circuit starts when a Trouble Ticket is opened by Company for a Hard Outage and concludes with restoration of the USPL circuit—i.e., the time that the Hard Outage condition no longer exists. The duration of each Hard Outage for a particular circuit is totaled at the end of each billing month and divided by the corresponding number of Hard Outage Trouble Tickets for that particular circuit.

#### C. Service Installation Period

Service Installation Period Standard and Calculation. The Service Installation Period Standard applies to speeds of up to 1.544 Mbps (i.e. T1 and below) but it does not apply to any individual case basis "(ICB") arrangements. Service Installation Period refers to the period of time to install the USPL circuit. The time period starts from the date that any prerequisite steps have been taken and Company has received and accepted all required information to provide and bill the circuit. The time period ends with successful test and Turn-up of the circuit.

Service Type	Service Installation (business days)*
Linear (excluding Linear DS3)	
Type1, 2 or 3	45
SONET	Not Available

\* If a national holiday is observed on a business day, such day is not considered a business day for purposes of the Service Installation Period Standard.

- 6. SERVICE LEVEL AGREEMENTS (CONT'D.):
  - 1. VBS III U.S. Private Line Service Level Agreement (Cont'd.)
    - III. Service Level Objective.

Verizon provides a network Latency service level objective of 10 milliseconds per 1,000 route miles. This service level objective applies only to the following USPL services: DS3, OC3, OC3c, OC12 and OC12c.

- IV. Qualification Process.
- A. General. Customer is eligible to receive credits if Company does not meet the Service Level Standards set above. To qualify for such credits, Customer must comply with the procedures set forth in this section. Credits are based on a stated percentage of the monthly recurring charge (MRC) for the service component in Customer's Agreement for which the Service Level Standard has not been met. Subject to the limitations set forth in section V. A. below, the maximum credit available to Customer under any combination of the Service Availability, Mean Time to Repair and Service Installation Period Standard within a single calendar month is 100 percent of the MRC for that service component for the month during which the Service Level Standard was not met. Company data and calculations will be used to determine if a Service Level Standard has not been met and a credit is due. In addition, Company will not issue credits pursuant to the Service Level Standard for more than 6 months in any 12 month period. Company will issue a credit due, net of discounts and taxes, within 90 days of its determination that a Service Level Standard was not met.
- B. Service Availability and MTTR Credit Process.

Customer must perform the following to become eligible or qualify for credit under either the Service Availability standard or the MTTR standard:

- Open a Trouble Ticket within four hours of first learning of a Hard Outage by using one of the following available methods:
  - a. Using the Customer Center portal on the Verizon Business website, or
  - b. Calling the Customer Service toll free number located on Customer's invoice, or
  - c. A trouble ticket is required to be opened within four hours of first learning of a Hard Outage; and
- Submit in writing the following information no later than thirty (30) days from the end of the calendar month in which the USPL service was restored:
  - a. All applicable Trouble Ticket numbers associated with each circuit experiencing Hard Outages;
  - b. The date and time the Trouble Ticket(s) were opened; and
  - c. The circuit ID number for each circuit that experienced the Hard Outage.

- 6. SERVICE LEVEL AGREEMENTS (CONT'D.):
  - 1. VBS III U.S. Private Line Service Level Agreement (Cont'd.)
    - IV. Qualification Process (Cont'd.).
    - B. Service Availability and MTTR Credit Process (Cont'd.)

Failure to comply with each of the preceding requirements may result in Company denying Customer's request for credit under either the Service Availability or MTTR Standard.

A Trouble Ticket is a Company document used to record network service issues. Trouble Tickets are time stamped when opened and closed. The time stamps are used to calculate whether (i) the Service Availability and MTTR standards under this SLA have been met, (ii) the issue reported is considered a Hard Outage claim; and (iii) the issue isn't excluded by the items listed in Section V below.

C. Service Installation Period Credit Process. To qualify for credit under the Service Installation Period Standard, Customer must make a written request for credit to Company within five business days of completed installation of the qualifying service. When making this credit request, the Customer must provide its Company Account Team with the following information: (1) the date the Service Installation Period commenced; (2) the date specified for installation in the Customer order; (3) the date installation was completed; (4) the circuit ID number for each circuit that was not installed within the Service Installation Period Standard; and, (5) the Customer's order number.

Company will use its order activation system(s) as the basis to determine if the Service Installation Period Service Level Standard has been met. In addition to the exclusions described in Section V. below, no credit is available under the Service Installation Period Standard for:

- A Customer-ordered installation date that is prior to the USPL installation interval;
- Installations outside of the continental United States;
- Delays resulting from inaccurate or incorrect order information;
- Customer order that is suspended due to Customer credit issues:
- Delays resulting from the unavailability of Customer's premises, Customer Premises Equipment (CPE) or facilities required to install.
- Delays resulting from changes to a previously-accepted service order by customer, its agents or vendors;
- Extended demarcation point of Customer's access circuit;
- Customer changes to a previously accepted service order;
- Installations in which any necessary prerequisite steps are not completed fully or successfully before USPL implementation begins.

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.):</u>
    - 1. VBS III U.S. Private Line Service Level Agreement (Cont'd.)
      - V. Credits
      - A. Service Availability and MTTR. During any monthly billing period in which Company fails to meet the Service Availability standard or MTTR standard for a specific Customer circuit, and provided the conditions set forth in this SLA have been met, Customer is eligible to receive a credit applied against the USPL MRC for that circuit, as follows. Access and backhaul charges are excluded from the USPL credit calculations and are not considered eligible for a credit:
        - For a failure to meet the Service Availability or MTTR standard for one month, Company will provide, as applicable, a Service Availability credit equal to 25 percent of the USPL MRC for that circuit, after application of all discounts.
        - For a failure to meet the Service Availability or MTTR standard for the same circuit for two consecutive months, Company will provide, as applicable, a Service Availability credit equal to 50 percent of the USPL MRC for that circuit, after application of all discounts, for the second month.
        - 3. For a failure to meet the Service Availability or MTTR standard for the same circuit for three consecutive months, either: (1) Company will provide a credit equal to 100 percent of the USPL MRC for that circuit, after application of all discounts, for the third month and for each consecutive month thereafter that Customer remains subscribed to the circuit and Company fails to satisfy the Service Availability or MTTR standard; or (2) Customer may terminate service for that circuit without incurring termination liability, except for charges incurred prior to service termination, if Customer notifies Company in writing of its intent to terminate the circuit no later than 30 days after the conclusion of the third month or any consecutive month in which the same Service Level Standard is not met for that circuit. After a failure of three consecutive months or more, Company, upon written notice to Customer, may terminate its performance obligations under this SLA.
        - If the Hard Outage carries over from one month to the next, the Hard Outage is deemed to have occurred in the month in which USPL circuit was restored.
        - 5. In addition, Company will not issue credits pursuant to this SLA for more than a total of six months in any 12-month period.
      - B. Service Installation Period. During any monthly billing period that Company fails to meet the Service Installation Period Standard, Company will provide a credit equal to 50 percent of the MRC, after application of all discounts.

- 6. SERVICE LEVEL AGREEMENTS (CONT'D.):
  - 1. VBS III U.S. Private Line Service Level Agreement (Cont'd.)
    - VI. **Exclusions**

In addition to any exclusions specified above, Service Level Standard calculations will not count as unavailable or repair time any time that results from any one or more of the following:

- Α. USPL Services that have been installed for less than one full calendar month;
- B. Any act or omission on the part of Customer, its contractors, vendors, agents or any other entity over which Customer exercises control or has the right to exercise control, other than acts or omissions of Company or Company approved 3<sup>rd</sup> party network or maintenance providers, including, but not limited to, the following:
  - Delays of any kind attributable to Customer;
  - Incorrect or incomplete information provided by Customer;

  - Periods during which Company or the Company-approved maintenance provider is denied access to the CPE or network components at the Customer location when such access is required;
  - Customers failure or refusal to release the circuit for testing:
  - Customer's unavailability when needed to close a Trouble Ticket;
  - Circuits where Customer (not Company) provides the local access
  - The manner in which Customer has used the circuit, including, but not limited to router configuration, power availability environmental conditions prevailing at Customer's location
  - Any third party equipment other than equipment furnished by LECs as part of the access services provisioned by Company via LEC facilities.
- C. Interruptions not reported by Customer, or for which no Trouble Ticket was opened, if required under Section III above;
- D. Proper electrical power is not available:
- E. A Force Majeure event, including but not limited to, acts of God, government regulation, labor strikes, natural disasters, national emergency, acts of terrorism, or other such events that are not reasonably within Company's control;
- F. Scheduled maintenance by Customer or entities under Customer's direction or control:
- G. Scheduled or emergency maintenance by Company;
- H. Customer's use of the USPL service in an unauthorized or unlawful manner;
- I. Network degradation, such as slow data transmission;

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

- 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.):</u>
  - 1. VBS III U.S. Private Line Service Level Agreement (Cont'd.)
    - VI. Exclusions (Cont'd.)
    - J. Service outages attributable to installation of a new circuit;
    - K. Delays resulting from order suspension due to credit issues involving Customer.
  - 2. <u>Pre-VBS III U.S. Private Line Service Level Agreement</u>
  - I. Scope
  - A. General.

This SLA applies to Company's U.S. Private Line (USPL) Service on an end-to-end basis, except where otherwise noted. Customer may receive credits for failure by Company to satisfy the "Service Availability", "Mean Time to Repair" or "Service Installation Period" standards as set forth below. This SLA is only available for customers who pay Verizon Business Service Pre-VBS I, VBS I and VBS II charges for USPL Service.

This SLA applies to the following USPL services: Linear services, including Analog, Digital Signal Level 0 (DS0), Fractional DS1, and Digital Signal Level 1 (DS1); and SONET Services, including Restorable DS3, STS-1, OC3, and OC3c, except except where noted elsewhere below. This SLA does not apply to Linear DS3 service or to any services made available on an ICB basis.

The Mean Time to Repair and Service Installation SLAs do not apply to SONET service. This SLA applies only to "Hard Outages", i.e., an inability to exchange data between Customer end points. Slow service or other service degradation is not considered a Hard Outage.

Except as set forth in this SLA, Company makes no claims regarding the availability or performance of USPL service.

Unless otherwise identified, the term SLA refers collectively to the following specific agreements set forth in this SLA:

Service Availability Mean Time to Repair Service Installation

- 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.)</u>:
  - 2. Pre-VBS III U.S. Private Line Service Level Agreement (Cont'd.)
  - I. Scope (Cont'd.)
  - B. Access Required.

Local access service is required for USPL and is provided by a MCI Legacy Company. The level of Service Availability and MTTR is dependent upon the access design and method used to deliver USPL Service. The term "Type" is used to label or categorize the various designs and methods used to provide local access. Local access types are as follows:

Type 1 circuits are those for which the local loop is furnished wholly via MCI Legacy Company facilities;

Type 2 circuits are those for which the local loop is furnished in part via MCI Legacy Company facilities; and,

Type 3 circuits are those for which the local loop is not furnished via MCI Legacy Company facilities, but are ordered and billed on Customer's behalf by Company

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.)</u>:
    - 2. <u>Pre-VBS III U.S. Private Line Service Level Agreement (Cont'd.)</u>
    - II. Service Level Agreements
    - A. Service Availability
      - 1. Service Availability Standards. The Service Availability SLA applies only to "Hard Outages", i.e., an inability to exchange data end to end. Service is not considered unavailable until at least 60 seconds of unavailability have elapsed during any 15 minute interval.

Service Type	Service Availability
Linear (excludes Linear DS3)	j
Type 1	100%
Type 2 or 3	99.8%
SONET	
Type 1	100%
Type 2 or 3	100% (IXC portion only)

2. Calculation: To calculate the percentage of Service Availability, add the total number of minutes that Customer's circuit was available to Customer to exchange data end-to-end during a billing period (i.e., not subject to a Hard Outage), plus any minutes during which the circuit was not available to exchange data between Customer end points, due to events described in Section V, Exclusions. Divide the sum by the total number of minutes in the billing month. Multiply the quotient by 100

The total number of minutes in a billing month is as follows:

Month in Days	Total Minutes
31 Day Month	44,640
30 Day Month	43,200
29 Day Month	41,760
28 Day Month	40,320

- B. Mean Time to Repair (MTTR)
  - 1. MTTR Standard. The Mean Time to Repair SLA applies only to "Hard Outages". Slow service or other service degradation is not considered a Hard Outage.

Service	MTTR
Type	(hours)
Linear (excluding Linear DS3)	
Type 1	2
Type 2 or 3	4
SONET	Not Available

- 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.):</u>
  - 2. Pre-VBS III U.S. Private Line Service Level Agreement (Cont'd.)
  - II. Service Level Agreements (Cont'd.)
  - B. Mean Time to Repair (MTTR) (Cont'd.)

Calculation: MTTR is the average time to repair all Hard Outages for a particular circuit that Customer has reported as described below over a calendar month period. The time to repair a circuit starts when a Trouble Ticket is opened by Company for a Hard Outage and concludes with restoration of the USPL circuit—i.e., the time that the Hard Outage condition no longer exists. The duration of each Hard Outage for a particular circuit is totaled at the end of each billing month and divided by the corresponding number of Hard Outage Trouble Tickets for that particular circuit.

#### C. Service Installation Period

Service Installation Period Standard and Calculation. The Service Installation Period SLA applies to speeds of up to 1.544 Mbps (i.e. T1 and below) but it does not apply to any individual case basis (ICB) arrangements. Service Installation Period refers to the period of time to install the USPL circuit. The time period starts from the date that any prerequisite steps have been taken and Company has received and accepted all required information to provide and bill the circuit. The time period ends with successful test and Turn-up of the circuit.

Service	Service
Туре	Installation
	(business days)*
Linear (excluding Linear DS3)	
Type 1	45
Type 2 or 3	45
SONET	Not Available

\* If a national holiday is observed on a business day, such day is not considered a business day for purposes of the Service Installation Period SLA.

- 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.):</u>
  - 2. Pre-VBS III U.S. Private Line Service Level Agreement (Cont'd.)
  - III. Qualification Process.
  - A. General. Customer is eligible to receive credits if Company does not meet the SLA standards set above. To qualify for such credits, Customer must comply with the procedures set forth in this section. Credits are based on a stated percentage of the monthly recurring charge (MRC) for the service component in Customer's Agreement for which the SLA standard has not been met. Subject to the limitations set forth in section IV. A. below, the maximum credit available to Customer under any combination of the Service Availability, Mean Time to Repair and Service Installation Period SLAs within a single calendar month is 100 percent of the MRC for that service component for the month during which the SLA standard was not met. Company data and calculations will be used to determine if an SLA standard has been missed and a credit is due. In addition, Company will not issue credits pursuant to the SLA for more than 6 months in any 12 month period. Company will issue a credit due, net of discounts and taxes, within 90 days of its determination that an SLA standard was not met.
  - B. Service Availability and MTTR Credit Process.

Customer must perform the following to become eligible for credit under either the Service Availability standard or the MTTR standard: Customer must perform the following to qualify for credit under either the Service Availability or the MTTR standard:

- Open a Trouble Ticket within four hours of first learning of a Hard Outage by using one of the following available methods:
  - Using the Customer Center portal on the Verizon Business website, or
  - Calling the Customer Service toll free number located on Customer's invoice, or

A trouble ticket is required to be opened within four hours of first learning of a Hard Outage; and

- 2 Submit in writing the following information no later than thirty (30) days from the end of the calendar month in which the USPL service was restored:
  - All applicable Trouble Ticket numbers associated with each circuit experiencing Hard Outages;
  - The date and time the Trouble Ticket(s) were opened; and
  - The circuit ID number for each circuit that experienced the Hard Outage.

Failure to comply with each of the preceding requirements may result in Company denying Customer's request for credit under either the Service Availability or MTTR SLA.

A Trouble Ticket is a Company document used to record network service issues. Trouble Tickets are time stamped when opened and closed. The time stamps are used to calculate whether (i) the Service Availability and MTTR standards under this SLA have been met, (ii) the issue reported is considered a Hard Outage claim; and (iii) the issue isn't excluded by the items listed in Section V below.

- 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.):</u>
  - 2. <u>Pre-VBS III U.S. Private Line Service Level Agreement (Cont'd.)</u>
  - III. Qualification Process (Cont'd.)
  - C. Service Installation Period Credit Process. To qualify for credit under the Service Installation Period SLA, Customer must make a written request for credit to Company within five business days of completed installation of the qualifying service. When making this credit request, the Customer must provide its Company Account Team with the following information: (1) the date the Service Installation Period commenced; (2) the date specified for installation in the Customer order; (3) the date installation was completed; (4) the circuit ID number for each circuit that was not installed within the SLA; and, (5) the Customer's order number.

Company will use its order activation system(s) as the basis to determine if the Service Installation Period SLA standard has been met. In addition to the exclusions described in Section V. below, no credit is available under the Service Installation Period SLA for:

- A Customer-ordered installation date that is prior to the USPL installation interval;
- Installations outside of the continental United States;
- Delays resulting from inaccurate or incorrect order information;
- Customer order that is suspended due to Customer credit issues;
- Delays resulting from the unavailability of Customer's premises, Customer Premises Equipment (CPE) or facilities required to install.
- Delays resulting from changes to a previously-accepted service order by customer, its agents or vendors;
- Extended demarcation point of Customer's access circuit;
- Customer changes to a previously accepted service order:
- Installations in which any necessary prerequisite steps are not completed fully or successfully before USPL implementation begins.

- IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)
  - 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.):</u>
    - 2. <u>Pre-VBS III U.S. Private Line Service Level Agreement (Cont'd.)</u>
    - IV. Credits
    - A. Service Availability and MTTR. During any monthly billing period in which Company fails to meet the Service Availability standard or MTTR standard for a specific Customer circuit, and provided the conditions set forth in this SLA have been met, Customer is eligible to receive a credit applied against the USPL MRC for that circuit, as follows. Access and backhaul charges are excluded from the USPL credit calculations and are not considered eligible for a credit:
      - 1. For a failure to meet the Service Availability or MTTR standard for one month, Company will provide, as applicable, a Service Availability credit equal to 25 percent of the USPL MRC for that circuit, after application of all discounts.
      - 2. For a failure to meet the Service Availability or MTTR standard for the same circuit for two consecutive months, Company will provide, as applicable, a Service Availability credit equal to 50 percent of the USPL MRC for that circuit, after application of all discounts, for the second month.
      - 3. For a failure to meet the Service Availability or MTTR standard for the same circuit for three consecutive months, either: (1) Company will provide a credit equal to 100 percent of the USPL MRC for that circuit, after application of all discounts, for the third month and for each consecutive month thereafter that Customer remains subscribed to the circuit and Company fails to satisfy the Service Availability or MTTR standard; or (2) Customer may terminate service for that circuit without incurring termination liability, except for charges incurred prior to service termination, if Customer notifies Company in writing of its intent to terminate the circuit no later than 30 days after the conclusion of the third month or any consecutive month in which the same SLA standard is not met for that circuit. After a failure of three consecutive months or more, Company, upon written notice to Customer, may terminate its performance obligations under this SLA.
      - 4. If the Hard Outage carries over from one month to the next, the Hard Outage is deemed to have occurred in the month in which USPL circuit was restored.
      - 5. In addition, Company will not issue credits pursuant to this SLA for more than a total of six months in any 12-month period.
    - B. Service Installation Period. During any monthly billing period that Company fails to meet the Service Installation Period SLA, Company will provide a credit equal to 50 percent of the MRC, after application of all discounts.

# IX. INTRASTATE U.S. PRIVATE LINE SERVICES (CONT'D.)

- 6. <u>SERVICE LEVEL AGREEMENTS (CONT'D.):</u>
  - 2. <u>Pre-VBS III U.S. Private Line Service Level Agreement (Cont'd.)</u>
  - V. Exclusions

In addition to any exclusions specified above, SLA standard calculations will not count as unavailable or repair time any time that results from any one or more of the following:

- A. USPL Services that have been installed for less than one full calendar month;
- B. Any act or omission on the part of Customer, its contractors, vendors, agents or any other entity over which Customer exercises control or has the right to exercise control, other than acts or omissions of Company or Company approved 3rd party network or maintenance providers, including, but not limited to, the following:
  - Delays of any kind attributable to Customer;
  - Incorrect or incomplete information provided by Customer;
  - CPE
  - Periods during which Company or the Company-approved maintenance provider is denied access to the CPE or network components at the Customer location when such access is required;
  - Customers failure or refusal to release the circuit for testing;
  - Customer's unavailability when needed to close a Trouble Ticket;
  - Circuits where Customer (not Company) provides the local access loop;
  - the manner in which Customer has used the circuit, including, but not limited to router configuration, power availability and environmental conditions prevailing at Customer's location
  - Any third party equipment other than equipment furnished by LECs as part of the access services provisioned by Company via LEC facilities.
- C. Interruptions not reported by Customer, or for which no Trouble Ticket was opened, if required under Section III. above;
- D. Proper electrical power is not available:
- E. A Force Majeure event, including but not limited to, acts of God, government regulation, labor strikes, natural disasters, national emergency, acts of terrorism, or other such events that are not reasonably within Company's control;
- F. Scheduled maintenance by Customer or entities under Customer's direction or control;
- G. Scheduled or emergency maintenance by Company;
- H. Customer's use of the USPL service in an unauthorized or unlawful manner;
- I. Network degradation, such as slow data transmission;
- J. Service outages attributable to installation of a new circuit;
- K. Delays resulting from order suspension due to credit issues involving Customer.

# X. STATE SPECIFIC PLANS AND PROGRAMS

### 1. <u>Verizon Business Service 90 Day Satisfaction Guarantee</u>

Offer: Subject to the Conditions below, a Customer signing a new Verizon Business service agreement ("Agreement") may terminate such Agreement (and all services under it) at any time within 90 days of the Agreement Effective Date by providing written notice of termination to Verizon within that 90-day period.

#### **Eligibility**: Customer:

- must sign the Agreement including this promotion, and submit it to Company; and,
- may not have had any Verizon billing for services received within the 90 days before signing the Agreement.

#### Other Conditions:

Customer must provide Company with at least 30 days written notice in advance of their requested date of termination in accordance with the Notice provision in the Agreement (longer notice is required for international service). Company will implement that termination within 60 days of the date it receives Customer notification.

Customers who terminate service by invoking this 90 Day Satisfaction Guarantee will be required to repay all credits, including installation credits, received up to the service termination date, as well as the pro-rata value of any promotional benefits for which Customer has not met the full requirements for those benefits (including without limitation any credits already received or charges waived).

Customer will have no obligation to fulfill any Annual Volume Commitment applicable under the Agreement.

#### X. STATE SPECIFIC PLANS AND PROGRAMS

#### 2. Verizon Business Service Install Guarantee

Offer: Subject to the Conditions below, a Customer signing a new Verizon Business service agreement ("Agreement") is eligible to receive a credit if Company fails to install an eligible service ordered under the Agreement and this tariff so that it is available to Customer for use on or before the date Company has told Customer it will be installed and available for Customer use ("Late Installation"). The credit amount will equal the amount paid by Customer for the installation of the service subject to the Late Installation, and will be applied against charges for interstate service under the Agreement, excluding third-party charges, pass-through charges and expedite charges.

### Eligibility:

- Customer must execute the Agreement including this plan, and submit it to Company;
- Customer must successfully submit a completed Installation Commitment Submission Form, using the online process established by Company for this purpose (https://customercenter.verizon.com/installguarantee), within 30 days of the date Company has told the Customer the service will be installed and available for Customer use;
- the Late Installation must not result from a Customer change to an order or any other Customer act or omission;
- eligible services must be provided and located entirely in the U.S. Mainland; and
- eligible services are those provided under the Agreement by an MCI Legacy Company.

### Other Conditions:

Customer may not receive any discounts based on term and volume commitments, excluding affinity program discounts, or the benefits of a Special Customer Arrangement (SCA) or Product Package, other than Product Packages Guide Types 13, 14, 15, 16, 18, 19, 20, 21, 22 and 23 as provided in the "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com/publications/service\_guide.

The credit amount is based on Company installation charges only. Vendor, LEC or other third-party installation charges are not included in the credit amount.

The credit may only be applied against invoices for services provided, under this Agreement, by MCI Legacy Company.

#### 3. Regional Checkbook 2006 Monthly Option Plan

Offer: Subject to the Conditions below, a Customer signing a new Verizon Business service agreement ("Agreement"), Customer will receive a monthly credit, not to exceed \$100,000 per Customer, equal to 10 percent of the Customer's Total Volume Commitment (defined as the Annual Volume Commitment multiplied by the number of years in the initial Term) of the Agreement (the "Checkbook Credit").

Customer will receive the first credit in the 3rd month following the Effective Date of the Agreement. The amount received will also include the credit amounts for months 1 and 2. The remaining credit amount will be divided into equal amounts for the remaining months of the initial Term (months 4 through 12 for a one year Term; months 4 through 24 for a two year Term and months 4 through 36 for a 3, 4, or 5 year Term).

#### Eligibility: Customer must:

- enroll in this Regional Checkbook 2006 Monthly Option Plan in a signed Agreement with the Company;
- execute a contract after April 18, 2006 with a minimum one-year Term under which Customer subscribes to one or more of the following MCI Legacy Company-provided services: U.S. Private Line Services, Verizon Business Services (VBS) II and VBS III Long Distance Voice Services, and/or Local Service-CLEC service; and,
- demonstrate to the Company's reasonable satisfaction that it will accept a competitor's offer in the absence of the Company providing the Regional Checkbook 2006 Monthly Option Plan.

#### Other Conditions:

Customer may not receive (i) any discounts or (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, (other than Product Package Guide Types 18, 19, 20, 21, 22 and 23) as provided in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="www.verizonbusiness.com/publications/service\_guide">www.verizonbusiness.com/publications/service\_guide</a>; (Companion Interstate Service"); Checkbook Promotion/Plan 2004; Regional Checkbook Promotion/Plan 2004; and Checkbook 2006 Monthly Option Promotion/Plan.

The Checkbook credit may not be applied against taxes, charges for unauthorized calls, prior outstanding balances owed to Company, termination or underutilization charges associated with term plans or program commitments, or disputed charges.

If Customer terminates the Agreement prior to the month the next credit is to be applied, Customer will not be eligible for that month's credit and any unused credit amount at the time of termination is forfeited.

The maximum cumulative credit that a Customer may receive under this promotion is \$100,000.

The credit may only be applied against invoices for services provided, under this Agreement, by an MCI Legacy Company.

#### 4. Checkbook 2006 Monthly Option Plan

Offer: Subject to the Conditions below, a Customer signing a new Verizon Business service agreement ("Agreement"), Customer will receive a monthly credit, not to exceed \$100,000 per Customer, equal to 5 percent of the Customer's Total Volume Commitment (defined as the Annual Volume Commitment multiplied by the number of years in the initial Term) of the Agreement (the "Checkbook Credit").

Customer will receive the first credit in the 3rd month following the Effective Date of the Agreement. The amount received will also include the credit amounts for months 1 and 2. The remaining credit amount will be divided into equal amounts for the remaining months of the initial Term (months 4 through 12 for a one year Term; months 4 through 24 for a two year Term and months 4 through 36 for a 3, 4, or 5 year Term).

#### Eligibility: Customer must:

- enroll in this Checkbook 2006 Monthly Option Plan in a signed Agreement with the Company;
- execute a contract after April 18, 2006 with a minimum one-year Term under which Customer subscribes to one or more of the following MCI Legacy Company-provided services: U.S. Private Line Services, Verizon Business Services (VBS) II and VBS III Long Distance Voice Services, and/or Local Service-CLEC service; and,
- demonstrate to the Company's reasonable satisfaction that it will accept a competitor's offer in the absence of the Company providing the Checkbook 2006 Monthly Option Plan.

#### Other Conditions:

Customer may not receive (i) any discounts or (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, (other than Product Package Guide Types 18, 19, 20, 21, 22 and 23) as provided in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="www.verizonbusiness.com/publications/service\_guide">www.verizonbusiness.com/publications/service\_guide</a>; (Companion Interstate Service"); Checkbook Promotion/Plan 2004; Regional Checkbook Promotion/Plan 2004; and Regional Checkbook 2006 Monthly Option Promotion/Plan.

The Checkbook credit may not be applied against taxes, charges for unauthorized calls, prior outstanding balances owed to Company, termination or underutilization charges associated with term plans or program commitments, or disputed charges.

If Customer terminates the Agreement prior to the month the next credit is to be applied, Customer will not be eligible for that month's credit and any unused credit amount at the time of termination is forfeited.

The credit may only be applied against invoices for services provided, under this Agreement, by an MCI Legacy Company.

#### 5. Verizon Business Services Billing Guarantee Plan

Offer: Subject to the Conditions below, a Customer signing a new Verizon Business service agreement ("Agreement") including an Eligible Service (defined below) will be eligible to receive a credit if, with respect to an Eligible Service, Company fails to do either of the following:

- (i) Respond to a Customer billing inquiry by sending Customer a Resolution Letter (defined below) addressing that inquiry within 45 days of the date Customer submits a Billing Inquiry Form (defined below); and/or,
- (ii) Provide Customer with an invoice that reflects the result of that resolution within the first two invoices following the date of the Resolution Letter:

The amount of the credit that a Customer can receive is indicated in the table below, based on the Annual Volume Commitment in Customer's Agreement.

Annual Volume Commitment	Credit
Less than \$60,000	\$ 500
\$60,000 - \$300,000	1,250
\$300,001 - \$600,000	2,500
Greater than \$600,000	5,000

<u>Definitions</u>: For purposes of this plan offering only, the following definitions apply:

Billing Accuracy Response: Company's written confirmation to Customer of receipt of a Billing Inquiry Form.

Billing Issue: An invoiced charge for MCI Legacy Company-provided service which Customer alleges to differ from the rate specified in Customer's written agreement for that service and which is not in the Customer's favor.

Eligible Service:,Long Distance Voice Services-CLEC and U.S. Private Line Services.

Resolution Letter: Company's written notice to Customer in response to a Billing Inquiry Form submission which specifies Company's determination, in the Company's sole discretion, of the rate set forth in Customer's written agreement for the charge which is the basis of Customer's billing Issue

#### Eligibility:

- Customer must successfully submit a completed Billing Inquiry Form, using the online process established by Company for this purpose (https://customercenter.verizon.com/billingguarantee), within 60 days of the date of the affected invoice, concerning invoiced charges that differ adversely from the charges due under Customer's Agreement and applicable tariffs; and

eligible services must be provided and located entirely in the U.S. Mainland.

5. <u>Verizon Business Services Billing Guarantee Plan</u> (Cont'd)

In order for a Billing Issue to be eligible:

Customer must notify the Company of the Billing Issue via a completed Billing Inquiry Form, which is found at a Company-designated Internet site, and which must be received by the Company within 60 days of the date of the invoice on which the Billing Issue appears;

the rate or charge which is the basis of a Billing Issue must be a charge for an Eligible Service; and,

the rate or charge which is the basis of a Billing Issue may not be, in the Company's sole determination:

- the subject of another Customer Billing Inquiry Form;
- a pass-through charge or a charge imposed by a third party charges, tax, or Governmental charge or surcharge or the subject of a claim of fraud; and/or
- one for which Customer has received an alternative credit or other credit to resolve the Billing Issue.

The Billing Inquiry Form must specify a single Billing Issue and:

the Customer's Company account number; the date of the invoice containing the Billing Issue; the service type of the Billing Issue; and, a description of the Billing Issue.

Customer must supply to Company any additional information requested by Company within three business days of the request.

#### Other Requirements:

Customer may not receive more than one credit for any and all invoices dated in the same month, regardless of the number of its agreements, services, or billing inquiries. If a similar issue arises in a subsequent month, Customer may submit a new Billing Inquiry Form for that subsequent month (subject to the one-credit-per-month limitation stated in the preceding sentence).

Company will determine whether any credit is due, the credit amount, and the account level at which the credit will be applied, in its sole discretion.

If the Agreement's Initial Term expires, or if Customer terminates the Agreement or the service to which the Billing Inquiry Form relates prior to the month the credit is to be applied, Customer will not be eligible for the credit and any unused credit amount at the time of termination of service will be forfeited by the Customer.

5. <u>Verizon Business Services Billing Guarantee Plan</u> (Cont'd)

The credit may only be applied against invoices for services provided, under this Agreement, by MCI Legacy Company.

Customer may not receive any discounts based on term and volume commitments, excluding affinity program discounts, or the benefits of a Special Customer Arrangement (SCA) or Product Package, other than Product Packages Guide Types 13, 14, 15, 16, 18, 19, 20, 21, 22 and 23 on Eligible Service as provided in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="https://www.verizonbusiness.com/publications/service\_guide">www.verizonbusiness.com/publications/service\_guide</a>; (Companion Interstate Service").

#### 6. Checkbook Plan 2004

Offer: Subject to the Conditions below, a Customer signing a new Verizon Business service agreement ("Agreement"), Customer will receive a credit, not to exceed \$100,000 per Customer, equal to 5% of the Total Volume Commitment (defined as the Annual Volume Commitment multiplied by the number of years in the initial Term) of the Agreement (the "Checkbook Credit") which Customer will receive as: i) a credit on an invoice or ii) a one-time deposit to the Customer's Fund account, depending on the version of this plan that is in the Agreement.

<u>Invoice Credit:</u> For Customers who elect to receive the credit as an invoice credit, the credit will be applied:

- in the sixth monthly period following the Effective Date of the Agreement for a Customer who commits to a 1-year Agreement Term ,
- in an amount which equals 50 percent of the credit amount in each of the sixth and eighteenth monthly periods following the Effective Date of the Agreement for a Customer who commits to a 2-year Agreement Term , and
- in an amount which equals 33.33 percent of the credit amount in each of the sixth, eighteenth and thirtieth monthly periods following the Effective Date of the Agreement for a Customer who commits to an Agreement Term which equals or exceeds three years.

<u>Fund Deposit:</u> For Customers who elect to receive the credit as a Fund deposit, the credit will be applied within the first two billing cycles following the Effective Date of the Agreement. Fund benefits are not transferable. Any and all tax liabilities and shipping costs arising from participation in the Fund are solely the responsibility of Customer. Company is not liable for products, services, and warranties, express or implied, of participating vendors. The Customer may convert its Fund account balance to invoice credits, which will be applied on a pro-rata basis to Customer's first invoice following the end of the annual period in which the Customer makes such request and in each subsequent 12 month period of the customer's Term. Fund deposits earned by Customer as a result of signing this Agreement are not renewable under this Agreement.

6. <u>Checkbook Plan 2004</u> (Cont'd)

# Eligibility: Customer must:

- enroll in this Checkbook Plan 2004 in a signed Agreement with the Company;
- execute a contract for service after December 1, 2003 under which Customer subscribes under term of service which equals or exceeds one year to U.S. Private Line Services, Verizon Business Services (VBS) II and VBS III Long Distance Voice Services, and/or MCI Legacy Company-provided exchange service; and,
- demonstrate to the Company's reasonable satisfaction that it will accept a competitor's offer in the absence of the Company providing the Checkbook Plan 2004.

#### Other Conditions:

Customer may not receive (i) any discounts or (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Product Package Guide Types 13, 14, 15, 16, 18, 19, 20, 21, 22 and 23 as provided in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="https://www.verizonbusiness.com/publications/service\_guide">www.verizonbusiness.com/publications/service\_guide</a>; (Companion Interstate Service").; and, Regional Checkbook Promotion 2004.

The Checkbook credit may not be applied against taxes, charges for unauthorized calls, prior outstanding balances owed to Company, termination or underutilization charges associated with term plans or program commitments, or disputed charges.

If Customer terminates all services under the Agreement prior to the month a credit is to be applied, the Customer will not be eligible to receive the credit and any unapplied credit amount at the time of termination of service will be forfeited by Customer. Standard Fund provisions apply.

#### 7. Regional Checkbook Plan 2004

Offer: Subject to the Conditions below, a Customer signing a new Verizon Business service agreement ("Agreement"), Customer will receive a credit, not to exceed \$100,000 per Customer, equal to 10% of the Total Volume Commitment (defined as the Annual Volume Commitment multiplied by the number of years in the initial Term) of the Agreement (the "Checkbook Credit") which Customer will receive as: i) a credit on an invoice or ii) a one-time deposit to the Customer's Fund account, depending on the version of this promotion that is in the Agreement.

Invoice Credit: For Customers who elect to receive the credit as an invoice credit, the credit will be applied:

- in the sixth monthly period following the Effective Date (as defined in the Agreement) of Customer's Agreement for a Customer who commits to a 1-year Agreement Term,
- in an amount which equals 50 percent of the credit amount in each of the sixth and eighteenth monthly periods following the Effective Date of the Agreement for a Customer who commits to a 2-year Agreement Term, and
- in an amount which equals 33.33 percent of the credit amount in each of the sixth. eighteenth and thirtieth monthly periods following the Effective Date of the Agreement for a Customer who commits to an Agreement Term which equals or exceeds three years.

Fund Deposit: For Customers who elect to receive the credit as a Fund deposit, the credit will be applied within the first two billing cycles following the Effective Date of the Agreement. Fund benefits are not transferable. Any and all tax liabilities and shipping costs arising from participation in the Fund are solely the responsibility of Customer. Company is not liable for products, services, and warranties, express or implied, of participating vendors. The Customer may convert its Fund account balance to invoice credits, which will be applied on a pro-rata basis to Customer's first invoice following the end of the annual period in which the Customer makes such request and in each subsequent 12 month period of the customer's Term. Fund deposits earned by Customer as a result of signing this Agreement are not renewable under this Agreement.

### Eligibility: Customer must:

- enroll in this Regional Checkbook Plan 2004 in a signed Agreement with the Company;
- execute a contract for MCI Legacy Company-provided service after December 1, 2003 under which Customer subscribes under term of service which equals or exceeds one year U.S. Private Line Services, Verizon Business Services (VBS) II and VBS III Long Distance Voice Services, and/or MCI Legacy Company-provided exchange service; and,
- demonstrate to the Company's reasonable satisfaction that it will accept a competitor's offer in the absence of the Company providing the Regional Checkbook Plan 2004 Plan.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

### 7. Regional Checkbook Plan 2004 (Cont'd)

#### Other Conditions:

Customer may not receive (i) any discounts or (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Product Package Guide Types 13, 14, 15, 16, 18, 19, 20, 21, 22 and 23 as provided in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="https://www.verizonbusiness.com/publications/service\_guide">www.verizonbusiness.com/publications/service\_guide</a>; (Companion Interstate Service"); and Checkbook Promotion/Plan 2004.

The Checkbook credit may not be applied against taxes, charges for unauthorized calls, prior outstanding balances owed to Company, termination or underutilization charges associated with term plans or program commitments, or disputed charges.

If Customer terminates all services under the Agreement prior to the month the credit is to be applied, the Customer will not be eligible to receive the credit and any unapplied credit amount at the time of termination of service will be forfeited by Customer. Standard Fund provisions apply

The credit may only be applied against invoices for services provided, under this Agreement, by MCI Legacy Company.

### 8. <u>Verizon Business Services III Optional Calling Plan I</u>

To receive the following benefits of Verizon Business Services III Option Calling Plan I, Customer must demonstrate to Verizon reasonable satisfaction that it will accept a competitor's offer of service in place of Verizon Business Services III if it does not receive the benefits of this plan. Customer may not receive any other rates, discounts or other benefits applicable to Verizon Business Services III.

#### **OUTBOUND:**

Origination Termin	ation			
<u>Type</u> <u>Type</u>	<u>1</u>	YEAR TERM	2 YEAR TERM	3 YEAR TERM
<b>Local Network Connection</b>	<b>Local Network Connection</b>		\$0.0360	\$0.0350
\$0.0345				
<b>Local Network Connection</b>	Dedicated	\$0.0360	\$0.0350	\$0.0345
<b>Local Network Connection</b>	Switched	\$0.0360	\$0.0350	\$0.0345
Dedicated	<b>Local Network Connection</b>		\$0.0360	\$0.0350
\$0.0345				
Dedicated	Dedicated or Switched	\$0.0360	\$0.0350	\$0.0345
Switched/Card	<b>Local Network Connection</b>	\$0.0750	\$0.0740	\$0.0730
Switched/Card	Dedicated	\$0.0750	\$0.0740	\$0.0730
Switched/Card	Switched	\$0.0750	\$0.0740	\$0.0730

### **INBOUND:**

Termination			
<u>Type</u>	1 YEAR	TERM2 YEAR TERM	3 YEAR TERM
<b>Local Network Connection</b>	\$0.0360	\$0.0350	\$0.0345
Dedicated	\$0.0360	\$0.0350	\$0.0345
Switched	\$0.0750	\$0.0740	\$0.0730
<b>Local Network Connection</b>	\$0.0360	\$0.0350	\$0.0345
Dedicated	\$0.0360	\$0.0350	\$0.0345
Switched	\$0.0750	\$0.0740	\$0.0730
	Type Local Network Connection Dedicated Switched Local Network Connection Dedicated	Type 1 YEAR Local Network Connection \$0.0360 Dedicated \$0.0360 Switched \$0.0750 Local Network Connection \$0.0360 Dedicated \$0.0360	Type         1 YEAR TERM 2 YEAR TERM           Local Network Connection         \$0.0360         \$0.0350           Dedicated         \$0.0360         \$0.0350           Switched         \$0.0750         \$0.0740           Local Network Connection         \$0.0360         \$0.0350           Dedicated         \$0.0360         \$0.0350

#### XI. **CURRENTLY OFFERED PROMOTIONS:**

#### 1. LD Voice-Combo Benefit Promotion

Offer: Verizon ILEC Local Customers contracting for Long Distance Voice Services provided by MCI Communications Services, Inc. d/b/a Verizon Business Services ("LD Service") for the first time will receive a credit (equivalent to two months Outbound LD Service usage, as described further below) in each succeeding 12-month period (not to exceed a total of three credits) following the date on which the LD Service pricing becomes effective, during the Term of its master agreement. The credit will be provided in accordance with the following table, with the month in which the LD Service pricing becomes effective as Month 1, and will be applied to Outbound LD Service usage charges (regardless of local carrier). The amount of the credit is equal to two times the Customer's average monthly Intra-LATA toll, intrastate and interstate outbound LD Service per-minute usage charges for the 3-month period specified in the following table:

> 3-Month Period for Calculating Average Outbound LD Service Monthly Per-Minute Usage, By Billing Platform

Month in which	All Billing Platforms	
Credit is Provided	(except the IXPlus platform)	IXPlus Billing Platform
Month 9	Months 6 through 8	Months 5 through 7
Month 21	Months 18 through 20	Months 17 through 19
Month 33	Months 30 through 32	Months 29 through 31
	Credit is Provided Month 9 Month 21	Credit is Provided (except the IXPlus platform) Month 9 Months 6 through 8 Month 21 Months 18 through 20

# Eligibility: The Customer must:

NA - - Ale de la la la la

- enroll in this promotion between June 1, 2008 and September 30, 2008;
- have new or existing Verizon ILEC Local Service;
- have signed and submitted a contract for Outbound LD Service by October 31. 2008: and
- request installation of new Outbound LD Service by November 30, 2008, to occur no later than December 31, 2008.

Other Conditions: Customer may not receive the benefits of Product Package Guide Type 1.

Notwithstanding the foregoing, if Customer terminates all Outbound LD Service upon which the credits under this promotion are calculated before all of the credits under this promotion are issued to Customer, any unissued credits are forfeited and not owed.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

# XI. CURRENTLY OFFERED PROMOTIONS:

# 2. LD Voice-Outbound 60 Promotion

Offer: Company Customers contracting for Long Distance Voice Services (LD Service) provided by MCI Communications Services, Inc. d/b/a Verizon Business Services for the first time will receive a credit (equal to two months Outbound LD Service usage, as described further below) in each succeeding 12-month period (not to exceed a total of three credits) following the date on which the LD Service pricing becomes effective for such Customer, during the Term of its master agreement. The credit will be provided in accordance with the following table, with the month in which the LD Service pricing becomes effective as Month 1, and will be applied to Outbound LD Service usage charges (regardless of local carrier). The amount of the credit is equal to two times the Customer's average monthly Intra-LATA toll, intrastate and interstate Outbound LD Service per-minute usage charges for the 3-month period specified in the following table:

12-Month Period	3-Month Period for Calculating Average Outbound LD Service Monthly Per-Minute Usage	Month in which Credit is Provided
Months 1 – 12	Months 6 through 8	Month 9
Months 13 – 24	Months 18 through 20	Month 21
Months 25 – 36	Months 30 through 32	Month 33

<u>Eligibility</u>: The Customer must sign and submit a service agreement (or amendment) for Outbound LD Service including this promotion between October 1, 2008 and March 31, 2009.

#### Other Conditions:

A Customer receiving the benefits of this promotion may not receive the benefits of Product Package Guide Type 1 and LD Voice Interstate 30 Promotion as provided in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="https://www.verizonbusiness.com/publications/service\_guide">www.verizonbusiness.com/publications/service\_guide</a> (Companion Interstate Service").

- This promotion is applicable only for Outbound LD Service within the 48 contiguous states with
  - the exception of intrastate usage in South Dakota, to which this promotion is not applicable.
- Existing Customers must have a minimum of 33 months remaining in the term of their agreement to receive 3 credits; 21 months remaining to receive 2 credits and at least 9 months remaining to receive 1 credit.

Notwithstanding the foregoing, if Customer terminates all Outbound LD Service upon which the credits under this promotion are calculated before all of the credits under this promotion are issued

to Customer, any unissued credits are forfeited and not owed.

MCI Communications Services, In	IC.
d/b/a Verizon Business Services	

FL PSC Tariff No. 6 Original Sheet No. 266

XII.	<b>EXPIRED</b>	DDOMO:	TIONIC.
AII.	EXPIRED	PRUMU	HUNNS:

Intentionally left blank.

### XIII. GRANDFATHERED SERVICES:

The following provides services and pricing programs which are no longer available to new enterprise business customers (i.e., non-mass markets) and are not otherwise described in other portions of this tariff. All of these grandfathered services are subject to the same terms and conditions applicable to the other services as set forth in this tariff.

Gaps in the numbering of sections indicate sections which are intentionally left blank. To a large extent, this is due to the desire to have the sections below match the section numbers to these services which previously were in Company tariff Nos. 3 and 5 to the extent applicable. The materials in Subsection A below were previously in tariff No. 3. The materials in Subsection B were previously in tariff No. 5.

Subsection A – Material previously in Company tariff No. 3

#### 3.5.4 Option D (MCI Prism I)

Option D is a one-way, multipoint service requiring the customer to originate calls via dedicated facilities between the customer's premises and MCI's terminal location and allowing the termination of calls via a combination of MCI-provided Interexchange facilities and local business telephone lines. The customer must arrange for the originating dedicated facilities between the customer's premises and MCI's terminal. Prism I has two pricing Ranges for total coverage on every line. All Option D calls are subject to one minute average connect time (i.e. total monthly minutes of use divided by total monthly calls must equal at least one minute). All Option D calls are rounded to the next higher 6-second increment. The usage charges for Option D are located in Section 4.1.4.

### 3.5.5 Option E (MCI Prism II)

Option E is a one-way, multipoint service requiring the customer to originate calls via dedicated facilities between the customer's premises and MCI's terminal location and allowing the termination of calls via a combination of MCI-provided Interexchange facilities and local business telephone lines. The customer must arrange for the originating dedicated facilities between the customer's premises and MCI's terminal. Prism II has two pricing Ranges for total coverage on every line. All Option E calls are subject to a one minute average connection time (i.e. total monthly minutes of use divided by total monthly calls must equal at least one minute). All Option E calls are rounded to the next higher 6-second increment. The usage charges for Option E are located in Section 4.1.5.

### 3.5.8 Option H (MCI 800 Service)

MCI 800 Service is an inward WATS service which permits interstate and intrastate calls to a customer's station in one location from stations in diverse geographical service areas, and in which the MCI 800 customer is billed for the calls rather than the call originators.

All Option H calls will be subject to an 18-second minimum initial period and additional 6-second increments. A customer of MCI 800 Service must choose between Switched WATS, Dedicated Access, and Business Lines for all lines for each 800 service terminating service group. All Option H calls are rounded to the next higher 6-second increment.

Unless otherwise indicated herein, the terms "MCI 800 Service," "888 Service," and/or "Toll Free Service"; and the terms "800 number," "888 number," and/or "Toll Free number" can be used interchangeably when used throughout this tariff.

In addition to the Usage Charges set forth in Section 4.1.8 following, charges for service options and special features will be billed as set forth under <a href="http://www.mci.com/service">http://www.mci.com/service</a>.

# 3.5 METERED USE SERVICE OPTIONS (Cont.)

#### 3.5.9 Option I (Vnet Service)

Vnet Service is a customized virtual network telecommunications service. It provides a unified service for multi-location companies, providing both communications and management features. Vnet is suitable for long distance calls between company locations, as well as all long distance calls to the entire U.S. and to MCI-served international locations.

Subscribers to Vnet take a Basic Feature Package under MCI's F.C.C. Tariff No. 1, which includes the availability of Dedicated and Shared Access and Termination; Usage Rates with associated Volume Discounts; international calling to MCI-service countries; Long Distance Directory Assistance; Universal Range Privilege; plus customer-defined private dialing plans, and a consolidated invoice for all Vnet calls to all locations. Additional optional features, which include Customer Information Management Access, Direct Termination Overflow, ID/Codes, Call Detail Tapes, Customized Range Privileges and Network Information Management System Access are also available to Vnet subscribers.

Vnet Service is provided as an interstate offering. The associated access lines, network usage, and features, other than specific intrastate usage charges are covered by MCI's F.C.C. Tariff No. 1. Specific intrastate usage charges are covered by this tariff in Section 4.1.9 and are being offered for Florida Vnet customers who have a requirement to make intrastate calls.

#### 3.5.10 Option J (MCI Vision) 1/

MCI Vision is an outbound customized telecommunications service which may include an inbound 800 service option using Business Line, WATS Access Line, or Dedicated Access Line Termination. It provides a unified service for single or multi-location companies using switched, dedicated, and card origination, and switched and dedicated termination. All rates and charges apply to all MCI Vision Programs unless otherwise specified.

\_\_\_\_

<sup>1/</sup> Beginning May 11, 1996, Option J (MCI Vision) will no longer be available to new subscribers unless otherwise noted herein.

# 3.5 METERED USE SERVICE OPTIONS(Cont.)

#### 3.5.10 Option J (MCI Vision) (Cont.)

Subscribers to Vision take the Vision Service Package which includes the availability of Dedicated, Switched, and Card Access and Dedicated, Switched, and WATS Access Line Termination, with Volume Discounts, Time-of-Day Discounts, Long Distance Directory Assistance, Accounting Codes, Universal Range Privileges, Call Detail, and either a consolidated invoice or location level invoices for all Vision calls from all locations. All Option J calls are subject to an 18-second minimum initial period and are rounded to the next higher six-second increment except for Operator Assisted calls, which are subject to a 60 second initial period and additional 60 second increments. All Option J Off-Peak Vision or MCI Vision Power Rate Off-Peak outbound calls are subject to a six second minimum initial period and additional six second increments. Inbound service ordered with Off-Peak Vision or the MCI Vision Power Rate Off-Peak Program are subject to an 18 second minimum initial period and additional six second increments.

The rules and regulations governing the provision of service for inbound 800 calls are identical to those for Option H, except where otherwise noted.

### 3.5.12 Option L (MCI 900 Service) 1/

MCI 900 Service is an inbound, long distance, voice-grade telecommunications service that permits callers to place long distance calls to Customer's stations in one location from stations in diverse geographical service areas. MCI 900 Service Customers may provide live or pre-recorded messages to callers using Customer's own equipment.

It is the customer's responsibility to provide answer supervision back to the MCI point of connection even when the MCI 900 Service is connected to switching equipment or a Customer-provided communications system. In such cases, the equipment or system must provide appropriate supervision so that the measure of chargeable time begins upon the delivery of the call to the customer's switching equipment or communications system and ends upon termination of the call.

. .

<sup>&</sup>lt;u>1</u>/ Effective/Beginning December 1, 2007, this service is grandfathered and is no longer available to new customers.

### 3.5.16 Option P (MCI HotelDirect) 1/

Option P is an outbound, customized telecommunications service which may include an inbound service option using Business Line, WATS Access Line, or Dedicated Access Line termination. It provides service for single and multi-location customers using switched and/or dedicated origination, and switched and/or dedicated termination. All outbound Option P calls are subject to a 30 second minimum initial period and are rounded to the next higher six second increment. All inbound Option P calls are subject to an 18 second minimum initial period and are rounded to the next higher six second increment. If the computed charge for a call includes a fraction of a cent, the fraction is rounded to the nearest whole cent. Option P is offered as an add-on to MCI's Interstate offering. To be eligible for service, a customer must subscribe to outbound service at all of its locations.

In addition to the usage charges set forth in section 4.1.16 herein, charges for terminating access lines, service option, and optional features will be billed in accordance with MCI's F.C.C. Tariff No. 1.

#### 3.5.18 Option R (MCI MASTERS)

MCI MASTERS is a one-way dial in-dial out multipoint service. All Option R calls are rounded to the next higher full minute. If the computed charge includes a fraction of a cent, the fraction is rounded down to the next whole cent for all intrastate calls. MCI MASTERS is available to students, faculty and administrators of educational institutions, to students, faculty, and administrators involved in a business relationship with other entities, and to employees of government agencies, provided that said institutions, entities and agencies have agreed to act as MCI's agent in connection with the provision of service.

Option R customers can place on-site or off-site calls. On-site calling, which requires no special access dialing sequences, originates from telecommunications equipment enrolled in MCI MASTERS which is located within the campus of participating educational institution or agency location or is routed through the participating educational institution's or agency location's switching equipment. Off-site calling requires the caller to access the service by dialing an MCI-provided 800 number. Off-site calls can originate from telecommunications equipment located anywhere within the state.

ISSUED: November 13, 2008

<sup>1/</sup> Beginning June 11, 1997, Option P will not be available to new subscribers.

#### 3.5.19 Option S (University Dial 1)

University Dial 1 is a one-way dial in-dial out multipoint service. Customers may originate and terminated calls from any telephone within the state. All Option S calls are rounded to the next higher full minute. If the computed charge includes a fraction of a cent, the fraction is rounded to the nearest whole cent for all calls. University Dial 1 is available to students, faculty, and administrators of educational institutions that agree to act as MCI's agent in connection with the provisions of this service.

University Dial 1 calls originate and terminate via local exchange carrier switched facilities. A customer may access service by: 1) designating MCI as its Primary Interexchange Carrier (PIC); or, 2) dialing an MCI Carrier Identification Code (CIC). A customer is not required to select MCI as its PIC in order to use this service. In all instances, access to service must be made via a Personal Identification Number (PIN), which will be used as the basis to invoice the customer. The assignment of a PIN establishes the user as a customer subject to all applicable provisions in this tariff which pertain to customers. Terms and conditions will be applied as set forth under

http://www.mci.com/service.

#### 3.5.22 Option V (hospitalityMCI) 1/

Option V is an outbound, customized telecommunications service which may include an inbound service option using a Business Line, WATS Access Line, or Dedicated Access Line Termination. Option V provides service for single and multi-location customers using switched and/or dedicated access origination and switched and/or dedicated termination. Option V may be used for long distance calls between customer locations and to other locations within the state. Upon enrollment a customer must commit to a term plan option and is bound by all terms and conditions as described in MCI's F.C.C. Tariff No. 1. All intrastate Option V calls are subject to an 18-second minimum initial period and are rounded to the next higher six second increment. If the computed charge for a call includes a fraction of a cent, the fraction us rounded to the nearest whole cent. When a hospitalityMCI call is established in one rate period and ends in another, the rate in effect in each rate period applies to the portion of the call occurring during that rate period. The rate for inbound calls is the rate for the time of day at the terminating location. To be eligible for service, a customer must subscribe to outbound service at all of its locations.

In addition to the usage charges set forth below, charges for terminating access lines, service options, discounts, surcharges, and optional features will be billed in accordance with MCl's F.C.C. Tariff No. 1.

The rules and regulations governing the provision of service for inbound calls are identical to those in Option H (MCI 800 Service), except where otherwise noted.

1/ Beginning June 11, 1997, Option V will not be available to new subscribers.

ISSUED: November 13, 2008

# 3.5 METERED USE SERVICE OPTIONS (Cont.)

#### 3.5.25 Option Y (networkMCI One)

Option Y (networkMCI One) is an outbound, customized telecommunications service which may include an inbound service option using Business Line or Dedicated Access Line termination. It provides a unified service for single or multi-location companies using switched, dedicated, and card origination, and switched and dedicated termination. Option Y is suitable for intrastate calls between company locations, as well as all other intrastate calls within the state.

Except as otherwise specified, all intrastate Option Y calls are subject to an 18-second minimum initial period and are rounded to the next higher six-second increment, except for Operator Assisted calls, which are subject to a 60-second initial period and additional 60-second increments.

If the computed charge includes a fraction of a cent, the fraction is rounded to the nearest whole cent.

Subscribers to networkMCI One receive the networkMCI One Service package which includes the availability of Dedicated, Switched, and Card Access and Dedicated and Switched Access Line Termination, networkMCI One Toll Free, Directory Assistance, Accounting Codes, Universal Range Privileges, Custom Calling Range Privileges, Call Detail, and either a consolidated invoice or location level invoices for all networkMCI One calls from all locations.

The rules and regulations governing the provision of service for toll free service are identical to those for Metered Use Service Option H (MCI 800 Service) except where otherwise specified.

#### 3.5.28 Option BB (MASTERS97)

MASTERS97 is an one-way dial-out multipoint service. Customers may originate calls from any telephone in the state. MASTERS97 is available to students, faculty, and administrators of educational institutions, provided that said institutions and entities have agreed to act as MCI's agent in connection with the provision of service. Dial-out calling, which requires no special access dialing sequences, originates from telecommunications equipment enrolled in MASTERS97 which is located within the campus of a participating educational institution or entity location or is routed through the participating educational institution's or entity location's switching equipment. MASTERS97 calls will be billed in 60-second increments and are rounded to the next higher full minute. If the computed charge includes a fraction of a cent, the fraction is rounded to the next whole cent.

#### 3.5.31 Option EE (networkMCI Contact one-number service)

one-number service is an inbound and outbound voice and data telecommunications service that permits calling from stations within Florida.

For each networkMCI Contact one-number service customer (Subscriber), MCI will designate a networkMCI Contact one-number service Number and networkMCI Contact a 6 digit security code which permits the Subscriber to manage service features and call routing features. The networkMCI Contact one-number service Number is a MCI designated Automatic Number Identification (ANI) provided by an affiliate of the company upon subscription to the service.

For each networkMCI Contact one-number service call, there are two parties: the "Subscriber", the party at whose designated location(s) the networkMCI Contact one-number service call terminates, and the "Calling Party", the party who places the call. The Subscriber is the customer and is subject to all applicable provisions in this tariff which pertain to customers.

Unless otherwise specified, all networkMCI Contact one-number service calls are subject to a 30-second minimum duration, with additional 6-second increments and are rounded to the next higher 6-second increment. If the computed charge for any call includes a fraction of a cent, the fraction is rounded to the nearest whole cent.

The following types of calls are not available with this service: Operator-assisted and Directory Assistance.

The Subscriber is not required to designate MCI as his or her Primary Interexchange Carrier (PIC).

Users of networkMCI Contact one-number service may have access to non-tariffed enhancements (e.g., information services) provided by MCI and/or other enhanced service providers.

#### 3.5.33 Option GG (On-Net Voice) 2/

Option GG is an outbound, customized telecommunications service that may include an inbound service option using Business Line/WATS or Dedicated Access Line terminations and/or a virtual private network. Option GG provides a service for single or multi-location companies using switched, dedicated and card origination, and switched and dedicated termination.

Except as otherwise specified, all domestic Option GG calls are subject to an 18-second minimum initial period and rounding to the next higher 6-second increment, except for Operator Assisted calls, which are subject to a 60-second initial period and rounding to the next 60-second increment.

If the computed charge includes a fraction of a cent, the fraction is rounded to the nearest whole cent.

The rules and regulations governing the provision of service for toll free (inbound) service are identical to those for Metered Use Service Option H (MCI 800 Service) except where otherwise specified.

This tariff is being offered to Florida customers that have a requirement to make intrastate calls. Intrastate service is provided in conjunction with interstate Option RR service and is available only to customers subscribing to interstate Option RR as provided in MCl's Interstate Tariff F.C.C. No. 1.

<sup>2/</sup> Beginning January 1, 2004, this service will no longer be available to new customers.

#### 3.5 METERED USE SERVICE OPTIONS (Cont.)

### 3.5.97 Option NNNN (Verizon Business Service I Flex T1

Eligibility: To be eligible, customers:

Must designate a Company affiliate as its exchange service carrier and the Company both as it interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

Must subscribe to the Verizon Business Service I Flex T1 exchange service as described in the

appropriate Company-Affiliate exchange service tariff ("Companion Local Service") concurrent with enrollment in this plan;

Must subscribe to service under Product Packages Guide Types 13, 14, 15 or 16 as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com; (Companion Interstate Services").

### 3.5.105 Option VVVV (Verizon Business Services II Flex T1 Plan) 1 Eligibility: To be eligible, customers:

Must designate a Company affiliate as its exchange service carrier and the Company both as it interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

Must subscribe to the Verizon Business Service Flex T1 exchange service as described in the the appropriate Company-Affiliate exchange service tariff ("Companion Local Service") concurrent with enrollment in this plan;

Must subscribe to service under Product Packages Guide Types 18, 19, 20 or 21 as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com; (Companion Interstate Services").

<sup>1/</sup> Beginning April 1, 2005, this service will no longer be available to new subscribers.

#### 3.6 DEDICATED LEASED LINE SERVICES 1/

Dedicated Leased Line Service is offered in the form of discrete MCI-provided interexchange communications facilities which are dedicated to the use of a specific customer and are billed at pre-determined fixed monthly rates. Dedicated Leased Line Services are available as set forth in the Service Availability Tables listed in this tariff. Dedicated Leased Line Service may be provided for Analog transmission and Digital transmission in the following modes:

#### 3.6.1 **Analog Service:**

- 3.6.1.1 Voice Grade Service is offered in increments of one or more voice grade channels, each with nominal bandwidth of 4 kHz. Such voice grade channels or combinations of voice grade channels may be utilized, on a permissive basis, for voice service, data service, facsimile service or combinations thereof, and may be interconnected with the facilities of other communications carriers and users for a customer's communications needs, such as voice connecting arrangements. MCI will provide, if required, a handset at each circuit end as part of this service.
- 3.6.1.2 Analog Data Service is offered at approximate data speeds of 0-300, 1200, 2400, 4800, 7200, 9600 or 19200 bits per second (b.p.s.).
- 3.6.1.3 Alternate Voice/Data Service is offered with the capability of voice grade communications and data communications on an alternate use basis. If required, MCI will provide a handset at each circuit end as part of this service.

Recurring and non-recurring charges for Analog Service are located in Section 4.2.1.

1/ Beginning September 1, 2005, this service will no longer be available to new subscribers.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

- XIII. GRANDFATHERED SERVICES: SUBSECTION A
- 3.6 DEDICATED LEASED LINE SERVICES (Cont.)
  - 3.6.2 Digital Service:
    - 3.6.2.1 Digital Data Service (DDS) is a MCI-provided point to point dedicated circuit used for simultaneous two-way transmission at synchronous speeds of 9.6, 56 and 641/ Kbps. For purposes of determining credit allowances in accordance with the regulations in Section 2.14.1.1 preceding, DDS is considered to be interrupted when either there has been a loss of continuity, or the error performance is below the design objective of 99.5% error free seconds measured over a continuous 24-hour period.
    - 3.6.2.2 Terrestrial Digital Service (TDS) is a point to point dedicated circuit used for simultaneous two-way transmission of serial bipolar, isochronous digital signals at a transmission speed of 1.544 Mbps. Credit will be allowed for interrupted service as provided under Section 2.14.1.1 preceding.
      - 3.6.2.2.1 Terrestrial Digital Service - 1.5 (TDS-1.5) Inter-Office Service

#### Monthly Rate Plan:

Under this Plan, IOC charges are determined by applying the per circuit and per mile Monthly Plan rates to individual IOCs. A customer may disconnect circuits or shift them to other plans subject to the regulations in Section 3.6.3 entitled "Notification Required from Customers".

#### **Fixed Term Plans:**

Under these plans, a customer may order individual circuits for fixed terms of 1, 2, 3, 4, or 5 years. The monthly recurring IOC charges under a particular plan are determined by deducting the applicable discount from the monthly charges calculated by using the per circuit and per mile rates of the Monthly Plan.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

<sup>1/</sup> For Clear Channel 64 Kbps circuit, the customer must provide B8ZS/clear channel compatible customer terminal equipment.

- XIII. GRANDFATHERED SERVICES: SUBSECTION A
- 3.6 <u>DEDICATED LEASED LINE SERVICES (Cont.)</u>
  - 3.6.2 <u>Digital Service (Cont.)</u>:
    - 3.6.2.2 Terrestrial Digital Service (TDS) (Cont.)
      - 3.6.2.2.1 <u>Terrestrial Digital Service 1.5 (TDS-1.5) Inter-Office Service (Cont.)</u>

A customer may terminate a Fixed Term Plan or Plans before the end of the committed term or terms without liability, if:

- a revision in this Tariff results in a higher IOC rate for the Plan to which has not given written consent;
- the customer enrolls the TDS-1.5 IOC in a new Fixed Term Plan which expires on or after the expiration date of the existing Plan;

Except as otherwise specified immediately above, if a customer terminates a Fixed Term Plan prior to its expiration, the customer shall pay termination charges for any unexpired portions of the term remaining after the notice period for disconnecting a circuit (see Section 3.6.3 entitled "Notification Required From Customers"). The customer is liable for termination charges equal to 100 percent of the applicable monthly recurring TDS-1.5 IOC charges for any remaining portions of the first year of the term, and 50 percent of the charges for the remainder of any subsequent years.

3.6.2.3 <u>Digital Private Line Service</u> is a point-to-point dedicated digital circuit used for voice and data communications at transmission speeds of up to 64 1/ kbps and Fractional T-1 speeds up to 768 kbps in 56 kbps or 64 kbps increments. This service can be accessed by either Analog Local Access or Digital Local Access (DS0), including Digital Data Local Access or T-1 Digital Access.

Recurring and non-recurring charges for Digital Services are located in Section 4.2.2.

3.6.2.4 <u>Terrestrial Digital Service - 45 (TDS-45)</u> is a point-to-point dedicated circuit between customer premises that is used for simultaneous two-way transmission of serial bipolar, isochronous digital signals at a transmission speed of 44.746 Mbps. One TDS-45 channel provides the equivalent information handling capacity of 28 TDS-1.5 channels or 672 voice equivalent circuits. TDS-45 may be used for any combination of voice and data communication.

\_

<sup>1/</sup> For Clear Channel 64 Kbps circuit, the customer must provide B8ZS/clear channel compatible customer terminal equipment.

# 3.6 <u>DEDICATED LEASED LINE SERVICES (Cont.)</u>

#### 3.6.3 Notification Required From Customers:

All customer requests to commence, change, or terminate a Fixed Term Pricing Plan must be made in writing and received by MCI no later than the 15th day of the month preceding the month in which the desired action is to take effect. All rate plan changes for existing circuits take effect on the first day of a calendar month.

A customer may cancel any order for an optional Fixed Term Pricing Plan provided that a written request to do so is received by MCI before the plan commences. A request to cancel an optional rate plan shall not be deemed to be a request to cancel any associated circuits, unless otherwise indicated in the written notification.

A customer must provide MCI with one month written notice to disconnect a circuit. All charges for dedicated leased line services and service components ordered under this tariff and provided by MCI, or by MCI as agent when acting in the customer's behalf, will apply for one month from the time notice is received or until the requested disconnection date, whichever is later. The charges will apply whether or not the customer uses the circuit.

A circuit may only be enrolled in one rate plan at a time. The Monthly Plan will be the applicable rate plan pursuant to which a circuit will be billed when no other plan has been selected by the customer, or when the selected Plan has not commenced, been canceled, or has expired. The Monthly Plan may also apply if the customer fails to comply with the requirements for notification as set forth above.

#### 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.4 Option D (MCI Prism I)

The usage charges for calls within the state of Florida are based on two ranges as shown in Table VI.

RANG	BUSINESS DAY	<u>EVENING</u>	NIGHT & WEEKEND
1 2	\$.1330 \$.1330	\$.1170 \$.1170	\$.0990 \$.0990
Access	Options:	MONTHLY (Per Channel)	INSTALLATION (Per Channel)
(A) (B) (C) (D)	T-1 Digital Access Analog Local Access Special Access Surcharge <sup>1/</sup> Access Coordination:	LEC Charge LEC Charge \$36.25	LEC Charge LEC Charge N/A
` '	T-1 Digital Access Analog Local Access	\$21.70 \$ 9.60	\$186.00 \$135.00
(E)	Central Office Connection: T-1 Digital Access Analog Local Access	\$62.00 \$16.50	\$279.00 \$ 75.45

CPE Components - If T-1 Digital Access is used, charges for CPE are those shown in Section 4.2.2.4 for Terrestrial Digital Service.

Monthly Service Charge: \$ 25.00 per Prism I invoice

#### Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

#### Optional Features:

(A) Accounting Codes (2-digit) \$15/99 codes (B) Call Records on Magnetic Tape \$100/Account

<sup>1/</sup> The Surcharge, imposed by the Local Exchange Carrier, will not apply to those customers who provide MCI with an Exemption Certificate (as defined herein).

### 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.4 Option D (MCI Prism I) (Cont.)

Non-Recurring Charges: Per Circuit or T-1 Channel Per Order

(A) Installation LEC Charge \$ 50

The Rates and Charges can be found in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com (Companion Interstate Service).

### 4.1.5 Option E (MCI Prism II)

The usage charges for calls within the state of Florida are based on two ranges as shown in Table VI.

<u>RANGE</u>	<b>BUSINESS DAY</b>	<u>EVENING</u>	NIGHT & WEEKEND
1	\$.1705	\$.1338	\$.0891
2	\$.1705	\$.1338	\$.0891

# Minimum Usage Charge: \$75.00/account for all usage

### Monthly Dedicated Access Line Charges:

(A)	MCI-provided access line	\$130.00
(B)	Customer-provided access line	\$ 50.00
(C)	Special Access Surcharge 1/	\$ 36.25

# Optional Features:

(A) Accounting Codes (2-digit) \$15/99 codes

(B) Call Records on Magnetic Tape \$100/Account

Non-Recurring Charges: Per Circuit Per Order

(A) Installation \$120 \$50

The Rates and Charges can be found in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="https://www.verizonbusiness.com">www.verizonbusiness.com</a> (Companion Interstate Service).

ISSUED: November 13, 2008

<sup>1/</sup> The Surcharge, imposed by the Local Exchange Carrier, will not apply to those customers who provide MCI with an Exemption Certificate (as defined herein).

### 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.8 Option H (MCI 800 Service)

#### 4.1.8.1 <u>Dedicated Termination Option</u>

#### Per minute usage charges:

<u>RANGE</u>	<b>BUSINESS DAY</b>	<u>EVENING</u>	NIGHT & WEEKEND
1	\$.1364	\$.1148	\$.0836
2	\$.1364	\$.1148	\$.0836

#### Terminating Access Line Charges:

T-1 Digital Access - Charges associated with T-1 Digital Access are shown in Section 4.2.2.2.

Analog Access - Charges associated with Analog Access are shown in Section 4.2.1.

	<u>MONTHLY</u>	INSTALLATION
Special Access Surcharge 1/:	29.62	N/A

Monthly Service Charge: \$50 per service group

# Volume Discounts:

0% on total monthly usage between \$0 and \$1000.00 7% on total monthly usage between \$1000.01 and \$10,000.00, plus 12% on total monthly usage between \$10,000.01 and \$30,000.00, plus 12% on total monthly usage above \$30,000.00

ISSUED: November 13, 2008

<sup>1/</sup> The Special Access Surcharge, imposed by the Local Exchange Carrier, will not apply to those customers who furnish MCI with an Exemption Certificated (as defined herein).

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.8 Option H (MCI 800 Service) (Cont.)

#### 4.1.8.2 Switched WATS Termination Option

# Per minute usage charges:

<u>RANGE</u>	<b>BUSINESS DAY</b>	<b>EVENING</b>	NIGHT & WEEKEND
1	\$.2008	\$.1444	\$.1028
2	\$.2008	\$.1444	\$.1028

# **Terminating Access Line Charges:**

(A) Monthly	\$ 35.00/line
(B) Installation	\$150.00/line
(C) Special Access Surcharge <u>1</u> /	\$ 36.25/line

Monthly Service Charge: \$20 per service group

### Volume Discounts:

0% on total monthly usage between \$0 and \$50.00 5% on total monthly usage between \$50.01 and \$350.00, plus 10% on total monthly usage between \$350.01 and \$1,350.00, plus 15% on total monthly usage above \$1,350.00

ISSUED: November 13, 2008

EFFECTIVE: November 14, 2008

<sup>1/</sup> The Special Access Surcharge, imposed by the Local Exchange Carrier, will not apply to those customers who furnish MCI with an Exemption Certificated (as defined herein).

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.8 Option H (MCI 800 Service) (Cont.)

# 4.1.8.3 Business Line Termination Option

Per minute usage charges:

RANGE	<b>BUSINESS DAY</b>	<b>EVENING</b>	NIGHT & WEEKEND
1	\$.2261	\$.2193	\$.2193
2	\$.2261	\$.2193	\$.2193

#### **Terminating Access Line Charges:**

Non-Recurring Charge: \$ 0 per account

Monthly Service Charge: \$20 per service group

#### Volume Discounts:

0% on total monthly usage between \$0 and \$50.00 5% on total monthly usage between \$50.01 and \$350.00, plus 10% on total monthly usage between \$350.01 and \$1,350.00, plus 15% on total monthly usage above \$1,350.00

# 4.1.8.4 Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

#### Other Non-Recurring Charges:

The Rates and Charges can be found in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="https://www.verizonbusiness.com">www.verizonbusiness.com</a> (Companion Interstate Service).

		PerLine/Channel	<u>Per Order</u>
(A)	Installation	Refer to Sections	\$50.00
		4.1.8.1 and 4.1.8.2	

#### 4.1.9 Option I (Vnet Service)

### **Promotional Offerings**

#### (A) <u>Vnet Service Dial Access Promotion</u>

Beginning May 11, 1996 and ending September 30, 1996, MCI will offer the following promotion via direct solicitation only to customers of Metered Use Service Option I (Vnet Service).

MCI will provide a usage credit or credits, the total of which may not exceed \$3,300 per participating location, according to the chart below.

# Switched Outbound

per Location	Requirement A	Requirement B Credit	
1 - 4	\$ 40	\$ 40	
5 - 8	80	80 1,100	
9 - 12	120	120 1,650	
13 - 16	160	160 2,200	
17 - 20	200	200 2,750	
21 - 24	240	240 3,300	

#### Where:

Requirement A reflects the minimum monthly interLATA usage averaged quarterly per participating location for twelve full months after the promotion location enrollment effective date; and Requirement B reflects the minimum monthly incremental (compared to the full month prior to the promotion location enrollment effective date) switched outbound intraLATA usage, excluding calling card usage, averaged quarterly per participating service location for twelve full months after the promotion location enrollment effective date.

For the purpose of this promotion, quarterly shall mean three-month periods beginning on a promotion location enrollment effective date.

Each customer's usage per location will be evaluated quarterly to determine if the above-stated requirements are satisfied. If the customer does not meet Requirements A and B in any quarter for 12 months after the promotion location enrollment effective date, the customer will be billed and required to pay an underutilization charge equal to the difference in the quarterly Requirement A and Requirement B minimums and the associated actual usage for that quarter.

Customers who terminate service at any location(s) prior to one year from the promotion location enrollment effective date will be billed and required to pay an early termination charge equal to the underutilization charge for all remaining quarters in that year (or pro rata portion thereof), not to exceed the promotional credits received to date for that location. The credit(s) will appear on the invoice reflecting the location's second full month's invoice after the promotion location enrollment effective date, with no credits being reflected prior to the October 1995 invoice. Unless otherwise specified, customers enrolled in a Special Customer Arrangement other than Types 5, 17, 19, 20, and 21, as described in MCI's F.C.C. Tariff No. 1, are not eligible to receive the benefits of this promotion.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.9 Option I (Vnet Service) (Cont.)

Promotional Offerings (Cont.)

#### (B) 1+ Local Toll Equal Access Promotion

Beginning June 11, 1997 and ending June 30, 1998, MCI will offer the following promotion to new and existing switched and dedicated customers of Option J (MCI Vision), Option I (MCI Vnet), and Option Y (networkMCI One) who enroll that do not currently use MCI as their local toll provider. Existing customers adding MCI Local Toll service to a new location are also eligible for the benefits of this promotion. Customers enrolling in this promotion will receive a credit equal to their second full month's discounted outbound Local Toll usage, excluding card usage, for all locations which have at least 1 ANI PIC'd to MCI for Local Toll. The credit will be applied to the third full month's invoice not to exceed \$5,000 per location.

Additionally, after twelve months Customers will receive credit, not to exceed \$5,000 per location equal to five percent of their discounted outbound Local Toll usage, excluding card usage, over the previous full 12 months for all locations which have at least 1 ANI PIC'd to MCI for Local Toll. This credit will be applied to the customer's 13th month's invoice.

Customers who terminate MCI service prior to receipt of the credits will not receive the credits associated with this promotion. Locations currently receiving the benefits of any other Local Toll promotion cannot receive the benefits of this promotion. Each customer location may enroll in this promotion once.

Unless otherwise specified, customers enrolled in Special Customer Arrangements other than types 5, 17, 19, 20, 21, 23 through 32, and 35 through 44 are not eligible to receive the benefits of this promotion.

In the event a Customer invokes the National "Service Guarantee" Promotion for any service provided in hereunder, Customers will be billed and Customer agrees to pay back to MCI the credits received for such service under this promotion.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.9 Option I (Vnet Service) (Cont.)

## Promotional Offerings (Cont.)

#### (C) <u>Vnet Short Haul Promotion</u>

Beginning May 1, 1997 and ending March 13, 1998, MCI will offer the following promotional rates to Vnet customers that enroll in this program and use dedicated access-switch termination to access the MCI network. MCI will offer the following rates on Intralata traffic originating and terminating within the specified mileage band in lieu of rates specified in Section 4.1.9.2.2 and 4.1.9.2.3.

<u>Mileage</u> <u>Rate</u> 0 - 50 \$.0400

The Short-Haul Intralata rates listed above are available throughout the remainder of the customer's current Vnet contract. For Intralata calls terminating outside of the specified mileage band, tariffed rates will apply. At the expiration of this customer contract, Intralata rates for all mileage bands will revert to Vnet tariffed rates. These promotional rates are not available to customer locations that fall within the Local Service Area as defined in MCI Metro Local Exchange Tariffs. Customers enrolling in this promotion are not eligible for the Investment Honors Promotion, NetworkMCI One SCA - The New Deal, or the Access Promotion Under Race To Success For National Accounts as described in MCI's FCC #1. Customers enrolling in this program may be eligible for all other Race To Success For National Accounts Promotions as described in MCI's FCC #1.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.9 Option I (Vnet Service) (Cont.)

#### 4.1.9.1 Access/Termination Methods

#### 4.1.9.1.1 <u>Dedicated Access/Termination</u>

<u>Analog Access</u> - Components associated with Analog Access and their relevant monthly and non-recurring charges are billed in accordance with MCI's F.C.C. Tariff No. 1, Section C-2.0211.

<u>T-1 Digital Access</u> - Components associated with T-1 Digital Access and their relevant monthly and non-recurring charges are billed in accordance with MCI's F.C.C. Tariff No. 1, Section C-2.0221.

4.1.9.1.2 <u>Shared (Switched) Access</u>: The following Shared Access methods are available to customers of Vnet Service:

<u>Vnet Dial "1" Access</u> - available in MCI-served equal access exchanges. Subscribers can dial Vnet 7-digit and 10-digit private dialing plan numbers, and 10-digit off-Vnet numbers. Subscribers must dial the prefix "700" before a 7-digit private calling plan number.

<u>Vnet Card Access</u> - available from any tone-dial phone via an "800" number. Subscribers can dial Vnet 7-digit and 10-digit private dialing plan numbers, and 10-digit off-Vnet numbers.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.9 Option I (Vnet Service)

## 4.1.9.2 Rates and Charges 1/

4.1.9.2.1 <u>Vnet Card Surcharge</u>: The following surcharge applies to all Vnet Card calls other than calls to Directory Assistance.

\$0.40 per call

4.1.9.2.2 <u>Business Day Rates</u> - The following rates apply 8 a.m. - 5 p.m. weekdays.

# (A) <u>InterLATA Usage Rates</u>

		D-SWITCHED	DEDICAT	ED-SWITCHE	ED	
	AND CA	<u> </u>	AND CA	<u>ARD</u>	<u>DEDICATED</u>	<u>-DEDICATED</u> <u>3</u> /
Mileage <u>2</u> /	First	Add'l	First	Add'l	First	Add'l
Band	<u>18 Sec.</u>	<u>6 Sec.</u>	18 Sec.	6 Sec.	<u>18 Sec.</u>	<u>6 Sec.</u>
0 -292	\$.0484	\$.0161	\$.0300	\$ .0100	\$ .0139	\$ .0046
293 - Over	\$.0484	\$.0161	\$ .0300	\$ .0100	\$ .0139	\$ .0046

# (B) <u>IntraLATA Usage Rates</u>

			SWITCHE	:D-DEDICATE	D		
	SWITCHE	D-SWITCHED	DEDICAT	ED-SWITCHE	D		
	AND CA	ARD	AND C	<u>ARD</u>	<b>DEDICATED</b>	-DEDICATED 3	3/
Mileage 2/	First	Add'l	First	Add'l	First	Add'l	
_Band_	<u> 18 Sec.</u>	6 Sec.	18 Sec.	<u> 6 Sec.</u>	<u> 18 Sec.</u>	<u> 6 Sec.</u>	
0 -292	\$.0340	\$.0113	\$ .0306	\$ .0102	\$ .0139	\$ .0046	
0 -292 293 - Over	\$.0340 \$.0340	\$.0113 \$.0113	\$ .0306 \$ .0306	\$ .0102 \$ .0102	\$ .0139 \$ .0139	\$ .0046 \$ .0046	
293 - OVEI	φ.0340	φ.υτιδ	φ.0300	φ.0102	φ.0139	φ.0040	

ISSUED: November 13, 2008

<sup>1/</sup> Usage charges for Vnet Card calls are based on Switched Access origination.

<sup>2/</sup> Mileage is calculated by using the formula presented on sheets 78 through 81 and the Vertical and Horizontal Coordinates as obtained by reference to AT&T's Tariff FCC No. 274.

<sup>3/</sup> Reflects the transport portion of the call.

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.9 Option I (Vnet Service) (Cont..)

## 4.1.9.2 Rates and Charges (Cont.) 1/

4.1.9.2.3 <u>Non-Business Day Rates</u> - The following rates apply at all times other than 8 a.m. - 5 p.m. weekdays.

# (A) InterLATA Usage Rates

	SWITCHE	D-SWITCHED		D-DEDICATEDED-SWITCHED	=	
	AND CA	<u>RD</u>	AND C	ARD_	<b>DEDICATED</b>	-DEDICATED 3/
Mileage 2/	First	Add'l	First	Add'l	First	Add'l
<u>Band</u>	<u>18 Sec.</u>	<u>6 Sec.</u>	<u>18 Sec.</u>	<u> 6 Sec.</u>	<u>18 Sec.</u>	<u> 6 Sec.</u>
0 -292	\$.0403	\$.0134	\$ .0207	\$ .0069	\$ .0129	\$ .0043
293 - Over	\$.0403	\$.0134	\$ .0207	\$ .0069	\$ .0129	\$.0043

# (B) IntraLATA Usage Rates

	SWITCHE AND CA	D-SWITCHED		D-DEDICATEI ED-SWITCHEI ARD		-DEDICATED 3/
Mileage <u>2</u> /	First	Add'l	First	Add'l	First	Add'l
<u>Band</u>	18 Sec.	6 Sec.	18 Sec.	6 Sec.	18 Sec.	<u>6 Sec.</u>
0 -292	\$.0295	\$.0098	\$ .0210	\$ .0070	\$ .0129	\$ .0043
293 - Over	\$.0295	\$.0098	\$ .0210	\$ .0070	\$ .0129	\$ .0043

## 4.1.9.2.4 Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

ISSUED: November 13, 2008

<sup>1/</sup> Usage charges for Vnet Card calls are based on Switched Access origination.

<sup>2/</sup> Mileage is calculated by using the formula presented on sheets 78 through 81 and the Vertical and Horizontal Coordinates as obtained by reference to AT&T's Tariff FCC No. 274.

<sup>3/</sup> Reflects the transport portion of the call.

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.9 Option I (Vnet Service) (Cont.)

4.1.9.3 <u>Directory Assistance</u> - An undiscounted charge per call will be applied to each Directory Assistance call, subject to the provisions set forth in Section 2.6.4 of this tariff.

## 4.1.9.4 Government Pricing Plan (GPP)

The following rates will be applied for customers of Type 11, Type 12, Type 13, and Type 14 GPPs as described in MCI's F.C.C. Tariff No. 1, Section C-16.

4.1.9.4.1	Switched - Switched Us	age Charges	
DAY Initial 18 Seconds	Additional 6 Seconds	NON-E Initial Additional Notational Notation 18 Seconds	
\$0.0343	\$0.0114	\$0.0214	\$0.0071
4.1.9.4.2	Dedicated - Switched ar	nd Switched - De	edicated Usage Charges
DAY		NON-E	DAY
Initial 18 Seconds	Additional 6 Seconds	Initial 18 Seconds	Additional 6 Seconds
\$0.0183	\$0.0061 R	\$0.0114	\$0.0038
4.1.9.4.3	Dedicated - Dedicated L	Jsage Charges	
DAY		NON-E	
Initial 18 Seconds	Additional 6 Seconds	Initial Addition 18 Seconds	nal <u>6 Seconds</u>
\$0.0131	\$0.0044	\$0.0092	\$0.0031

Operator Assistance - For intrastate calls placed with the assistance of an MCI operator, the rates and surcharges in Sections 4.1.1.3(B) and 4.1.1.3(A) respectively, will apply.

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.9 Option I (Vnet Service) (Cont.)

#### 4.1.9.6 Optional Features:

### 4.1.9.6.1 Market Area Calling 1/

Vnet Market Area Calling: Offers intralata, outbound, voice only calling within the customer's Market Area for calls using dedicated access-switched termination and dedicated access-dedicated termination This call type is not available to customer locations that fall within the local service areas of Perrine, Miami, Homestead, North Dade, Fort Lauderdale, Coral Springs, Deerfield Beach, Pompano Beach, Hollywood, Debary, East Orange, Geneva, Orlando, Oviedo, Sanford, Lake Mary, Tampa, Clearwater, Plant City, Zephyrhills, St. Petersburg, New Port Richey, Hudson, Tarpon Springs, Boca Raton, Delray Beach, and Boynton Beach.

The rates specified in sections 4.1.9.2.2 and 4.1.9.2.3 intraLATA (dedicated-switched and dedicated -dedicated) will apply to all Market Area calls originating and terminating within the Customer's Market Area. A customers Market Area is defined as the dominant Local Exchange Carrier's (LEC) Local calling area specified in the LEC's Local Exchange Tariff for the customer's exchange area for the specific location implementing Market Area Calling.

In addition, Vnet discounts as outlined in MCI's FCC Tariff No. 1, Section C-3.097 will apply. If a customer subscribes to a Special Customer Arrangement with intrastate discounts or rates then those rates and discounts will apply to Market Area Calling. SCA Types 1 and 4 are not eligible for Market Area Calling.

ISSUED: November 13, 2008

<sup>1/</sup> Beginning November 15, 1998, Market Area Calling will no longer be available to new subscribers.

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.9 Option I (Vnet Service) (Cont.)

## 4.1.9.6 Optional Features:

# 4.1.9.6.1 <u>Market Area Calling</u>

Customers must have Branch Exchange (PBX) or Hybrid Key System and originate Market Area Calling traffic via dedicated access facilities to use this feature.

The Market Area Calling feature cannot be used to originate 911, Directory Assistance, operator services calls, toll free or to terminate inbound local calls. Customers must maintain LEC-provided lines or trunks to originate and terminate these call types. This feature is subject to the Company's existing capacity and network availability.

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) 1/

## **Promotional Offerings**

#### MCI Vision Dial Access Promotion

Beginning May 11, 1996 and ending September 30, 1996, MCI will offer the following promotion via direct solicitation only to customers of Metered Use Service Option J (MCI Vision).

MCI will provide a usage credit or credits, the total of which may not exceed \$3,300 per participating location, according to the chart below.

Switched Outbound ANIs Enrolled per Location	Requirement A	Requirement B	<u>Credit</u>
1 - 4	\$ 40	\$ 40	\$ 550
5 - 8	80	80	1,100
9 - 12	120	120	1,650
13 - 16	160	160	2,200
17 - 20	200	200	2,750
21 - 24	240	240	3,300

\_

<sup>1/</sup> Beginning May 11, 1996, Option J (MCI Vision) will no longer be available to new subscribers unless otherwise noted herein.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) 1/

## **Promotional Offerings**

MCI Vision Dial Access Promotion (Cont.)

Where:

Requirement A reflects the minimum monthly interLATA usage averaged quarterly per participating location for twelve full months after the promotion location enrollment effective date; and Requirement B reflects the minimum monthly incremental (compared to the full month prior to the promotion location enrollment effective date) switched outbound intraLATA usage, excluding calling card usage, averaged quarterly per participating service location for twelve full months after the promotion location enrollment effective date.

For the purpose of this promotion, quarterly shall mean three-month periods beginning on a promotion location enrollment effective date.

Each customer's usage per location will be evaluated quarterly to determine if the above-stated requirements are satisfied. If the customer does not meet Requirements A and B in any quarter for 12 months after the promotion location enrollment effective date, the customer will be billed and required to pay an underutilization charge equal to the difference in the quarterly Requirement A and Requirement B minimums and the associated actual usage for that quarter.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

Promotional Offerings (Cont.)

#### MCI Vision Dial Access Promotion (Cont.)

Customers who terminate service at any location(s) prior to one year from the promotion location enrollment effective date will be billed and required to pay an early termination charge equal to the underutilization charge for all remaining quarters in that year (or pro rata portion thereof), not to exceed the promotional credits received to date for that location.

The credit(s) will appear on the invoice reflecting the location's second full month's invoice after the promotion location enrollment effective date, with no credits being reflected prior to the October 1995 invoice.

Unless otherwise specified, customers enrolled in a Special Customer Arrangement other than Types 5, 17, 19, 20, and 21, as described in MCI's F.C.C. Tariff No. 1, are not eligible to receive the benefits of this promotion.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

## Promotional Offerings (Cont.)

#### 1+ Local Toll Equal Access Promotion

Beginning June 11, 1997 and ending June 30, 1998, MCI will offer the following promotion to new and existing switched and dedicated customers of Option J (MCI Vision), Option I (MCI Vnet), and Option Y (networkMCI One) who enroll that do not currently use MCI as their local toll provider. Existing customers adding MCI Local Toll service to a new location are also eligible for the benefits of this promotion. Customers enrolling in this promotion will receive a credit equal to their second full month's discounted outbound Local Toll usage, excluding card usage, for all locations which have at least 1 ANI PIC'd to MCI for Local Toll. The credit will be applied to the third full month's invoice not to exceed \$5,000 per location.

Additionally, after twelve months Customers will receive credit, not to exceed \$5,000 per location equal to five percent of their discounted outbound Local Toll usage, excluding card usage, over the previous full 12 months for all locations which have at least 1 ANI PIC'd to MCI for Local Toll. This credit will be applied to the customer's 13th month's invoice.

Customers who terminate MCI service prior to receipt of the credits will not receive the credits associated with this promotion. Locations currently receiving the benefits of any other Local Toll promotion cannot receive the benefits of this promotion. Each customer location may enroll in this promotion once.

Unless otherwise specified, customers enrolled in Special Customer Arrangements other than types 5, 17, 19, 20, 21, 23 through 32, and 35 through 44 are not eligible to receive the benefits of this promotion.

In the event a Customer invokes the National "Service Guarantee" Promotion for any service provided in hereunder, Customers will be billed and Customer agrees to pay back to MCI the credits received for such service under this promotion.

#### 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.) 1/

#### 4.1.10.1 Service Availability

Option J is available as follows:

- Analog Local Access for Vision is available in the MCI Metropolitan Area Terminal Α. locations set forth in Section C-7, Table I.
- T-1 Digital Access for Vision is available in the MCI Metropolitan Area Terminal B. locations set forth in Section C-7, Table I.
- C. Dial "1" Access for Vision is available from all exchanges which have been converted to equal access in the cities listed in Section C-7, Table II.
- D. Vision customers can terminate calls to all cities within the State of Florida.
- E. Intrastate Vision Service is an add-on to the interstate MCI Vision Service provided in this Company's Tariff F.C.C. No. 1. Customers reselling or rebilling intrastate Vision Service must have a certificate of Public Convenience and Necessity as an IXC from the Florida Public Service mission.

#### 4.1.10.2 Access/Termination Methods and Charges

#### 4.1.10.2.1 **Outbound Service**

# **Dedicated Access/Termination**

Analog Access - Components associated with Analog Access and their relevant monthly and non-recurring charges are billed in accordance with MCI's F.C.C. Tariff No. 1, Section C-2.0211.

T-1 Digital Access - Components associated with T-1 Digital Access and their relevant monthly and non-recurring charges are billed in accordance with MCI's F.C.C. Tariff No. 1, Section C-2.0221.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

Customers currently enrolled in the Vision RPP Plan, as described in MCI's F.C.C. Tariff No. 1, will receive 1/ the intrastate usage rates and discounts effective on 2/1/95.

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision)

## 4.1.10.2 Access/Termination Methods and Charges (Cont.)

## 4.1.10.2.1 Outbound Service (Cont.)

## **Shared Access Methods**

- (A) Vision Card Access Available from any touch-tone phone in Florida via an "800" number. There is a \$ .95 surcharge for each Vision Card Call.
- (B) MCI Vision Power Rate Card Access There is a \$ .90 surcharge for each MCI Vision Power Rate Card call. 1/

# 4.1.10.2.2 <u>Inbound 800 Service</u>

The charges in Section 4.1.8 apply.

1/ The MCI Vision Power Rate Program will continue to be available to customers after May 11, 1996.

ISSUED: November 13, 2008

EFFECTIVE: November 14, 2008

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

# 4.1.10.3 Rates and Charges

4.1.10.3.1 <u>Outbound Service 1/2/</u>

# (A) InterLATA Usage Rates

Mileage	Switched <u>Access</u>	Dedicated	Vision Card
<u>Band</u>		<u>Access</u>	<u>Access</u>
0-100	\$ .1612	\$ .1031	\$ .1917
101+	\$ .1612	\$ .1031	\$ .1917

# (B) IntraLATA Usage Rates

Mileage	Switched	-	Dedicated	Vision Card
<u>Band</u>	Access		<u>Access</u>	Access
0-100	\$ .1161		\$ .1025	\$ .1989
101+	\$ .1161		\$ .1025	\$ .1989

For Vision customers participating in MCI's Interstate Vision Regional Program, the following intrastate usage rates will apply. In addition, the Vision Card Access surcharge found in Section 4.1.10.2.1.(A) will apply to Card usage:

Switched/Card Access	<u>Dedicated Access</u>
\$0.1790	\$0.1160

ISSUED: November 13, 2008

Vision customers participating in MCI's Interstate Tariff Off-Peak Vision Program will receive a 10% discount off of Switched and Dedicated intrastate Vision tariff rates. All Off-Peak Vision Program outbound calls are subject to a six-second minimum initial period and additional six-second increments.

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

## 4.1.10.3 Rates and Charges (Cont.)

## 4.1.10.3.2 <u>Inbound 800 Service</u>

<u>Per Minute Usage Charges</u>: Inbound calls will be priced at the rates specified below.

Business Line/

Switched WATS Access
Line Termination
Dedicated Access
Line Termination

\$ .2106 \$ .1364

## 4.1.10.3.3 MCI Vision Power Rate Program 1/

# (A) MCI Vision Power Rate - InterLATA

Inbound/Outbound Switched Access	\$.1728
Inbound/Outbound Dedicated Access	\$.1046
MCI Vision Power Rate Card Access	\$.1728

# (B) MCI Vision Power Rate - IntraLATA

Inbound/Outbound Switched Access	\$.1345
Inbound/Outbound Dedicated Access	\$.1008
MCI Vision Power Rate Card Access	\$.1345

1/ The MCI Vision Power Rate Program will continue to be available to customers after May 11, 1996.

ISSUED: November 13, 2008

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

# 4.1.10 Option J (MCI Vision) (Cont.)

#### 4.1.10.3 Rates and Charges Cont.)

# 4.1.10.3.3 MCI Vision Power Rate Program

(C) MCI Vision Power Rate Off-Peak - The following per minute usage rates will apply for both outbound and inbound usage. For MCI Vision Off-Peak Power Rate Card Access, the per minute usage rates found in section 4.1.10.3.1(A), as well as the per call surcharge found in Section 4.1.10.2.1(B) will apply.

# InterLATA Usage Rates

	Switched Access	Dedicated <u>Access</u>
Peak	\$0.2010	\$0.1228
Off-Peak	\$0.1206	\$0.0867

## **IntraLATA Usage Rates**

	Switched Access	Dedicated <u>Access</u>
Peak	\$0.1514	\$0.1178
Off-Peak	\$0.0942	\$0.0832

- 4.1.10.3.4 <u>Directory Assistance</u> An undiscounted charge per call will be applied to each Directory Assistance call, and subject to the provisions set forth in Section 2.6.4 herein.
- 4.1.10.3.5 Operator Assistance For intrastate calls placed with the assistance of an MCI operator, the rates and surcharges in Sections 4.1.1.3(B) and 4.1.1.3(A) respectively, will apply.

## 4.1.10.3.6 Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

#### 4.1.10.4 Discounts 1/

<u>Time-of-Day Discounts</u> - A 20 percent discount will apply to all inbound and outbound traffic occurring during the Evening and Night/Weekend rate periods (i.e. anytime except 8AM - 5PM weekdays). This discount does not apply to usage by customers of MCI Vision Power Rate. When a Vision call is established in one rate period and ends in another, the rate in effect in each rate period applies to the portion of the call occurring during that rate period. The rate for inbound calls is the rate for the time of day at the terminating location.

<u>Dedicated Termination Discount</u> <u>2</u>/ - Upon a customer's written or verbal request, all Vision Switched, Dedicated, and Card Access traffic, which is transmitted between locations of a customer and terminated by means of dedicated termination facilities will receive a discount of \$0.02 per minute. This discount will be applied before any other discounts.

Vision customers who subscribe to Intrastate Dedicated Leased Line Services and who also subscribe to MCI's Interstate Vision VIP Plus are eligible to receive discounts, as described in MCI's F.C.C. Tariff No. 1, on these intrastate data services.

The Dedicated Termination Discount will continue to be available to customers after May 11, 1996.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

#### 4.1.10.4 Discounts (Cont.)

<u>Volume Discounts</u> - Volume Discounts, as set forth below, are available to those Vision customers whose total monthly combined inbound 800 and outbound usage and surcharge (not including the service fee or optional features charged) equal or exceeds \$1,000.00 per invoice. The discounts shown apply on an incremental basis and only for amounts in excess of each usage threshold. This discount does not apply to usage by customers of MCI Vision Power Rate.

Total Monthly Usage	Volume Discount
\$ 0.00 - \$ 999.99	0%
\$ 1,000.00 - \$ 4,999.99	5%
\$ 5,000.00 - \$ 9,999.99	8%
\$ 10,000.00 - \$ 19,999.99	11%
\$ 20,000.00 +	14%

<u>Domestic Optimizer Discount</u> - A discount of 10 percent will be applied to all outbound Vision calls which are made to the single area code with the highest total usage (in dollars) in the period for which a bill is being calculated. This discount is calculated and applied at the location level. Directory Assistance and MCI Vision Power Rate calls are excluded from this discount. This discount is calculated on usage to that area code before all other discounts other than the Time of Day and Dedicated Termination Discounts are applied.

#### MCI Vision IntraLATA Usage Incentive 1/

#### Option B

New and existing Vision customers with intraLATA monthly usage as a result of accessing MCI intraLATA service via a PBX will be given a credit of up to \$250.00 towards MCI intraLATA usage. In order to be eligible for this credit, each location must generate a minimum of \$150.00 in intraLATA usage at least once during the first three months of use following enrollment in the program.

The credit will be applied on the fourth month's invoice following enrollment in the program. Customers who terminate MCI service prior to receipt of the credit or fail to meet the \$150.00 Vision outbound intraLATA requirements, will not receive the credit associated with this program.

Each location may enroll in Option B only once. Locations which previously enrolled in the Vision IntraLATA Usage Incentive are not eligible.

<sup>1/</sup> The MCI Vision IntraLATA Usage Incentive will continue to be available to customers after May 11, 1996.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

# 4.1.10.5 Vision Service Package Fee 1/

The Vision Service Package, to which all Vision customers subscribe, includes the availability of Dedicated and/or Switched Access with Volume Discounts, inbound 800, Time-of-Day Discounts, and Long Distance Directory Assistance, plus the following:

## For Inbound 800 and Outbound Service:

Call Detail on Paper and/or Magnetic Tape - Call detail records, including the prediscounted price for each call. One type of Call detail medium is included in this Package Fee; however, the customer may receive both Paper and Magnetic Tape at these additional costs: \$25 one-time set up fee, \$25 per month.

#### For Outbound Service:

Accounting Codes (up to 7-digits) - A customer can specify that Accounting Codes are to be dialed from specific dedicated access groups and Dial "1" telephone numbers (ANIs).

Universal Calling Range Privileges - A customer can specify the type of Vision calls allowable for users on each Dedicated Access Line group, for each Vision Card authorization code and for each Dial "1" ANI (originating telephone number) and for each ID code. The range privileges are defined as follows:

- 0 = no calls allowed
- 2 = the United States
- 3 = Range 2 plus all other North American Numbering Plan Locations <u>2</u>/ and all MCI served international countries
- 4 = Range 2 plus all other North American Numbering Plan Locations <u>2</u>/

\_

<sup>1/</sup> The Vision Service Package feature will continue to be available to customers after May 11, 1996.

<sup>2/</sup> The North American Numbering Plan Locations includes the continental U.S., Hawaii, Alaska, Canada, Mexico, Puerto Rico and the U.S. Virgin Islands.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

## 4.1.10.5 Vision Service Package Fee (Cont.)

The monthly MCI Vision Service Package Fee is \$50 per customer per month for those customers whose monthly usage is \$450 or less. The MCI Vision Service Package Fee for customers whose monthly usage exceeds \$450 but is less than \$500 will be charged the difference between the customers' actual usage in that month and \$500. Customers billing whose monthly usage exceeds \$500 will not be charged a monthly MCI Vision Service Package Fee.

All customers billing over \$500 per month will pay no monthly MCI Vision Service Package Fee.

# 4.1.10.6 Optional Features 1/

#### 4.1.10.6.1 For Outbound Service

<u>Direct Termination Overflow (DTO)</u> <u>2</u>/: Allows the customer to designate that calls to specific terminating dedicated trunk groups are to be alternately routed over shared MCI facilities or to another dedicated trunk group, if all the trunks in the target terminating trunk group are busy.

<u>Interswitch DTO Call Segment Charges</u>: Customer pays for two call segments for calls where the originating MCI Switch and the serving MCI Switch of the target terminating trunk group are different.

1/ Optional Features will continue to be available to customers after May 11, 1996.

2/ For inbound 800 service, these charges apply.

ISSUED: November 13, 2008

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

#### 4.1.10.6 Optional Features (Cont.)

### 4.1.10.6.1 For Outbound Service (Cont.)

Segment 1: From the originating point to target terminating dedicated

trunk group. Switched Access, Dedicated Access, or Vision Card Access charges (Section 4.1.10) and the Dedicated Termination Discount (Section 4.1.10) will

apply.

Segment 2: From the target terminating dedicated trunk group to

destination telephone number or terminating dedicated trunk group. Dedicated Access charges (Section 4.1.10) will apply if destination telephone number or terminating dedicated trunk group is over 100 miles from target terminating dedicated trunk group. The Dedicated

Termination Discount will be taken if applicable.

<u>Intraswitch DTO Call Segment Charges</u>: Customer pays for one call segment where the originating MCI Switch and the serving MCI Switch of the target terminating trunk group are the same.

From the originating switch to destination telephone number or to terminating dedicated trunk groups, Switched Access, Dedicated Access, or Vision Card Access charges and the Dedicated Termination Discount (Section 4.1.10) will be assessed where applicable for intrastate calls.

MCI Vision Invoicing 1/ - A customer must choose between two options of invoicing. With Consolidated Invoicing, a single monthly invoice will be rendered for all locations of a Corporate Enterprise. With the Location Level Invoicing option, each customer's account under the Corporate Enterprise will receive an individual invoice and will be responsible for remitting payment for usage and charges associated with that particular account. The Corporate Enterprise must accept contingent financial responsibility for all invoice charges to accounts electing this option.

MONTHLY

RECURRING INSTALLATION

Consolidated Invoicing \$0 \$0

Location Level Invoicing \$25/Corporate \$3/Invoice

Enterprise

4.1.10.6.2 Additional optional features must be purchased from MCl's F.C.C. Tariff No. 1.

1/ For inbound 800 service, these invoicing options apply.

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

#### 4.1.10.7 Vision Term Incentive

Beginning April 15, 1995 and ending June 14, 1995, MCI will offer the following as a response to competitive circumstances affecting a specific prospect and, as such, is available to that prospect and to all similarly situated prospects. Specifically, this incentive is responsive to an offer made by another interexchange carrier to an MCI prospect, when it can demonstrate to MCI's reasonable satisfaction that it will accept the competitor's offer in the absence of any further inducement from MCI to subscribe to MCI Metered Use Service Option J. Eligible customers must also be enrolled in the Vision Term Plan offered in MCI Tariff F.C.C. No. 1.

The following per minute rates on metered usage charges for intrastate inbound and outbound traffic based on term commitment and access type will apply.

	2 Ye		3 Year	
Switched	<u>Peak</u>	<u>Off-Peak</u>	<u>Peak</u>	Off-Peak
Outbound: Inbound:	\$.2545 \$.2636	\$.2291 \$.2382	\$.2455 \$.2545	\$.2218 \$.2309
<u>Dedicated</u>				
Outbound: Inbound:	\$.1455 \$.1546	\$.1273 \$.1364	\$.1364 \$.1455	\$.1200 \$.1291
<u>Card</u>	\$.4545	\$.4545	\$.4545	\$.4545

No surcharge will apply for MCI Vision card calls. Time of Day, Dedicated Termination, Volume Discounts, and Optimizer discounts will not apply. The customer will be charged standard tariffed rates for all other Vision usage.

If the percentage of the Customer's intrastate usage (as measured in minutes of use) of MCI Vision (outbound service only) is greater that 50 percent of the Customer's total usage, the Customer will pay a surcharge of \$.05 per minute for all intrastate usage.

Customers participating in this program are not eligible to receive the benefits of any other incentive programs shown in this tariff.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

#### 4.1.10.8 Vision Non-Term Incentive

Beginning April 15, 1995 and ending June 14, 1995 MCI will offer the following as a response to competitive circumstances affecting a specific prospect and, as such, is available to that prospect and to all similarly situated prospects. Specifically, this incentive is responsive to an offer made by another carrier to an MCI prospect, when the prospect can demonstrate to MCI's reasonable satisfaction that it will accept the competitor's offer in the absence of any further inducement from MCI to subscribe to MCI Metered Use Service Option J.

The following per minute rates on metered usage charges for intrastate inbound and outbound traffic will apply.

Switched		Dedicated		
	<u>Peak</u>	Off-Peak	<u>Peak</u>	Off-Peak
Outbound: Inbound:	\$.2727 \$.2818	\$.2727 \$.2818	\$.1636 \$.1727	\$.1636 \$.1727
<u>Card</u>	\$.4545	\$.4545	\$.4545	\$.4545

No surcharges will apply for Vision intrastate calling card calls. Time of Day, Dedicated Termination, Volume Discounts, and Optimizer discounts will not apply.

The customer will be charged the rates and receive the discounts specified herein for one year.

If the percentage of the Customer's intrastate usage (as measured in minutes of use) of MCI Vision (outbound service only) is greater that 50 percent of the Customer's total usage, the Customer will pay a surcharge of \$.05 per minute for all intrastate usage.

Customers participating in this program are not eligible to receive the benefits of any other incentives shown in this tariff.

# 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

## 4.1.10.9 Government Pricing Plan (GPP) 1/

The following rates will be applied for customers of Type 11, Type 12, Type 13, and Type 14 GPPs as described in MCI's F.C.C. Tariff No. 1, Section C-16.

# 4.1.10.9.1 <u>Switched Usage Rates</u>

DAY	DAY NON-DAY		PΑΥ
Initial	Additional	Initial	Additional
18 Seconds	6 Seconds	18 Seconds	6 Seconds
\$0.0562	\$0.0187	\$0.0422	\$0.0141

#### 4.1.10.9.2 <u>Dedicated Usage Rates</u>

DAY		NON-	NON-DAY	
Initial 18 Seconds	Additional 6 Seconds	Initial <u>18 Seconds</u>	Additional 6 Seconds	
\$0.0360	\$0.0120	\$0.0285	\$0.0095	

\_\_\_

<sup>1/</sup> Government Pricing Plans will continue to be available to customers after May 11, 1996.

ISSUED: November 13, 2008

- XIII. GRANDFATHERED SERVICES: SUBSECTION A
- 4.1 METERED USE SERVICE (Cont.)
  - 4.1.12 Option L (MCI 900 Service) 1/
    - 4.1.12.1 Terminating Access Line Charges: An MCI 900 customer must use Dedicated Access at each 900 Service terminating location. The access must provide hardware answer supervision.

#### **Dedicated Termination**

T-1 Digital Access - Charges associated with T-1 Digital Access are billed in accordance with the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com (Companion Interstate Service).

Special Access Surcharge 2/ - Applicable for Dedicated Terminating Access to MCI 900 Service.

Dedicated Termination: The monthly charges for Special Access Surcharge for each voice grade equivalent channel are shown in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com (Companion Interstate Service). (No non-recurring charges apply.)

4.1.12.2Per Minute Usage Charges: The rates will be in effect seven days a week and at all times of the day. Calls will be billed at a minimum of 30 seconds and rounded thereafter, to the next higher six second increment.

<u>1ST 30 SEC</u>	ADD'L 6 SEC
\$0.1550	\$0.0310

Effective/Beginning December 1, 2007, this service is grandfathered and is no longer available to new 1/ customers.

EFFECTIVE: November 14, 2008

<sup>2/</sup> The Special Access Surcharge, imposed by the Local Exchange Carrier, will not apply to these customers who furnish MCI with an Exemption Certificate (as defined herein).

#### 4.1 METERED USE SERVICE (Cont.)

- 4.1.12 Option L (MCI 900 Service) (Cont.)
  - 4.1.12.3 Volume Discounts: Volume Discounts, as specified below, will apply to all 900 Service per minute usage charges based on the total monthly usage (interstate and intrastate combined).

Monthly Usage	Incremental Volume Discount
\$ 0.00 - 69,999.99	0%
\$ 70,000.00 - 139,999.99	6%
\$ 140,000.00 +	12%

#### Example of application:

Eligible Gross Monthly Usage	Discount %	Usage <u>Discount \$</u>	Invoiced
\$120,000,00	6%	\$7 200	\$112 800

4.1.12.4 Feature Charges: All optional features will be billed in accordance with MCl's F.C.C. Tariff No. 1, Section C-3.152.

# 4.1.12.5 Service Availability

Service may be limited in certain areas as determined by MCI's F.C.C. Tariff No. 1.

# 4.1 METERED USE SERVICE (Cont.)

#### 4.1.16 Option P (MCI HotelDirect) 1/

## 4.1.16.1 Per Minute Usage Charges

## A. <u>Outbound Service</u>

Switched	Off-Peak	<u>Dedicate</u>	<u>d</u>
Peak		<u>Peak</u>	Off-Peak
\$0.1755	\$0.1404	\$0.1134	\$0.0907

#### B. Inbound Service

<u>Switched</u>		<u>Dedicated</u>	
<u>Peak</u>	Off-Peak	<u>Peak</u>	Off-Peak
\$0.2034	\$0.1627	\$0.1247	\$0.0998

- C. <u>Operator Assistance</u>: Service is provided according to the rovisions and rates as described in Section 4.1.1.
- D. <u>Calling Card Access</u>: Calling Card Access rates and surcharges are charged pursuant to the terms and conditions as outlined in MCl's F.C.C. Tariff No. 1.
- E. <u>Directory Assistance</u>: An undiscounted charge per call will be applied to each Directory Assistance call, subject to the provisions set forth in Section 2.6.4 herein.

## F. Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

ISSUED: November 13, 2008

<sup>1/</sup> Beginning June 11, 1997, Option P will not be available to new subscribers.

# 4.1 METERED USE SERVICE (Cont.)

#### 4.1.18 Option R (MCI MASTERS)

## 4.1.18.1 Usage Charges

On-Site: The following per minute usage rates apply:

<u>DAY</u> <u>EVENING</u> <u>NIGHT/WEEKEND</u>

\$0.2207 \$0.1644 \$0.1204

Off-Site: The following per minute usage rates apply:

<u>DAY</u> <u>EVENING</u> <u>NIGHT/WEEKEND</u>

\$0.30 \$0.15 \$0.09

#### Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any

charge element is \$.01.

- 4.1.18.2 <u>Directory Assistance:</u> An undiscountable charge will be applied to each call requesting Directory Assistance subject to the provisions set forth in Section 2.6.4 herein.
- 4.1.18.3 <u>DialHome</u>: DialHome allows MCI MASTERS customers to be eligible to place collect calls from on- or off-site telephones. The called party who accepts charges for the DialHome call is the customer for service under this Tariff. Customers will be charged the Off-Site rates listed above and the access surcharge in Section 4.1.18.4 for each such call.

<u>Access Charges:</u> A per call surcharge of \$.75 will apply to each MCI MASTERS off-site call. The surcharge will also apply when placing Directory Assistance calls.

Optional Calling Plan: MCI MASTERS customers may enroll in the following Optional Calling Plan:

4.1.18.5 Off-Site Calling Plan Option 1: A variation of Option R MCI MASTERS, Off-Site Calling Plan Option 1 offers off-site calling within the state. In lieu of the per minute usage rates specified in section 4.1.18.1, customers enrolled in the Off-Site Calling Plan Option 1 will be charged the following per minute rates for off-site calls.

<u>DAY</u>	<u>EVENING</u>	NIGHT/WEEKEND
\$0.2538	\$0.1891	\$0.1385

In addition, in lieu of the per call surcharges specified in section 4.1.18.4 customers enrolled in the Off-Site Calling Plan Option 1 will be charged the following per call surcharge for off- site calls.

Per Call Surcharge \$0.40

\_, \_\_, ... . \_

\_ . . .

# 4.1 METERED USE SERVICE (Cont.)

#### 4.1.19 Option S (University Dial 1)

# 4.1.19.1 Usage Charges

The following per minute usage charges apply to all calls originating and terminating within the state:

<u>Peak</u> <u>Off-Peak</u> \$0.2200 \$0.1000

Peak rates apply from 7:00 a.m. - 6:59 p.m., Monday through Friday; Off-Peak rates apply at all other times.

#### Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

#### 4.1.19.2 Features

4.1.19.2.1 <u>Directory Assistance</u>: Directory Assistance will be provided as specified in Section 2.6.4.

<u>Personal Identification Number (PIN)</u>: Each ANI will be assigned one or more four-digit PINs which must be used to access service. Invoices will be provided at the PIN level. There is no limit to the number of PINs assigned to an ANI. There is no additional charge for this feature.

# 4.1 METERED USE SERVICE (Cont.)

#### 4.1.22 Option V (hospitalityMCI) 1/

# 4.1.22.1 Per Minute Usage Charges 2/

(A) <u>Outbound Service</u> - The following per minute charges will apply to all outbound calls made within the state based on access type and time of day.

	<u>Peak</u>	Off-Peak
Switched Access and Card	\$0.1476	\$0.1098
Dedicated Access	\$0.0895	\$0.0665

(B) <u>Inbound Service</u> - The following per minute charges apply to inbound service calls as described in Section 3.5.8, based on termination type and time of day.

\$0.1098
\$0.0665
_

- (C) <u>Card Access Surcharges</u> A per call surcharge applies to each hospitalityMCI calling card call, other than calls to Directory Assistance, in addition to the rates in Section 4.1.22.1(A) above. The applicable surcharge can be found in MCI's F.C.C. Tariff No. 1.
- (D) Other Charges Customers may receive Inbound Features as described in Section 3.5.8 at the rates described in that section. In all instances, no discounts apply.
- (E) <u>Directory Assistance</u> An undiscounted charge per call will be applied to each Directory Assistance call, subject to the rate and provisions set forth in Section 2.6.4 herein.
- (F) Operator Services For intrastate calls placed with the assistance of an MCI operator, the rates and surcharges described in 4.1.1.3(B) and 4.1.1.3(A) respectively, apply.
- (G) Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

#### 4.1.22.2 Service Availability

hospitalityMCI service is available to customers from all cities within the state.

<sup>1/</sup> Beginning June 11, 1997, Option V will not be available to new subscribers.

<sup>2/</sup> Peak rates will apply between 8 a.m. and 5 p.m. weekdays, and Off-Peak rates will apply at all other times.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) 1/

## **Promotional Offerings**

#### MCI Vision Dial Access Promotion

Beginning May 11, 1996 and ending September 30, 1996, MCI will offer the following promotion via direct solicitation only to customers of Metered Use Service Option J (MCI Vision).

MCI will provide a usage credit or credits, the total of which may not exceed \$3,300 per participating location, according to the chart below.

Switched Outbound ANIs Enrolled per Location	Requirement A	Requirement B	<u>Credit</u>
1 - 4	\$ 40	\$ 40	\$ 550
5 - 8	80	80	1,100
9 - 12	120	120	1,650
13 - 16	160	160	2,200
17 - 20	200	200	2,750
21 - 24	240	240	3,300

# Where:

Requirement A reflects the minimum monthly interLATA usage averaged quarterly per participating location for twelve full months after the promotion location enrollment effective date; and Requirement B reflects the minimum monthly incremental (compared to the full month prior to the promotion location enrollment effective date) switched outbound intraLATA usage, excluding calling card usage, averaged quarterly per participating service location for twelve full months after the promotion location enrollment effective date.

For the purpose of this promotion, quarterly shall mean three-month periods beginning on a promotion location enrollment effective date.

Each customer's usage per location will be evaluated quarterly to determine if the above-stated requirements are satisfied. If the customer does not meet Requirements A and B in any quarter for 12 months after the promotion location enrollment effective date, the customer will be billed and required to pay an underutilization charge equal to the difference in the quarterly Requirement A and Requirement B minimums and the associated actual usage for that quarter.

\_

<sup>&</sup>lt;u>1</u>/ Beginning May 11, 1996, Option J (MCI Vision) will no longer be available to new subscribers unless otherwise noted herein.

## 4.1 METERED USE SERVICE OPTIONS (Cont.)

#### 4.1.10 Option J (MCI Vision) (Cont.)

#### Promotional Offerings (Cont.)

#### MCI Vision Dial Access Promotion (Cont.)

Customers who terminate service at any location(s) prior to one year from the promotion location enrollment effective date will be billed and required to pay an early termination charge equal to the underutilization charge for all remaining quarters in that year (or pro rata portion thereof), not to exceed the promotional credits received to date for that location.

The credit(s) will appear on the invoice reflecting the location's second full month's invoice after the promotion location enrollment effective date, with no credits being reflected prior to the October 1995 invoice.

Unless otherwise specified, customers enrolled in a Special Customer Arrangement other than Types 5, 17, 19, 20, and 21, as described in MCI's F.C.C. Tariff No. 1, are not eligible to receive the benefits of this promotion.

## 4.1.25 Option Y (networkMCI One)

#### **Promotional Offerings**

## (A) <u>networkMCI One Short Haul Promotion</u>

Beginning May 15, 1997 and ending March 13, 1998 MCI will offer the following promotional rates to networkMCI One customers that enroll in this program and use dedicated access-switch termination to access the MCI network. MCI will offer the following rates on Intralata outbound traffic originating and terminating within the specified mileage band in lieu of those rates specified in Section 4.1.25.3.

<u>Mileage</u>	<u>Rate</u>		
0 - 50	\$.0400		

The Short-Haul Intralata rates listed above are available throughout the remainder of the customer's current networkMCI One contract. Rates for Intralata calling terminating outside of the specified mileage band may vary during this period. At the expiration of this customer contract, Intralata rates for all mileage bands will revert to networkMCI One tariffed rates. These promotional rates are not available to customer locations that fall within the Local Service Area as defined in MCI Metro Local Exchange Tariffs. Customers enrolling in this promotion are not eligible for the networkMCi One Sign-Up Bonus, Install Waiver, New Access Promotion, or Access Investment Plan Under Race for Success, but may enroll in networkMCI One Power Portfolio as described in MCI's FCC #1.

#### 4.1 METERED USE SERVICE (Cont.)

#### 4.1.25 Option Y (networkMCI One)

## 4.1.25.1 Access/Termination Methods and Charges (Outbound and Inbound Service)

#### (A) **Dedicated Access/Termination**

### T-1 Digital Access Termination

T-1 Digital Access is available with networkMCI One. Monthly and non-recurring charges are specified in Section 4.2.2.2 as in effect on the date of the invoice.

In addition to the charges associated with these components, MCI will assess charges for the following option, if applicable:

T-1 Digital Access Features: Access Integration as specified in Section C-2.02216 of MCI's Tariff F.C.C. No. 1.

In addition, a networkMCI One customer may allow another networkMCI One customer to terminate calls on the first customer's network via dedicated access. This feature is available at the request of the networkMCI One customer on whose network the calls are to terminate. Such requests must be submitted to MCI in writing. Other customers authorized to make such dedicated access calls must dial a private dialing plan number.

#### Analog Access/Termination

Analog Local Access is available with networkMCl One. Monthly and nonrecurring charges are specified in Section 4.2.1 as in effect on the date of the invoice.

In addition to the charges associated with these components, MCI will assess charges for the following option, if applicable:

Analog Access Functions: Access Integration as specified in Section C-2.02115 of MCI's Tariff F.C.C. No. 1.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

- XIII. GRANDFATHERED SERVICES: SUBSECTION A
- 4.1 METERED USE SERVICE (Cont.)
- 4.1 METERED USE SERVICE (Cont.)
  - 4.1.25 Option Y (networkMCI One) (Cont.)
    - 4.1.25.1 Access/Termination Methods and Charges (Outbound and Inbound Service (Cont.)
      - (B) Shared Access

# networkMCI One Switched Outbound Location Minimum Charge:

The minimum charge applies to each networkMCI One customer location using switched access. Each customer's networkMCI One usage at each location, during each month, must equal or exceed \$5 or the customer will be charge the difference between the customer's actual usage charges and \$5. All networkMCI One usage and surcharges associated with each such location will apply in satisfaction of the minimum charge.

## networkMCI One Card Access

NetworkMCI One Card Access (via a networkMCI One Card) may originate and terminate from any touch-tone phone in the state via a toll free number.

<u>networkMCI One Card Surcharge</u>: The networkMCI One Card per call surcharge of \$0.85 will apply.

<u>networkMCI One Toll Free Remote Access</u>: Allows access to the customer's networkMCI One network via a customer-specific toll free number from anywhere in the state. This feature is available for on-net calls, with both 7- and 10-digit dialing plans, via switched access only. This feature is not available via networkMCI One Card Access.

<u>networkMCI One Toll Free Access Surcharge</u>: For networkMCI One Toll Free Access Remote calls which originate within the state, the applicable per call surcharge of \$0.25 will apply. This per-call charge applies in addition to the applicable charges specified in Section 4.1.25.3.

## 4.1 METERED USE SERVICE (Cont.)

## 4.1.25 Option Y (networkMCI One) (Cont.)

## 4.1.25.2 Inbound Service

The access methods and charges specified in Section 4.1.8 apply for networkMCI One Toll Free Service unless otherwise specified.

4.1.25.3 Per Minute Usage Charges: The following per-minute rates apply to outbound networkMCI One and inbound networkMCI One Toll Free Service usage within the state. Outbound networkMCI One usage charges are based on origination type. Inbound networkMCI One Toll Free Service usage charges are based on termination type.

Access Type	InterLATA Rate	IntraLATA Rate
Switched Origination or Termination	\$0.1419	\$0.1127
Dedicated Origination or Termination	\$0.0836	\$0.0836
networkMCI One Card Access	\$0.1419	\$0.1127

## Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

- 4.1.25.4 Operator Assistance: Operator Assistance is available for outbound calls made by customers using switched, including networkMCI One Card Access, and dedicated access only. Applicable usage rates and surcharges are as specified in Sections 4.1.1.3(B) and 4.1.1.3(A), respectively.
- 4.1.25.5 <u>Directory Assistance</u>: An undiscountable per call charge will be applied to each call requesting Directory Assistance for number within the state. This charge is as specified in Section 2.6.4 herein.
- 4.1.25.6 <u>Discounts and Optional Features</u>: Additional Discounts and Optional Features not listed in this tariff, must be purchased out of MCI's Tariff F.C.C. No. 1.

#### 4.1.25.6.1 <u>Market Area Calling 1/</u>

networkMCI One Market Area Calling: Offers intralata, outbound, voice only calling within the customer's Market Area for calls using dedicated access-switched termination and dedicated access-dedicated termination. This call type is not available to customer locations that fall within the local service areas of Perrine, Miami, Homestead, North Dade, Fort Lauderdale, Coral Springs, Deerfield Beach, Pompano Beach, Hollywood, Debary, East Orange, Geneva, Orlando, Oviedo, Sanford, Lake Mary, Tampa, Clearwater, Plant City, Zephyrhills, St. Petersburg, New Port Richey, Hudson, Tarpon Springs, Boca Raton, Delray Beach, and Boynton Beach.

The rates specified in section 4.1.25.3 IntraLATA (Dedicated origination or Termination) will apply to all Market Area calls originating and terminating within the Customer's Market Area. A customers Market Area is defined as the dominant Local Exchange Carrier's (LEC) local calling area specified in the LEC's Local Exchange Tariff for the customer's exchange area for the specific location implementing Market Area Calling.

1/ Beginning November 15, 1998, Market Area Calling will no longer be available to new subscribers.

# 4.1 METERED USE SERVICE (Cont.)

#### 4.1.25 Option Y (networkMCI One) (Cont.)

#### 4.1.25.6 Discounts and Optional Features (Cont.)

#### 4.1.25.6.1 Market Area Calling (Cont.)

In addition, networkMCI One discounts as outlined in MCI's FCC Tariff No. 1, Section C-3.41 (excluding the Dedicated Termination Discounts) will apply. If a customer subscribes to a Special Customer Arrangement with intrastate discounts or rates then those rates and discounts will apply to Market Area Calling. SCA Types 1 and 4 are not eligible for Market Area Calling.

Customers must have Branch Exchange (PBX) or Hybrid Key System and originate Market Area calling traffic via dedicated access facilities to use this feature.

The Market Area Calling feature cannot be used to originate 911, Directory Assistance, operator services calls, toll free or to terminate inbound local calls. Customers must maintain LEC-provided lines or trunks to originate and terminate these call types. This feature is subject to the Company's existing capacity and network availability.

#### 4.1.25.7 <u>Non-Recurring Charges</u>:

The Rates and Charges can be found in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="https://www.verizonbusiness.com">www.verizonbusiness.com</a> (Companion Interstate Service).

Per circuit or T-1 channel installation charges will be based on the type of Local Access used and can be found in Section 4.2.

# 4.1.28 Option BB (MASTERS97)

Usage Charges: The following per minute usage rates apply:

<u>Day</u>	<u>Evening</u>	Night/Weekend
\$0.15	\$0.15	\$0.15

#### Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

<u>Directory Assistance:</u> An undiscountable charge of \$1.25 per call will be applied to each call requesting Directory Assistance in the state, subject to the provisions set forth in Section 2.6.4 herein.

## 4.1. METERED USE SERVICE (Cont.)

#### 4.1.31 Option EE (networkMCI Contact one-number service)

#### **Basic Features**

networkMCI Contact one-number service subscribers receive the following:

<u>Basic Routing</u>: Each Subscriber is assigned an ANI for the routing of calls placed by Calling Parties to the Subscriber.

<u>Direct Calling</u>: The Subscriber may designate either of two Direct Calling methods:

- 1. <u>Predefined Call Routing</u>: Permits a Subscriber to pre-establish in priority up to three telephone numbers.
- 2. <u>Scheduled Routing</u>: Permits a Subscriber to pre-establish in priority call routing capabilities based on a Subscriber-defined schedule (e.g., day-of-week, day-of-year, time-of-day) for up to two alternate telephone numbers for the completion of calls placed by Calling Parties to the subscriber.

<u>Final Routing</u>: Permits a Subscriber to designate various options for the call routing location. If calls are not completed at any of the Direct Calling alternate numbers, the Calling Party's call will be routed to the Subscriber-defined final routing location.

Override Routing: Permits Subscriber to override Basic Routing so that the call is routed directly to a location specified by the Subscriber. This feature can be activated and deactivated at any time by the Subscriber. When deactivated, Basic Routing is reactivated.

## 4.1. METERED USE SERVICE (Cont.)

#### 4.1.31 Option EE (networkMCI Contact one-number service) (Cont.)

#### Basic Features (Cont.)

<u>Call Screening</u>: Permits a Subscriber to either accept a call or to route a call to a specific destination instead of receiving the call. This feature can be activated and deactivated at any time by the Subscriber.

<u>Default Routing</u>: Permits a Subscriber to designate call routing for Calling Parties who do not respond to Basic Routing.

<u>Outbound Dialing</u>: Permits a Subscriber to access her or his networkMCI Contact one-number service number to call telephone numbers within Florida. Permits a Subscriber to designate and forward calls placed to her or his networkMCI Contact one-number service to telephone numbers within Florida.

#### **USAGE CHARGES**

<u>ANI Access Service</u>: The following usage charges will apply to calls to networkMCI Contact One-Number Service ANI Numbers:

## **IntraLATA**

- The Subscriber will be charged \$0.15 per minute for the calling party's intrastate intraLATA usage.
- The Subscriber will be charged \$0.20 per minute for the subscriber's intrastate intraLATA
  outbound dialing usage used to designate and forward calls placed to the subscriber's
  networkMCI Contact one-number service ANI.
- 3. The Subscriber will be charged \$0.20 per minute for intrastate intraLATA outbound dialing usage.

#### <u>InterLATA</u>

- The Subscriber will be charged \$0.15 per minute for the calling party's intrastate interLATA usage.
- The Subscriber will be charged \$0.20 per minute for the subscriber's intrastate interLATA
  outbound dialing usage used to designate and forward calls placed to the subscriber's
  networkMCI Contact one-number service ANI.

The Subscriber will be charged \$0.20 per minute for intrastate interLATA outbound dialing usage.

## 4.1. METERED USE SERVICE (Cont.)

#### 4.1.31 Option EE (networkMCI Contact one-number service) (Cont.)

#### **OTHER**

1. The subscriber will be charged \$.15 per call for the calling party's usage which originates and terminates within an area that has networkMCI Contact one-number service ANI access availability as described in the Service Availability Section.

The Subscriber will be charged \$.20 per call for the subscriber's outbound dialing usage, which originates and terminates within an area that has networkMCI Contact one-number service ANI access availability as described in the Service Availability Section, and is used to designate and forward calls placed to the subscriber's networkMCI Contact one- number service ANI.

# **USAGE CHARGES**

#### OTHER (Cont'd)

3. The Subscriber will be charged \$.20 per call for outbound dialing which originates and terminates within an area that has networkMCI Contact one-number service ANI access availability as described in the Service Availability Section.

The Subscriber will be billed in 60-Second increments for calls Which terminate to a numeric or Alphanumeric Display Device.

# Calling Assistance:

For Calling Party Usage which originates and terminates within Florida the Subscriber will be charged a \$1.15 per-call surcharge for routing assistance provided to the Calling Party.

The Subscriber will be charged a \$1.15 per-call surcharge for routing assistance for Outbound Dialing calls within Florida.

#### Service Availability

Option EE is available as follows:

A. networkMCI Contact one-number ANI Access is available in Boca Raton, FL, Ft. Lauderdale, FL, Miami, FL, Orlando, FL, Tampa, FL, and WinterPark, FL.

Option EE customers can terminate calls to all cities within the Florida.

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services)

#### **PROMOTIONAL OFFERINGS**

## **SCA Promotion**

In order to flow through cost reductions as required by the Florida Public Services Commission to customers with special customer arrangements in effect as of November 1, 2005, which SCAs, in the absence of this promotion, would prevent them from receiving the benefits of that flow-through, MCI will reduce the rates charged to those customers under those SCAs as set forth below. MCI may end this promotional rate reduction when it has completed flowing through the required amounts.

# Option 1

## Outbound:

Origination	Termination	Rate
<u>Type</u>	<u>Type</u>	Reduction
Local Network Connection	Local Network Connection	\$0.0169
Local Network Connection	Dedicated	0.0169
Local Network Connection	Switched	0.0169
Dedicated	Local Network Connection	0.0082
Dedicated	Dedicated	0.0082
Dedicated	Switched	0.0082
Switched/Card	Local Network Connection	0.0169
Switched/Card	Dedicated	0.0169
Switched//Card	Switched	0.0169

## Inbound:

Origination	Termination	Rate
<u>Type</u>	<u>Type</u>	Reduction
Local Network Connection	Local Network Connection	\$0.0169
Local Network Connection	Dedicated	0.0082
Local Network Connection	Switched	0.0169
Switched	Local Network Connection	0.0169
Switched	Dedicated	0.0082
Switched	Switched	0.0169

# 4.1. METERED USE SERVICE (Cont.)

# 4.1.33 Option GG (MCI WorldCom On-Net Services)

# **PROMOTIONAL OFFERINGS**

SCA Promotion (CONT.)

# Option 2

# Outbound:

Origination	Termination	Rate
<u>Type</u>	Type	Reduction
Local Network Connection	Local Network Connection	\$0.0169
Local Network Connection	Dedicated	0.0086
Local Network Connection	Switched	0.0169
Dedicated	Local Network Connection	0.0082
Dedicated	Dedicated	0.0000
Dedicated	Switched	0.0082
Switched/Card	Local Network Connection	0.0169
Switched/Card	Dedicated	0.0086
Switched//Card	Switched	0.0169

# Inbound:

Origination	Termination	Rate
Type	<u>Type</u>	Reduction
Local Network Connection	Local Network Connection	\$0.0169
Local Network Connection	Dedicated	0.0082
Local Network Connection	Switched	0.0169
Switched	Local Network Connection	0.0169
Switched	Dedicated	0.0082
Switched	Switched	0.0169

#### 4.1. METERED USE SERVICE (cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services) 1/

#### **Usage Charges**

#### **Definitions**

The following definitions will apply for all usage rates:

"Local Network Connection": A switched connection between a customer's premises and an MCI Point of Presence which is provided by MCI or an MCI affiliate.

"Dedicated": A non-switched connection between a customer's premises and an MCI Point of Presence.

"Switched": A switched connection between a customer's premises and an MCI POP which is not provided by MCI or an MCI affiliate.

Card: Call origination using an Option GG Card.

# **Intrastate Usage Rates**

Outbound Rates: The following per-minute rates will apply to Option GG usage. Usage charges are based on origination type.

Termination	Per-Minute
<u> Type</u>	Rate
Local Network Connection	\$0.0396
Dedicated	0.0396
Switched	0.0496
Local Network Connection	0.0423
Dedicated	0.0423
Switched	0.0523
Local Network Connection	0.0869
Local Network Connection	0.0969
Dedicated	0.0869
Switched	0.0969
	Type  Local Network Connection Dedicated Switched Local Network Connection Dedicated Switched Local Network Connection Local Network Connection Dedicated

<sup>1/</sup> Beginning January 1, 2004, this service will no longer be available to new subscribers.

## 4.1. METERED USE SERVICE (Cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services) (Cont.)

### Intrastate Usage Rates (Cont.)

<u>Inbound Rates:</u> The following per-minute rates will apply to Option GG usage. Usage charges are based on termination type.

OriginationType	TerminationType	Per-Minute <u>Rate</u>
Local Network Connection	Local Network Connection	\$0.0396
Local Network Connection	Dedicated	\$0.0423
Local Network Connection	Switched	0.0869
Switched	Local Network Connection	0.0496
Switched	Dedicated	0.0523
Switched	Switched	0.0969

#### Minimum Charge

The Company does not invoice charges in fractions of a cent; the minimum charge for any charge element is \$.01.

<u>Usage Rates for Operator Assisted Calling:</u> The following charges apply to outbound Operator Assisted calls made by customers using switched (including Option GG Card Access), dedicated access, and local network connection access. For calls originating over Local Network Connection facilities (as defined herein), an origination credit of \$0.0460 per minute will apply. For calls terminating over Local Network Connection facilities (as defined herein), a termination credit of \$0.01 per minute will apply. The origination and termination credits described above do not apply to Option GG card origination.

## 4.1. METERED USE SERVICE (Cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services) (Cont.)

Usage Rates for Operator Assisted Calling (Cont.)

<u>Classification (A)</u>: For calls within classification (a), as specified in Section 4.1.1.3 (excluding collect calls and calls which are billed by a third party), the usage rates found in Section 4.1.1.3 will apply.

<u>Classifications (C) and (D):</u> For calls within classifications (c) and (d), as specified in Section 4.1.1.3 (excluding collect calls and calls which are billed by a third party), the usage rates found in Section 4.1.1.3 will apply.

Operator Services Surcharges: Except as specified below, for all calls in classifications (a), (c) and (d), as specified in Section 4.1.1.3 (excluding collect calls and calls which are billed by a third party), the undiscountable surcharges, as set forth in Section 4.1.1.3, will apply.

Operator Station \$2.10 Person to Person \$3.90		<u>Surcharge</u>
Operator Dialed \$1.15	Person to Person	\$3.90

<u>Directory Assistance:</u> An undiscountable charge per call will be applied to each call requesting Directory Assistance, subject to the rate and provisions set forth in Section 2.6.4 herein.

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services) (Cont.)

Usage Rates for Operator Assisted Calling (Cont.)

#### Card Access

Option GG Access (via an Option GG Card) may originate from any touch-tone phone in the United States via a toll free number. Usage charges as indicated in Section 4.1.33 will apply.

The following per-call surcharge will apply to each Option GG Card call, other than calls to Directory Assistance:

# **Surcharge**

Per Call \$0.90

<u>Toll Free Remote Access:</u> Allows access to the customer's Option GG network via a customer specific toll free number from anywhere in the U.S. This feature is available for calling within the customer's private network, with both 7- and 10-digit dialing plans, via switched access only. This feature is not available via Option GG Card Access.

The following charge applies to each Option GG Toll Free Remote Access call which originates and terminates within the state. This per-call charge applies in addition to applicable usage charges as indicated in Section 4.1.33:

Surcharge

Per Call \$0.25

## 4.1. METERED USE SERVICE (Cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services)

## Term Plan

The Option GG Term Plan is a term plan, in lieu of all other tariffed term plans. Customers who subscribe to Option GG through the Option GG Term Plan are subject to the terms and conditions outlined for Option RR in MCl's Tariff F.C.C. No. 1 except for the discount on eligible intrastate charges as defined below.

Eligible Intrastate charges is the customer's total intrastate usage of the following, after the application of promotional and other discounts: intrastate inbound and outbound usage and Option GG Card, Option GG Remote Access (including surcharges). Charges for the following are not included as eligible intrastate charges and will not receive Option GG Term plan Discounts: Directory Assistance usage and surcharges and Operator Assisted usage and surcharges.

Term Commitment/Discounts: The following discounts will apply to the eligible intrastate charges based on the customer's term commitment.

Term Commitment	<u>Discount</u>
1 year 2 year 3 year 4 year 5 year	3% 6% 9% 9% 9%

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services)

#### On-Net Plus Plan

Eligibility: To be eligible for this program, customers:

- must designate a company affiliate as its exchange service carrier and the company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intralata toll calling;
- must receive interstate service under Special Customer Arrangement (SCA) Guide Type
   1, 2, 3, 4, 5, 6, 7, 8, or 9 as described in the MCI's "Service Publication and Price Guide" located on the Company's website at <a href="https://www.verizonbusiness.com">www.verizonbusiness.com</a>
- must be new business customers or existing business customers who is eligible for renewal of their contracts;

# **Usage Charges:**

The following per-minute usage rates will apply to Intrastate Usage. Customers will be charged the following per minute rates for Inbound and Outbound Intrastate Service.

#### Outbound:

	1 Year	2 Year
Termination Type	Per Minute Rate	Per Minute Rate
Local Network Connection	\$0.0000	\$0.0000
Dedicated	\$0.0447	\$0.0423
Switched	\$0.0447	\$0.0423
Local Network Connection	\$0.0470	\$0.0445
Dedicated/Switched	\$0.0470	\$0.0445
Local Network Connection	\$0.0891	\$0.0851
Dedicated	\$0.0891	\$0.0851
Switched	\$0.0891	\$0.0851
Local Network Connection	\$0.0447	\$0.0423
Dedicated	\$0.0470	\$0.0445
Switched	\$0.0891	\$0.0851
Local Network Connection	\$0.0447	\$0.0423
Dedicated	\$0.0470	\$0.0445
Switched	\$0.0891	\$0.0851
	Local Network Connection Dedicated Switched Local Network Connection Dedicated/Switched Local Network Connection Dedicated Switched  Local Network Connection Dedicated Switched Local Network Connection Dedicated Switched Local Network Connection Dedicated	Termination Type Local Network Connection Dedicated \$0.0447 Switched \$0.0447 Local Network Connection Dedicated/Switched \$0.0470 Local Network Connection Dedicated \$0.0891 Switched \$0.0891 Switched \$0.0891  Local Network Connection \$0.0891 Switched \$0.0891  Local Network Connection \$0.0447 Dedicated \$0.0470 Switched \$0.0891 Local Network Connection \$0.0447 Dedicated \$0.0891 Local Network Connection \$0.0447 Switched \$0.0891 Local Network Connection \$0.0447 Switched \$0.0447 Dedicated \$0.0470

## 4.1. METERED USE SERVICE (Cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services)

#### Agent Program

Eligibility: To be eligible for this program, the Customer:

must subscribe to this program through a Company-designated agent;

may not receive any discounts or the benefits of any promotional offering;

may not receive service under any other term plan;

must subscribe to the WorldCom On-Net Voice Services Agent Program as described in WorldCom's "Service Publication and Pricing Guide" located on WorldCom's website at www.worldcom.com; and,

must subscribe to exchange service provided by the Company or a Company affiliate.

Term Commitment and Renewal Options: The Customer must commit to service for a term of either one or two years (Initial Term). The term of service will automatically renew on a month-to-month basis (Extended Term) upon expiration of its Initial Term and Extended Term(s), unless either the Customer or the Company provides written notification to terminate the term, which must be received by the other party no more than 60 days prior to the expiration of the existing term.

Usage Charges: The following per-minute rates apply during each monthly period of a Customer's term of service intrastate outbound usage which originates via Local Network Connection and terminates via switched and Toll Free usage that originates switched and terminates Local Network Connection.

Per-Minute Rate: 0.0690

Classifications, Practices and Regulations:

Termination: Either the Customer or the Company may terminate this program for Cause. For purposes of this program only: (i) as to payment of invoices, "Cause" shall mean the Customer's failure to pay any invoice within 30 days after the date of the invoice; and, (ii) for all other matters, "Cause" shall mean a breach by the other party of any material provision of this program, provided that written notice of the breach has been given to the breaching party, and the breach has not

been cured within 30 days after delivery of such notice.

## 4.1. METERED USE SERVICE (Cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services)

#### Intrastate Plus Service

Eligibility: To be eligible for this program, customers:

must designate a company affiliate as its exchange service carrier and the company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intralata toll calling;

must receive interstate service under a New Special Customer Arrangement (SCA) Guide Type 6, 7, 8, 9 or 10 as described in the WorldCom's "Service Publication and Price Guide" located on the Company's website at <a href="https://www.worldcom.com">www.worldcom.com</a>;

## **Usage Charges:**

The following per-minute usage rates will apply to Intrastate Usage. Customers will be charged the following per minute rates for Inbound and Outbound Intrastate Service.

ORIGINATION TYPE: OUTBOUND	TERMINATION TYPE: OUTBOUND	1 YEAR TERM	2 YEAR TERM	13 YEAR TERM
LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION	\$0.0000	\$0.0000	\$0.0000
LOCAL NETWORK CONNECTION	DEDICATED	\$0.0536	\$0.0521	\$0.0505
LOCAL NETWORK CONNECTION	SWITCHED	\$0.0536	\$0.0521	\$0.0505
DEDICATED	LOCAL NETWORK CONNECTION	\$0.0536	\$0.0521	\$0.0505
DEDICATED	DEDICATED OR SWITCHED	\$0.0536	\$0.0521	\$0.0505
SWITCHED / CARD	LOCAL NETWORK CONNECTION	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	DEDICATED	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	SWITCHED	\$0.0917	\$0.0890	\$0.0863
ORIGINATION TYPE: INBOUND	TERMINATION TYPE: OUTBOUND	1 YEAR TERM	2 YEAR TERM	1 3 YEAR TERM
ORIGINATION TYPE: INBOUND LOCAL NETWORK CONNECTION	TERMINATION TYPE: OUTBOUND LOCAL NETWORK CONNECTION	1 YEAR TERM \$0.0536	2 YEAR TERM \$0.0521	1 3 YEAR TERM \$0.0505
LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION	\$0.0536	\$0.0521	\$0.0505
LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION DEDICATED	\$0.0536 \$0.0536	\$0.0521 \$0.0521	\$0.0505 \$0.0505
LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION DEDICATED SWITCHED	\$0.0536 \$0.0536 \$0.0917	\$0.0521 \$0.0521 \$0.0890	\$0.0505 \$0.0505 \$0.0863
LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION SWITCHED / CARD	LOCAL NETWORK CONNECTION DEDICATED SWITCHED LOCAL NETWORK CONNECTION	\$0.0536 \$0.0536 \$0.0917 \$0.0536	\$0.0521 \$0.0521 \$0.0890 \$0.0521	\$0.0505 \$0.0505 \$0.0863 \$0.0505

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.33 Option GG (MCI WorldCom On-Net Services)

Intrastate Plus Service

Intrastate Plus Optional Calling Plan I

To receive the following benefits of Intrastate Plus Optional Calling Plan I, Customer must demonstrate to MCI reasonable satisfaction that it will accept a competitor's offer of service in place of Intrastate Plus if it does not receive the benefits of this plan. Customer may not receive any other rates, discounts or other benefits applicable to Intrastate Plus.

ORIGINATION TYPE: OUTBOUND	TERMINATION TYPE: OUTBOUND	1 YEAR TERM	2 YEAR TERM	13 YEAR TERM
LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION	\$0.0000	\$0.0000	\$0.0000
LOCAL NETWORK CONNECTION	DEDICATED	\$0.0505	\$0.0505	\$0.0505
LOCAL NETWORK CONNECTION	SWITCHED	\$0.0505	\$0.0505	\$0.0505
DEDICATED	LOCAL NETWORK CONNECTION	\$0.0505	\$0.0505	\$0.0505
DEDICATED	DEDICATED OR SWITCHED	\$0.0505	\$0.0505	\$0.0505
SWITCHED / CARD	LOCAL NETWORK CONNECTION	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	DEDICATED	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	SWITCHED	\$0.0917	\$0.0890	\$0.0863
ORIGINATION TYPE: INBOUND	TERMINATION TYPE: OUTBOUND	1 YEAR TERM	2 YEAR TERM	13 YEAR TERM
LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION	\$0.0505	\$0.0505	\$0.0505
LOCAL NETWORK CONNECTION	DEDICATED	\$0.0505	\$0.0505	\$0.0505
LOCAL NETWORK CONNECTION	SWITCHED	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	LOCAL NETWORK CONNECTION	\$0.0505	\$0.0505	\$0.0505
SWITCHED / CARD	DEDICATED	\$0.0505	\$0.0505	\$0.0505
SWITCHED / CARD	SWITCHED	\$0.0917	\$0.0890	\$0.0863

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.91 Option NNNN (Verizon Business Service I) (Cont.)

# Term Plan

The Option NNNN Term Plan is a term plan, in lieu of all other tariffed term plans. Customers who subscribe to Option NNNN through the Option NNNN Term Plan are subject to the terms and conditions outlined for Verizon Business Service I in the Company's "Service Publication and Price

Guide" located on the Company's website at www.verizonbusiness.com, except for the discount on eligible intrastate charges as defined below.

Eligible Intrastate charges is the customer's total intrastate usage of the following, after the application of promotional and other discounts: intrastate inbound and outbound usage and Option NNNN Card, Option NNNN Remote Access (including surcharges). Charges for the following are not included as eligible intrastate charges and will not receive Option NNNN Term plan Discounts: Directory Assistance usage and surcharges and Operator Assisted usage and surcharges.

Term Commitment/Discounts: The following discounts will apply to the eligible intrastate charges based on the customer's term commitment.

Term Commitment	<u>Discount</u>
1 year	3%
2 year	6%
3 year	9%
4 year	9%
5 year	9%

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.91 Option NNNN (Verizon Business Service I) (Cont.)

# Verizon Business Services I Optional Calling Plan

To receive the following benefits of Verizon Business Services I Option Calling Plan I, Customer must demonstrate to Verizon reasonable satisfaction that it will accept a competitor's offer of service in place of Verizon Business Services I if it does not receive the benefits of this plan. Customer may not receive any other rates, discounts or other benefits applicable to Verizon Business Services I.

#### **OUTBOUND:**

Origination	Termination			
<u>Type</u>	Type	<b>1 YEAR TERM</b>	<b>2 YEAR TERM</b>	3 YEAR TERM
Local Network Connection	Local Network Connection	\$0.0350	\$0.0350	\$0.0350
Local Network Connection	Dedicated	0.0350	\$0.0350	\$0.0350
Local Network Connection	Switched	0.0350	\$0.0350	\$0.0350
Dedicated	Local Network Connection	0.0350	\$0.0350	\$0.0350
Dedicated	Dedicated or Switched	0.0350	\$0.0350	\$0.0350
Switched/Card	Local Network Connection	0.0650	\$0.0650	\$0.0650
Switched/Card	Dedicated	0.0650	\$0.0650	\$0.0650
Switched/Card	Switched	0.0650	\$0.0650	\$0.0650

# INBOUND:

<u>/I</u>

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.95 Option RRRR (Verizon Business Service I Flex T1)

#### Definitions:

For this purpose of Verizon Business Services I Flex T1, the following definitions apply:

"Eligible Intrastate Service" is defined as Verizon Business Services I Voice Outbound Service (Excluding Card) usage that originated and terminates in one state.

"Eligible Interstate Usage" is defined as Verizon Business Services | Outbound Service (Excluding

card) usage that originates in the U.S. Mainland and Hawaii and terminated in the US Mainland, Alaska, Hawaii, Puerto Rico, the U.S. Virgin Islands, Guam and CNMI.

"New customers" are customers who at the time of subscription to the plan, are not receiving service under this tariff and the document containing the Company Local and Interstate Service.

"Renewing customers" are customers who at the time of subscription to the plan: (i) be receiving Company service under a term of service commitment with fewer than 3 monthly periods remaining in the term; and (ii) enter into a new term of service for Company service.

#### Offering:

In each monthly period in which the customer is eligible for this plan: (i) the Company will waive per-minute usage charges for the customer's Outbound service usage per Flex T1, not to exceed the following minutes of use, based on the customer's Product Package term of service; and, (ii) the Customer will be billed standard tariffed Verizon Business Service I per-minute usage charges

for the customer's Outbound service usage per Flex T1 which exceeds the following minutes of use, based on the customer's Product Package term of service:

## Term of Service (Years) Minutes of Use

1	200
2	500
3+	1,000

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.95 Option RRRR (Verizon Business Service I Flex T1) (Cont.)

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For new and renewing customers who disconnect Companion Local Service offered in MCImetro Access Transmissions Services LLC d/b/a Verizon Access Transmission Services, FL PSC Tariff No. 1 Companion Interstate Service under The Guide and Companion Intrastate Service offered under this tariff will terminate and the customer will be automatically subscribed to the Verizon Business Services I Voice Services under The Guide and Verizon Business Services I Voice Service under this tariff.

For new and renewing customers renewing who disconnect Companion Local Service offered in MCImetro Access Transmissions Services LLC d/b/a Verizon Access Transmission Services, FL PSC Tariff No. 1 and Companion Intrastate Service, Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to Verizon Business Services I Voice Services under The Guide and Verizon Business Services I under this tariff.

#### Other Conditions:

Services under this plan are not eligible to receive the benefits of any discounts or promotions including any term plan discounts other than Verizon Business Services I discounts for any LD usage over the allotted free minutes listed above.

The Verizon Business Services I Flex T1 cannot be used in conjunction with the Verizon Business

Services I Local and Long Distance Plan or the Verizon Business Services I Local and Long Distance Line Solution.

## 4.1. METERED USE SERVICE (Cont.)

#### 4.1.101 Option XXXX (Verizon Business Services II) (Cont.)

#### **Term Plan**

The Option XXXX Term Plan is a term plan, in lieu of all other tariffed term plans. Customers who subscribe to Option NNNN through the Option XXXX Term Plan are subject to the terms and conditions outlined for Verizon Business Service I in the Company's "Service Publication and Price

Guide" located on the Company's website at www.verizonbusiness.com, except for the discount on eligible intrastate charges as defined below.

Eligible Intrastate charges is the customer's total intrastate usage of the following, after the application of promotional and other discounts: intrastate inbound and outbound usage and Option XXXX Card, Option XXXX Remote Access (including surcharges). Charges for the following are not included as eligible intrastate charges and will not receive Option XXXX Term plan Discounts: Directory Assistance usage and surcharges and Operator Assisted usage and surcharges.

Term Commitment/Discounts: The following discounts will apply to the eligible intrastate charges based on the customer's term commitment.

Term Commitment	<u>Discount</u>
1 year	3%
2 year	6%
3 year	9%
4 year	9%
5 year	9%

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.101 Option XXXX (Verizon Business Services II) (Cont.)

# Verizon Business Services II Optional Calling Plan I

To receive the following benefits of Verizon Business Services II Option Calling Plan I, Customer must demonstrate to Verizon reasonable satisfaction that it will accept a competitor's offer of service in place of Verizon Business Services II if it does not receive the benefits of this plan. Customer may not receive any other rates, discounts or other benefits applicable to Verizon Business Services II.

#### **OUTBOUND:**

Origination		Termination			
<u>Type</u>		<u>Type</u>	1 YEAR TERM	2 YEAR TERM	3 YEAR TERM
Local Network Connecti	on	Local Network Connection	\$0.0360	\$0.0350	\$0.0345
Local Network Connecti	on	Dedicated	0.0360	\$0.0350	\$0.0345
Local Network Connecti	on	Switched	0.0360	\$0.0350	\$0.0345
Dedicated	Loc	al Network Connection	0.0360	\$0.0350	\$0.0345
Dedicated	Dec	dicated or Switched	0.0360	\$0.0350	\$0.0345
Switched/Card	Loc	al Network Connection	0.0750	\$0.0740	\$0.0730
Switched/Card	Dec	dicated	0.0750	\$0.0740	\$0.0730
Switched/Card	Swi	itched	0.0750	\$0.0740	\$0.0730

#### **INBOUND:**

Origination	Termination			
Type_	Type_	1 YEAR TERM	2 YEAR TERM	3 YEAR TERM
<b>Local Network Connection</b>	Local Network Connection	\$0.0360	\$0.0350	\$0.0345
<b>Local Network Connection</b>	Dedicated	0.0360	\$0.0350	\$0.0345
<b>Local Network Connection</b>	Switched	0.0750	\$0.0740	\$0.0730
Switched/Card Lo	cal Network Connection	0.0360	\$0.0350	\$0.0345
Switched/Card De	edicated	0.0360	\$0.0350	\$0.0345
Switched/Card Sw	vitched	0.0750	\$0.0740	\$0.0730

#### 4.1. METERED USE SERVICE (Cont.)

#### 4.1.103 Option ZZZZ (Verizon Business Services II Flex T1 Plan) 1/

#### Definitions:

For this purpose of Verizon Business Services Flex T1, the following definitions apply:

"Eligible Intrastate Service" is defined as Verizon Business Services II Voice Outbound Service (Excluding Card) usage that originated and terminates in one state.

"Eligible Interstate Usage" is defined as Verizon Business Services II Outbound Service (Excluding card) usage that originates in the U.S. Mainland and Hawaii and terminated in the US Mainland, Alaska, Hawaii, Puerto Rico, the U.S. Virgin Islands, Guam and CNMI.

"New customers" are customers who at the time of subscription to the plan are not receiving service under this tariff and the document containing the Company Local and Interstate Service.

"Renewing customers" are customers who at the time of subscription to the plan: (i) be receiving Company service under a term of service commitment with fewer than 3 monthly periods remaining in the term; and (ii) enter into a new term of service for Company service.

# Offering:

In each monthly period in which the customer is eligible for this plan: (i) the Company will waive per-minute usage charges for the customer's Outbound service usage per Flex T1, not to exceed the following minutes of use, based on the customer's Product Package term of service; and, (ii) the Customer will be billed standard tariffed Verizon Business Service II per-minute usage charges

for the customer's Outbound service usage per Flex T1 which exceeds the following minutes of use, based on the customer's Product Package term of service:

<u>: (Years)</u> Minutes of Use
200
500
1,000

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

<sup>1/</sup> Beginning April 1, 2005, this service will no longer be available to new subscribers.

# 4.1. METERED USE SERVICE (Cont.)

#### 4.1.103 Option ZZZZ (Verizon Business Services II Flex T1 Plan) (Cont.)

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For new and renewing customers who disconnect Companion Local Service offered in MCImetro Access Transmissions Services LLC d/b/a Verizon Access Transmission Services, FL PL No. 2 Companion Interstate Service under The Guide and Companion Intrastate Service offered under this tariff will terminate and the customer will be automatically subscribed to the Verizon Business Services

II Voice Services under The Guide and Verizon Business Services II Voice Service under this tariff.

For new and renewing customers renewing who disconnect Companion Local Service offered in MCImetro Access Transmissions Services LLC d/b/a Verizon Access Transmission Services, FL PL

No. 2 and Companion Intrastate Service, Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to Verizon Business Services II Voice Services under The Guide and Verizon Business

Services II under this tariff.

#### Other Conditions:

Services under this plan are not eligible to receive the benefits of any discounts or promotions including any term plan discounts other than Verizon Business Services II discounts for any LD usage over the allotted free minutes listed above.

The Verizon Business Services Flex T1 cannot be used in conjunction with the Verizon Business Services II Local and Long Distance Plan.

# 4.1. METERED USE SERVICE (Cont.)

### 4.1.109 Option GGGGG (Verizon Business Services III) (Cont.)

Term Plan

The VBS III Term Plan is a term plan, in lieu of all other tariffed term plans. Customers who subscribe to VBS III through the VBS III Term Plan are subject to the terms and conditions outlined for Verizon Business Service III in the Company's "Service Publication and Price Guide" located on the Company's website at www.verizonbusiness.com/publications/service\_guide, except for the discount on eligible intrastate charges as defined below.

Eligible Intrastate charges is the customer's total intrastate usage of the following, after the application of promotional and other discounts: intrastate inbound and outbound usage and VBS III, VBS III Remote Access (including surcharges). Charges for the following are not included as eligible intrastate charges and will not receive VBS III Term plan Discounts: Directory Assistance usage and surcharges and Operator Assisted usage and surcharges.

Term Commitment/Discounts: The following discounts will apply to the eligible intrastate charges based on the customer's term commitment.

Term Commitment	Discount
1 year 2 year 3 year 4 year 5 year	3% 6% 9% 9%
,	

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

### 3) DIGITAL DATA SERVICE (DDS) Millennium Tiered Intrastate Access Promotion

Beginning May 1, 2000 and ending November 30, 2000, MCI WorldCom will offer the following promotion to new and existing customers of Dedicated Lease Line Service in Florida who install a new Digital Data Service local access circuit that is associated with use of new intrastate MCI WorldCom Communication Inter Office Channel service in California and who have a commitment to MCI WorldCom Communication, Inc., services of one year or longer.

The dedicated access circuit must be associated with an MCI WorldCom Communications Inter-Office Channel circuit that has a minimum length of 100 miles to be eligible for this promotion.

In lieu of standard tariffed monthly recurring charges for Digital Data Local Access service of 56/64 Kbps, customers enrolled in this promotion will receive the following rates based on the term of service and the Local Access Channel Monthly Recurring charges referenced in MCI WorldCom Communications, Inc. FL PSC TARIFF NO.3, Section 4.2.2.1 (Tariff).

Monthly Recurring Per-Local Access Channel Charge Digital Data Service Access Circuits

	(\$1-300	(\$300-800
TERM	Tariff)	Tariff)
1 YEAR	\$150.00	\$200.00
2 YEAR	\$142.50	\$190.00
3 YEAR	\$138.00	\$184.00
4 YEAR	\$135.00	\$180.00
5 YEAR	\$127.50	\$170.00

To receive the benefits of this promotion Customers must place an order for installation of all circuits which will receive the benefits of this promotion by November 30, 2000. All circuits which receive the benefits of this promotion must be installed by February 28, 2001.

Customers enrolled in this promotion are not eligible for any other promotions or discounts on the Local Access Channel Charges affected by the promotional rates.

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

#### PROMOTIONAL OFFERINGS

# 4) TERRESTRIAL DIGITAL SERVICE (DS1) Millennium Tiered Intrastate Access Promotion

Beginning May 1, 2000 and ending November 30, 2000, MCI WorldCom will offer the following promotion to new and existing customers of Dedicated Lease Line Service in Florida who install a new Terrestrial Digital Service local access circuit that is associated with use of new intrastate MCI WorldCom Communication Inter Office Channel service in Florida and who have a commitment to MCI WorldCom Communication, Inc., services of one year or longer.

The dedicated access circuit must be associated with an MCI WorldCom Communications Inter-Office Channel circuit that has a minimum length of 100 miles to be eligible for this promotion.

In lieu of standard tariffed monthly recurring charges for Terrestrial Digital Local Access service, customers enrolled in this promotion will receive the following rates based on the term of service, and the Local Access Channel Monthly Recurring charges referenced in MCI WorldCom Communications, Inc. FL PSC TARIFF NO.3, Section 4.2.2.2 (Tariff).

# Monthly Recurring Per-Local Access Channel Charge Terrestrial Digital Service Access Circuits

(\$1-500 TERM Tariff)	(\$500-1000 Tariff)	(\$1000-1500 Tariff)	(\$1500-2000 Tariff)
1 YEAR \$237.50	\$332.50	\$475.00	\$712.50
2 YEAR \$225.00	\$315.00	\$450.00	\$675.00
3 YEAR \$200.00	\$280.00	\$400.00	\$600.00
4 YEAR \$195.00	\$273.00	\$390.00	\$585.00
5 YEAR \$190.00	\$266.00	\$380.00	\$570.00

To receive the benefits of this promotion Customers must place an order for installation of all circuits which will receive the benefits of this promotion by November 30, 2000. All circuits which receive the benefits of this promotion must be installed by February 28, 2001.

Customers enrolled in this promotion are not eligible for any other promotions or discounts on the Local Access Channel Charges affected by the promotional rates.

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u> 1/

# 4.2.1 <u>Analog Services</u>: <u>Access Functions</u>:

# (A) <u>Local Access Channel Terminations</u>

	1	Monthly	Monthly	
Mileage		Fixed	Per Mile	Install
AT Hub -0		\$63.80	\$0.00	\$550.00
1-8		\$92.30	\$1.65	\$550.00
9-25		\$92.30	\$1.60	\$550.00
26+		\$92.30	\$1.55	\$550.00

# (B) Access Coordination

As of December 1, 1998, the following charges apply, per local access channel, only for circuits installed prior to October 1, 1998 and only if either: (1) MCI orders the local access channel; or (2) in the case of customer-provided access, if the customers has not provided written notice to MCI that the customer accepts responsibility for the design, installation coordination and ongoing coordination of testing and trouble resolution for the local access channel.

<u>MONTHLY</u>	<u>INSTALLATION</u>	
\$19.40	\$174.00	

# (C) <u>Central Office Connection</u>

As of December 1, 1998, the following monthly recurring and non-recurring charges will apply, per connection, only for circuits installed prior to October 1, 1998, whether access is MCI provided or customer-provided.

<u>MONTHLY</u>	<u>INSTALLATION</u>	
\$19.45	\$215.00	

# (D) <u>Network Connection Charge</u>

As of December 1, 1998, the monthly recurring and non-recurring charges will apply, per connection, only for circuits installed on or after October 1, 1998, if access is customer-provided. Applicable rates are located in MCI F.C.C. Tariff No. 8, Section C.0125.

<sup>1/</sup> Beginning September 1, 2005, this service will no longer be available to new subscribers.

# 4.2 <u>DEDICATED LEASED LINE SERVICES (Cont.)</u>

4.2.1 Analog Services: (Cont.)

Access Functions:

**Inter-Office Channel Functions** 

(A) <u>Inter-Office Channel Mileage Charges (per channel)</u>: <u>1</u>/

<u>MILEAGE</u>	<u>FIXED</u>	MONTHLY PER MILE	INSTALLATION
0	\$100.00	N/A	
1-50	\$237.60	\$0.34	N/A
51-100	261.90	0.34	
101+	281.30	0.34	

ISSUED: November 13, 2008

EFFECTIVE: November 14, 2008

<sup>1/</sup> For the IOC on a point to point private line service circuit for which the two points terminate at the same MCI Point of Presence, the mileage is 0.

#### 4.2 **DEDICATED LEASED LINE SERVICES**

4.2.2 Digital Services:

#### 4.2 **DEDICATED LEASED LINE SERVICES**

#### 4.2.2 **Digital Services:**

#### 4.2.2.1 <u>Digital Data Service (DDS)</u>

ACCESS FUNCTIONS: (Cont.)

#### (B) Access Coordination:

As of December 1, 1998, the following charges apply, per local access channel, only for circuits installed prior to October 1, 1998 and only if either: (1) MCI orders the local access channel; or (2) in the case of customer-provided access, if the customers has not provided written notice to MCI that the customer accepts responsibility for the design, installation coordination and ongoing coordination of testing and trouble resolution for the local access channel.

SPEED	MONTHLY	INSTALLATION
2.4/4.8/9.6 kbps	\$28.86	\$276.00
56/64 kbps	\$28.86	\$287.00

#### (C) Central Office Connection (COC) (per connection)

As of December 1, 1998, the following monthly recurring and non-recurring charges will apply, per connection, only for circuits installed prior to October 1, 1998, whether access is MCI provided or customer-provided.

SPEED	MONTHLY	INSTALLATION
2.4/4.8/9.6 kbps	\$16.61	\$167.00
56/64 kbps	\$41.54	\$252.00

1/

Digital Data Service (DDS) is available only for circuits installed prior to December 1, 1998.

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

4.2.2 <u>Digital Services</u>: (Cont.)

# 4.2.2.1 <u>Digital Data Service (DDS) (Cont.)</u>

ACCESS FUNCTIONS: (Cont.)

(C) <u>DDS Multiplex (required, one per Central Office Connection)</u>

<u>SPEED</u>	<u>MONTHLY</u>	<u>INSTALLATION</u>
0.99.6 kbps	\$ 0.00	N/A
56/64 kbps	\$ 0.00	N/A

(D) Clear Channel 64 kbps DDS is available only with T1 local access that is equipped with B8ZS line coding from the Local Exchange Carrier. T1 access functions are referenced in Section 4.2.2.2.

# (E) <u>Network Connection Charge</u>

As of December 1, 1998, the monthly recurring and non-recurring charges will apply, per connection, only for circuits installed on or after October 1, 1998, if the access is customer-provided. Applicable rates are located in MCI F.C.C Tariff No. 8, Section C.0325.

ISSUED: November 13, 2008

# XIII. GRANDFATHERED SERVICES: SUBSECTION A

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

#### 4.2.2 <u>Digital Services (Cont.)</u>

## 4.2.2.1 <u>Digital Data Service (DDS) (Cont.)</u>

# Inter-Office Channel Mileage Charges 1/

## (A) <u>2.4/4.8/9.6 kbps</u>

		<b>MONTHLY</b>	
<b>MILEAGE</b>	<u>FIXED</u>	PER MILE	<u>INSTALLATION</u>
0	\$100.00	N/A	
1- 50	\$149.32	\$ 1.48	N/A
51-100	\$149.32	\$ 1.48	
101+	\$149.32	\$ 1.48	

# (B) <u>56/64 kbps</u>

		<u>MONTHLY</u>	
<b>MILEAGE</b>	<u>FIXED</u>	PER MILE	<u>INSTALLATION</u>
0	\$100.00	N/A	
1- 50	\$567.43	\$ 5.40	N/A
51-100	\$567.43	\$ 5.40	
101+	\$567.43	\$ 5.40	

# **Inter-Office Channel Options:**

(A) Route Diversity - Charges will be developed and filed on an individual case basis.

<sup>1/</sup> Network Pricing Plan terms are pursuant to MCI's F.C.C. Tariff No. 1. Specific discounts are listed in Section 4.2.2.5 herein. For the IOC on a point to point private line service circuit for which the two points terminate at the same MCI Point of Presence, the mileage is 0.

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

#### 4.2.2 Digital Services (Cont.)

#### 4.2.2.2 <u>Terrestrial Digital Service (TDS 1.5)</u>

#### Access Functions:

# (A) Local Access Channel Terminations

	Monthly	Monthly	
Mileage	Fixed	Per Mile	<u>Install</u>
At Hub – 0	\$281.80	\$0.00	\$1,490.00
1-8	\$341.55	\$24.25	\$1,490.00
9-25	\$341.55	\$22.35	\$1,490.00
26+	\$341.55	\$20.50	\$1,490.00

# (B) Access Coordination

As of December 1, 1998, the following charges will apply, per local access channel, only for circuits installed prior to October 1, 1998 and only if either: (1) MCI orders the local access channel; or (2) in the case of customer-provided access, if the customers has not provided written notice to MCI that the customer accepts responsibility for the design, installation coordination and ongoing coordination of testing and trouble resolution for the local access channel.

MONTHLY \$63.05 INSTALLATION \$215.00

# (C) Central Office Connection

As of December 1, 1998, the following monthly recurring and non-recurring charges will apply, per connection, only for circuits installed prior to October 1, 1998, whether access is MCI provided or customer-provided.

<u>MONTHLY</u>	<u>INSTALLATION</u>
\$156.20	\$340.00

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

### 4.2.2 <u>Digital Services (Cont.)</u>

### 4.2.2.2 Terrestrial Digital Service (TDS 1.5) (Cont.)

Access Functions: (Cont.)

(C) Office Options:

MONTHLY INSTALLATION

M24 MultiplexTo demultiplex a TDS
circuit into 24 voice
and/or data channels

M44 MultiplexTo demultiplex a TDS
circuit into two 1.544
mbps channels

# (D) Network Connection Charge

As of December 1, 1998, the monthly recurring charge and non-recurring charges will apply, per connection, only for circuits installed on or after October 1, 1998, if access is customer-provided. Applicable rates are located in MCI F.C.C. Tariff No. 8, Section C.0225.

#### 4.2 **DEDICATED LEASED LINE SERVICES**

#### 4.2.2 Digital Services (Cont.)

#### 4.2.2.2 <u>Terrestrial Digital Service (TDS 1.5)</u>

#### Inter-Office Channel Rates (per channel) 1/:

# Monthly Rate Plan

<u>Mileage</u>	Fixed Charge	Per Mile
0	\$200.00	N/A
1-50	\$970.00	\$7.76
51-100	\$970.00	\$7.76
101+	\$970.00	\$7.76

# Fixed Term Plans: 2/

Percent discount off Monthly Plan Inter-Office Channel Fixed and Per Mile Rates:

One Year	17%
Two Year	19%
Three Year	22%
Four Year	26%
Five Year	31%

# Inter-Office Channel Options:

Route Diversity - Charges will be developed and filed on an individual case (A) basis.

<sup>1/</sup> Network Pricing Plan terms are pursuant to MCl's F.C.C. Tariff No. 1. Specific discounts are listed in Section 4.2.2.5 herein. For the IOC on a point to point private line service circuit for which the two points terminate at the same MCI Point of Presence, the mileage is 0.

<sup>2/</sup> Fixed Term Plan are available only for circuits installed prior to December 1, 1998.

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

#### 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.3 <u>Digital Private Line Service</u>

#### Access Functions

Local Access for this service can be acquired through one of the following types of analog or digital access (DS0): Analog Local Access, T-1 Digital Access or DDS Local Access.

# (A) Access Coordination:

As of December 1, 1998, the following charges will apply, per local access channel, only for circuits installed prior to October 1, 1998 and only if either: (1) MCI orders the local access channel; or (2) in the case of customer-provided access, if the customers has not provided written notice to MCI that the customer accepts responsibility for the design, installation coordination and ongoing coordination of testing and trouble resolution for the local access channel.

Access Type	<u>MONTHLY</u>	INSTALLATION
Analog	\$19.40	\$174.00
DS0 Hubless (64 kbps & below)	\$28.86	\$174.00
T-1 Access (Channelized & Unchannelized)	\$63.05	\$215.00

#### 4.2 **DEDICATED LEASED LINE SERVICES**

#### 4.2.2 Digital Services (Cont.)

### 4.2.2.3 <u>Digital Private Line Service (Cont.)</u>

#### Access Functions (Cont.)

#### (B) Special Access Surcharge 1/

As of December 1, 1998, circuits installed prior to October 1, 1998, a \$29.63 Special Access Surcharge will be applied to each voice grade equivalent circuit end (per Local Channel for Analog Access) when the access line is provided by a Local Exchange Carrier. Customers who use their access line for both intrastate and interstate calling pay only one surcharge per access line a month. The surcharge will not apply to those customers who furnish MCI with an Exemption Certification.

(C) Digital Private Line Service at Fractional T-1 speed of multiple 56 kbps is available only with T-1 access from the Local Exchange Carrier. Service with Fractional T-1 speeds in multiples of 64 kbps is available only with T-1 access with B8ZS line coding.

#### (D) **Network Connection Charge**

As of December 1, 1998, the monthly recurring and non-recurring charges will apply, per connection, only for circuits installed on or after October 1, 1998, if access is customer-provided. Applicable rates are located in MCI F.C.C. Tariff No. 8, Section C.0125 (Analog), C.0425 (DS0 Hubless), and C.0225 (T-1).

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

Surcharge, imposed by the Local Exchange Carrier, will not apply to those customers who furnish MCI with 1/ an Exemption Certification (as defined herein).

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

# 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.3 <u>Digital Private Line Service (Cont.)</u>

# **Digital Private Line Central Office Functions**

# (A) <u>Central Office Connection (COC) per connection</u>

As of December 1, 1998, the following monthly recurring and non-recurring charges will apply, per connection, only for circuits installed prior to October 1, 1998, whether access is MCI provided or customer-provided.

Access Type	<u>MONTHLY</u>	INSTALLATION
Analog	\$19.45	\$215.00
DS0 Hubless (64kbps & below)	\$19.45	\$215.00
T-1 Access (Channelized & Unchannelized)	\$156.20	\$340.00

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

# 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.3 <u>Digital Private Line Service (Cont.)</u>

# Digital Private Line Central Office Functions (Cont.)

# (A) <u>Inter-Office Channel (IOC) 1/</u> Mileage Charges (per IOC)

<u></u>	<del> </del>	MONTHLY	
DATA SPEED	<b>MILEAGE</b>	<u>FIXED</u>	PER MILE
	0	\$100.00	N/A
64 kbps & below	1-50	\$261.90	\$ 0.34
(DS0)	51-100	261.90	0.34
	101+	281.30	0.34
	0	\$200.00	N/A
112 kbps or	1-50	\$528.65	\$ 0.61
Clear Channel 128 kbps	51-100	528.65	φ 0.61
Clear Charmer 120 kbps	101+	528.65	0.61
	1017	526.05	0.01
	0	\$200.00	N/A
168 kbps or	1-50	\$776.00	\$ 0.89
Clear Channel 192 kbps	51-100	\$776.00	\$ 0.89
·	101+	\$776.00	\$ 0.89
	0	<b>#200.00</b>	N1/A
00.4 leber a re	0	\$200.00	N/A
224 kbps or	1-50	\$1,003.95 \$4,003.05	\$ 1.15 © 4.45
Clear Channel 256 kbps	51-100	\$1,003.95	\$ 1.15
	101+	\$1,003.95	\$ 1.15
	0	\$200.00	N/A
280 kbps or	1-50	\$1,246.45	\$ 1.45
Clear Channel 320 kbps	51-100	\$1,246.45	\$ 1.45
	101+	\$1,246.45	\$ 1.45
	_		
00011	0	\$200.00	N/A
336 kbps or	1-50	\$1,459.85	\$ 1.70
Clear Channel 384 kbps	51-100	\$1,459.85	\$ 1.70
	101+	\$1,459.85	\$ 1.70

<sup>1/</sup> Network Pricing Plan terms are pursuant to MCI's F.C.C. Tariff No. 1. Specific discounts are listed in Section 4.2.2.5 herein. For the IOC on a point to point private line service circuit for which the two points terminate at the same MCI Point of Presence, the mileage is 0.

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

# 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.3 <u>Digital Private Line Service (Cont.)</u> <u>Digital Private Line Inter-Office Service (Cont.)</u> 1/

# (A) Inter-Office Channel (IOC) Mileage Charges (per IOC) (Cont.)

DATA SPEED  392 kbps or Clear Channel 448 kbps	MILEAGE 0 1-50 51-100 101+	MONTHLY FIXED \$200.00 \$1,658.70 \$1,658.70 \$1,658.70	PER MILE N/A \$ 1.94 \$ 1.94 \$ 1.94
448 kbps or Clear Channel 512 kbps	0 1-50 51-100 101+	\$200.00 \$1,838.15 \$1,838.15 \$1,838.15	N/A \$ 2.15 \$ 2.15 \$ 2.15
504 kbps or Clear Channel 576 kbps	0 1-50 51-100 101+	\$200.00 \$2,012.75 \$2,012.75 \$2,012.75	N/A \$ 2.36 \$ 2.36 \$ 2.36
560 kbps or Clear Channel 640 kbps	0 1-50 51-100 101+	\$200.00 \$2,167.95 \$2,167.95 \$2,167.95	N/A \$ 2.55 \$ 2.55 \$ 2.55
616 kbps or Clear Channel 704 kbps	0 1-50 51-100 101+	\$200.00 \$2,308.60 \$2,308.60 \$2,308.60	N/A \$ 2.72 \$ 2.72 \$ 2.72
672 kbps or Clear Channel 768 kbps	0 1-50 51-100 101+	\$200.00 \$2,434.70 \$2,434.70 \$2,434.70	N/A \$ 2.88 \$ 2.88 \$ 2.88

The Interoffice Channel (IOC) monthly fixed and per mile rates for Digital Private Line Service bandwidths above 768 kbps will be charged according to the TDS-1.5 rates listed herein.

<sup>1/</sup> Network Pricing Plan terms are pursuant to MCl's F.C.C. Tariff No. 1. Specific discounts are listed in Section 4.2.2.5. herein.

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

# 4.2.2 <u>Digital Services (Cont.)</u>

#### 4.2.2.4 Terrestrial Digital Service - 45 (DS3)

#### Access Functions:

# (A) Local Access Channel Terminations

	Monthly	Monthly	
Mileage	Fixed	Per Mile	Install
AT Hub -0	\$3,600.00	\$0.00	\$600.00
1-4	\$4,900.00	\$120.00	\$600.00
5-8	\$4,900.00	\$120.00	\$600.00
9-25	\$5,050.00	\$120.00	\$600.00
26+	\$5,300.00	\$120.00	\$600.00

#### (B) Access Coordination

As of December 1, 1998, the following charges will apply, per local access channel, only for circuits installed prior to October 1, 1998 and only if either: (1) MCI orders the local access channel; or (2) in the case of customer-provided access, if the customers has not provided written notice to MCI that the customer accepts responsibility for the design, installation coordination and ongoing coordination of testing and trouble resolution for the local access channel.

Monthly Installation \$ 145.50 \$ 548.00

# (C) Central Office Connection (COC) - (per connection)

As of December 1, 1998, the following monthly recurring and non-recurring charges will apply, per connection, only for circuits installed prior to October 1, 1998, whether access is MCI provided or customer-provided.

Monthly Installation \$ 485.00 \$ 548.00

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

# 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.4 Terrestrial Digital Service - 45 (TDS-45) (Cont.)

#### Access Functions:

# (C) <u>Inter-Office Channel (IOC) Rates - (per Inter-Office Channel)</u>

A TDS-45 Inter-Office Channel (IOC) connects two designated MCI terminals on MCI's digital fiber-optic network.

<u>Mileage</u>	Fixed Rate	Per Mile
1 - 50	\$ 12,125.00	\$ 33.95
51 - 100	\$ 12,125.00	\$ 33.95
101 +	\$ 12,125.00	\$ 33.95

# (D) <u>Network Connection Charge</u>

As of December 1, 1998, the monthly recurring and non-recurring charges will apply, per connection, only for circuits installed on or after October 1, 1998, if access is customer-provided. Applicable rates are located in MCI F.C.C. Tariff No. 8, Section C.0526.

- XIII. GRANDFATHERED SERVICES: SUBSECTION A
- 4.2 <u>DEDICATED LEASED LINE SERVICES</u>
  - 4.2.2 <u>Digital Services (Cont.)</u>
    - 4.2.2.5 <u>Incidental Non-Recurring Charges</u>

The Rates and Charges can be found in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="https://www.verizonbusiness.com">www.verizonbusiness.com</a> (Companion Interstate Service).

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

#### 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.6 Network Pricing Plans (NPP) 1/

Under these plans, a customer commits to a minimum monthly Inter-Office Channel revenue level over a selected term. NPP terms will be applied pursuant to MCI's F.C.C. Tariff No. 1, Section C-2.025.

# 4.2.2.6.1 <u>Analog Service NPPs</u>

Network Pricing Plans are available at minimum monthly Inter-Office Channel revenue levels of \$2,000, \$5,000, \$10,000, \$25,000, \$50,000, \$75,000, \$100,000, \$200,000, \$350,000, \$500,000, \$750,000, and \$1,000,000 over terms of 1, 2, 3, 4, or 5 years.

# All Speeds

Revenue					
Commitment	<u>1 yr</u>	2 yrs	<u>3 yrs</u>	<u>4 yrs</u>	<u>5 yrs</u>
\$ 2,000	2%	5%	6%	8%	10%
\$ 5,000	3%	6%	7%	9%	11%
\$ 10,000	4%	7%	8%	10%	12%
\$ 25,000	5%	8%	9%	11%	13%
\$ 50,000	6%	9%	10%	12%	14%
\$ 75,000	7%	10%	11%	13%	15%
\$ 100,000	8%	11%	12%	14%	16%
\$ 200,000	9%	12%	13%	15%	17%
\$ 350,000	10%	13%	14%	16%	18%
\$ 500,000	10%	13%	14%	16%	18%
\$ 750,000	10%	13%	14%	16%	18%
\$1,000,000	10%	13%	14%	16%	18%

<sup>1/</sup> Network Pricing Plans with minimum monthly IOC revenue levels of \$75,000 or greater are available only for circuits installed prior to December 1, 1998.

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

# 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.6 Network Pricing Plans (NPP) (Cont.)

# 4.2.2.6.2 <u>Digital Data Service NPPs</u>

Network Pricing Plans are available at minimum monthly Inter-Office Channel revenue levels of \$2,000, \$5,000, \$10,000, \$25,000, \$50,000, \$75,000, \$100,000, \$200,000, \$350,000, \$500,000, \$750,000, and \$1,000,000 over terms of 1, 2, 3, 4, or 5 years.

# All Speeds

Revenue Commitment	<u>1 yr</u>	<u>2 yrs</u>	<u>3 yrs</u>	4 yrs	<u>5 yrs</u>
\$ 2,000	3%	5%	8%	10%	12%
\$ 5,000	5%	7%	10%	12%	14%
\$ 10,000	7%	10%	13%	15%	17%
\$ 25,000	10%	14%	16%	18%	20%
\$ 50,000	13%	17%	19%	21%	22%
\$ 75,000	14%	18%	21%	22%	23%
\$ 100,000	15%	19%	22%	23%	24%
\$ 200,000	17%	22%	27%	28%	29%
\$ 350,000	18%	23%	28%	29%	30%
\$ 500,000	19%	24%	29%	30%	32%
\$ 750,000	20%	25%	30%	32%	35%
\$1,000,000	21%	26%	31%	35%	40%

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

#### 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.6 Network Pricing Plans (NPP) (Cont.)

# 4.2.2.6.3 <u>Terrestrial Digital Service (TDS 1.5)</u>

Network Pricing Plans are available at minimum monthly Inter-Office Channel revenue levels of \$2,000, \$5,000, \$10,000, \$25,000, \$50,000, \$75,000, \$100,000, \$200,000, \$350,000, \$500,000, \$750,000, and \$1,000,000 over terms of 1, 2, 3, 4, or 5 years.

<u>3 yrs                                   </u>	<u>2 yrs</u>	<u>1 yr</u>	Revenue Commitment
8% 10% 12%	7%	4%	\$ 2,000
9% 11% 13%	8%	5%	\$ 5,000
10% 12% 14%	9%	6%	\$ 10,000
11% 13% 15%	10%	7%	\$ 25,000
12% 14% 16%	11%	8%	\$ 50,000
13% 15% 17%	12%	9%	\$ 75,000
14% 16% 18%	13%	10%	\$ 100,000
15% 17% 19%	14%	11%	\$ 200,000
16% 18% 20%	15%	12%	\$ 350,000
16% 18% 20%	15%	12%	\$ 500,000
16% 18% 20%	15%	12%	\$ 750,000
16% 18% 20%	15%	12%	\$1,000,000
10%     12%     14       11%     13%     15       12%     14%     16       13%     15%     17       14%     16%     18       15%     17%     19       16%     18%     20       16%     18%     20       16%     18%     20       16%     18%     20       16%     18%     20	9% 10% 11% 12% 13% 14% 15% 15%	6% 7% 8% 9% 10% 11% 12% 12%	\$ 10,000 \$ 25,000 \$ 50,000 \$ 75,000 \$ 100,000 \$ 200,000 \$ 350,000 \$ 500,000 \$ 750,000

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

#### 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.6 Network Pricing Plans (NPP) (Cont.)

# 4.2.2.6.4 <u>Digital Private Line Service</u>

Network Pricing Plans are available at minimum monthly Inter-Office Channel revenue levels of \$2,000, \$5,000, \$10,000, \$25,000, \$50,000, \$75,000, \$100,000, \$200,000, \$350,000, \$500,000, \$750,000, and \$1,000,000 over terms of 1, 2, 3, 4, or 5 years.

2.4/4.8/9.6, 56/64 kbps Speeds

Revenue Commitment	<u>1 yr</u>	<u>2 yrs</u>	<u>3 yrs</u>	4 yrs	<u>5 yrs</u>
\$ 2,000	2%	5%	6%	8%	10%
\$ 5,000	3%	6%	7%	9%	11%
\$ 10,000	4%	7%	8%	10%	12%
\$ 25,000	5%	8%	9%	11%	13%
\$ 50,000	6%	9%	10%	12%	14%
\$ 75,000	7%	10%	11%	13%	15%
\$ 100,000	8%	11%	12%	14%	16%
\$ 200,000	9%	12%	13%	15%	17%
\$ 350,000	10%	13%	14%	16%	18%
\$ 500,000	10%	13%	14%	16%	18%
\$ 750,000	10%	13%	14%	16%	18%
\$1,000,000	10%	13%	14%	16%	18%

# 4.2 <u>DEDICATED LEASED LINE SERVICES</u>

# 4.2.2 <u>Digital Services (Cont.)</u>

# 4.2.2.6 Network Pricing Plans (NPP) (Cont.)

# 4.2.2.6.4 <u>Digital Private Line Service (Cont.)</u>

Fractional T-1 - 112 kbps and up to 768 kbps speeds

Revenue Commitment	<u>1 yr</u>	<u>2 yrs</u>	<u>3 yrs</u>	<u>4 yrs</u>	<u>5 yrs</u>
\$ 2,000	13%	16%	17%	18%	19%
\$ 5,000	16%	18%	19%	20%	21%
\$ 10,000	19%	20%	21%	22%	23%
\$ 25,000	21%	22%	23%	24%	25%
\$ 50,000	23%	24%	25%	26%	27%
\$ 75,000	25%	26%	27%	28%	29%
\$ 100,000	27%	28%	29%	30%	31%
\$ 200,000	28%	29%	30%	31%	32%
\$ 350,000	29%	30%	31%	32%	33%
\$ 500,000	30%	31%	32%	33%	34%
\$ 750,000	31%	32%	33%	34%	35%
\$1,000,000	32%	33%	34%	35%	36%

- XIII. GRANDFATHERED SERVICES: SUBSECTION A
- 4.3 EXPERIMENTAL SERVICES
- 4.3.1 MCI Coin Telephone Service

Collection of Charges at Coin Telephones

Charges for calls at coin telephones will be the total charge as provided in Section 4.1.1.3 (A and B) herein, and are computed and rounded to the nearest multiple of \$.05.

# 4.5 SPECIAL SERVICE ARRANGEMENTS

Upon appropriate circumstances, the Company may enter into Special Service Arrangements furnished in lieu of existing tariff offerings. All rates and charges for specific service(s) provided will be negotiated on an individual case basis. In such circumstances, the rates, terms and conditions applicable to such special contractual offerings will be specified below, with cross-reference to the particular contract involved.

#### 4.5.1 University of Florida

- A. This tariff offering is a Special Service Arrangement (SSA) for Vnet Service and University Calling Card Service being provided to the University of Florida in response to the Request for Proposal No. 93P93.
- B. The Vnet Service being provided under this Special Service Arrangement is described in Section 3.5.9 of this tariff.
- C. The University Calling Card Service is a customized Calling Card Service which is offered to the University of Florida pursuant to the terms and conditions contained in the MCI Response. The service is a one-way, dial-in-dial-out multipoint service. Subscribers can originate calls from any tone-dial phone via an "800" number.

#### Regulations

The General Conditions, Specifications and Special Requirements of the Request for Proposal No. 93P93 shall be considered as part of this Special Service Arrangement.

#### Rates and Charges

A. <u>Vnet Service</u>

<u>BUSINESS DAY</u> <u>EVENING</u> <u>NIGHT/WEEKEND</u>

**\$.1080 \$.0756 \$.0756** 

B. University Calling Card Service

<u>BUSINESS DAY</u> <u>EVENING</u> <u>NIGHT/WEEKEND</u>

\$.1583 \$.1413 \$.1413

# 5.1 SPECIAL SERVICE ARRANGEMENTS

Upon appropriate circumstances, the Company may enter into Special Service Arrangements furnished in lieu of existing tariff offerings. All rates and charges for specific service(s) provided will be negotiated on an individual case basis. In such circumstances, the rates, terms and conditions applicable to such special contractual offerings will be specified below, with cross-reference to the particular contract involved.

# 5.5.1 University of Florida

- A. This tariff offering is a Special Service Arrangement (SSA) for Vnet Service and University Calling Card Service being provided to the University of Florida in response to the Request for Proposal No. 93P93.
- B. The Vnet Service being provided under this Special Service Arrangement is described in Section 3.5.9 of this tariff.
- C. The University Calling Card Service is a customized Calling Card Service which is offered to the University of Florida pursuant to the terms and conditions contained in the MCI Response. The service is a one-way, dial-in-dial-out multipoint service. Subscribers can originate calls from any tone-dial phone via an "800" number.

#### Regulations

The General Conditions, Specifications and Special Requirements of the Request for Proposal No. 93P93 shall be considered as part of this Special Service Arrangement.

# Rates and Charges

A. <u>Vnet Service</u>

BUSINESS DAY EVENING NIGHT/WEEKEND

\$.1080 \$.0756 \$.0756

B. University Calling Card Service

BUSINESS DAY EVENING NIGHT/WEEKEND

\$.1583 \$.1413 \$.1413

#### SECTION 5- OBSOLETE SERVICES

# 5.1 <u>SPECIAL SERVICE ARRANGEMENTS (CONT.)</u>

5.5.3 This tariff offering provides MCI Vision Dedicated Access for Florida Southern College under special agreement. The following non-distance sensitive rate will apply to all time periods.

Intrastate Usage Charge:

Outbound Dedicated Access \$0.08

<u>Term</u>: Five (5) Year Contract

#### 5.5.4 Shands Hospital at the University of Florida

This tariff offering provides Vnet Service, MCI 800 Service using Dedicated Line (800 DAL) or Common Business Line (800 CBL) termination, and Dedicated Leased Line Service under special agreement. The following non-distance sensitive rates will apply to all time of day periods.

# Usage Rates:

#### **Vnet Service**

Switched/Dedicated

<u>Dedicated/Switched</u> <u>Switched/Switched</u>

\$0.0840 \$0.0840

#### MCI 800 Service

 800 DAL Peak
 800 DAL Off-Peak
 800 CBL

 \$ 0.0950
 \$ 0.750
 \$ 0.1400

<u>Dedicated Leased Line Service</u> - For Terrestrial Digital Service (TDS 1.5) Interoffice Channel (IOC), the following rate applies for up to three TDS-1.5 circuits on a monthly basis:

\$ 1,150.00 per TDS 1.5

Local loops will be provided at tariffed rates. MCI shall waive installation charges incurred by Customer for installation of all MCI T-1's under this Agreement. In addition, MCI will waive monthly recurring access charges Access Coordination (AC) and Central Office Connection (COC) for up to four (4) T-1 Local Loops for the term of the Agreement. Such discount shall be exclusive of access/egress (or related) charges, any non-recurring charges imposed in the Tariff, taxes and surcharges, to be applied to Customers monthly T-1 charges. The combined charges for AC, COC, and local loop (prior to the APP discount) must be less than \$600.00 per T-1 in order for MCI to provide the T-1's to Customer free of charge.

<u>Term</u>: 36 months with option to extend for two additional twelve month terms upon thirty days written notice to MCI.

#### TABLE I

# METERED USE SERVICE AVAILABILITY

#### Option H (MCI 800 Service),

MCI 800 Service with Switched WATS termination, and MCI Prism Plus are available in all exchanges in the following cities which have been converted to equal access.

Bartow Blountstown Boca Raton Bradenton Bronson Brooksville

Chiefland

Clearwater

Cocoa

Daytona Beach Eau Gallie Ft. Lauderdale Ft. Pierce Gainesville

Mandarin

Melbourne Miami Milton

New Smyrna Beach

Niceville Orlando Pahokee

Palatka Panama City

> Pensacola Pomona Park

Port Richey Port Saint Lucie Riviera Beach

# TABLE I

# METERED USE SERVICE AVAILABILITY

# Option H (MCI 800 Service).

Haines City Heathrow Hudson Hutchinson Island Jacksonville Key Largo Lynn Haven St. Augustine Sabastian Sarasota Tallahassee Tampa Titusville Vero Beach Weekiwachee Spring West Palm Beach Zephyrhills

#### **TABLE II**

#### METERED USE SERVICE AVAILABILITY

#### OPTION D (MCI PRISM I), AND OPTION E (MCI PRISM II)

(A) MCI Prism I, and MCI Prism II are available for call origination via Analog Local Access and for call termination in the following cities. In addition, where facilities permit, T-1 Digital Access used in conjunction with Analog Dedicated Leased Line Service, MCI 800, Vnet, MCI Prism I and Vision is available from the following cities.

Auburndale

Avon Park

Boca Raton

Chipley

Clearwater

Cocoa

Daytona Beach

Deerfield Beach

Ft. Lauderdale

Ft. Myers

Ft. Pierce

Gainesville

Heathrow

Jacksonville

Lynn Haven

Mandarin

Melbourne

Miami

Niceville

Ocala

Orlando

Pensacola St. Augustine

St. Petersburg

Sarasota

Tallahassee

Tampa

West Palm Beach

# **TABLE II**

# METERED USE SERVICE AVAILABILITY

# OPTION D (MCI PRISM I) and OPTION E (MCI PRISM II)

(B) Option D, and E are available for call termination in the following cities.

Lakeland Panama City

#### **TABLE V**

#### PART A - DEDICATED LEASED LINE SERVICE AVAILABILITY FOR DIGITAL SERVICE

Digital Data and Terrestrial Digital Service are available between the following cities. T-1 Digital Access provided in conjunction with Analog Dedicated Leased Line Service, MCI 800 (Dedicated Termination), Vnet, MCI Prism I, MCI Prism II, and Vision Service is also available in these cities.

# MCI METROPOLITAN AREA TERMINAL CITY LOCATIONS (TERMINAL LOCATIONS)

Auburndale 1/2/

Avon Park 1/2/3/

**Boca Raton** 

Chipley 1/3/

Clearwater 1/2/

Cocoa

Daytona Beach 1/2/

Ft. Lauderdale <u>1</u>/ <u>2</u>/ <u>3</u>/

Fort Myers 1/3/

Ft. Pierce 1/2/

Gainesville 2/

Heathrow

Jacksonville 1/2/

Mandarin

Melbourne 4/

Miami 1/2/

Ocala 3/

Orlando 1/2/ Pensacola 1/2/

Riviera Beach

Sarasota

St. Augustine

St. Cloud 2/

St. Petersburg 1/2/

Tallahassee 1/

Tampa 1/2/

Vero Beach 3/

West Palm Beach 1/2/

- <u>1</u>/ This numbering denotes availability of the fiber-only routing option of TDS-1.5 Service.
- This numbering denotes availability of Clear Channel 64 kbps. DDS which is available only with T-1 local 2/ access that is equipped with B8ZS line coding from the Local Exchange Carrier.
- DDS only (TDS-1.5 not available) <u>3</u>/
- 4/ TDS-1.5 only (DDS not available)

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

#### TABLE V

#### PART B - DEDICATED LEASED LINE SERVICE AVAILABILITY FOR ANALOG SERVICE

Dedicated Leased Line Analog Service and Digital Private Line Service are available between the following locations within the State of Florida.

# MCI METROPOLITAN AREA TERMINAL CITY LOCATIONS (TERMINAL LOCATIONS)

Auburndale

Avon Park 1/2/

Blountstown

**Boca Raton** 

Chipley 2/

Clearwater 1/

Cocoa

Daytona Beach

Ft. Lauderdale 1/

Fort Myers

Ft. Pierce 2/

Gainesville 1/

Heathrow

Jacksonville 1/

Lynn Haven

Mandarin

Melbourne 1/2/

Miami 1/

Niceville

Ocala

Orlando 1/

Pensacola 1/

St. Augustine

St. Petersburg 1/

Sarasota

Tallahassee 1/

Tampa <u>1</u>/

West Palm Beach 1/

ISSUED: November 13, 2008

<sup>1/</sup> This numbering denotes the availability of Clear Channel Digital Private Line Service with speeds at increments of 64 kbps.

DDS 2.4, 4.8. 9.6, and 56 kbps. local access is not available in these locations for Digital Private Line Service.

# **TABLE VI**

# METERED USE SERVICE AVAILABILITY

# MCI PRISM I, PRISM II AND MCI 800 SERVICE - RANGES

The following chart denotes the Range designations for calls from each Option D, Option E, and Option H originating area code to each terminating area code with the state of Florida.

ORIGINATING AREA CODE	TERMINATING AREA COD			CODE
	<u>305</u>	<u>407</u>	<u>813</u>	<u>904</u>
305	1	1	1	2
407	1	1	1	1
813	1	3	1	1
904	2	1	1	1

Subsection B – Material previously in Company tariff No. 5.

# 3.9 Message Toll Service (MTS)

MTS service is a measured use, full time service and is offered on a monthly basis, utilizing intercity communications facilities shared among multiple users. The individual customer's basic monthly charges for the use of such intercity communications facilities are based upon the time of day, the total minutes the customer utilizes such facilities and the distance of each call. For each call under the MTS option, the minimum charge shall be the applicable charge for one minute of use with use in excess of one minute during a call charged at the applicable rate per minute with the fraction, if any, of the last minute of each call rounded up to the next highest whole minute, unless stated otherwise.

# 3.9.1 Message Toll Service (MTS) Options

#### A. Plan 75

Plan 75 is a two-way calling, multipoint service allowing the customer calling capabilities via customer-provided business lines. This is a flat-rated, time-of-day, non-mileage-sensitive service. Plan 75 is available in both equal and non-equal access areas. The product is recommended for small to medium sized customer.

#### B. Plan 76

Plan 76 is a two-way calling, flat-rated, non-time-of-day, non-mileage-sensitive, outbound WATS-like product available in both equal and non-equal access areas allowing the customer calling capabilities via customer-provided business lines. The product is recommended for small to medium sized customers. Discounts are available depending upon volume of usage.

# C. Plan 78

Plan 78 is a one-way, direct-dial, time-of-day-sensitive, non-mileage-sensitive service utilizing 1 + or dial-up access making use of shared access lines connecting the customer to the Company. Discounts are available depending upon volume of usage.

# 1. <u>In equal exchange access areas:</u>

In central offices where equal exchange access is available, customers who have established accounts with the Company may access MTS on Plan 78 on a direct dial basis or by dialing a local access number.

#### 2. In areas not having equal exchange access areas:

In central offices where equal access is not available, customers who have established accounts with the Company may use MTS on Plan 78 by dialing a local access number.

Subsection B – Material previously in Company tariff No. 5.

# 3.9 Message Toll Service (MTS) (Cont'd)

#### D. Plan 91

Plan 91 is a one-way, direct-dial, time-of-day-sensitive, non-mileage-sensitive service utilizing 1+ or dial-up access, making use of shared access lines connecting the customer to the Company. Plan 91 is designed only for equal access customers. The first minute of use, or a fraction thereof, is rated higher than each additional minute of use. Discounts are available depending upon volume or usage. Fractions of the last minute of each call are rounded up to the next highest one-half of a minute.

# E. Plan 17

Plan 17 is a time-of-day, non-mileage-sensitive, special rate plan developed solely for the employees of the Company. This plan was terminated on July 8, 1980, however, employees working for the Company before the plan was terminated are still allowed to use this rate plan until such time as they are no longer employed by the Company.

# 3.9 <u>Message Toll Service (MTS)</u> (Continued)

# 3.9.1 Message Toll Service (MTS) Options (Continued)

#### F. Dial USA

A one-way direct dial service utilizing 1+ or dial-up access, making use of common shared access lines connecting the customer with the Company facilities. In central offices where equal access is not available, customers may use MTS service by dialing a 7-digit access number.

# 1. <u>In Equal Exchange Access Areas</u>

In central offices where equal exchange access is available, customers who have established accounts with the Company may access MTS on a direct dial basis.

#### 2. In Areas Not Having Equal Exchange Access Areas

- (a) In central offices where equal access is not available, customers who have established accounts with the Company may use MTS Service by dialing a 7 digit access number.
- (b) Customers who use 7 digit access in connection with automatic dialing equipment furnished by the Company in non-converted exchanges may be exempted from this limitation on new or existing service in areas where the Company determines that the likelihood of fraudulent use of the service using such equipment is not significant.

# G. RingAmerica Service 1/

RingAmerica is a block-of-time calling plan designed for the residential customer. RingAmerica customers are charged a base monthly fee for up to sixty (60) minutes of evening and night/weekend calling.

# H. TropiCall

TropiCall is a direct dial, time-of-day sensitive, non-distance sensitive service offering, utilizing 1+ or dial-up access over Feature Group D, equal access facilities. This product is designed for residential customers. Customers are charged a flat rate per minute of usage.

# 3.9 <u>Message Toll Service (MTS)</u> (Continued)

# 3.9.1 Message Toll Service (MTS) Options (Continued)

#### I. Message Telephone Service

This service arrangement allows a customer to originate calls in areas with Equal Access capabilities serviced by the Company by presubscribing to the service. This service is also available to customers in areas without Equal Access capabilities through use of a non-travel authorization code and Company-provided switched access facilities.

The applicable usage rate depends upon the distance between originating and terminating points and the rate period(s) in which the call occurs. Discounts are given based upon the customer's usage volume.

# J. <u>Dial Up-WATS</u>

This service arrangement allows a customer to originate calls in areas with Equal Access capabilities served by the Company by pre-subscribing to the service, or by a non-travel authorization code which in limited instances is provided to customers who are not served by an Equal Access office. Calls can be terminated at any point in the state.

The applicable usage rate depends upon the distance between originating and terminating points and the rate period(s) in which the call occurs. Discounts are given based upon the customer's usage volume. A monthly service charge applies to each customer account.

\_

<sup>1/</sup> Beginning September 5, 2003, this service will no longer be available to new subscribers.

# 3.9 <u>Message Toll Service (MTS)</u> (Continued)

# 3.9.1 Message Toll Service (MTS) Options (Continued)

# J. <u>Dial Up-WATS</u> (Continued)

Dial Up-WATS rates are effective when the customer's billing information is entered into the billing system. Until such time MTS rates will apply. A customer who fails to pay any charges owed to the Company as prescribed in Section 2 will be terminated as a Dial Up-WATs customer. If the customer continues to access the network, they will be billed at MTS rates. Additionally, the customer will not be eligible for any volume discounts. Customers who have not established an account with the Company but utilize the network through casual access by dialing a 5-digit code will not be eligible for Dial Up-WATS rates.

# K. New Precision Service

This service arrangement allows a customer to originate calls in areas with equal access capabilities served by the Company by pre-subscribing to the service, or by use of a non-travel authorization code which in limited instances is provided to customers who are not served by an equal access office.

New Precision Service rates are effective when customer's account is established in the company's billing system. Applicable usage rates depend on the customer's monthly usage volume. Customers who do not have an established account or who's account has been disconnected pursuant to Section 2.8, but utilize the network will be billed at Custom Call 100 rates. Volume discounts will not apply.

# L. <u>Precision Plus</u>

This service arrangement allows a single location customer to originate long distance calls in areas served by the Company. Access to Precision Plus is obtained by presubscribing the customer's telephone to the Company, or by utilizing a non-travel authorization code which in limited instances is provided to customers who are not in equal access areas. Calls can be terminated at any point in Florida.

Precision Plus has no monthly fee or minimum usage requirement. Precision Plus customers may not subscribe to the Company's Advanced or Ultimate Call Manager billing programs, may not participate in a Fixed Period discount plan, and may not be an 800 or WATS service customer.

# O. RingAmerica Plus 1/

RingAmerica Plus is a flat-rated service offering available to existing RingAmerica customers. Customers are charged the per minute rate, for all time periods, located in Section 4.3.15 for all intrastate calls.

\_

 $<sup>\</sup>underline{1}$ / Beginning September 5, 2003, this service will no longer be available to new subscribers.

# 3.10 <u>Wide Area Telecommunications Service (WATS)</u>

WATS service is an outbound service requiring the customer to originate calls via a dedicated access facility between the customer's premises and the Company's POP and allowing the completion of calls via the Company's interexchange facilities and the facilities leased from other carriers. The dedicated access facility is provided by the Local Exchange Carrier. The Company will act as agent for the customer in ordering and installation of such facilities. For each call under WATS option the minimum charge shall be the applicable charge for one minute of use with use in excess of one minute during a call charged at the applicable rate per minute, with the fraction, if any, of the last minute of each call rounded up to the next highest tenth of one minute, unless stated otherwise. For plans that offer a volume discount, interstate, intrastate and international calls (except directory acceptance calls) will be aggregated to determine the Customer's volume level achieved in a monthly billing cycle.

# 3.10.1 WATS Service Options

#### A. Plan 5

Plan 5 is a one-way, non-time-of-day sensitive, outbound WATS service that utilizes dedicated access facilities provided by the Local Exchange Carrier. This product is designed for small to medium users. Volume discounts are available depending upon usage.

#### B. Plan 72

Plan 72 is a one-way, banded, non-time-of-day-sensitive, outbound WATS service that utilizes dedicated access facilities provided by the local exchange carrier. This product is designed for medium to large users. The minimum charge for a call is one-half minute.

# C. <u>Plan 90</u>

Plan 90 is a one-way, time-of-day-sensitive, outbound WATS service that utilizes access facilities provided by the Local Exchange Carrier. This product is designed for large users. Volume discounts are available depending upon usage. The minimum charge for a call is one-half minute.

#### D. Dedicated WATS

This service arrangement allows a customer to originate calls via dedicated access facilities connected to the Company locations and terminate calls via access facilities at any point within the state. Intrastate service is an add-on to the interstate Dedicated WATS Service.

The applicable usage rate depends upon the distance between originating and terminating points and the rate period(s) in which the call occurs. A minimum average time requirement (MATR) call duration of 30 seconds applies to a customer's monthly domestic calls. Discounts are given based upon the customer's usage volume.

# 3.10 Wide Area Telecommunications Service (WATS) (Cont.)

#### E. Precision Fit Service

This service arrangement is provided in the State of Florida in conjunction with the Company's interstate Precision Fit Service. With this complement, a customer may originate calls via dedicated access facilities. It is the responsibility of the customer to arrange for the local dedicated access interconnection between its premise and the Company's service points.

Applicable usage rates depend upon the LATA in which a call terminates, the rate period(s) in which the call occurs and upon the total minutes of use obtained within each rate period during the billing period.

A minimum usage requirement of \$5000 applies during each billing period. If a customer fails to reach the minimum usage level for a particular month, the difference between the customer's actual usage and the minimum requirement will be billed as a separate charge on that month's invoice.

#### 3.11 800 Service

Company 800 Service is an inbound service originating on feature group facilities provided by the Local Exchange Carrier (LEC) and terminating on a Regular Business Line or a Special Access Line (SAL). This service enables the customer to receive 800 service calls at their residence or place of business.

# 3.11.1 800 Service Options

#### A. Plan 50

Plan 50 is both an interstate and intrastate service that allows calls to be terminated on feature group facilities provided by the Local Exchange Carrier network. This product is designated for small to medium users. For each call under this plan, the minimum charge shall be the applicable charge for one minute of use with use in excess of one minute during a call charged at the applicable rate per minute, with the fraction, if any, of the last minute rounded up to the next highest tenth of one minute.

#### B. Plan 65

Plan 65 is both an interstate and intrastate service that allows calls to be terminated on a dedicated facility provided by the Local Exchange Carrier. This product is designed for large users. For each call under this plan, the minimum charge shall be the applicable charge for one minute of use with use in excess of one minute during a call charged at the applicable rate per minute, with the fraction, if any, of the last minute rounded up to the next highest one-half of one minute. Volume discounts may apply. Interstate, Intrastate and International calls will be aggregated to determine the Customer's volume level achieved in a monthly billing cycle.

#### C. Metromedia 800 Service

This service permits intrastate calls to be delivered to a customer's location in Florida from stations located throughout the state. The 800 Service customer is billed for the calls rather than the calling party. Different service options are available depending on the type of access connection between the customer and the Company. The switched termination option permits calls to be delivered to a customer's exchange access line(s) via the switched access facilities of the Company. The dedicated termination option permits calls to be delivered over dedicated access lines connecting Company locations to a customer's premises. This service may also be billed under the product name Performance 800.

The applicable usage rate depends upon the customer's service option, the distance between originating and terminating points and the rate period(s) in which the call occurs. Discounts are given based upon the customer's usage volume and the applicability of any optional billing plan discounts. A minimum usage requirement applies to dedicated termination arrangements utilizing analog access lines.

# 3.12 <u>Complementary Services</u>

#### 3.12.1 800 Travel Service

800 Travel Service is a measured use, non-distance-sensitive, non-time-of-day sensitive, outbound communications service whereby customers can access Company's switch via an 800 number, and upon entry of a valid authorization code and phone number complete a call from any touch tone phone. All customers of the Company will, upon request, be given an authorization code to use with the Company's 800 Travel Service.

# 3.12.2 OnLine Card Service

OnLine calling card is available to residential and commercial Company customers. Non-Company customers may also subscribe to OnLine as a stand-alone service. Customers may place domestic and international long distance calls using this service.

OnLine access can be from either a tone generating or rotary-dial telephone and is gained by dialing the Company's 800 access number. Calls originating from rotary phones will be completed with the assistance of operators. Per minute usage rates apply. A retroactive volume discount is applied to the total usage portion of the monthly statement from all time of day periods.

The time of day rate periods for OnLine Service will be Peak and Off-Peak. Peak rates apply to all calls that occur between 8AM and 5PM Monday through Friday, except on Company-recognized holidays. Off-Peak rates apply to all other calls.

# 3.12.3 Metromedia Calling Card

This service arrangement is offered in the state of Florida as a complement to the Company's interstate Metromedia Calling Card Service. It allows customers to originate call from any point within the state through use of a Metromedia Calling Card via the Company's 800 Universal Access Number. Calls may be terminated to any point within the state via shared use facilities.

Applicable usage rates depend upon the service options selected by the customer. Two options are available. Under Option A, calls are billed in one minute increments. Under Option B, calls are billed in six second increments and subject to a per call surcharge. The surcharge depends on whether the call originates from a LATA selected by the customer--Home Base or from a point outside of the customer's home base LATA, Travel.

All usage rates are time-of-day and distance sensitive. Usage volume discounts apply.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.12 <u>Complementary Services</u> (Cont.)

# 3.12.4 <u>Directory Assistance</u>

Directory Assistance is available to customers of any of the Company's services. A charge will apply to each Directory Assistance call. The charge applies to each call regardless of whether the Directory Assistance Bureau is able to furnish the requested telephone number. One request may be made on each Directory Assistance call.

Directory Assistance charges will not count towards any volume discounts. 1/ Directory Assistance calls from Longer Distance Niteline customers will be billed separately from accumulated Niteline usage.

#### 3.12.5 Operator Services

Operator Services will be offered to the Company's shared access service customers served from equal access offices, and to users accessing the company's services through public pay phones, customer-provided stations, correctional facilities, or other transient locations. The first minute of use, or fraction thereof, is rated higher than each additional minute of use. Fractions of the last minute of each call are rounded up to the next highest minute of use.

Directory Assistance charges will be included with other usage charges when calculating volume discounts and in satisfying minimum usage requirements for all Metromedia services except USA 1200, where such charges will not count toward the minimum usage requirement.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.12 <u>Complementary Services</u> (Continued)
  - 3.12.5 Operator Services (Continued)

Charges for Operator Services may be billed to a customer's commercial credit card account or local exchange company (LEC) calling card account, or to the calling station, called station or a designated third party station. Charges may not be billed to public pay phones or customer-provided stations, or to stations outside the United States.

# A. <u>Call types</u>:

- 1. <u>Customer Dialed Calling Card</u> Calls completed through an automated interface and billed to a telephone company calling card by the calling party, without the intervention of a live or automated operator.
- Operator Station Calls completed with the assistance of a company operator on a station-to-station basis. Charges may be billed to the customer's commercial credit card or LEC calling card account, or to the calling station, called station or a designated third party station. Collect calls will be billed if the called party accepts the call. For other call types, charges will be incorrect if the called station answers.
- 3. Person-to-Person Calls completed with the assistance of a company operator to a particular person, station, department or PBX extension specified by the calling party. Charges may be billed to the customer's commercial credit card or LEC calling card account, or to the calling station, called station or a designated third party station. Charges will be incurred only if the designated called party accepts the call.
- 4. <u>Operator Dialed Surcharge</u> This surcharge applies in addition to service charges described above to Operator Station and Person-to-Person calls in which the caller has the capacity to dial a number but rather has the operator dial instead.
- 5. <u>Pay Telephone Set Use Fee</u> This fee applies in addition to service charges described above on calls made by users accessing pay telephones.

The charges for calls completed with the assistance of an operator will be per minute usage charges billed in one minute increments, plus a per call surcharge dependent upon the type of call completed. Operator surcharges are not discounted. There will be no charges when the call cannot be completed or, when a Person-to-Person call is requested, the designated person, station, department or PBX extension cannot be reached.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.12 <u>Complementary Services</u> (Continued)
  - 3.12.6 LEC Billed Measured Service

LEC Billed Measured Service calling includes calls made by Customers without an established account with the Company. For purposes of this tariff, the Company considers a call to be "casual" if dialed:

- Using an accepted Company access code (e.g., 10XXX from a line not presubscribed to the Company; or
- From a line presubscribed to the Company (i.e., when the customer does not have an established account and billing relationship with WorldCom or another carrier using the WorldCom network).

LEC Billed Measured Service calls accepted by the Company will be billed at the Company's Dial USA per minute rates. In addition to per minute rates, a surcharge will apply to each call as provided in Section 4.6.4 of this tariff. Volume discounts will not apply to usage billed by the LEC.

# 3.12 <u>Complementary Services</u> (Continued)

# 3.12.7 <u>Credit Card Reader Phones</u>

Individuals may access Preferred Calling Service from specifically identified Credit Card Reader Phones located in the cities below. A non-discounted access surcharge will apply to each call that is billed to certain valid major credit cards, 1/ in addition to the usage rates listed in Section 4. An authorization code will not be required for individuals utilizing Credit Card Reader Phones. Usage billed to credit card accounts may not be used to satisfy any individual's minimum usage requirements related to other Company-provided services.

Credit Card Reader Phones are available in the following Metropolitan areas: Miami, Tampa

#### 3.13 Acclaim Family of Services

#### 3.13.1 Description of Acclaim Services

Acclaim is a family of service offerings that offers a unified service for single or multi-location customers. The Acclaim package includes the availability of dedicated and switched access termination with peak and off peak rates, volume discounts, accounting codes, call detail, and a consolidated invoice for all locations. Peak and off peak rates, volume discounts, and applicable recurring and non-recurring charges can be found in Section 4. All fractional per call charges will be rounded to the nearest whole cent.

Acclaim I is available as an outgoing switched product with origination via FGD, equal access lines. Acclaim II is available as an outgoing dedicated product via customer-provided dedicated access line (DAL) or T-1 access. Acclaim III is an 800 product via regular business lines. Acclaim IV is an 800 product via customer-provided dedicated access line (DAL) or T-1 service.

#### 3.13.2 Acclaim Extended Service Plan ("ESP")

Acclaim services are available to customers through an Extended Service Plan ("ESP") option if the customer agrees to commit to such service for a term of twelve (12) or twenty-four (24)  $\underline{1}$ / months and to generate a minimum monthly gross usage as outlined in Section 4.7.2.A. Customers who elect an Acclaim ESP option are subject to the following:

- A. Customers must indicate what Acclaim service or services are to be included in the ESP. The discounts provided under this option become effective with the first full month's usage.
- B. Customers must reach the minimum usage requirement associated with each Acclaim ESP option by the third invoice period and monthly thereafter.

ISSUED: November 13, 2008

EFFECTIVE: November 14, 2008

<sup>1/</sup> The twenty-four (24) month Extended Service Plan ("ESP") option is available only to customers who subscribe to Acclaim II and/or Acclaim IV services.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.13 Acclaim Family of Services (Continued)
  - 3.13.2 Acclaim Extended Service Plan ("ESP") (Continued)
    - C. Customers who do not generate the monthly minimum required in a given month will be billed the monthly minimum in lieu of the actual usage for that month. No discounts will be applied if usage does not reach required ESP option monthly minimum.
    - D. Customers who terminate service prior to the end of the term of commitment will be billed the minimum usage requirement times the number of full or partial months remaining in the term in one lump sum. This charge will not apply to Acclaim customers who convert from an Acclaim service to another Company service with equal or greater term and minimum requirement commitments.
    - E. All customer requests to commence or terminate Acclaim ESP must be made in writing to the Company and received no later than the last day of the month preceding the month in which the desired action is to take effect. The customer must provide written notification to cancel the ESP which must be received by the Company not less than thirty (30) days prior to the expiration of the term commitment. If such notification is not received by the Company within this timeframe, the Acclaim ESP agreement will be automatically renewed for a new term of commitment.

# 3.16 HotelNet Services

Products offered in this Section are marketed under the name of HotelNet Services. They are designed specifically for hotels, motels and other call aggregators whose traffic distribution is such that not more than 50% of their interstate minutes of use occur within the day rate period; that is, from 8 a.m. to, but not including, 5 p.m., Monday through Friday excluding holidays.

For purposes of determining rates, discounts and surcharges, all intrastate, interstate, international and directory assistance calls will contribute toward determining maximum levels under schedules described in this section. However, usage or revenues obtained from one service will not contribute toward obtaining qualifying usage or revenue for any other service described herein or elsewhere in this tariff. For purposes of billing, each HotelNet service is treated as a stand-alone product.

### 3.16.1 HotelNet WATS

#### A. Description

HotelNet WATS provides customers the option to originate calls by either switched or dedicated access arrangements. Customers subscribing to the HotelNet shared access option will be billed in accordance with the rates, terms and conditions applicable to the Company's Dial-Up WATS service as described in Section 3.9.1.J proceeding. Customers subscribing to the HotelNet WATS dedicated option will be billed in accordance with the rates, terms and conditions applicable to the Company's Dedicated WATS service as described in Section 3.10.1.D preceding.

# B. <u>Ultimate Call Manager Discount</u> (UCMD)

In addition to the usage volume discounts associated with each service option, an Ultimate Call Manager Discount (UCMD) of 5% will be applied to the customer's total usage after standard volume discounts are applied. However, if the customer's business day interstate minutes of use exceeds 50% of total interstate usage, a UCMD of zero percent will be applied for the month. The UCMD will appear on the customer's invoice as a single line item.

### 3.16.2 HotelNet Precision

### A. <u>Description</u>

HotelNet Precision provides customers the option to originate calls by either switched or dedicated access arrangements as described below and to terminate calls via shared use facilities to any point within the state.

Applicable usage rates and charges depend upon the access option selected by the customer. Rates for both options depend upon an account's total minutes of use per billing period. The rate associated with the highest total minutes of use obtained during a billing period will be the rate used for the billing of all calls for the period. However, if the customer's business day interstate minutes of use exceeds 50% of total interstate usage, a per minute surcharge of \$0.015 will be added to total billed minutes of interstate usage for the period.

Calls originated via dedicated access arrangements are subject to a minimum average time requirement (MATR) of 30 seconds per call per month. Dedicated access arrangements are also subject to order, access and installation charges as described in the Company's FCC Tariff.

## 3.17 ExactCall

# 3.17.1 Description

This service is provided in conjunction with the Company's interstate service. It allows the customer to select outward, inward and/or travel service options as described below. Volume discounts are applied to each service option based on total ExactCall usage for a billing period.

# 3.17.2 Service Arrangements

- A. <u>ExactCall PLUS</u> This service option is an outward calling switched access arrangement available to customers who presubscribe to the Company's service. Usage rates are time-of-day sensitive. Calls are billed in six second increments. A minimum average time requirement (MATR) of thirty seconds per call applies during a billing period.
- B. <u>ExactCall WATS</u> This service option is an outward calling dedicated access arrangement. ExactCall WATS arrangements are subject to installation and other service charges. Usage rates are time-of-day sensitive. Calls are billed in six second increments. A minimum average time requirement (MATR) of thirty seconds per call applies during a billing period.
- C. <u>ExactCall 800</u> This service option is an inward calling termination arrangement available via switched and dedicated access facilities. Usage rates are time-of-day sensitive. Calls are billed in six second increments. A minimum average time requirement (MATR) of thirty seconds per call applies during a billing period. 800 Service number charges apply as described in Section 4.5.4.F. 800 WATS Service (dedicated) arrangements are subject to installation and other service charges as described in Section 4.5.4.
- D. <u>ExactCall Card</u> This service option is an outward calling shared access arrangement that allows a customer to originate calls through use of a Company calling card via an 800 Universal Access Number. Usage rates are time-of-day sensitive. Calls are billed in six second increments subject to a one minute per call minimum. Metromedia Calling Card optional features as described in Section 3.12.3 are also available to ExactCall customers.
- E. <u>ExactCall MTS</u> This service option is an outward calling shared access arrangement available to customers who presubscribe to the Company service. Usage rates are time-of-day sensitive. Calls are billed in one minute increments. This service is also marketed under the names "Take-20" and HomePlus".

# 3.19 Performance 4000 Services 1/

## 3.19.1 Description

Performance 4000 Service offers a unified service for single or multi-location customers using both switched and dedicated inbound and outbound service, a calling card and international service. Performance 4000 has been designed especially for the larger customer billing over \$7,500 a month, with the availability of peak and off-peak rates, calling card services and offering a discount for customers willing to sign a term commitment of either twelve (12), twenty-four (24) or thirty-six (36) months and/or a willingness to commit to a specific dollar volume of monthly minimum usage. There are five (5) levels of Performance 4000 which are described as follows:

- Level I Requires a monthly minimum billing commitment of \$7,500; if the minimum is not reached the customer is charged the difference.
- Level II Requires a monthly minimum billing commitment of \$15,000; if the minimum is not reached the customer is charged the difference.
- Level III Requires a monthly minimum billing commitment of \$25,000; if the minimum is not reached the customer is charged the difference.
- Level IV Requires a monthly minimum billing commitment of \$50,000; if the minimum is not reached the customer is charged the difference.
- Level V Requires a monthly minimum billing commitment of \$75,000; if the minimum is not reached the customer is charged the difference.

Multiple services and/or multiple locations using Performance 4000's services can contribute to the overall monthly minimum commitment. Domestic 800, Domestic and International outbound calls contribute to determining monthly minimum usage. Operator Services usage is not included in determining monthly minimum usage.

Customers must reach the minimum monthly usage requirement associated with their selected Performance 4000 Level by the fourth invoice period and monthly thereafter.

Peak and off peak rates, rate periods, billing increments and applicable recurring and non-recurring charges can be found in Section 4 - Rates and Charges.

ISSUED: November 13, 2008

<sup>1/</sup> Effective June 1, 1995, Performance 4000 Service will no longer be offered to new customers.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.19 Performance 4000 Services (Cont.)
  - 3.19.2 <u>Service Arrangements</u>
    - A. <u>Performance 4000 Term Plan</u> Performance 4000 Services are available to customers through a Term Plan option if the customer agrees to commit to such service for a term of twelve (12), twenty-four (24) or thirty-six (36) months. Customers who elect the Term Plan Option will receive a discount off their selected option's domestic rates. Customers who elect the Performance 4000 Term Plan option are subject to the following:
      - 1. Customers must indicate what Performance 4000 Level is to be included in the term plan. The rates provided under this option become effective with the first full month's usage.
      - Customers who fall below the monthly minimum usage requirement for the plan selected will be billed the difference between actual usage and the monthly minimum required.
      - 3. Customers who terminate service prior to the end of the term of commitment will be billed a termination liability consisting of the monthly minimum of the selected plan option multiplied by the number of months remaining in the commitment, due and payable upon termination in a lump sum. This charge will not apply to customers who convert from a Performance 4000 service to another Company service with equal or greater term and minimum usage requirement commitments.

# 3.19 <u>Performance 4000 Services</u> (Continued)

# 3.19.2 <u>Service Arrangements</u> (Continued)

# A. <u>Performance 4000 Term Plan</u> (Continued)

- 4. All customer requests to commence or terminate a Performance 4000 Term Plan must be made in writing to the Company and received no later than thirty (30) days prior to the then existing term expiration date. If such notification is not received by the Company within this timeframe, the Performance Term Plan agreement will be automatically renewed for a new term of commitment. The customer has ninety (90) days from the date service is made available in which to notify the Company in writing of the customer's desire to cancel this agreement without further obligation.
- 5. Performance 4000 Term Plan is not available with Operator Services, Ultimate Call Manager, other Fixed Period Discount Plans or any promotions not associated with this plan.

## 3.22 InteleNet Service

# 3.22.1 <u>Description</u>

InteleNet offers a full service voice product for single or multi-location customers using switched or dedicated, and calling card origination and switched or dedicated toll-free termination. The InteleNet package includes a single flat rate for both peak and off-peak dedicated and switched usage and offers discounts for customers willing to sign a term commitment of month to month, twelve (12) or twenty-four (24) months and/or a willingness to commit to a specific dollar volume of monthly minimum usage.

The minimum monthly commitment will be based on net charges for service (after discounts have been applied) and must be reached by the fourth invoice period and monthly thereafter. Multiple services and/or multiple locations using InteleNet can contribute to the overall monthly commitment; however, the customer must allocate the minimum by service and location. All inbound and outbound domestic and international, calling card, intrastate service and internet access charges, including local services (if applicable) contribute to determining the monthly minimum usage with the exception of enhanced fax, conference calling and non-recurring charges.

If a customer's monthly service usage charges for any month in the term are less than the monthly commitment agreed upon, the customer shall pay the company the difference between the monthly usage charges and the monthly commitment in addition to charges for service.

# 3.22.2 <u>Service Arrangements</u>

When the customer has completed the term commitment, service will automatically be renewed for additional term(s) of the agreement unless the Company has received the customer's written notice to terminate the service on not less than sixty (60) days prior to the then existing term expiration date. Customers who terminate service prior to the end of the term in any manner other than stated will be liable for a service termination charge of an amount equal to the greater of the following, unless customer converts to another Company service with equal or greater term and minimum usage commitment:

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.22 <u>InteleNet Service</u> (Continued)
  - 3.22.2 <u>Service Arrangements</u> (Continued)
    - A. If the termination becomes effective prior to the completion of the first year of the Customer Term, then the charge shall be an amount equal to the monthly billing commitment times the number of months remaining in Term through the expiration of the first year. If applicable, twenty-five percent (25%) of the balance remaining (monthly billing commitment times the number of remaining months in the Term beyond the first year) will also be included.
    - B. If the termination becomes effective after the completion of the first year, then the charge shall be an amount equal to twenty-five percent (25%) of the balance remaining (monthly billing commitment times the number of months remaining in the Term).

In addition, the customer will be required to repay any promotional credits that were applied, and any charges reimbursed or waived.

Upon execution of the InteleNet Service Agreement, the customer has ninety (90) days from the date service is made available in which to notify the Company in writing, either by certified or registered mail (return receipt requested), of customer's desire to cancel the Service Agreement without penalty or further obligation, except for charges incurred up to the date of termination, in the event that quality deficiencies solely caused by the Company in the provision of telecommunications service hereunder are demonstrated by customer to affect adversely and materially customer's telecommunications applications (such a termination under this clause constituting a Termination for Cause). A Termination for Cause shall not be effective unless customer has reported trouble on an ANI or circuit-specific basis to (and received corresponding trouble ticket from) the appropriate Company Support Center and a period of not less than thirty (30) days after receipt of customer's written notice of termination has elapsed during which the Company fails to correct such quality deficiencies. Provided, nothing contained herein shall impose any liability on Company and customer's sole remedy shall be termination of the affected service as described.

Customer's InteleNet Service Agreement is subject to the general terms, and conditions and rates of this Tariff and/or Service or Credit Application forms executed in connection with the services provided herein.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.22 <u>InteleNet Service</u> (Continued)

#### 3.22.3 InteleNet Association

The InteleNet Association program is a benefit package offered in conjunction with InteleNet, which allows the individual users who are members or employees of the participating organization to receive additional product discounts off of domestic interstate usage. Members who elect to participate in the InteleNet Association will receive InteleNet products and Calling card service.

To qualify for InteleNet association an organization generally must be a: (1) Trade association representing businesses or individuals in an industry, profession or business classification; (2) Business with franchises, agents, distributors, or multiple representatives; (3) Non profit organization; or, (4) Buying group not organized merely to buy the Company's long distance for resale. The qualifying organization agrees to meet the following criteria within six (6) months of undertaking to qualify and thereafter maintain, and enters into a written agreement with the Company for marketing of the Company's services. The group's members who have subscribed to the Company's services throughout the group must have aggregate billings, net of taxes; promotional credits and surcharges of at least \$5,000 per month.

The Company will render monthly statements to the individual members and the statement received will show all appropriate discounts. All sums due from members are the sole property of the Company, and the Company shall have sole right to collect, enforce collection and settle such sums. The InteleNet member group shall receive a monthly report from the Company listing members of the group who subscribe to the Company service under this program.

Unless otherwise specified in this tariff, member's usage of Company service under this plan cannot be used to qualify for any other benefits under this tariff or under other arrangements between the Company and third parties who undertake to market the Company's services. All other conditions of InteleNet will apply to InteleNet Association.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.23 <u>Total Solution Gold<sub>sm</sub> Service</u>

# 3.23.3 <u>Total Solution Gold Term<sub>sm</sub> Plans</u>

Total Solution Gold<sub>sm</sub> Service is available at 1 or 2 year Term Plan rates. Term Plans are subject to a minimum monthly usage level of \$500 for shared access or \$2,500 for dedicated access after 90 days of initial service. Should the Customer's actual usage fall below the monthly minimum usage commitment, the Customer is required to pay the Company a fee equal to the difference between the Customer's actual usage and the Customer's minimum monthly commitment. Term Plan threshold calculations will include usage charges from all intrastate and interstate switched outbound and inbound usage products (directory assistance and operator assistance are not contributory).

Charges such as taxes, late payment fees and other service non-recurring and monthly recurring charges will not be included in the Term Plan usage threshold calculation. A Customer who terminates a Term Plan prior to the term's expiration will be required to pay in one lump sum an amount equaling the monthly minimum service usage charge times the number of months remaining in the term.

A Customer may terminate their Term Plan without liability under the following conditions:

- A. The customer requests, and remains on, a new Term Plan for a term equal to or greater than the time remaining on their current Company Term Plan.
- B. The Customer provides written notification to cancel the Term Plan to the Company postmarked within 90 days of the commencement of the Term Plan. The Customer is responsible for payment of services used during this period.

Term Plans will be automatically renewed unless the Customer notifies the Company in writing to cancel the renewal within 30 days of expiration of the current Term Plan.

# 3.24 Total Solution Series 100<sub>sm</sub> Service

# 3.24.1 Description

Total Solution Series 100sm Service is an offering that allows Customers to select a commitment level and shared and dedicated outbound and inbound service arrangements, obtain term plan discounts, and receive Peak and Off-Peak pricing (Off-Peak pricing only applies to intraLATA outbound service where available). Total Solution Series 100sm calls are flat rated, with discounts off the base rates available according to the commitment level and term plan selected by the Customer.

# 3.24.2 <u>Service Arrangement</u>

The Outbound Calling Service arrangement is a long distance switch-based service, which is available on either a shared or dedicated access basis. The usage is billed on a flat rated basis. Outbound Calling Service calls are billed in 6 second increments, with an initial billing period of 18 seconds per call.

The Inbound Calling Service arrangement is a toll free switched access service that permits calls to be completed at the Customer's location without charge to the calling party. This service is available on either a shared or dedicated basis. Access to the service is gained by an end user dialing a 10 digit telephone number (e.g., 800+NXX-XXXX) which will terminate at the Customer's location. The usage is billed on a flat rated basis. Toll Free Service calls are billed in 6 second increments, with an initial billing period of 18 seconds per call. In addition, there is a non-recurring and monthly recurring Toll Free Service number charge, which will be applied either on a "per Toll Free Service number" basis for shared access or on a "per routing arrangement" basis for dedicated access.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.24 Total Solution Series 100<sub>sm</sub> Service (Cont.)

## 3.24.3 Commitment Levels and Term Plans

Total Solution Series 100sm Service is available on a month to month basis or on an optional 1 or 2 year Term Plan. Discounts off the base rates are available to the Customer according to the commitment level and Term Plan selected by the Customer.

Total Solution Series  $100_{\rm sm}$  Service is subject to a minimum monthly usage commitment (after the third full month following the initial installation of service for new Customers) based on the Customer's selected commitment level. The commitment level is calculated from the monthly total of usage generated from the following Company products: shared and dedicated domestic interstate and intrastate outbound and inbound service, international outbound, intraLATA, local service, and calling card. Charges such as taxes, late payment fees or other service non-recurring and monthly recurring charges will not be included in the total usage amount calculation, as well as usage charges from Directory Assistance, Operator Service, voice mail and conference calling.

Should the Customer's actual usage fall below the monthly minimum usage commitment, the Customer is required to pay the Company a fee equal to the difference between the Customer's actual usage and the Customer's minimum monthly commitment. A Customer who terminates a Term Plan prior to the term's expiration will be required to pay in one lump sum an amount equaling the selected monthly commitment level times the number of months remaining in the term, plus an amount equal to any promotional credit, or discount, or waiver, if applicable, that was provided to the Customer.

A Customer may terminate their Term Plan without liability under the following conditions:

- A. The Customer requests, and remains on, a new Term Plan for a term and commitment level equal to or greater than their current Company Term Plan and commitment level.
- B. A new Customer (who was not receiving services through the Company prior to execution of the Term Plan) provides written notification to cancel the Term Plan to the Company postmarked within 90 days of the commencement of the Term Plan. The Customer is responsible for payment of services used during this period.

Term Plans will be automatically renewed unless the Customer notifies the Company in writing to cancel the renewal within 30 days of expiration of the current Term Plan.

# 3.25 MFS Inteleplan

### 3.25.1 Description

MFS Inteleplan is a communications service that is available for use by residential customers twenty-four (24) hours a day. MFS Inteleplan is offered in exchanges whereby the Customer's local telephone lines are presubscribed by the local exchange company (LEC) to the Company's MFS Inteleplan, such that "1+ the 10-digit number" calls are automatically routed to the Company's network. Customers may originate MFS Inteleplan from locations served by the Company, and may terminate in all locations within the state of Florida.

# 3.25.2 Service Arrangement

MFS Inteleplan is a flat-rated calling plan that is non-distance sensitive for direct-dialed long distance calling. Charges are based on the time period (Peak/Off-peak) when the call is placed. Peak and Off-peak time periods are defined in Section 4.19 of this tariff. Calls are billed in one-minute increments, with an initial billing period of one minute. A monthly recurring surcharge will apply to Customer accounts with a usage (including Directory Assistance) of less than \$25.00. Customer accounts with usage in excess of \$500.00 per month will be assessed a surcharge. The entire month's charges will be recalculated at the new rate level.

Directory Assistance and Operator Services are available to MFS Inteleplan subscribers subject to the provisions of Sections 3.12.4 and 3.12.5 of this tariff.

## 3.26 MFS Intelenet Calling Card Service 1/

## 3.26.1 Description

MFS Intelenet Calling Card Service is provided to Customers for use when away from their established service location. New customers of the Company's interexchange services will automatically be issued two calling cards. These cards will initially be subject to a \$25.00 per month usage limit. Upon notification to the Company by the Customer, the monthly usage limit may be increased. Access to the service is gained by dialing Company-designated toll free access number (eg. 800-NXX-XXXX), plus the Customer's/User's MFS Intelenet Calling Card authorization number and the called telephone number. The service includes a Customer-programmable speed dialing capability for up to eight telephone numbers, and a capability whereby a Customer may establish 2-to-6 digit, verified or unverified accounting codes for use with the MFS Intelenet Calling Card, at no additional charge. The MFS Intelenet Calling Card can also be used to place an operator-assisted and directory assistance calls, subject to the application of additional charges.

Beyond these standard features, the MFS Intelenet Calling Card includes the following enhanced features: conference calling, MFS InteleNet Voice Mail access, voice massaging, and news and information access. Use of these enhanced features is subject to separate charges. MFS Intelenet Calling Card calls are billed in six second increments, with a one minute minimum. This service is offered with Peak and Off-Peak pricing. A description of the additional features are listed below.

### A. Operator-Assisted Calls

The MFS Intelenet Calling Card can be used to place MFS Intelenet operator-assisted calls. The surcharges are applied per call, in addition to the standard usage charges.

1/ MFS Intelenet Calling Card Service will no longer be offered to new customers as of January 1, 1998

# 3.26 MFS Intelenet Calling Card Service (Continued)

# 3.26.1 <u>Description</u> (Continued)

#### B. Directory Assistance Calls

The MFS Intelenet Calling card can be used to place calls for Directory Assistance. A flat charge will apply per requested number (Requested Number Charge). At the Customer's option, the Company will automatically place a call to the requested number. For calls completed in this manner, a Call Completion Charge and the Standard Usage Charges will apply in addition to the Requested Number Charge.

# C. Enhanced Feature Charges

Enhanced features are available for use as described below. Enhanced features charges apply in lieu of standard usage charges. Usage charges are billed in six second increments with a one minute minimum.

- Conference Calling Allows a User to establish a conference call by accessing the conference operator. Charges apply per established line per minute of usage.
- 2. <u>Voice Mail Access</u> Allows a User to access MFS Intelenet Voice Mail and to place return calls without having to hang-up and initiate a new calling card call.
- 3. <u>Voice Messaging</u> Allows the User to leave up to a three minute voice recorded message that is stored for future delivery when the called number is busy or if there is no answer.
- 4. <u>News and Information</u> Provides access to news, weather, sports, financial information and other features.

### 3.30.2 Association Saver/Alumni Saver Program

The Company will render monthly statements to the individual members and the statement received will show all appropriate discounts. All sums due from members are the sole property of the Company, and the Company shall have the sole right to collect, enforce collection and settle such sums. The Association Saver or Alumni Saver member group shall receive a monthly report from the Company listing members of the group who subscribe to the Company's service under this program.

Unless otherwise specified in this tariff, members' usage of Company service under this plan cannot be used to qualify for any other benefits under this tariff or under other arrangements between the Company and third parties who undertake to market the Company's services.

# 3.30.3 FundRaiser

New customers who elect to participate in the FundRaiser program will receive Acclaim! products or RingAmerica block-of-time calling plan; and OnLine Calling Card service.

To qualify for FundRaiser an organization generally must be a: (1) Trade association representing businesses or individuals in an industry, profession or business classification; (2) Business with franchises, agents, distributors, or multiple representatives; (3) Non-profit organization or; (4) Buying group not organized merely to buy the Company's long distance for resale; and agrees to meet the following set of criteria within six (6) months of undertaking to qualify and thereafter maintain them, and enters into a written agreement with the Company for the marketing of the Company's services. The group's members who have subscribed to the Company's services through the group must have aggregate billings, net of taxes, promotional credits and surcharges of at least \$5,000 per month.

If the sponsor electing the FundRaiser plan meets the above criteria, the sponsor will receive a monthly contribution to its organization based on its members usage (usage for each member is added together to determine the payment schedule). If the organization chooses to print its own FundRaiser brochures it will receive an additional contribution. The sponsor's members may receive additional product discounts with the Acclaim! family of products. Unless otherwise specified in this tariff or in the agreement, customers' usage of Company service under this plan cannot be used to qualify for any other benefits under this tariff or under other arrangements between the Company and third parties who undertake to market the Company's services.

# 3.30.5 MCI WorldCom On-Net Service - Voice 1/

#### Α. **Description**

MCI WorldCom On-Net is an outbound, customized telecommunications service that may include an inbound service option using Business Line/WATS or Dedicated Access Line terminations and/or a virtual private network. MCI WorldCom On-Net provides a service for single or multi-location companies using switched, dedicated and card origination, and switched termination. This service is suitable for calling between company locations.

The rules and regulations governing the provision of service for toll free (inbound) service are identical to those for Toll-Free Service as filed in this Tariff, except where otherwise specified.

At the Customer's request and where feasible, the Company may permit the Customer to obtain other forms of access to the Customer's inbound service terminating location, or the Company may procure other forms of access to such location. All applicable recurring and non-recurring charges for such service will be calculated on an individual case basis, in accordance with the charges assessed by the Local Exchange Carrier or other access provider. The installation and monthly charges for any interface equipment associated with such access that is provided by the Company shall be calculated on an individual case basis. If the Company procures access for the Customer from the Local Exchange Carrier, the Customer's use of such access shall be in conformity with the regulations and other terms and conditions under which the Local Exchange Carrier provides such access.

This tariff is being offered to Florida customers that originate and terminate calls within the State. Intrastate service is provided in conjunction with interstate service and is available only to customers subscribing to interstate MCI WorldCom On-Net service as provided in WorldCom's interstate Tariff FCC No. 1 and FCC No. 2.

#### B. Definitions applicable to MCI WorldCom On-Net Service

The following definitions will apply for all usage rates:

"Local Net Connection": A switched connection between a customer premises and a WorldCom Point of Presence which is provided by WorldCom or a WorldCom affiliate. "Dedicated": A non-switched connection between a customer premise and a WorldCom Point of Presence.

"Switched": A switched connection between a customer premise and a WorldCom POP that is not provided by WorldCom or a WorldCom affiliate.

<sup>1/</sup> Beginning January 1, 2004, this service will no longer be available to new subscribers.

# 3.30.5 MCI WorldCom On-Net Service - Voice 1/

# C. MCI WorldCom On-Net Term Plan

The MCI WorldCom On-Net Term Plan is a term plan, in lieu of all other tariffed term plans. Customers who subscribe to this service through the Term Plan are subject to the terms and conditions following:

- 1. Eligible Intrastate charges is the customer's total intrastate usage of the following, after the application of promotional and other discounts: intrastate inbound and outbound usage. Charges for the following are not included as eligible intrastate charges and will not receive Term plan Discounts: Directory Assistance usage and surcharges and Operator Assisted usage and surcharges.
- 2. Term Commitment and Renewal Options: A customer must commit to service for a term of service of either one, two, three, four, or five years. The term of service will commence no earlier than the fifteenth of the billing month in which the customer subscribes to the plan. A plan will automatically renew for an equivalent term and volume commitment upon expiration of its term unless the customer provides written notification to cancel the service Term Plan, which must be received by WorldCom no more than 30 days after expiration of the existing term. If the customer cancels the existing term plan within 30 days after expiration of the existing term, the customer will receive the discounts for which the customer qualifies during the 30-day period following the expiration of the existing term of service.

## 3. Early Termination Charges:

- (a) Cancellation or Discontinuance Without Liability: If: (i) the customer's use of WorldCom service under the service Term Plan equals or exceeds the Customer's equivalent annualized minimum volume commitment and (ii) at the time of termination the customer is enrolled in a new service Term Plan with a volume commitment which equals or exceeds the Customer's existing volume commitment, the customer may terminate service under the Term Plan without liability as follows: (i) the customer may terminate service at any time during the last three months of the term of service if the customer's new service Term Plan's term commitment is one year; or (ii) the customer may terminate service at any time during the last six months of the term of service if the customer's new service Term Plan's term commitment is equal to or greater than two years.
- (b) <u>Cancellation or Discontinuance With Liability:</u> Discontinuance of all services furnished under the Term Plan prior to the expiration of the committed term of service constitutes discontinuance of the plan and the customer will be billed and required to pay an early termination charge equal to the Underutilization Charge for the year of termination plus 50 percent of each annual volume commitment for each year remaining in the unfulfilled term of service.

# D. Term Commitment/Discounts

The discounts set forth in Section 4.23.2 of this tariff will apply to the eligible intrastate charges based on the customer's term commitment.

# 3.30 Other Service Offerings (Continued)

## 3.30.5 MCI WorldCom On-Net Service – Voice (Continued)

## E. LD and Local Online Calling Plan 1/

Eligibility: To be eligible for this plan, customers:

must subscribe to this plan via a Company-designated Internet site;

must designate a Company affiliate as its exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the LD and Local Online Calling Plan as described in the MCI WORLDCOM COMMUNICATIONS, INC. TARIFF F.C.C. NO. 6 ("Companion Interstate Service") and to certain exchange service as described in the appropriate Company-affiliate exchange service tariff ("Companion Local Service") concurrent with enrollment in this plan;

may not receive service under a Special Customer Arrangement;

#### Definitions:

For the purposes of this plan, the following definitions apply:

"Eligible Intrastate Service" is defined as MCI WorldCom On-Net Service-Voice Outbound Service (excluding Card) usage and Inbound Service usage that originates and terminates in one state.

"Eligible Interstate Service" is defined as MCI WorldCom On Net Services Option 1 Outbound Service (excluding Card) usage that originates in the U.S. Mainland and Hawaii and terminates in the U.S. Mainland, Alaska, Hawaii, Puerto Rico, the U.S. Virgin Islands, Guam, and CNMI and MCI WorldCom On Net Services Option 1 Inbound Service usage that originates in the U.S Mainland, Alaska, Hawaii, Puerto Rico, the U.S. Virgin Islands, Guam, and CNMI and terminates in the U.S. Mainland and Hawaii.

"Existing customers" are customers who, at the time of subscription to the plan, are receiving service under this tariff and the tariff containing the Companion Interstate Service.

"New customers" are customers who, at the time of subscription to the plan, are not receiving service under this tariff and the tariff containing the Companion Interstate Service.

<sup>1/</sup> Beginning January 1, 2003, this service will no longer be available to new subscribers.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.6 MCI WorldCom On-Net Service Voice (Continued)
    - E. <u>LD and Local Online Calling Plan</u> (Cont.)

Monthly Recurring Charges:

A monthly recurring charge will apply for each Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Local Service. Offerings A and B are available on a per-Local Line basis and Offerings C and D are on a per-Local Trunk basis. The Monthly recurring charges can be found in Section 4.23.5.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - E. LD and Local Online Calling Plan (Cont.)

#### Benefits:

- a. Commencement of Service: The following provisions will apply during the period following plan enrollment prior to installation of Companion Local Service. Customers will be charged \$0.05 per minute for Eligible Intrastate Service.
- b. Eligible Interstate Usage and Companion Intrastate Service Allotments: Upon installation of intrastate service and Companion Local Service, the customer will receive an allotment of minutes per monthly period, as follows, that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period, based on Offering:

Offerings Allotment (Minutes) per Local trunk or per Local line)

- A 500
- B 1,000
- C 1,000
- D 2.000

Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment. When a call under this offering begins prior to the completion of customer's monthly allotment and ends after completion of the allotment, the customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - E. LD and Local Online Calling Plan (Cont.)
  - b. (Cont.)

For customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be pro-rated for that billing cycle.

Companion Local Service Allotments: Customers who subscribe to Companion
 Local Service will also receive unlimited exchange service usage as described in
 the Companion Local Service tariffs.

## d. Additional Benefits:

The company will waive the monthly recurring service fee per service group charges set forth in Section 3.5.15 for access associated with service provided under this plan.

Eligible Interstate Usage and Eligible Intrastate Service calls will be billed in 60second increments.

<u>Termination of Service</u>: The following provisions will apply to customers who terminate this plan, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect only Companion Local Service, the Companion Interstate Service under F.C.C. No. 6 and LD and Local Online Calling Plan under this tariff will terminate. Customers will then be automatically re-subscribed to the service offering under this tariff and F.C.C No. 6 to which the customer was subscribed at the time of subscription to this plan under this tariff and the Companion Interstate Service.

For existing customers who disconnect from Companion Local Service and LD and Local Online Calling Plan under this tariff, the Companion Interstate Service and LD and Local Online Calling Plan under this tariff will terminate. Customers will then be automatically re-subscribed to the service offering under F.C.C. No. 6 to which the customer was subscribed at the time of subscription to this plan.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - E. LD and Local Online Calling Plan (Cont.)

For new customers who disconnect only Companion Local Service, the Companion Interstate Service under F.C.C. No. 6 and LD and plan service under this tariff will terminate. Customers will then be automatically subscribed to MCI WorldCom On-Net Service under this tariff and F.C.C No. 6.

For new customers who disconnect Companion Local Service and plan service under this tariff, the Companion Interstate Service under F.C.C. No. 6 and plan service under this tariff will terminate. Customers will then be automatically subscribed to MCI WorldCom On-Net Service under F.C.C No. 6 for interstate long distance and MCI WorldCom On-Net Service under this tariff for Intrastate IntraLATA service only.

#### Other Conditions:

Charges under this plan will not be calculated in satisfaction of any usage volume requirement.

The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.

Customers subscribing to this plan may not receive the benefits of any discounts or promotions including any term plan discounts.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - F. On-Net Plus Plan

Eligibility: To be eligible for this program, customers:

- must designate a company affiliate as its exchange service carrier and the company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intralata toll calling;
- must receive interstate service under Special Customer Arrangement (SCA)
  Guide Type 1, 2, 3, 4, 5, 6, 7, 8, or 9 as described in the
  WorldCom's "Service Publication and Price Guide" located on the Company's
  website at www.worldcom.com;
- must be new business customers or existing business customers who is eligible for renewal of their contracts.

Usage Charges can be found in Section 4.23.6.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - G. Agent Program

Eligibility: To be eligible for this program, the Customer:

must subscribe to this program through a Company-designated agent;

may not receive any discounts or the benefits of any promotional offering;

may not receive service under any other term plan;

must subscribe to the WorldCom On-Net Voice Services Agent Program as described in WorldCom's "Service Publication and Pricing Guide" and,

must subscribe to exchange service provided by the Company or a Company affiliate.

Term Commitment and Renewal Options: The Customer must commit to service for a term of either one or two years (Initial Term). The term of service will automatically renew on a month-to-month basis (Extended Term) upon expiration of its Initial Term and Extended Term(s), unless either the Customer or the Company provides written notification to terminate the term, which must be received by the other party no more than 60 days prior to the expiration of the existing term.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - G. Agent Program (Cont.)

Usage Charges: The per-minute rates located in Section 4.23.7 apply during each monthly period of a Customer's term of service intrastate outbound usage which originates via Local Network Connection and terminates via switched and Toll Free usage that originates switched and terminates Local Network Connection.

Classifications, Practices and Regulations:

Termination: Either the Customer or the Company may terminate this program for Cause. For purposes of this program only: (i) as to payment of invoices, "Cause" shall mean the Customer's failure to pay any invoice within 30 days after the date of the invoice; and, (ii) for all other matters, "Cause" shall mean a breach by the other party of any material provision of this program, provided that written notice of the breach has been given to the breaching party, and the breach has not been cured within 30 days after delivery of such notice.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - H. Local and Long Distance Service Plus Plan/Local and Long Distance Service-Trunk Solution/Local and Long Distance Service-Line Solution II

Eligibility: To be eligible for this plan, customers:

must designate a Company affiliate as its exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the Local and Long Distance Service Plus Plan/Local and Long Distance Service Trunk Solution/Local and Long Distance Service-Line Solution II as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.worldcom.com; ("Companion Interstate Service") and to certain exchange service as described in the appropriate Company-affiliate exchange service tariff ("Companion Local Service") concurrent with enrollment in this plan;

must subscribe to service under Special Customer Arrangement SCA Guide Types 1, 2, 3, 4, 5, 6, 7, 8, 9, 10, 11, or 12 as described in The Guide.

#### Definitions:

For the purposes of this plan, the following definitions apply:

"Eligible Intrastate Service" is defined as MCI WorldCom On-Net Service-Voice Outbound Service (excluding Card) usage that originates and terminates in one state.

"Eligible Interstate Usage" is defined as MCI WorldCom On Net Services Option 1 Outbound Service (excluding Card) usage that originates in the U.S. Mainland and Hawaii and terminates in the U.S. Mainland, Alaska, Hawaii, Puerto Rico, the U.S. Virgin Islands, Guam, and CNMI.

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - H. Local and Long Distance Service Plus Plan/Local and Long Distance Service-Trunk Solution/Local and Long Distance Service-Line Solution II

"Existing customers" are customers who, at the time of subscription to the plan, are receiving service under this tariff and the document containing the Companion Interstate Service.

"New customers" are customers who, at the time of subscription to the plan, are not receiving service under this tariff and the document containing the Companion Interstate Service.

# Monthly Recurring Charges:

A monthly recurring charge will apply for each Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service.

Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per-Local T-1 or ISDN-PRI basis. The following Monthly recurring charges apply:

<u>Offering</u>	Monthly Recurring Charge (per line, trunk, T-1, or ISDN-PRI)	
Α	Boca Raton	\$55.00
	Jacksonville	\$55.00
	Miami	\$55.00
	Orlando (BellSouth)	\$55.00
	Orlando (Sprint)	\$49.50
	Tampa	\$60.50
	Rest of State	\$66.00
В	\$65.00	
С	\$1,540.00	

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - H. Local and Long Distance Service Plus Plan/Local and Long Distance Service-Trunk Solution/Local and Long Distance Service-Line Solution II

## Benefits:

- a. Offering A: Upon installation of Companion Interstate Service and Companion
   Local Service, the customer will receive unlimited Eligible Intrastate Service,
   Eligible Interstate Usage and unlimited exchange service usage as described in
   the Companion Local Service tariffs or other appropriate governing document.
- Offering B and Offering C: Upon installation of Companion Interstate Service and b. Companion Local Service, the customer will receive an allotment of 1,250 minutes per monthly period per-Local Trunk or Per-Local T-1 that may be used for (i) Eligible Interstate Usage and (ii) Eligible Intrastate Service in that monthly period. Customers whose usage exceeds this allotment in any monthly period will be charged \$0.05 per minute for Eligible Intrastate Service that exceeds the allotment (Overage Usage Charge). When a call under this offering begins prior to the completion of customer's monthly allotment and ends after completion of the allotment, the customer will be charged \$0.05 per minute for Eligible Intrastate Service for the portion of the call occurring after completion of the allotment. For customers enrolling in this plan after the first day of a billing cycle, whose allotment begins after the first day of a billing cycle or whose enrollment in this plan ends before the last day of the billing cycle, the allotment will be prorated for that billing cycle. The portion of a Customer's allotment that is not exhausted at the end of a monthly billing period will not carry over to the next month.

In addition, the customer will receive unlimited exchange service usage as described in Companion Local Service tariffs or other appropriate governing document.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - H. Local and Long Distance Service Plus Plan/Local and Long Distance Service-Trunk Solution/Local and Long Distance Service-Line Solution II

## Discounts: 1/

These discounts are identical to, and shall not be in addition to, any discounts applicable to Companion Interstate Service and Companion Intrastate
Service.

For Offering A, B and C, the Company will provide a 5, 10 or 15 percent discount on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, 2) commit to a new term of service that equals or exceeds 1 year for a 5 percent discount, 2 years for a 10 percent discount, and 3 years for a 15 percent discount.

<u>Discounts:</u> These discounts are identical to, and shall not be in addition to, discounts applicable to the Local and Long Distance Service Plus Plan/Local and Long Distance Service-Trunk Solution/Local and Long Distance Service-Line Solution II pricing plan for intrastate and interstate long distance service.

A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount.

ISSUED: November 13, 2008

<sup>1/</sup> Effective August 1, 2006, these discounts will no longer be available for new customers.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - H. Local and Long Distance Service Plus Plan/Local and Long Distance Service-Trunk Solution/Local and Long Distance Service-Line Solution II

# Early Termination:

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only offered in MCImetro Access Transmission Services, Inc., F.P.S.C. Price List No. 2, the plan service offered under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service offered in MCImetro Access Transmission Services, Inc., F.P.S.C. Price List No. 2 and Companion Intrastate Service, Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

For new customers who disconnect Companion Local Service only offered in MCImetro Access Transmission Services, Inc., F.P.S.C. Price List No. 2, Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to WorldCom On Net Voice Services Option 1 under The Guide and MCI WorldCom On-Net Service-Voice under this tariff.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - H. Local and Long Distance Service Plus Plan/Local and Long Distance Service-Trunk Solution/Local and Long Distance Service-Line Solution II

For new customers who disconnect Companion Local Service offered in MCImetro Access Transmission Services, Inc., F.P.S.C. Price List No. 2 and Companion Intrastate Service, Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to WorldCom On Net Voice Services Option 1 under The Guide and MCI WorldCom On-Net Service-Voice under this tariff for intrastate interLATA service only.

# Other Conditions:

Services under this plan may not receive the benefits of any discounts or promotions including any term plan discounts except that customers will receive the benefits of the On-Net Plus Program for Domestic Intrastate Inbound Service and Intrastate Card Service Usage.

If Customer receives service under SCA Guide Types 2, 3, 4, 5, 6, 7, 8, 9, 10, 11, and 12 subscribes to domestic long distance service in combination with this plan, any discounts applicable to long distance service will apply to Overage Usage Charges.

Customers who subscribe to service via a company-designated Internet site will receive Electronic Billing invoicing only.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - H. Local and Long Distance Service Plus Plan/Local and Long Distance Service-Trunk Solution/Local and Long Distance Service-Line Solution II

The following disclaimers apply to Local and LD Service Plus Program Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY WORLDCOM. WorldCom will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

#### 3.30 Other Service Offerings (Continued)

# 3.30.5 MCI WorldCom On-Net Service – Voice (Continued)

#### I. Affinity 1 Program

1. Eligibility: To be eligible for this plan, customers:

> must designate a company affiliate as its exchange service carrier and the company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intralata toll calling;

must subscribe to the Affinity 1 Program as described in the service publication and pricing guide found at www.worldcom.com;

must pay a rate that is described as a non-residential, commercial, or business rate in the applicable local exchange service tariff for switched service; and,

may not receive service under a Special Customer Arrangement.

2. Definitions: For purposes of this plan, the following definitions apply:

> "Eligible Intrastate Service" is defined as WorldCom On-Net Service Voice Outbound Service usage and Inbound Service usage that originate and terminates in one state.

3. **Usage Charges:** 

> Customers will be charged the per minute rates located in Section 4.23.8 for Eligible inbound and outbound Intrastate Service.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - J. Affinity 2 Program1.

**Eligibility:** To be eligible for this plan, customers:

- must designate a company affiliate as its exchange service carrier and the company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intralata toll calling;
- must subscribe to the Affinity 2 Program as described in the service publication and pricing guide found at www.worldcom.com;
- must pay a rate that is described as a non-residential, commercial, or business rate in the applicable local exchange service tariff for switched service; and,
- may not receive service under a Special Customer Arrangement.
- 2. <u>Definitions:</u> For purposes of this plan, the following definitions apply:

"Eligible Intrastate Service" is defined as WorldCom On-Net Service Voice Outbound Service usage and Inbound Service usage that originate and terminates in one state.

3. Usage Charges:

Customers will be charged the per minute rates located in Section 4.23.9 for Eligible inbound and outbound Intrastate Service.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)

K. Local and Long Distance Line Solution Service

Eligibility: To be eligible for this plan, customers:

must designate a Company affiliate as its exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the Local and Long Distance-Line Solution Service as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="www.worldcom.com">www.worldcom.com</a>; ("Companion Interstate Service") and to certain exchange service as described in the appropriate Company-affiliate exchange service tariff ("Companion Local Service") concurrent with enrollment in this plan; must subscribe to service under Special Customer Arrangement SCA Guide Types 2, 3, 4, 5, 6, 7, 8, 9, or 10 as described in The Guide.

<u>Definitions:</u>For the purposes of this plan, the following definitions apply:

"Eligible Intrastate Service" is defined as MCI WorldCom On-Net Service-Voice Outbound Service (excluding Card) usage that originates and terminates in one state.

"Eligible Interstate Usage" is defined as MCI WorldCom On Net Services Option 1 Outbound Service (excluding Card) usage that originates in the U.S. Mainland and Hawaii and terminates in the U.S. Mainland, Alaska, Hawaii, Puerto Rico, the U.S. Virgin Islands, Guam, and CNMI. "Existing customers" are customers who, at the time of subscription to the plan, are receiving service under this tariff and the document containing the Companion Interstate Service.

New customers" are customers who, at the time of subscription to the plan, are not receiving service under this tariff and the document containing the Companion Interstate Service.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - K. Local and Long Distance Line Solution Service (Cont.)

## Monthly Recurring Charges:

A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Local Service.

Offering Monthly Recurring Charge (unlimited) Unlimited \$60.00

# Discounts: 1

These discounts are identical to, and shall not be in addition to, any discounts applicable to Companion Interstate Service and Companion Intrastate Service.

Customers committing to a new term of service that equals or exceeds 1 year for a 5 percent discount, 2 years for a 10 percent discount, or 3 years for a 15 percent discount will receive applicable discounts applied to Eligible Charges, in lieu of all other discounts.

Discounts: These discounts are identical to, and shall not be in addition to, discounts applicable to the Local and Long Distance Line Solution Service pricing plan for intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount.

<sup>&</sup>lt;u>1</u>/ Effective August 1, 2006, these discounts will no longer be available for new customers.

- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - K. Local and Long Distance Line Solution Service (Cont.)

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only offered in MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services , FPSC Price List No. 2, the plan service offered under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service offered in MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services , FPSC Price List No. 2 and Companion Intrastate Service, Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

For new customers who disconnect Companion Local Service only offered in MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services, FPSC Price List No. 2, Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to WorldCom On Net Voice Services Option 1 under The Guide and MCI WorldCom On-Net Service-Voice under this tariff.

For new customers who disconnect Companion Local Service offered in MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services , FPSC Price List No. 2 and Companion Intrastate Service, Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to WorldCom On Net Voice Services Option 1 under The Guide and MCI WorldCom On-Net Service-Voice under this tariff for intrastate interLATA service only.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - K. Local and Long Distance Line Solution Service (Cont.)

Other Conditions:

Services under this plan, including Interstate Inbound Service and Interstate Card Service may not receive the benefits of any discounts or promotions including any term plan discounts except that customers will receive the benefits of the On-Net Plus Program for Domestic Intrastate Inbound Service and Intrastate Card Service Usage.

Customers who subscribe to service via a company-designated Internet site will receive Electronic Billing invoicing only.

The following disclaimers apply to Local and Long Distance-Line Solution Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY WORLDCOM. WorldCom will install the Linebased Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.5 MCI WorldCom On-Net Service Voice (Continued)
    - L. Intrastate Plus Service

Eligibility: To be eligible for this program, customers:

must designate a company affiliate as its exchange service carrier and the company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intralata toll calling;

must receive interstate service under a New Special Customer Arrangement (SCA) Guide Type 6, 7, 8, 9 or 10 as described in the WorldCom's "Service Publication and Price Guide" located on the Company's website at www.worldcom.com;

**Usage Charges:** 

The per-minute usage rates as located in Section 4.23.10 will apply to Intrastate Usage.

1. Intrastate Plus Optional Calling Plan I

To receive the benefits of Intrastate Plus Optional Calling Plan I, Customer must demonstrate to MCI reasonable satisfaction that it will accept a competitor's offer of service in place of Intrastate Plus if it does not receive the benefits of this plan. Customer may not receive any other rates, discounts or other benefits applicable to Intrastate Plus. The rate are located in Section 4.23.10.1.

### 3.30 Other Service Offerings (Continued)

### 3.30.6 MCI WorldCom Business Solutions

### A. Description

MCI WORLDCOM Business Solutions is a switched, outbound, customized telecommunications service that may include an inbound service option. Outbound service may originate via switched WATS/Business Line Access and calling card access; inbound service, if elected, may terminate via switched WATS/Business Line Access. Unless otherwise specified, all calls are subject to an 30 second initial period and rounding to the next 6 second increment, except for Operator Assisted calls, which are subject to a 60-second initial period and rounding to the next 60-second increment. If the computed charge includes a fraction of a cent, the fraction is rounded to the nearest whole cent.

The rules and regulations governing the provision of service for toll free (inbound) service are identical to those for Toll Free Service. This service is being offered to Florida customers that may have a requirement to make intrastate calls. Intrastate services is provided in conjunction with Interstate Business Solutions service and is available only to customers subscribing to interstate Business Solutions service provided in MCI WorldCom's Interstate Tariff.

### B. Definitions

The following definitions will apply for all usage rates.

"Switched": A switched connection between a customer's premises and an MCI WorldCom POP which is not provided by MCI WorldCom or an MCI Worldcom affiliate.

"Card": Call origination using a Calling Card.

### 3.30 Other Service Offerings (Continued)

### 3.30.7 Agency Program A

Customers who subscribe to service through an agent of the Company with whom prior arrangements have been made will be charged the rates specified in Section 4.28.1.

### A. Description

Agency Program A offers 1+, dedicated outbound, inbound, and calling card service during all time of day rate periods. Customers must commit to a one-year term of service with a monthly minimum volume requirement (MVR) of either \$100 or \$1000. Customers who sign up with a \$1000 monthly minimum volume requirement must have at least one dedicated line used to access this service.

## B) Term Plans

Term plan customers are also subject to the following provisions:

Underutilization: For each monthly period of a customer's term of service in which a customer's Agency Program A usage charges (including intrastate usage, but not including any other applicable charges imposed by the Company or a third party) fail to equal or exceed the applicable monthly minimum volume requirement, the customer will be billed and required to pay an amount equal to the difference between the customer's actual usage and the applicable monthly minimum.

Termination with Liability: If a customer terminates service prior to the expiration of the customer's term of service under this plan, the customer will be billed and required to pay an amount equal to the applicable monthly minimum for each monthly remaining in the customer's term of service at the time of actual termination.

These charges will apply in addition to all incurred usage charges.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 3.30 Other Service Offerings (Continued)
  - 3.30.7 Agency Program A (Cont.)
    - C. Billing Increments
    - 1+, dedicated outbound, and inbound calls will be subject to an 18-second minimum duration. Calls that are more than 18 seconds will be billed in 6-second increments. Calling card calls will be subject to a 18-second minimum call duration, and calls that are more than 18 seconds will be billed in 6-second increments. If the computed charge for a call includes a fraction of a cent, the charge will be rounded to the nearest whole cent.
    - D) Rates:
    - 1+, Dedicated Outbound and Inbound Service: Customers will be charged the per minute rates specified in Section 4.28.1 for 1+, dedicated outbound, and inbound intrastate/intralata calls.

Switched Dedicated Outbound/Inbound \$0.1013 \$0.0761

Calling Card: Customers will be charged \$0.069 per minute for calling card calls. A per call surcharge of \$0.75 will apply to all calls.

## 3.30 Other Service Offerings (Continued)

### 3.30.11 Verizon Business Services Versatile T1 Plan 1/

Verizon Business Services Versatile T1 ("Versatile T1") is an integrated solution for businesses. Customer must order a minimum of 8 circuits, 4 of which must be local lines or trunks, and one of which must be Internet Dedicated Access, and a maximum of 24 circuits based on Packages A, B, or C as detailed below.

### **Eligibility**

Customer must subscribe to service under a Verizon Business Services Agreement, Guide Type 17, with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at www.verizonbusiness.com.

Customer must also subscribe to the Verizon Business Services Versatile T1 offering as described in MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services, F.P.S.C. Price List No. 2 ("Companion Intrastate Service").

Customer is not eligible to receive the benefits of any other program or promotion related to local, long distance, conferencing or internet dedicated access services.

### Monthly Recurring Charges

Customer will incur monthly recurring charges as described in MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services, F.P.S.C. Price List No. 2 ("Companion Intrastate Service").

Additional Charges and Calling Card rates can be found in Section 4.32.

### Benefits

Package A: Customers who subscribe to Versatile T1 service with a minimum 8 circuits or a maximum 12 circuits will receive up to 3000 Long Distance minutes which includes Toll Free, interstate / intrastate, and unlimited Local.

Package B: Customers who subscribe to Versatile T1 service with a minimum 8 circuits or a maximum 12 circuits will receive up to 6000 Long Distance minutes which includes Toll Free, interstate / intrastate, and unlimited Local.

Package C: Customers who subscribe to Versatile T1 service with a minimum 8 circuits or a maximum 12 circuits will receive up to 9000 Long Distance minutes which includes Toll Free, interstate / intrastate, and unlimited Local.

### Additional benefits

In addition, customers will receive the benefits of the Companion Intrastate Service as described in MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services, F.P.S.C. Price List No. 2.

<sup>1/</sup> Beginning October 1, 2005, Versatile T1 will no longer be available to new subscribers and existing customers will no longer be able to place new orders..

## 3.31 Optional Features

Various optional features are made available which may be limited to certain service arrangements. These optional features may be provided at no extra charge, or may be subject to monthly and non-recurring charges. A description of these features and their availability follows. Charges for these features can be found in Section 4.20.

### 3.31.1 Accounting Codes

Permits the customer to allocate usage charges to separate cost centers by entering a two, three or four digit number at the end of the dialing sequence.

### 3.31.2 Customer Account Codes

Sub-accounting codes on a customers bill, determined by the customer's input of three additional numbers after dialing a telephone number, are available at no charge to the customer.

### 3.31.3 Abbreviated Dialing

After switching center dial tone is secured, allows the caller to use four-digit abbreviated dialing codes in place of the standard seven-digit codes. This feature is available to customers with dedicated access facilities.

### 3.31.4 Project Billing

After securing switching center dial tone, the caller dials in the standard manner. A subsequent tone-burst signal alerts the caller to the need of dialing additional digits or a tone-code number to signify the specific sub-billing account or project. This feature is available to customers with dedicated access facilities.

### 3.31.5 Call Records on Magnetic Tape

Upon request, the Company will provide detailed monthly billing call records on magnetic tape. Special Service Charges may apply on a case-by-case basis to set up various customer requested program changes. This feature is available to all customers.

## 3.31 Optional Features (Continued)

## 3.31.6 1 - Plus Screening

Allows the customer to dial 1 on all long distance numbers. Dial 1 action is not appropriate when utilizing the Company switch. This feature will automatically remove the one from the dialed number so that the Company switch may process the call. This feature is available to customers with dedicated access facilities.

### 3.31.7 Area Code Screening

Gives the customer the ability to block access to certain customer designated area codes. This feature is available to customers with dedicated access facilities.

### 3.31.8 Verified Account Codes

Allows the customer to assign 2 or 3 digit account codes that must be validated by the Company switch before a call will be completed.

### 3.31.9 Prefix NPA Addition

The Company switch requires 10 digits in order to complete a local call. With this feature, the switch will insert the 3 digit local area code (NPA).

### 3.31.10 Longer Distance Autodialer Optional Features

The following optional features are available to Autodialer customers.

- A. <u>Speed Numbers</u> 2 digit access to frequently called numbers Up to 50 numbers can be programmed.
- B. <u>Zero "+" Screening</u> Blocks operator assisted and international calls at the option of the customer.
- C. Call Routing Will route calls to alternative carrier if the Company's trunks are busy.
- D. <u>NPA/NXX Blocking</u> Specific NPA's and/or NXX's can be blocked at the customer's option. Not available if the Speed Number feature is utilized.
- E. <u>Travel Code</u> Usage charges associated with a travel code will appear on the same account as the Autodialer charges. The travel code will allow the customer to use the service at locations other than the Autodialer Service location when traveling.

# 4.3 MTS Rates

### 4.3.1 Plan 17

A. Per Minute Rates:

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$0.1941 \$0.1339 \$0.1060

### 4.3.2 Plan 75

A. Per Minute Rates:

<u>Day Evening</u> <u>Night/Weekend</u> \$0.2085 \$0.1395 \$0.1195

## 4.3.3 Plan 76

A. <u>Per Minute Rates</u>:

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$0.20 \$0.20

## B. Monthly Volume Discount:

Total Monthly Usage	Discount
\$0.00 - \$200.99	0%
\$201.00 - \$299.99	3%
\$300.00 - \$399.99	5%
\$400.00 - \$599.99	7%
\$600.00 - \$999.99	10%
\$1,000.00 +	15%

## 4.3 MTS Rates (Continued)

### 4.3.4 Plan 78

A. Per Minute Rates:

<u>Day</u>	<u>Evening</u>	Night/Weekend
\$0.1835	\$0.1225	\$0.1095

### B. <u>Monthly Volume Discount</u>:

Total Monthly Usage	Discount
\$0.00 - \$75.00	0%
\$75.01 - \$125.00	2%
\$125.01 - \$200.00	3%
\$200.01 +	5%

## 4.3.5 Plan 91

A. <u>Per Minute Rates</u>:

D	av	Evening		-Night/Weekend	1
ost	-			J	
1 <sup>st</sup> Min	Add'l Min	1 <sup>st</sup> Min	Add'l Min	1 <sup>st</sup> Min	Add'l Min
	00.4005	<del></del> -	00.1055		
\$0.2425	\$0.1995	\$0.1625	\$0.1255	\$0.1325	\$0.1123

### B. Monthly Volume Discount:

Total Monthly Usage	Discount
\$0.00 - \$75.00	0%
\$75.01 - \$150.00	2%
\$150.01 - \$249.99	3%
\$250.01 - \$9,999.99	5%

### 4.3.7 RingAmerica 1/

RingAmerica customers are charged a base monthly fee of \$7.00 for up to sixty (60) minutes of evening and night/weekend calling. Additional evening, night/weekend calling will be billed at a per hour rate of \$6.00 for each sixty (60) minutes or \$.10 cents per minute.

As an optional feature Daytime calling is available to RingAmerica customers at a ten (10%) percent discount off the Dial USA Day rates found in Section 4.3.6.

 $<sup>\</sup>underline{1}$ / Beginning September 5, 2003, this service will no longer be available to new subscribers.

## 4.3 MTS Rates (Continued)

### 4.3.8 TropiCall

### A. Rate Schedule

-Rate Per Minute of Use-Peak Off Peak \$0.22 \$0.12

### B. <u>Billing Increments</u>

Calls are rated at an initial one (1) minute increment and in six (6) second increments thereafter.

# 4.3.9 <u>Message Telephone Service</u>

## A. <u>Per Minute Rates</u>

	[	Day	Evening		-Night/Weeken	d
<u>Mileage</u>	1 <sup>st</sup> Min	Add'l Min	1 <sup>st</sup> Min	Add'l Min	1 <sup>st</sup> Min	Add'l Min
0 - 10	\$.1791	\$.0887	\$.1318	\$.0665	\$.0846	\$.0444
11 - 22	\$.2488	\$.1576	\$.1841	\$.1182	\$.1244	\$.0788
23 - 55	\$.2488	\$.2157	\$.1841	\$.1592	\$.1343	\$.1133
56 - 124	\$.2492	\$.2172	\$.1845	\$.1629	\$.1388	\$.1172
125 - 292	\$.2497	\$.2236	\$.1849	\$.1678	\$.1413	\$.1222
293 - 430	\$.2497	\$.2266	\$.1849	\$.1697	\$.1428	\$.1222
431+	\$.2522	\$.2344	\$.1867	\$.1758	\$.1463	\$.1271

## B. <u>Billing Increments</u>

Calls are billed in one minute increments after an initial minimum billable period of one minute.

### 4.3 MTS Rates (Continued)

### 4.3.9 Message Telephone Service (Continued)

### C. Volume Discounts

Discounts will be applied to a customer's usage charges based upon the customer's total interstate, intrastate and directory assistance MTS usage charges for the account for the billing period. International calls will not be included in determining a customer's total usage charges, and will not receive discounts under this schedule. Discount percentages increase as total usage charges increase, and vary by rate period. The discount percentage corresponding to a given rate period and total usage level will be applied to all usage in that rate period.

<u>Usage</u>	<u>Day</u>	Evening	Night/Weekend
\$0 - \$24.99	0%	0%	0%
\$25 - \$49.99	1%	0%	0%
\$50 - \$99.99	2%	1%	0%
\$100 - \$199.99	4%	1%	0%
\$200 +	5%	2%	0%

### 4.3.10 Dial Up-WATS

## A. <u>Usage Charges</u>

### 1. Per Minute Rates

<u>Peak</u>	<u>Off-Peak</u>
\$0.1815	\$0.1361
\$0.1815	\$0.1361
\$0.1815	\$0.1361
	\$0.1815

### 4.3 MTS Rates (Continued)

### 4.3.10 <u>Dial Up-WATS</u> (Continued)

### A. Usage Charges (Continued)

### 2. Rate Periods

The time of day rate periods for Dial Up-WATS will be Peak and Off-Peak. Peak rates apply to all calls that occur between 8AM and 5PM Monday through Friday, except on Company-recognized holidays. Off Peak rates apply to all other calls, including holidays.

### B. Billing Increments

Calls are billed in six second increments after an initial minimum billable period of six seconds.

## C. Volume Discounts

Discounts will be applied to a customer's usage charges based upon the customer's total interstate and intrastate Dial Up-WATS usage charges for the account for the billing period. Directory assistance and international charges will not be included in determining a customer's total usage charges, and will not receive discounts under this schedule. The discount percentage corresponding to a given usage level will be applied to all usage for that billing period.

<u>Usage</u>	Discount
\$0 - \$499.99	0%
\$500 - \$999.99	3%
\$1000 - \$1999.99	5%
\$2000 - \$3999.99	8%
\$4000 +	11%

#### 4.3 MTS Rates (Continued)

### 4.3.11 New Precision Service

#### Α. **Usage Charges**

Per minute usage rates for a billing period apply seven days a week, 24 hours per day.

<u>Rate</u>
\$.2145
\$.1865
\$.1715
\$.1615
\$.1590

### B. Billing Increments

Calls are billed in six second increments after an initial minimum billable period of six seconds.

### 4.3.12 Precision Plus

#### A. **Usage Charges**

1. Per Minute Usage Rates

\$0.1700 per minute

2. Rate Periods:

Rates apply 24 hours a day, 7 days a week.

#### B. Billing Increments

Calls are billed in six second increments.

### 4.3.15 RingAmerica Plus 1/

- A. **Usage Charges**
- 1. Per Minute Usage Rates \$0.10 per minute
- 2. Rate Periods:

Rates apply 24 hours a day, 7 days a week.

<sup>1/</sup> Beginning September 5, 2003, this service will no longer be available to new subscribers

# 4.4 WATS Services

## 4.4.1 Plan 5

A. Per Minute Rate: \$.1995

B. Monthly Volume Discount:

Total Monthly Usage	<u>Discount</u>
\$ 0.00 - \$599.99	5%
\$600.00 - \$1,199.99	10%
\$1,200.00 +	15%

## 4.4.2 Plan 72

A. Per Minute Rates: 1/

InterLATA IntraLATA \$.1765 \$.1685

## 4.4.3 Plan 90

A. <u>Per Minute Rates</u>:

<u>Day</u>	Evening	Night/Weekend
\$.1835	\$.1505	\$.1150

## B. <u>Monthly Volume Discount</u>:

Total Monthly Usage	<u>Discount</u>
\$1.00 - \$499.99	0%
\$500.00 - \$999.99	5%
\$1,000.00 - \$1,999.99	10%
\$2,000.00 - \$2,999.99	12%
\$3,000.00 +	15%

<sup>1/</sup> No monthly volume discounts apply.

## 4.4 <u>WATS Services</u> (Continued)

### 4.4.4 <u>Dedicated WATS</u>

### A. Usage Charges

### 1. Per Minute Rates

<u>Mileage</u>	<u>Day</u>	<u>Evening</u>	Night/Weekend
0 - 292		\$.0985	\$.0840
293 +	\$.1365	\$.1130	\$.0935

## 2. Rate Periods

Day, Evening and Night/Weekend rate periods will apply as defined in Section 4.2. On holidays, evening rates will apply unless a lower rate would normally apply.

## B. Billing Increments

Calls are billed in increments of six seconds after an initial minimum billable period of six seconds.

## C. <u>Volume Discounts</u>

Discounts will be applied to a customer's usage charges based upon the customer's total interstate and intrastate usage charges for the account for the billing period. Directory assistance and international calls will not be included in determining a customer's total usage and will not receive discounts under this schedule. The discount percentages corresponding to a total usage level will be applied to all usage in a given rate period.

### 4.4 <u>WATS Services</u> (Continued)

### 4.4.4 <u>Dedicated WATS</u> (Continued)

### C. <u>Volume Discounts</u> (Continued)

<u>Usage</u>	<u>Day</u>	Evening	Night/Weekend
\$0 - \$4999.99	0%	0%	0%
\$ 5,000 - \$9,999.99	5%	5%	3%
\$10,000 - \$24,999.99	11%	11%	3%
\$25,000 - \$34,999.99	13%	13%	3%
\$35,000 +	18%	13%	3%

### Example of Usage Volume Discounts

If a customer's usage during a billing period totals \$21,000, usage volume discounts associated with the \$10,000 to \$24,999.99 usage level will apply for each respective rate period as follows:

Rate Periods	<u>Usage</u>	Applicable Discount	<u>Discount</u>
Day	\$15,000	11%	\$1650
Evening	\$5,000	11%	\$550
Night	\$1,000	3%	\$30
Total	\$21,000		\$2,230

### D. Minimum Average Time Requirement

If an account's total interstate and intrastate usage divided by the number of domestic calls is less than 30 seconds, the customer's usage charges will be determined by multiplying the number of calls by 30 seconds to arrive at billable usage, and multiplying that usage amount by the average billable rate for the calls made during that period to determine total usage charges.

## 4.4 <u>WATS Services</u> (Continued)

## 4.4.5 Precision Fit Service

### A. <u>Usage Charges</u>

## 1. Per Minute Usage Rates

	D	AY Rate Period-		
	0 Minutes	150,001	300,001	Non-Day
Terminating LATA	to 150,000	to 300,000	or More	Rate Period
448 - Pensacola	\$0.0990	\$0.0965	\$0.0941	\$0.0825
450 - Panama City	\$0.0990	\$0.0965	\$0.0941	\$0.0825
452 - Jacksonville	\$0.0940	\$0.0917	\$0.0893	\$0.0825
454 - Gainesville	\$0.0990	\$0.0965	\$0.0941	\$0.0825
456 - Daytona Beach	\$0.0990	\$0.0965	\$0.0941	\$0.0825
458 - Orlando	\$0.0990	\$0.0965	\$0.0941	\$0.0817
460 - S.E. Florida	\$0.0940	\$0.0917	\$0.0893	\$0.0825
939 - Ft. Myers	\$0.0990	\$0.0965	\$0.0941	\$0.0817
952 - Gulf Coast	\$0.0990	\$0.0965	\$0.0941	\$0.0817
953 - Tallahassee	\$0.0990	\$0.0965	\$0.0941	\$0.0817

## 2. Rate Periods

Day rates apply from 8AM to, but not including, 5PM weekdays. Non-day rates apply during all other times and on holidays unless a lower rate would apply.

### B. Billing Increments

Calls are billed in six second increments with a minimum average time requirement (MATR) during each billing period of thirty (30) seconds per call.

#### 4.5 800 Services

#### 4.5.1 Plan 50

A. Per Minute Rates:

> Night/Weekend Day Evening \$0.28 \$0.24 \$0.24

B. Monthly Line Charge: \$15.00

C. Non-Recurring Installation Fee: \$49.00

#### 4.5.2 Plan 65

A. Per Minute Rates:

> Day Evening Night/Weekend \$0.2255 \$0,2200 \$0.2485

B. Monthly Line Charge: \$15.00 per number

C. \$38.00 per number Non-Recurring Installation Fee:

D. Monthly Volume Discount:

Total Monthly Usage	Discount
\$0.00 - \$249.99	0%
\$250.00 - \$499.99	2%
\$500.00 - \$949.99	5%
\$950.00 +	7%

#### 4.5.4 Metromedia 800 Service

Α. Per Minute Rates

> The applicable rate schedule is dependent upon the service option utilized to deliver the call to the customer.

## **Shared Termination:**

Evening Day Night/Weekend \$0.2163 \$0.1854 \$0.1494

### **Dedicated Termination**:

Day Evening Night/Weekend \$0.1385 \$0.1221 \$0.1030

The monthly service fee will be waived for each month in which usage exceeds \$10.00. Also, the monthly 1/ service fee can be waived by Association groups and Alumni programs.

## 4.5 <u>800 Services</u> (Continued)

### 4.5.4 Metromedia 800 Service (Continued)

### B. Rate Periods

Day, Evening and Night/Weekend rate periods will apply as defined in Section 4.2. On holidays, Evening rates will apply unless a lower rate would normally apply. The rate period is determined by the local time at the customer's location.

# C. <u>Billing Increments</u>

Calls are billed in increments of six seconds and subject to a minimum average time requirement of 30 seconds per call during a billing period.

### D. Volume Discounts

Discounts will be applied to a customer's usage charges based upon the customer's total interstate and intrastate 800 Service usage charges for an account for the billing period. Different discount tables are applicable to the Shared termination and Dedicated Termination options. Discount percentages increase as total usage increases, and vary by rate period. The discount percentage corresponding to a given rate period and total usage level will be applied to all usage in that rate period.

### 1. Shared Termination

<u>Usage</u>	<u>Day</u>	Evening	Nights/Weekends
\$0 - \$249.99	0.0%	0.0%	0.0%
\$250 - \$499.99	3.5%	2.0%	2.0%
\$500 - \$999.99	8.5%	5.0%	5.0%
\$1000 +	12.5%	9.0%	6.5%

#### 4.5 800 Services (Continued)

#### 4.5.4 Metromedia 800 Service (Continued)

#### D. Volume Discounts (Continued)

#### 2. **Dedicated Termination**

<u>Usage</u>	Day	Evening	Nights/Weekends
\$0 - \$999.99	0.0%	0.0%	0.0%
\$1,000 - \$2,499.99	5.0%	3.0%	3.0%
\$2,500 - 4,999.99	6.5%	3.0%	3.0%
\$5,000 - \$9,999.99	8.0%	5.0%	5.0%
\$10000 +	12.0%	6.0%	6.0%

### E. Minimum Usage Requirement

There is a minimum average usage requirement of \$125.00 per activated voice grade equivalent line in a customer's dedicated access line group. If actual usage does not meet or exceed the minimum usage amount, this minimum amount will be billed in lieu of the actual usage charges.

#### F. 800 Service Number Charge

There is a non-recurring charge and monthly charge associated with the establishment and maintenance of a customer's 800 routing information in the Company's network. The charges apply to each 800 number provisioned.

Shared Termination: Dedicated Termination:	Non-recurring \$50.00 No Charge	Monthly Recurring \$20.00 No Charge for first number; \$20 for each
		additional number.

### 4.5 <u>800 Services</u> (Continued)

### 4.5.5 800 Directory Listing

		Monthly
	Set-Up	Recurring
	Charge	Charge
Nationwide 800 Directory Listing (per 800 number)	\$15.00	\$12.50
Expedite Directory Listing - Major	\$25.00	N/C
Expedite Directory Listing - Minor	\$20.00	N/C

## 4.5.6 Responsible Organization Charges

Where the Company serves as a Responsible Organization (Resp. Org.) for a non-Company 800 Service customer, the Company will pass on the tariffed Local Exchange Carrier charges for SMS/800 database and related services. In addition, the following Company charges will apply:

		Monthly
	Set-Up	Recurring
<u>Service</u>	<u>Charge</u>	<u>Charge</u>
Set-Up 800 Number (per 800 number)	\$50.00	\$1.00
Modify 800 Record (add/change 800 number)		
Vertical or Enhanced Features	\$25.00	N/C

## 4.5.7 <u>Takeback and Transfer (TBX)</u>

Takeback and Transfer is a toll-free service which allows Call Centers to transfer a toll-free call in progress to another remote site using pre-defined keypad commands and transfer digits. The service will be billed at the usage rates for Intelenet Service as filed in Section 4.16 and the following additional charges will apply:

Se	

One Time Installation Charge:	\$500.00
Monthly Recurring Charge per 8XX No.:	\$80.00
Change Charge:	\$100.00
Per Call Surcharge:	\$0.04

## 4.6 <u>Complementary Services</u> (Continued)

### 4.6.2 <u>Metromedia Calling Card</u>

### A. Option A - Full Minute Billing

## 1. Per Minute Rates 1/

		Day	Evening		-Night/Weeken	d
Mileage Band	1 <sup>st</sup> Min	Add'l Min	1 <sup>st</sup> Min	Add'l Min	1 <sup>st</sup> Min	Add'l Min
0 - 10	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500
11 - 22	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500
23 - 55	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500
56 - 24	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500
25 - 292	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500
293 - 430	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500
431 +	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500	\$.2500

## 2. Per Call Surcharge

A surcharge of \$.25 applies to each call.

## 3. <u>Volume Discounts</u>

Discounts will be applied to a customer's day usage charges based upon the customer's total interstate and intrastate usage for the billing period. For example, if a customer's usage for the billing period totaled \$550, the customer would qualify for a 10% volume discount on day use totaling \$55 for the month.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 4.6 <u>Complementary Services</u> (Continued)
  - 4.6.2 <u>Metromedia Calling Card</u> (Continued)
    - A. Option A Full Minute Billing (Continued)
      - 3. <u>Volume Discounts</u> (Continued)

Total Monthly Usage	Day
\$0 - \$49.99	0%
\$50 - \$99.99	2%
\$100 - \$199.99	5%
\$200 - \$499.99	8%
\$ 500 +	10%

- B. Option B Six Second Billing
  - 1. Per Minute Rates

<u>Mileage</u>	<u>Day</u>	<u>Evening</u>	Night/Weekend
1 - 10	\$.2500	\$.2500	\$.2500
11 - 20	\$.2500	\$.2500	\$.2500
21 - 40	\$.2500	\$.2500	\$.2500
41 - 70	\$.2500	\$.2500	\$.2500
71 -100	\$.2500	\$.2500	\$.2500
101 -150	\$.2500	\$.2500	\$.2500
151 -330	\$.2500	\$.2500	\$.2500
331 -430	\$.2500	\$.2500	\$.2500
431 -925	\$.2500	\$.2500	\$.2500
926+	\$.2500	\$.2500	\$.2500

### 4.6 <u>Complementary Services</u> (Continued)

### 4.6.2 <u>Metromedia Calling Card</u> (Continued)

### B. Option B - Six Second Billing (Continued)

### 3. Volume Discounts

Discounts will be applied to a customer's day, evening, night and weekend usage charges based upon the customer's total interstate and intrastate usage for the billing period. For example, if a customer's usage totaled \$550 for the billing period, the customer would qualify for a volume discount of 4% on day use and 3% for evening, night and weekend use.

<u>Usage</u>	<u>Day</u>	Evening	Night/Weekend
\$0 - \$249.99	0.0%	0.0%	0.0%
\$250 - \$499.99	2.0%	1.0%	1.0%
\$500 - \$999.99	4.0%	3.0%	3.0%
\$1,000 - \$4,999.99	7.0%	6.0%	6.0%
\$ 5,000 - \$9,999.99	10.0%	9.0%	9.0%
\$10,000 - \$14,999.99	11.0%	10.0%	10.0%
\$15,000 - \$24,999.99	12.0%	11.0%	11.0%
\$25,000 +	13.0%	12.0%	12.0%

### 4. Per Call Surcharges

The following per call surcharges apply based on the originating LATA of the call.

Home Base \$.25 per call Travel \$.25 per call

### 4.6 <u>Complementary Services</u> (Continued)

### 4.6.2 Metromedia Calling Card (Continued)

### C. Operator Assisted Calls

Company Calling Card operator assisted calls are billed to the customer's Company Calling Card number. Surcharges apply in addition to all other surcharges applicable for this service arrangement.

Station-to-Station \$0.40 per call Person-to-Person \$1.18 per call

### 4.6.3 800 Travel Service

### A. Per Minute Rates:

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$0.39 \$0.39

### 4.6.4 LEC Billed Measured Service

## A. <u>Usage Charges</u>

- 1. Reference per minute rates applicable to Dial USA 1/ found in Section 4.3.6.
- 2. In addition to per minute rates, a \$0.60 surcharge applies to each call.

ISSUED: November 13, 2008

<sup>1/</sup> Former Metromedia customers who have the existing service arrangements prior to September 15, 1993 will be billed per minute rates applicable to Custom Call 100 found in Section 6.1.23.

## 4.6 <u>Complementary Services (Continued)</u>

### 4.6.5 Operator Services

Operator Service charges include per minute rates and per call surcharges. No time-of-day/day-of-week discounts apply to the Operator Services per call surcharges.

## A. <u>Per Minute Usage Rates</u>:

### 1. <u>IntraLATA/Intrastate</u>

	Day		Ev	Evening		Night/Weekend	
Rate Mileage	1 <sup>st</sup> Min	Add'l Min	<u>1<sup>st</sup> Min</u>	Add'l Min	1 <sup>st</sup> Min	Add'l Min	
1 - 10	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	
11 - 22	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	
23 - 55	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	
56 – 124	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	
125 - 292	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	
293 - 430	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	
431 - 624	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	

## (a) Per Call Surcharges:

The following per call surcharges apply in addition to per minute rates found in Section 4.6.5.A (1) above.

LEC Calling Card	\$1.75
Bong Credit Card	\$1.75
Station-to-Station	\$1.75
Billed to Third Party	\$1.75
Person-to-Person	\$3.25

## 4.6 <u>Complementary Services (Continued)</u>

### 4.6.5 Operator Services (Continued)

## A. <u>Per Minute Usage Rates (Continued)</u>

## 2. <u>InterLATA/Intrastate</u>

	Day		Ev	ening	Night/V	Veekend
Rate Mileage	1 <sup>st</sup> Min	Add'l Min	<u>1<sup>st</sup> Min</u>	Add'l Min	1 <sup>st</sup> Min	Add'l Min
1- 10	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000
11- 22	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000
23- 55	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000
56-124	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000
125-292	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000
293-430	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000
431-624	\$.3000	\$.3000	\$.3000	\$.3000	\$.3000	\$.300

## (a) Per Call Surcharges:

The following per call surcharges apply in addition to per minute rates found in Section 4.6.5.A (2) above.

LEC Calling Card Bong Credit Card Station-to-Station	\$1.75 \$1.75 \$1.75
Billed to Third Party	\$1.75
Person-to-Person	\$3.25

# 4.6 <u>Complementary Services</u> (Continued)

# 4.6.5 Operator Services (Continued)

# C. <u>Application of Service Charges and Surcharges</u>

TYPE OF CALL	OPERATOR SERVICE <u>CHARGE</u>	OPERATOR DIALED SURCHARGE
Dialed Station (customer dialed 1+)	No	No
Customer Dialed Calling Card Station (0+) Operator Station (customer dialed 0+)	Yes	No
collect, billed to third number	Yes	No
Operator Station (operator dialed 0-) collect, billed to third number, sent paid Operator Station (operator dialed 0-)	Yes	Yes
billed to a calling card	Yes	No
Person to Person (customer dialed 0+) collect, billed to third number, calling card, sent paid	Yes	No
Person to Person (operator dialed 0-) collect,	165	NO
billed to third number, sent paid	Yes	Yes
Person to Person (operator dialed 0-), billed to a calling card	Yes	No
Real Time Rated (customer dialed 0+)	Voo	No
coin paid, time and charges Real Time Rated (operator dialed 0-)	Yes	No
coin paid, time and charges	Yes	Yes

### 4.6 <u>Complementary Services</u> (Continued)

### 4.6.6 <u>Directory Assistance Rates</u> 1/

A. Customers will be billed the following undiscounted) per call charge for each Directory Assistance call unless otherwise stated in the specific product rates and charges.

Per Call Charge

Business \$1.40 Residential \$0.65

B. Customers will be billed the following per call charge when they access the Company Operator and request Directory Assistance information.

Operator Assisted Directory Assistance Per Call Charge \$1.00

## 4.6.7 <u>Credit Card Reader Phones</u>

\$ .66 per call

### 4.7 <u>Acclaim Family of Services</u>

## 4.7.1 Acclaim Services

### A. Rate Periods

The time of day rate periods for Answer Services will be Peak and Off-Peak. Peak rates apply to all calls that occur between 8AM and 5PM Monday through Friday, except on Company-recognized holidays. Off-Peak rates apply to all other calls.

### B. Billing Increments

All Acclaim products are billed in six (6) second initial increments and are rounded to the next higher six (6) second increment.

### C. Per Minute Rate Schedule

Per minute rates apply as shown below. In addition, a volume discount will apply to each month's total Peak usage as outlined in Section 4.7.1.D following. The volume discount is based on total usage of Acclaim products.

Accla	aim I	Accla	im III
<u>Peak</u>	Off Peak	<u>Peak</u>	Off Peak
\$0.1472	\$0.1099	\$0.1570	\$0.1256
Accla	aim II	Accla	im IV
<u>Peak</u>	Off Peak	<u>Peak</u>	Off Peak
\$0.1021	\$0.0785	\$0.1060	\$0.0824

<sup>1/</sup> Customers who have existing service arrangements billed on the Metromedia billing system will be charged \$0.64 per call for Directory Assistance.

\_

### Acclaim Family of Services (Continued) 4.7

#### Acclaim Services (Continued) 4.7.1

#### D. Volume Discount

		% Disco	unt	
Usage Level \$	Acclaim I	Acclaim II	Acclaim III	Acclaim IV
\$0 - \$249.99	0%	0%	0%	0%
\$250 - \$499.99	5.4%	7.4%	0%	0%
\$500 - \$749.99	6.8%	8.7%	0%	0%
\$750 - \$499.99	8.1%	9.9%	3.6%	1.7%
\$1,500 - \$2,499.99	9.4%	11.5%	7.1%	3.4%
\$2,500 - \$4,999.99	11.5%	13.3%	10.5%	5.6%
\$5,000 - \$9,999.99	12.8%	14.9%	14.1%	7.3%
\$10,000-\$24,999.99	14.1%	16.5%	17.7%	9.0%
\$25,000-\$49,999.99	14.1%	16.5%	21.3%	10.7%
\$50,000 +	14.1%	16.5%	24.9%	12.4%
E. Recurring Char	rges <u>1</u> /			
1. Validat	ed Project Acct (	Code: Monthly		\$10.00

1.	Validated Project Acct Code; Monthly	\$10.00
2.	800 Number Charge (per 800 #)	\$10.00
3.	800 Message Referral	
	(a) MCIW 800 to Non-MCIW 800 (3 mos)	\$ 0.00
	(b) Non-MCIW 800 to MCIW 800 Message	
	Referral (3 months)	\$25.00
4.	Real Time Automatic Number Identification	
	(ANI) service	\$200.00
5.	Magnetic Tape	\$15.00
6.	Customized Account Codes	\$50.00

Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access <u>1</u>/ shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

## 4.7 <u>Acclaim Family of Services</u> (Continued)

## 4.7.1 <u>Acclaim Services</u> (Continued)

## F. Non-Recurring Charges 1/

1. 2. 3. 4. 5.	Dedic Each T-1 Ir	ated Project Acct Codes (set up) cated Access Line (1st Line) Installation Additional Special Access Line installation (per T-1) Features	\$15.00 \$200.00 \$50.00 \$450.00
	(a)	Area Code Routing (per 800 #)	\$50.00
	(b)	Area Code Blocking (per 800 #)	\$50.00
	(c)	Dialed Number Identification Service (per 800#)	\$50.00
	(d)	Time of Day Routing (per 800 #)	\$50.00
	(e)	Day of Week Routing (per 800 #)	\$50.00
	(f)	Emergency Restoration (per 800 #)	\$50.00
	(g)	Emergency Re-routing (per 800 #)	\$50.00
	(h)	Real Time ANI; setup (per trk grp)	\$350.00
6.	800 F	Referral	
	(a)	MCIW 800 to Non-MCIW 800; setup	\$100.00
	(b)	MCIW 800 to Non-MCIW 800;	
		add'l 3 month period	\$100.00
	(c)	Non-MCIW 800 to MCIW 800 Message referral, setup	\$25.00
7.	Magn	netic Tape (setup)	\$50.00
8.	Custo	omized Account Codes; setup	\$50.00
9.	Admi	nistrative Change (per order)	\$15.00
10. Ex	pedite (	Order (per order)	\$100.00

<sup>1/</sup> Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 4.7 <u>Acclaim Family of Services</u> (Continued)
  - 4.7.2 Acclaim Extended Service Plan ("ESP")

Acclaim services are available to customers through Extended Service Plan ("ESP") option if customer agrees to commit to service for a term of twelve (12) or twenty-four (24) 1/ months and generate a minimum monthly gross usage. Customers who elect an ESP option are subject to the following in addition to those conditions listed in Section 3.13.2:

- A. Customers who elect ESP for Acclaim I or III must meet the minimum usage requirement of \$500 per month per account. Customers who elect ESP for Acclaim II or IV must meet the minimum usage requirement of \$1,000 per month per account.
- B. Customers who subscribe to Acclaim through an ESP option will receive the following discounts in addition to those described in Section 4.7.1.D.
- C. This additional discount is applied to all domestic usage and is given only to customers who have met the monthly minimum usage requirement.
- D. Total intrastate, interstate, and international usage contributes to discount level.
- E. The method of calculation to determine the ESP discount is explained in Section 4.7.2.F following.

	Acclaim I or III		
Monthly Usage (\$)	<u>Peak</u>	Off-Peak	
\$0 - \$499.99	0%	0%	
\$500 - \$1,499.99	3%	4%	
\$1,500 - \$4,999.99	4%	4%	
\$5,000 +	5%	4%	

ISSUED: November 13, 2008

<sup>1/</sup> The twenty-four (24) month Extended Service Plan ("ESP") option is only available to customers who subscribe to Acclaim II and/or Acclaim IV services.

### 4.7 <u>Acclaim Family of Services</u> (Continued)

### 4.7.2 <u>Acclaim Extended Service Plan ("ESP")</u> (Continued)

### E. (Continued)

L. (Oontinaca)						
,		Acclaim II or IV				
			-12-Months		24-l	Months
Monthly Usage (\$)	Peak		Off-Peak	Peak	Off-Peak	
\$0 - \$999.99		0%	0%	<u> </u>	0%	0%
\$1000 - \$1,499	.99	3%	4%		6%	7%
\$1,500 - \$4,999	9.99	4%	4%		7%	7%
\$5,000 +		5%	4%		8%	7%

### F. Method of Calculation for ESP Discount

Determine the total monthly usage (all domestic and international usage combined). For the purpose of figuring the ESP discount, the percentage that corresponds to this total usage will apply. The ESP discount is applied to domestic usage after any Acclaim discount (Section 4.7.1.D) has been applied.

### **EXAMPLE**

### Assumptions:

Customer subscribes to Acclaim I service.

## Calculation of ESP Discount:

1. Determine appropriate discount percentage that applies. For a customer on the twelve (12) month plan, the \$1400 usage level (above) qualifies for the following ESP discounts:

Peak Usage (\$600) = 3% Off Peak Usage (\$500) = 4%

- 2. The ESP discount applies to domestic usage only; i.e., [( $$600 48.60 \ \underline{1}$ /) x 3% = \$16.54] + [ $$500 \times 4\% = $20.00$ ].
- 3. Subtract the amount calculated in step 2 from the total Peak and Off-Peak usage respectively; i.e. [\$600 48.60 = \$551.40] + [\$500 20.00 = \$480.00].

ISSUED: November 13, 2008

<sup>1/</sup> Acclaim I discount to be applied to Peak usage only.

#### 4.10 **HotelNet Services**

## 4.10.1 HotelNet WATS

#### A. **Usage Charges**

#### 1. **Shared Access Option**

Reference per minute rates and volume discounts applicable to Dial-Up WATS Service appearing in Section 4.3.10.

#### 2. **Dedicated Access Option**

Reference per minute rates and volume discounts applicable to Dedicated WATS Service appearing in Section 4.4.4.

Note: A UCMD of 5% may be applied to total usage after standard discounts are applied provided time-of-day traffic qualifications are met.

### 4.10.2 HotelNet Precision

#### A. **Usage Charges**

#### 1. Shared Access Option - Per Minute Rates

Total Usage In Minutes	<u>Rate</u>
0 - 249	\$.1750
250 - 2,399	\$.1650
2400 - 4,799	\$.1625
4800 -17,999	\$.1575
18,000 & Over	\$.1525

#### 2. Dedicated Access Option - Per Minute Rates

Total Minutes of Use	Rate
0 - 25,000	\$.1100
25,001 - 50,000	\$.1075
50,001 -200,000	\$.1000
200,001 -300,000	\$.0975
300,001 +	\$.0950

Note: Per minute rates are subject to a surcharge of \$.15 per minute if interstate day minutes of use exceeds 50% of total interstate minutes of use during the billing period.

#### 4.11 **ExactCall**

### 4.11.1 Usage Charges

#### A. Per Minute Rates:

<u>Peak</u>	Off-Peak
\$.1800 \$.1300	\$.1440 \$.1170
ψ. 1000	ψ.1170
\$.2225	\$.1780
\$.1470	\$.1176
\$.2500	\$.2500
\$.2475	\$.1400
	\$.1800 \$.1300 \$.2225 \$.1470 \$.2500

#### B. Rate Periods

The time of day rate periods for ExactCall will be Peak and Off-Peak. Peak rates apply to all calls that occur between 8AM and 5PM Monday through Friday, except on Companyrecognized holidays. Off-Peak rates apply to all other calls.

#### C. Billing Increments

For ExactCall PLUS, ExactCall WATS and ExactCall 800 service options, calls are billed in six second increments. For ExactCall Card service option, calls are billed in six second increments subject to a one minute per call minimum. For ExactCall MTS service option, calls are billed in one minute increments.

#### D. Volume Discounts

The following option specific volume discounts apply to each service option based exclusively on total ExactCall usage for a billing period.

<u>Usage</u>	<b>PLUS</b>	<u>WATS</u>	800 SHARED	800 WATS	CARD	MTS
\$0 - \$49	0%	0%	0%	0%	0%	0%
\$50 - \$249	0%	0%	0%	0%	2%	0%
\$250 - \$499	7%	0%	0%	0%	2%	0%
\$500 - \$999	7%	0%	7%	0%	5%	0%
\$1,000 - \$1,499	7%	0%	9%	0%	7%	0%
\$1,500 - \$1,999	9%	0%	9%	0%	7%	0%
\$2,000 - \$3,999	9%	0%	11%	0%	7%	0%
\$4,000 - \$4,999	12%	10%	11%	5%	7%	0%
\$5,000 - \$7,499	12%	10%	11%	5%	10%	0%
\$7,500 - \$9,999	12%	13%	11%	8%	10%	0%
\$10,000 - \$14,999	14%	16%	11%	10%	12%	0%
\$15,000 - \$19,999	14%	18%	11%	12%	13%	0%
\$20,000 - \$24,999	14%	18%	11%	12%	13%	0%
\$25,000+	14%	23%	11%	15%	13%	0%

<sup>1/</sup> A surcharge of \$.25 applies to each call.

#### 4.13 Performance 4000 Service

### 4.13.1 Usage Charges

#### A. Rate Periods

The time of day rate periods for Performance 4000 services will be Peak and Off-Peak. Peak rates apply to all calls that occur from 7AM to 7PM Monday through Friday, except on Company-recognized holidays. Off-Peak rates apply to all other calls.

#### B. Billing Increments

Performance 4000 Services are billed in six (6) second initial increments and are rounded to the next higher six (6) second increment. A minimum average time requirement (MATR) of thirty (30) seconds per call applies during a billing period for switched outbound and calling card. Performance 4000 Calling Card is billed at an initial sixty (60) second increment and rounded to the next higher sixty (60) second increment thereafter.

#### C. Per Minute Rate Schedules

Performance 4000 Level I and II Customers will receive the following rates:

#### 1. InterLATA Rate Schedules

#### Switched Per Minute Rates 1/ (a)

	Month to	12 Month	24/36 Month
	<u>Month</u>	<u>Term</u>	<u>Term</u>
Outbound	\$.1516	\$.1467	\$.1419
Inbound	\$.1615	\$.1467	\$.1419

EFFECTIVE: November 14, 2008 ISSUED: November 13, 2008

Performance 4000 Levels III, IV and V Customers will receive a 5% discount off the rates listed in (1) and 1/ (2) above.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 4.13 <u>Performance 4000 Service</u> (Continued)
  - 4.13.1 <u>Usage Charges</u> (Continued)
    - C. <u>Per Minute Rate Schedules</u> (Continued)
      - 1. <u>InterLATA Rate Schedules</u> (Continued)
        - (b) <u>Dedicated Per Minute Rates</u> 1/

	Month to	12 Month	24/36 Month
	<u>Month</u>	Term	<u>Term</u>
Outbound	\$.1075	\$.1027	\$.0978
Inbound	\$.1174	\$.1027	\$.0978

- 2. <u>IntraLATA Rate Schedules</u> (BellSouth areas only)
  - (a) Switched Per Minute Rates 1/

	Month to	12 Month	24/36 Month
	<u>Month</u>	<u>Term</u>	<u>Term</u>
Outbound	\$.1180	\$.1136	\$.1093
Inbound	\$.1443	\$.1311	\$.1268

(b) <u>Dedicated Per Minute Rates</u> 1/

	Month to	12 Month	24/36 Month
	<u>Month</u>	<u>Term</u>	<u>Term</u>
Outbound	\$.0998	\$.0950	\$.0903
Inbound	\$.1140	\$.0998	\$.0950

ISSUED: November 13, 2008

<sup>1/</sup> Performance 4000 Levels III, IV and V Customers will receive a 5% discount off the rates listed in (1) and (2) above.

# 4.13 <u>Performance 4000 Service</u> (Continued)

### 4.13.1 <u>Usage Charges</u> (Continued)

### D. Calling Card Service

### 1. <u>Direct Dial Rate Schedule</u>

(a) Per Minute Rates applying to all time periods and Levels:

Month to Month Rate: \$.2500 1, 2 or 3 Year Term Plan Rate: \$.2500

(b) Per Call Surcharge

A \$0.25 surcharge applies to each call.

(c) <u>Billing Increments</u> - Usage is billed at an initial sixty (60) second increment and rounded to the next higher sixty (60) second increment thereafter.

### E. Monthly Recurring Charges

Performance 4000 service is an interstate service offering with customers having the option of using the service for intrastate calls. Accordingly, Monthly Recurring Charges apply as described in Performance 4000 interstate service in Section 7.8.2.E of the Company's Tariff FCC No. 2.

# F. <u>Non-Recurring Charges</u>

Non-Recurring Charges apply as described in Performance 4000 interstate service in Section 7.8.2.F of the Company's Tariff FCC No. 2.

### 4.16 InteleNet Service

### 4.16.1 Rate Periods

<u>Peak</u> - All calls that occur between 7:00 A.M. through 7:00 P.M. Monday through Friday, except on Company recognized holidays.

Off Peak - All calls that occur between 7:01 P.M. through 6:59 A.M. Monday through Thursday, and all calls between 7:01 P.M. Friday and 6:59 A.M. Monday and Company-recognized holidays.

### 4.16.2 Billing Increments

InteleNet is billed in eighteen (18) second initial increments and is rounded to the next higher six (6) second increment with the exception of calling card which is billed in sixty (60) second initial increments and rounded to the next higher six (6) second increment. All fractional per call charges will be rounded to the nearest whole cent.

### 4.16.3 Per Minute Base Rate Schedule - InteleNet

Peak/Off Peak

Switched Services \$0.0981 Dedicated Services \$0.0747

- 4.16.4 <u>Discounts Applicable</u> A ten percent (10%) discount will apply towards intrastate usage.
- 4.16.5 InteleNet Calling Card Service

A. Per Minute Rates applying to all time periods: \$0.2600

B. Per Call Surcharge: \$0.4000

### 4.16.6 Monthly Recurring/Non-Recurring Charges

InteleNet for intrastate use is offered in connection with the Company's interstate InteleNet service. Accordingly, monthly recurring and non-recurring charges are found in the Company's interstate tariff, FCC No. 2, Section 6.3.1.

# 4.17 <u>Total Solution Gold Service</u>

### 4.17.1 Rate Periods

Peak and Off-Peak rate periods are as follows:

Non-Holiday		To But Not	
Rate Periods	<u>From</u>	Including	<u>Days</u>
Peak	8:00 a.m.	5:00 p.m.	Mon-Fri
Off-Peak	5:00 p.m.	8:00 a.m.	Mon-Fri
	8:00 a.m.	8:00 a.m.	Sat-Sun
	8:00 a.m.	8:00 a.m.	Holidays

Holidays: On Christmas Day (Dec. 25), New Years Day (Jan. 1), Memorial Day, Independence Day (July 4), Labor Day (first Monday in Sept.) and Thanksgiving Day (fourth Thursday in Nov.) the Off-Peak Period rate applies unless a lower rate would normally apply.

### 4.17.2 Usage Charges

### A. Outbound Calling Service

### 1. Shared - Per Minute Rates

Usage Per Month	<u>1 Year</u>	2 Year
\$0 - \$249	\$0.16 <del>7</del> 9	\$0.1623
\$250 - \$999	\$0.1679	\$0.1623
\$1,000 - \$2,499	\$0.1679	\$0.1623
\$2,500 - \$4,999	\$0.1679	\$0.1623
\$5,000 - \$9,999	\$0.1679	\$0.1623
\$10,000 - \$19,999	\$0.1679	\$0.1623
\$20,000 +	\$0.1679	\$0.1623

### 2. <u>Dedicated - Per Minute Rates</u>

Usage Per Month	<u>1 Year</u>	2 Year
\$0 - \$249	\$0.1175	\$0.1119
\$250 - \$999	\$0.1175	\$0.1119
\$1,000 - \$2,499	\$0.1175	\$0.1119
\$2,500 - \$4,999	\$0.1175	\$0.1119
\$5,000 - \$9,999	\$0.1175	\$0.1119
\$10,000 - \$19,999	\$0.1175	\$0.1119
\$20,000 +	\$0.1175	\$0.1119

### B. Inbound Calling Service

# 1. <u>Shared - Per Minute Rates</u>

Usage Per Month	<u>1 Year</u>	2 Year
\$0 - \$249	\$0.1679	\$0.1623
\$250 - \$999	\$0.1679	\$0.1623
\$1,000 - \$2,499	\$0.1679	\$0.1623
\$2,500 - \$4,999	\$0.1679	\$0.1623
\$5,000 - \$9,999	\$0.1679	\$0.1623
\$10,000 - \$19,999	\$0.1679	\$0.1623
\$20,000 +	\$0.1679	\$0.1623

### 4.17 <u>Total Solution Gold Service</u> (Continued)

### 4.17.2 <u>Usage Charges</u> (Continued)

### B. <u>Inbound Calling Service</u> (Continued)

### 2. <u>Dedicated - Per Minute Rates</u>

Usage Per Month	<u>1 Year</u>	2 Year
\$0 - \$249	\$0.1175	\$0.1119
\$250 - \$999	\$0.1175	\$0.1119
\$1,000 - \$2,499	\$0.1175	\$0.1119
\$2,500 - \$4,999	\$0.1175	\$0.1119
\$5,000 - \$9,999	\$0.1175	\$0.1119
\$10,000 - \$19,999	\$0.1175	\$0.1119
\$20,000 +	\$0.1175	\$0.1119

### 4.18 Total Solution Series 100 Service

### 4.18.1 Rate Periods

Peak and Off-Peak rate periods are as follows:

Non-Holiday		To But Not	
Rate Periods	<u>From</u>	Including	<u>Days</u>
Peak	8:00 a.m.	5:00 p.m.	Mon-Fri
Off-Peak	5:00 p.m.	8:00 a.m.	Mon-Fri
	8:00 a.m.	8:00 a.m.	Sat-Sun
	8:00 a.m.	8:00 a.m.	Holidays

<u>Holidays</u>: On Christmas Day (Dec. 25), New Years Day (Jan. 1), Memorial Day, Independence Day (July 4), Labor Day (first Monday in Sept.) and Thanksgiving Day (fourth Thursday in Nov.) the Off-Peak Period rate applies unless a lower rate would normally apply.

### 4.18.2 <u>Usage Charges</u>

A. Outbound Calling Service

Base Rate Per Minute

Shared \$ 0.1768 Dedicated \$ 0.1231

B. <u>Inbound Calling Service</u>

Shared \$ 0.1880 Dedicated \$ 0.1343

C. <u>Toll-Free Number Charge</u>

	<u>Monthly Recurring</u>	Non-Recurring
Shared charge per toll free #	\$10.00	\$15.00
Dedicated charge, per routing arrange	\$40.00	\$50.00

Shared Outhound Discount

# 4.18 <u>Total Solution Series 100 Service</u> (Continued)

# 4.18.3 Discounts

The following discounts on per minute base rates for qualifying usage are based on the monthly commitment level and Term Plan selected by the Customer. Customers committing to Term Plans will receive an additional discount off the applicable Commitment Level discounted rate for qualifying interstate usage. To qualify for Term Plan Discounts, shared access Customers must select a minimum commitment level of at least \$250.00 per month, and dedicated access Customers must select a minimum commitment level of at least \$1,000.00 per month.

Α.	Shared Outbound Disc	<u>count</u>		
	Monthly Commitment	Month to	Term Pla	an
	Level	Month_	<u>1 Year</u>	2 Year
	\$0	0.00%	N/A	N/A
	\$250	0.00%	5.00%	10.00%
	\$500	0.00%	5.00%	10.00%
	\$1,000	0.00%	5.00%	10.00%
	\$2,500	0.00%	5.00%	10.00%
	\$5,000	0.00%	5.00%	10.00%
	\$10,000	0.00%	5.00%	10.00%
	\$15,000	0.00%	5.00%	10.00%
B.	<b>Dedicated Outbound D</b>	<u>viscount</u>		
	Monthly Commitment	Month to	Term Pla	an
	Level	<u>Month</u>	<u>1 Year</u>	2 Year
	\$0	0.00%	N/A	N/A
	\$250	0.00%	N/A	N/A
	\$500	0.00%	N/A	N/A
	\$1,000	0.00%	5.00%	10.00%
	\$2,500	0.00%	5.00%	10.00%
	\$5,000	0.00%	5.00%	10.00%
	\$10,000	0.00%	5.00%	10.00%
	\$15,000	0.00%	5.00%	10.00%
C.	Shared Inbound Discor			
	Monthly Commitment	Month to	Term Pla	
	Level	<u>Month</u>	<u>1 Year</u>	2 Year
	\$0	0.00%	N/A	N/A
	\$250	0.00%	10.65%	15.36%
	\$500	0.00%	10.65%	15.36%
	\$1,000	0.00%	10.65%	15.36%
	\$2,500	0.00%	10.65%	15.36%
	\$5,000	0 00%	10.65%	15.36%
	\$10,000	0.00%	10.65%	15.36%
_	\$15,000	0.00%	10.65%	15.36%
D.	Dedicated Inbound Dis			
	Monthly Commitment	Month to	Term Pla	
	Level	Month	1 Year	2 Year
	\$0	0.00%	N/A	N/A
	\$250	0.00%	N/A	N/A
	\$500	0.00%	N/A	N/A
	\$1,000	0.00%	12.92%	17.50%
	\$2,500	0.00%	12.92%	17.50%
	\$5,000	0.00%	12.92%	17.50%
	\$10,000	0.00%	12.92%	17.50%
	\$15,000	0.00%	12.92%	17.50%

### 4.19 MFS Inteleplan

### 4.19.1 Rate Periods

Peak and Off-Peak rate periods are as follows:

	To But Not	
<u>From</u>	<u>Including</u>	<u>Days</u>
7:00 a.m.	7:00 p.m.	Mon-Fri
7:00 p.m.	7:00 a.m.	Mon-Fri
All Hours		Sat-Sun
All Hours		Holidays
	7:00 a.m. 7:00 p.m. All Hours	From Including 7:00 a.m. 7:00 p.m. 7:00 p.m. 7:00 a.m. All Hours

<u>Holidays</u>: On Christmas Day (Dec. 25), New Years Day (Jan. 1), Memorial Day, Independence Day (July 4), Labor Day (first Monday in Sept.) and Thanksgiving Day (fourth Thursday in Nov.) the Off-Peak Period rate applies unless a lower rate would normally apply.

# A. Per Minute Rates

<u>Peak</u> <u>Off-Peak</u> \$ 0.2300 \$ 0.1100

B. <u>Surcharge</u>

Per Minute \$ 0.0200

C. Monthly Recurring \$3.00/Per Account

### 4.20 MFS Intelenet Calling Card Service

### 4.20.1 Rate Periods

Peak and Off-Peak rate periods are as follows:

Non-Holiday		To But Not	
Rate Periods	<u>From</u>	Including	<u>Days</u>
Peak	8:00 a.m.	5:00 p.m.	Mon-Fri
Off-Peak	5:00 p.m.	8:00 a.m.	Mon-Fri
	8:00 a.m.	8:00 a.m.	Sat-Sun
	8:00 a.m.	8:00 a.m.	Holidays

<u>Holidays</u>: On Christmas Day (Dec. 25), New Years Day (Jan. 1), Memorial Day, Independence Day (July 4), Labor Day (first Monday in Sept.) and Thanksgiving Day (fourth Thursday in Nov.) the Off-Peak Period rate applies unless a lower rate would normally apply.

\$0.25

# 4.20.2 <u>Usage Charges</u>

A.	Standard	Usage	Charges
----	----------	-------	---------

Per minute of use Peak Rate Off-Peak Discount

Off-Peak Discount 0%
Per Call Surcharge \$0.40

### B. Operator Assisted Calls

Person-to-Person surcharge \$3.50 Station-to-Station surcharge \$0.60

### C. Directory Assistance

Requested Number Charge \$1.40 Call Completion Charge \$0.50

### D. Enhanced Feature Charges

### Conference Calling

Per established line	\$2.00
Per minute of usage per line	\$0.26
Voice Mail Access per minute of usage	\$0.26
Voice Messaging per call	\$1.50
News and Information per minute of usage	\$0.28

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 4.23 MCI WorldCom On-Net Service Voice 1/

### 4.23.1 Usage Rates

- A. <u>Billing Increments</u> Except as otherwise specified, all calls are subject to an 18-second minimum initial period and rounding to the next higher 6-second increment, except for Operator Assisted calls, which are subject to a 60-second initial period and rounding to the next 60-second increment. If the computed charge includes a fraction of a cent, the fraction is rounded to the nearest whole cent.
- B. <u>Per Minute Rates</u> The following per-minute usage rates will apply to outbound and inbound usage. Outbound usage charges are based on origination type, and inbound usage charges are based on termination type. Card is not available for inbound usage.

<u>Outbound Rates:</u> The following per-minute rates will apply. Usage charges are based on origination type.

Origination <u>Type</u>	Termination <u>Type</u>	Per-Minute <u>Rate</u>
Local Network Connection 2/	Local Network Connection	\$0.0396
Local Network Connection	Dedicated	0.0396
Local Network Connection	Switched	0.0496
Dedicated	Local Network Connection	0.0423
Dedicated	Dedicated	0.0423
Dedicated	Switched	0.0523
Switched	Local Network Connection	0.0869
Card	Local Network Connection	0.0969
Switched/Card	Dedicated	0.0869
Switched/Card	Switched	0.0969

ISSUED: November 13, 2008

<sup>1/</sup> Beginning January 1, 2004, this service will no longer be available to new subscribers.

<sup>2/</sup> Local Network Connection rates will apply to all calls with local origination.

### 4.23 MCI WorldCom On-Net Service - Voice

### 4.23.1 Usage Rates

<u>Inbound Rates:</u> The following per-minute rates will apply. Usage charges are based on termination type.

OriginationType	Termination <u>Type</u>	Per-Minute <u>Rate</u>
Local Network Connection	Local Network Connection	\$0.0396
Local Network Connection	Dedicated	\$0.0423
Local Network Connection	Switched	0.0869
Switched	Local Network Connection	0.0496
Switched	Dedicated	0.0523
Switched	Switched	0.0969

For outbound calls with Local termination or inbound calls with Local origination, a credit of \$0.01 per minute will apply. Dedicated origination is not available for inbound calls.

C. <u>Directory Assistance</u> - A charge of \$1.40 per call will be applied to all customers of this Service requesting Directory Assistance for numbers within the state.

### 4.23.2 Term Commitment/Discounts

The following discounts will apply to the eligible intrastate charges based on the customer's term commitment.

Term Commitment	Discount
1 year	3%
2 year	6%
3 year	9%
4 year	9%

### 4.23 MCI WorldCom On-Net Service - Voice

### 4.23.3 Non-Recurring Charges:

Installation

The following non-recurring charges will apply:

Per Order Per Location \$50 \$25

The Rates and Charges can be found in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="https://www.verizonbusiness.com">www.verizonbusiness.com</a> (Companion Interstate Service).

### 4.23.4 Monthly Recurring Charges:

The following monthly recurring charges will apply:

- A. <u>Shared Access Charge:</u> A \$5.00 monthly minimum charge will apply to each customer location using switched access. The Customer will be charged the difference between the Customer's actual usage charges (including surcharges) and \$5.00 per location during each monthly billing period.
- B. <u>Service Fee:</u> The following monthly service fee per service group per toll-free number will apply:

Business Line Terminations: \$20.00 Dedicated Terminations: \$50.00

### 4.23.5 Card Surcharge

The per-call surcharge as located in the MCI Communications Services, Inc. d/b/a Verizon Business Services, FL PSC Tariff No. 3, Section 4.1.33 will apply to each Card call, other than calls to Directory Assistance:

# 4.23.5 LD and Local Online Calling Plan

Monthly Recurring Charges:

A monthly recurring charge will apply for each Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Local Service. Offerings A and B are available on a per-Local Line basis and Offerings C and D are on a per-Local Trunk basis. The following Monthly recurring charges apply:

Offering	Monthly Recurring Charge
per line or per trunk)	
A	\$39.00
В	\$65.00
C	\$59.00
D	\$99.00

### 4.23 MCI WorldCom On-Net Service - Voice

### 4.23.6 On-Net Plus Plan

### **Usage Charges:**

The following per-minute usage rates will apply to Intrastate Usage. Customers will be charged the following per minute rates for Inbound and Outbound Intrastate Service.

### Outbound:

Outbouria.			
		1 Year	2 Year
Origination Type	Termination Type	Per Minute Rate	Per Minute Rate
Local Network Connection	Local Network Connection	\$0.0000	\$0.0000
Local Network Connection	Dedicated	\$0.0447	\$0.0423
Local Network Connection	Switched	\$0.0447	\$0.0423
Dedicated	Local Network Connection	\$0.0470	\$0.0445
Dedicated	Dedicated/Switched	\$0.0470	\$0.0445
Switched/Card	Local Network Connection	\$0.0891	\$0.0851
Switched/Card	Dedicated	\$0.0891	\$0.0851
Switched/Card	Switched	\$0.0891	\$0.0851
Inbound:			
Local Network Connection	Local Network Connection	\$0.0447	\$0.0423
Local Network Connection	Dedicated	\$0.0470	\$0.0445
Local Network Connection	Switched	\$0.0891	\$0.0851
Switched/Card	Local Network Connection	\$0.0447	\$0.0423
Switched/Card	Dedicated	\$0.0470	\$0.0445
Switched/Card	Switched	\$0.0891	\$0.0851

### 4.23.7 Agent Program

Per Minute Usage Rate: \$0.0690

### 4.23.8 Affinity 1 Program

Customers will be charged the following per minute rates for Eligible inbound and outbound Intrastate Service:

Switched \$0.110

### 4.23.9 Affinity 2 Program

Customers will be charged the following per minute rates for Eligible inbound and outbound Intrastate Service:

Switched \$0.110

# 4.23 MCI WorldCom On-Net Service - Voice

# 4.23.10 Intrastate Plus Service

The following per-minute usage rates will apply to Intrastate Usage. Customers will be charged the following per minute rates for Inbound and Outbound Intrastate Service.

ORIGINATION TYPE: OUTBOUND	TERMINATION TYPE: OUTBOUND	1 YEAR TERM	2 YEAR TER	M 3 YEAR TERM
LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION	\$0.0000	\$0.0000	\$0.0000
LOCAL NETWORK CONNECTION	DEDICATED	\$0.0536	\$0.0521	\$0.0505
LOCAL NETWORK CONNECTION	SWITCHED	\$0.0536	\$0.0521	\$0.0505
DEDICATED	LOCAL NETWORK CONNECTION	\$0.0536	\$0.0521	\$0.0505
DEDICATED	DEDICATED OR SWITCHED	\$0.0536	\$0.0521	\$0.0505
SWITCHED / CARD	LOCAL NETWORK CONNECTION	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	DEDICATED	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	SWITCHED	\$0.0917	\$0.0890	\$0.0863
ORIGINATION TYPE: INBOUND	TERMINATION TYPE: INBOUND	1 YEAR TERM	2 YEAR TER	M 3 YEAR TERM
LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION	\$0.0536	\$0.0521	\$0.0505
LOCAL NETWORK CONNECTION	DEDICATED	\$0.0536	\$0.0521	\$0.0505
LOCAL NETWORK CONNECTION	SWITCHED	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	LOCAL NETWORK CONNECTION	\$0.0536	\$0.0521	\$0.0505
SWITCHED / CARD	DEDICATED	\$0.0536	\$0.0521	\$0.0505
SWITCHED / CARD	SWITCHED	\$0.0917	\$0.0890	\$0.0863

# 4.23.10.1 <u>Intrastate Plus Optional Calling Plan I</u> The following per minute rates apply:

ORIGINATION TYPE: OUTBOUND	TERMINATION TYPE: OUTBOUND	1 YEAR TERM	2 YEAR	TERM 3 YEAR TERM
LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION	\$0.0000	\$0.0000	\$0.0000
LOCAL NETWORK CONNECTION	DEDICATED	\$0.0505	\$0.0505	\$0.0505
LOCAL NETWORK CONNECTION	SWITCHED	\$0.0505	\$0.0505	\$0.0505
DEDICATED	LOCAL NETWORK CONNECTION	\$0.0505	\$0.0505	\$0.0505
DEDICATED	DEDICATED OR SWITCHED	\$0.0505	\$0.0505	\$0.0505
SWITCHED / CARD	LOCAL NETWORK CONNECTION	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	DEDICATED	\$0.0917	\$0.0890	\$0.0863
SWITCHED / CARD	SWITCHED	\$0.0917	\$0.0890	\$0.0863
ORIGINATION TYPE: OUTBOUND	TERMINATION TYPE: OUTBOUND	1 YEAR TERM	2 YEAR	TERM 3 YEAR TERM
ORIGINATION TYPE: OUTBOUND LOCAL NETWORK CONNECTION	TERMINATION TYPE: OUTBOUND LOCAL NETWORK CONNECTION	1 YEAR TERM \$0.0505	2 YEAR 5	TERM 3 YEAR TERM \$0.0505
LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION	\$0.0505	\$0.0505	\$0.0505
LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION DEDICATED	\$0.0505 \$0.0505	\$0.0505 \$0.0505	\$0.0505 \$0.0505
LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION	LOCAL NETWORK CONNECTION DEDICATED SWITCHED	\$0.0505 \$0.0505 \$0.0917	\$0.0505 \$0.0505 \$0.0890	\$0.0505 \$0.0505 \$0.0863
LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION LOCAL NETWORK CONNECTION SWITCHED / CARD	LOCAL NETWORK CONNECTION DEDICATED SWITCHED LOCAL NETWORK CONNECTION	\$0.0505 \$0.0505 \$0.0917 \$0.0505	\$0.0505 \$0.0505 \$0.0890 \$0.0505	\$0.0505 \$0.0505 \$0.0863 \$0.0505

### 4.24 Optional Features

### 4.24.1 Accounting Codes

No charge

### 4.24.2 Abbreviated Dialing

There is a non-recurring charge of \$50 for each request for Abbreviated Dialing Codes. A maximum of 25 codes will be provided for each request (i.e., if the customer requests 28 than the non-recurring charge will apply twice).

### 4.24.3 Project Billing

\$50 per access line group per request to install or modify.

### 4.24.4 Call Records on Magnetic Tape

Monthly Recurring Charge: \$100 per Magnetic Tape Installation Charge: \$500 per account

### 4.24.5 1-Plus Screening

Installation Charge: \$50 per access line group 1/

# 4.24.6 Area Code Screening

\$100 per request to install or modify.

### 4.24.7 Verified Account Codes

There is a monthly recurring charge of \$25 per order of verified account codes. A maximum of 25 codes will be issued per order. There is an installation charge of \$200 per order.

### 4.24 Optional Features (Continued)

### 4.24.8 Prefix NPA Addition

Installation Charges: \$100 per order

# 4.24.9 Autodialer Features

No charge

 $\underline{1}$ / The installation charge will not apply if the 1-Plus Screening option is ordered prior to installation.

### 4.27 MCI WorldCom Business Solutions

### 4.27.1 Usage Rates

4.27.1.1 Outbound/Inbound Rates: The following per-minute rate will apply: \$0.1050

Calling Card: Card Access may originate from any touch-one phone in the United States via a toll free number. Usage charges as indicated in Section 4.27.1.1 will apply. A \$0.50 per-call surcharge will also apply.

- 4.27.1.3 Operator Assisted Calling: The charges found in section 4.6.5 apply to outbound Operator Assisted calls (excluding collect calls and calls which are billed by a third party) made by customers using switched (including card access) access.
- 4.27.1.4 Directory Assistance: an undiscountable charge per call will be applied to each call requesting Directory Assistance, subject to the rate and provisions set forth in Section 4.6.6.
- 4.28 Agency Program A
- 4.28.1 Usage Rates
- 1+, Dedicated Outbound and Inbound Service: Customers will be charged the following per minute rates for 1+, dedicated outbound, and inbound intrastate/intralata calls.

Switched Dedicated

Outbound/Inbound \$0.0950 \$0.0579

Calling Card: Customers will be charged \$0.0950 per minute for calling card calls. A per call surcharge of \$0.75 will apply to all calls.

Calls are billed in 6 second increments with a 60 second minimum.

### 4.30 <u>D Street Plus Voice Service Calling Plan</u>

### 4.30.1 Usage Rates

Customers will be charged the following per minute rates for Eligible inbound and outbound Intrastate Service.

Switched \$0.0743 Dedicated \$0.0301

 $\underline{1}$ / Beginning November 1, 2002, this service will no longer be available to new subscribers.

ISSUED: November 13, 2008 EFFECTI

### 4.35 CALLING PLANS

# 1. Verizon Loyalty Plus I Plan 1/

Offer: Existing Company customers who simultaneously order a new eligible MCI Legacy Company service "unit" (see table below), excluding upgrades and conversions on existing service and enroll that unit in this plan will receive the following benefit: a one-time credit in the amount indicated in the table below, applied to the monthly recurring charges for the new MCI Legacy Company service unit. The credit will be applied to the second full-month's invoice following activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer's Fund account. To receive the benefits of this plan, each eligible service unit must be active, with no pending cancellation request, at the time the credit is applied. Except for this one-time credit, this plan does not change the terms and conditions that apply to the new service unit.

Customers will also receive a waiver of the non-recurring Company-billed installation charges for installation following plan enrollment of all eligible service units identified below. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

This plan applies only with respect to new eligible MCI Legacy Company provided service units ordered for installation within 30 days of order (or by Company's quoted installation date if later), by a Customer with at least one MCI Legacy Company service which has been actively billing for at least 90 days prior to enrollment in this plan.

Customer may enroll any eligible service unit, but any individual unit may be enrolled only once. Customer may not receive: (i) any discounts, (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Special Customer Arrangements (SCA) Guide Types 2, 3,4,5,6,7,8 or 9 or Produce Package SCA Guide Types 13, 14, 15, 16, 18, 19, 20 and 21, or (iii) the benefits of Verizon Business Services I Flex T1 Promotion, Regional Plus Frame Relay Promotion and Competitive Voice 2 Promotion as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="www.verizonbusiness.com">www.verizonbusiness.com</a> on any enrolled unit.

### TABLE OF ELIGIBLE COMPANY SERVICE UNITS AND CREDIT AMOUNTS

Product Group	Sub-product	Service Unit	Credit Value	Company Installation Charge Waiver (Yes/No)
Voice				
	Local and Long Distance Service-Trunk Solution	Trunk	\$50	Υ

<sup>1/</sup>Beginning July 1, 2007, this service will no longer be available to new subscribers.

# 4.35 CALLING PLANS (CONT.)

### Verizon Loyalty Plus II Plan 1/

Offer: Existing Company customers who simultaneously order a new eligible MCI Legacy Company service "unit" (see table below), excluding upgrades and conversions on existing service and enroll that unit in this plan will receive the following benefit: two credits in the amount indicated in the table below, applied to the monthly recurring charges for the new MCI Legacy Company service unit. The credits will be applied to the second full-month's and the sixth full-month's invoice following activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer's Fund account. To receive the benefits of this plan, each eligible service unit must be active, with no pending cancellation request, at the time the credits are applied. Except for these two credits, this plan does not change the terms and conditions that apply to the new service unit.

Customers will also receive a waiver of the non-recurring Company-billed installation charges for installation following plan enrollment of all eligible service units identified below. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

This plan applies only with respect to new eligible MCI Legacy Company provided service units ordered for installation within 30 days of order (or by Company's quoted installation date if later), by a Customer with at least one MCI Legacy Company service which has been actively billing for at least 90 days prior to enrollment in this plan.

Customer may enroll any eligible service unit, but any individual unit may be enrolled only once. Customer may not receive: (i) any discounts, (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Special Customer Arrangements (SCA) Guide Types 2, 3,4,5,6,7,8 or 9 or Produce Package SCA Guide Types 13, 14, 15, 16, 18, 19, 20 and 21, or (iii) the benefits of Verizon Business Services I Flex T1 Promotion, Regional Plus Frame Relay Promotion and Competitive Voice 2 Promotion as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="www.verizonbusiness.com">www.verizonbusiness.com</a> on any enrolled unit.

TABLE OF FLIGIBLE COMPANY SERVICE UNITS AND CREDIT AMOUNTS

Product Group	Sub-product	Service Unit	Credit Value	Company Installation Charge Waiver (Yes/No)
Voice				
	Local and Long Distance Service-Trunk Solution	Trunk	\$50	Y

<sup>1/</sup> Beginning February 1, 2007, this service will no longer be available to new subscribers.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 4.35 CALLING PLANS (CONT.)

# Verizon Loyalty Plus III Plan 1/

Offer: Existing Company customers who simultaneously order a new eligible MCI Legacy Company service "unit" (see table below), excluding upgrades and conversions on existing service and enroll that unit in this plan will receive the following benefit: three credits in the amount indicated in the table below, applied to the monthly recurring charges for the new MCI Legacy Company-provided service unit. The credits will be applied to the second full-month's, sixth full-month's, and twelfth full-month's invoice following activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer's Fund account. To receive the benefits of this plan, each eligible service unit must be active, with no pending cancellation request, at the time the credits are applied. Except for these three credits, this plan does not change the terms and conditions that apply to the new service unit.

Customers will also receive a waiver of the non-recurring Company-billed installation charges for installation following plan enrollment of all eligible service units identified below. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

This plan applies only with respect to new eligible MCI Legacy Company provided service units ordered for installation within 30 days of order (or by Company's quoted installation date if later), by a Customer with at least one MCI Legacy Company service which has been actively billing for at least 90 days prior to enrollment in this plan.

Customer may enroll any eligible service unit, but any individual unit may be enrolled only once. Customer may not receive: (i) any discounts, (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Special Customer Arrangements (SCA) Guide Types 2, 3,4,5,6,7,8 or 9 or Produce Package SCA Guide Types 13, 14, 15, 16, 18, 19, 20 and 21, or (iii) the benefits of Verizon Business Services I Flex T1 Promotion, Regional Plus Frame Relay Promotion and Competitive Voice 2 Promotion as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <a href="www.verizonbusiness.com">www.verizonbusiness.com</a> on any enrolled unit.

# TABLE OF ELIGIBLE COMPANY SERVICE UNITS AND CREDIT AMOUNTS

Product Group	Sub-product	Service Unit	Credit Value	Company Installation Charge Waiver (Yes/No)
Voice				
	Local and Long Distance Service-Trunk Solution	Trunk	\$50	Y

<sup>1/</sup> Beginning September 1, 2006, this service will no longer be available to new subscribers.

### 5.1 <u>Promotional Offerings and Rates (Continued)</u>

### 5.1.25 Long Distance Voice – Intrastate Assistance Promotion

Subject to the Conditions of Eligibility and Other Conditions below, a new customer ("Customer") of Verizon Long Distance Voice Services ordering Domestic Outbound Intrastate service ("Promotional Service") originating in the state will receive the following promotional rates as shown below for each eligible circuit ("Promotional Circuit"), as applicable, for the Term of the Agreement:

1-Year Term		2-Year Term		3-Year Term	
Call	Call Type		Call Type		Туре
Orig/Term Orig/Term Orig/Term Orig/Term		Orig/Term	Orig/Term	Orig/Term	
Dedicated/	Switched/	Dedicated/	Switched/	Dedicated/	Switched/
Switched*	Switched	Switched*	Switched	Switched*	Switched
\$0.0360	\$0.0600	\$0.0350	\$0.0589	\$0.0345	\$0.0586

### Conditions Of Eligibility

Dedicated/Switched origination/termination rates listed above also apply to the following call type combinations: Dedicated/Dedicated; Switched/Dedicated; Local/Local; Local/Dedicated; Dedicated/Local; Switched/Local and Local/Switched.

### Conditions of Eligibility

- 1 Verizon's standard expedite fees will apply to expedited orders for Promotional Service.
- 2. Customer must enroll between October 11, 2007 and June 30, 2008.
- 3. Customer must execute and submit an agreement with Verizon by July 31, 2008 that includes the Promotional Service.

### Other Conditions

- 1. Promotional Circuits receiving the benefits of this promotion may not receive the benefits of the following: LD Voice 300, 500 or 800 Minute Packages for Business Lines & Trunks; LD Voice Package.
- 2. Unless stated otherwise, the promotional rates/discounts set forth herein are in lieu of all other discounts.

### SECTION 6 - OBSOLETE SERVICES

Service offerings listed herein are classified as obsolete. Obsolete services are furnished subject to all the Rules and Regulations of the tariff the same as would be applicable if the service offerings were not obsolete. No new installations for these services will be made. Existing customers may remain with the service unless the Company provides thirty (30) days written notice that the service will be discontinued. Obsolete services with no subscribed customers will be removed from the tariff with prior Commission approval.

### 6.1.6 Standard PLUS WATS

### A. <u>Description</u>

Standard PLUS is an outbound WATS-like product available in both equal and non-equal access areas allowing the customer calling capabilities via customer-provided business lines. The product is recommended for small to medium sized customers.

# B. <u>Per Minute Rates</u>:

<u>Day</u>	<u>Evening</u>	Night/Weekend
\$.2150	\$.1650	\$.1290

- C. <u>Billing Increments</u> Calls are rated at an initial thirty (30) seconds and in six (6) second increments thereafter.
- D. <u>Monthly Volume Discounts</u> A volume discount is applied to the total usage portion of the monthly statement for each service category as specified below and is calculated on an incremental amount basis.

Total Monthly Usage	<u>Discount</u>
\$125 - \$1,000	10%
\$1,000 - \$2,000	15%
Over \$2,000	20%

### E. Non-Recurring Charges

1.	Account Set-Up Fee	\$ 0.00

2. Conversion Charge/Account (applies to customers converting from Basic Service) \$ 25.00

3. Magnetic Tape Billing \$200.00

4. Validated Acct Codes; setup \$ 20.00

### F. Monthly Recurring Charges

1.	Account Fee	\$ 0.00

2. Magnetic Tape Billing \$ 75.00

3. Validated Account Codes; \$ 20.00

#### MTS Service (Continued) 6.1

#### 6.1.7 **LaserPLUS**

#### A. Description

LaserPLUS is a two-way calling, multipoint service allowing the customer calling capabilities via customer-provided business lines. LaserPLUS is available both equal and non-equal Access Areas.

#### Per Minute Rates: B.

Per Minute of Use Charges

<u>Taper</u>	Day	Evening	Night/Weekend
0 - 10 hours	\$.2150	\$.1650	\$.1290
10.1 - 50 hours	\$.1935	\$.1485	\$.1161
50.1 - 150 hours	\$.1828	\$.1403	\$.1097
150 hours & over	\$.1720	\$.1320	\$.1032

C. Billing Increments - Calls are rated at an initial thirty (30) seconds and in six (6) second increments thereafter.

#### D. Non-Recurring Charges

6.

1.	Order Processing Charge (per acct)	\$ 10.00
2.	Conversion Charge (per account, for those custo	omers converting from MTS only) \$ 25.00
3.	Administrative (per order)	\$20.00
4.	Magnetic Tape call detail (per acct)	\$200.00
5.	Validated Account Codes (per acct)	\$ 20.00

\$ 10.00

ISSUED: November 13, 2008 EFFECTIVE: November 14, 2008

Non-Validated Account Codes (per acct)

# 6.1 <u>MTS Service</u> (Continued)

### 6.1.7 <u>LaserPLUS</u> (Continued)

### E. Monthly Recurring Charges

1.	Account Codes (per account)	\$ 10.00
2.	Validated Account Codes (per acct)	\$ 20.00
3.	Magnetic Tape call detail (per acct)	\$ 75.00
4.	Non-Validated Account Codes(per acct)	\$ 10.00

### 6.1.8 Advanced LaserPLUS

### A. Description

Advanced LaserPLUS is a two-way calling, multipoint service available in both equal and non-equal access areas allowing the customer calling capabilities via customer-provided business lines. The product is recommended for small to medium sized customers.

# B. Rate Schedule

### Per Minute of Use Charge

<u>Day</u>	<u>Evening</u>	Night/Weekend
\$0.1950	\$0.1600	\$0.1250

C. <u>Billing Increments</u> - Calls are rated at an initial thirty (30) seconds and in six (6) second increments thereafter.

# 6.1 MTS Service (Continued)

### 6.1.8 Advanced LaserPLUS (Continued)

Monthly Usage

D. <u>Monthly Volume Discount</u> - A volume discount is applied to the total usage portion of the monthly statement as specified below and is calculated on an incremental amount basis. Inter- Intrastate usage may be combined to reach usage level.

Discount

	\$150 - \$2,000	\$1,999.99 plus	5% 15%
E.	Non-re	curring Charges	
	1.	Set-up fee (per account	·)
	2.	Conversion Charge (per customers converting fr	

	customers converting nom with only)	Ψ25.00
3.	Conversion Charge (per account, for those customers converting from LaserPLUS only)	\$50.00
4.	Magnetic Tape	200.00 <u>1</u> /
5.	Validated Project Account Coding	\$20.00

<sup>6.</sup> Non-Validated Project Account Coding \$ 5.00

ISSUED: November 13, 2008

\$ 0.00

\$25.00

<sup>1/</sup> For Association Saver customers the non-recurring fee for Magnetic Tape Billing is \$100.00. The monthly recurring fee is \$50.00.

# 6.1 MTS Service (Continued)

# 6.1.8 <u>Advanced LaserPLUS</u> (Continued)

### F. Monthly Recurring Charges

1.	Account Charge	\$ 0.00
2.	Account Codes (per account)	\$ 5.00
3.	Magnetic Tape Billing	\$75.00 <u>1</u> /
4.	Validated Project Account Coding	\$20.00
5.	Non-Validated Project Accounting Coding	\$ 5.00

NOTE: The monthly account charge will be waived for each month in which usage exceeds \$100.00.

### 6.1.9 InWATS USA

### A. Per Minute Rate Schedules:

1. Step I: All originating points 0-30 miles from the Carrier's Switching Station.

<u>Day</u>	Evening	Night/Weekend
\$0.345	\$0.330	\$0.310

ISSUED: November 13, 2008

<sup>1/</sup> For Association Saver customers the non-recurring fee for Magnetic Tape Billing is \$100.00. The monthly recurring fee is \$50.00.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 6.1 MTS Service (Continued)
  - 6.1.9 InWATS USA (Continued)
    - A. Per Minute Rate Schedules (Continued)
      - 2. Step II: All originating points over 30 miles from the Carrier's Switching Station.

Night/Weekend Day Evening \$0.395 \$0.375 \$0.360

- Billing Increments Calls are timed and charged with the first full minute of each call В. being a full minute charge and additional time is billed in six (6) second increments thereafter.
- C. Monthly Recurring Service Fee: \$10.00
- 6.1.10 Band 0 WATS
  - Α. Per Minute Rate: \$0.1850
  - B. Billing Increments - Calls are timed and charged with the first full minute of each call being a full minute charge and additional time is billed in six (6) second increments thereafter.
  - C. Minimum Monthly Usage Charge: \$175.00
- 6.1.11 American Business Plan
  - A. Description

American Business Plan ("ABP") is WATS-like product available in both equal and non-equal access areas utilizing customer-provided business lines. ABP is a calling plan designed for small to medium sized commercial customers who are charged a flat per minute rate.

### 6.1 MTS Service (Continued)

### 6.1.11 American Business Plan (Continued)

B. <u>Per Minute Rate</u>: \$0.190

C. <u>Billing Increments</u> - Calls are rated at an initial thirty (30) seconds and in six (6) second increments thereafter.

### 6.1.12 UltraSaver PLUS

### A. <u>Description</u>

UltraSaver PLUS is a two-way calling, multipoint service allowing the customer calling capabilities via customer-provided business lines. Customers are charged a flat rate per minute with volume discounts at specified usage levels.

### B. <u>Per Minute Rates</u>:

Day	<u>Evening</u>	Night/Weekend		
\$0.1950	\$0.1800	\$0.1600		

- C. <u>Billing Increments</u> Calls are rated at an initial thirty (30) seconds and in six (6) second increments thereafter.
- D. <u>Monthly Volume Discount</u> A retroactive volume discount is applied to domestic Day usage over \$400 when the customer reaches the usage levels indicated below. Note: Usage in all time frames will be included to reach discount levels.

<u>\$ Volume</u>	<u>Discount Percentage</u>
\$400 - \$799.99	2%
\$800 - \$999.99	4%
\$1,000 - \$1,499.99	6%
\$1,500 - \$1,999.99	8%
\$2,000 and over	10%

# 6.1 MTS Service (Continued)

### 6.1.13 Concise WATS

### A. Description

Concise WATS is a one-way direct dial service utilizing 1+ access in equal access areas. This WATS-like product is available as a LEC billed service in areas where the Company has billing agreements with the LEC.

### B. <u>Per Minute Rates</u>:

<u>Peak</u> <u>Off-Peak</u> \$0.2025 \$0.1550

C. <u>Billing Increments</u> - Calls are rated at an initial thirty (30) seconds and in six (6) second increments thereafter.

### 6.1.14 Prestige I

# A. <u>Description</u>

Prestige I is a switched outbound service utilizing the customer's local access line. Customers are charged a flat rate per minute with volume discounts based on actual usage. Rates and volume discounts can be found in Section 4. All fractional per call charges will be rounded to the nearest whole cent.

B. Rate Schedule (Per Minute of Use)

<u>Peak</u> <u>Off Peak</u> \$0.2000 \$0.1600

C. <u>Billing Increments</u> - Calls are rated at an initial thirty (30) seconds and in six (6) second increments thereafter.

### 6.1 MTS Service (Continued)

### 6.1.14 Prestige I (Continued)

### D. Volume Discount

A retroactive volume discount is applied to the total Peak usage portion of the monthly statement as specified below. Total interstate, intrastate and international usage may be combined to reach volume discount levels.

Monthly Usage	Discount %		
\$0 - \$49.99	0%		
\$50 - \$149.99	1.0%		
\$150 - \$299.99	2.5%		
\$300 - \$499.99	3.5%		
\$500 +	5.0%		

# E. <u>Multi-Service Discount</u>

Customers subscribing to Prestige I in conjunction with Prestige II will receive an additional 2% discount for each month in which combined total monthly usage exceeds \$150.00. This discount is applied to Peak use only.

# 6.1.15 Treasure Services

### A. Description

Treasure Call/Treasure WATS is an outbound distance sensitive, time-of-day sensitive service offering. Service is provided to customers over Feature Group D facilities.

### 6.1 MTS Service (Continued)

### 6.1.14 Prestige I (Continued)

### B. Treasure Call

### 1. Per Minute Rate Schedule

	Busine	ess Day	Eve	ning	Night/W	/eekend
<u>Mileage</u>	<u>1st Min</u>	Addl Min	<u>1st Min</u>	Addl Min	1st Min	Addl Min
1- 10	\$0.1900	\$0.0900	\$0.1425	\$0.0675	\$0.0950	\$0.0450
11- 22	\$0.2600	\$0.1600	\$0.1950	\$0.1200	\$0.1350	\$0.0800
23- 55	\$0.2600	\$0.2190	\$0.1950	\$0.1643	\$0.1450	\$0.1150
56-124	\$0.2605	\$0.2205	\$0.1954	\$0.1654	\$0.1495	\$0.1190
125-292	\$0.2610	\$0.2270	\$0.1958	\$0.1703	\$0.1520	\$0.1240
293-430	\$0.2610	\$0.2300	\$0.1958	\$0.1725	\$0.1535	\$0.1240
431-624	\$0.2635	\$0.2380	\$0.1976	\$0.1785	\$0.1570	\$0.1290

2. <u>Billing Increments</u> - Calls are billed in one (1) minute increments.

### 3. Rebate Program

Each month customers are eligible to receive a rebate check equal to ten (10%) percent of their total previous month's paid monthly usage. For every twelvementh (12) period of continuous service, customers are eligible to receive an additional rebate check equal to five (5%) percent of the combined previous twelve-month total usage. This rebate is calculated on an annual basis.

# C. TreasureWATS

### 1. Per Minute Rate Schedule

Per minute rates apply as found in Section 4.1.14.A proceeding.

# 6.1 MTS Service (Continued)

### 6.1.14 Prestige I (Continued)

- C. <u>TreasureWATS</u> (Continued)
  - 2. <u>Billing Increments</u> Calls are rated at an initial thirty (30) seconds and in six (6) second increments thereafter.

### 3. Rebate Program

Each month customers are eligible to receive a rebate check equal to twelve (12%) percent of their total previous month's paid monthly usage. For every twelve-month (12) period of continuous service, customers are eligible to receive an additional rebate check based on the average of the combined previous twelve-month total usage. This rebate is calculated on an annual basis.

Average Monthly	
Usage	Percent (%)
(12 Months)	of Rebate
\$0 - \$100	5%
\$101 - \$200	6%
\$201 - \$300	7%
\$301 - \$400	8%
\$401 - \$600	9%
\$601 - \$700	10%
\$701 - \$999	11%
\$1,000 +	12%

# 6.1 MTS Service (Continued)

# 6.1.16 FiberCom MTS

### A. Per Minute Rates:

	Day		Evening		Night/Weekend	
<u>Miles</u>	1st Min	Addl Min	1st Min	Addl Min	1st Min	Addl Min
1 - 10	\$.1728	\$.1516	\$.1053	\$.1027	\$.0896	\$.0896
11 - 22	\$.2592	\$.1516	\$.1625	\$.1202	\$.1005	\$.1005
23 - 55	\$.3744	\$.2591	\$.2364	\$.1684	\$.1120	\$.1120
56 - 124	\$.4128	\$.2925	\$.2612	\$.1933	\$.1312	\$.1312
125 - 292	\$.4608	\$.3167	\$.2927	\$.2058	\$.1404	\$.1404
293 - 430	\$.4608	\$.3167	\$.2927	\$.2058	\$.1404	\$.1404
431 +	\$.4608	\$.3167	\$.2927	\$.2058	\$.1404	\$.1404

B. Billing Increments - Calls are billed in full minute increments.

C. <u>Monthly Service Charge</u>: None

D. <u>Monthly Minimum Usage</u>: \$5.00

# 6.1.17 FiberCom I

### A. <u>Description</u>

FiberCom I is a switched service designed for the large business with more than \$500.00 of monthly long distance usage.

### 6.1 MTS Service (Continued)

# 6.1.17 FiberCom I (Continued)

B. Per Minute Rates:

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1692 \$.1405 \$.1010

C. <u>Billing Increments</u> - Calls are billed in 1/10 minute increments with an 18-second minimum.

D. <u>Monthly Service Charge</u>: \$60.00

E. <u>Monthly Minimum Usage</u>: \$500.00

### 6.1.18 Sterling I

### A. Description

Sterling is a switched access service designed for business customers. Available through equal access or dial up with authorization code.

B. <u>Per Minute Rate Schedule</u>

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1788 \$.1400 \$.1200

- C. <u>Billing Increments</u> Six (6) second increment billing beginning with first six (6) seconds.
- D. Monthly Minimum Usage: None

1/ 10% discount on all usage in excess of \$2,000 per month

### 6.1 MTS Service (Continued)

# 6.1.19 Sterling

### A. Description

Sterling is a switched access service designed for business customers. Available through equal access or dial up with authorization code.

### B. <u>Per Minute Rate Schedule</u>

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1882 \$.1600 \$.1223

- C. <u>Billing Increments</u> Six (6) second increment billing beginning with first six (6) seconds.
- D. <u>Monthly Minimum Usage</u>: None

### 6.1.20 Elite 1 (29)

### A. Description

Elite 1 is a switched access service designed for business customers which requires no minimum usage. This service is available through equal access or dial up with authorization code.

### B. Per Minute Rate Schedule

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1500 \$.1400 \$.1400

C. <u>Billing Increments</u> - Six (6) second incremental billing beginning with first six (6) seconds.

### 6.1.21 Elite 2

### A. Description

Elite 2 is a switched access service designed for business customers with usage over \$150. This service is available through equal access or dial up with authorization code.

### B. Per Minute Rate Schedule

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1550 \$.1200 \$.1200

C. Billing Increments - Six (6) second incremental billing beginning with first six (6) seconds.

### 6.1.22 Express Services

### A. Description

Express Services are flat rate switched access services for business customers. Express Services are available through equal access or dial up with authorization code.

### B. Express 15

# 1. Per Minute Rate Schedule

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1526 \$.1500 \$.1500

2. <u>Billing Increments</u> - Six (6) second incremental billing beginning with the first six (6) seconds.

- 6.1 MTS Service (Continued)
  - 6.1.22 Express Services (Continued)
    - C. Express 16
      - 1. Per Minute Rate Schedule

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u>

**\$.1626 \$.1500 \$.1500** 

- 2. <u>Billing Increments</u> Six (6) second incremental billing beginning with the first six (6) seconds.
- D. Express 17
  - 1. Per Minute Rate Schedule

<u>Day Evening Night/Weekend</u> \$.1726 \$.1500 \$.1500

- 2. <u>Billing Increments</u> Six (6) second incremental billing beginning with the first six (6) seconds.
- E. Express 18
  - 1. Per Minute Rate Schedule

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1826 \$.1500 \$.1500

- 2. <u>Billing Increments</u> Six (6) second incremental billing beginning with the first six (6) seconds.
- F. Express 19
  - 1. Per Minute Rate Schedule

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1926 \$.1500 \$.1500

- 2. <u>Billing Increments</u> Six (6) second incremental billing beginning with the first six (6) seconds.
- G. Express 20
  - 1. <u>Per Minute Rate Schedule</u>

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.2026 \$.1500 \$.1500

- 2. <u>Billing Increments</u> Six (6) second incremental billing beginning with the first six (6) seconds.
- H. Express 62
  - Per Minute Rate Schedule

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1500 \$.1400 \$.1400

2. <u>Billing Increments</u> - Six (6) second incremental billing beginning with the first six (6) seconds.

### 6.1 MTS Service (Continued)

# 6.1.22 Express Services (Continued)

. Express 82

1. Per Minute Rate Schedule

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1750 \$.1494 \$.1143

2. <u>Billing Increments</u> - Six (6) second incremental billing beginning with the first six (6) seconds.

### J. Express 83

Per Minute Rate Schedule

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1850 \$.1400 \$.1400

2. <u>Billing Increments</u> - Six (6) second incremental billing beginning with the first six (6) seconds.

### K. Express 92

1. <u>Per Minute Rate Schedule</u>

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.1690 \$.1400 \$.1400

2. <u>Billing Increments</u> - Six (6) second incremental billing beginning with the first six (6) seconds.

# 6.1.23 <u>Custom Call</u> 100

### A. <u>Description</u>

This service arrangement allows the customer to originate calls in areas with equal access capabilities by pre-subscribing to the service. This service is also provided to subscribers of Company's Longer Distance Service interstate offering. The applicable rate depends upon the call mileage and the time of day the call originated. Discounts are given based on the customer's usage volume.

N I :- I- 4 / A / - - I - - - - - I

B. Per Minute Rates

	D	oay	Evening		Night/Weekend	
Mileage Band	1st Min	Addl Min	1st Min	Addl Min	1st Min	Addl Min
0- 10	\$.190	\$.090	\$.124	\$.059	\$.076	\$.036
11- 22	\$.280	\$.160	\$.182	\$.104	\$.112	\$.064
23- 55	\$.400	\$.270	\$.260	\$.176	\$.160	\$.108
56- 124	\$.440	\$.310	\$.286	\$.202	\$.176	\$.124
125- 292	\$.490	\$.330	\$.319	\$.215	\$.196	\$.132
293- 430	\$.490	\$.350	\$.332	\$.228	\$.204	\$.140
431+	\$.510	\$.350	\$.332	\$.228	\$.204	\$.140

- C. <u>Billing Increments</u> Calls are billed in one minute increments.
- D. <u>Monthly Volume Discounts</u>

Discounts will be applied to a Custom Call 100 customer's usage charge based on the total eligible day period usage charge per account for that month. Eligible usage includes all day period charges, including both interstate and intrastate calling and Preferred Calling usage, and excludes charges for Directory Assistance, Operator Services and per call surcharges. The discount corresponding to a certain volume level will apply to all eligible usage.

### 6.1 MTS Service (Continued)

# 6.1.23 Custom Call 100 (Continued)

# D. <u>Monthly Volume Discounts</u> (Continued)

<u>unt</u>

Longer Distance Service customers will receive the following usage volume discounts, applicable to usage in all rate periods. To calculate the discount, multiply the usage in each discount band by the applicable discount rate and sum.

Usage Charge	<u>Discount</u>
\$0 - \$14.99	0%
\$15 - \$74.99	2%
\$75 - \$199.99	5%
\$200 - Up	10%

# 6.1.24 Preferred Calling

### A. Description

This service arrangement allows customers to originate calls through use of an authorization code via Company-provided switched access facilities by dialing 950-0488, a local access number or a designated "800" access number. The applicable usage rate depends on the type of access utilized. A call originated by dialing 950-0488 or a local access number is billed at time of day and distance sensitive per minute rates. No surcharge applies. A call originated by dialing the "800" number is billed per minute rates with a surcharge applicable to each call. Discounts are given based on the customer's usage volume.

## 6.1 MTS Service (Continued)

## 6.1.24 Preferred Calling (Continued)

### B. <u>950-type access</u>

## Per Minute Rates

	D	)ay	Evening		Night/Weekend-	
Mileage Band	1st Min	Addl Min	1st Min	Addl Min	1st Min	Addl Min
0- 10	\$.2152	\$.1018	\$.1632	\$.0663	\$.1267	\$.0599
11- 22	\$.3175	\$.2062	\$.2401	\$.1102	\$.1866	\$.1065
23- 55	\$.4563	\$.3095	\$.3450	\$.2348	\$.2788	\$.1818
56- 124	\$.5024	\$.3498	\$.3800	\$.2650	\$.2957	\$.2056
125- 292	\$.5512	\$.3763	\$.4203	\$.2851	\$.3265	\$.2215
293- 430	\$.5719	\$.3917	\$.4341	\$.2968	\$.3371	\$.2306
431+	\$.5782	\$.3964	\$.4378	\$.3005	\$.3397	\$.2332

## C. <u>800-type access</u>

The following rates apply to calls originated via 800-type connections.

	Per Call
Surcharge:	\$.75

Per Minute Rates	Initial Minute	Additional Minute
Day:	\$.250	\$.250
Evening:	\$.250	\$.250
Night/Weekend:	\$.250	\$.250

- D. Billing Increments Calls are billed in one minute increments.
- E. <u>Volume Discount</u> The volume discounts for Custom Call 100, preceding, apply to charges for Preferred Calling.

### 6.1 MTS Service (Continued)

#### 6.1.25 Longer Distance Niteline

#### A. Description

This service arrangement flat charges for all usage in the Evening and Night/Weekend time periods, measured in quarter-hour increments following the initial one hour per month. Calls are originated through use of an authorization code via Company-provided shared access facilities by dialing 950-0488 or a local access number. Call time is accumulated in one minute increments.

## B. <u>Usage Charge</u>

Intrastate usage will be combined with interstate usage to determine the charge. The definition of time periods will apply as defined in 4.1.

Calls made during the Evening and Night/Weekend time period:

Fixed Monthly Charge (includes 1 hour of usage) \$11.95
Each additional 15 minute period or part thereof \$2.50

Calls made during the Day time period:

-Charge per minute \$ 0.35

#### 6.1.26 Longer Distance Autodialer

## A. <u>Description</u>

This service arrangement permits the customer to access Custom Call 100 service without entering the access number and authorization code. There is a minimum monthly usage charge associated with this service arrangement.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 6.1 MTS Service (Continued)
  - 6.1.26 Longer Distance Autodialer (Continued)
    - B. <u>Usage Charge</u> The usage charge is that for Custom Call 100.
    - C. Minimum Usage Charge

There is a Billing Usage Adjustment Charge for subscribers who do not provide at least \$250 in billed usage per account during any given billing period. The billing adjustment charge will not be prorated and will apply in full in the event that service is provided for less than a month in the initial or final billing period. The charge is determined as follows:

- 1. The Billing Usage Adjustment Charge will not apply if the customer's usage charges in that billing period are greater than or equal to \$250
- 2. If the customer's usage charges are \$50 or less, a Usage Adjustment Charge of \$50 will apply in addition to the customer's usage charges.
- 3. Where the customer's usage charges are less than \$250, but greater than \$50 the Billing Usage Adjustment Charge will be determined by applying a \$.25 surcharge for each dollar that the customer's usage is below \$250. For example, if the customer's usage is \$200, this usage is \$50 below \$250. Therefore to determine the Billing Usage Adjustment Surcharge, multiply \$50 by \$.25. The result is \$12.50.
- The Autodialer Billing Usage Adjustment Charge may be satisfied by aggregating the usage from all the autodialer systems and travel codes in the customer's account.

### 6.1 MTS Service (Continued)

#### 6.1.27 Performance Service

#### A. Description

This service arrangement is a one-way, dial-in-dial-out multipoint service that allows the customer to originate and terminate calls via the Carrier provided telephone lines. Customers who presubscribe to the Carrier may do so on a direct dial basis, by dialing 10011, via a 7 digit access number or through operator assistance. All direct dial calls are rounded to the next highest full minute.

## B. <u>Usage Charges</u>

Each Performance Service call is charged at a rate for the first and each additional minute or fraction of a minute from the following applicable Rates Schedules as determined by the actual distance in miles between the metropolitan areas in which the call originates and terminates. The rate tables below incorporate the volume discounts as set forth in Section 6.1.27.C.

### DAY RATE SCHEDULE

	\$0 -	- \$49	\$50	- \$99	\$100 -	- \$199	\$2	00 +
<u>Mileage</u>	1st Min	Addl Min						
1-10	\$.1862	\$.0873	\$.1825	\$.0856	\$.1769	\$.0830	\$.1732	\$.0812
11-16	\$.2744	\$.1552	\$.2689	\$.1521	\$.2607	\$.1475	\$.2552	\$.1444
17-22	\$.2744	\$.1552	\$.2689	\$.1521	\$.2607	\$.1475	\$.2552	\$.1444
23-30	\$.3345	\$.2228	\$.3278	\$.2183	\$.3178	\$.2117	\$.3111	\$.2072
31-40	\$.3365	\$.2242	\$.3298	\$.2197	\$.3197	\$.2130	\$.3129	\$.2085
41-55	\$.3365	\$.2242	\$.3298	\$.2197	\$.3197	\$.2130	\$.3129	\$.2085
56-70	\$.3475	\$.2300	\$.3406	\$.2254	\$.3301	\$.2185	\$.3232	\$.2139
71-124	\$.3475	\$.2300	\$.3406	\$.2254	\$.3301	\$.2185	\$.3232	\$.2139
125-196	\$.3555	\$.2446	\$.3484	\$.2397	\$.3377	\$.2324	\$.3306	\$.2275
197-292	\$.3555	\$.2446	\$.3484	\$.2397	\$.3377	\$.2324	\$.3306	\$.2275
293-430	\$.3555	\$.2446	\$.3484	\$.2397	\$.3377	\$.2324	\$.3306	\$.2275
430 +	\$.3718	\$.2546	\$.3644	\$.2495	\$.3532	\$.2419	\$.3458	\$.2368

## 6.1 MTS Service (Continued)

430 +

\$.1800

\$.1295

\$.1800

## 6.1.27 Performance Service (Continued)

## B. <u>Usage Charges</u> (Continued)

		ı	EVENING R	ATE SCHEE	ULE			
	\$0 -	- \$49	\$50	- \$99	\$100 -	- \$199	\$2	00 +
<u>Mileage</u>	1st Min	Addl Min	1st Min	Addl Min	1st Min	Addl Min	1st Min	Addl Min
1-10	\$.1419	\$.0631	\$.1419	\$.0631	\$.1391	\$.0618	\$.1391	\$.0618
11-16	\$.2092	\$.1155	\$.2092	\$.1155	\$.2050	\$.1132	\$.2050	\$.1132
17-22	\$.2092	\$.1155	\$.2092	\$.1155	\$.2050	\$.1132	\$.2050	\$.1132
23-30	\$.2444	\$.1672	\$.2444	\$.1672	\$.2395	\$.1639	\$.2395	\$.1639
31-40	\$.2469	\$.1689	\$.2469	\$.1689	\$.2420	\$.1655	\$.2420	\$.1655
41-55	\$.2469	\$.1689	\$.2469	\$.1689	\$.2420	\$.1655	\$.2420	\$.1655
56-70	\$.2606	\$.1780	\$.2606	\$.1780	\$.2554	\$.1745	\$.2554	\$.1745
71-124	\$.2606	\$.1780	\$.2606	\$.1780	\$.2554	\$.1745	\$.2554	\$.1745
25-196	\$.2615	\$.1827	\$.2615	\$.1827	\$.2563	\$.1790	\$.2563	\$.1790
197-292	\$.2615	\$.1827	\$.2615	\$.1827	\$.2563	\$.1790	\$.2563	\$.1790
293-430	\$.2615	\$.1827	\$.2615	\$.1827	\$.2563	\$.1790	\$.2563	\$.1790
430 +	\$.2764	\$.1949	\$.2764	\$.1949	\$.2709	\$.1910	\$.2709	\$.1910
		NUOLITA	MEEKEND F	ATE OOUE				
	Φ0			RATE SCHE		<b>#</b> 400	Φ0	00.
N 4:1	•	- \$49		- \$99	\$100 -	•	•	00 +
Mileage	1st Min	Addl Min	1st Min	Addl Min	1st Min	Addl Min	1st Min	Addl Min
1-10	\$.0949	\$.0425	\$.0949	\$.0425	\$.0949	\$.0425	\$.0949	\$.0425
11-16	\$.1399	\$.0760	\$.1399	\$.0760	\$.1399	\$.0760	\$.1399	\$.0760
17-22	\$.1399	\$.0760	\$.1399	\$.0760	\$.1399	\$.0760	\$.1399	\$.0760
23-30	\$.1633	\$.1117	\$.1633	\$.1117	\$.1633	\$.1117	\$.1633	\$.1117
31-40	\$.1633	\$.1117	\$.1633	\$.1117	\$.1633	\$.1117	\$.1633	\$.1117
41-55	\$.1633	\$.1117	\$.1633	\$.1117	\$.1633	\$.1117	\$.1633	\$.1117
56-70	\$.1725	\$.1176	\$.1725	\$.1176	\$.1725	\$.1176	\$.1725	\$.1176
71-124	\$.1725	\$.1176	\$.1725	\$.1176	\$.1725	\$.1176	\$.1725	\$.1176
125-196	\$.1750	\$.1230	\$.1750	\$.1230	\$.1750	\$.1230	\$.1750	\$.1230
197-292	\$.1750	\$.1230	\$.1750	\$.1230	\$.1750	\$.1230	\$.1750	\$.1230
293-430	\$.1775	\$.1246	\$.1775	\$.1246	\$.1775	\$.1246	\$.1775	\$.1246

\$.1295

\$.1800

\$.1295

\$.1800

\$.1295

### 6.1 MTS Service (Continued)

## 6.1.27 Performance Service (Continued)

#### C. Monthly Volume Discounts

Performance Service calls shall be entitled to additional volume discounts applied to the customer's total Performance Service dollar usage in each rate period below based upon the Customer's total dollar usage level attained at all rate periods.

Dollar Usage Level	Rate Period Volume				
Discount percent	<u>Day</u>	Evening	Night/Weekend		
\$00.00 - \$49.99	0.0%	0.0%	0.0%		
\$50.00 - \$99.99	2.0%	0.0%	0.0%		
\$100.00 - \$199.99	5.0%	2.0%	0.0%		
\$200 +	7.0%	2.0%	0.0%		

#### 6.1.28 Performance Plus Service

#### A. <u>Description</u>

This service arrangement is a one-way, dial-in-dial-out multipoint service that allows the customer to originate and terminate calls via the Carrier provided telephone lines. To obtain Service, customer's premises must be located in Equal Access exchange areas and customer must obtain access to Carrier via Equal Access dial plan arrangement through the local exchange telephone carrier. Customers may also use authorization codes to make Performance Plus Calls.

### B. <u>Usage Charges</u>

All Performance Plus calls are charged at a rate for each six seconds or fraction thereof, subject to a minimum of 30 seconds per call. The rates below incorporate the volume discounts set forth in Section 6.1.28.C.

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 6.1 MTS Service (Continued)
  - 6.1.28 Performance Plus Service (Continued)
    - B. <u>Usage Charges</u> (Continued)

## DAY RATE SCHEDULE / PER MINUTE RATES

<u>Mileage</u>	\$0-\$249	\$250-\$499	\$500-\$999	\$1000-\$1999	\$2000-\$499	99 \$5000+
1-10	\$.1887	\$.1642	\$.1548	\$.1472	\$.1397	\$.1321
11-16	\$.1887	\$.1642	\$.1548	\$.1472	\$.1397	\$.1321
17-22	\$.2091	\$.1819	\$.1715	\$.1631	\$.1547	\$.1464
23-30	\$.2683	\$.2334	\$.2200	\$.2093	\$.1986	\$.1878
31-40	\$.2683	\$.2334	\$.2200	\$.2093	\$.1986	\$.1878
41-55	\$.2901	\$.2524	\$.2379	\$.2263	\$.2147	\$.2031
56-70	\$.2901	\$.2524	\$.2379	\$.2263	\$.2147	\$.2031
71-124	\$.3253	\$.2831	\$.2668	\$.2538	\$.2408	\$.2277
125-196	\$.3515	\$.3058	\$.2883	\$.2742	\$.2601	\$.2461
197-292	\$.3548	\$.3087	\$.2909	\$.2767	\$.2625	\$.2484
293-430	\$.3548	\$.3087	\$.2909	\$.2767	\$.2625	\$.2484
430+	\$.3548	\$.3087	\$.2909	\$.2767	\$.2625	\$.2484

### **EVENING RATE SCHEDULE / PER MINUTE RATES**

<u>Mileage</u>	<u>\$0-\$249</u>	<u>\$250-\$499</u>	<u>\$500-\$999</u>	<u>\$1000-\$1999</u>	\$2000-\$4999	\$5000+
1-10	\$.1040	\$.1040	\$.1040	\$.0988	\$.0988	\$.0988
11-16	\$.1040	\$.1040	\$.1040	\$.0988	\$.0988	\$.0988
17-22	\$.1238	\$.1238	\$.1238	\$.1176	\$.1176	\$.1176
23-30	\$.1782	\$.1782	\$.1782	\$.1693	\$.1693	\$.1693
31-40	\$.1782	\$.1782	\$.1782	\$.1693	\$.1693	\$.1693
41-55	\$.1937	\$.1937	\$.1937	\$.1841	\$.1841	\$.1841
56-70	\$.1937	\$.1937	\$.1937	\$.1841	\$.1841	\$.1841
71-124	\$.2159	\$.2159	\$.2159	\$.2051	\$.2051	\$.2051
125-196	\$.2308	\$.2308	\$.2308	\$.2192	\$.2192	\$.2192
197-292	\$.2340	\$.2340	\$.2340	\$.2223	\$.2223	\$.2223
293-430	\$.2340	\$.2340	\$.2340	\$.2223	\$.2223	\$.2223
430+	\$.2340	\$.2340	\$.2340	\$.2223	\$.2223	\$.2223

### 6.1 MTS Service (Continued)

## 6.1.28 Performance Plus Service (Continued)

#### B. Usage Charges (Continued)

## NIGHT/WEEKEND RATE SCHEDULE / PER MINUTE RATES

<u>Mileage</u>	\$0-\$249	\$250-\$499	\$500-\$999	\$1000-\$1999	\$2000-\$499	9 <u>\$5000+</u>
1-10	\$.0990	\$.0990	\$.0990	\$.0990	\$.0990	\$.0990
11-16	\$.0990	\$.0990	\$.0990	\$.0990	\$.0990	\$.0990
17-22	\$.0990	\$.0990	\$.0990	\$.0990	\$.0990	\$.0990
23-30	\$.1148	\$.1148	\$.1148	\$.1148	\$.1148	\$.1148
31-40	\$.1148	\$.1148	\$.1148	\$.1148	\$.1148	\$.1148
41-55	\$.1247	\$.1247	\$.1247	\$.1247	\$.1247	\$.1247
56-70	\$.1247	\$.1247	\$.1247	\$.1247	\$.1247	\$.1247
71-124	\$.1396	\$.1396	\$.1396	\$.1396	\$.1396	\$.1396
125-196	\$.1498	\$.1498	\$.1498	\$.1498	\$.1498	\$.1498
197-292	\$.1518	\$.1518	\$.1518	\$.1518	\$.1518	\$.1518
293-430	\$.1518	\$.1518	\$.1518	\$.1518	\$.1518	\$.1518
430+	\$.1518	\$.1518	\$.1518	\$.1518	\$.1518	\$.1518

### C. Volume Discounts

Volume Discounts are available to those customers who have established Performance Plus Service accounts with Carrier and whose total Performance Plus Service, Performance Service and 950 access usage charges equal or exceed \$250.00. Performance Service and 950 accessed calls will continue to be rated at Performance Service rates. The discounts will be applied to the customer's monthly invoice for the Performance Plus Service, Performance Service calls and will be based on the total monthly Performance Plus Service, and Performance Service usage incurred in each rate period.

### 6.1 MTS Service (Continued)

## 6.1.28 Performance Plus Service (Continued)

#### C. <u>Volume Discounts</u> (Continued)

Monthly Performance		Discou	nts (%)
Plus Service Usage	<u>Day</u>	<b>Evenings</b>	Nights/Weekend
\$0.00 - \$249.99	0.0%	0.0%	0.0%
\$250.00 - \$499.99	13%	0.0%	0.0%
\$500.00 - \$999.99	18%	0.0%	0.0%
\$1000.00 -\$1999.00	22%	5%	0.0%
\$2000.00 -\$4999.00	26%	5%	0.0%
\$5000.00 +	30%	5%	0.0%

A \$15.00 dollar monthly charge shall apply to each Performance Plus Service account. This charge will be waived for partial billing cycles in the first and last month of service.

#### 6.1.29 Precision Service

### A. <u>Description</u>

This service arrangement utilizes switched access services. Precision Service calls are measured in six second increments, subject to a minimum call duration of six (6) seconds. The applicable rate for Precision Service rates are not distance or time-of-day sensitive. In addition to usage charges, a monthly fee or minimum usage level also applies. The usage level will be based upon the total monthly (billing cycle) usage charges for interstate and intrastate Precision Service and Performance Service.

### 6.1 MTS Service (Continued)

## 6.1.29 <u>Precision Service</u> (Continued)

#### B. Usage Charges

	Monthly Fee	Rate/Min.
Option I	\$20.00	\$.2050
Option II	\$50.00	\$.1950
Option III	\$75.00	\$.1900
Option IV	\$ 0.00	\$.1800

## C. <u>Billing Increments</u>

Precision Service is billed in six second increments, subject to a minimum call duration of 6 seconds.

#### D. Minimum Usage Requirement

A minimum monthly usage level requirement of \$4,000.00 applies for Option IV Service. Precision Service usage charges only apply towards the minimum requirement.

#### 6.1.30 MLD Message Telephone Service

### A. Description

This service arrangement allows a customer to originate calls in areas with Equal Access capabilities served by the Company by pre-subscribing to the service, or, for casual use, by dialing a 5 digit (10XXX) code. This service is also available to customers in areas without equal access capabilities through use of a non-travel authorization code and Company-provided switched access facilities. Calls can be terminated at any point in the state.

The applicable usage rate depends upon the distance between originating and terminating points and the rate period(s) in which the call occurs. Discounts are given based upon the customer's usage volume.

### 6.2 <u>Dedicated WATS Services</u>

### 6.2.1 <u>LaserEXPRESS USA</u>

#### A. Description

LaserEXPRESS USA is a one-way, virtually banded, outbound WATS service that utilizes dedicated access facilities provided by the Local Exchange Carrier. This product is designed for large users.

B. Rate Schedule (Per Minute of Use)

Called Destination	<u>Day</u>	Evening	Night/Weekend
All Florida Exchanges	\$.1590	\$0.129	\$0.099

C. <u>Billing Increments</u> - Calls are rated at an initial eighteen (18) seconds and in six second increments thereafter.

## D. Volume Discount

A volume discount is applied to the total intrastate usage charges billed in a month under the LaserEXPRESS USA option per location.

Monthly Usage Charges	<u>Day</u>	Evening	Night/Weekend
\$0.00 - \$2,999.99	0%	0%	0%
\$3,000.00 - \$3,999.99	7%	4%	4%
\$4,000.00 - \$4,999.99	9%	5%	5%
\$5,000.00 - \$9,999.99	10%	6%	6%
\$10,000.00 - \$4,999.99	12%	8%	8%
\$15,000.00 - \$49,999.99	14%	10%	10%
\$50,000.00 - \$149,999.99	18%	10%	10%
\$150,000.00 and above	20%	10%	10%

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 6.2 <u>Dedicated WATS Services</u> (Continued)
  - 6.2.1 <u>LaserEXPRESS USA (Continued)</u>
    - E. Minimum Usage Charge per location: \$1,350.00
    - F. <u>Minimum Number of Lines in Service</u>

Each location is subject to a minimum of 3 lines in service at all times.

## G. Non-Recurring Charges 1/

1.	First dedicated access line installed at one time	\$200.00
2.	Each add'l dedicated access line installed at same time	\$50.00
3.	Conversion charge, per line	\$100.00
4.	Expedite Order (per order)	\$100.00
5.	Cancellation of Order (per order)	\$130.00
6.	Administrative (per order)	\$20.00
7.	T-1 Installation	\$660.00
8.	Predefined Account Code Set-up	\$20.00
9.	Customized Account Code Set-up	\$100.00
	(must have a minimum of monthly usage of \$7500.00)	
10.	Call detail (paper copy)	\$20.00
11.	Magnetic Tape of Call detail	\$200.00
12.	Account Code Summary	\$ 0.00

ISSUED: November 13, 2008

<sup>1/</sup> Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

## 6.2 <u>Dedicated WATS Services</u> (Continued)

## 6.2.1 <u>LaserEXPRESS USA (Continued)</u>

#### H. Monthly Recurring Charges 1/

1.	Access Port Charge (per circuit)	\$55.00
2.	Project Accounting Codes (001-999)	\$20.00
3.	Predefined Account Codes	\$30.00
4.	Customized Account Codes	\$50.00
5.	Speed Numbers (each)	\$ 0.50
6.	T-1 Port Charge	\$400.00
7.	Call detail (paper copy)	\$20.00
8.	Magnetic Tape of Call detail	\$75.00
9.	Account Code Summary	\$20.00

#### 6.2.2 HotelEXPRESS

#### A. Description

A variation of LaserEXPRESS USA, is available to customers whose distribution of total monthly traffic is less than 55% for the business day rate period. If the percent of business day traffic exceeds 55%, usage charges are assessed using the rates associated with the standard LaserEXPRESS USA service or the HotelEXPRESS minimum monthly usage, whichever is higher. Three options are available, HotelEXPRESS A, B and C.

#### B. Minimum Period

All options are subject to a one (1) year minimum commitment. Converting from one option to another is not permitted, HotelEXPRESS requires the customer to originate calls via dedicated facilities obtained from the Local Exchange Company (LEC).

<sup>1/</sup> Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

### 6.2 <u>Dedicated WATS Services</u> (Continued)

### 6.2.2 <u>HotelEXPRESS</u> (Continued)

#### C. <u>Minimum Usage Requirements</u>

Usage is subject to a Minimum Average Time Requirement (MATR), a Minimum Minutes of Use Requirement and a Minimum Usage Requirement. The MATR is one (1) minute per completed call and is applied to all messages in a service group. The Minimum Minutes of Use and the Minimum Usage are assessed in accordance with the HotelEXPRESS option that is chosen.

# D. <u>Monthly Usage Charges</u>

	Rate/minute	Minutes of Use	<u>Minimum Usage</u>
HotelEXPRESS A	\$.160	50,000	\$8,000.00
HotelEXPRESS B	\$.155	80,000	\$12,400.00
HotelEXPRESS C	\$.150	100,000	\$15,000.00

E. <u>Billing Increments</u> - HotelEXPRESS is billed in six (6) second increments.

### F. Non-Recurring Charges 1/

1.	Expedite Order (per order)	\$100.00
2.	Cancellation of order (per order)	\$130.00
3.	Administrative (per order)	\$20.00
4.	Disconnect Fee (prior to fulfilling min. commitment)	\$500.00
5.	1st Dedicated Access Line installed	\$200.00
6.	Each Add'l Access Line installed	\$50.00
7.	T-1 installation	\$660.00
8.	Call Detail (paper copy)	\$20.00
9.	Magnetic Tape call detail	\$200.00

<sup>1/</sup> Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

#### 6.2 <u>Dedicated WATS Services</u> (Continued)

#### 6.2.2 HotelEXPRESS (Continued)

#### G. Monthly Recurring Charges 1/

1.	Account Codes	\$20.00
2.	Account Code Summary	\$20.00
3.	Call Detail (paper copy)	\$20.00
4.	Magnetic Tape call detail	\$75.00

#### 6.2.3 Corporate LaserEXPRESS

#### Α. Description

Corporate LaserEXPRESS is a one-way, virtually banded, outbound WATS service that utilizes Special or T-1 access. This product is designed for high volume users.

#### B. Rate Schedule

### Rate Per Minute of Use

Called Destination	<u>Day</u>	<u>Evening</u>	Night/Weekend
All Florida Exchanges	\$0.1401	\$0.1082	\$0.0876

C. Billing Increments - Calls are rated on an initial eighteen (18) seconds and in six (6) second increments thereafter.

Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access 1/ shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

## 6.2 <u>Dedicated WATS Services</u> (Continued)

## 6.2.3 <u>Corporate LaserEXPRESS</u> (Continued)

#### D. Non-Recurring Charges 1/

1. 2.	First Line (Special Access) Each Add'l Special Access Line	\$200.00 <u>2/</u> \$50.00 <u>2/</u>
3.	First Line (T-1 Access, per T-1)	\$450.00 <u>2/</u>
4.	Call Detail	\$20.00 <u>2/</u>
5.	Magnetic Tape	\$200.00 <u>3</u> /
6.	Conversion Processing Fee (per order)	\$150.00
7.	Expedite Order (per order)	\$200.00
8.	Cancellation of Order (per order)	\$150.00
9.	Administrative (per order)	\$30.00
10.	Custom Account Coding	\$100.00
11.	Validated Account Coding; set up	\$20.00
12.	Non-Validated Account Coding; setup	\$15.00

ISSUED: November 13, 2008

<sup>1/</sup> Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

Those customers electing service under the Association Saver plan will not be charged the non-recurring fees normally associated with Corporate LaserEXPRESS.

<sup>3/</sup> For Association Saver customers, the non-recurring charge for magnetic tape billing is \$100.00.

### 6.2 <u>Dedicated WATS Services</u> (Continued)

### 6.2.3 <u>Corporate LaserEXPRESS</u> (Continued)

#### E. Monthly Recurring Charges 1/

4	0 11 5 4 11	<b>***</b>
1.	Call Detail	\$20.00
2.	Account Coding	\$15.00
3.	Magnetic Tape	\$75.00 <u>2</u>
4.	Service Fee	\$ 0.00
5.	Custom Account Coding	\$50.00
6.	Validated Account Coding	\$30.00
7.	Non-Validated Account Coding	\$15.00

#### 6.2.4 LaserWATS I

#### A. Description

LaserWATS I is a one-way, virtually banded, outbound WATS service that utilizes dedicated access facilities provided by the Local Exchange Carrier. This product is designed for medium to large users.

B. Rate Schedule (Rate Per Minute of Use)

<u>Day</u>	Evening	Night/Weekend
\$0.2134	\$0.1600	\$0.1250

Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

For those customers electing service under the Association Saver plan the monthly recurring charge for Magnetic Tape Billing will be \$50.00.

### 6.2 <u>Dedicated WATS Services</u> (Continued)

### 6.2.4 <u>LaserWATS I</u> (Continued)

#### C. Non-Recurring Charges 1/

1.	Installation Circuit Charge (per circuit)	\$50.00
2.	Expedite Order (per order)	\$100.00
3.	Administrative (per order)	\$20.00

## D. Monthly Recurring Charges 1/

1.	Access port charge per line (billed in advance	of service) \$30.00
2.	Project Accounting Codes	\$20.00
3.	Speed Numbers (per number)	\$ 0.50
4.	Magnetic tape of call detail	\$85.00
5.	Minimum usage charge per circuit	\$300.00

#### 6.2.5 LaserWATS II

## A. <u>Description</u>

A one-way InterLATA multipoint service requiring a customer to originate calls via a WATS Access Line (WAL) provided by the Local Exchange Carrier. This product is available only in Southern Bell exchange areas that are Company-served equal access areas. This product is designed for medium users.

<sup>1/</sup> Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

## 6.2 <u>Dedicated WATS Services</u> (Continued)

## 6.2.5 <u>LaserWATS II</u> (Continued)

## B. Rate Schedule

## 1. Rate Per Minute of Use

Called Destination	Day	Evening	Night/Weekend
All Florida Exchanges	\$0.265	\$0.170	\$0.125

## 2. <u>Monthly Volume Discounts</u>

Total Monthly Usage	All Rate Periods
\$0 - \$349.99	0%
\$350 - \$699.99	2%
\$700 - \$999.99	3.5%
\$1000 +	5%

## C. <u>Non-Recurring Charges</u>

1.	Call detail (paper copy); per acct	\$10.00
2.	Magnetic tape call detail	\$200.00
3.	Administrative (per order)	\$20.00

## D. Recurring Charges

1.	Call detail (paper copy); per acct	\$10.00
2.	Account Codes (per customer)	\$20.00
3.	Magnetic tape call detail	\$75.00

### 6.2 <u>Dedicated WATS Services</u> (Continued)

### 6.2.6 CUSTOM WATS

#### A. Description

This service arrangements permits the customer to originate calls via a dedicated WATS access line connecting his premises to a Local Exchange Company (LEC) WATS serving office. This line is obtained from and billed by the LEC. Calling is permitted to all points with the state. Inter-LATA calls are handled and billed by the Company, intra-LATA calls are handled and billed by the LEC. The applicable rate depends on the time of day the call originated. Discounts are given based on a customer's usage volume. A monthly service charge and an installation charge apply to this service.

#### B. Usage Charge

	<u>Day</u>	Evening	Night/Weekend
Per Minute	\$.2668	\$.1753	\$.1225

C. <u>Billing Increments</u> - Inter-LATA calls are billed in six second increments after an initial one minute period.

### D. Monthly Volume Discounts

Discounts will be applied to the customer's usage charges based upon the total inter-LATA usage charge per account for the billing period. The discount is expressed as a percentage reduction in the inter-LATA usage charge for that amount of the customer's usage charge that falls within each discount band.

InterLATA Usage Charge	Discount
\$0 - \$500.00	0%
\$500.01-\$1000.00	2%
Over \$1000.00	5%

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 6.2 <u>Dedicated WATS Services</u> (Continued)

#### 6.2.7 CUSTOM WATS 100

#### A. Description

This service arrangement allows the customer to originate calls via equal access facilities connecting the customer's premises to a suitably equipped local exchange company (LEC) office or via switched access facilities provided by the Company. The applicable rate depends on the time of day the call originated. Discounts are given based on a customer's usage volume. A monthly service charge applies to each billing account.

CUSTOM WATS 100 rates are effective when the customer's billing information is entered into the billing system. Until such time Custom Call 100 rates will apply. A customer who fails to pay any charges owed to the Company as prescribed in Section 5 will be terminated as an CUSTOM WATS 100 customer. If the customer continues to access the Company's Network, they will be billed at Custom Call 100 rates. Additionally, the customer will not be eligible for any volume discount. Customers who have no established account with the Company but utilize the Company's network through casual access by dialing 10488 will not be eligible for CUSTOM WATS 100 rates.

#### B. Per Minute Rates:

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$.2436 \$.1827 \$.1267

 Billing Increments - Calls are billed in 30 second increments after an initial 30 second period.

### 6.2 <u>Dedicated WATS Services</u> (Continued)

## 6.2.7 CUSTOM WATS 100 (Continued)

#### D. Monthly Volume Discounts

Discounts will be applied to the customer's intrastate usage charges based upon the total CUSTOM WATS 100 intrastate usage charge per account for the billing period. The discount is expressed as a percentage reduction in the intrastate usage charge for that amount of the customer's intrastate usage charge that falls within each discount band.

Intrastate Usage Charge	Discount
\$200.00 - \$499.99	10%
\$500.00 - \$999.99	15%
\$000.00 +	21%

## E. Monthly Service Charge

Per Account \$10.00

The monthly service charge will not apply when the initial billing period is less than a full month or any billing period in which no CUSTOM WATS 100 calls were made, the monthly charge will apply in full in the event the final billing period is less than a full month.

### 6.2 <u>Dedicated WATS Services</u> (Continued)

#### 6.2.8 CUSTOM WATS 400

#### A. Description

This service arrangement allows the customer to originate calls via dedicated DS-1 access facilities\* connected to a Company serving office and terminate calls via switched access facilities at any point within the state. The customer may elect to have his service arrangement configured to include the Private Network Option. This permits the customer to complete calls to his other company served locations also having the Private Network Option via the dedicated access lines connecting the called location to the Company switch. Different rate schedules are employed to reflect the location of the called point and the type of egress facilities used. The customer is subject to monthly service charges, usage charges, and access and installation charges.

#### B. Usage Rate Schedule

Calls completed via dedicated access facilities to the called location are designated as Private Network Option calls. Calls completed via shared access facilities are designated as Standard calls.

### 1. Standard Calls

		Rate per Minute			
Mileage Band	<u>Day</u>	Evening	Night/Weekend		
0-292	\$.1170	\$.0970	\$.0760		
293-430	\$.1320	\$.1070	\$.0820		
431+	\$.1450	\$.1180	\$.0900		

- (a) Calls are billed in six second increments.
- Private Network Option Calls The rates for Private Network Option Calls are those for Standard calls as listed above, less the following amounts.

### 6.2 <u>Dedicated WATS Services</u> (Continued)

### 6.2.8 CUSTOM WATS 400 (Continued)

#### C. Monthly Service Charge

Per access line group: \$50

<u>Time Period</u> <u>Discount per Minute</u>

Day \$.03 Evening, Night \$.02

## 6.2.9 One Step WATS Service

#### A. Description

This service arrangement is a one-way, multipoint service requiring the customer to originate calls via dedicated facilities between his premises and Carrier terminal location. All calls may be subject to a 30 second minimum average time requirement (MATR). All calls are rounded to the next highest 6-second increment.

#### B. Usage Charges

All One Step WATS calls are charged at a rate for each six seconds or fraction thereof subject to a minimum of 6 seconds per call. The rates below incorporate the volume discounts set forth in Section 6.2.12.C.

## DAY RATE SCHEDULE / PER MINUTE RATES

<u>Mileage</u>	<u>\$0-\$499</u>	\$500-\$9 <u>99</u>	<u>\$1000-\$1999</u>	\$2000-\$3999	\$4000+
0 - 55	\$.1925	\$.1867	\$.1829	\$.1771	\$.1694
56-124	\$.2025	\$.1964	\$.1924	\$.1863	\$.1782
125+	\$.2125	\$.2061	\$.2019	\$.1955	\$.1870

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 6.2 <u>Dedicated WATS Services</u> (Continued)
  - 6.2.9 One Step WATS Service (Continued)
    - B. Usage Charges (Continued)

## EVENING/NIGHT/WEEKEND RATE SCHEDULE / PER MINUTE RATES

<u>Mileage</u>	<u>\$0-\$499</u>	<u>\$500-\$999</u>	<u>\$1000-\$1999</u>	\$2000-\$3999	<del>\$4000+</del>
0 - 55	\$.1444	\$.1400	\$.1372	\$.1328	\$.1271
56-124	\$.1519	\$.1473	\$.1443	\$.1397	\$.1337
125+	\$.1594	\$.1546	\$.1514	\$.1466	\$.1403

- C. A \$10.00 monthly charge shall apply to each One Step WATS account. This charge will be waived for partial billing cycles in the first and last month of service.
- D. Monthly Volume Discounts

Volume discounts are available to those customers who have established One Step WATS accounts with Carrier and whose total One Step WATS usage in a billing cycle equals or exceeds \$500.00. The discount will be applied to the customers monthly invoice for One Step WATS and will be based on usage incurred for all rate periods.

		Discount Percent		
Dollar Usage Level	<u>Day</u>	Evening/Night/Weekend		
\$0.00 - \$499.99	0.0%	0.0%		
\$500.00 - \$999.99	3.0%	3.0%		
\$1,000.00 - \$1,999.99	5.0%	5.0%		
\$2,000.00 - \$3,999.99	8.0%	8.0%		
\$4,000.00 +	12.0%	12.0%		

## 6.2 <u>Dedicated WATS Services</u> (Continued)

### 6.2.10 Performance WATS

#### A. Description

This service arrangement is a one-way outward service requiring the customer to originate calls via a dedicated facility or WATS access lines between the customer premises and Carrier's terminal location and allowing the completion of calls anywhere in the state of Florida, the Continental United States, Alaska, Hawaii, Puerto Rico and the U.S. Virgin Islands. Customers may utilize their dedicated facility to access Carrier's International service or other services. Such usage, however, shall not enter into calculations of charges hereunder. Performance WATS customers will receive a per billing cycle statement indicating the total usage per time-of-day and day-of-week rate period.

#### B. Usage Charges

Average Hours/	Day	Evening	Night/Weekend	
Usage Per Line	Cost/Minute	Cost/Minute	Cost/Minute	
0-19.99	\$0.1725	\$0.144	<b>5</b>	\$0.0905
20-39.99	\$0.1715	\$0.141	0	\$0.0905
40-59.99	\$0.1675	\$0.134	0	\$0.0905
60-79.99	\$0.1506	\$0.122	20	\$0.0905
80-99.99	\$0.1430	\$0.113	35	\$0.0905
100-119.99	\$0.1392	\$0.112	25	\$0.0905
120 +	\$0.1363	\$0.112	20	\$0.0905

## C. <u>Billing Increments</u>

Performance WATS is billed in minutes of use. All calls are billed in six second increments and are rounded up to the nearest six second increment. Performance WATS calls are subject to a minimum average time requirement ("MATR") of thirty (30) seconds per call per billing cycle (i.e. total per billing cycle calls must equal an average of at least thirty (30) seconds).

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 6.2 <u>Dedicated WATS Services</u> (Continued)
  - 6.2.10 Performance WATS (Continued)
    - D. Monthly maintenance fee per dedicated access line:

Per Month: \$50.00

E. Minimum usage charge per dedicated access line up to a maximum of four (4) lines:

<u>Per Month</u>: \$125.00

- F. Customers who have more than one access line in a service group will have their usage allocated in such a manner as to cause the greatest number of access lines to satisfy the minimum usage requirement.
- G. Non-recurring administrative charge per initial or subsequent order: \$150.00

#### 6.2.11 Precision WATS Service

#### A. <u>Description</u>

This service arrangement is a one-way outward service requiring the customer to originate calls via a dedicated facility or WATS access line between the Customer's premises and Carrier's terminal location and allows the completion of calls, anywhere in the state of Florida, the Continental United States, Alaska, Hawaii, Puerto Rico and the U.S. Virgin Islands and to access Carrier's international service or other services. Precision WATS customers will receive a per billing cycle statement indicating the total usage by Precision. Precision WATS is time-of-day and day-of-week sensitive.

### 6.2 <u>Dedicated WATS Services</u> (Continued)

### 6.2.11 Precision WATS Service

#### B. Usage Charges

The usage charges below incorporate the volume discounts as set forth in 6.2.14.D.

## Per Minute Rates

		9600-	24,000-	39,000	54,000	
Minutes of Use	<u>0-9599</u>	<u>23,999</u>	<u>38,999</u>	<u>53,999</u>	<u>77,999</u>	<u>78,000</u> +
Dayf	\$.1300	\$.1170	\$1170	\$.1144	\$.1118	\$.1092
Evening/Night/	\$.1300	\$.1170	\$1170	\$.1144	\$.1118	\$.1092
Weekend						

## C. <u>Billing Increments</u>

Precision WATS is billed in six second increments and are rounded up to the nearest six second increment. Precision WATS calls are subject to a minimum average time requirement ("MATR") of thirty (30) seconds per call per billing cycle (i.e. total per billing cycle calls must equal an average of at least thirty (30) seconds).

### D. Monthly Volume Discounts

Volume Discounts apply to Precision WATS usage only

<u>Minutes</u>	Volume Discount %
\$0 - \$9,599	0.0%
\$9,600 - \$23,999	10.0%
\$24,000 - \$38,999	10.0%
\$39,000 - \$53,999	12.0%
\$54,000 - \$77,999	14.0%
\$78,000 +	16.0%

E. Monthly maintenance fee per dedicated line:

Per Month: \$50.00

F. Minimum usage charge per dedicated access line up to a maximum of two lines:

Per Month/Per Line: \$125.00

Customers who have more than one access line in a service group will have their usage allocated in such a manner as to cause the greatest number of access/lines to satisfy the minimum usage requirement.

G. Non recurring charges per initial or subsequent order: \$150.00

### 6.3 <u>800 Services</u>

### 6.3.1 <u>Inbound 800</u>

#### A. Description

Dial-up, multipoint service, allowing parties wishing to contact the customer toll-free to access the carrier's switch for any city outside the local calling area. This option will allow customers to utilize the service by receiving calls from parties outside the local calling area in Florida at no charge to the parties placing the calls.

## B. <u>Per Minute Rates</u>:

<u>Day</u>	<u>Evening</u>	<u>Night</u>
\$.1649	\$.1349	\$.1049

C. Billing Increments - Calls are billed in six (6) second increments after initial minute.

D. Monthly service charge: None

E. <u>Monthly minimum usage</u>: \$10.00

F. <u>Cancellation</u>: \$50.00

## 6.3.2 <u>LaserPLUS 800</u>

#### A. Description

LaserPLUS 800 is an intrastate service that allows calls to be terminated on feature group facilities provided by the Local Exchange Carrier Network. This product is designed for small to medium users.

### 6.3 <u>800 Services</u> (Continued)

### 6.3.2 <u>LaserPLUS 800</u> (Continued)

B. Per Minute Rates:

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$0.1950 \$0.1800 \$0.1600

## C. Monthly Volume Discount Schedule

\$0 - \$40.99 0% \$50 - \$349.99 5% \$350 + 15%

#### D. Non-recurring Charges

1. Basic Number Assignment/Service Order (per 800 number) \$75.00

2. Administrative Change

\$20.00

Routing Number Change (per record change)

- 800 Number Change (per 800 number)

E. <u>Monthly Service Charge</u> \$10.00

#### 6.3.3 LaserEXPRESS 800

### A. <u>Description</u>

LaserEXPRESS 800 is an intrastate service that allows calls to be terminated on a dedicated facility provided by the Local Exchange Network. This product is designed for large users.

#### 6.3 800 Services (Continued)

#### 6.3.3 LaserEXPRESS 800 (Continued)

#### B. Per Minute Rates:

Call Termination	<u>Day</u>	Evening	Night/Weekend
All Company-Served	\$0.1350	\$0.1125	\$0.0925
FL locations			

#### C. Monthly Volume Discount

Total Monthly Usage **Discount** \$500 - \$9,999.99 10% \$10,000 plus 15%

#### D. Non-Recurring Charges 1/

1.	1st dedicated access line installed at one time	\$200.00
2.	Each add'l dedicated access line installed at same time	\$50.00
3.	Expedite Order (per order)	\$100.00
4.	Cancellation of Order (per order)	\$130.00
5.	Administrative (per order)	\$20.00
6.	T-1 Installation	\$450.00
7.	Routing Number Change (per SAL/T-1)	
	(a) Routing Assignment	\$40.00
	(b) Service Order	\$43.00

<sup>&</sup>lt;u>1</u>/ Dedicated access shall be provided by the Local Exchange Carrier (LEC) and the use of such access shall conform with the regulations and terms and conditions under which the Local Exchange Carrier provides such access.

### 6.3 <u>800 Services</u> (Continued)

## 6.3.3 <u>LaserEXPRESS 800</u> (Continued)

D. Non-Recurring Charges (Continued)

8.	800 Number Change (per 800 #)	
	(a) Number Assignment	\$30.00
	(b) Service Order	\$79.00
9.	Activation of COMMAND ROUTING (per 800 #)	\$100.00
	(This charge is incurred to activate the COMMAND	
	ROUTING arrangement and is incurred every	
	time the service is activated) 1/	
10.	Deactivation of COMMAND ROUTING (per 800 #)	\$100.00
	(This charge is incurred to for each request to	
	"deactivate" COMMAND ROUTING	
	arrangement), 2/ per request	
11.	Surcharge 3/ (for Business Line termination)	\$12.00
	(per VF line/hour or fraction thereof)	-

ISSUED: November 13, 2008

EFFECTIVE: November 14, 2008

Activation of COMMAND ROUTING will occur within one (1) hour after notification during Company business days, Monday - Friday 8am - 5pm, & within three (3) hours after notification at all other times.

<sup>&</sup>lt;u>2/</u> Deactivation will occur during the same time frames as Activation, however, the Company will stop the charges at the time notification of Deactivation is made.

The "Surcharge" will apply from the time the service is activated until the customer notifies the Company to deactivate the service. A minimum usage period of one (1) hour applies.

## 6.3 <u>800 Services</u> (Continued)

### 6.3.4 <u>Prestige 800</u>

#### A. Description

This product will enable small business customers to receive "toll-free" 800 service calls at their place of business. The customer will be assigned an 800 telephone number to receive calls that are paid for by the customer rather than the calling party.

B. Rate Schedule (Per Minute of Use)

Day All Other Times

\$0.220 \$0.195

- C. <u>Billing Increments</u> Usage is billed in one (1) minute increments.
- D. Monthly Service Fee: \$2.50 1/

## 6.3.5 Prestige II

#### A. Description

Prestige II is an inbound 800 service which enables customers to receive "toll-free" 800 service calls at their residence or place of business. Customers are charged a flat rate per minute.

B. Rate Schedule (Per Minute of Use)

<u>Peak</u> <u>Off Peak</u> \$0.2200 \$0.1950

ISSUED: November 13, 2008

<sup>1/</sup> The monthly service fee will be waived for each month in which usage exceeds \$10.00.

## 6.3 <u>800 Services</u> (Continued)

### 6.3.5 Prestige II (Continued)

C. Billing Increments - Calls are rated at one (1) minute increments.

D. Monthly Service Fee (per 800 #): \$5.00

#### E. Multi-Service Discount

Customers subscribing to Prestige II in conjunction with Prestige I will receive an additional 2% discount for each month in which combined total monthly usage exceeds \$150.00. This discount is applied to Peak usage only.

#### 6.3.6 Incoming 800

#### A. Description

Incoming 800 service is available to customers who wish to provide a toll-free line for their incoming callers. The caller dials an 800 number that connects the caller with the customer.

## B. Rate Per Minute

<u>Day</u>	<u>Evening</u>	Night/Weekend
\$.1980	\$.1620	\$.1380

C. <u>Billing Increments</u> - Calls are billed in six (6) second increments with a one minute minimum.

D. Monthly Service Charge: \$20.00 1/

E. <u>Installation Charge</u>: \$50.00

<sup>1/</sup> The service fee will be waived for each month in which account usage exceeds \$25.00.

293-430

430 +

\$.2600

\$.2700

\$.2535

\$.2633

## XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)

## 6.4.2 <u>Metromedia Travel Service</u>

## A. <u>Description</u>

This service arrangement is a one-way, dial-in-dial-out multipoint service that allows the customer to originate and terminate calls via the Carrier provided telephone lines from anywhere in the State of Florida. All travel calls are rounded to the next highest full minute. Service may be accessed through either a seven digit universal access telephone number or a 1-800 telephone number. 950 Accessed Calls requiring operator assistance are subject to a per call surcharge.

## B. <u>Usage Charges</u>

#### 1. 950 Access Type

Each Travel Service call accessed via 950 Access is charged at the rates set forth below. These rates incorporate the volume discounts as set forth in section 6.4.2.D.

DAY RAT	E SCHEDU	LE / PER MINU	TE RATES			
Mileage	\$0-\$99	\$100-\$249	\$250-\$999	\$1000-\$2499	\$2500-4999	\$5000 <b>+</b>
1-10	\$.1900	\$.1853	\$.1805	\$.1758	\$.1710	\$.1615
11-22	\$.2700	\$.2633	\$.2565	\$.2498	\$.2430	\$.2295
23-40	\$.3300	\$.3218	\$.3135	\$.3053	\$.2970	\$.2805
41-55	\$.3300	\$.3218	\$.3135	\$.3053	\$.2970	\$.2805
56-70	\$.3300	\$.3218	\$.3135	\$.3053	\$.2970	\$.2805
71-124	\$.3300	\$.3218	\$.3135	\$.3053	\$.2970	\$.2805
125-196	\$.3400	\$.3315	\$.3135	\$.3145	\$.3060	\$.2890
197-292	\$.3400	\$.3315	\$.3230	\$.3145	\$.3060	\$.2890
293-430	\$.3900	\$.3803	\$.3230	\$.3608	\$.3510	\$.3315
430 +	\$.3900	\$.3803	\$.3705	\$.3608	\$.3510	\$.3315
EVENING	RATE SCH	HEDULE / PER N	JINI ITE RATES	3		
Mileage	\$0-\$99	\$100-\$249	\$250-\$999	\$1000-\$2499	\$2500-4999	\$5000 +
1-10	\$.1400	\$.1365	\$.1330	\$.1295	\$.1260	\$.1190
11-22	\$.1600	\$.1560	\$.1520	\$.1480	\$.1440	\$.1360
23-40	\$.2100	\$.2048	\$.1995	\$.1943	\$.1890	\$.1785
41-55	\$.2200	\$.2145	\$.2090	\$.2035	\$.1980	\$.1870
56-70	\$.2300	\$.2243	\$.2185	\$.2128	\$.2070	\$.1955
71-124	\$.2300	\$.2243	\$.2185	\$.2128	\$.2070	\$.1955
125-196	\$.2500	\$.2438	\$.2375	\$.2313	\$.2250	\$.2125
197-292	\$.2500	\$.2438	\$.2375	\$.2313	\$.2250	\$.2125

\$.2470

\$.2565

\$.2405

\$.2498

\$.2350

\$.2430

\$.2210

\$.2295

### 6.4 Other Services (Continued)

## 6.4.2 <u>Metromedia Travel Service</u> (Continued)

B. <u>Usage Charges</u> (Continued)

1. <u>950 Access Type</u> (Continued)

### NIGHT/WEEKEND RATE SCHEDULE / PER MINUTE RATES

<u>Mileage</u>	\$0-\$99	\$100-\$249	\$250-\$999	\$1000-\$2499	\$2500-4999	\$5000 +
1-10	\$.1000	\$.0975	\$.0950	\$.0925	\$.0900	\$.0850
11-22	\$.1300	\$.1268	\$.1235	\$.1203	\$.1170	\$.1105
23-40	\$.1500	\$.1463	\$.1425	\$.1388	\$.1350	\$.1275
41-55	\$.1500	\$.1463	\$.1425	\$.1388	\$.1350	\$.1275
56-70	\$.1600	\$.1560	\$.1520	\$.1480	\$.1440	\$.1360
71-124	\$.1600	\$.1560	\$.1520	\$.1480	\$.1440	\$.1360
125-196	\$.1700	\$.1658	\$.1615	\$.1573	\$.1530	\$.1445
197-292	\$.1700	\$.1658	\$.1615	\$.1573	\$.1530	\$.1445
293-430	\$.2000	\$.1950	\$.1900	\$.1850	\$.1800	\$.1700
430 +	\$.2000	\$.1950	\$.1900	\$.1850	\$.1800	\$.1700

#### 2. 800 Access Type

Each Travel Service call accessed via 800 Service is charged at \$0.75 per call plus the per minute rate listed below.

	Per Minute	Each Additional Minute
Day	\$0.36	\$0.28
Evening	\$0.23	\$0.23
Night/Weekend	\$0.20	\$0.20

#### D. Monthly Volume Discounts

Travel calls shall be entitled to additional volume discounts applied to the total Travel dollar usage attained. The surcharge for 800 accessed calls is not subject to discounts.

TOTAL USAGE	<b>VOLUME DISCOUNT</b>
\$0 - \$99.99	0.0%
\$100.00 - \$249.99	2.5%
\$250.00 - \$999.99	5.0%
\$1,000.00 - \$2,499.99	7.5%
\$2,500.00 - \$4,999.99	10.0%
\$5,000.00 +	15.0%

#### 6.4.3 LataNET

### A. <u>Description</u>

LataNET is a one-way inter and intrastate tiered WATS service utilizing Special or T-1 access to connect the customer to the Company's national fiber optic network. This service has been designed for other common carriers, resellers, large corporations, hotel properties, state governments and other high-end telecommunications users.

### B. <u>Per Minute Rates</u>:

	<u>Day</u>	<u>Evening</u>	Night/ Weekend
All Florida Exchanges	\$0.1120	\$0.0850	\$0.0850

# 6.4 Other Services (Continued)

#### 6.4.3 <u>LataNET</u> (Continued)

- C. <u>Billing Increments</u> Calls are rated at an initial eighteen (18) seconds and in six (6) second increments thereafter.
- D. <u>Minimum Requirements</u> The customer is subject to a 90 day minimum requirement with a minimum usage requirement of at least \$4,000 per T-1 per month.

#### E. Non-Recurring Charges

1.	Installation, per T-1	\$450.00
	(This charge will be in addition	
	to the associated LEC charges)	
2.	Special Access Line, 1 <sup>st</sup> line	\$200.00
3.	Special Access Line, Each Add'l	\$50.00
4.	Optional Call Detail	\$20.00
5.	Validated Account Codes; set up	\$20.00
6.	Non-Validated Account Codes; set up	\$ 5.00

#### F. Monthly Recurring Charges

1.	Optional Call Detail	\$20.00
2.	Validated Account Codes	\$20.00
3.	Non-Validated Account Codes	\$ 5.00

#### 6.4.4 Lata EXPRESS II

# A. <u>Description</u>

Lata EXPRESS II is a one-way inter- and intrastate tiered WATS service utilizing T-1 access to connect the customer to the Company's national fiber optic network. This service has been designed for other common carriers, resellers, large corporations, hotel properties, state governments and other high-end telecommunications users.

# 6.4 Other Services (Continued)

#### 6.4.4 <u>Lata EXPRESS II</u> (Continued)

B. Rate Schedule

Rate Per Minute of Use

<u>Day</u> <u>Evening</u> <u>Night/Weekend</u> \$0.0975 \$0.0825 \$0.0825

- C. <u>Billing Increments</u> Calls are rated at an initial eighteen (18) seconds and in six (6) second increments thereafter.
- D. <u>Minimum Requirements</u> The customer is subject to a 90 day minimum requirement with a minimum usage requirement of at least \$10,000 per T-1 per month.

# E. <u>Non-Recurring Charges</u>

1.	Installation, per T-1	\$450.00
	(This charge will be in addition to	
	the associated LEC charges.)	
2.	Optional Call Detail	\$20.00
3.	Validated Account Codes; set-up	\$20.00
4.	Non-Validated Account Codes; set-up	\$ 5.00

#### F. Monthly Recurring Charges

1.	Optional Call Detail	\$20.00
2.	Validated Account Codes	\$20.00
3.	Non-Validated Account Codes	\$ 5.00

# 6.4 Other Services (Continued)

#### 6.4.5 TFN Travel Service

#### A. Description

This service allows the customer to call an 800 access number and twelve-digit authorization code to gain access to the company's network from any touch tone phone.

#### B. Per Minute Rates:

Day Evening Night/Weekend

\$.3610 \$.2660 \$.2280

1. Surcharge per call: \$.45

C. Billing Increments - Calls are billed in full minute increments.

## 6.4.7 WorldMark Service

#### A. General Description

WorldMark Service offers a unified service for single or multi-location customers using switched, dedicated, and WorldMark calling card origination and switched or dedicated 800 termination. The WorldMark package includes the availability of outbound, inbound (800) products with peak and off peak rates and a non time-of-day sensitive, calling card product, as well as, offering the most comprehensive performance guarantees available. Intrastate service is offered in conjunction with interstate service.

WorldMark Service offers cross-contributory product volume discounts on switched and dedicated, outbound and inbound(800) products based on combined usage of multiple WorldMark Service products, including domestic switched and dedicated, outbound and inbound (800) product usage, calling card and switched and dedicated WorldMark data product usage. Total intrastate, interstate and international usage contributes to discount level. The discount will apply to WorldMark domestic interstate rates only.

WorldMark Service offers an additional discount on switched and dedicated outbound and 800 products for customers willing to sign a term commitment.

There are four (4) Options of WorldMark available to customers, each of which are described below: 1/

Option A - Requires no monthly minimum billing commitment for month-to-month customers; \$15,000 monthly minimum billing commitment for term customers.

\_

<sup>1/</sup> If the minimum is not reached for Option A term and Options B through D, the customer will be charged for the difference.

# 6.4 Other Services (Continued)

#### 6.4.7 WorldMark Service (Continued)

A. <u>General Description</u> (Continued)

Option B - Requires a monthly minimum billing commitment of \$25,000.

Option D - Requires a monthly minimum billing commitment of \$50,000.

Requires a monthly minimum billing commitment of \$75,000.

Multiple services and/or multiple customer locations using WorldMark Service can contribute to individual product volume discounts and the overall monthly minimum commitment. WorldMark Operator Services and Directory Assistance usage is not included in determining product volume discounts and monthly minimum usage.

#### B. Service Arrangements

# 1. Term Commitment Option

WorldMark is available to customers through a term commitment option if the customer agrees to commit to such service for a term of twelve (12), twenty-four (24) or thirty-six (36) months. Customers who elect a term commitment will receive an additional discount off their selected option's domestic interstate outbound and 800 rates. (The Company's interstate rates are provided in its FCC Tariff No. 2 as filed with the Federal Communications Commission.) Customers who elect the term commitment are subject to the following:

- (a) Customers must indicate what WorldMark service or services are to be included in the term commitment. The discounts provided under this option become effective with the first full month's usage.
- (b) Upon execution of this agreement, the customer has ninety (90) days from the date service is made available in which to notify the Company in writing, either by certified or registered mail (return receipt requested), of the customer's desire to cancel this agreement without further obligation.
- (c) Customers who terminate service prior to the end of the term of commitment in any manner other than stated in (4) following, will be liable for a cancellation penalty equal to the monthly minimum billing commitment times the number of months remaining in the customer commitment period through the expiration of the first year. If applicable, 25% of the balance remaining (monthly minimum billing commitment times the number of remaining months in the contract beyond the first year) will also be included (per service type). If the termination becomes effective after the completion of the first year, then the charge shall be equal to 25% of the balance remaining (monthly minimum billing commitment times the number of months remaining in the contract). The foregoing cancellation penalty will apply unless the customer immediately converts to another Company service with equal or greater term and minimum usage requirement commitments, and Customer continues to use such Company service for a period of months equal to or greater than the number of months remaining in the term during which the conversion takes place.

#### 6.4 Other Services (Continued)

- 6.4.7 WorldMark Service (Continued)
  - B. <u>Service Arrangements</u> (Continued)
    - 1. Term Commitment Option (Continued)
      - (d) All customer requests to commence or terminate a WorldMark term commitment must be made in writing, either by certified mail (<u>return</u> <u>receipt requested</u>), to the Company and received no later than thirty (30) days prior to the then existing term expiration date. If such notification is not received by the Company within this timeframe, the WorldMark term agreement will be automatically renewed for a new term of commitment.
      - (e) A customer who cancels their agreement prior to the expiration will be required to repay any promotional credits that were given in addition to other termination charges as noted above.

#### C. WorldMark Service Guarantees

The following is a list of guarantees for each product offered under this service that will automatically compensate the customer when stated product performance levels are not met.

The customer's right to receive the WorldMark Guarantees as set forth will depend upon the Company's receipt of customer's full payment of each month's invoice. In the event the customer's account becomes past due, the Company reserves the right to void all rights to WorldMark's Service Satisfaction Guarantees and Service Restoration Guarantees as defined herein.

#### 1. WorldMark 1+ Switched/Dedicated Service

<u>Installation Interval Guarantee</u> - WorldMark customers of switched outbound service will be installed within five (5) full business days from the date the customer signed the service order. WorldMark customers of dedicated outbound service (limited to WorldCom Tier One cities) will be installed within eighteen (18) full business days from the date the customer signed the service order.

<u>Customer Compensation</u> - For each day the WorldMark switched or dedicated outbound installation interval is exceeded, a credit for each day's usage for the affected service will be issued. The average daily usage amount is calculated from the first complete billing period. In addition, customers will receive credit for their installation and first month's access line charges.

<u>Service Satisfaction Guarantee</u> - LDDS WorldCom will pay the local access and interexchange carrier service installation charges to return the customer to their previous carrier and circuit configuration if they are not satisfied after one hundred twenty (120) days with our service. Reimbursement will be limited to the levels of service in the previous configuration.

<u>Service Restoration Guarantee</u> - LDDS WorldCom guarantees that it will restore outbound service interruptions within thirty (30) minutes from the time the customer initiates the trouble call.

#### 6.4 Other Services (Continued)

- 6.4.7 WorldMark Service (Continued)
  - C. <u>WorldMark Service Guarantees</u> (Continued)
    - 1. <u>WorldMark 1+ Switched/Dedicated Service</u> (Continued)

<u>Customer Compensation</u> - If the restoration interval is exceeded, MCI WorldCom will automatically issue a credit equal to one (1) day's average usage charges. The average daily usage amount is calculated from the first complete billing period. Limited to one (1) credit per day, not to exceed thirty (30) credits in one (1) month.

#### 2. WorldMark Toll-Free Service

<u>Installation Interval Guarantee</u> - WorldMark customers of switched inbound service will be installed within five (5) full business days from the date the customer signed the service order. WorldMark customers of dedicated inbound service (limited to WorldCom Tier One cities) will be installed within eighteen (18) full business days from the date the customer signed the service order.

<u>Customer Compensation</u> - For each day the WorldMark switched or dedicated inbound installation interval is exceeded, a credit for each day's usage for the affected service will be issued. The average daily usage amount is calculated from the first complete billing period. Limited to one (1) credit per day, not to exceed thirty (30) credits in one (1) month. In addition, the monthly WorldMark toll-free service fee will be credited and dedicated customers will receive credit for their installation and first month's access line charges.

Additional Number Installation Guarantee - Additional toll-free numbers will be added to existing WorldMark switched or dedicated inbound service within one (1) full business day. This guarantee is not applicable if the additional number requires the installation of additional dedicated access lines or if the FCC-imposed restrictions on the availability of toll-free numbers necessitates a delay beyond LDDS WorldCom's control. In addition, LDDS WorldCom makes no warranty on our ability to overcome delays caused by competitors when porting existing numbers.

<u>Customer Compensation</u> - For each day the additional number installation interval is exceeded, a credit for each day's usage will be issued for the affected service. In addition, the monthly WorldMark toll-free service fee will be credited for the affected service.

<u>Feature Installation Interval Guarantee</u> - New WorldMark toll-free service feature configurations will be installed in one hundred twenty (120) minutes or less from the time a request is received by our WorldMark Customer Service Support Center. Toll-free feature change requests are limited to three (3) per hour, limited to ten (10) changes per feature.

<u>Feature Reconfiguration Guarantee</u> - Predetermined WorldMark toll-free service feature configurations will be implemented in sixty (60) minutes or less from the time a request is received by our WorldMark Customer Support Center. Toll-free feature change requests are limited to three (3) per hour, limited to ten (10) changes per feature.

# 6.4 Other Services (Continued)

- 6.4.7 WorldMark Service (Continued)
  - C. <u>WorldMark Service Guarantees</u> (Continued)

#### 2. <u>WorldMark Toll-Free Service</u> (Continued)

<u>Customer Compensation</u> - If the service or feature installation and/or reconfiguration interval is exceeded, the customer will automatically receive a credit equal to one day's average usage charges for WorldMark toll-free service and WorldMark toll-free monthly recurring service and feature charges for the affected service.

<u>Service Satisfaction Guarantee</u> - MCI WorldCom will pay the local access and interexchange carrier service installation charges to return the customer to their previous carrier and circuit configuration if they are not satisfied after one hundred twenty (120) days with our service. Reimbursement will be limited to the levels of service in the previous configuration.

<u>Service Restoration Guarantee</u> - MCI WorldCom guarantees that it will restore inbound service interruptions within thirty (30) minutes from the time the customer initiates the trouble call.

<u>Customer Compensation</u> - If the restoration interval is exceeded, MCI WorldCom will automatically issue a credit equal to one (1) day's average usage charges. The average daily usage amount is calculated from the first complete billing period. Limited to one (1) credit per day, not to exceed thirty (30) credits in one (1) month.

#### 3. WorldMark Calling Card Service

<u>Installation Interval Guarantee</u> - MCI WorldCom will deliver WorldMark calling cards within five (5) business days from the date the service order is signed. Upon signing a service order for WorldMark Calling Cards, a complimentary twenty (20) minute PhonePass prepaid card will be issued for each calling card ordered.

<u>Customer Compensation</u> - If the installation interval is exceeded, a \$15.00 credit will be issued for each distinct WorldMark Calling Card number.

<u>Service Restoration Guarantee</u> - MCI WorldCom guarantees that it will restore WorldMark Calling Card service interruptions within thirty (30) minutes or less, upon notification to the WorldMark Customer Support Center.

<u>Customer Compensation</u> - MCI WorldCom will automatically issue a \$15.00 credit if WorldMark Calling Card Service is not restored within thirty (30) minutes. Limit one (1) credit per account per day, not to exceed thirty (30) credits in one (1) month.

<u>Card Security Guarantee</u> - If the customer's WorldMark Calling Card is deactivated due to toll fraud, they may call our WorldMark Customer Support Center and receive a free twenty (20) minute PhonePass prepaid card number to use until we issue another calling card.

#### 6.4 Other Services (Continued)

- 6.4.7 WorldMark Service (Continued)
  - C. <u>WorldMark Service Guarantees</u> (Continued)
    - 4. WorldMark Service Liability

With thirty (30) days prior written notice, the Company reserves the right to discontinue providing WorldMark Service to a customer if the Company issues five (5) or more restoration credits in a thirty (30) calendar period.

- (a) In the event the Company cancels WorldMark Service in accordance with above, the Company will reimburse the customer the charges directly related to the switch back (the "Qualifying Charges") based on the customer's prior network configuration, not to exceed the amount paid for installation of WorldMark Service.
- (b) Any WorldMark Switched Back Reimbursement due hereunder as described above, shall be due and payable to customer on or before sixty (60) calendar days from the time the Company notifies the customer of the cancellation of WorldMark Services.
- (c) Payment of the WorldMark Switch back Reimbursement may at the customer's option, be made by either (i) a credit against the customer's then-current charges for services provided by the Company, if any; or, (ii) a check drawn on a United States bank and payable in United States dollars. Said payment may be withheld if customer is in past due status with the Company.

In addition to the liability provisions specifically provided in this section, all other Rules and Regulations for liability as filed in Section 2 will also apply to WorldMark Services.

#### D. Usage Charges

#### 1. Rate Periods

<u>Peak - All calls that occur between 7 A.M. through 6:59 P.M., Monday through Friday, except on Company-recognized holidays.</u>

Off Peak - All calls that occur between 7 P.M. through 6:59 A.M., Monday through Friday, and all calls between 7 P.M. Friday and 6:59 A.M. Monday and Company-recognized holidays.

# 2. <u>Billing Increments</u>

Domestic switched and dedicated access, outbound and inbound (800) product calls are billed in eighteen (18) second initial increments and additional increments of six (6) seconds. WorldMark Calling Card is billed at an initial sixty (60) second increment and rounded to the next higher six (6) second increment thereafter. All fractional per call charges will be rounded to the nearest whole cent.

# 3. <u>WorldMark Rate Schedules</u>

Per minute intrastate base rates for WorldMark Service are the same for all Options (A-D) and apply as shown below:

#### (a) Switched Per Minute Rates

Peak/Off Peak

Switched Outbound \$0.1250 Switched 800 \$0.1300

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 6.4 Other Services (Continued)
  - 6.4.7 WorldMark Service (Continued)
    - D. Usage Charges (Continued)
      - 3. WorldMark Rate Schedules (Continued)
        - (b) **Dedicated Per Minute Rates**

Peak/Off Peak **Dedicated Outbound** \$0.0900 Dedicated 800 \$0.0950

- WorldMark Calling Card Service 4.
  - Direct Dial Rate Schedule (a)
    - (1) Per Minute rates applying to all time periods:

Per Minute rate: \$0.2500

- (2) Billing Increments - Domestic calls are billed in sixty (60) second initial increments and additional increments of six (6) seconds.
- (3) Per Call Surcharge

Per Call Surcharge: \$0.2500 (i)

The following rates and surcharge will apply to calls that (ii) default to a live operator.

> Per Minute rate: \$0.2500 Per Call Surcharge: \$0.5000

- XIII. GRANDFATHERED SERVICES: SUBSECTION B (CONT.)
- 6.4 Other Services (Continued)
  - 6.4.7 WorldMark Service (Continued)
    - D. <u>Usage Charges</u> (Continued)
      - 4. <u>WorldMark Calling Card Service</u> (Continued)
        - (b) Operator Assisted Rate Schedule
          - (1) If a customer chooses to access an operator to place a call, the call will be billed at the per minute rate below:

Per Minute Rate: \$0.5000

- (2) <u>Billing Increments</u> All WorldMark Calling Card calls placed with the assistance of an operator will be billed in one (1) minute increments.
- (3) <u>Per Call Surcharge</u> The following surcharge will apply in addition to the per minute rates above.

Per Call Charge
\$1.50
\$3.00
\$0.50

# E. WorldMark Service and Feature Charges

WorldMark service is an interstate service offering with customers having the option of using the service for intrastate calls. Accordingly, a description of the chargeable services and features, including monthly recurring, non-recurring, change and per minute surcharges associated with WorldMark Service can be found in Section 7.11.6 of the Company's Tariff FCC No. 2.

#### 6.4 Other Services (Continued)

#### 6.4.8 Agency WATS/Multi-Location Plus

Agency WATS is a business line service designed for agencies or franchises for use by individual agents or agent/owners. Agency WATS can be sold as a multi location plan and is referred to in such instances as Multi-Location Plus. Said agents will receive reduced rates in lieu of commissions paid to the parent organization. The agents will also receive the Corporate Travel Plan rates under this service plan.

There will be a six month ramp up period for this service to reach a minimum of \$4,000 in aggregated billing. If this minimum is not reached within the first six months of service, all accounts will be converted to the Company's Standard PLUS WATS with associated rates and charges.

# A. Per Minute Rates

Business Day	<u>Evening</u>	Night/Weekend
\$0.1766	\$0.1455	\$0.1455

#### B. <u>Non-recurring Charges</u>

Validated Project Account Codes; set up \$20.00

# C. Recurring Charges

1.	Monthly Fee	\$ 0.00
2.	Project Account Codes	\$ 0.00
3.	Validated Project Account Codes; Mthly	\$10.00

#### D. Billing Increments

Timing for all calls will include a 30 second minimum duration and additional time is billed in six (6) second increments thereafter.

#### 6.5 Intermedia Services

#### 6.5.1 Intermedia USA III Switched Service

Intermedia USA III Switched Service is an On-Net or Off-Net, inbound and outbound switched service offered to business customers at a single flat rate. Customers may also elect to subscribe to inbound or outbound service only. This service permits the end user to route 8XX traffic based on the time of day, and to accept or block calls based on the originating NPA or NPA/NXX. A volume discount is offered to customers whose monthly long distance billing (including Intermedia's intrastate/interstate/international and calling card charges) is \$750 or above. Calls are billed in six second increments with a six second minimum.

Intrastate service is only provided in conjunction with the Company's interstate offering. A monthly minimum of \$100 applies, and customers whose monthly usage is less than the minimum will be billed the minimum charge. Intermedia USA III Switched Service requires a one (1) year term commitment. Any customer who requests over 40 8XX numbers will be charged an additional \$10 per month per 8XX number over 40. The charges above are cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service.

The following enhanced service features are available with Intermedia USA III Switched Service. Interstate non-recurring and recurring charges may apply.

- 1. Message Referral
- 2. Call Area Selection
- 3. Geographic Routing
- 4. Time of Day Routing
- 5. Day of Week Routing
- 6. Day of Year (Holiday) Routing
- 7. Percent Allocation

Intermedia USA III Switched Service

	<u>Inbound</u>	Outbound
Rate per Minute, Base:	\$0.1050	\$0.1050
Rate Per Minute, Discounted:	\$0.0945	\$0.0945

Discount rate is applied to base rate when account's aggregate billing reaches \$750.

# 6.5 <u>Intermedia Services (Cont.)</u>

#### 6.5.2 Intermedia USA III Dedicated Service

Intermedia USA III Dedicated Service is an Off-Net, inbound and outbound dedicated service offered to business customers at a single flat rate. Customers may also elect to subscribe to inbound or outbound service only. A volume discount is offered to customers whose monthly long distance billing (including Intermedia's intrastate/interstate/international and calling card charges) is \$7,500 or above. This service permits the end user to route 8XX traffic based on the time of day, and to accept or block calls based on the originating NPA or NPA/NXX. The service also enables the end user who utilizes multiple dedicated 8XX numbers on the same trunk to identify the specific 8XX number dialed by the caller. Each 8XX number and its associated calls are identified separately on customer invoices. Calls are billed in six second increments with a six second minimum.

Intrastate service is only provided in conjunction with the Company's interstate offering. A monthly minimum billing of \$2,500 applies, and customers whose monthly usage is less than the minimum will be billed the minimum charge. Intermedia USA III Dedicated Service requires a one (1) year term commitment. Any customer who requests over 40 8XX numbers will be charged an additional \$10 per month per 8XX number over 40. The charges above are cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service.

The following enhanced service features are available with Intermedia USA III Dedicated Service. Interstate non-recurring and recurring charges may apply.

- Message Referral
- 2. Call Area Selection
- 3. Geographic Routing
- 4. Time of Day Routing
- 5. Day of Week Routing
- 6. Day of Year (Holiday) Routing
- 7. Percent Allocation
- 8. Dialed Number Identification Service (DNIS)
- Real-time ANI

#### 6.5 <u>Intermedia Services (Cont.)</u>

#### 6.5.3 Intermedia USA III Dedicated Service

# Intermedia USA III Dedicated Service

	Inbound	Outbound
Rate per Minute, Base:	\$0.1050	\$0.0950
Rate Per Minute, Discounted:	\$0.0955	\$0.0855

Discount rate is applied to base rate when account's aggregate billing reaches \$7,500.

#### 6.5.4 Single T Dedicated Long Distance Service

Single T Dedicated Long Distance service is an On-Net, inbound and outbound dedicated service offered to business customers who subscribe to certain Intermedia local exchange service offerings which utilize a high capacity dedicated connection between the Company's switching location and the Customer's premises. Service is also provided to branch locations, but via Intermedia Switched Long Distance service. Volume and term discounts apply. Volume discounts are based upon total primary and branch location per minute long distance usage charges (including Intermedia intrastate, interstate, and international services) calculated at the base rate. The achieved discount rate applies to all Single T per minute usage charges. Calls are billed in six second increments with a six second minimum.

Intrastate service is only provided in conjunction with the Company's interstate offering. A minimum monthly billing of \$500 applies, and customers whose monthly usage is less than the minimum will be billed the minimum charge. An initial term contract of one year is required. Any customer who requests over 40 8XX numbers will be charged an additional \$10 per month per 8XX number over 40. The charges above are cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service.

The following enhanced service features are available with Single T Dedicated Long Distance service. Interstate non-recurring and recurring charges may apply.

- 1. Geographic Routing
- 2. Time of Day Routing
- 3. Day of Week Routing
- 4. Day of Year (Holiday) Routing
- 5. Percent Allocation
- 6. Custom Blocking

<sup>\*</sup>Services and rates listed on this page are available only to existing subscribers as of November 29, 2000.

#### 6.5 <u>Intermedia Services (Cont.)</u>

# 6.5.4 <u>Single T Dedicated Long Distance Service - Grandfathered\* (Cont.)</u>

Base rate	<u>Inbound</u> \$ 0.1000	Net Effective Rate/Min. Ou	tbound 0900	Net Effective Rate/Min.
<u>Dase rate</u>	ψ 0.1000	Ψ0.	0300	
<u>Volume</u>	Discount		Discour	<u>nt</u>
\$0-\$2,499.99	0.0%	\$ 0.1000	0.0%	\$ 0.0900
\$2,500-\$7,499.99	5.0%	\$ 0.0950	5.0%	\$ 0.0855
\$7,500-\$9,999.99	10.0%	\$ 0.0900	10.0%	\$ 0.0810
\$10,000-\$14,999.99	15.0%	\$ 0.0850	15.0%	\$ 0.0765
\$15,000-\$19,999.99	20.0%	\$ 0.0800	20.0%	\$ 0.0720
\$20,000+	25.0%	\$ 0.0750	25.0%	\$ 0.0675

<sup>\*</sup>Services and rates listed on this page are available only to existing subscribers as of November 29, 2000.

### 6.5.5 Long Distance Only T-1 - Grandfathered\*

The Company's Long Distance Only T-1 service is an On-Net, inbound and outbound dedicated service offered to business customers. Service is also provided to branch locations, but via switched access. Volume discounts apply and are based upon total primary and branch location per minute long distance usage charges (including Intermedia intrastate, interstate, and international services) calculated at the base rate. The achieved discount rate applies to all Long Distance Only T-1 per minute usage charges. Calls are billed in six second increments with a six second minimum.

Intrastate service is only provided in conjunction with the Company's interstate offering. An initial term contract of one year is required, and the minimum monthly billing volume below will apply for the remainder of the term if canceled by the Customer prior to expiration. A monthly minimum of \$2,500 applies, and customers whose monthly usage is less than the minimum will be billed the minimum amount. Any customer who requests over 40 8XX numbers will be charged an additional \$10 per month per 8XX number over 40. The charges above are cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service.

The following enhanced service features are available with Long Distance Only T-1 service. Interstate non-recurring and recurring charges may apply.

- Geographic Routing
- 2. Time of Day Routing
- 3. Day of Week Routing
- 4. Day of Year (Holiday) Routing
- 5. Percent Allocation
- 6. Custom Blocking

<sup>\*</sup>Services and rates listed on this page are available only to existing subscribers as of November 29, 2000.

# 6.5 <u>Intermedia Services (Cont.)</u>

# 6.5.5 Long Distance Only T-1 - Grandfathered\* (Cont.)

Base rate	<u>Inbound</u> \$ 0.1000	Net Effective Rate/Min.	Outbound \$0.0900	Net Effective Rate/Min.
Volume Disco	<u>ount</u>		Discount	
\$0-\$2,499.99	0.0%	\$ 0.1000	0.0%	\$ 0.0900
\$2,500-\$7,499.99	5.0%	\$ 0.0950	5.0%	\$ 0.0855
\$7,500-\$9,999.99	10.0%	\$ 0.0900	10.0%	\$ 0.0810
\$10,000-\$14,999.99	15.0%	\$ 0.0850	15.0%	\$ 0.0765
\$15,000-\$19,999.99	20.0%	\$ 0.0800	20.0%	\$ 0.0720
\$20,000+	25.0%	\$ 0.0750	25.0%	\$ 0.0675

<sup>\*</sup>Services and rates listed on this page are available only to existing subscribers as of November 29, 2000.

#### 6.5.6 Unified Long Distance Services

Unified Long Distance Service is an outbound service offered to business customers. Unified Long Distance Service utilizes switched access. Intrastate service is provided in conjunction with the Company's interstate offering. Pricing for Unified Long Distance Service is based upon the monthly (or annual, if selected) revenue (including outbound, inbound, calling card, operator service, and directory assistance in all jurisdictions) and term commitment selected by the customer. Customers whose monthly (or annual, if selected) billing is less than the commitment amount will be billed the actual usage charges, plus the difference between actual usage charges and the committed amount. This minimum charge is cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service. Usage eligible to retire the customer commitment is from all of the following Unified Voice Services at any customer location:

Unified Long Distance Service (Switched Outbound Service)

Unified Toll-Free Service (Switched Inbound Service)

Unified Direct Long Distance Service (Dedicated Outbound Service)

Unified Direct Toll-Free Service (Dedicated Inbound service)

Unified Extended Direct Long Distance Service (Dedicated Outbound Service)

Unified Extended Direct Toll-Free Service (Dedicated Inbound Service)

Unified Single T Long Distance Service (Single T Outbound)

Unified Single T Toll-Free Service (Single T Inbound)

Unified Calling Card

Unified Directory Assistance

**Unified Operator Services** 

Unified Direct Global service (Dedicated International)

Unified Extended Direct Global Services (Dedicated International)

Unified Global Services (Switched International)

Unified Extended Global services (Switched International)

#### 6.5 <u>Intermedia Services (Cont.)</u>

# 6.5.7 <u>Unified Long Distance Services</u>

Month-to-month, one year, two year, and three year term plans are available to subscribers of Unified Long Distance Service. The customer's liability for early termination of a term plan is the monthly revenue commitment (annual divided by 12, if selected) times the months remaining in the term plan. Calls are billed in six-second increments with a six-second per call minimum.

Rate per Minute, Rate 1 (\$0-\$999.99 revenue commitment): \$0.0945 Rate Per Minute, Rate 2 (\$1,000+ revenue commitment): \$0.0851

# 6.5.8 Unified Toll-Free Service

Unified Toll Free Service is an inbound (i.e., 8XX) calling service offered to business customers. Unified Toll- Free Service utilizes switched access. Intrastate service is provided in conjunction with the Company's interstate offering. Pricing for Unified Toll-Free Service is based upon the monthly (or annual, if selected) long distance billing (including outbound, inbound, calling card, operator service, and directory assistance in all jurisdictions) and term commitment selected by the customer. Customers whose monthly (or annual if selected) billing is less than the commitment amount will be billed the actual usage charges, plus the difference between actual usage charges and the committed amount. This minimum charge is cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service. Usage eligible to retire the customer commitment is from all of the following Unified Voice Services at any customer location:

Unified Long Distance Service (Switched Outbound Service)

Unified Toll-Free Service (Switched Inbound Service)

Unified Direct Long Distance Service (Dedicated Outbound Service)

Unified Direct Toll-Free Service (Dedicated Inbound service)

Unified Extended Direct Toll-Free Service (Dedicated Inbound Service)

Unified Single T Long Distance Service (Single T Outbound)

Unified Single T Toll-Free Service (Single T Inbound)

Unified Calling Card

Unified Directory Assistance

**Unified Operator Services** 

Unified Direct Global service (Dedicated International)

Unified Extended Direct Global Services (Dedicated International)

Unified Global Services (Switched International)

Unified Extended Global services (Switched International)

#### 6.5 <u>Intermedia Services (Cont.)</u>

# 6.5.8 Unified Toll-Free Service (Cont.)

Month-to-month, one year, two year, and three year term plans are available to customers of Unified Toll Free Service. The customer's liability for early termination of a term plan is the monthly revenue commitment (annual divided by 12, if selected) times the months remaining in the term plan. Calls are billed in six-second increments with a six-second per call minimum. Any customer who requests over 40 8XX numbers will be charged \$10 per month (cross-jurisdictional) per 8XX number over 40.

Rate per Minute, Rate 1 (\$0-\$999.99 revenue commitment): \$0.0945 Rate Per Minute, Rate 2 (\$1,000+ revenue commitment): \$0.0851

#### 6.5.9 Unified Direct Long Distance Service

Unified Direct Long Distance Service is an outbound service offered to business customers. Unified Direct Long Distance Service utilizes dedicated access. The customer is responsible for establishing the appropriate dedicated access connection(s) between the customer premises and the Company's switching location and all associated charges. Intrastate service is provided in conjunction with the Company's interstate offering. Pricing for Unified Direct Long Distance Service is based upon the monthly (or annual if selected) long distance billing (including outbound, inbound, calling card, operator service, and directory assistance in all jurisdictions) and term commitment selected by the customer. The minimum monthly revenue commitment for this plan is \$2,000. Customers whose monthly (or annual if selected) billing is less than the commitment amount will be billed the actual usage charges, plus the difference between actual usage charges and the committed amount. This minimum charge is cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service. Usage eligible to retire the customer commitment is from all of the following Unified Voice Services at any customer location:

Unified Long Distance Service (Switched Outbound Service)

Unified Toll-Free Service (Switched Inbound Service)

Unified Direct Long Distance Service (Dedicated Outbound Service)

Unified Direct Toll-Free Service (Dedicated Inbound service)

Unified Extended Direct Long Distance Service (Dedicated Outbound Service)

Unified Extended Direct Toll-Free Service (Dedicated Inbound Service)

Unified Single T Long Distance Service (Single T Outbound)

Unified Single T Toll-Free Service (Single T Inbound)

Unified Calling Card

**Unified Directory Assistance** 

**Unified Operator Services** 

Unified Direct Global service (Dedicated International)

Unified Extended Direct Global Services (Dedicated International)

Unified Global Services (Switched International)

Unified Extended Global services (Switched International)

Month-to-month, one year, two year, and three year term plans are available to subscribers of Unified Direct Long Distance Service. The customer's liability for early termination of a term plan is the monthly revenue commitment (annual divided by 12, if selected) times the months remaining in the term plan. Calls are billed in six-second increments with a six-second per call minimum.

Rate per Minute, Rate 1 (\$2,000-\$9,999.99 revenue commitment): \$0.0700 Rate Per Minute, Rate 2 (\$10,000+ revenue commitment): \$0.0550

#### 6.5 Intermedia Services (Cont.)

#### 6.5.10 Unified Direct Toll Free Service

Unified Direct Toll-Free Service is an inbound toll-free (i.e., 8XX) calling service offered to business customers. Unified Direct Toll-Free Service utilizes dedicated access. The customer is responsible for establishing the appropriate dedicated access connection(s) between the customer premises and the Company's switching location and all associated charges. Intrastate service is provided in conjunction with the Company's interstate offering. Pricing for Unified Direct Toll-Free Service is based upon the monthly (or annual if selected) long distance billing (including outbound, inbound, calling card, operator service, and directory assistance in all jurisdictions) and term commitment selected by the customer. The minimum monthly revenue commitment for this plan is \$2,000. Customers whose monthly (or annual if selected) billing is less than the commitment amount will be billed the actual usage charges, plus the difference between actual usage charges and the committed amount. This minimum charge is crossjurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service. Usage eligible to retire the customer commitment is from all of the following Unified Voice Services at any customer location:

Unified Long Distance Service (Switched Outbound Service)

Unified Toll-Free Service (Switched Inbound Service)

Unified Direct Long Distance Service (Dedicated Outbound Service)

Unified Direct Toll-Free Service (Dedicated Inbound service)

Unified Extended Direct Long Distance Service (Dedicated Outbound Service)

Unified Extended Direct Toll-Free Service (Dedicated Inbound Service)

Unified Single T Long Distance Service (Single T Outbound)

Unified Single T Toll-Free Service (Single T Inbound)

Unified Calling Card

Unified Directory Assistance

**Unified Operator Services** 

Unified Direct Global service (Dedicated International)

Unified Extended Direct Global Services (Dedicated International)

Unified Global Services (Switched International)

Unified Extended Global services (Switched International)

Month-to-month, one year, two year, and three year term plans are available to subscribers of Unified Direct Toll-Free Service. The customer's liability for early termination of a term plan is the monthly revenue commitment (annual divided by 12. if selected) times the months remaining in the term plan. Calls are billed in six-second increments with a six-second per call minimum. Any customer who requests over 40 8XX numbers will be charged \$10 per month (cross-jurisdictional) per 8XX number over 40.

The following enhanced service features are available with Unified Direct Toll-Free Service. Nonrecurring and recurring charges may apply.

- 1. Dialed Number Identification service (DNIS)
- 2. Real-time ANI

Rate per Minute, Rate 1 (\$2,000-\$9,999.99 revenue commitment): \$0.0700 Rate Per Minute, Rate 2 (\$10,000+ revenue commitment): \$0.0550

# 6.5 <u>Intermedia Services (Cont.)</u>

# 6.5.11 <u>Unified Extended Direct Long Distance Service</u>

Unified Extended Direct Long Distance Service is an outbound service offered to business customers. Unified Extended Direct Long Distance Service utilizes dedicated access. The customer is responsible for establishing the appropriate dedicated access connection(s) between the customer premises and the Company's switching location and all associated charges. Intrastate service is provided in conjunction with the Company's interstate offering. Pricing for Unified Extended Direct Long Distance Service is based upon the monthly (or annual if selected) long distance billing (including outbound, inbound, calling card, operator service, and directory assistance in all jurisdictions) and term commitment selected by the customer. The minimum monthly revenue commitment for this plan is \$2,000. Customers whose monthly (or annual if selected) billing is less than the commitment amount will be billed the actual usage charges, plus the difference between actual usage charges and the committed amount. This minimum charge is cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service. Usage eligible to retire the customer commitment is from all of the following Unified Voice Services at any customer location:

Unified Long Distance Service (Switched Outbound Service)

Unified Toll-Free Service (Switched Inbound Service)

Unified Direct Long Distance Service (Dedicated Outbound Service)

Unified Direct Toll-Free Service (Dedicated Inbound service)

Unified Extended Direct Long Distance Service (Dedicated Outbound Service)

Unified Extended Direct Toll-Free Service (Dedicated Inbound Service)

Unified Single T Long Distance Service (Single T Outbound)

Unified Single T Toll-Free Service (Single T Inbound)

Unified Calling Card

Unified Directory Assistance

**Unified Operator Services** 

Unified Direct Global service (Dedicated International)

Unified Extended Direct Global Services (Dedicated International)

Unified Global Services (Switched International)

Unified Extended Global services (Switched International)

Month-to-month, one year, two year, and three year term plans are available to subscribers of Unified Extended Direct Long Distance Service. The customer's liability for early termination of a term plan is the monthly revenue commitment (annual divided by 12, if selected) times the months remaining in the term plan. Calls are billed in six-second increments with a six-second per call minimum.

Rate per Minute, Rate 1 (\$2,000-\$9,999.99 revenue commitment): \$0.0950 Rate Per Minute, Rate 2 (\$10,000+ revenue commitment): \$0.0855

# 6.5 <u>Intermedia Services (Cont.)</u>

# 6.5.12 <u>Unified Extended Direct Toll Free Service</u>

Unified Extended Direct Toll-Free Service is an inbound toll-free (i.e., 8XX) calling service offered to business customers. Unified Extended Direct Toll-Free Service utilizes dedicated access. The customer is responsible for establishing the appropriate dedicated access connection(s) between the customer premises and the Company's switching location and all associated charges. Intrastate service is provided in conjunction with the Company's interstate offering. Pricing for Unified Extended Direct Toll-Free Service is based upon the monthly (or annual, if selected) long distance billing (including outbound, inbound, calling card, operator service, and directory assistance in all jurisdictions) and term commitment selected by the customer. The minimum monthly revenue commitment for this plan is \$2,000. Customers whose monthly (or annual if selected) billing is less than the commitment amount will be billed the actual usage charges, plus the difference between actual usage charges and the committed amount. This minimum charge is cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service. Usage eligible to retire the customer commitment is from all of the following Unified Voice Services at any customer location:

Unified Long Distance Service (Switched Outbound Service); Unified Toll-Free Service (Switched Inbound Service); Unified Direct Long Distance Service (Dedicated Outbound Service)
Unified Direct Toll-Free Service (Dedicated Inbound service); Unified Extended Direct Long Distance Service (Dedicated Outbound Service); Unified Extended Direct Toll-Free Service (Dedicated Inbound Service); Unified Single T Long Distance Service (Single T Outbound)
Unified Single T Toll-Free Service (Single T Inbound); Unified Calling Card
Unified Directory Assistance; Unified Operator Services; Unified Direct Global service (Dedicated International)
Unified Global Services (Switched International); Unified Extended Global services (Switched International)

Month-to-month, one year, two year, and three year term plans are available to subscribers of Unified Extended Direct Toll-Free Service. The customer's liability for early termination of a term plan is the monthly revenue commitment (annual divided by 12, if selected) times the months remaining in the term plan. Calls are billed in six-second increments with a six-second per call minimum. Any customer who requests over 40 8XX numbers will be charged \$10 per month (cross-jurisdictional) per 8XX number over 40. The following enhanced service features are available with Unified Extended Direct Toll-Free Service. Non-recurring and recurring charges may apply.

- Message Referral
- 2. Call Area Selection
- 3. Geographic Routing
- 4. Time of Day Routing
- 5. Day of Week Routing
- 6. Day of Year
- 7. Dialed Number Identification service (DNIS)
- 8. Real-Time ANI

Rate per Minute, Rate 1 (\$2,000-\$9,999.99 revenue commitment): \$0.0950 Rate Per Minute, Rate 2 (\$10,000+ revenue commitment): \$0.0855

# 6.5 <u>Intermedia Services (Cont.)</u>

# 6.5.12 Unified Single T Long Distance Service

Unified Single T Long Distance Service is an outbound service offered to business customers who subscribe to certain Intermedia local exchange service offerings which utilize a high capacity dedicated connection between the Company's switching location and the customer's premises. Unified Single T Long Distance Service utilizes the dedicated connection in concert with the customer's local exchange service. The customer is responsible for establishing the appropriate dedicated access connections between the customer premises and the Company's switching location and all associated charges. Intrastate service is provided in conjunction with the Company's interstate offering. Pricing for Unified Single T Long Distance Service is based upon the annual, revenue (including outbound, inbound, calling card, operator service, and directory assistance in all jurisdictions) and term commitment selected by the customer. The minimum monthly commitment for this plan is \$250. Customers whose monthly (or annual if selected) billing is less than the commitment amount will be billed the actual usage charges, plus the difference between actual usage charges and the committed amount. This minimum charge is cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service. Usage eligible to retire the customer commitment is from all of the following Unified Voice Services at any customer location:

Unified Long Distance Service (Switched Outbound Service)

Unified Toll-Free Service (Switched Inbound Service)

Unified Direct Long Distance Service (Dedicated Outbound Service)

Unified Direct Toll-Free Service (Dedicated Inbound service)

Unified Extended Direct Long Distance Service (Dedicated Outbound Service)

Unified Extended Direct Toll-Free Service (Dedicated Inbound Service)

Unified Single T Long Distance Service (Single T Outbound)

Unified Single T Toll-Free Service (Single T Inbound)

**Unified Calling Card** 

Unified Directory Assistance

**Unified Operator Services** 

Unified Direct Global service (Dedicated International)

Unified Extended Direct Global Services (Dedicated International)

Unified Global Services (Switched International)

Unified Extended Global services (Switched International)

One year, two year, and three year term plans are available to subscribers of Unified Single T Long Distance Service. The customer's liability for early termination of a term plan is the monthly revenue commitment (annual divided by 12, if selected) times the months remaining in the term plan. Calls are billed in six-second increments with a six-second per call minimum.

Rate per Minute, Rate 1 (\$250-\$4,999.99 revenue commitment): \$0.0700 Rate Per Minute, Rate 2 (\$5,000+ revenue commitment): \$0.0550

# 6.5 <u>Intermedia Services (Cont.)</u>

#### 6.5.13 Unified Single T Toll Free Service

Unified Single T Toll-Free Service is an inbound toll-free (i.e., 8XX) calling service offered to business customers who subscribe to certain Intermedia local exchange service offerings which utilize a high capacity dedicated connection between the Company's switching location and the customer's premises. Unified Single T Toll-Free Service utilizes the dedicated connection in concert with the customer's local exchange service. The customer is responsible for establishing the appropriate dedicated access connections between the customer premises and the Company's switching location and all associated charges. Intrastate service is provided in conjunction with the Company's interstate offering. Pricing for Unified Single T Toll-Free Service is based upon the monthly (or annual, if selected) revenue (including outbound, inbound, calling card, operator service, and directory assistance in all jurisdictions) and term commitment selected by the customer. The minimum monthly commitment for this plan is \$250. Customers whose monthly (or annual if selected) billing is less than the commitment amount will be billed the actual usage charges, plus the difference between actual usage charges and the committed amount. This minimum charge is cross-jurisdictional, i.e., charges will not be applied in each jurisdiction, but rather one charge will apply in connection with all relevant jurisdictions of the service. Usage eligible to retire the customer commitment is from all of the following Unified Voice Services at any customer location:

Unified Long Distance Service (Switched Outbound Service)

Unified Toll-Free Service (Switched Inbound Service)

Unified Direct Long Distance Service (Dedicated Outbound Service)

Unified Direct Toll-Free Service (Dedicated Inbound service)

Unified Extended Direct Long Distance Service (Dedicated Outbound Service)

Unified Extended Direct Toll-Free Service (Dedicated Inbound Service)

Unified Single T Long Distance Service (Single T Outbound)

Unified Single T Toll-Free Service (Single T Inbound)

**Unified Calling Card** 

Unified Directory Assistance

**Unified Operator Services** 

Unified Direct Global service (Dedicated International)

Unified Extended Direct Global Services (Dedicated International)

Unified Global Services (Switched International)

Unified Extended Global services (Switched International)

Month-to-month, one year, two year, and three year term plans are available to subscribers of Unified Single T Toll-Free Service. The customer's liability for early termination of a term plan is the monthly revenue commitment (annual divided by 12, if selected) times the months remaining in the term plan. Calls are billed in six-second increments with a six-second per call minimum. Any customer who requests over 40 8XX numbers will be charged \$10 per month (cross-jurisdictional) per 8XX number over 40.

The following enhanced service features are available with Unified Single T Toll-Free Service. Non-recurring and recurring charges may apply.

- Dialed Number Identification service (DNIS)
- 2. Real-Time ANI

Rate per Minute, Rate 1 (\$250-\$4,999.99 revenue commitment): \$0.0700 Rate Per Minute, Rate 2 (\$5,000+ revenue commitment): \$0.0550

# 6.5 <u>Intermedia Services (Cont.)</u>

#### 6.5.14 Unified Directory Assistance

Unified Directory Assistance is available to Intermedia Customers. A Directory Assistance charge applies to each call to the Directory Assistance Bureau. Up to two requests may be made on each call to Directory Assistance. The Directory Assistance charge applies to each call regardless of whether the Directory Assistance Bureau is able to furnish the requested telephone number.

#### 6.5.15 Payphone Compensation Charge

In order to recover the Company's expenses to comply with the FCC's pay telephone compensation plan effective on October 7, 1997 (FCC 97-371), an undiscountable per call charge is applicable to all interstate, intrastate, and international calls that originate from any domestic pay telephone used to access Intermedia services. This surcharge, which is in addition to standard tariffed usage charges and any applicable service charges and surcharges associated with Intermedia service, applies for the use of the instrument used to access Intermedia service and is unrelated to the Intermedia service accessed from the pay telephone.

Pay telephones include coin-operated and coinless phones owned by local telephone companies, independent companies and other interexchange carriers. The Payphone Compensation Charge applies to the initial completed call and any reoriginated call (i.e., using the "#" symbol).

Whenever possible, the Payphone Compensation Charge will appear on the same invoice containing the usage charges for the surcharged call. In cases where proper pay telephone coding digits are not transmitted to the Company prior to completion of a call, the Payphone Compensation Charge may be billed on a subsequent invoice after the Company has obtained information from a carrier that the originating station is an eligible pay telephone.

The Payphone Compensation Charge does not apply to calls placed from pay telephones at which the Customer pays for service by inserting coins during the progress of the call.

# 6.5 <u>Intermedia Services (Cont.)</u>

#### 6.5.16 Audio Conferencing

Audio conferencing provides full-service audio conferencing functionality, with audio bridges, plus ancillary support services.

#### **Product Description**

Intermedia will offer four different audio conferencing services with features. The features and functions of the Conference Calling service are listed below:

**Corporate Call A:** Participants call into a standard phone number or Toll Free number at the scheduled time of the conference. The conference administrator will greet each participant, ask their name and the conference they are scheduled to attend. The chairperson is billed on a per line per minute rate.

**Corporate Call B:** A Company chairperson provides the names and numbers of all participants at the time the call is reserved. The conference administrator calls the participants and places them in the conference together. The chairperson is billed back on a per line per minute rate.

**Call Express:** Participants call into a standard phone number or Toll Free number at the scheduled time of the conference. The participants are connected to an automated greeting. Callers will be given voice prompts and then automatically dropped into the conference call. The chairperson is billed on a per line per minute rate.

**On Demand:** Participants call into one standard Toll Free number. Each subscriber is issued a chairperson PIN and a participant pin. No interaction with a conference operator is required to schedule the conference calls. The chairperson is billed on a per line per minute rate.

**Event:** Participants call into a standard phone number or Toll Free number at the scheduled time of the conference. The conference administrator will greet them and escort them into the conference. This conference call used for very large conference calls and/or special events.

Various enhanced services, in addition to those listed herein, may be requested at the time of the conference set-up:

**Smart Polling (Q & A):** An efficient method for conducting question and answer sessions. "Listen only" participants in a conference call may register their questions simply by pressing 14 on a touch-tone phone. The participants are announced into the conference in order that the requests are received.

**Conference Recording:** Playback your recorded conference call via a toll free number. Digitally records the teleconference, in its entirety or any portion you choose. Following the call, individuals may access the recording by dialing a designated number and entering a password. The system allows the listener to fast-forward, pause and receive on-line help.

**Broadcast Fax:** Broadcast fax service, when you need to provide hard-copy information to any number of participants simultaneously, before or after a conference.

#### 6.5 <u>Intermedia Services (Cont.)</u>

#### 6.5.17 Unified Enhanced Calling Card

Intermedia's Enhanced Calling Card allows customers to make intrastate long distance calls when they are away from home or the office, and the calls will be billed to their card number. Intrastate service is provided in conjunction with the Company's interstate offering.

Calls are billed in six (6) second increments after a thirty (30) second per call minimum. All partial increments are rounded up to the next whole increment.

#### 6.5.17.1 Calling Card Standard Service Components

The features and functions of the Calling Card are listed below:

**Voice Prompts:** Whenever a caller accesses Intermedia's Card Service they will be guided by voice prompts. For the experienced user, the system will permit DTMF cut-though, which means they can dial over the voice prompts and not have to wait for the prompts to be completed.

**14 Digit Authorization Code**: All authorization codes will be 14 digits, consisting of a 10 digit card number plus a 4 digit PIN.

**Re-Dial:** This feature will allow the caller to return to the previous voice prompt if they should misdial during any part of the calling process, by holding down the \* key, the cardholder will initiate this feature.

**Credit Limit:** A credit limit is established when the card is issued, preventing any additional calls to be made once the credit limit threshold has been reached. This is a real-time monitoring feature. When the cardholder reaches their credit limit threshold, however, the current call is not terminated. The next time they use their card a message will be played stating that the credit limit has been exceeded. It will prevent additional calls and request that the caller contact Intermedia Customer Service to rectify the problem.

**# Re-Originate:** The cardholder has the ability to make sequential calls without having to hangup. By pressing the # key for 2 seconds, the caller will be prompted to enter the next number they wish to dial.

**No Domestic Calling Surcharge:** Intermedia does not charge a surcharge for any domestic or international terminations. There will be a payphone surcharge, however.

**Speech Recognition:** Speech recognition offers customers the ability to simply speak information into the telephone rather than enter the DTMF digits.

**Speed Dial:** Supports storage of up to 9 frequently dialed numbers

**Voice Message Delivery:** Messages can be recorded and delivered immediately or some date in the future

Accounting Codes: Supports unverified accounting codes to track usage and allocate costs

Conference Calling: Provides the ability to have up to a 6 person conference call

**Directory Assistance:** Supports the ability to connect the card user with an operator to obtain telephone number

# 6.5 <u>Intermedia Services (Cont.)</u>

# 6.5.18 Unified Toll Free Plus PIN

Unified Toll-Free plus PIN (800+PIN) service is a switched toll-free member of the Unified Long Distance family. Calls to subscribers are made by dialing the toll-free access number and then entering the subscriber's 6-digit Personal Identification Number (PIN). PINs are generated by the Company and are not customer selectable. This service is not available on a stand-alone basis, but may be used in connection with any existing Intermedia long distance service(s) as an additional service for an existing customer. Calls are billed in whole minute increments. Partial increments are rounded up to the next full increment. Intrastate service is provided only in conjunction with the Company's interstate offering.

There is no specific minimum or specific commitment for this product. There is, however, the requirement of a \$250 minimum monthly total account commitment with a minimum term of 1 year. An existing account with less than a \$250 commitment level that adds this service must increase the total account commitment to \$250. Accounts that are not Unified Long Distance must be converted to a volume/term account with a minimum commitment of \$250 per month and minimum term of 1 year. The customer's liability for early termination of a term plan is the monthly revenue commitment times the months remaining in the term plan. Usage revenue generated by Unified Toll-Free plus PIN is applicable to the subscriber's total account commitment.

# 6.5.19 Low Billing Surcharge

In addition to all other nonrecurring, monthly recurring, or usage charges identified in Company tariffs, a monthly Low Billing Surcharge applies to stand-alone business accounts with total monthly long distance usage billing of less than \$100, and to residential accounts with total monthly long distance usage billing of less than \$25. This charge is cross-jurisdictional, i.e., the charge will not be applied in each jurisdiction of service within an account, but rather one charge will apply in connection with all relevant jurisdictions.

# 6.5.20 Unified Calling Card

The Unified Calling Card is a service offered to Customers for intrastate and interstate calling. Customers must input a valid Personal Identification Number (PIN) when originating Calling Card calls. Calls are billed in one (1) minute increments after the initial minimum period of one (1) minute. There are no nonrecurring or monthly recurring charges. No calling card surcharge applies for use of this service, although a Payphone Compensation Charge will apply for use of the telephone instrument used to access the service.

\*Services and rates listed on this page are available only to existing subscribers as of November 29, 2000.

# 6.5 <u>Intermedia Services (Cont.)</u>

#### 6.5.21 Standard Features

Standard Features referred to in the preceding sections of this tariff refer to the following, all of which are provided at no additional charge:

<u>Speed Dialing Codes</u> - A three digit code dialed after the Authorization Code which permits the customer to reach a commonly called long distance number without dialing the long distance number.

Additional Authorization Codes - Five (5) digit code used, usually, in areas that are still not equipped for equal access, to enable ICI to identify the use of the service on the customer's account.

Generic Project Codes - A two (2), three (3), or four (4) digit code assigned by the customer and dialed after the long distance number. Project codes are not verified except as provided under "Verified Project Codes." Project codes are printed on the customer bill as part of the call detail record.

<u>Project Code Report</u> - A report which consists of the complete call detail for the current month, sorted and subtotaled by project code.

<u>Verified Project Codes</u> - A group of 2, 3, or 4 digit project codes within which a predesignated quantity of codes are valid. Assignment of valid codes is random. Calls placed with the remaining invalid codes will not be completed. Validation is accomplished by ICI's network switch.

<u>AT&T Call Cost Report</u> - A report prepared at the time of monthly billing which lists all calls and equivalent AT&T cost, according to current AT&T tariffs on file, and approved by the appropriate regulatory body.

<u>Potential Call Abuse Report</u> - A report prepared at the time of monthly billing which can list the 20 longest calls made during a billing cycle, or all calls placed during the Evening/Weekend time periods during a billing cycle. A combination of both reports can also be provided on request by a customer.

<u>Automatic Dialer Equipment</u> - A device installed by ICI at the customer's premise which when activated, performs most or all of the "pre destination number." Automated Dialers are generally not provided where Equal Access is available.

#### 6.5.22 Unified Operator Services

Unified Operator Service is provided to customers who presubscribe to this service for intrastate and interstate calling. Intrastate rates are specified in this tariff. Various billing arrangements are available with Intermedia's operator-assisted service including Calling Card, Commercial Credit Card, Collect, Person-to-Person and Third Party. Service is provided from presubscribed, dedicated or shared use access lines. Calls are billed in one minute increments. Operator surcharges apply as appropriate.

#### 6.6 Special Services (Cont.)

6.6.2 Type 1 Service

#### A. Usage Charges

Usage charges are billed in 6 second increments and are subject to a minimum average time requirement, (MATR) of one (1) minute per completed call. This is not on each call but is applied to all messages in the service group. If the average per call usage is less than one minute, then one minute per message will be applied to the service group.

1. Rate Schedule Per Minute of Use All Florida Exchanges

\$.0599

# 2. Rules and Requirements

The General Rates, Conditions, Specifications and Special Requirements of ITB No. DGS 90/91-047 shall be considered as part of this Type 1 Service and only for the State of Florida. The Company shall not change the rates in 4.25.1.A.1 unless ordered to do so by the FPSC or to the extent that intrastate terminating access charges are increased by local exchange carriers, If intrastate terminating access charges are increased, the Company will pass through such cost increases directly by revising the minute of use charges listed in 4.25.1.A.1 (above).

#### Call Detail

Call Detail shall be provided only via magnetic tape.

(a) Monthly Charge (per tape) \$75.00(b) Non-recurring charge \$200.00